## JYU DISSERTATIONS 410

Keijo Mönkkönen

Integral Geometry and Unique Continuation Principles



JYU DISSERTATIONS 410

Keijo Mönkkönen

# Integral Geometry and Unique Continuation Principles

Esitetään Jyväskylän yliopiston matemaattis-luonnontieteellisen tiedekunnan suostumuksella julkisesti tarkastettavaksi elokuun 13. päivänä 2021 kello 12.

Academic dissertation to be publicly discussed, by permission of the Faculty of Mathematics and Science of the University of Jyväskylä, on August 13, 2021 at 12 o'clock noon.



JYVÄSKYLÄ 2021

Editors Joonas Ilmavirta Department of Mathematics and Statistic, University of Jyväskylä Timo Hautala Open Science Centre, University of Jyväskylä

Copyright © 2021, by University of Jyväskylä

ISBN 978-951-39-8780-0 (PDF) URN:ISBN:978-951-39-8780-0 ISSN 2489-9003

Permanent link to the online version of this publication: http://urn.fi/URN:ISBN:978-951-39-8780-0

#### Foreword

I wish to thank my supervisor Joonas Ilmavirta for his support and guidance during my years as a PhD student of mathematics. He has helped me to become a little bit more mathematician after my short career in theoretical physics, but not to lose the physicist's way of thinking and calculating without thinking. I also want to thank Mikko Salo for offering me this rare but great opportunity to study inverse problems in his internationally recognized inverse problems group. I want to express my gratitude to the Department of Mathematics and Statistics of University of Jyväskylä for providing me a fruitful working environment in 2017–2021.

I thank Angkana Rüland for agreeing to be my opponent at the public examination of my dissertation. I also want to thank the pre-examiners Jürgen Frikel and Venky Krishnan for their valuable feedback. I wish to thank my colleagues Jesse Railo and Giovanni Covi for many inspiring moments in mathematics and outside mathematics. Sharing the office room with Jesse has been very beneficial for my research and career, and Giovanni's work on fractional inverse problems convinced me to become interested in that area of mathematics. I also thank Gunther Uhlmann for collaboration in a very interesting article on higher order fractional Calderón problems. I want to thank all the members in Mikko Salo's inverse problems group, former and present, for the friendly, supporting and encouraging spirit that has always existed in our group.

Finally, I want to thank my family and friends for their support in my long and undirected journey in the academic world.

Jyväskylä, July 1, 2021 Department of Mathematics and Statistics University of Jyväskylä Keijo Mönkkönen

#### LIST OF INCLUDED ARTICLES

This dissertation consists of an introductory part and the following seven articles:

- [A] J. Ilmavirta and K. Mönkkönen. Unique continuation of the normal operator of the x-ray transform and applications in geophysics. *Inverse Problems*, 36(4):045014, 2020.
- [B] G. Covi, K. Mönkkönen and J. Railo. Unique continuation property and Poincaré inequality for higher order fractional Laplacians with applications in inverse problems. *Inverse Probl. Imaging*, 15(4):641–681, 2021.
- [C] J. Ilmavirta and K. Mönkkönen. X-ray Tomography of One-forms with Partial Data. SIAM J. Math. Anal., 53(3):3002–3015, 2021.
- [D] G. Covi, K. Mönkkönen, J. Railo and G. Uhlmann. The higher order fractional Calderón problem for linear local operators: uniqueness. Preprint (2020), arXiv:2008.10227.
- [E] J. Ilmavirta, K. Mönkkönen and J. Railo. On tensor decompositions and algebraic structure of the mixed and transverse ray transforms. Preprint (2020), arXiv:2009.01043.
- [F] K. Mönkkönen. Boundary rigidity for Randers metrics. Ann. Fenn. Math., 2020. To appear.
- [G] J. Ilmavirta and K. Mönkkönen. Partial data problems and unique continuation in scalar and vector field tomography. Preprint (2021), arXiv:2103.14385.

The author has participated actively in the research of the joint articles [A, B, C, D, E, G].

#### Abstract

In this thesis we study inverse problems in integral geometry and non-local partial differential equations. We will study these rather different areas of mathematical inverse problems by using the theory of non-local fractional operators. This thesis mainly focuses on proving different kind of unique continuation results of fractional operators which are then used to prove uniqueness results for fractional Calderón problems and partial data problems in scalar and vector field tomography.

The introductory part of the thesis contains a general introduction and review of inverse problems arising in medical and seismic imaging. The included articles are divided into three classes which are then presented in their own sections and studied in different levels of detail.

In the articles [A, B, C, G] we consider partial data problems in the X-ray tomography of scalar and vector fields. In the first article [A] we prove unique continuation for certain Riesz potentials and apply it to partial data problems of scalar fields. In the second article [B] we prove unique continuation results for higher order fractional Laplacians which are then used in proving uniqueness for partial data problems of vector fields and we prove unique continuation of the normal operator of vector fields which implies uniqueness for the partial data problems. In the seventh article [G] we generalize the unique continuation result of fractional Laplacians proved in [B] and use it to prove uniqueness for partial data problems of scalar and vector fields, extending the partial data results of the articles [A, B, C] to more general cases.

In the articles [B, D] we consider higher order fractional Calderón problems. In the second article [B] we use the unique continuation of higher order fractional Laplacians to prove uniqueness for the Calderón problem of the higher order fractional (magnetic) Schrödinger equation. In the fourth article [D] we generalize the uniqueness result proved in [B] to include general lower order local perturbations of the fractional Laplacian.

In the articles [E, F] we consider the travel time tomography problem and its different linearized versions. In the fifth article [E] we study mixing ray transforms which are generalizations of the geodesic ray transform. We prove solenoidal injectivity results for them in various different cases. In the sixth article [F] we study the boundary rigidity problem on certain non-reversible Finsler manifolds which are also called Randers manifolds. We prove that if the Randers metric consists of a boundary rigid Riemannian metric and a closed 1-form, then the boundary distances determine the Randers metric uniquely up to a natural gauge.

#### TIIVISTELMÄ

Tässä väitöskirjassa tutkitaan integraaligeometrian ja epälokaalien osittaisdifferentiaaliyhtälöiden inversio-ongelmia. Näitä melko erilaisia matemaattisia inversio-ongelmia tutkitaan käyttämällä apuna epälokaalien fraktionaalisten operaattoreiden teoriaa. Väitöskirja keskittyy pääosin todistamaan fraktionaalisten operaattoreiden erilaisia yksikäsitteisen jatkon tuloksia, joita käytetään todistaessa yksikäsitteisyyttä fraktionaalisille Calderónin ongelmille sekä skalaari- ja vektorikenttien tomografian osittaisen datan ongelmille.

Väitöskirjan johdantokappale sisältää yleisen tason johdatuksen sekä kirjallisuuskatsauksen lääketieteellisessä ja seismisessä kuvantamisessa esiintyviin inversio-ongelmiin. Väitöskirjaan sisällytetyt artikkelit on jaettu kolmeen luokkaan, jotka esitellään omissa kappaleissaan ja joita tarkastellaan yksityiskohtien osalta monella eri tasolla.

Artikkelit [A, B, C, G] käsittelevät osittaisen datan ongelmia skalaari- ja vektorikenttien röntgentomografiassa. Ensimmäisessä artikkelissa [A] todistetaan yksikäsitteinen jatko tietyille Rieszin potentiaaleille ja sitä sovelletaan skalaarikenttien osittaisen datan ongelmiin. Toisessa artikkelissa [B] todistetaan yksikäsitteisen jatkon tuloksia korkeamman kertaluvun fraktionaalisille Laplace-operaattoreille ja niitä käytetään *d*-tasomuunnosten osittaisen datan ongelmien yksikäsitteisyyden todistamisessa. Kolmannessa artikkelissa [C] tutkitaan vektorikenttien osittaisen datan ongelmia ja todistetaan vektorikenttien normaalioperaattorin yksikäsitteinen jatko, josta seuraa yksikäsitteisyys osittaisen datan ongelmille. Seitsemännessä artikkelissa [G] yleistetään artikkelissa [B] todistettu fraktionaalisen Laplace-operaattorin yksikäsitteisen jatkon tulos ja sitä käytetään skalaari- ja vektorikenttien osittaisen datan ongelmien yksikäsitteisyyden todistamisessa, laajentaen artikkeleiden [A, B, C] osittaisen datan tuloksia yleisempiin tapauksiin.

Artikkelit [B, D] käsittelevät korkeamman kertaluvun fraktionaalisia Calderónin ongelmia. Toisessa artikkelissa [B] käytetään korkeamman kertaluvun fraktionaalisten Laplace-operaattoreiden yksikäsitteistä jatkoa todistaessa yksikäsitteisyyttä korkeamman kertaluvun fraktionaalisen (magneettisen) Schrödingerin yhtälön Calderónin ongelmalle. Neljännessä artikkelissa [D] yleistetään artikkelin [B] yksikäsitteisyystulos fraktionaalisen Laplaceoperaattorin yleisille alempiasteisille lokaaleille perturbaatioille.

Artikkelit [E, F] käsittelevät matka-aikatomografiaa ja sen linearisoituja versioita. Viidennessä artikkelissa [E] tutkitaan sekoitussädemuunnoksia, jotka ovat geodeettisen sädemuunnoksen yleistyksiä. Niille todistetaan solenoidisia injektiivisyystuloksia monissa eri tilanteissa. Kuudennessa artikkelissa [F] tutkitaan reunajäykkyysongelmaa tietyillä ei-reversiibeleillä Finslermonistoilla, joita kutsutaan myös Randers-monistoiksi. Artikkelissa todistetaan, että jos Randers-metriikka koostuu reunajäykästä Riemannin metriikasta ja suljetusta 1-muodosta, niin Randers-metriikka määräytyy reunaetäisyyksistään luonnollista mittaa vaille yksikäsitteisesti.

### Contents

Foreword	i
List of included articles	ii
Abstract	iii
Tiivistelmä	iv
1. Introduction	2
1.1. On the articles of this thesis	3
1.2. Inverse problems and forward problems	4
1.3. X-ray tomography of scalar and vector fields	5
1.3.1. X-ray tomography of scalar fields	5
1.3.2. X-ray tomography of vector fields	7
1.4. Electrical impedance tomography and its non-local versions	8
1.4.1. The Calderón problem	8
1.4.2. The fractional Calderón problem	9
1.5. Travel time tomography and its linearization	10
1.5.1. The boundary rigidity problem	10
1.5.2. Linearized versions of the boundary rigidity problem	12
2. X-ray tomography with partial data: [A, B, C, G]	14
2.1. Notation	16
2.2. Main results	18
3. Higher order fractional Calderón problems: [B, D]	20
3.1. Notation	23
3.2. Main results	24
4. Travel time tomography on Riemannian and Finsler manifolds:	
$[\mathrm{E,F}]$	27
4.1. Notation	30
4.2. Main results	32
References	35
Included articles	43

1

#### 1. INTRODUCTION

This thesis is about mathematical inverse problems and the main focus is in proving uniqueness results for different problems arising in tomography. As the title of the thesis suggests, some of the inverse problems appear in integral geometry. However, there are also included inverse problems which do not strictly fit under this category, but they are related to problems in integral geometry via unique continuation principles of non-local operators.

The inverse problems studied in this thesis can be roughly divided into three classes:

- (I1) The travel time tomography problem and its linearized versions
- (I2) Partial data problems in X-ray tomography
- (I3) Fractional Calderón problems.

The classes (I1) and (I2) belong to integral geometry. In fact, problems in (I2) are linearized travel time tomography problems in Euclidean background with partial data. Hence (I2) can be seen as a subset of (I1). The class (I3) belongs to non-local partial differential equations and at first sight has nothing to do with the classes (I1) and (I2). But there is a way to get from (I3) to (I1), namely using the "intermediate step" (I2).

A unifying theme between fractional Calderón problems (I3) and partial data problems in X-ray tomography (I2) is the use of unique continuation properties of non-local operators in proving uniqueness results. The central operator of this thesis is the fractional Laplace operator  $(-\Delta)^s$ ,  $s \in (-n/2, \infty) \setminus \mathbb{Z}$ , and many of the main theorems of this thesis are unique continuation results of  $(-\Delta)^s$  or corollaries of them. The unique continuation of  $(-\Delta)^s$  is used to prove Runge approximation and hence uniqueness for fractional Calderón problems. As a special case of fractional Laplacians we have the normal operators of different X-ray transforms whose unique continuation properties are then used to prove uniqueness for various partial data problems arising in the X-ray tomography of scalar and vector fields.

This introductory part is organized in the following way. We first discuss in section 1.1 how the different articles of this thesis are related to each other. Then we give a gentle introduction to inverse problems and forward problems in section 1.2, and in sections 1.3-1.5 we review the main three classes of inverse problems (I1)–(I3) which are studied in this thesis. In sections 2–4 we go through the main theorems of the included articles. In the beginning of each section we first introduce the inverse problem and give the main results in a general level. We then go through the needed notation in sections 2.1–4.1 before giving the main theorems with all technical details in sections 2.2–4.2. Section 4 can be read independently of sections 2 and 3. Section 3 can also be read independently of section 2 if one first goes through the notation in section 2.1. 1.1. On the articles of this thesis. In figure 1 we have illustrated the connection between the different articles of this thesis. In most of the articles we study inverse problems with partial data: these include fractional Calderón problems (articles [B, D]), X-ray tomography with partial data (articles [A, B, C, G]) and linearized travel time tomography with "half-local" data (article [A]). Unique continuation of fractional Laplacians has a crucial role in proving uniqueness for partial data problems studied in this thesis. Fractional Laplacians arise in fractional Calderón problems and also in X-ray tomography in the form of different normal operators. Problems in X-ray tomography in turn can be seen as linearized travel time tomography problems in Euclidean background.



FIGURE 1. A graph illustrating the relation between the different articles of this thesis.

In fractional Calderón problems the task is to recover the potential (and more generally a perturbation) of the fractional Schrödinger equation in a bounded domain by doing measurements in the exterior of the domain. These problems are studied in the articles [B, D] and treated in section 3. In X-ray tomography we want to determine a scalar field (or a vector field) when we know its integrals over lines which intersect a given nonempty open set. This is studied in the articles [A, B, C, G] and treated in section 2. In travel time tomography one wants to recover the speed of sound (and more generally Riemannian metric or Finsler norm) by measuring travel times (geodesic distances) on the boundary of a compact manifold. This problem and its linearized versions (the geodesic ray transform and its generalizations) are studied in the articles [A, E, F] and treated in sections 2 and 4.

A remark from the point of view of graph theory: the (connected) graph presented in figure 1 has a Hamilton cycle, i.e. a closed walk such that every vertex is visited exactly once. The graph also has an Euler trail, i.e. a walk such that every edge is traversed exactly once. Formally, this "proves" that the articles of this thesis are closely related to each other. However, the graph does not admit an Euler tour (a closed walk which is an Euler trail) since not every vertex has even degree [14].

1.2. Inverse problems and forward problems. Inverse problems are practical or abstract problems which arise for example in medical and seismic imaging [62, 70, 78, 108, 112, 113, 114, 129, 157]. Inverse problems are often encountered when making indirect measurements. In such situations we have an object we cannot or do not want to access by invasive methods. In medical imaging the object can be a patient we want to study without doing surgical operations, and in seismic imaging the object can be the planet Earth whose deep interior we cannot reach by any practical means. The common task in both cases is that one wants to deduce the interior features of some object by making measurements on the boundary or in the exterior of the object. Usually we have some physical model which tells us how the interior properties of the object affect the measurements we make on the boundary or in the exterior. The goal is to use this physical model to deduce the interior properties of the object from the boundary or exterior measurements. The boundary and exterior measurements are often called just data.

Inverse problems are opposite to what we call direct problems or forward problems. Let us consider an example from X-ray tomography to illustrate the difference. In X-ray tomography one shoots X-rays through an object and studies the attenuation pattern of the X-rays. The attenuation of the X-rays is determined by the interior properties (the position-dependent attenuation coefficient) of the object. In the direct problem one knows the attenuation of the object and wants to determine the attenuation pattern of the X-rays. When the initial intensity of the X-rays is known, then one can easily calculate the final intensity of the X-rays by using a simple physical model [112]. Roughly saying, the direct problem corresponds to putting values for parameters in an equation and computing the result.

Inverse problems are much harder since they "operate" in the opposite direction. For example, in medical imaging one wants to determine the attenuation of the object instead of the attenuation pattern of the X-rays which can be easily measured. Since one also can control the initial intensity of the X-rays we have indirect information about the attenuation, i.e. we know the total attenuation of the X-rays and want to determine the attenuation of the object from that data. It turns out that the total attenuation corresponds to the integrals of the attenuation function along lines which intersect the object [112]. The inverse problem is to invert this integral transform which is also called the X-ray transform. The inversion of the X-ray transform is a much harder task than solving the forward problem where we already know the interior features of the object (the attenuation) and just have to calculate the end result (the final intensity of the X-rays).

Uniqueness, stability and reconstruction are important properties in the study of inverse problems. Uniqueness means that the inverse problem has a unique solution. In other words, if two objects produce the same boundary or exterior data, then they must have the same interior features. Reconstruction means that there is some way (e.g. an algorithm or formula) so that one can compute the desired physical quantity related to the interior properties of the object from the boundary or exterior data. Stability is related to how much measurement errors affect uniqueness or reconstruction. Since in practice there is always some noise in measurements, stability is important in showing that the reconstructed quantity is not too far away from the true value of that quantity. These three properties are not independent of each other since uniqueness usually follows from reconstruction and stability.

Uniqueness and stability have a connection to Hadamard's formulation of a well-posed problem [59, 60]. A mathematical problem related to a physical phenomenon is called well-posed, if the problem has unique solution which is stable with respect to the measured data (the solution depends continuously on the data) [62, 78, 108, 112, 129]. If the solution fails to exist, the solution is not unique or the solution does not depend continuously on the data, the problem is said to be ill-posed. Forward problems are often well-posed, but inverse problems tend to be ill-posed. Usually the reason for ill-posedness of inverse problems is that they lack stability which causes difficulties in numerical reconstruction [78, 80, 108, 114, 129].

In this thesis we mainly focus on uniqueness, i.e. in most of our theorems we show that the inverse problem has unique solution. Even if we do not get stability or a reconstruction formula for the problem, uniqueness is important in practical applications. Uniqueness for example increases the reliability of the results obtained in X-ray tomography when we only have a "small amount" of measurement data available.

#### 1.3. X-ray tomography of scalar and vector fields.

1.3.1. X-ray tomography of scalar fields. X-ray tomography is a commonly used method in medical imaging to study interiors of objects. The main goal is to determine the attenuation of the object when one knows the initial and final intensities of X-rays, i.e. the total attenuation of the X-rays. The attenuation can be modelled as a scalar function  $f: \mathbb{R}^n \to \mathbb{R}$ . If the rays propagate parallel to x-axis, then the intensity I of the X-rays satisfies the differential equation I'(x) = -f(x)I(x) and the total attenuation corresponds to the line integral [112]

(1) 
$$\ln\left(\frac{I_0}{I_1}\right) = \int_{\gamma} f \mathrm{d}s$$

where  $I_0$  is the initial intensity and  $I_1$  is the final intensity of the X-rays, and  $\gamma$  is a line along which the X-ray beam propagates. The inverse problem in X-ray tomography is to solve f in equation (1) using different lines  $\gamma$  when the left-hand side of the equation is known.

The previous discussion motivates us to define the operator  $X_0$  as

(2) 
$$X_0 f(\gamma) = \int_{\gamma} f \mathrm{d}s$$

where  $\gamma$  is a line in  $\mathbb{R}^n$  and  $f: \mathbb{R}^n \to \mathbb{R}$  is a scalar field. The operator  $X_0$  is called the X-ray transform of scalar fields and in two dimensions  $X_0$  is also known as the Radon transform. The inverse problem is to invert the operator  $X_0$  in equation (2) and it was first studied by Johann Radon [126]. Theoretical and practical applications to computerized tomography were studied by Cormack and Hounsfield [27, 28, 112]. There are formulas for the inversion of  $X_0$  some of which involve the normal operator  $N_0$  of the X-ray transform [64, 112, 129, 149]. The normal operator  $N_0 = X_0^* X_0$  is defined as first applying the X-ray transform and then back-projecting  $X_0 f$  from the space of all lines to a function in  $\mathbb{R}^n$  using the adjoint  $X_0^*$  of the X-ray transform. Hence  $N_0$  is a useful auxiliary operator which maps functions on  $\mathbb{R}^n$  to functions on  $\mathbb{R}^n$  and one can study the X-ray transform  $X_0$  using its normal operator  $N_0$ .

The inversion formulas for  $X_0$  assume that we know the integrals of fover all lines in  $\mathbb{R}^n$ . In practical applications we only have access to a small subset of lines, and in that case we have a partial data problem. One such partial data problem is to uniquely determine f everywhere in  $\mathbb{R}^n$  from its X-ray data on all lines intersecting a given open set  $V \subset \mathbb{R}^n$ . The integrals alone cannot determine f uniquely and one has to make additional assumptions [79, 112]. The partial data problem has unique solution, if  $f|_V$  vanishes [29, 79],  $f|_V$  is piecewise constant or piecewise polynomial [79, 162] or if  $f|_V$  is real analytic [77]. A complementary partial data result is the Helgason support theorem where one has access to lines which do not intersect a given compact and convex set and the problem is to determine the scalar field uniquely outside that set [64]. Partial data problems are in general much harder to treat than problems with full data because the reconstruction is not stable anymore and there can be artefacts in the images even if the problem admits a unique solution. In such cases we have "invisible singularities" [83, 84, 112, 124, 125].

We can generalize the transform  $X_0$  from lines to affine *d*-dimensional planes where 0 < d < n. The *d*-plane transform  $\mathcal{R}_d$  is defined as [64]

(3) 
$$\mathcal{R}_d f(A) = \int_A f(x) \mathrm{d}m(x)$$

where A is an affine d-dimensional plane, m is the d-dimensional Hausdorff measure and  $f: \mathbb{R}^n \to \mathbb{R}$  is a scalar field. The case d = 1 corresponds to the X-ray transform  $X_0$  and the case d = n - 1 is often called the Radon transform which coincides with the X-ray transform in two dimensions [64, 112, 129]. As before, the inverse problem is to invert the transform  $\mathcal{R}_d$  in equation (3). There is an inversion formula in terms of the normal operator  $\mathcal{N}_d$  of the *d*-plane transform which is defined in a similar way as in the case of the X-ray transform [64]. One example of partial data results for *d*-plane transforms is the Helgason support theorem where one knows the integrals of the scalar field over all *d*-planes which do not intersect a given compact and convex set [64].

The *d*-plane transform (also called the *k*-plane transform in some works) has been extensively studied after the pioneering work by Fuglede [49] and Helgason [63]. See for example [2, 55, 64, 68, 85, 127] and the works by Rubin [130, 131, 132, 133, 134].

1.3.2. X-ray tomography of vector fields. X-ray tomography is also used in the imaging of moving fluids which is based on Doppler backscattering or acoustic travel time measurements. If  $h: \mathbb{R}^n \to \mathbb{R}^n$  is a vector field which represents the flow field of a moving fluid, then after a linearization procedure one ends up studying the transform [115, 116]

(4) 
$$X_1 h(\gamma) = \int_{\gamma} h \cdot d\overline{s}.$$

The operator  $X_1$  is called the X-ray transform of vector fields and it has applications for example in medical ultrasound imaging [73, 75, 140, 148]. The inverse problem is to invert the operator  $X_1$  in equation (4).

Unlike in the scalar case we have a natural gauge: the gradients of scalar fields which vanish at infinity are always in the kernel of  $X_1$ . For this reason one can determine the vector field h only up to potential fields from its X-ray transform, i.e. one can only determine the solenoidal part  $h^s$  in the Helmholtz decomposition  $h = h^s + \nabla \phi$  where  $h^s \colon \mathbb{R}^n \to \mathbb{R}^n$  is a vector field such that div  $h^s = 0$  and  $\phi$  is a scalar field [140, 143, 149]. The solenoidal part can be uniquely determined from the full X-ray data and there is an inversion formula in terms of the normal operator  $N_1 = X_1^* X_1$  of the Xray transform of vector fields where  $X_1^*$  is the adjoint operator (or backprojection) [75, 115, 143, 148, 149].

Like in the scalar case, one can also study X-ray tomography of vector fields with partial data. The main goal in such problems is to determine the solenoidal part of the vector field from its partial X-ray data. Examples of such partial data results include cases where one knows the integrals of the vector field over lines which intersect a certain type of curve [42, 128, 159] or which are parallel to a finite set of planes [75, 139, 142]. There is also a vectorial version of the Helgason support theorem where one knows the integrals of the vector field over all lines not intersecting a given convex and compact set [149].

#### 1.4. Electrical impedance tomography and its non-local versions.

1.4.1. The Calderón problem. Electrical impedance tomography (EIT) is an imaging method which has applications in geophysics and medical imaging [78, 108, 157]. EIT is based on the conductivity equation and the inverse problem is known as the Calderón problem. In the Calderón problem we have an object whose electrical properties we want to deduce by making boundary measurements. In particular, we want to determine the conductivity inside the object by applying voltages on the boundary and measuring the induced currents on the boundary which depend on the electrical properties of the interior of the object.

We can model the object as a bounded domain  $\Omega \subset \mathbb{R}^n$  with sufficiently regular boundary  $\partial \Omega$ . The conductivity equation is [157]

(5) 
$$\begin{cases} \nabla \cdot (\eta \nabla u) = 0 \text{ in } \Omega \\ u|_{\partial \Omega} = f \end{cases}$$

where f is the potential on the boundary, u is the induced potential in  $\Omega$ and  $\eta$  is the electrical conductivity of  $\Omega$  which is assumed to be sufficiently smooth positive function. The measurements are encoded in the Dirichletto-Neumann (DN) map  $\Lambda_{\eta}$  which tells how the electrical properties of the interior induce normal currents on the boundary when one applies the voltage f on the boundary. More specifically, one can write  $\Lambda_{\eta}f = (\eta \partial_{\nu} u)|_{\partial\Omega}$ where  $\nu$  is the outer unit normal on  $\partial\Omega$ . The inverse problem is to determine the conductivity  $\eta$  in equation (5) by applying different boundary values f (voltages) and measuring the induced currents  $\Lambda_{\eta}f$ . In particular, the uniqueness problem is the following: if  $\Lambda_{\eta_1}f = \Lambda_{\eta_2}f$  for all boundary values f, does it follow that  $\eta_1 = \eta_2$ ? This problem was first studied mathematically by Alberto Calderón and the inverse problem is therefore known as the Calderón problem [19].

Using the substitution  $\tilde{u} = \sqrt{\eta}u$  one can convert the conductivity equation (5) to the following Schrödinger equation [110, 154, 157]

(6) 
$$\begin{cases} (-\Delta + q)\tilde{u} = 0 \text{ in } \Omega \\ \tilde{u}|_{\partial\Omega} = \tilde{f}. \end{cases}$$

Here  $q = (\Delta \sqrt{\eta})/\sqrt{\eta}$  now corresponds to the electric potential in  $\Omega$  and  $\tilde{f} = \sqrt{\eta} f$ . The DN map  $\Lambda_q$  for equation (6) can be written as  $\Lambda_q \tilde{f} = \partial_{\nu} \tilde{u}|_{\partial\Omega}$  assuming  $\partial\Omega$  is regular enough. The interpretation of the DN map is as in the conductivity equation: the DN map tells how the applied voltage on the boundary induces normal currents on the boundary via the electrical properties of the interior of the object. The inverse problem now is to determine the potential q in equation (6) by applying different boundary values  $\tilde{f}$  (voltages) and measuring the induced currents  $\Lambda_q \tilde{f}$ . The uniqueness problem is as for the conductivity equation: if  $\Lambda_{q_1} \tilde{f} = \Lambda_{q_2} \tilde{f}$  for all boundary values  $\tilde{f}$ , does it follow that  $q_1 = q_2$ ? One standard tool in proving uniqueness for the Calderón problem of the conductivity equation (5) and

Schrödinger equation (6) is the construction of complex geometrical optics solutions [8, 19, 153, 154, 157].

1.4.2. The fractional Calderón problem. One can study the non-local version of the Schrödinger equation (6) as follows. One replaces the Laplacian  $-\Delta$  with the fractional Laplacian  $(-\Delta)^s$  which is the pseudodifferential operator

$$(-\Delta)^s u = \mathcal{F}^{-1}(|\cdot|^{2s} \hat{u}), \quad s \in (-n/2, \infty) \setminus \mathbb{Z}.$$

The fractional Laplacian is a non-local operator in contrast to the ordinary Laplacian: the value  $(-\Delta)^s u(x)$  depends on the values of u everywhere in  $\mathbb{R}^n$ while  $-\Delta u(x)$  depends only on the values of u in a small neighborhood of  $x \in \mathbb{R}^n$ . For example, the normal operator of the X-ray transform  $N_0$  (and more generally the normal operator of the d-plane transform  $\mathcal{N}_d$ ) is the fractional Laplacian  $(-\Delta)^{-1/2}$  (more generally  $(-\Delta)^{-d/2}$ ) up to a constant factor. In addition to integral geometry fractional Laplacians arise also in non-local diffusion [16, 45, 54] and in fractional quantum mechanics [90, 91].

Replacing  $-\Delta$  with  $(-\Delta)^s$  where  $s \in (0,1)$  we obtain the fractional Schrödinger equation introduced in [54]

(7) 
$$\begin{cases} ((-\Delta)^s + q)u = 0 \text{ in } \Omega \\ u|_{\Omega_e} = f \end{cases}$$

where  $\Omega_e = \mathbb{R}^n \setminus \overline{\Omega}$  is the exterior of the bounded domain  $\Omega \subset \mathbb{R}^n$ . For such non-local equation (7) it is more natural to consider exterior values  $u|_{\Omega_e} = f$  instead of boundary values. The DN map  $\Lambda_q$  maps the "nonlocal voltage" f to a non-local version of the normal current [54]: under stronger assumptions one can write  $\Lambda_q f = (-\Delta)^s u|_{\Omega_e}$ . In the fractional Calderón problem one wants to determine the potential q in equation (7) by applying different exterior values f and measuring the induced "exterior currents"  $\Lambda_q f$ . The uniqueness problem is similar as in the local case: if  $\Lambda_{q_1} f = \Lambda_{q_2} f$  for all exterior values f, does it follow that  $q_1 = q_2$ ? The fractional Calderón problem for equation (7) was first studied by Ghosh, Salo and Uhlmann [54].

In fractional Calderón problems instead of constructing complex geometrical optics solutions one can exploit the non-locality of the equation and especially the non-local behaviour of the operator  $(-\Delta)^s$ . One has the following unique continuation property of fractional Laplacians [54]: if  $s \in (0, 1)$ and  $(-\Delta)^s u|_V = u|_V = 0$  for some nonempty open set  $V \subset \mathbb{R}^n$ , then u = 0. Clearly such property cannot hold for local operators such as  $-\Delta$ . The unique continuation of  $(-\Delta)^s$  is in essential role in proving uniqueness for fractional Calderón problems [13, 21, 30, 54].

After the seminal work [54] there have been numerous results for different variants of the fractional Calderón problem: these include stability and instability results [135, 136], uniqueness under single measurement [53], magnetic versions of the fractional Schrödinger equation [30, 96, 97, 98], lower order local and non-local perturbations [13, 21], semilinear equations [87, 88], fractional conductivity and heat equations [31, 89, 137] and equations arising from a non-local Schrödinger-type elliptic operator [20, 52].

#### 1.5. Travel time tomography and its linearization.

1.5.1. The boundary rigidity problem. In seismic travel time tomography the objective is to study the interior properties of the Earth by measuring travel times of seismic waves on the surface of the Earth [22, 62, 144, 152]. It is impossible to access the deep interior of the Earth by any practical means and the only way to obtain information is by doing indirect measurements on the surface. The travel times of seismic waves depend on the speed of sound in the medium where the wave propagates. Therefore the travel times contain indirect information about the physical properties of the Earth.

The Earth can be modelled as a three-dimensional compact manifold M with boundary  $\partial M$  (e.g. a closed ball). Assuming that the medium is isotropic the speed of sound depends only on position and it becomes a positive scalar function  $c: M \to (0, \infty)$ . The travel time of a seismic wave or ray can be expressed as the line integral [22]

(8) 
$$T = \int_{\gamma} \frac{\mathrm{d}s}{c}$$

where  $\gamma$  is the ray path. The travel time tomography problem or inverse kinematic problem is to solve the speed of sound c in equation (8) when the travel times T measured on the surface are known.

The travel time tomography problem was studied first in 1900s by Herglotz, Wiechert and Zoeppritz [65, 160]. They solved the problem assuming that the speed of sound is radial c = c(r) and satisfies the Herglotz condition

(9) 
$$\frac{\mathrm{d}}{\mathrm{d}r}\left(\frac{r}{c(r)}\right) > 0.$$

Under these assumptions the solution reduces to the inversion of an Abeltype integral transform [117, 144]. The Herglotz condition (9) is equivalent to the condition that the travel times in equation (8) are finite [36]. In geometrical terms, the Herglotz condition (9) means that one can foliate the manifold M with strictly convex hypersurfaces (i.e. spheres) [152].

The travel time tomography problem can be formulated in a more geometrical way. The speed of sound c determines the Riemannian metric  $g_c = c^{-2}(x)e$  where e is the Euclidean metric. By Fermat's principle the rays propagate along geodesics of the metric  $g_c$  and the travel times correspond to lengths of these geodesics [22]. The inverse problem is to determine the scalar function c, or equivalently the metric  $g_c$ , from the lengths of all geodesics connecting points on the boundary  $\partial M$ . One sees that the problem is highly non-linear since the geodesics depend on the function c (or the metric  $g_c$ ).

One can study the above geometric problem in a more general case: if g is a Riemannian metric, determine g from the distances between boundary

points (boundary distances) given by g. This geometric inverse problem is known as the boundary rigidity problem [152]. In particular, one problem of interest is the uniqueness problem: if two Riemannian metrics  $g_1$  and  $g_2$ give the same boundary distances, does it follow that  $g_1 = g_2$ ? The answer is no in general since there is a gauge: if  $g_2 = \Psi^* g_1$  where  $\Psi: M \to M$  is a diffeomorphism which is identity on the boundary, then  $g_1$  and  $g_2$  give the same boundary distances [152]. Hence without further restrictions one can determine the metric only up to a boundary preserving diffeomorphism.

The boundary rigidity problem is a difficult non-linear inverse problem and it has been solved only in certain special cases where the manifold admits strictly convex foliation [150, 152] or the manifold is known to be simple (a generalization of a Euclidean ball). Boundary rigidity holds for simple subspaces of Euclidean space [57] and simple subspaces of symmetric spaces of constant negative curvature [12]. In two dimensions examples include simple subspaces of the open hemisphere [104] and simple spaces of negative curvature [33]. If the Riemannian metrics on a compact simple Riemannian manifold are in the same conformal class, then the distances between boundary points determine the metric uniquely, i.e. the diffeomorphism  $\Psi$  becomes identity in this case [34, 109, 152]. In general, compact simple Riemannian manifolds are known to be boundary rigid in two dimensions [122], but it is conjectured that boundary rigidity holds for compact simple Riemannian manifolds of any dimension [103].

In the travel time tomography problem one usually assumes that the speed of sound c is isotropic, i.e. it only depends on position. However, anisotropies have been observed in the shallow crust, upper mantle and inner core of the Earth [32, 46, 144]. Therefore it is reasonable to consider c as a function on the tangent bundle  $c: TM \to (0, \infty)$  so that the dependence on the direction of propagation can be taken into account. If the sound speed is anisotropic, then the seismic rays propagate along geodesics of a Finsler norm and we need Finsler geometry to treat the anisotropies [7, 161]. The travel time tomography problem can then be expressed as a boundary rigidity problem on Finsler manifolds where the fiberwise inner product depends not only on position but also on direction.

The boundary rigidity problem is much harder in the Finslerian case since there are non-isometric Finsler norms which give the same boundary distances [17, 25, 26, 72]. This means that in general Finsler norms are not rigid in the same way as Riemannian metrics. However, some rigidity results are known in certain special cases. Projectively flat Finsler norms on compact convex domains of  $\mathbb{R}^2$  are uniquely determined by their boundary distances [4, 5, 86]. When we restrict ourselves to Finsler norms which are relevant in seismology, we can expect more rigidity: one can use the collection of boundary distance maps to determine the differential and topological structures of Finsler manifolds [38], and the broken scattering relation determines the isometry class of reversible Finsler manifolds which admit strictly convex foliation [37]. 1.5.2. Linearized versions of the boundary rigidity problem. Let us study the linearization of the boundary rigidity problem. Let  $\epsilon > 0$  and  $s \in (-\epsilon, \epsilon)$ . Assume that  $g^s$  is a family of Riemannian metrics which all give the same boundary distances where  $g^0$  corresponds to a known "background metric". When we linearize the boundary rigidity problem, we calculate the derivative of the boundary distances at s = 0. Since these distances do not depend on the parameter s we obtain [143]

$$0 = \int_{a}^{b} \frac{\partial g_{ij}^{s}(\gamma_{0}(t))}{\partial s} \bigg|_{s=0} \dot{\gamma}_{0}^{i}(t) \dot{\gamma}_{0}^{j}(t) \mathrm{d}t$$

where  $\gamma_0: [a, b] \to M$  is a geodesic of the base manifold  $(M, g^0)$  connecting two boundary points. If the variations  $g^s$  are conformal, i.e.  $g^s = f_s g_0$ where  $f_s: M \to \mathbb{R}$  is a family of positive scalar functions such that  $f_0 = 1$ , then the linearization leads to

$$0 = \int_{a}^{b} \frac{\partial f_{s}(\gamma_{0}(t))}{\partial s} \bigg|_{s=0} \mathrm{d}t.$$

The previous observations motivate us to study the kernel of the geodesic ray transform of symmetric *m*-tensor fields where  $m \ge 0$ . The geodesic ray transform of a scalar field  $f: M \to \mathbb{R}$  (or 0-tensor field) on a Riemannian manifold (M, g) is defined as

(10) 
$$\mathcal{I}_0 f(\gamma) = \int_{\tau_\gamma^-}^{\tau_\gamma^+} f(\gamma(t)) \mathrm{d}t$$

where  $\gamma : [\tau_{\gamma}^{-}, \tau_{\gamma}^{+}] \to M$  is a geodesic defined on the maximal interval  $[\tau_{\gamma}^{-}, \tau_{\gamma}^{+}]$ which can be finite or infinite. More generally, the geodesic ray transform of a symmetric (covariant) *m*-tensor field *h* is  $(m \ge 1)$ 

(11) 
$$\mathcal{I}_m h(\gamma) = \int_{\tau_{\gamma}^-}^{\tau_{\gamma}^+} h_{i_1 \dots i_m}(\gamma(t)) \dot{\gamma}^{i_1}(t) \cdots \dot{\gamma}^{i_m}(t) \mathrm{d}t$$

where  $h_{i_1...i_m}(x)$  are the components of the *m*-tensor field h in local coordinates and we have used the Einstein summation convention (repeated indices which appear both as a subscript and superscript are implicitly summed over). The geodesic ray transform  $\mathcal{I}_m$  can be seen as a generalization of the Euclidean X-ray transform since in Euclidean space geodesics are lines. However, Funk studied the geodesic ray transform of scalar fields on the sphere  $S^2 \subset \mathbb{R}^3$  (also known as the Funk transform) before Radon introduced the Euclidean X-ray transform or Radon transform [50, 51, 64].

The inverse problem in geodesic ray tomography is to determine the *m*tensor field *h* (or the scalar field *f*) from its integrals along geodesics, i.e. we want to invert the operator  $\mathcal{I}_m$  in equation (11) (or in equation (10)). As in the case of vector fields in  $\mathbb{R}^n$  there is a gauge for *m*-tensor fields of order  $m \geq 1$ : if *h* is the symmetrized covariant derivative of an m-1-tensor field which vanishes on the boundary (or at infinity), then *h* is in the kernel of  $\mathcal{I}_m$ . Therefore one can only determine the solenoidal part of the *m*-tensor field from its geodesic ray transform [69, 120, 143]; if this can be done we say that  $\mathcal{I}_m$  is solenoidally injective (or s-injective) on *m*-tensor fields.

The solenoidal injectivity of  $\mathcal{I}_m$  has been widely studied and we list only some special cases here: comprehensive treatment can be found in the reviews [69, 120]. If (M, g) is a compact simple Riemannian manifold, then the geodesic ray transform is injective on scalar fields and s-injective on 1forms [6, 109]. Solenoidal injectivity is known for tensor fields of any order on two-dimensional compact simple manifolds [119], on simply connected compact manifolds with strictly convex boundary and non-positive curvature [118, 123, 143] and on non-compact Cartan–Hadamard manifolds under certain decay conditions on the tensor fields and on the curvature [94, 95]. If  $n \geq 3$  and m = 0, 1, 2, 4, then solenoidal injectivity follows from foliation condition by strictly convex hypersurfaces [41, 151, 158]. There are also some partial data results for scalar and tensor fields under restrictions on the Riemannian metric [81, 151, 158]. We also mention that one of the basic general tools in studying solenoidal injectivity of  $\mathcal{I}_m$  is an energy estimate also known as the Pestov identity [69, 109, 120].

An interesting generalization of the geodesic ray transform in two dimensions is the mixed ray transform [40, 143] (12)

$$L_{k,l}h(\gamma) = \int_{\tau_{\gamma}^-}^{\tau_{\gamma}^+} h_{i_1\dots i_k j_1\dots j_l}(\gamma(t))(\dot{\gamma}(t)^{\perp})^{i_1}\cdots(\dot{\gamma}(t)^{\perp})^{i_k}\dot{\gamma}^{j_1}(t)\cdots\dot{\gamma}^{j_l}(t)\mathrm{d}t$$

where  $\dot{\gamma}(t)^{\perp}$  denotes the rotation of  $\dot{\gamma}(t)$  by 90 degrees counterclockwise and k + l = m. The mixed ray transform  $L_{k,l}$  arises in the linearization of the elastic travel time tomography problem [39, 40, 143]. If k = 0, then  $L_{k,l}$  reduces to the geodesic ray transform  $\mathcal{I}_m$ . When l = 0, we have the transverse ray transform [143]

(13) 
$$\mathcal{I}_m^{\perp}h(\gamma) = \int_{\tau_{\gamma}^-}^{\tau_{\gamma}^+} h_{i_1\dots i_m}(\gamma(t))(\dot{\gamma}(t)^{\perp})^{i_1}\cdots(\dot{\gamma}(t)^{\perp})^{i_m} \mathrm{d}t.$$

The mixed and transverse ray transforms in equations (12) and (13) can be extended to higher dimensions n > 2, but they become tensor-valued transforms [39, 143]. On two-dimensional orientable manifolds one can study the mixed ray transform by reducing it to the geodesic ray transform using rotations [40, 143].

The transverse ray transform was first studied by Braun and Hauck in two-dimensional Euclidean space with applications to flame analysis [15, 113, 141, 147]. Other applications of the transverse ray transform include diffraction tomography [99], polarization tomography [143] and photoelasticity [61]. The kernel of the transverse ray transform is known in  $\mathbb{R}^2$  and on higher dimensional manifolds  $(n \geq 3) \mathcal{I}_m^{\perp}$  is even injective under certain conditions [43, 113, 143]. There are also partial data results for the transverse ray transform [1, 82]. For the mixed ray transform some results related to solenoidal injectivity are known in  $\mathbb{R}^2$ , on two- and three-dimensional compact simple manifolds, and on manifolds satisfying certain curvature estimates [39, 40, 43, 143].

### 2. X-RAY TOMOGRAPHY WITH PARTIAL DATA: [A, B, C, G]

In the articles [A, B, C, G] we study partial data problems arising in the Xray tomography of scalar and vector fields. The basic question in such problems is the following: can we say something about the scalar field  $f : \mathbb{R}^n \to \mathbb{R}$ if we know the integrals of f (the X-ray transform  $X_0 f$ ) on all lines intersecting a given nonempty open set  $V \subset \mathbb{R}^n$ ? We have focused in the uniqueness problem: if  $X_0 f = 0$  on all lines intersecting V, does it follow that f = 0? In general, the knowledge of the integrals is not enough to determine funiquely [112] and therefore one has to put additional assumptions on f.

We have studied the partial data problem under different assumptions. In the most general case we assume that f satisfies a constant coefficient partial differential equation in V in a weak sense. If P is a polynomial, we let P(D)be the constant coefficient partial differential operator induced by P, i.e. we consider the partial derivatives D as variables in P. For example, the polynomial  $P(\xi) = \xi_1^2 + \ldots + \xi_n^2$  corresponds to the Laplacian  $P(D) = -\Delta$ . The main idea of the partial data problem is illustrated in figure 2: if

The main idea of the partial data problem is illustrated in figure 2: if  $X_0 f = 0$  on all lines intersecting V and  $P(D)f|_V = 0$  for some constant coefficient partial differential operator P(D), does it follow that f = 0 everywhere? Using the linearity of  $X_0$  and P(D), and the commutativity of distributional derivatives we see that this is indeed a uniqueness problem in the following sense: if  $f_1$  and  $f_2$  are scalar fields such that  $P_1(D)f_1|_V = P_2(D)f_2|_V = 0$  and  $X_0f_1 = X_0f_2$  on all lines intersecting V, does it follow that  $f_1 = f_2$  in all of  $\mathbb{R}^n$ ? The partial data problem can be reduced to a unique continuation problem of the normal operator  $N_0$  of the X-ray transform: if  $N_0f|_V = P(D)f|_V = 0$ , does it follow that f = 0 everywhere?

More generally, one can replace lines with d-planes in the partial data problem of scalar fields. In this way we obtain a partial data problem for the d-plane transform  $\mathcal{R}_d$ : if  $\mathcal{R}_d f = 0$  on all d-planes intersecting V and  $P(D)f|_V = 0$ , is it true that f = 0? The partial data problem for vector fields is formulated analogously as in the scalar case. However, for vector fields the problem is naturally formulated in terms of the curl (or the exterior derivative) of the vector field: if  $X_1 h = 0$  on all lines which intersect V and P(D)(dh) = 0 where h is a vector field and dh its curl, does it follow that dh = 0? By the Poincaré lemma this is equivalent to that the solenoidal part of h vanishes [67, 100, 143]. As in the case of the X-ray transform of scalar fields, the partial data problems for  $\mathcal{R}_d$  and  $X_1$  can be reduced to the corresponding unique continuation problems of the normal operator  $\mathcal{N}_d$  of the d-plane transform and the normal operator  $N_1$  of the X-ray transform



FIGURE 2. The partial data problem for the X-ray transform of scalar fields in its most general form as we have studied. Here  $V \subset \mathbb{R}^n$  is a nonempty open set, P(D) is a constant coefficient partial differential operator and  $\gamma$  is a line which intersects V.

of vector fields. We focus on studying the partial data problems from the point of view of the unique continuation of the different normal operators.

In the article [A] we study the partial data problem for  $X_0$  under the assumption  $f|_V = 0$ . The main result of the article [A] is a unique continuation property of Riesz potentials which correspond to fractional Laplacians with negative exponents. The Riesz potential of a scalar function  $f: \mathbb{R}^n \to \mathbb{R}$  is defined as the convolution  $I_{\alpha}f = f * |\cdot|^{-\alpha}$  where  $\alpha < n$  (see section 2.1). The main theorem is the following: if  $I_{\alpha}f$  vanishes to infinite order at some point  $x_0 \in V$  where the exponent  $\alpha$  satisfies some conditions and  $f|_V = 0$ , then f = 0. This implies a unique continuation result for the normal operator  $N_0$ : if  $N_0 f|_V = f|_V = 0$ , then f = 0. The unique continuation of  $N_0$  can then be used to prove uniqueness for the partial data problem: if  $X_0 f = 0$ on all lines intersecting V and  $f|_V = 0$ , then f = 0. We also provide an application of the partial data result to linearized travel time tomography in Euclidean background.

In the article [B] we study the partial data problem for the *d*-plane transform  $\mathcal{R}_d$  in the case  $f|_V = 0$ . This is a generalization of the problem studied in the article [A] where we considered the case d = 1. As in the article [A], the partial data problem is studied using the normal operator  $\mathcal{N}_d$  of the *d*-plane transform. One of the main results of the article [B] is a unique continuation property of fractional Laplacians: if  $(-\Delta)^s f|_V = f|_V = 0$  where  $s \in (-n/2, \infty) \setminus \mathbb{Z}$ , then f = 0. When *d* is odd, this implies a unique continuation result for  $\mathcal{N}_d$ : if  $\mathcal{N}_d f|_V = f|_V = 0$  and *d* is odd, then f = 0. The unique continuation of  $\mathcal{N}_d$  then implies uniqueness for the partial data problem: if d is odd,  $\mathcal{R}_d f = 0$  on all d-planes intersecting V and  $f|_V = 0$ , then f = 0. When d is even and  $\mathcal{R}_d f = 0$  on all lines which intersect V, we can locally invert the d-plane data to obtain that  $f|_V = 0$ .

The article [C] considers the partial data problem for  $X_1$  under the assumption  $dh|_V = 0$ . The approach is similar as in the scalar case, and the main result of the article [C] is a unique continuation property of  $N_1$ : if  $N_1h$ vanishes to infinite order at some point in V and  $dh|_V = 0$ , then dh = 0. This unique continuation result is proved by reducing it to a unique continuation problem of  $N_0$  treated in the article [A]. The unique continuation of  $N_1$  can then be used to prove uniqueness for the partial data problem: if  $X_1h = 0$  on all lines which intersect V and  $dh|_V = 0$ , then dh = 0. This is equivalent to that  $h = d\phi$  for some scalar field  $\phi$  by the Poincaré lemma, or to that the solenoidal part of h vanishes. In the article [C] we also obtain partial data results for the matrix-weighted X-ray transform of vector fields which special case, the Euclidean transverse ray transform, we study in two dimensions.

The article [G] is a continuation of the articles [A, C] and the integral geometry part of the article [B]. In particular, we extend the assumptions  $f|_V = 0$  and  $dh|_V = 0$  in the partial data problems of scalar and vector fields to the more general cases  $P(D)f|_V = 0$  and  $P(D)(dh)|_V = 0$ where P(D) is a constant coefficient partial differential operator induced by the polynomial P as above. The main result of the article [G] is a unique continuation property of fractional Laplacians which generalizes the unique continuation result proved in the article [B]: if  $(-\Delta)^s f|_V = P(D)f|_V = 0$ where  $s \in (-n/2, \infty) \setminus \mathbb{Z}$  and P(D) is any constant coefficient partial differential operator, then f = 0. This unique continuation result directly implies a corresponding unique continuation property for  $\mathcal{N}_d$ : if d is odd and  $\mathcal{N}_d f|_V = P(D)f|_V = 0$ , then f = 0. Using reduction to the scalar case one also obtains a unique continuation result for  $N_1$ : if  $N_1h|_V = P(D)(dh)|_V =$ 0, then dh = 0. These unique continuation results for  $\mathcal{N}_d$  and  $N_1$  then imply uniqueness for the most general partial data problems we have studied: if  $\mathcal{R}_d f = 0$  on all d-planes intersecting V where d is odd and  $P(D)f|_V = 0$ (or  $X_1h = 0$  on all lines intersecting V and  $P(D)(dh)|_V = 0$ ), then f = 0(respectively dh = 0).

2.1. Notation. Let us first introduce some notation before giving the main theorems. We will follow the notation conventions of the references [64, 102, 106, 112, 143, 149, 155].

We write f for a scalar function or distribution. The space of tempered distributions is denoted by  $\mathscr{S}'(\mathbb{R}^n)$ . We let  $\mathscr{O}'_C(\mathbb{R}^n) \subset \mathscr{S}'(\mathbb{R}^n)$  be the space of rapidly decreasing distributions. It contains as a subset all compactly supported distributions  $\mathcal{E}'(\mathbb{R}^n) \subset \mathscr{O}'_C(\mathbb{R}^n)$  and all continuous functions which decrease faster than any polynomial at infinity  $C_{\infty}(\mathbb{R}^n) \subset \mathscr{O}'_C(\mathbb{R}^n)$ . The fractional  $L^2$ -Sobolev space of order  $r \in \mathbb{R}$  is defined as

$$H^{r}(\mathbb{R}^{n}) = \{ f \in \mathscr{S}'(\mathbb{R}^{n}) : \mathcal{F}^{-1}(\langle \cdot \rangle^{r} \hat{f}) \in L^{2}(\mathbb{R}^{n}) \}$$

where  $\langle x \rangle = (1 + |x|^2)^{1/2}$ ,  $\hat{f} = \mathcal{F}(f)$  is the Fourier transform of tempered distributions and  $\mathcal{F}^{-1}$  is the inverse Fourier transform. These spaces are nested, i.e.  $H^r(\mathbb{R}^n) \hookrightarrow H^t(\mathbb{R}^n)$  continuously when  $r \geq t$ , and one can identify  $H^{-r}(\mathbb{R}^n)$  with the dual  $(H^r(\mathbb{R}^n))^*$  for every  $r \in \mathbb{R}$ . We let  $H^{-\infty}(\mathbb{R}^n) = \bigcup_{r \in \mathbb{R}} H^r(\mathbb{R}^n)$  so that  $\mathcal{O}'_C(\mathbb{R}^n) \subset H^{-\infty}(\mathbb{R}^n) \subset \mathcal{S}'(\mathbb{R}^n)$ . The fractional Laplacian is defined via Fourier transform

$$(-\Delta)^s f = \mathcal{F}^{-1}(|\cdot|^{2s} \hat{f}), \quad s \in (-n/2, \infty) \setminus \mathbb{Z}.$$

We have that  $(-\Delta)^s f$  defines a tempered distribution for  $f \in \mathscr{O}'_C(\mathbb{R}^n)$  when  $s \in (-n/2, \infty) \setminus \mathbb{Z}$ , and for  $f \in H^r(\mathbb{R}^n)$  when  $s \in (-n/4, \infty) \setminus \mathbb{Z}$ .

The fractional Laplacian has a connection to Riesz potentials. Let  $\alpha \in \mathbb{R}$ such that  $\alpha < n$ . We define the Riesz potential  $I_{\alpha} \colon \mathscr{O}'_{C}(\mathbb{R}^{n}) \to \mathscr{S}'(\mathbb{R}^{n})$  as  $I_{\alpha}f = f * h_{\alpha}$  where the kernel is  $h_{\alpha}(x) = |x|^{-\alpha}$ . If in addition  $0 < \alpha < n$ , then  $I_{\alpha} = (-\Delta)^{-s}$  up to a constant factor with  $s = (n-\alpha)/2$ . On the other hand, if -n/2 < s < 0, then we can write  $(-\Delta)^{s}f = I_{2s+n}f$  up to a constant factor. We say that  $I_{\alpha}f$  vanishes to infinite order at a point  $x_{0} \in \mathbb{R}^{n}$ , if  $I_{\alpha}f$  is smooth in a neighborhood of  $x_{0}$  and  $\partial^{\beta}(I_{\alpha}f)(x_{0}) = 0$  for all multi-indices  $\beta \in \mathbb{N}^{n}$ .

We let  $\mathcal{P}$  be the set of all polynomials on  $\mathbb{R}^n$  with complex coefficients excluding the zero polynomial  $P \equiv 0$ . If  $P \in \mathcal{P}$  is a polynomial of degree  $m \in \mathbb{N}$ , then it can be identified with the constant coefficient partial differential operator P(D) of order  $m \in \mathbb{N}$  by writing  $P(D) = \sum_{|\alpha| \leq m} a_{\alpha} D^{\alpha}$  where  $a_{\alpha} \in \mathbb{C}, D^{\alpha} = D_1^{\alpha_1} \cdots D_n^{\alpha_n}, D_j = -i\partial_j$  and  $\alpha = (\alpha_1, \ldots, \alpha_n) \in \mathbb{N}^n$  is a multi-index so that  $|\alpha| = \alpha_1 + \ldots + \alpha_n$ . If  $V \subset \mathbb{R}^n$  is a nonempty open set, we define the set of admissible functions  $\mathcal{A}_V$  by setting

$$\mathcal{A}_V = \{ f \in H^{-\infty}(\mathbb{R}^n) : P(D)f|_V = 0 \text{ for some } P \in \mathcal{P} \}.$$

One can see that the set  $\mathcal{A}_V \subset H^{-\infty}(\mathbb{R}^n)$  forms a vector space.

The X-ray transform of scalar fields is denoted by  $X_0$  and it takes a function f and integrates it over lines. The normal operator is  $N_0 = X_0^* X_0$ where  $X_0^*$  is the adjoint of  $X_0$ . If f is a distribution, then  $X_0 f$  and  $N_0 f$  are defined by duality. More generally, we denote by  $\mathcal{R}_d$  the d-plane transform of scalar fields. The transform  $\mathcal{R}_d$  takes a scalar field f and integrates it over d-dimensional planes where 0 < d < n. The normal operator of the d-plane transform is  $\mathcal{N}_d = \mathcal{R}_d^* \mathcal{R}_d$  where  $\mathcal{R}_d^*$  is the adjoint of  $\mathcal{R}_d$ . If f is a distribution, then  $\mathcal{R}_d f$  and  $\mathcal{N}_d f$  are defined by using duality.

We denote by h a vector field or vector-valued distribution. We write  $h \in (\mathcal{E}'(\mathbb{R}^n))^n$  if  $h = (h_1, \ldots, h_n)$  where  $h_i \in \mathcal{E}'(\mathbb{R}^n)$  for all  $i = 1, \ldots, n$ . The exterior derivative or curl of h is a matrix whose components are  $(dh)_{ij} = \partial_i h_j - \partial_j h_i$ . The X-ray transform of vector fields is denoted by  $X_1$  and it maps a vector field to its line integrals. The normal operator is  $N_1 = X_1^* X_1$ 

where  $X_1^*$  is the adjoint of  $X_1$ . If h is a vector-valued distribution, then both  $X_1h$  and  $N_1h$  are defined by duality.

2.2. Main results. The following two theorems are the main results of the article [A]. The first one is a unique continuation result for Riesz potentials and the second one is a partial data result for the X-ray transform of scalar fields.

**Theorem 2.1** ([A, Theorem 1.1]). Let  $\alpha = n - 1$  or  $\alpha \in \mathbb{R} \setminus \mathbb{Z}$  and  $\alpha < n$ where  $n \geq 2$ . Let  $f \in \mathcal{E}'(\mathbb{R}^n)$ ,  $V \subset \mathbb{R}^n$  any nonempty open set and  $x_0 \in V$ . If  $f|_V = 0$  and  $I_{\alpha}f$  vanishes to infinite order at  $x_0$ , then f = 0.

**Theorem 2.2** ([A, Theorem 1.2]). Let  $V \subset \mathbb{R}^n$  be any nonempty open set where  $n \geq 2$ . If  $f \in \mathcal{E}'(\mathbb{R}^n)$  satisfies  $f|_V = 0$  and  $X_0 f$  vanishes on all lines that intersect V, then f = 0.

Theorem 2.1 is proved by showing that one can obtain all the polynomials in a certain form by taking finite linear combinations of the derivatives of the integral kernel  $h_{\alpha}$  in  $I_{\alpha}f = f * h_{\alpha}$ . The density of polynomials in the space of smooth functions then gives the claim since  $f \in \mathcal{E}'(\mathbb{R}^n)$  belongs to the dual of that space. We give multiple proofs for theorem 2.2. Two proofs reduce the partial data problem to a unique continuation problem of normal operator: if  $X_0f = 0$  on all lines intersecting V, then  $N_0f|_V = 0$ . The normal operator  $N_0$  can be seen as the Riesz potential  $I_{n-1}$  up to a constant factor, or equivalently, as the fractional Laplacian  $(-\Delta)^{-1/2}$  up to a constant factor. The partial data result then follows from theorem 2.1, or by using the unique continuation of fractional Laplacians which is proved in [54]. The third proof works directly at the level of the X-ray transform and is based on angular Fourier series and density of polynomials.

In addition, we provide an application of theorem 2.2 to linearized travel time tomography in Euclidean background. In particular, we show how one can use global shear wave splitting data to uniquely determine the difference of the S-wave speeds in weak anisotropy. We also show in the article [A] how one can use "half-local" measurements of travel times to uniquely determine the conformal factor in the linearization; this is a partial data result where we measure travel times of seismic waves in a small open subset of the surface of the Earth, but the waves can emanate from anywhere on the surface.

In the article [B] we generalize the unique continuation and partial data results proved in [A] for scalar fields to *d*-plane transforms. The following two theorems are the main results of the integral geometry part of the article [B].

**Theorem 2.3** ([B, Corollary 1]). Let  $n \ge 2$  and let f belong to either  $\mathcal{E}'(\mathbb{R}^n)$ or  $C_{\infty}(\mathbb{R}^n)$ . Let  $d \in \mathbb{N}$  be odd such that 0 < d < n. If  $\mathcal{N}_d f|_V = 0$  and  $f|_V = 0$  for some nonempty open set  $V \subset \mathbb{R}^n$ , then f = 0.

**Theorem 2.4** ([B, Corollary 2]). Let  $n \ge 2$ ,  $V \subset \mathbb{R}^n$  a nonempty open set and  $f \in C_{\infty}(\mathbb{R}^n)$  or  $f \in \mathcal{E}'(\mathbb{R}^n)$ . Let  $d \in \mathbb{N}$  be odd such that 0 < d < n. If  $f|_V = 0$  and  $\mathcal{R}_d f = 0$  for all d-planes intersecting V, then f = 0. Theorem 2.3 is proved by using a unique continuation property of fractional Laplacians which is proved in the same article [B] (see theorem 3.1). Unique continuation of fractional Laplacians can be used since the normal operator  $\mathcal{N}_d$  of the *d*-plane transform corresponds to the fractional Laplacian  $(-\Delta)^{-d/2}$  up to a constant factor. The unique continuation of  $\mathcal{N}_d$  is then used to prove theorem 2.4. For this reason we have to assume that *d* is odd: theorem 2.3 does not hold if *d* is even since in that case  $\mathcal{N}_d$  is the inverse of a local operator. However, if *d* is even, then the partial data problem for the *d*-plane transform is locally uniquely solvable: if  $\mathcal{R}_d f = 0$  on all lines intersecting *V* and *d* is even, then  $f|_V = 0$ .

In the article [C] we generalize the above partial data results to vector fields. The following two main theorems of the article [C] are similar to theorems 2.1 and 2.2.

**Theorem 2.5** ([C, Theorem 1.1]). Let  $h \in (\mathcal{E}'(\mathbb{R}^n))^n$  and  $V \subset \mathbb{R}^n$  some nonempty open set where  $n \geq 2$ . If  $dh|_V = 0$  and  $N_1h$  vanishes to infinite order at  $x_0 \in V$ , then  $h = d\phi$  for some  $\phi \in \mathcal{E}'(\mathbb{R}^n)$ .

**Theorem 2.6** ([C, Theorem 1.2]). Let  $h \in (\mathcal{E}'(\mathbb{R}^n))^n$  and  $V \subset \mathbb{R}^n$  some nonempty open set where  $n \geq 2$ . Assume that  $dh|_V = 0$ . Then  $X_1h$  vanishes on all lines intersecting V if and only if  $h = d\phi$  for some  $\phi \in \mathcal{E}'(\mathbb{R}^n)$ .

Instead of assuming that  $N_1h$  vanishes to infinite order in theorem 2.5 we could require that  $d(N_1h)$  vanishes componentwise to infinite order at some point  $x_0 \in V$ . This weaker condition implies the claim since theorem 2.5 is proved by using theorem 2.1 and the fact that  $d(N_1h) = N_0(dh)$  holds componentwise up to a constant factor. We provide two alternative proofs for theorem 2.6. The first proof directly uses the unique continuation of the normal operator  $N_1$  in theorem 2.5. The second proof is based on Stokes' theorem and theorem 2.2. Both proofs use the same idea: from the assumptions we deduce that dh = 0 and the Poincaré lemma implies that  $h = d\phi$ for some scalar field  $\phi$ .

In the article [C] we also study the matrix-weighted X-ray transform of vector fields  $X_A = X_1 \circ A$  where A is a smooth invertible matrix field. Similar results as in theorems 2.5 and 2.6 are obtained for the transform  $X_A$ . As a special case of the transform  $X_A$  we obtain results for the Euclidean transverse ray transform in two dimensions.

In the article [G] we generalize the partial data and unique continuation results obtained in the articles [A, B, C]. The partial data results are proved by using the following unique continuation property of fractional Laplacians which is a generalization of the unique continuation result we proved in the article [B].

**Theorem 2.7** ([G, Theorem 1.1]). Let  $n \ge 1$ ,  $s \in (-n/4, \infty) \setminus \mathbb{Z}$  and  $f \in \mathcal{A}_V$ where  $V \subset \mathbb{R}^n$  is some nonempty open set. If  $(-\Delta)^s f|_V = 0$ , then f = 0. If  $f \in \mathscr{O}'_C(\mathbb{R}^n) \cap \mathcal{A}_V$ , then the claim holds for  $s \in (-n/2, \infty) \setminus \mathbb{Z}$ . The condition  $f \in \mathcal{A}_V$  means that  $f \in H^r(\mathbb{R}^n)$  for some  $r \in \mathbb{R}$  and  $P(D)f|_V = 0$  for some constant coefficient partial differential operator P(D). Theorem 2.7 is proved by using the unique continuation result of fractional Laplacians proved in [B] (see theorem 3.1) for the scalar field P(D)f. The assumptions and locality of P(D) imply the conditions  $P(D)f|_V = (-\Delta)^s (P(D)f)|_V = 0$  and hence f has to satisfy the global partial differential equation P(D)f = 0 which has only trivial solutions in the class of admissible functions  $\mathcal{A}_V$ .

As before, the unique continuation of fractional Laplacians in theorem 2.7 can be used to prove partial data results for scalar and vector fields. The following two theorems of the article [G] are generalizations of theorems 2.2 and 2.6.

**Theorem 2.8** ([G, Theorem 1.4]). Let  $n \ge 2$  and  $f \in \mathcal{E}'(\mathbb{R}^n) \cap \mathcal{A}_V$  or  $f \in C_{\infty}(\mathbb{R}^n) \cap \mathcal{A}_V$  where  $V \subset \mathbb{R}^n$  is some nonempty open set. If  $X_0 f = 0$  on all lines intersecting V, then f = 0.

**Theorem 2.9** ([G, Theorem 1.7]). Let  $n \ge 2$  and  $h \in (\mathcal{E}'(\mathbb{R}^n))^n$  such that  $(dh)_{ij} \in \mathcal{A}_V$  for all i, j = 1, ..., n where  $V \subset \mathbb{R}^n$  is some nonempty open set. If  $X_1h = 0$  on all lines intersecting V, then dh = 0. Especially,  $h = d\phi$  for some  $\phi \in \mathcal{E}'(\mathbb{R}^n)$ .

Theorems 2.8 and 2.9 are proved in the following way. Theorem 2.7 implies a corresponding unique continuation result for the normal operator  $N_0$ . The unique continuation of  $N_0$  is then used to prove the partial data result in theorem 2.8. Further, using again the fact that  $d(N_1h) = N_0(dh)$  holds componentwise up to a constant factor we can prove a unique continuation property for  $N_1$ , which in turn implies the partial data result in theorem 2.9. When d is odd, theorem 2.7 implies a corresponding unique continuation result for the normal operator  $\mathcal{N}_d$ , which in turn implies a similar partial data result as in theorem 2.8 for the d-plane transform  $\mathcal{R}_d$ .

3. HIGHER ORDER FRACTIONAL CALDERÓN PROBLEMS: [B, D]

In the articles [B, D] we study uniqueness for higher order fractional Calderón problems. Let  $\Omega \subset \mathbb{R}^n$  be an open bounded set,  $\Omega_e = \mathbb{R}^n \setminus \overline{\Omega}$ its exterior and  $s \in (0, \infty) \setminus \mathbb{Z}$ . We consider the Calderón problem for the fractional Schrödinger equation

(14) 
$$\begin{cases} ((-\Delta)^s + q)u = 0 \text{ in } \Omega \\ u|_{\Omega_e} = f \end{cases}$$

and for the more general equation involving lower order local perturbations of the fractional Laplacian

(15) 
$$\begin{cases} ((-\Delta)^s + P(x,D))u = 0 \text{ in } \Omega \\ u|_{\Omega_e} = f \end{cases}$$

20

where P(x, D) is a variable coefficient partial differential operator of order  $m \in \mathbb{N}$ . We can write P(x, D) as

$$P(x,D) = \sum_{|\alpha| \le m} a_{\alpha}(x) D^{\alpha}$$

where the coefficients  $a_{\alpha} = a_{\alpha}(x)$  are functions in  $\Omega$  (or more generally Sobolev multipliers in  $\mathbb{R}^n$ ). We assume that m < 2s so that P(x, D) can be considered as a lower order perturbation to  $(-\Delta)^s$ . We see that equation (14) is a special case of equation (15) and the potential q can be treated as a zeroth order perturbation to  $(-\Delta)^s$ .

The inverse problem for equations (14) and (15) is illustrated in figure 3. Formally, we put some "non-local voltage" in the open set  $W_1 \subset \Omega_e$  and measure "non-local currents" in the open set  $W_2 \subset \Omega_e$ . More precisely, the fractional Calderón problem for equation (15) is formulated as follows: if the DN maps  $\Lambda_{P_1}$  and  $\Lambda_{P_2}$  agree in  $W_2$  for all exterior values  $f \in C_c^{\infty}(W_1)$ , does it follow that the partial differential operators  $P_1$  and  $P_2$  are equal in  $\Omega$ ? This problem was first introduced by Ghosh, Salo and Uhlmann in their seminal work [54] where the authors studied equation (14) in the case  $s \in (0, 1)$ . We can think the inverse problem as a partial data problem: instead of having data in the full exterior  $\Omega_e$  we only have information or make measurements in the (possibly small) open subsets  $W_1, W_2 \subset \Omega_e$ .



FIGURE 3. The fractional Calderón problem in its most general form as we have studied. Here  $\Omega \subset \mathbb{R}^n$  is a bounded open set and  $\Omega_e = \mathbb{R}^n \setminus \overline{\Omega}$  its exterior. The "measurements" are done in the (possibly disjoint) subsets  $W_1, W_2 \subset \Omega_e$  of the exterior.

The basic tools in proving uniqueness for fractional Calderón problems are the unique continuation property and fractional Poincaré inequality for the fractional Laplacian  $(-\Delta)^s$ . We already saw the importance of unique continuation of fractional Laplacians in partial data problems of scalar and vector fields. In fractional Calderón problems the unique continuation of  $(-\Delta)^s$ implies Runge approximation: one can approximate functions in certain Sobolev spaces arbitrarily well by solutions of the fractional equation under study (see section 3.2). The fractional Poincaré inequality is a norm estimate involving the  $L^2$ -norms of a function and its fractional Laplacian, and it is an important inequality in proving well-posedness for the forward problem (the coercivity of the bilinear form). These basic tools (unique continuation and Poincaré inequality) were proved in [54] in the case  $s \in (0, 1)$ .

In the article [B] we study the higher order fractional Calderón problem for equation (14) when  $s \in (0, \infty) \setminus \mathbb{Z}$ . We prove higher order unique continuation result for fractional Laplacians: if  $(-\Delta)^s u|_V = u|_V = 0$  for some nonempty open set  $V \subset \mathbb{R}^n$  and  $s \in (-n/2, \infty) \setminus \mathbb{Z}$ , then u = 0. This generalizes the result proved in [54]. We also prove higher order fractional Poincaré inequality for  $s \in (0,\infty) \setminus \mathbb{Z}$  which says that the L<sup>2</sup>-norm of u can be bounded from above by the  $L^2$ -norm of  $(-\Delta)^s u$ . We provide five possible proofs for the Poincaré inequality and some of the proofs also give information about the constant in the inequality. Using the Poincaré inequality we prove wellposedness of the forward problem, and unique continuation of  $(-\Delta)^s$  implies Runge approximation for equation (14). Using Runge approximation and the so-called Alessandrini identity (see section 3.2) for suitable test functions we prove uniqueness for the inverse problem: if  $W_1, W_2 \subset \Omega_e$  are some open sets such that the DN maps satisfy  $\Lambda_{q_1} f = \Lambda_{q_2} f$  in  $W_2$  for all exterior values  $f \in C_c^{\infty}(W_1)$ , then  $q_1 = q_2$  in  $\Omega$ . This is done for certain singular potentials q which can be viewed as Sobolev multipliers. We also study the magnetic counterpart of equation (14) (the higher order fractional magnetic Schrödinger equation) in the article [B] and prove uniqueness (up to a gauge) under certain assumptions on the electric and magnetic potentials, generalizing the results in [30] to higher order cases.

The article [D] is a continuation of the article [B] in the sense that we replace the potential q in equation (14) with a general lower order local perturbation P(x, D). In the article [D] we study the fractional Calderón problem for equation (15) when  $s \in (0, \infty) \setminus \mathbb{Z}$  and m < 2s. We consider two different classes of coefficients  $a_{\alpha}$  of the partial differential operator P(x, D): coefficients which belong to certain  $L^{\infty}$ -Bessel potential spaces, and coefficients which are certain Sobolev multipliers. The same tools that we develop in the article [B], i.e. the higher order unique continuation property and fractional Poincaré inequality for  $(-\Delta)^s$ , are applicable in proving uniqueness in the article [D]. In addition to the Poincaré inequality we also need the Kato-Ponce inequality in proving well-posedness of the forward problem. The Kato-Ponce inequality is a fractional Leibnitz rule in terms of  $L^p$ -norms [56, 58, 76]. As in the article [B], the unique continuation of  $(-\Delta)^s$  implies Runge approximation for equation (15). Using the Runge approximation and the corresponding Alessandrini identity for suitable test functions we prove uniqueness for the inverse problem: if  $W_1, W_2 \subset \Omega_e$  are some open sets such that the DN maps satisfy  $\Lambda_{P_1} f = \Lambda_{P_2} f$  in  $W_2$  for all exterior values  $f \in C_c^{\infty}(W_1)$ , then  $P_1 = P_2$  in  $\Omega$ . This uniqueness result is shown for both classes of coefficients  $a_{\alpha}$ , i.e. coefficients with bounded fractional derivatives and coefficients which are Sobolev multipliers.

3.1. Notation. We use the same notation that we introduced in section 2.1, but we also introduce some additional notation. We follow the notation conventions of the references [11, 23, 101, 102, 106, 156].

If  $1 \leq p \leq \infty$ , we define the fractional  $L^p$ -Bessel potential space of order  $r \in \mathbb{R}$  as

$$H^{r,p}(\mathbb{R}^n) = \{ u \in \mathscr{S}'(\mathbb{R}^n) : \mathcal{F}^{-1}(\langle \cdot \rangle^r \hat{u}) \in L^p(\mathbb{R}^n) \}$$

and we equip it with the norm

$$\left\|u\right\|_{H^{r,p}(\mathbb{R}^n)} = \left\|\mathcal{F}^{-1}(\langle\cdot\rangle^r \hat{u})\right\|_{L^p(\mathbb{R}^n)}.$$

These spaces are nested, i.e.  $H^{r,p}(\mathbb{R}^n) \hookrightarrow H^{t,p}(\mathbb{R}^n)$  continuously when  $r \geq t$ . We see that  $H^{0,p}(\mathbb{R}^n) = L^p(\mathbb{R}^n)$ . If  $\Omega \subset \mathbb{R}^n$  is an open set, then we define the spaces  $H^{r,p}(\Omega)$  as restrictions

$$H^{r,p}(\Omega) = \{ u |_{\Omega} : u \in H^{r,p}(\mathbb{R}^n) \}$$

and we use the quotient norm

$$||w||_{H^{r,p}(\Omega)} = \inf\{||u||_{H^{r,p}(\mathbb{R}^n)} : u \in H^{r,p}(\mathbb{R}^n) \text{ such that } u|_{\Omega} = w\}.$$

It follows that the inclusions  $H^{r,p}(\Omega) \hookrightarrow H^{t,p}(\Omega)$  are continuous when  $r \geq t$ . The spaces  $H^{r,p}(\Omega)$  are not to be confused with the Sobolev-Slobodeckij spaces  $W^{r,p}(\Omega)$  which are defined by using weak derivatives of  $L^p$ -functions and which in general are different from the Bessel potential spaces we have introduced [44]. If  $r \geq 0$  and p = 2, then  $H^{r,2}(\mathbb{R}^n) = W^{r,2}(\mathbb{R}^n)$  and  $H^{r,2}(\Omega) = W^{r,2}(\Omega)$  when  $\Omega$  is a Lipschitz domain.

The following spaces are special cases of the above Bessel potential spaces

$$\begin{split} H^{r,p}_F(\mathbb{R}^n) &= \{ u \in H^{r,p}(\mathbb{R}^n) : \operatorname{spt}(u) \subset F \} \\ \widetilde{H}^{r,p}(\Omega) &= \text{ closure of } C^{\infty}_c(\Omega) \text{ with respect to the norm } \|\cdot\|_{H^{r,p}(\mathbb{R}^n)} \\ H^{r,p}_0(\Omega) &= \text{ closure of } C^{\infty}_c(\Omega) \text{ with respect to the norm } \|\cdot\|_{H^{r,p}(\Omega)} \end{split}$$

where  $F \subset \mathbb{R}^n$  is some closed set. Observe that  $\widetilde{H}^{r,p}(\Omega) \subset H^{r,p}(\mathbb{R}^n)$  and  $H_0^{r,p}(\Omega) \subset H^{r,p}(\Omega)$ . One also sees that  $\widetilde{H}^{r,p}(\Omega) \subset H_0^{r,p}(\Omega)$  and  $\widetilde{H}^{r,p}(\Omega) \subset H_{\overline{\Omega}}^{r,p}(\Omega)$ . When p = 2, we simply write  $H^{r,2}(\mathbb{R}^n) = H^r(\mathbb{R}^n)$ ,  $H^{r,2}(\Omega) = H^r(\Omega)$  and so on. It follows that  $(\widetilde{H}^r(\Omega))^* = H^{-r}(\Omega)$  and  $(H^r(\Omega))^* = \widetilde{H}^{-r}(\Omega)$  for any open set  $\Omega \subset \mathbb{R}^n$  and  $r \in \mathbb{R}$ . If in addition  $\Omega$  is a Lipschitz domain and  $r \geq 0$  such that  $r \notin \{\frac{1}{2}, \frac{3}{2}, \frac{5}{2} \dots\}$ , then  $\widetilde{H}^r(\Omega) = H_0^r(\Omega)$ .

We define the space of Sobolev multipliers  $M(H^r \to H^t) \subset \mathcal{D}'(\mathbb{R}^n)$  by saying that the distribution  $f \in \mathcal{D}'(\mathbb{R}^n)$  belongs to  $M(H^r \to H^t)$  if the multiplier norm

 $||f||_{r,t} = \sup\{|\langle f, uv\rangle| : u, v \in C_c^{\infty}(\mathbb{R}^n), ||u||_{H^r(\mathbb{R}^n)} = ||v||_{H^{-t}(\mathbb{R}^n)} = 1\}$ 

is finite. We let  $M_0(H^r \to H^t)$  be the closure of  $C_c^{\infty}(\mathbb{R}^n)$  in  $M(H^r \to H^t)$ with respect to the norm  $\|\cdot\|_{r,t}$ . The elements of the space  $M(H^r \to H^t)$ are called Sobolev multipliers since each  $f \in M(H^r \to H^t)$  induces a map  $m_f \colon H^r(\mathbb{R}^n) \to H^t(\mathbb{R}^n)$  defined as

$$\langle m_f(u), v \rangle = \langle f, uv \rangle$$

for all  $u \in H^r(\mathbb{R}^n)$  and  $v \in H^{-t}(\mathbb{R}^n)$ . As a special case of multipliers we write  $Z^{-s}(\mathbb{R}^n) = M(H^s \to H^{-s})$  and  $Z_0^{-s}(\mathbb{R}^n) = M_0(H^s \to H^{-s})$  whose elements we also call singular potentials.

We say that 0 is not a Dirichlet eigenvalue of the operator  $(-\Delta)^s + q$ , if the following condition holds:

(16) If  $u \in H^s(\mathbb{R}^n)$  solves  $((-\Delta)^s + q)u = 0$  in  $\Omega$  and  $u|_{\Omega_e} = 0$ , then u = 0.

Analogously, we say that 0 is not a Dirichlet eigenvalue of the operator  $(-\Delta)^s + P(x,D)$  if condition (16) holds when q is replaced with the partial differential operator P(x,D). When the forward problem for equation (14) is well-posed, we can define the DN map  $\Lambda_q \colon H^s(\Omega_e) \to (H^s(\Omega_e))^*$  as  $\langle \Lambda_q f_1, f_2 \rangle = B_q(u_{f_1}, f_2)$  where  $B_q(\cdot, \cdot)$  is the bilinear form associated to equation (14) and  $u_{f_1}$  is the unique solution to equation (14) with exterior value  $u|_{\Omega_e} = f_1$ . The DN map  $\Lambda_P$  for equation (15) is defined similarly.

3.2. Main results. One of the main theorems of the article [B] is the following unique continuation property of fractional Laplacians.

**Theorem 3.1** ([B, Theorem 1.1]). Let  $n \ge 1$ ,  $s \in (-n/4, \infty) \setminus \mathbb{Z}$  and  $u \in H^r(\mathbb{R}^n)$  where  $r \in \mathbb{R}$ . If  $(-\Delta)^s u|_V = 0$  and  $u|_V = 0$  for some nonempty open set  $V \subset \mathbb{R}^n$ , then u = 0. The claim holds also for  $s \in (-n/2, -n/4] \setminus \mathbb{Z}$  if  $u \in H^{r,1}(\mathbb{R}^n)$  or  $u \in \mathcal{O}'_C(\mathbb{R}^n)$ .

Theorem 3.1 is proved by reducing the claim to the case  $s \in (0, 1)$  and using the unique continuation result proved in [54]. The reduction can be done by using the simple relation  $(-\Delta)^k(-\Delta)^s = (-\Delta)^s(-\Delta)^k = (-\Delta)^{s+k}$ when  $k \in \mathbb{N}$  and  $s \in (-n/2, \infty) \setminus \mathbb{Z}$ . The assumptions on s in theorem 3.1 are put so that  $(-\Delta)^s$  is a non-local operator and  $(-\Delta)^s u$  is well-defined as a tempered distribution. We also prove in the article [B] many other versions of the unique continuation of  $(-\Delta)^s$  in different Sobolev spaces, including homogeneous Sobolev spaces and certain Bessel potential spaces.

The next theorem of the article [B] is called the (fractional) Poincaré inequality. It has an essential role in proving well-posedness for the forward problems of equations (14) and (15).

**Theorem 3.2** ([B, Theorem 1.2]). Let  $n \ge 1$ ,  $s \ge t \ge 0$ ,  $K \subset \mathbb{R}^n$  a compact set and  $u \in H^s_K(\mathbb{R}^n)$ . There exists a constant c = c(n, K, s) > 0 such that

(17) 
$$\left\| (-\Delta)^{t/2} u \right\|_{L^2(\mathbb{R}^n)} \le c \left\| (-\Delta)^{s/2} u \right\|_{L^2(\mathbb{R}^n)}$$

24

In well-posedness we only need the cases t = 0 and  $s \in (0, \infty) \setminus \mathbb{Z}$  of theorem 3.2. Note that the inequality (17) holds for all exponents  $s \ge t \ge 0$ , not just fractional ones. The interpretation of theorem 3.2 is that the norms of lower order derivatives of u are bounded from above by the norms of higher order derivatives of u. When t = 0 and s = 1, then the inequality (17) reduces to the classical Poincaré inequality.

We provide five different proofs for theorem 3.2. Two of the simplest proofs are based on Fourier analysis: the first uses splitting of frequencies on the Fourier side and the second uses uncertainty inequalities proved in [48]. Two other proofs are based on a reduction argument similar to what we did in proving the unique continuation of higher order fractional Laplacians. The fifth proof considers the case  $s \ge 1$  and it uses interpolation in homogeneous Sobolev spaces and the classical Poincaré inequality. This proof also gives an explicit constant for the inequality: if  $s \ge 1$  and  $u \in \tilde{H}^s(\Omega)$ , then in theorem 3.2 we can take  $c = C^{s-t}$  where C is the classical Poincaré constant. This is expected since on the left-hand side of equation (17) we take t derivatives and on the right-hand side we take s derivatives.

The next theorem of the article [B] gives uniqueness for the higher order fractional Schrödinger equation with a singular potential.

**Theorem 3.3** ([B, Theorem 1.3]). Let  $n \ge 1$ ,  $\Omega \subset \mathbb{R}^n$  a bounded open set,  $s \in (0,\infty) \setminus \mathbb{Z}$ , and  $q_1, q_2 \in Z_0^{-s}(\mathbb{R}^n)$  such that 0 is not a Dirichlet eigenvalue of the operators  $(-\Delta)^s + q_i$ . Let  $W_1, W_2 \subset \mathbb{R}^n \setminus \overline{\Omega}$  be open sets. If the DN maps for the equations  $(-\Delta)^s u + m_{q_i}(u) = 0$  in  $\Omega$  satisfy  $\Lambda_{q_1} f|_{W_2} = \Lambda_{q_2} f|_{W_2}$ for all  $f \in C_c^\infty(W_1)$ , then  $q_1|_{\Omega} = q_2|_{\Omega}$ .

The proof of theorem 3.3 follows from Runge approximation for equation (14) and choosing suitable test functions in the Alessandrini identity. The Runge approximation says that we can approximate functions in  $\tilde{H}^s(\Omega)$ arbitrarily well by solutions of the fractional Schrödinger equation (14), and it can be proved by using the unique continuation of  $(-\Delta)^s$  in theorem 3.1 and the well-posedness of the forward problem. The Alessandrini identity is an integral identity showing how the DN maps  $\Lambda_{q_i}$  and the corresponding potentials  $q_i$  are related in terms of exterior values f and solutions  $u_f$  of equation (14).

The article [D] generalizes the higher order fractional Schrödinger equation studied in the article [B] to include more general lower order local perturbations. The following two theorems are the main results of the article [D].

**Theorem 3.4** ([D, Theorem 1.1]). Let  $\Omega \subset \mathbb{R}^n$  be a bounded open set where  $n \geq 1$ . Let  $s \in (0, \infty) \setminus \mathbb{Z}$  and  $m \in \mathbb{N}$  be such that 2s > m. Let

$$P_j = \sum_{|\alpha| \le m} a_{j,\alpha} D^{\alpha}, \quad j = 1, 2,$$

be partial differential operators of order m where  $a_{j,\alpha} \in M_0(H^{s-|\alpha|} \to H^{-s})$ such that 0 is not a Dirichlet eigenvalue of the operators  $(-\Delta)^s + P_j$ . Given any two open sets  $W_1, W_2 \subset \mathbb{R}^n \setminus \overline{\Omega}$ , suppose that the DN maps  $\Lambda_{P_j}$  for the equations  $((-\Delta)^s + P_j)u = 0$  in  $\Omega$  satisfy

$$\Lambda_{P_1} f|_{W_2} = \Lambda_{P_2} f|_{W_2}$$

for all  $f \in C_c^{\infty}(W_1)$ . Then  $P_1|_{\Omega} = P_2|_{\Omega}$ .

**Theorem 3.5** ([D, Theorem 1.2]). Let  $\Omega \subset \mathbb{R}^n$  be a bounded Lipschitz domain where  $n \geq 1$ . Let  $s \in (0, \infty) \setminus \mathbb{Z}$  and  $m \in \mathbb{N}$  be such that 2s > m. Let

$$P_j(x,D) = \sum_{|\alpha| \le m} a_{j,\alpha}(x) D^{\alpha}, \quad j = 1, 2,$$

be partial differential operators of order m with coefficients  $a_{j,\alpha} \in H^{r_{\alpha},\infty}(\Omega)$ where

$$r_{\alpha} = \begin{cases} 0 & if \qquad |\alpha| - s < 0, \\ |\alpha| - s + \delta & if \quad |\alpha| - s \in \{1/2, 3/2, \ldots\}, \\ |\alpha| - s & if \qquad otherwise \end{cases}$$

for any fixed  $\delta > 0$  and assume that 0 is not a Dirichlet eigenvalue of the operators  $(-\Delta)^s + P_j(x, D)$ . Given any two open sets  $W_1, W_2 \subset \mathbb{R}^n \setminus \overline{\Omega}$ , suppose that the DN maps  $\Lambda_{P_j}$  for the equations  $((-\Delta)^s + P_j(x, D))u = 0$  in  $\Omega$  satisfy

$$\Lambda_{P_1} f|_{W_2} = \Lambda_{P_2} f|_{W_2}$$
  
for all  $f \in C_c^{\infty}(W_1)$ . Then  $P_1(x, D) = P_2(x, D)$ .

It is not known whether the spaces  $M_0(H^{s-|\alpha|} \to H^{-s})$  and  $H^{r_\alpha,\infty}(\Omega)$  are contained in each other. If this is not the case, then theorems 3.4 and 3.5 are distinct and neither claim implies the other. In theorem 3.4 we consider multipliers which can be approximated in the multiplier norm by smooth compactly supported functions and for this reason we do not need to assume anything about the boundary of  $\Omega$ . In theorem 3.5 we have put some conditions on  $\partial\Omega$  and for the exponent  $r_\alpha$  which are needed in proving wellposedness in the case of coefficients with bounded fractional derivatives. The assumptions that 0 is not a Dirichlet eigenvalue and 2s > m (i.e. we consider perturbations to  $(-\Delta)^s$ ) are also crucial in both theorems when proving wellposedness of the forward problem. It follows that  $M(H^{s-|\alpha|} \to H^{-s}) = \{0\}$ if  $s - |\alpha| < -s$ . Partly because of this reason theorem 3.4 is formulated only for 2s > m since the multiplier coefficients for higher order derivatives are zero, i.e.  $a_{\alpha} = 0$  for all  $|\alpha| > 2s$ .

Theorems 3.4 and 3.5 are proved in the same way, even though the exact details are a little bit different. The proofs follow the same ideas as in the article [B] where we proved uniqueness for zeroth order perturbations, and we see that theorem 3.3 is in fact a special case of theorem 3.4. The

well-posedness of the forward problem is proved by using the higher order fractional Poincaré inequality in theorem 3.2 and interpolation inequality in non-homogeneous Sobolev spaces. In the case of coefficients with bounded fractional derivatives we also need the Kato–Ponce inequality in proving well-posedness. The higher order unique continuation of  $(-\Delta)^s$  in theorem 3.1 together with well-posedness implies Runge approximation for equation (15) (and for the adjoint equation of (15)): one can approximate functions in  $\tilde{H}^s(\Omega)$  arbitrarily well by solutions of equation (15). We can prove uniqueness for the inverse problem by using the Runge approximation and suitable test functions in the Alessandrini identity which gives the relation between the DN maps  $\Lambda_{P_i}$  and the partial differential operators  $P_i$  in terms of exterior values f and solutions  $u_f$  of equation (15) (and the adjoint equation of (15)). It is important to notice that in theorems 3.4 and 3.5 we recover the coefficients  $a_{\alpha}$  uniquely and there is no gauge in contrast to the perturbed local Schrödinger equation [71, 111, 138].

Note that even though we consider partial differential operators in theorems 3.4 and 3.5, the results apply for more general local linear operators. In fact, Peetre's theorem implies that any local linear operator  $L: C_c^{\infty}(\Omega) \to C_c^{\infty}(\Omega)$  which satisfies  $\operatorname{spt}(Lf) \subset \operatorname{spt}(f)$  for all  $f \in C_c^{\infty}(\Omega)$  can be identified with a partial differential operator [105, 121]. Hence our results hold for any such local operator satisfying the assumptions in theorems 3.4 and 3.5.

# 4. Travel time tomography on Riemannian and Finsler Manifolds: [E, F]

In the articles [E, F] we study the travel time tomography or boundary rigidity problem and its linearized versions on Riemannian manifolds and more general Finsler manifolds where the fiberwise inner product depends on direction. The basic idea of the boundary rigidity problem is illustrated in figure 4. Suppose we have two Finsler norms  $F_1$  and  $F_2$  (which can be for example two Riemannian metrics) on a manifold M with boundary  $\partial M$ . We assume that between any two boundary points  $x, x' \in \partial M$  there is unique geodesic  $\gamma_i$  of the Finsler norm  $F_i$  going from x to x'. The length of the geodesic  $\gamma_i$  with respect to  $F_i$  is denoted by  $L_{F_i}(\gamma_i)$  and it gives the (not necessarily symmetric) distance from  $x \in \partial M$  to  $x' \in \partial M$ . The boundary rigidity problem is the following: if the Finsler norms  $F_1$  and  $F_2$  give the same distances between all boundary points  $x, x' \in \partial M$ , does it follow that  $F_1 = F_2$  up to a natural gauge?

If the Finsler norms  $F_i$  are induced by Riemannian metrics  $g_i$  (the fiberwise inner product does not depend on direction), then the natural gauge is a boundary preserving diffeomorphism: if  $g_2 = \Psi^* g_1$  where  $\Psi \colon M \to M$  is a diffeomorphism such that  $\Psi|_{\partial M} = \text{Id}$ , then  $g_1$  and  $g_2$  give the same boundary distances. For a special class of non-reversible Finsler norms called Randers metrics the gauge is similar: if  $F_1 = F_g + \beta$  where  $F_g$  is a Finsler norm



FIGURE 4. An illustration of the boundary rigidity problem on Finsler manifolds (M, F). Here  $x, x' \in \partial M$  are two boundary points,  $\gamma_1$  and  $\gamma_2$  are the unique geodesics of the Finsler norms  $F_1$  and  $F_2$  connecting x to x', and  $L_{F_i}(\gamma_i)$  denotes the length of the geodesic  $\gamma_i$  with respect to  $F_i$  (adapted from [F, Figure 1]).

induced by the Riemannian metric g and  $\beta$  is a 1-form whose norm with respect to g is small, then  $F_1$  and  $F_2 = \Psi^* F_1 + d\phi$  give the same boundary distances where  $\Psi \colon M \to M$  is a diffeomorphism which is identity on the boundary and  $\phi$  is a scalar field vanishing on the boundary (and  $d\phi$  is considered as a small perturbation to  $\Psi^* F_1$ ).

On Riemannian manifolds (M, g) the linearization of the boundary rigidity problem leads to the geodesic ray transform  $\mathcal{I}_m$  of symmetric (covariant) m-tensor fields [143]: if h and h' are two symmetric m-tensor fields such that  $\mathcal{I}_m h = \mathcal{I}_m h'$ , does it follow that h = h' up to a natural gauge? When  $m \geq 1$ , the gauge is given by the derivative of a lower order tensor field: if  $h' = h + \sigma \nabla v$  where h is symmetric m-tensor field, v is an m - 1-tensor field vanishing on the boundary (or at infinity) and  $\sigma \nabla$  is the symmetrized covariant derivative, then  $\mathcal{I}_m h = \mathcal{I}_m h'$ . Since the problem is linear ( $\mathcal{I}_m$  is a linear operator) it is enough to study the kernel of  $\mathcal{I}_m$ : if h is a symmetric m-tensor field such that  $\mathcal{I}_m h = 0$ , does it follow that  $h = \sigma \nabla v$  where v is an m - 1-tensor field vanishing on the boundary (or at infinity)? If this is true for all sufficiently regular symmetric m-tensor fields, we say that  $\mathcal{I}_m$  is solenoidally injective (or s-injective).

In the article [E] we study the mixed ray transform and more general mixing ray transforms on Riemannian manifolds. These integral transforms are generalizations of the geodesic ray transform and they arise in the linearization of the elastic travel time tomography problem [39, 40, 143]. The main focus in the article [E] is on the algebraic properties of mixing ray transforms and decompositions of tensor fields with respect to these transforms. We have various corollaries of a main idea how to study the kernel characterization and solenoidal injectivity of the mixing ray transforms using correct notion of symmetry and reduction.

The mixing ray transform of *m*-tensor fields  $(m \ge 1)$  is defined as the composition  $\mathcal{I}_A h = (\mathcal{I}_m \circ A)h$  where  $\mathcal{I}_m$  is the geodesic ray transform of *m*-tensor fields and A is a smooth linear invertible map on *m*-tensor fields (see section 4.1). If A is the identity map, then  $\mathcal{I}_A$  reduces to the geodesic ray transform  $\mathcal{I}_m$ . One can think that the transform  $\mathcal{I}_A$  first rotates the tensor field h and then takes the geodesic ray transform of the rotated *m*-tensor field Ah.

The mixing ray transforms are matrix-weighted geodesic ray transforms and they have a different kind of kernel than the geodesic ray transform. We prove in the article [E] that every *m*-tensor field *h* can be written as the direct sum  $h = \hat{\sigma}_A h + (h - \hat{\sigma}_A h)$  where  $\hat{\sigma}_A$  is the symmetrization map with respect to the transform  $\mathcal{I}_A$  (see section 4.1) and  $h - \hat{\sigma}_A h \in \text{Ker}(\mathcal{I}_A)$ . Here  $\hat{\sigma}_A h$  is the "symmetric part" of *h* and  $h - \hat{\sigma}_A h$  is the "trivial part" of *h* from the point of view of the transform  $\mathcal{I}_A$ . We show that if  $\mathcal{I}_m$  is s-injective on symmetric *m*-tensor fields and  $\mathcal{I}_A h = 0$ , then  $\hat{\sigma}_A h = \hat{\sigma}_A \nabla^A v$ for some m - 1-tensor field *v* vanishing on the boundary (or at infinity) where  $\nabla^A = A^{-1} \circ \nabla$  is the weighted covariant derivative associated to  $\mathcal{I}_A$ . This property is referred as the solenoidal injectivity of  $\mathcal{I}_A$  and it allows us to write the kernel of  $\mathcal{I}_A$  as the direct sum  $\text{Ker}(\mathcal{I}_A) = \text{Im}(\mathcal{H}) \oplus \text{Im}(\hat{\sigma}_A \nabla^A)$ where  $\mathcal{H} = \text{Id} - \hat{\sigma}_A$  is the projection onto the "trivial part" of  $\text{Ker}(\mathcal{I}_A)$  (see sections 4.1 and 4.2).

In addition to solenoidal injectivity results we prove in the article [E] numerous corollaries of the algebraic approach to mixing ray transforms and related transforms such as the mixed ray transform and the light ray transform. For example, we show that previous results for the light ray transform on Lorentzian manifolds and the mixed ray transform on simple Riemannian manifolds in [40, 47] can be seen as solenoidal injectivity results when we have a correct notion of symmetry. We also prove some stability results for the mixed ray transform, and show that the geodesic ray transform and the transverse ray transform together determine 1-forms uniquely on certain two-dimensional compact and non-compact manifolds.

In the article [F] we study the boundary rigidity problem for certain non-reversible Finsler norms called Randers metrics. Finsler norms are nonnegative functions on the tangent bundle  $F: TM \to [0, \infty)$  so that for every  $x \in M$  the map  $y \mapsto F(x, y)$  is a positively homogeneous norm in  $T_xM$ . The Finsler norm F is reversible, if F(x, -y) = F(x, y) for all  $x \in M$ and  $y \in T_xM$ . In this case the map  $y \mapsto F(x, y)$  defines a norm in  $T_xM$ . In general, the distance function given by F is not necessarily symmetric
in contrast to the Riemannian distance function. Finsler norms induce a fiberwise inner product which depends not only on position but also on direction. Riemannian metrics are a special case of reversible Finsler norms where the inner product does not depend on direction.

Randers metrics are Finsler norms of the form  $F = F_g + \beta$  where  $F_g$  is a Finsler norm induced by the Riemannian metric g and  $\beta$  is a 1-form whose norm with respect to g is small enough. Randers metrics are non-reversible since F(x, -y) = F(x, y) for all  $x \in M$  and  $y \in T_x M$  if and only if  $\beta \equiv 0$ . Randers metrics arise naturally in Zermelo's navigation problem [10, 146]. Roughly saying, Zermelo's problem asks what is the shortest path in time for a moving object to travel from point A to point B when an external force field is acting on the object. Basic example is a ship which is sailing on a sea under the influence of wind or current.

In the article [F] we prove two boundary rigidity results. If F is a Finsler norm and  $x, x' \in \partial M$ , denote by  $d_F(x, x')$  the (non-symmetric) geodesic distance from x to x' (see section 4.1). The first theorem is the following: if  $F_1$  and  $F_2$  are Finsler norms of the form  $F_i = F_{r,i} + \beta_i$  where  $F_{r,i}$  is a reversible Finsler norm and  $\beta_i$  is a closed 1-form  $(d\beta_i = 0)$  such that  $d_{F_1}(x,x') = d_{F_2}(x,x')$  for all  $x,x' \in \partial M$ , then  $\beta_2 = \beta_1 + \mathrm{d}\phi$  where  $\phi$ is a scalar field vanishing on the boundary and  $d_{F_{r,1}}(x, x') = d_{F_{r,2}}(x, x')$ for all  $x, x' \in \partial M$ . This is done by using projective equivalence of the Finsler norms  $F_i$  and  $F_{r,i}$ : since the 1-form  $\beta_i$  is closed  $F_i$  and  $F_{r,i}$  have the same geodesics as point sets, and the geodesics of  $F_i$  remain geodesics (as point sets) if their orientation is reversed. The second theorem is a boundary rigidity result for Randers metrics and it is a corollary of the first theorem: if  $F_1 = F_{g_1} + \beta_1$  and  $F_2 = F_{g_2} + \beta_2$  are Randers metrics where  $g_1$  and  $g_2$  are boundary rigid Riemannian metrics,  $\beta_i$  is a closed 1-form and  $d_{F_1}(x, x') = d_{F_2}(x, x')$  for all  $x, x' \in \partial M$ , then  $F_2 = \Psi^* F_1 + d\phi$  where  $\phi$  is a scalar field vanishing on the boundary and  $\Psi \colon M \to M$  is a diffemorphism which is identity on the boundary. In other words, the equality of the boundary distances implies that the Randers metrics  $F_1$  and  $F_2$  are equal up to the natural gauge. Using Zermelo's navigation problem we provide an application of the second theorem to seismology where the seismic wave propagates in a moving medium.

4.1. Notation. Let us first go through the notation used in the article [E]. We follow the notation conventions of the references [92, 93, 95, 120, 143]. We will use the Einstein summation convention so that every repeated index appearing both as a subscript and superscript is implicitly summed over.

Let M be an n-dimensional smooth manifold where  $n \geq 2$ . We usually assume that M is compact and has a boundary  $\partial M$  or that M is non-compact without boundary. If (M, g) is a Riemannian manifold, we denote by K(x)the Gaussian curvature at  $x \in M$ . We say that a compact Riemannian manifold (M, g) with boundary is simple (or that the Riemannian metric g is simple) if it is non-trapping (maximal geodesics have finite length), geodesics have no conjugate points and the boundary  $\partial M$  is strictly convex with respect to g (the second fundamental form on  $\partial M$  is positive definite). A compact simple manifold is always diffeomorphic to a ball. We say that a non-compact manifold (M, g) without boundary is a Cartan–Hadamard manifold if it is simply connected, complete and its sectional curvature is nonpositive. Cartan–Hadamard manifolds are diffeomorphic to  $\mathbb{R}^n$  and basic examples are the Euclidean space and hyperbolic spaces.

Let  $m \geq 1$ . We denote by  $\mathfrak{X}(T_m M)$  the space of all covariant *m*-tensor fields and  $S_m M \subset \mathfrak{X}(T_m M)$  is the space of symmetric covariant *m*-tensor fields. The notations  $C^{\infty}(T_m M) := C^{\infty}(\mathfrak{X}(T_m M))$  and  $C^{\infty}(S_m M)$  mean that the corresponding tensor fields are smooth. The pointwise norm of a covariant *m*-tensor field *h* is  $|h|_{g_x} = \sqrt{g_x(h,h)}$  where  $g_x(\cdot, \cdot)$  is the fiberwise inner product of *m*-tensor fields. We define the following sets of polynomially and exponentially decaying tensor fields which are mainly used on Cartan– Hadamard manifolds

$$\begin{split} E_{\eta}(T_m M) &= \{h \in C^1(T_m M) : \\ & |h|_{g_x} \leq Ce^{-\eta d(x,o)} \text{ for some } C > 0\}, \\ E_{\eta}^1(T_m M) &= \{h \in C^1(T_m M) : \\ & |h|_{g_x} + |\nabla h|_{g_x} \leq Ce^{-\eta d(x,o)} \text{ for some } C > 0\}, \\ P_{\eta}(T_m M) &= \{h \in C^1(T_m M) : \\ & |h|_{g_x} \leq C(1 + d(x,o))^{-\eta} \text{ for some } C > 0\}, \\ P_{\eta}^1(T_m M) &= \{h \in C^1(T_m M) : \\ & |h|_{g_x} \leq C(1 + d(x,o))^{-\eta} \text{ and} \\ & |\nabla h|_{g_x} \leq C(1 + d(x,o))^{-\eta-1} \text{ for some } C > 0\} \end{split}$$

where  $o \in M$  is a fixed point and  $\eta > 0$ .

Let (M, g) be a non-trapping compact Riemannian manifold with boundary  $\partial M$ . Let  $x \in \partial M$  and  $\xi \in T_x M$  be an inward-pointing unit vector. Denote by  $\gamma_{x,\xi}$  the geodesic starting at x in the direction  $\xi$  and let  $\tau(x,\xi)$ be the first time when the geodesic hits the boundary again. The geodesic ray transform of a sufficiently regular *m*-tensor field h is defined as

$$\mathcal{I}_m h(x,\xi) = \int_0^{\tau(x,\xi)} h_{i_1\dots i_m}(\gamma_{x,\xi}(t)) \dot{\gamma}_{x,\xi}^{i_1}(t) \cdots \dot{\gamma}_{x,\xi}^{i_m}(t) \mathrm{d}t.$$

Similarly, if (M, g) is a Cartan–Hadamard manifold and  $x \in M$  and  $\xi \in T_x M$  has unit length, then we define

$$\mathcal{I}_m h(x,\xi) = \int_{-\infty}^{\infty} h_{i_1\dots i_m}(\gamma_{x,\xi}(t)) \dot{\gamma}_{x,\xi}^{i_1}(t) \cdots \dot{\gamma}_{x,\xi}^{i_m}(t) \mathrm{d}t$$

whenever the m-tensor field h decays rapidly enough at infinity. By completeness geodesics are defined on all times on Cartan–Hadamard manifolds.

We define  $A: C^{\infty}(T_m M) \to C^{\infty}(T_m M)$  as a smooth linear invertible map on *m*-tensor fields which operates as

$$(Ah)_x(\xi_1,\ldots,\xi_m) = h_x(A_1(x)\xi_1,\ldots,A_m(x)\xi_m)$$

where  $\xi_i \in T_x M$  and each  $A_i(x)$  is a linear bijection in  $T_x M$ . Such map A is called a mixing of degree  $m \ge 1$ . If A is a mixing of degree m, we define the mixing ray transform  $\mathcal{I}_A$  by setting  $\mathcal{I}_A = \mathcal{I}_m \circ A$  where  $\mathcal{I}_m$  is the geodesic ray transform of m-tensor fields. On orientable two-dimensional manifolds an important special case of the mixing ray transforms is the mixed ray transform  $L_{k,l} = \mathcal{I}_m \circ A_{k,l}$  where the components  $A_i$  of the mixing  $A_{k,l}$ satisfy  $A_i = \star$  when  $i = 1, \ldots, k$  and  $A_i = \text{Id}$  when  $i = k + 1, \ldots, k + l = m$ . Here  $\star$  is the Hodge star operating on 1-forms (and hence on vector fields via the musical isomorphisms) and on orientable two-dimensional manifolds it corresponds to rotation by 90 degrees counterclockwise.

If A is a mixing of degree m, we define the generalized symmetrization operator  $\hat{\sigma}_A = A^{-1} \circ \sigma \circ A$  where  $\sigma$  is the usual symmetrization of tensor fields. Then  $\hat{\sigma}_A$  is a projection onto  $A^{-1}(S_m M)$  and we have the direct decomposition  $h = \hat{\sigma}_A h + (h - \hat{\sigma}_A h)$  where  $\hat{\sigma}_A h \in A^{-1}(S_m M)$  and  $h - \hat{\sigma}_A h \in \text{Ker}(\mathcal{I}_A)$ . We denote by  $\nabla^A$  the weighted covariant derivative  $\nabla^A = A^{-1} \circ \nabla$ . We say that the mixing ray transform  $\mathcal{I}_A$  is s-injective on a compact Riemannian manifold (M, g) with boundary, if for every  $h \in C^{\infty}(T_m M)$  we have that  $\mathcal{I}_A h = 0$  if and only if  $\hat{\sigma}_A h = \hat{\sigma}_A \nabla^A v$  for some  $v \in C^{\infty}(S_{m-1}M)$ vanishing on the boundary.

Then we shortly introduce the additional notation used in the article [F]; these basic notions of Finsler geometry can be found in [3, 9, 24, 145].

Let  $F: TM \to [0, \infty)$  be a Finsler norm and denote by  $F_r$  a reversible Finsler norm, i.e.  $F_r(x, -y) = F_r(x, y)$  for all  $x \in M$  and  $y \in T_x M$ . If g is a Riemannian metric, then it defines a reversible Finsler norm  $F_g$  as  $F_g(x, y) = \sqrt{g_{ij}(x)y^iy^j}$ . We denote by  $\beta$  a smooth 1-form and say that  $\beta$  is closed, if  $d\beta = 0$  where d is the exterior derivative of differential forms. We define the dual norm of  $\beta$  as  $\|\beta\|_{F^*} = \sup_{x \in M} F^*(x, \beta_x)$  where  $F^*$  is the co-Finsler norm in  $T^*M$ . More specifically,  $F^*(x, \beta_x) = \sup_{y \in T_x M, F(x,y)=1} \beta_x(y)$ . If Fis a Finsler norm and  $\beta$  is a 1-form such that  $\|\beta\|_{F^*} < 1$ , then  $F + \beta$  also defines a Finsler norm.

We define admissible Finsler norms as follows: F is admissible, if for any two points  $x, x' \in \partial M$  there exists unique geodesic  $\gamma$  of F going from x to x'having finite length. When F is admissible, we define the map  $d_F(\cdot, \cdot) : \partial M \times$  $\partial M \to [0, \infty)$  as  $d_F(x, x') = L_F(\gamma)$  where  $L_F(\gamma)$  is the length of the geodesic  $\gamma$  with respect to F. In general the map  $d_F(\cdot, \cdot)$  is not symmetric. We say that the Riemannian metrics  $g_1$  and  $g_2$  on M are boundary rigid, if  $d_{g_1}(x, x') = d_{g_2}(x, x')$  for all  $x, x' \in \partial M$  if and only if  $g_2 = \Psi^* g_1$  where  $\Psi : M \to M$  is a diffeomorphism which is identity on the boundary.

4.2. Main results. In the article [E] we study linearized travel time tomography. We have numerous corollaries of the algebraic approach to mixing ray

32

transforms and here we only present the most important results considering solenoidal injectivity. The first result says that s-injectivity of one mixing ray transform implies s-injectivity for all mixing ray transforms.

**Theorem 4.1** ([E, Corollary 3.4]). Let  $m \ge 1$  and (M,g) be a compact Riemannian manifold with boundary so that the transform  $I_A$  is s-injective for some A of degree m. Then  $I_{\widetilde{A}}$  is s-injective for all  $\widetilde{A}$  of degree m.

Theorem 4.1 holds in all dimensions  $n \geq 2$  and it is proved by using the definition of s-injectivity and the properties of mixings A and the corresponding projections  $\hat{\sigma}_A$ . The following theorem is a special case of theorem 4.1 in two dimensions.

**Theorem 4.2** ([E, Corollary 4.1]). Let  $m \ge 1$ . Let (M,g) be a compact two-dimensional orientable Riemannian manifold with boundary such that the geodesic ray transform is s-injective on  $C^{\infty}(S_m M)$  and let  $h \in C^{\infty}(T_m M)$ . Then  $L_{k,l}h = 0$  if and only if  $\widehat{\sigma}_{A_{k,l}}h = \widehat{\sigma}_{A_{k,l}}\nabla^{A_{k,l}}v$  for some  $v \in C^{\infty}(S_{m-1}M)$  vanishing on the boundary  $\partial M$ .

S-injectivity of the geodesic ray transform is known for example on compact simple surfaces [119] and on simply connected compact surfaces with strictly convex boundary and non-positive sectional curvature [118, 143]. Hence we obtain many new s-injectivity results for the mixed ray transform in two dimensions using theorem 4.2. The assumption that (M, g) is a twodimensional orientable manifold is needed so that the mixed ray transform is well-defined, i.e. we can use the Hodge star  $\star$  to rotate vector fields.

The next theorem of the article [E] shows that s-injectivity holds for the mixed ray transform also on certain non-compact Cartan–Hadamard manifolds.

**Theorem 4.3** ([E, Corollary 4.2]). Let (M, g) be a two-dimensional Cartan– Hadamard manifold and let  $m \ge 1$ . The following claims are true:

- (a) Let  $-K_0 \leq K \leq 0$  for some  $K_0 > 0$  and  $h \in E^1_{\eta}(T_m M)$  for some  $\eta > \frac{3}{2}\sqrt{K_0}$ . Then  $L_{k,l}h = 0$  if and only if  $\widehat{\sigma}_{A_{k,l}}h = \widehat{\sigma}_{A_{k,l}}\nabla^{A_{k,l}}v$  for some  $v \in S_{m-1}M$  such that  $v \in E_{\eta-\epsilon}(T_{m-1}M)$  for all  $\epsilon > 0$ .
- (b) Let  $K \in P_{\kappa}(M)$  for some  $\kappa > 2$  and  $h \in P_{\eta}^{1}(T_{m}M)$  for some  $\eta > 2$ . Then  $L_{k,l}h = 0$  if and only if  $\widehat{\sigma}_{A_{k,l}}h = \widehat{\sigma}_{A_{k,l}}\nabla^{A_{k,l}}v$  for some  $v \in S_{m-1}M \cap P_{\eta-1}(T_{m-1}M)$ .

Theorem 4.3 follows from the corresponding s-injectivity result for the geodesic ray transform proved in [95]. Before we can use the results in [95] we show that the mixing  $A_{k,l}$  in the mixing ray transform  $L_{k,l} = \mathcal{I}_m \circ A_{k,l}$  maps tensor fields in  $E_{\eta}^1(T_m M)$  to tensor fields in  $E_{\eta}^1(T_m M)$ , and similarly tensor fields in  $P_{\eta}^1(T_m M)$  to tensor fields in  $P_{\eta}^1(T_m M)$ . We do not need to assume orientability in theorem 4.3 since Cartan–Hadamard manifolds are always orientable.

Theorem 4.2 implies that we can write the kernel of the mixed ray transform on compact orientable surfaces with boundary admitting s-injectivity of the geodesic ray transform as the direct sum

(18) 
$$\operatorname{Ker}(L_{k,l}|_{C^{\infty}(T_mM)}) = \operatorname{Im}(\mathcal{H}|_{C^{\infty}(T_mM)}) \oplus \operatorname{Im}(\widehat{\sigma}_{A_{k,l}} \nabla^{A_{k,l}}|_Y)$$

where  $\mathcal{H} = \mathrm{Id} - \hat{\sigma}_{A_{k,l}}$  is the projection onto the trivial part of  $\mathrm{Ker}(L_{k,l})$  and  $Y = \{v \in C^{\infty}(S_{m-1}M) : v|_{\partial M} = 0\}$ . Similar decomposition as in (18) holds for non-compact Cartan–Hadamard manifolds by theorem 4.3 using the sets of polynomially and exponentially decaying tensor fields.

In the article [F] we study the non-linear travel time tomography or boundary rigidity problem. The following theorem is the first main result of the article [F].

**Theorem 4.4** ([F, Theorem 1.3]). Let M be a compact and simply connected smooth manifold with boundary. For  $i \in \{1, 2\}$  let  $F_i = F_{r,i} + \beta_i$  be admissible Finsler norms where  $F_{r,i}$  is an admissible and reversible Finsler norm and  $\beta_i$ is a smooth closed 1-form such that  $\|\beta_i\|_{F_{r,i}^*} < 1$ . Then the following are equivalent:

- (i)  $d_{F_1}(x, x') = d_{F_2}(x, x')$  for all  $x, x' \in \partial M$ .
- (ii) There is unique scalar field  $\phi$  vanishing on the boundary such that  $\beta_2 = \beta_1 + d\phi$ , and  $d_{F_r,1}(x, x') = d_{F_r,2}(x, x')$  for all  $x, x' \in \partial M$ .

One can take  $F_r$  to be for example a simple Riemannian metric in theorem 4.4 since they are admissible and reversible. Since  $F_r$  is reversible for any curve  $\gamma$  we can obtain  $L_{F_r}(\gamma)$  from the symmetric part and  $\int_{\gamma} \beta$  from the antisymmetric part of the length functional  $L_F(\gamma)$ . In other words, the data for  $\beta$  and  $F_r$  "decouple". Closedness of the 1-form  $\beta$  is in essential role in proving theorem 4.4:  $d\beta = 0$  implies that  $F_r$  and  $F = F_r + \beta$  have the same geodesics up to orientation preserving reparametrizations, and geodesics of F remain geodesics as point sets when their parametrization is reversed. Simply connectedness of M implies that  $\beta_i = d\phi_i$  for some scalar field  $\phi_i$  and this fact also plays a role in the proof.

The next theorem is the second main result of the article [F] and it gives a boundary rigidity result for certain Randers metrics.

**Theorem 4.5** ([F, Theorem 1.5]). Let M be a compact and simply connected smooth manifold with boundary. For  $i \in \{1, 2\}$  let  $F_i = F_{g_i} + \beta_i$  be admissible Finsler norms where  $g_i$  is an admissible Riemannian metric and  $\beta_i$  is a smooth closed 1-form such that  $\|\beta_i\|_{g_i} < 1$ . Assume that  $(M, g_i)$  is boundary rigid. Then the following are equivalent:

- (a)  $d_{F_1}(x, x') = d_{F_2}(x, x')$  for all  $x, x' \in \partial M$ .
- (b) There is unique scalar field  $\phi$  vanishing on the boundary and a diffeomorphism  $\Psi$  which is identity on the boundary such that  $\beta_2 = \beta_1 + d\phi$ and  $g_2 = \Psi^* g_1$ .

34

(c) There is unique scalar field  $\phi$  vanishing on the boundary and a diffeomorphism  $\Psi$  which is identity on the boundary such that  $\beta_2 = \Psi^* \beta_1 + d\phi$  and  $g_2 = \Psi^* g_1$ .

Theorem 4.5 is proved by using theorem 4.4 and the rigidity assumption on the Riemannian metrics  $g_i$ . Theorem 4.5 part (c) implies that  $F_2 = \Psi^* F_1 + d\phi$ . Finsler norms satisfying such relation are sometimes called almost isometric Finsler norms and the diffeomorphism  $\Psi : (M, F_2) \to (M, F_1)$ is called an almost isometry [18, 35, 66, 74]. We note that  $\Psi$  cannot be an isometry since this would require that  $\Psi^*\beta_1 = \beta_2$ . Theorem 4.5 can be seen as a generalization of the Riemannian boundary rigidity results to nonreversible Randers manifolds. If n = 2, then one can take  $g_i$  to be a simple Riemannian metric in theorem 4.5 since in two dimensions simple Riemannian metrics are boundary rigid [122].

Theorem 4.5 has the following application in seismology. Assume that  $M = \overline{B}(0, R) \subset \mathbb{R}^n$  is a closed ball of radius R > 0 equipped with the Riemannian metric  $g = c^{-2}(r)e$  where c = c(r) is a radial sound speed satisfying the Herglotz condition

$$\frac{\mathrm{d}}{\mathrm{d}r}\bigg(\frac{r}{c(r)}\bigg)>0,\quad r\in[0,R],$$

and e is the Euclidean metric. In addition, let us assume that g has no conjugate points so that g becomes a simple Riemannian metric [107, 152]. Suppose that the seismic wave propagates in a moving medium which velocity field is given by the vector field W. Using Zermelo's navigation problem and a first-order approximation we obtain that if the scaled flow field  $W/c^2$  is irrotational  $(d(W/c^2) = 0)$ , then one can uniquely determine the speed of sound c and the velocity field W up to potential fields from travel time measurements of seismic waves which are done on the boundary  $\partial M = S^{n-1}(0, R) \subset \mathbb{R}^n$ .

#### References

- A. Abhishek. Support theorems for the transverse ray transform of tensor fields of rank m. J. Math. Anal. Appl., 485(2):123828, 2020.
- [2] A. Abouelaz. The d-plane Radon transform on the torus T<sup>n</sup>. Fract. Calc. Appl. Anal., 14(2):233-246, 2011.
- [3] T. Aikou and L. Kozma. Global aspects of Finsler geometry. In D. Krupka and D. Saunders, editors, *Handbook of Global Analysis*, pages 1–39. Elsevier, Amsterdam, 2008.
- [4] R. Alexander. Planes for which the lines are the shortest paths between points. *Illinois J. Math.*, 22(2):177–190, 1978.
- [5] R. V. Ambartzumian. A note on pseudo-metrics on the plane. Z. Wahrscheinlichkeitstheorie verw. Gebiete, 37(2):145–155, 1976.
- [6] Y. E. Anikonov and V. G. Romanov. On uniqueness of determination of a form of first degree by its integrals along geodesics. J. Inverse Ill-Posed Probl., 5(6):487–490, 1997.
- [7] P. L. Antonelli, A. Bóna, and M. A. Slawiński. Seismic rays as Finsler geodesics. Nonlinear Anal. Real World Appl., 4(5):711–722, 2003.

- [8] K. Astala and L. Päivärinta. Calderón's Inverse Conductivity Problem in the Plane. Ann. of Math. (2), 163(1):265–299, 2006.
- [9] D. Bao, S.-S. Chern, and Z. Shen. An Introduction to Riemann-Finsler Geometry. Springer-Verlag, first edition, 2000.
- [10] D. Bao, C. Robles, and Z. Shen. Zermelo navigation on Riemannian manifolds. J. Differential Geom., 66(3):377–435, 2004.
- [11] J. Bergh and J. Löfström. Interpolation Spaces: An Introduction. Grundlehren der mathematischen Wissenschaften. Springer-Verlag Berlin Heidelberg, first edition, 1976.
- [12] G. Besson, G. Courtois, and S. Gallot. Entropies et rigidités des espaces localement symétriques de courbure strictement négative. *Geom. Funct. Anal.*, 5(5):731–799, 1995.
- [13] S. Bhattacharyya, T. Ghosh, and G. Uhlmann. Inverse problem for fractional-Laplacian with lower order non-local perturbations. *Trans. Amer. Math. Soc.*, 374(5):3053–3075, 2021.
- [14] J. Bondy and U. Murty. Graph Theory with Applications. North-Holland, 1976.
- [15] H. Braun and A. Hauck. Tomographic Reconstruction of Vector Fields. *IEEE Trans. Signal Process.*, 39(2):464–471, 1991.
- [16] C. Bucur and E. Valdinoci. Nonlocal Diffusion and Applications, volume 20 of Lecture Notes of the Unione Matematica Italiana. Springer, [Cham]; Unione Matematica Italiana, Bologna, 2016.
- [17] D. Burago and S. Ivanov. Boundary rigidity and filling volume minimality of metrics close to a flat one. Ann. of Math. (2), 171(2):1183–1211, 2010.
- [18] J. Cabello and J. A. Jaramillo. A functional representation of almost isometries. J. Math. Anal. Appl., 445(2):1243–1257, 2017. A special issue of JMAA dedicated to Richard Aron.
- [19] A. P. Calderón. On an inverse boundary value problem. *Comput. Appl. Math.*, 25:133–138, 2006. Reprint of the original work by A. P. Calderón published by the Brazilian Mathematical Society (SBM) in ATAS of SBM (Rio de Janeiro), pp. 65-73, 1980.
- [20] X. Cao, Y.-H. Lin, and H. Liu. Simultaneously recovering potentials and embedded obstacles for anisotropic fractional Schrödinger operators. *Inverse Probl. Imaging*, 13(1):197–210, 2019.
- [21] M. Cekić, Y.-H. Lin, and A. Rüland. The Calderón problem for the fractional Schrödinger equation with drift. *Calc. Var. Partial Differential Equations*, 59(3):Paper No. 91, 46, 2020.
- [22] V. Cerveny. Seismic Ray Theory. Cambridge University Press, 2001.
- [23] S. N. Chandler-Wilde, D. P. Hewett, and A. Moiola. Sobolev spaces on non-Lipschitz subsets of  $\mathbb{R}^n$  with application to boundary integral equations on fractal screens. *Integral Equations Operator Theory*, 87(2):179–224, 2017.
- [24] S.-S. Chern and Z. Shen. *Riemann-Finsler Geometry*. World Scientific, 2005.
- [25] B. Colbois, F. Newberger, and P. Verovic. Some smooth Finsler deformations of hyperbolic surfaces. Ann. Global Anal. Geom., 35(2):191–226, 2009.
- [26] D. Cooper and K. Delp. The marked length spectrum of a projective manifold or orbifold. Proc. Am. Math. Soc., 138(9):3361–3376, 2010.
- [27] A. M. Cormack. Representation of a Function by Its Line Integrals, with Some Radiological Applications. J. Appl. Phys., 34(9):2722–2727, 1963.
- [28] A. M. Cormack. Representation of a Function by Its Line Integrals, with Some Radiological Applications. II. J. Appl. Phys., 35(10):2908–2913, 1964.
- [29] M. Courdurier, F. Noo, M. Defrise, and H. Kudo. Solving the interior problem of computed tomography using a priori knowledge. *Inverse Problems*, 24(6):065001, 2008.

- [30] G. Covi. An inverse problem for the fractional Schrödinger equation in a magnetic field. *Inverse Problems*, 36(4):045004, 24, 2020.
- [31] G. Covi. Inverse problems for a fractional conductivity equation. Nonlinear Anal., 193:111418, 18, 2020.
- [32] K. C. Creager. Anisotropy of the inner core from differential travel times of the phases PKP and PKIKP. *Nature*, 356:309–314, 1992.
- [33] C. B. Croke. Rigidity for surfaces of non-positive curvature. Comment. Math. Helv., 65(1):150–169, 1990.
- [34] C. B. Croke. Rigidity and the distance between boundary points. J. Differential Geom., 33(2):445-464, 1991.
- [35] A. Daniilidis, J. A. Jaramillo, and F. Venegas M. Smooth semi-Lipschitz functions and almost isometries between Finsler manifolds. J. Funct. Anal., 279(8):108662, 2020.
- [36] M. V. de Hoop and J. Ilmavirta. Abel transforms with low regularity with applications to X-ray tomography on spherically symmetric manifolds. *Inverse Problems*, 33(12):124003, 2017.
- [37] M. V. de Hoop, J. Ilmavirta, M. Lassas, and T. Saksala. A foliated and reversible Finsler manifold is determined by its broken scattering relation. 2020. arXiv:2003.12657.
- [38] M. V. de Hoop, J. Ilmavirta, M. Lassas, and T. Saksala. Determination of a compact Finsler manifold from its boundary distance map and an inverse problem in elasticity. 2020. arXiv:1901.03902.
- [39] M. V. de Hoop, T. Saksala, G. Uhlmann, and J. Zhai. Generic uniqueness and stability for mixed ray transform. 2019. arXiv:1909.11172.
- [40] M. V. de Hoop, T. Saksala, and J. Zhai. Mixed ray transform on simple 2dimensional Riemannian manifolds. Proc. Amer. Math. Soc., 147(11):4901–4913, 2019.
- [41] M. V. de Hoop, G. Uhlmann, and J. Zhai. Inverting the local geodesic ray transform of higher rank tensors. *Inverse Problems*, 35(11):115009, 2019.
- [42] A. Denisjuk. Inversion of the x-ray transform for 3D symmetric tensor fields with sources on a curve. *Inverse Problems*, 22(2):399–411, 2006.
- [43] E. Y. Derevtsov and I. Svetov. Tomography of tensor fields in the plain. Eurasian J. Math. Comput. Appl., 3(2):24–68, 2015.
- [44] E. Di Nezza, G. Palatucci, and E. Valdinoci. Hitchhiker's guide to the fractional Sobolev spaces. Bull. Sci. Math., 136(5):521–573, 2012.
- [45] Q. Du, M. Gunzburger, R. B. Lehoucq, and K. Zhou. Analysis and approximation of nonlocal diffusion problems with volume constraints. *SIAM Rev.*, 54(4):667–696, 2012.
- [46] A. M. Dziewonski and D. L. Anderson. Preliminary reference Earth model. Phys. Earth Planet. Inter., 25(4):297–356, 1981.
- [47] A. Feizmohammadi, J. Ilmavirta, and L. Oksanen. The Light Ray Transform in Stationary and Static Lorentzian Geometries. J. Geom. Anal., 2020.
- [48] G. B. Folland and A. Sitaram. The uncertainty principle: a mathematical survey. J. Fourier Anal. Appl., 3(3):207–238, 1997.
- [49] B. Fuglede. An Integral Formula. Math. Scand., 6:207–212, 1958.
- [50] P. Funk. Über Flächen mit lauter geschlossenen geodätischen Linien. Math. Ann., 74(2):278–300, 1913.
- [51] P. Funk. Über eine geometrische Anwendung der Abelschen Integralgleichung. Math. Ann., 77(1):129–135, 1915.
- [52] T. Ghosh, Y.-H. Lin, and J. Xiao. The Calderón problem for variable coefficients nonlocal elliptic operators. *Comm. Partial Differential Equations*, 42(12):1923–1961, 2017.

- [53] T. Ghosh, A. Rüland, M. Salo, and G. Uhlmann. Uniqueness and reconstruction for the fractional Calderón problem with a single measurement. J. Funct. Anal., 279(1):108505, 42, 2020.
- [54] T. Ghosh, M. Salo, and G. Uhlmann. The Calderón problem for the fractional Schrödinger equation. Anal. PDE, 13(2):455–475, 2020.
- [55] F. B. Gonzalez. On the range of the Radon d-plane transform and its dual. Trans. Amer. Math. Soc., 327(2):601–619, 1991.
- [56] L. Grafakos and S. Oh. The Kato-Ponce inequality. Comm. Partial Differential Equations, 39(6):1128-1157, 2014.
- [57] M. Gromov. Filling Riemannian manifolds. J. Differential Geom., 18(1):1–147, 1983.
- [58] A. Gulisashvili and M. A. Kon. Exact Smoothing Properties of Schrödinger Semigroups. Amer. J. Math., 118(6):1215–1248, 1996.
- [59] J. Hadamard. Le problème de Cauchy et les équations aux dérivées partielles linéaires hyperboliques. Hermann, Paris, 1932.
- [60] J. Hadamard. La théorie des équations aux dérivées partielles. Éditions scientifiques, Peking, 1964.
- [61] H. Hammer and B. Lionheart. Application of Sharafutdinov's Ray Transform in Integrated Photoelasticity. J. Elasticity, 75(3):229–246, 2004.
- [62] A. H. Hasanoğlu and V. G. Romanov. Introduction to Inverse Problems for Differential Equations. Springer International Publishing, 1st edition, 2017.
- [63] S. Helgason. Differential operators on homogeneous spaces. Acta Math., 102(3-4):239-299, 1959.
- [64] S. Helgason. Integral Geometry and Radon transforms. Springer, New York, 2011.
- [65] G. Herglotz. Über die Elastizität der Erde bei Berücksichtigung ihrer variablen Dichte. Zeitschr. für Math. Phys., 52:275–299, 1905.
- [66] J. Herrera and M. A. Javaloyes. Stationary-Complete Spacetimes with non-standard splittings and pre-Randers metrics. J. Geom. Phys, 163:104120, 2021.
- [67] J. Horváth. Topological Vector Spaces and Distributions. Vol. I. Addison-Wesley Publishing Co., Reading, Mass.-London-Don Mills, Ont., 1966.
- [68] J. Ilmavirta. On Radon transforms on tori. J. Fourier Anal. Appl., 21(2):370–382, 2015.
- [69] J. Ilmavirta and F. Monard. Integral geometry on manifolds with boundary and applications. In R. Ramlau and O. Scherzer, editors, *The Radon Transform: The First 100 Years and Beyond.* de Gruyter, 2019.
- [70] V. Isakov. Inverse Source Problems. Mathematical surveys and monographs. American Mathematical Society, 1990.
- [71] V. Isakov. Inverse Problems for Partial Differential Equations. Applied Mathematical Sciences. Springer International Publishing, 3rd edition, 2017.
- [72] S. Ivanov. Local monotonicity of Riemannian and Finsler volume with respect to boundary distances. *Geom. Dedicata*, 164(1):83–96, 2013.
- [73] T. Jansson, M. Almqvist, K. Stråhlén, R. Eriksson, G. Sparr, H. W. Persson, and K. Lindström. Ultrasound Doppler vector tomography measurements of directional blood flow. *Ultrasound Med. Biol.*, 23(1):47–57, 1997.
- [74] M. A. Javaloyes, L. Lichtenfelz, and P. Piccione. Almost isometries of non-reversible metrics with applications to stationary spacetimes. J. Geom. Phys., 89:38–49, 2015.
- [75] P. Juhlin. Principles of Doppler Tomography. Technical report, Center for Mathematical Sciences, Lund Institute of Technology, S-221 00 Lund, Sweden, 1992.
- [76] T. Kato and G. Ponce. Commutator Estimates and the Euler and Navier-Stokes Equations. Comm. Pure Appl. Math., 41(7):891–907, 1988.
- [77] E. Katsevich, A. Katsevich, and G. Wang. Stability of the interior problem with polynomial attenuation in the region of interest. *Inverse Problems*, 28(6):065022, 2012.

38

- [78] A. Kirsch. An Introduction to the Mathematical Theory of Inverse Problems. Applied Mathematical Sciences. Springer International Publishing, 3rd edition, 2021.
- [79] E. Klann, E. T. Quinto, and R. Ramlau. Wavelet methods for a weighted sparsity penalty for region of interest tomography. *Inverse Problems*, 31(2):025001, 2015.
- [80] H. Koch, A. Rüland, and M. Salo. On instability mechanisms for inverse problems. 2020. arXiv:2012.01855.
- [81] V. P. Krishnan. A support theorem for the geodesic ray transform on functions. J. Fourier Anal. Appl., 15(4):515–520, 2009.
- [82] V. P. Krishnan, R. K. Mishra, and S. K. Sahoo. Microlocal inversion of a 3dimensional restricted transverse ray transform on symmetric tensor fields. J. Math. Anal. Appl., 495(1):124700, 2021.
- [83] V. P. Krishnan and E. T. Quinto. Microlocal Analysis in Tomography. In O. Scherzer, editor, *Handbook of Mathematical Methods in Imaging*, pages 847–902. Springer, New York, 2015.
- [84] P. Kuchment, K. Lancaster, and L. Mogilevskaya. On local tomography. *Inverse Problems*, 11(3):571–589, 1995.
- [85] Å. Kurusa. A characterization of the Radon transform's range by a system of PDEs. J. Math. Anal. Appl., 161(1):218–226, 1991.
- [86] Å. Kurusa and T. Odor. Boundary-rigidity of projective metrics and the geodesic X-ray transform. 2020. Preprint.
- [87] R.-Y. Lai and Y.-H. Lin. Global uniqueness for the fractional semilinear Schrödinger equation. Proc. Amer. Math. Soc., 147(3):1189–1199, 2019.
- [88] R.-Y. Lai and Y.-H. Lin. Inverse problems for fractional semilinear elliptic equations. 2020. arXiv:2004.00549.
- [89] R.-Y. Lai, Y.-H. Lin, and A. Rüland. The Calderón problem for a space-time fractional parabolic equation. SIAM J. Math. Anal., 52(3):2655–2688, 2020.
- [90] N. Laskin. Fractional quantum mechanics and Lévy path integrals. Phys. Lett. A, 268(4-6):298–305, 2000.
- [91] N. Laskin. Fractional Quantum Mechanics. World Scientific Publishing Co. Pte. Ltd., Hackensack, NJ, 2018.
- [92] J. M. Lee. Introduction to Smooth Manifolds. Springer-Verlag New York, second edition, 2012.
- [93] J. M. Lee. Introduction to Riemannian Manifolds. Springer International Publishing, second edition, 2018.
- [94] J. Lehtonen. The geodesic ray transform on two-dimensional Cartan-Hadamard manifolds. 2016. arXiv:1612.04800.
- [95] J. Lehtonen, J. Railo, and M. Salo. Tensor tomography on Cartan–Hadamard manifolds. *Inverse Problems*, 34(4):044004, 2018.
- [96] L. Li. A Semilinear Inverse Problem For The Fractional Magnetic Laplacian. 2020. arXiv:2005.06714.
- [97] L. Li. The Calderón problem for the fractional magnetic operator. *Inverse Problems*, 36(7):075003, 14, 2020.
- [98] L. Li. Determining the magnetic potential in the fractional magnetic Calderón problem. *Comm. Partial Differential Equations*, 2020. Published online.
- [99] W. R. B. Lionheart and P. J. Withers. Diffraction tomography of strain. *Inverse Problems*, 31(4):045005, 2015.
- [100] S. Mardare. On Poincaré and de Rham's theorems. Rev. Roumaine Math. Pures Appl., 53(5-6):523–541, 2008.
- [101] V. G. Maz'ya and T. O. Shaposhnikova. Theory of Sobolev Multipliers. Springer, First edition, 2009.
- [102] W. McLean. Strongly Elliptic Systems and Boundary Integral Equations. Cambridge University Press, Cambridge, 2000.

- [103] R. Michel. Sur la rigidité imposée par la longueur des géodésiques. Invent. Math., 65(1):71–83, 1981.
- [104] R. Michel. Restriction de la distance géodésique à un arc et rigidité. Bull. Soc. Math. France, 122(3):435–442, 1994.
- [105] M. Mišur. A Refinement of Peetre's Theorem. Results Math., 74(4):199, 2019.
- [106] D. Mitrea. Distributions, Partial Differential Equations, and Harmonic Analysis. Universitext. Springer International Publishing, 2nd edition, 2018.
- [107] F. Monard. Numerical Implementation of Geodesic X-Ray Transforms and Their Inversion. SIAM J. Imaging Sci., 7(2):1335–1357, 2014.
- [108] J. L. Mueller and S. Siltanen. Linear and Nonlinear Inverse Problems with Practical Applications. Society for Industrial and Applied Mathematics, Philadelphia, PA, 2012.
- [109] R. G. Mukhometov. The reconstruction problem of a two-dimensional Riemannian metric, and integral geometry (Russian). *Dokl. Akad. Nauk SSSR*, 232(1):32–35, 1977.
- [110] A. I. Nachman. Global Uniqueness for a Two-Dimensional Inverse Boundary Value Problem. Ann. of Math., 143(1):71–96, 1996.
- [111] G. Nakamura, Z. Q. Sun, and G. Uhlmann. Global identifiability for an inverse problem for the Schrödinger equation in a magnetic field. *Math. Ann.*, 303(3):377– 388, 1995.
- [112] F. Natterer. The Mathematics of Computerized Tomography, volume 32 of Classics in Applied Mathematics. Society for Industrial and Applied Mathematics (SIAM), Philadelphia, PA, 2001. Reprint of the 1986 original.
- [113] F. Natterer and F. Wübbeling. Mathematical Methods in Image Reconstruction. SIAM, Philadelphia, 2001.
- [114] F. D. M. Neto and A. J. da Silva Neto. An Introduction to Inverse Problems with Applications. Springer-Verlag Berlin Heidelberg, 1st edition, 2013.
- [115] S. J. Norton. Tomographic Reconstruction of 2-D Vector Fields: Application to Flow Imaging. *Geophys. J. Int.*, 97(1):161–168, 1989.
- [116] S. J. Norton. Unique Tomographic Reconstruction of Vector Fields Using Boundary Data. *IEEE Trans. Image Process.*, 1(3):406–412, 1992.
- [117] R. L. Nowack. Tomography and the Herglotz-Wiechert inverse formulation. Pure Appl. Geophys., 133(2):305–315, 1990.
- [118] G. P. Paternain and M. Salo. A sharp stability estimate for tensor tomography in non-positive curvature. *Math. Z.*, 2020.
- [119] G. P. Paternain, M. Salo, and G. Uhlmann. Tensor tomography on surfaces. Invent. Math, 193(1):229–247, 2013.
- [120] G. P. Paternain, M. Salo, and G. Uhlmann. Tensor tomography: Progress and challenges. *Chin. Ann. Math. Ser. B*, 35(3):399–428, 2014.
- [121] J. Peetre. Une caractérisation abstraite des opérateurs différentiels. Math. Scand., 7:211–218, 1959.
- [122] L. Pestov and G. Uhlmann. Two dimensional compact simple Riemannian manifolds are boundary distance rigid. Ann. of Math. (2), 161(2):1093–1110, 2005.
- [123] L. N. Pestov and V. A. Sharafutdinov. Integral geometry of tensor fields on a manifold of negative curvature. Sib. Math. J., 29(3):427–441, 1988.
- [124] E. Quinto. Singularities of the X-Ray Transform and Limited Data Tomography in R<sup>2</sup> and R<sup>3</sup>. SIAM J. Math. Anal., 24(5):1215–1225, 1993.
- [125] E. Quinto. Artifacts and Visible Singularities in Limited Data X-Ray Tomography. Sens. Imaging, 18, 2017.
- [126] J. Radon. Über die Bestimmung von Funktionen durch ihre Integralwerte längs gewisser Mannigfaltigkeiten. Akad. Wiss., 69:262–277, 1917.
- [127] J. Railo. Fourier analysis of periodic Radon transforms. J. Fourier Anal. Appl., 26(4):Paper No. 64, 27, 2020.

- [128] K. Ramaseshan. Microlocal Analysis of the Doppler Transform on ℝ<sup>3</sup>. J. Fourier Anal. Appl., 10(1):73–82, 2004.
- [129] A. G. Ramm and A. I. Katsevich. The Radon Transform and Local Tomography. CRC Press, Boca Raton, First edition, 1996.
- [130] B. Rubin. Convolution-backprojection method for the k-plane transform, and Calderón's identity for ridgelet transforms. Appl. Comput. Harmon. Anal., 16(3):231-242, 2004.
- [131] B. Rubin. Reconstruction of functions from their integrals over k-planes. Israel J. Math, 141(1):93–117, 2004.
- [132] B. Rubin. On some inversion formulas for Riesz potentials and k-plane transforms. Fract. Calc. Appl. Anal., 15(1):34–43, 2012.
- [133] B. Rubin. Weighted norm inequalities for k-plane transforms. Proc. Amer. Math. Soc., 142(10):3455–3467, 2014.
- [134] B. Rubin. Norm estimates for k-plane transforms and geometric inequalities. Adv. Math, 349:29–55, 2019.
- [135] A. Rüland and M. Salo. Exponential instability in the fractional Calderón problem. Inverse Problems, 34(4):045003, 21, 2018.
- [136] A. Rüland and M. Salo. The fractional Calderón problem: low regularity and stability. Nonlinear Anal., 193:111529, 56, 2020.
- [137] A. Rüland and M. Salo. Quantitative approximation properties for the fractional heat equation. Math. Control Relat. Fields, 10(1):1–26, 2020.
- [138] M. Salo. Recovering first order terms from boundary measurements. J. Phys.: Conf. Ser., 73:012020, 2007.
- [139] T. Schuster. The 3D Doppler transform: elementary properties and computation of reconstruction kernels. *Inverse Problems*, 16(3):701–722, 2000.
- [140] T. Schuster. The importance of the Radon transform in vector field tomography. In R. Ramlau and O. Scherzer, editors, *The Radon Transform: The First 100 Years* and Beyond. de Gruyter, 2019.
- [141] A. Schwarz. Multi-tomographic flame analysis with a schlieren apparatus. Meas. Sci. Technol., 7(3):406–413, 1996.
- [142] V. Sharafutdinov. Slice-by-slice reconstruction algorithm for vector tomography with incomplete data. *Inverse Problems*, 23(6):2603–2627, 2007.
- [143] V. A. Sharafutdinov. Integral geometry of tensor fields. Inverse and Ill-posed Problems Series. VSP, Utrecht, 1994.
- [144] P. M. Shearer. Introduction to Seismology. Cambridge University Press, 3rd edition, 2019.
- [145] Z. Shen. Lectures on Finsler Geometry. World Scientific, 2001.
- [146] Z. Shen. Finsler Metrics with K=0 and S=0. Canad. J. Math., 55(1):112–132, 2003.
- [147] G. Sparr and K. Stråhlén. Vector field tomography, an overview. Technical report, Centre for Mathematical Sciences, Lund Institute of Technology, Lund, Sweden, 1998.
- [148] G. Sparr, K. Stråhlén, K. Lindström, and H. W. Persson. Doppler tomography for vector fields. *Inverse Problems*, 11(5):1051–1061, 1995.
- [149] P. Stefanov and G. Uhlmann. Microlocal Analysis and Integral Geometry (working title). 2018. Draft version.
- [150] P. Stefanov, G. Uhlmann, and A. Vasy. Boundary rigidity with partial data. J. Amer. Math. Soc., 29(2):299–332, 2016.
- [151] P. Stefanov, G. Uhlmann, and A. Vasy. Inverting the local geodesic x-ray transform on tensors. J. Anal. Math., 136(1):151–208, 2018.
- [152] P. Stefanov, G. Uhlmann, A. Vasy, and H. Zhou. Travel Time Tomography. Acta Math. Sin. (Engl. Ser.), 35:1085–1114, 2019.
- [153] J. Sylvester and G. Uhlmann. A uniqueness theorem for an inverse boundary value problem in electrical prospection. *Comm. Pure Appl. Math.*, 39(1):91–112, 1986.

- [154] J. Sylvester and G. Uhlmann. A Global Uniqueness Theorem for an Inverse Boundary Value Problem. Ann. of Math., 125(1):153–169, 1987.
- [155] F. Trèves. Topological Vector Spaces, Distributions and Kernels. Academic Press, New York-London, 1967.
- [156] H. Triebel. Interpolation Theory, Function Spaces, Differential Operators. North-Holland Publishing Company, 1978.
- [157] G. Uhlmann. Inverse problems: seeing the unseen. Bull. Math. Sci., 4(2):209–279, 2014.
- [158] G. Uhlmann and A. Vasy. The inverse problem for the local geodesic ray transform. *Invent. Math*, 205(1):83–120, 2016.
- [159] L. B. Vertgeim. Integral geometry problems for symmetric tensor fields with incomplete data. J. Inverse Ill-Posed Probl., 8(3):355–364, 2000.
- [160] E. Wiechert and K. Zoeppritz. Über Erdbebenwellen. Nachr. Königl. Ges. Wiss. Göttingen, 4:415–549, 1907.
- [161] T. Yajima and H. Nagahama. Finsler geometry of seismic ray path in anisotropic media. Proc. R. Soc. A, 465(2106):1763–1777, 2009.
- [162] J. Yang, H. Yu, M. Jiang, and G. Wang. High-order total variation minimization for interior tomography. *Inverse Problems*, 26(3):035013, 2010.

Included articles

# Unique continuation of the normal operator of the x-ray transform and applications in geophysics

J. Ilmavirta and K. Mönkkönen

First published in *Inverse Problems* 36, no. 4 (2020) https://doi.org/10.1088/1361-6420/ab6e75

https://arxiv.org/abs/1909.05585

Copyright © IOP Publishing

## [A]

### UNIQUE CONTINUATION OF THE NORMAL OPERATOR OF THE X-RAY TRANSFORM AND APPLICATIONS IN GEOPHYSICS

JOONAS ILMAVIRTA AND KEIJO MÖNKKÖNEN

ABSTRACT. We show that the normal operator of the X-ray transform in  $\mathbb{R}^d$ ,  $d \geq 2$ , has a unique continuation property in the class of compactly supported distributions. This immediately implies uniqueness for the X-ray tomography problem with partial data and generalizes some earlier results to higher dimensions. Our proof also gives a unique continuation property for certain Riesz potentials in the space of rapidly decreasing distributions. We present applications to local and global seismology. These include linearized travel time tomography with half-local data and global tomography based on shear wave splitting in a weakly anisotropic elastic medium.

#### 1. INTRODUCTION

Linearized travel time tomography of shear waves reduces mathematically to a version of the X-ray tomography problem under a suitable model. We are interested in shear wave splitting of waves travelling through the mantle, leading us to a partial data problem. The partial data problem of the X-ray transform can then be reduced to a unique continuation problem of the normal operator of the X-ray transform. We study the unique continuation property of the normal operator mathematically and apply it to show that our partial data problems arising from geophysics have unique solutions.

Consider the following X-ray tomography problem with partial data. Assume we have a compactly supported function or distribution f on  $\mathbb{R}^d$ ,  $d \ge 2$ , and an open set  $V \subset \mathbb{R}^d$ . Suppose we only know the integrals of f over the lines through V and the values of f in V. Does this information determine f uniquely? In terms of the X-ray transform X, if  $Xf(\gamma) = 0$  for all lines  $\gamma$  intersecting V and  $f|_V = 0$ , is it true that f = 0? The answer is positive and even more is true.

The partial data problem can be recast into a unique continuation problem of the normal operator  $N = X^*X$  of the X-ray transform. In other words, if  $Nf|_V = 0$  and  $f|_V = 0$ , does it imply that f = 0? The answer is 'yes', and we prove a stronger unique continuation property for N where we only require that Nfvanishes to infinite order at some point in V. The proof also applies to some Riesz potentials of rapidly decreasing distributions. As a corollary we get the uniqueness result for the X-ray tomography problem with partial data.

It is well known that the partial data problem or region of interest (ROI) problem has important applications in medical imaging (see e.g. [21, 22, 34, 57, 58]). We

Date: January 17, 2020.

<sup>2010</sup> Mathematics Subject Classification. 86A22, 44A12, 46F12.

Key words and phrases. Inverse problems, X-ray tomography, normal operator, unique continuation, theoretical seismology.

Department of Mathematics and Statistics, University of Jyväskylä, P.O. Box 35 (MaD) FI-40014 University of Jyväskylä, Finland; joonas.ilmavirta@jyu.fi.

Department of Mathematics and Statistics, University of Jyväskylä, P.O. Box 35 (MaD) FI-40014 University of Jyväskylä, Finland; kematamo@student.jyu.fi.

introduce two possibly new applications in theoretical seismology. Namely, we show that one can uniquely solve a linearized travel time problem with receivers only in a small open subset of the Earth's surface. In addition, we describe how to use shear wave (S-wave) splitting measurements to determine the difference of the S-wave speeds. See section 1.2 for details on these applications.

Similar partial data results are known in  $\mathbb{R}^2$  for compactly supported smooth functions, compactly supported  $L^1$ -functions and compactly supported distributions [4, 21, 22]. Our method of proof applies to all dimensions  $d \ge 2$ . An important novelty is in looking at the partial data result from the point of view of unique continuation of the normal operator. The theorem can be seen as a complementary result to the Helgason support theorem (see lemma 2.3) where one requires that the lines do not intersect the set in question. Our result can also be seen as a unique continuation property for the inverse operator of the fractional Laplacian  $(-\Delta)^s$ .

We present two alternative proofs for the partial data problem. The first proof uses the unique continuation property of the normal operator of the X-ray transform. The second proof is more direct and uses spherical symmetry. However, both proofs rely on a similar idea, differentiation of an integral kernel and density of polynomials. We also present an alternative proof for the unique continuation of the Riesz potential which is based on unique continuation of the fractional Laplacian.

1.1. The main results. Denote by  $\mathcal{D}(\mathbb{R}^d)$  the set of compactly supported smooth functions and by  $\mathcal{D}'(\mathbb{R}^d)$  the space of all distributions in  $\mathbb{R}^d$ ,  $d \geq 2$ . Also denote by  $\mathcal{E}'(\mathbb{R}^d)$  the set of compactly supported distributions in  $\mathbb{R}^d$ . Let  $\alpha = d - 1$  or  $\alpha \in \mathbb{R} \setminus \mathbb{Z}$  and  $\alpha < d$ . We define the Riesz potential  $I_{\alpha}f = f * h_{\alpha}$  for  $f \in \mathcal{E}'(\mathbb{R}^d)$  where  $h_{\alpha}(x) = |x|^{-\alpha}$  and the convolution is understood in the sense of distributions. If  $\alpha = d - 1$ , then  $I_{\alpha}$  reduces to the normal operator of the X-ray transform up to a constant factor 2. We say that  $I_{\alpha}f$  vanishes to infinite order at a point  $x_0$  if  $\partial^{\beta}(I_{\alpha}f)(x_0) = 0$  for all  $\beta \in \mathbb{N}^d$ . Our main result is the following (see also theorem 5.1 and theorem 5.2).

**Theorem 1.1.** Let  $f \in \mathcal{E}'(\mathbb{R}^d)$ ,  $V \subset \mathbb{R}^d$  any nonempty open set and  $x_0 \in V$ . If  $f|_V = 0$  and  $I_{\alpha}f$  vanishes to infinite order at  $x_0$ , then f = 0. In particular, this holds for the normal operator of the X-ray transform.

The condition  $f|_V = 0$  guarantees that  $I_{\alpha}f$  is smooth in a neighborhood of  $x_0$ . The pointwise derivatives  $\partial^{\beta}(I_{\alpha}f)(x_0)$  therefore exist, see the proof of theorem 1.1 for details. The condition of vanishing derivatives at a point only makes sense under the assumption that f vanishes (or is smooth) in V.

Theorem 1.1 can be seen as a unique continuation property of the Riesz potential  $I_{\alpha}$ . The result resembles a strong unique continuation property but the roles in the decay conditions are interchanged. As an immediate corollary we obtain the following partial data results for the X-ray tomography problem. The first one is similar compared to the uniqueness results in [21, 22]. For the definition of the X-ray transform on distributions, see section 3.

**Theorem 1.2.** Let  $V \subset \mathbb{R}^d$  be any nonempty open set. If  $f \in \mathcal{E}'(\mathbb{R}^d)$  satisfies  $f|_V = 0$  and Xf vanishes on all lines that intersect V, then f = 0.

**Corollary 1.3.** Let R > r > 0 and  $f \in \mathcal{E}'(\mathbb{R}^d)$  such that  $\operatorname{spt}(f) \subset \overline{B}(0,R) \setminus B(0,r)$ . If X f vanishes on all lines that intersect B(0,r), then f = 0.

**Corollary 1.4.** Let  $\Omega \subset \mathbb{R}^d$  be a bounded, smooth and strictly convex set and  $\Sigma \subset \partial \Omega$  any nonempty open subset of its boundary. If  $f \in \mathcal{E}'(\mathbb{R}^d)$  is supported in  $\overline{\Omega}$  and its X-ray transform vanishes on all lines that meet  $\Sigma$ , then f = 0.

Proofs of the theorems and corollaries can be found in section 2.3 (see also the alternative proofs in section 5). Some of our assumptions are crucial for the theorems to be true. Theorem 1.2 is clearly false if d = 1. The function f cannot be determined from its integrals over the lines through the ROI only [22, 34, 48]. Thus one needs some information of f in the open set V which the lines all meet. Especially we need the assumption  $f|_V = 0$  when we use the Kelvin transform and density of polynomials. Our proof also exploits the assumption of compact support which is motivated by the physical setting and is needed to define the Riesz potential on distributions. However, one can relax that assumption to rapid decay at infinity (see theorem 5.1 and theorem 5.2). Theorem 1.2 and corollaries 1.3 and 1.4 have important applications in theoretical seismology and medical imaging. This is discussed in more depth in the next section.

1.2. **Applications.** Our results have theoretical applications in seismology. Applications include linearization of anisotropies in S-wave splitting and linearized travel time tomography. Even though there exist many different types of seismic data, we only use linearized travel time data without reflections in our models. For the following treatment of splitting of S-waves we refer to [5, 28, 29, 45, 47].

In linear elasticity in  $\mathbb{R}^3$  there are three polarizations of seismic waves which correspond to the eigenvectors of the symmetric Christoffel matrix. The eigenvalues correspond to wave speeds. In the isotropic case the largest eigenvalue is simple with the eigenvector parallel to the direction of propagation, corresponding to a P-wave. The other eigenvalue is degenerate with eigenvectors orthogonal to the P-wave polarization. These eigenvectors correspond to S-waves. In anisotropic medium this degeneracy is typically lost and the degenerate S polarization splits to two quasi-S (qS) polarizations. The data in the imaging method based on S-wave splitting is the arrival time difference between the two qS-waves.

One common type of anisotropy is hexagonally symmetric anisotropy. This means that there is a preferred direction or a symmetry axis and the velocities vary only with the angle from the axis, i.e. there is rotational symmetry. For example sedimentary layering and aligned crystals or cracks can cause hexagonal anisotropy. If the seismic wavelength is substantially larger than the layer or crack spacing, then the material appears to be anisotropic [1]. The widely used one-dimensional Preliminary Reference Earth Model (PREM) indicates this kind of anisotropy between the depths 80–220 km in the upper mantle [10, 47]. In the PREM-model the symmetry axis is radial and all the physical parameters of the Earth depend only on the depth. Anisotropies have also been observed in the shalow crust and in the inner core where the fastest direction is parallel to the rotation axis of the Earth [6, 47].

Our results pertain to so-called weak anisotropy, where we consider the anisotropy as a small perturbation to an isotropic reference model. In the isotropic background model S-waves have a speed  $c_0(x)$  for all directions and polarizations. When we add a small anisotropic perturbation, the speeds become  $c_i(x, v) = c_0(x) + \delta c_i(x, v)$ , i = 1, 2. Here  $v \in S^2$  is the direction of propagation of the wave. In the linearized regime  $|\delta c_i| \ll |c_0|$  we have

$$\frac{1}{c_i(x,v)} = \frac{1}{c_0(x) + \delta c_i(x,v)} \approx \frac{1}{c_0(x)} - \frac{\delta c_i(x,v)}{c_0^2(x)}.$$

If we only measure small differences in the arrival times, our data is roughly

$$\delta t \approx \int_{\gamma} \frac{\mathrm{d}s}{c_1(x,v)} - \int_{\gamma} \frac{\mathrm{d}s}{c_2(x,v)} \approx \int_{\gamma} \frac{\delta c_2(x,v) - \delta c_1(x,v)}{c_0^2(x)} \mathrm{d}s.$$

Thus upon linearization, the data is the X-ray transform of  $c_0^{-2}(\delta c_2 - \delta c_1)$ . To simplify this problem, we assume the function to depend on x but not on v. If the splitting occurs in a layer near the surface (see figure 1), we are in the setting of

corollary 1.3. The corollary implies that the linearized shear wave splitting data determines  $\delta c_2 - \delta c_1$  and thus  $c_2 - c_1$  uniquely in the outermost layer.



FIGURE 1. A highly simplified picture of the setting in the linearized model. The splitting can occur at every interface but we only care about the splitting near the surface with smallest difference in the arrival times. There may exist different polarization states during the propagation of the initial wave, we only assume that the second to last part is an S-type wave. Our data consists purely of the branched parts of the waves.

Travel time tomography has a close relationship to the boundary rigidity problem where the aim is to reconstruct the metric of a manifold from boundary distance measurements [50, 53]. In seismology these distances correspond to travel times of seismic waves which are assumed to propagate along geodesics or straightest possible paths in the manifold. This problem is highly nonlinear and difficult to solve in full generality. Thus it is relevant to consider the first-order approximation and linearize the problem. When we linearize the general travel time tomography problem assuming our manifold to be  $\mathbb{R}^d$  and that the variations in the metric are conformally Euclidean, the geodesics become lines and the problem reduces to the X-ray tomography problem of a scalar function.

Linearized travel time tomography motivates the following application of observing earthquakes by seismic arrays on the surface of the Earth. In the context of corollary 1.4 one can ideally think that some open set of the surface is covered densely by seismometers (see figure 2). One detects earthquakes only in this set and measures travel times of seismic waves originating anywhere on the surface. In geometrical terms, our geodesics have one endpoint in this open set and the other endpoint can freely vary. In contrast to "local data" where both endpoints are in the small set, we call this setting "half-local data". The interesting question then is whether this limited set of travel time data can determine the inner structure of the Earth uniquely. When we do the usual conformal linearization in the Euclidean background, we end up with partial X-ray tomography problem of a scalar function. Corollary 1.4 then tells that in principle one can use these kind of seismic arrays to uniquely determine the conformal factor in the linearization.

5



FIGURE 2. The setting as in corollary 1.4. Here  $\Sigma$  (thick) represents the seismic array where one measures the travel times of seismic waves and  $\Omega$  represents the Earth.

In addition to theoretical seismology one important application is medical imaging, see [21, 22, 34, 57, 58] and the references therein. Suppose we want to reconstruct a specific part of the human body, a region of interest (ROI). Is it possible to reconstruct the image by shooting X-rays only through the ROI? If this was possible it would be unnecessary to give a higher dose of X-rays to the patient and radiate regions outside the ROI which do not contribute significantly to the image. We can interpret the function f in theorem 1.2 as the attenuation of X-rays which to a good approximation travel along straight lines inside a body. Somehow surprisingly theorem 1.2 tells us that if we know the values of f in a small open set inside the ROI and the integrals of f over the lines going through the ROI, then f is uniquely determined everywhere (see figure 3). It is important to note that arbitrary attenuation cannot be determined from the line integrals only even in the ROI but one can always recover the singularities in the ROI [22, 34, 48].

1.3. Related results. The partial data problem for the X-ray transform has been solved earlier in  $\mathbb{R}^2$  under a variety of assumptions [4, 21, 22, 55]. The uniqueness result is known for  $C_c^{\infty}$ -functions and compactly supported  $L^1$ -functions if one assumes the knowledge of f inside an open set in the ROI [4, 22]. One also obtains uniqueness without knowing the exact values of f in the ROI; if f is piecewise constant or piecewise polynomial in the ROI, then the X-ray data determines funiquely [22, 55]. If f is polynomial in the ROI, then one obtains stability as well [21]. Closest to our theorem is the uniqueness result in [21] (see also [22] where the authors mention in the proof of lemma 2.4 that their method applies also to compactly supported distributions which are piecewise constant in the ROI. According to that result, if  $f \in \mathcal{E}'(\mathbb{R}^2)$  integrates to zero over all lines intersecting V and  $f|_V$  is real analytic, then f = 0.

Our result for the partial data problem uses stronger assumption  $f|_V = 0$ . This assumption is needed so that the Kelvin transformed function will be compactly supported and we can use density of polynomials. However, our theorem applies to any dimension  $d \ge 2$ . Another difference is in the point of view; we consider the normal operator and observe that the same result holds for a larger class of



FIGURE 3. Basic idea of ROI-tomography in the context of theorem 1.2. Here V is the region of interest and  $U \subset V$  some open subset. If one knows the attenuation f in U and the integrals of f over the lines through V, then one can construct f uniquely from the data.

Riesz potentials. Also our alternative proofs (theorem 5.1 and theorem 5.2) imply uniqueness for the partial data problem without assumption of compact support, rapid decay at infinity is enough. We remark that the X-ray data alone does not uniquely determine the attenuation in general. One cannot even construct  $C_c^{\infty}$ functions only from the integrals but one can always recover the singularities, which is equivalent with recovering the function up to a smooth error [22, 34, 48].

Unlike in [4, 21] our method is very unstable and concrete reliable reconstructions are basically hopeless. Our instability comes from the differentiation of the data and approximation of test functions by polynomials up to arbitrary order. However, our method of proof is not the only reason for instability. Instability is an intrinsic property of partial data problems. When we have limited X-ray data it is not guaranteed that we can see all the singularities of f from the data. Singularities which are invisible in the microlocal sense are related to the instability of inverting ffrom its limited X-ray data [25, 33, 35, 36]. See also [24, 37] for discussion of which part of the wave front set is visible in limited data tomography. Even though our theorem loses stability it gives uniqueness which is relevant for applications.

Our theorem is related to travel time tomography and the inverse kinematic problem. For a review of these, see [50, 53] and also [17, 54] for the original works by Herglotz, Wiechert and Zoeppritz. Specifically our result is a contribution to local and global theoretical seismology (see section 1.2). For example one can uniquely determine the difference of the anisotropic perturbations of the S-wave speeds by measuring the arrival time differences of the split S-waves. From the point of view of ROI tomography these seismic applications are new to the best of our knowledge.

It is also worth mentioning that our result is in a sense complementary to the famous support theorem by Helgason (see lemma 2.3). Helgason's theorem states that if  $C \subset \mathbb{R}^d$  is a convex compact set and  $f \in \mathcal{E}'(\mathbb{R}^d)$  such that  $f|_C = 0$  and the X-ray transform Xf vanishes on all lines not meeting C, then f = 0. Compared to theorem 1.2, Helgason's result uses complementary data but gives the same conclusion. Helgason's theorem holds also for rapidly decreasing continuous functions; our partial data result is true for this function class as well (see section 5 and the discussion after theorem 5.1).

Our theorem has a connection to the fractional Laplacian  $(-\Delta)^s$ . The operator  $(-\Delta)^s$  can be defined in many equivalent ways and one way is to consider it

as the inverse of a Riesz potential [26]. In our notation  $I_{\alpha}f = (-\Delta)^{-s}f$  where  $s = (d - \alpha)/2$  assuming  $0 < \alpha < d$ . For example from equation (2) we see that in Euclidean space the normal operator of the X-ray transform N is the inverse of the fractional Laplacian  $(-\Delta)^{1/2}$ . Thus our result can be seen as a unique continuation property for the operator  $(-\Delta)^{-\delta/2}$  where  $\delta$  is any positive non-integer or  $\delta = 1$ . There are several unique continuation results for the operator  $(-\Delta)^s$ when 0 < s < 1 and they have been recently used in fractional Calderón problems [14, 15, 39, 42]. One version of our theorem can be proved using unique continuation of  $(-\Delta)^s$  (see theorem 5.2). The fractional Laplacian even admits a strong unique continuation property if one assumes more regularity from the function [11, 41]. Here "strong" means that the function does not need to be zero in an open set, it only has to vanish to infinite order at some point. Theorem 1.1 has similar vanishing assumption for  $I_{\alpha}f$  instead of f. There are also (strong) unique continuation results for the higher order Laplacian  $(-\Delta)^t$  where t is a positive non-integer exponent [12, 13, 56].

In Euclidean space one can reconstruct a compactly supported distribution uniquely from its X-ray transform [48]. There even exist explicit inversion formulas using the formal adjoint  $X^*$  and the normal operator N. It is also known that the X-ray transform is injective on compact simple Riemannian manifolds with boundary [19]. Interesting injectivity results considering seismic applications have been obtained for conformally Euclidean metrics which satisfy the Herglotz condition [7]. See also how the length spectrum can be obtained from the Neumann spectrum of the Laplace-Beltrami operator or from the toroidal modes on these kind of manifolds in three dimensions [8]. This has a connection to the free oscillations of the Earth.

There are some partial data results for certain manifolds. If (M, g) is a twodimensional compact simple Riemannian manifold with boundary and a real-analytic metric g, then one can reconstruct  $L^2$ -functions locally from their geodesic X-ray transform [23]. In dimensions  $d \geq 3$  one can relax the analyticity condition to smoothness using a convexity assumption on the boundary [52]. Furthermore one can even invert the X-ray transform locally in a stable way and obtain a reconstruction formula based on Neumann series. Both of the results in [23, 52] rely on microlocal analysis. One can also locally invert, up to potential fields, tensors of order 1 and 2 near a strictly convex boundary point [49]. We remark that there is a similar distinction between analyticity and smoothness for the injectivity of the weighted X-ray transform in Euclidean space. When d = 2 the analyticity of the weight is required for injectivity while in higher dimensions smoothness is enough [2, 3, 48].

1.4. **Organization of the paper.** We begin our treatment by proving the main results in section 2. We also discuss the assumptions used in the results and applications. In section 3 we recall some basic theory of distributions and integral geometry in  $\mathbb{R}^d$ . Section 4 is devoted to the proof of lemma 2.2 which says that one can express all the polynomials in a certain form as a finite linear combination of the derivatives of the kernel of the Riesz potential  $I_{\alpha}$ . Section 5 contains alternative proofs for theorem 1.1 and theorem 1.2.

Acknowledgements. J.I. was supported by the Academy of Finland (decision 295853) and K.M. was supported by Academy of Finland (Centre of Excellence in Inverse Modelling and Imaging, grant numbers 284715 and 309963). We thank Maarten de Hoop and Todd Quinto for discussions. We also thank Mikko Salo for pointing out the connection between our result and the unique continuation of

the fractional Laplacian. We are grateful to the anonymous referees for insightful remarks and suggestions.

#### 2. Proofs of the main results

2.1. An overview of the proof. The rough idea of the proof of theorem 1.1 is the following. We may assume that  $x_0 = 0$ . The function  $I_{\alpha}f$  is smooth in V, and by assumption all of its derivatives vanish at the origin. By a convolution argument these derivatives can be computed explicitly. The vanishing of these derivatives amounts to f integrating to zero against a set of functions. After a change of variables and suitable rescaling, one can use density of polynomials to show that this set is dense. Therefore f has to vanish.

The proofs of the corollaries are more straightforward. Detailed proofs of these main results are given in section 2.3 below. The reader who is not familiar with the theory of distributions and integral geometry can first read section 3. See section 5 for alternative proofs of theorems 1.1 and 1.2.

2.2. Auxiliary results. In this section we give a few auxiliary results which are needed in our proofs. The first one is a known theorem in distribution theory.

**Lemma 2.1** ([51, p.160 Corollary 4]). Let  $\Omega \subset \mathbb{R}^d$  be an open set. Then the polynomials form a dense subspace of  $\mathcal{E}(\Omega)$ .

Recall the kernel of the Riesz potential  $h_{\alpha}(x) = |x|^{-\alpha}$ . The next lemma is proved in section 4.

**Lemma 2.2.** If  $d \ge 2$  and  $\alpha > d-2$  or  $\alpha \in \mathbb{R} \setminus \mathbb{Z}$ , then for any polynomial p one can express the product  $p(K(x))h_{\alpha}(x)$  as a finite linear combination of derivatives of  $h_{\alpha}$ . Here  $K(x) = x |x|^{-2}$  is the Kelvin transform.

We also need the following support theorem to prove corollary 1.4. The proof can be found for example in [16, 48].

**Lemma 2.3** (Helgason's support theorem). Let  $C \subset \mathbb{R}^d$  be a compact convex set and  $f \in \mathcal{E}'(\mathbb{R}^d)$ . If Xf vanishes on all lines not meeting C, then  $\operatorname{spt}(f) \subset C$ .

2.3. **Proofs of the results.** Now we are ready to prove our main theorem and its corollaries. Let  $d \ge 2$ . Recall the definition of the Riesz potential  $I_{\alpha}f = f * h_{\alpha}$  for  $f \in \mathcal{E}'(\mathbb{R}^d)$  where  $\alpha = d - 1$  or  $\alpha \in \mathbb{R} \setminus \mathbb{Z}$  and  $\alpha < d$ . The kernel  $h_{\alpha}$  has an expression  $h_{\alpha}(x) = |x|^{-\alpha}$ . We denote by K the Kelvin transform  $K(x) = x |x|^{-2}$ . See section 3 for basic results on distribution theory used in the proof.

Proof of theorem 1.1. We have to show that if  $f \in \mathcal{E}'(\mathbb{R}^d)$  and  $V \subset \mathbb{R}^d$  is any nonempty open set such that  $f|_V = 0$  and  $\partial^\beta (I_\alpha f)(x_0) = 0$  for some  $x_0 \in V$  and all  $\beta \in \mathbb{N}^d$ , then f = 0. Because the problem is translation invariant we can assume that  $x_0 = 0$ . Since f has compact support and it vanishes in a neighborhood of the origin, we have that  $\operatorname{spt}(f) \subset A$  for some open annulus A centered at the origin. Let  $g \in \mathcal{D}(\mathbb{R}^d)$  be a symmetric smooth version of  $h_\alpha$  such that  $g|_A = h_\alpha|_A$ . Choosing small enough  $\epsilon > 0$  we have  $I_\alpha f|_{B(0,\epsilon)} = (f * g)|_{B(0,\epsilon)}$  where  $f * g \in \mathcal{D}(\mathbb{R}^d)$  by lemma 3.4. Since  $I_\alpha f$  vanishes to infinite order at 0 lemmas 3.4 and 3.5 give us  $\partial^\beta (f * g)(0) = (f * (\partial^\beta g))(0) = \left\langle f, \tau_0 \partial^\beta g \right\rangle = \left\langle f, \partial^\beta g \right\rangle = 0$  for all multi-indices  $\beta \in \mathbb{N}^d$ . Since g is symmetric we get the condition  $\langle f, \partial^\beta g \rangle = 0$ .

Let  $\eta \in C_c^{\infty}(A)$  be such that  $\eta = 1$  in spt(f). By lemma 3.1 and the definition of restriction  $f|_A$  we have  $0 = \langle f, \partial^{\beta}g \rangle = \langle f, \eta \partial^{\beta}g \rangle = \langle f|_A, \eta \partial^{\beta}g \rangle$ . Since  $g|_A = h_{\alpha}|_A$  by lemma 2.2 we obtain all the polynomials p in the form  $p(K(x))h_{\alpha}(x)$  restricted to A by taking finite linear combinations of the derivatives of g. Using linearity we obtain  $\langle f|_A, \eta h_{\alpha}(p \circ K) \rangle = 0$  for all polynomials p. Taking the pullback we

get  $\langle f|_A \circ K, \eta_1 p \rangle = 0$  where  $\eta_1 = ((\eta |J_{K^{-1}}|^{-1}) \circ K)h_{\alpha}^{-1}$ . Let  $\psi \in \mathcal{E}(K^{-1}(A))$ . By lemma 2.1 there exists a sequence of polynomials  $p_k$  such that  $p_k \to \psi$  in  $\mathcal{E}(K^{-1}(A))$ . This implies  $\eta_1 p_k \to \eta_1 \psi$  in  $\mathcal{E}(K^{-1}(A))$  because  $\operatorname{spt}(\eta_1) \subset \subset K^{-1}(A)$ . Since  $f|_A \circ K \in \mathcal{E}'(K^{-1}(A))$  by continuity  $\langle \eta_1(f|_A \circ K), \psi \rangle = \langle f|_A \circ K, \eta_1 \psi \rangle = 0$ , i.e.  $\eta_1(f|_A \circ K) = 0$ . But now  $\eta_1 \neq 0$  in  $K^{-1}(\operatorname{spt}(f)) = \operatorname{spt}(f|_A \circ K)$  and hence  $f|_A \circ K = 0$  by lemma 3.3. Again using lemma 3.2 we obtain  $f|_A = 0$  which implies f = 0.

As an immediate consequence we obtain the proofs for the X-ray tomography problem with partial data.

Proof of theorem 1.2. We have to show that if  $f \in \mathcal{E}'(\mathbb{R}^d)$  and  $V \subset \mathbb{R}^d$  is any nonempty open set such that  $f|_V = 0$  and  $Xf|_{\Gamma_V} = 0$  where  $\Gamma_V$  is the set of all lines that intersect V, then f = 0. We can assume that V is a ball centered at the origin. Let  $\varphi \in \mathcal{D}(V)$ . From the definition of the normal operator of the X-ray transform we obtain  $\langle Nf, \varphi \rangle = \langle Xf, X\varphi \rangle = 0$  since  $X\varphi \in \mathcal{D}(\Gamma_V)$ . Hence  $Nf|_V = 0$ and the claim follows from theorem 1.1 by taking  $\alpha = d - 1$ .

Proof of corollary 1.3. We have to show that if R > r > 0 and  $f \in \mathcal{E}'(\mathbb{R}^d)$  such that  $\operatorname{spt}(f) \subset \overline{B}(0, R) \setminus B(0, r)$  and Xf vanishes on all lines that meet B(0, r), then f = 0. Take a nonempty open set  $V \subset C B(0, r)$ . Then we have  $f|_V = 0$  and Xf vanishes on all lines that intersect V. Theorem 1.2 implies that f = 0.  $\Box$ 

Proof of corollary 1.4. Let  $\Omega \subset \mathbb{R}^d$  be a bounded, smooth and strictly convex set and  $\Sigma \subset \partial \Omega$  nonempty open subset of the boundary. We have to show that if  $f \in \mathcal{E}'(\mathbb{R}^d)$  is supported in  $\overline{\Omega}$  and Xf vanishes on all lines that meet  $\Sigma$ , then f = 0. We can assume that  $\Sigma$  is connected by passing to a connected component. Denote by  $\operatorname{ch}(\Sigma)$  the convex hull of  $\Sigma$  (see figure 4). By the Helgason support theorem (lemma 2.3) the function f vanishes in  $\operatorname{ch}(\Sigma)$ . Take open set  $V \subset \operatorname{ch}(\Sigma)$ ,  $V \neq \emptyset$ . Then  $f|_V = 0$  and Xf vanishes on all lines that intersect V. We can apply theorem 1.2 to conclude that f = 0.



FIGURE 4. Idea of the proof of corollary 1.4. Here  $\Sigma$  (thick arc) is a connected open subset of  $\partial\Omega$  and ch( $\Sigma$ ) (segment) its convex hull. Helgason's support theorem (lemma 2.3) implies that f vanishes in ch( $\Sigma$ ) and then theorem 1.2 is used for the dashed set V to conclude that f = 0.

2.4. Discussion of assumptions and methods. We assume that  $f|_V = 0$  so as to ensure that  $I_{\alpha}f|_V$  is smooth and the differentiation makes sense. For this purpose alone it would have been enough to assume that  $f|_V$  is smooth. However, if  $f|_V$  is non-zero, our method of proof appears to become untractable. Especially the Kelvin transformed function  $f \circ K$  is not compactly supported anymore and we can not use density of polynomials in the proof. If  $f|_V$  is polynomial (or real analytic) and d = 2, the method of [21] can be applied to prove the partial data result for the X-ray transform directly. Our method has the additional freedom that the Riesz potential need not be exactly the normal operator and that the dimension is not restricted to two. Moreover, in the physical application of shear wave splitting in the mantle, only the anisotropy in the mantle will matter and the perturbation can thus be taken to be supported outside the core.

The assumptions in theorem 1.1 are not optimal. The assumption of compact support is needed to define the Riesz potential  $I_{\alpha}$  on distributions and is crucial in the proof when we use density of polynomials. However, compact support can be replaced with rapid decay at infinity (see theorem 5.1 and theorem 5.2). Theorem 1.2 is clearly false if d = 1. Also one cannot construct arbitrary  $C_c^{\infty}$ -functions from the integrals over the lines through the ROI only [22, 34, 48]. Therefore one needs some information of the function f in the open set V; our method of proof especially requires the assumption  $f|_V = 0$ . In corollary 1.4 it is enough to assume that only the subset  $\Sigma \subset \partial\Omega$  is strictly convex and the convex hull of the rest of the boundary does not cover all of  $\Sigma$ . The constraint  $\alpha < d$  comes from the requirement that the kernel  $h_{\alpha}$  determines a distribution. The other constraints for  $\alpha$  come from the proof of lemma 2.2.

It would be interesting to know whether we could weaken the decay assumption in theorem 1.2 in the smooth case. Does there exist  $f \in C^{\infty}(\mathbb{R}^d)$  such that  $f|_V = 0$ and Xf = 0 for all lines through V but f is not identically zero? By theorem 5.1 the result in theorem 1.2 holds when f decreases faster that any polynomial at infinity. There also exists a counterexample for the Helgason support theorem where the function does not decay rapidly enough [16, 34]. Since our theorem is similar in spirit, we would expect a counterexample also in our case.

The normal operator of the X-ray transform  $N = X^*X$  is an elliptic pseudodifferential operator. Therefore it would be natural to try methods of microlocal analysis to prove our main theorem. But the usual microlocal approach does not work here in the following sense. First, if we do the identification  $f \sim g$  if and only if  $f - g \in C^{\infty}(\mathbb{R}^d)$ , then the claim of theorem 1.1 is not true. Namely, the assumptions  $f|_V \in C^{\infty}(V)$  and  $Nf|_V \in C^{\infty}(V)$  do not imply that necessarily  $f \in C^{\infty}(\mathbb{R}^d)$ . Thus our result is not true modulo  $C^{\infty}$ . Second, from the assumptions of theorem 1.2 it is clear that some of the singularities of f are not visible in the data. These invisible singularities are usually difficult to reconstruct from the limited set of data [35, 36, 37]. The surprising thing here is that even though our data is local and smooth, we can still recover a distribution.

Our theorem considers the unique continuation of the normal operator of the X-ray transform. It is then natural to ask the following question: when does the normal operator of the geodesic X-ray transform on a manifold satisfy the unique continuation property? At the moment no results are known expect in the Euclidean case. Also there does not exist any simple relationship between the normal operator and the fractional Laplacian on general manifolds. In the context of seismic applications, it would be very beneficial to generalize the result to manifolds which are equipped with a conformally Euclidean metric satisfying the Herglotz condition [17, 54]. For example the widely used model of spherically symmetric

Earth (PREM model) satisfies the Herglotz condition to a good accuracy excluding discontinuity zones [10, 47]. But our method of proof seems to fit only to the Euclidean case, i.e. to zero curvature. Our proof was heavily based on a density argument using polynomials and polynomials were obtained by differentiating the kernel of the Riesz potential. Our preliminary calculations suggest that we cannot obtain all the polynomials even in the constant negative curvature case. In fact the procedure fails in the very first steps: we cannot even construct polynomials of order 2. Therefore we would need a different approach if we wanted to generalize our result to non-Euclidean manifolds.

There is another proof for theorem 1.2 which is based on spherical symmetry and angular Fourier series (see section 5.3). This method could perhaps generalize to some sort of spherically symmetric manifolds but it is not studied in a great detail yet. The big problem of general manifolds is that one cannot do explicit calculations. Especially we would need to express the Chebyshev polynomials in a nice form and show properties of them. The integral kernel is known in the conformally Euclidean case [7]. However, the issue becomes to calculate the derivatives of the kernel up to any order since the idea in the alternative proof is also to obtain all the polynomials and use density.

In section 1.2 we studied the applications of our results to seismology. We discussed about a model where we measure arrival time differences of split S-waves in a thin annulus. We did a linearization of the anisotropies of the S-wave speeds in isotropic background and made an (artificial) assumption that the difference of the perturbations is independent of direction of propagation. One could also consider a more general linearization in the elastic theory. This means that we have a known isotropic elastic model and a small anisotropic perturbation in the stiffness tensor  $c_{ijkl}$  to be determined from travel time measurements. It is shown in [46] that this kind of linearization leads to the X-ray tomography problem of a tensor field of degree 4 for P-waves. For S-waves one needs to study the so-called mixed ray transform of tensors of arbitrary order on 2-dimensional compact simple Riemannian manifolds with boundary [9]. But there are no known partial data results for the mixed ray transform. These would be highly beneficial and interesting considering applications in seismology.

If one treats the annulus as a thin layer with respect to the radius of the Earth ("flat Earth"), the situation resembles the X-ray tomography problem in a periodic slab  $[0, \epsilon] \times \mathbb{T}^2$ ,  $\epsilon > 0$ . There is a kernel characterization for the X-ray transform of  $L^2$ -regular tensor fields of any order on periodic slabs of type  $[0, 1] \times \mathbb{T}^d$  where d is any non-negative integer [20]. In particular the X-ray transform has a nontrivial kernel even for scalar fields in contrast to our result.

#### 3. INTEGRAL GEOMETRY AND DISTRIBUTIONS

3.1. Distribution theory. Let us review some basic distribution theory. A more detailed treatment can be found in a number of introductory books on distribution theory and functional analysis, e.g. [18, 31, 40, 43, 51]. This introduction is included for the benefit of readers less familiar with the theory and for the sake of easy reference later on. All the lemmas of this subsection are either well known or trivial and are therefore not proven.

Consider an open domain  $\Omega \subset \mathbb{R}^d$ . We denote by  $\mathcal{E}(\Omega)$  the space of all smooth functions  $\Omega \to \mathbb{C}$  and by  $\mathcal{D}(\Omega)$  the subspace consisting of compactly supported functions. These spaces are equipped with the topology of uniform convergence of derivatives of any order on compact sets. The topological duals of these function spaces are denoted by  $\mathcal{E}'(\Omega)$  and  $\mathcal{D}'(\Omega)$ , respectively, and their elements are called distributions. The space  $\mathcal{E}'(\Omega)$  can be identified with the subspace of  $\mathcal{D}'(\Omega)$  consisting of compactly supported distributions.

A multi-index  $\beta = (\beta_1, \ldots, \beta_d) \in \mathbb{N}^d$  is a *d*-tuplet of natural numbers. We use the convention that  $0 \in \mathbb{N}$ . We write  $|\beta| \coloneqq \beta_1 + \ldots + \beta_d$  and

$$\partial^{\beta} = \left(\frac{\partial}{\partial x_1}\right)^{\beta_1} \cdots \left(\frac{\partial}{\partial x_d}\right)^{\beta_d}$$

The distributional derivative of order  $\beta$  of  $u \in \mathcal{D}'(\Omega)$  is defined so that

$$\left\langle \partial^{\beta} u, \varphi \right\rangle = (-1)^{|\beta|} \left\langle u, \partial^{\beta} \varphi \right\rangle$$

for all  $\varphi \in \mathcal{D}(\Omega)$  and similarly for  $\mathcal{D}'$  and  $\mathcal{D}$  replaced with  $\mathcal{E}'$  and  $\mathcal{E}$ .

The value of a distribution evaluated at a test function only depends on the values of the test functions in the support of the distribution as stated in the next lemma.

**Lemma 3.1.** Let  $\Omega \subset \mathbb{R}^d$  be open and  $u \in \mathcal{E}'(\Omega)$ . If  $\psi_1, \psi_2 \in \mathcal{E}(\Omega)$  are such that  $\psi_1|_{\operatorname{spt}(u)} = \psi_2|_{\operatorname{spt}(u)}$ , then  $\langle u, \psi_1 \rangle = \langle u, \psi_2 \rangle$ . The corresponding result also holds with  $\mathcal{E}'$  and  $\mathcal{E}$  replaced with  $\mathcal{D}'$  and  $\mathcal{D}$ .

It will be convenient to make a change of variables for distributions. Let  $F: \Omega_1 \to \Omega_2$  be a  $C^{\infty}$ -diffeomorphism between two domains  $\Omega_1, \Omega_2 \subset \mathbb{R}^d$ . The pullback  $F^*u = u \circ F \in \mathcal{D}'(\Omega_1)$  of  $u \in \mathcal{D}'(\Omega_2)$  is defined so that

$$\langle u \circ F, \varphi \rangle = \langle u, (\varphi \circ F^{-1}) | J_{F^{-1}} | \rangle$$

for all  $\varphi \in \mathcal{D}(\Omega_1)$ . Here  $|J_{F^{-1}}|$  denotes the absolute value of the Jacobian determinant of  $F^{-1}$ . The same definition can be applied to  $u \in \mathcal{E}'(\Omega_2)$  with  $\varphi \in \mathcal{E}(\Omega_1)$ . The supports behave naturally under pullbacks as stated in the next lemma.

**Lemma 3.2.** Let  $\Omega_1, \Omega_2 \subset \mathbb{R}^d$  be open and  $F: \Omega_1 \to \Omega_2$  be a  $C^{\infty}$ -diffeomorphism. If  $u \in \mathcal{D}'(\Omega_2)$ , then  $\operatorname{spt}(u \circ F) = F^{-1}(\operatorname{spt}(u))$ . In particular, u = 0 if and only if  $u \circ F = 0$ .

We will make use of the Kelvin transform or the inversion  $K \colon \mathbb{R}^d \setminus \{0\} \to \mathbb{R}^d \setminus \{0\}$  given by  $K(x) = |x|^{-2} x$ . The Kelvin transform is its own inverse.

Any element of the spaces  $\mathcal{E}(\Omega)$ ,  $\mathcal{E}'(\Omega)$ ,  $\mathcal{D}(\Omega)$ , and  $\mathcal{D}'(\Omega)$  can be multiplied by an element of  $\mathcal{E}(\Omega)$ . Such multiplication has an injectivity property we will need:

**Lemma 3.3.** Let  $\Omega \subset \mathbb{R}^d$  be open,  $u \in \mathcal{E}'(\Omega)$  and  $g \in C^{\infty}(\Omega)$  such that  $g \neq 0$  in  $\operatorname{spt}(u)$ . Then u = 0 if and only if gu = 0.

For test functions  $\varphi \in \mathcal{E}(\mathbb{R}^d)$  we define translation  $\tau_{x_0}$  by  $x_0 \in \mathbb{R}^d$  so that  $(\tau_{x_0}\varphi)(x) = \varphi(x-x_0)$ . The reflection  $\tilde{\varphi}$  is defined by  $\tilde{\varphi}(x) = \varphi(-x)$ . Naturally  $\tau_{x_0}\varphi, \tilde{\varphi} \in \mathcal{E}(\mathbb{R}^d)$ . Translations and reflections can be defined on distributions by duality.

Convolutions can also be defined for distributions (see e.g. [43]):

**Lemma 3.4.** Let  $u \in \mathcal{D}'(\mathbb{R}^d)$  and  $\varphi \in \mathcal{D}(\mathbb{R}^d)$ . Then  $u * \varphi$  has a representative  $g_1 \in \mathcal{E}(\mathbb{R}^d)$  which is given by the formula  $g_1(x) = \langle u, \tau_x \widetilde{\varphi} \rangle$ . Additionally, if  $v \in \mathcal{E}'(\mathbb{R}^d)$ , then  $v * \varphi$  has a representative  $g_2 \in \mathcal{D}(\mathbb{R}^d)$  which is given by the formula  $g_2(x) = \langle v, \tau_x \widetilde{\varphi} \rangle$ .

**Lemma 3.5.** Let  $u \in \mathcal{E}'(\mathbb{R}^d)$  and  $v \in \mathcal{D}'(\mathbb{R}^d)$ . Then  $u * v \in \mathcal{D}'(\mathbb{R}^d)$  is defined via the formula

$$\langle u \ast v, \varphi \rangle = \langle u, \widetilde{v} \ast \varphi \rangle$$

for all  $\varphi \in \mathcal{D}(\mathbb{R}^d)$ , and for every  $\beta \in \mathbb{N}^d$  the derivatives satisfy

$$\partial^{\beta}(u * v) = (\partial^{\beta}u) * v = u * (\partial^{\beta}v)$$

in the sense of distributions.

3.2. Integral geometry and the normal operator. In this section we introduce basic theory of integral geometry in  $\mathbb{R}^d$ . For this we mainly follow the books [16, 34, 48], see also [38]. We define the Riesz potential  $I_{\alpha}$  and discuss about its connection to the normal operator of the X-ray transform N.

Denote by  $\Gamma$  the set of all oriented lines in  $\mathbb{R}^d$ . The X-ray transform of a function f is the map  $Xf \colon \Gamma \to \mathbb{R}$ ,

$$Xf(\gamma) = \int_{\gamma} f \mathrm{d}s$$

for all lines  $\gamma \in \Gamma$  assuming that the integrals exists. The integrals are finite whenever f decays fast enough at infinity. If the lines are parametrized by the set

$$\{(z,\theta): \theta \in S^{d-1}, \ z \in \theta^{\perp}\},\$$

the X-ray transform may be written as

$$Xf(z,\theta) = \int_{\mathbb{R}} f(z+s\theta) \mathrm{d}s.$$

It is a continuous linear map  $X : \mathcal{D}(\mathbb{R}^d) \to \mathcal{D}(\Gamma)$ . The set  $\Gamma$  can be freely identified with  $TS^{d-1}$ . As  $\Gamma$  is a smooth manifold, the test function and distribution spaces on it can be defined similarly to the Euclidean setting.

The formal adjoint  $X^* \colon \mathcal{E}(\Gamma) \to \mathcal{E}(\mathbb{R}^d)$  is given by

$$X^*\psi(x) = \int_{S^{d-1}} \psi(x - (x \cdot \theta)\theta, \theta) \mathrm{d}\theta.$$

The function  $\psi$  can be interpreted as a function in the set of all lines. The value  $X^*\psi(x)$  is obtained by integrating  $\psi$  over all lines going through the point x. The formal adjoint does not preserve compact supports, but the integrals in its definition are taken over compact sets.

The operators X and  $X^*$  can be defined on distributions by duality. That is,  $X: \mathcal{E}'(\mathbb{R}^d) \to \mathcal{E}'(\Gamma)$  and  $X^*: \mathcal{D}'(\Gamma) \to \mathcal{D}'(\mathbb{R}^d)$  are defined so that they satisfy

$$\langle Xf,\eta\rangle = \langle f,X^*\eta\rangle$$

for all  $f \in \mathcal{E}'(\mathbb{R}^d)$  and  $\eta \in \mathcal{E}(\Gamma)$ , and

$$\langle X^*g,\varphi\rangle = \langle g,X\varphi\rangle$$

for all  $g \in \mathcal{D}'(\Gamma)$  and  $\varphi \in \mathcal{D}(\mathbb{R}^d)$ . We say that Xf vanishes on all lines which intersect an open set V, if  $Xf|_{\Gamma_V} = 0$  as a distribution where  $\Gamma_V$  is the set of all parametrized lines intersecting V.

It is often convenient to study the X-ray transform X by way of its normal operator  $N = X^*X$ . This is not suited for all partial data scenarios and our proof in section 5.3 works directly at the level of X, but we make use of the normal operator elsewhere. Due to the mapping properties established above, the normal operator maps  $N: \mathcal{E}'(\mathbb{R}^d) \to \mathcal{D}'(\mathbb{R}^d)$ . It is a pseudodifferential operator of order -1, but our problem is not well suited for a microlocal approach as discussed in section 2.4. For a test function  $f \in \mathcal{D}(\mathbb{R}^d)$  the normal operator can be expressed conveniently as [48]

(1) 
$$Nf(x) = 2 \int_{\mathbb{R}^d} \frac{f(y)}{|x-y|^{d-1}} dy = 2(f * |\cdot|^{1-d})(x).$$

The convolution formula holds for a distribution  $f \in \mathcal{E}'(\mathbb{R}^d)$  by a duality argument, and it holds also for continuous functions which decrease rapidly enough at infinity [16].

The normal operator of the X-ray transform can be inverted by the formula [48]

(2) 
$$f = c_d (-\Delta)^{1/2} N f, \quad c_d = (2\pi |S^{d-2}|)^{-1}$$

for any  $f \in \mathcal{E}'(\mathbb{R}^d)$ . Here the fractional Laplacian  $(-\Delta)^s$  is defined via the inverse Fourier transform  $(-\Delta)^s f = \mathcal{F}^{-1}(|\cdot|^{2s} \hat{f})$  and it is a non-local operator. As can be seen in equation (2), the normal operator of the X-ray transform is essentially  $(-\Delta)^{-1/2}$  and is inverted by  $(-\Delta)^{1/2}$ .

Let  $h_{\alpha}(x) = |x|^{-\alpha}$  where  $\alpha = d - 1$  or  $\alpha \in \mathbb{R} \setminus \mathbb{Z}$  and  $\alpha < d$ . We define the Riesz potential  $I_{\alpha} \colon \mathcal{E}'(\mathbb{R}^d) \to \mathcal{D}'(\mathbb{R}^d)$  as

(3) 
$$\langle I_{\alpha}f,\varphi\rangle = \langle f*h_{\alpha},\varphi\rangle$$

for all  $f \in \mathcal{E}'(\mathbb{R}^d)$  and  $\varphi \in \mathcal{D}(\mathbb{R}^d)$ . When  $\alpha < d$  then  $h_\alpha$  is locally integrable and thus defines a tempered distribution. The convolution between two distributions is well-defined when at least one of them has compact support. This implies that  $I_\alpha f$ is always defined as a distribution when  $f \in \mathcal{E}'(\mathbb{R}^d)$ . Especially if  $f \in \mathcal{D}(\mathbb{R}^d)$ , then

$$I_{\alpha}f(x) = \int_{\mathbb{R}^d} \frac{f(y)}{|x-y|^{\alpha}} \mathrm{d}y$$

We call  $h_{\alpha}$  the kernel of the Riesz potential  $I_{\alpha}$ . If  $\alpha = d - 1$ , then equation (3) defines the normal operator of the X-ray transform N up to a constant factor 2, see equation (1). Extensive treatment of Riesz potentials can be found in many books, see e.g. [16, 27, 32, 44].

#### 4. Proof of Lemma 2.2

In this section we give a rather technical proof of lemma 2.2. The proof is based on induction and algebraic relations between certain functions and their derivatives.

Proof of lemma 2.2. We need to show that if  $d \ge 2$  and  $\alpha > d-2$  or  $\alpha \in \mathbb{R} \setminus \mathbb{Z}$ , then for any polynomial p one can express  $p(x|x|^{-2})|x|^{-\alpha}$  as a finite linear combination of derivatives of  $h_{\alpha}(x) = |x|^{-\alpha}$ . Let us denote

$$A_i = x_i B, \quad B = |x|^{-2} \text{ and } C = |x|^{-\alpha}.$$

Then one can calculate the relations

$$\partial_j A_i = \delta_{ij} B - 2A_i A_j, \quad |A|^2 = B \text{ and } \partial_i C = -\alpha A_i C.$$

Let us also define

$$D_{i_1\dots i_n} = A_{i_1} \cdot \dots \cdot A_{i_n} \cdot C = \left(\prod_{l=1}^n A_{i_l}\right)C, \quad i_k \in \{1,\dots,d\}.$$

We would like to express  $D_{i_1...i_n}$  for all  $n \in \mathbb{N}$  as a finite linear combination of derivatives of  $h_{\alpha}$ . The constant polynomials are given by  $h_{\alpha}$  itself. The first derivative is

$$\partial_i h_\alpha(x) = -\alpha x_i \left| x \right|^{-\alpha - 2} = -\alpha D_i.$$

Whence  $D_i$  can be obtained from first-order derivatives of  $h_{\alpha}$ . Differentiating  $D_i$  gives

$$\partial_j D_i = \delta_{ij} BC - (2 + \alpha) D_{ij}$$

and the divergence is

$$\sum_{i=1}^{d} \partial_i D_i = (d-2-\alpha)BC.$$

Combining these we obtain

$$D_{ij} = \frac{1}{2+\alpha} \left( \left( \frac{\delta_{ij}}{d-2-\alpha} \sum_{i=1}^d \partial_i D_i \right) - \partial_j D_i \right).$$

We have thus expressed the terms  $D_{ij}$  as a finite linear combination of the derivatives of the terms  $D_i$  which were multiples of the first-order derivatives of  $h_{\alpha}$ . Hence  $D_{ij}$  can be expressed as a finite linear combination of second-order derivatives of  $h_{\alpha}$ .

We claim that  $D_{i_1...i_n}$  is a finite linear combination of *n*th order derivatives of  $h_{\alpha}$  for all  $n \in \mathbb{N}$  and we have shown this for n = 0, 1, 2. The lemma follows from this claim. Let us assume that the claim holds for some  $m - 1 \in \mathbb{N}$ . Then  $D_{i_1...i_{m-1}}$  is a finite linear combination of (m-1)th order derivatives of  $h_{\alpha}$ . Thus  $\partial_{i_m} D_{i_1...i_{m-1}}$  is a finite linear combination of *m*th order derivatives of  $h_{\alpha}$  and a calculation shows that

(4) 
$$\partial_{i_m} D_{i_1 \dots i_{m-1}} = (2 - 2m - \alpha) D_{i_1 \dots i_m} + \sum_{j=1}^{m-1} \left( \delta_{i_m i_j} BC \prod_{\substack{l=1\\l \neq j}}^{m-1} A_{i_l} \right).$$

Let us then calculate the divergence from equation (4). We get

(5) 
$$\sum_{i_k=1}^d \partial_{i_k} D_{i_1...i_k...i_{m-1}} = (d-m-\alpha)BC \prod_{\substack{l=1\\l\neq k}}^{m-1} A_{i_l}.$$

From equations (4) and (5) we obtain the following expression for  $D_{i_1...i_m}$ 

$$\frac{1}{2 - 2m - \alpha} \left( \partial_{i_m} D_{i_1 \dots i_{m-1}} - \frac{1}{d - m - \alpha} \sum_{j=1}^{m-1} \left( \delta_{i_m i_j} \sum_{i_j=1}^d \partial_{i_j} D_{i_1 \dots i_j \dots i_{m-1}} \right) \right)$$

which is by the induction assumption a finite linear combination of mth order derivatives of  $h_{\alpha}$ . Thus the claim follows for all  $n \in \mathbb{N}$ .

#### 5. Alternative proofs of the main theorems

In this section we give alternative proofs to our main theorems, theorem 1.1 and theorem 1.2. We believe that presenting several proofs opens more possibilities to generalize the results and gives more tools for solving similar unique continuation problems and partial data problems.

We prove theorem 1.1 under the stronger assumption  $I_{\alpha}f|_{V} = 0$  for a slightly larger class of distributions, i.e. rapidly decreasing distributions. We do it in two alternative ways. First proof is based on convolution approximation and density of polynomials. The second approach uses the unique continuation property of the fractional Laplacian. The second proof is short since it relies on a strong result. The unique continuation of  $(-\Delta)^{s}$ ,  $s \in (0, 1)$ , is based on technical results about Carleman estimates and Caffarelli-Silvestre extensions [15].

We then prove theorem 1.2 first for compactly supported smooth functions using angular Fourier series and density argument based on differentiation of an integral kernel. By a standard mollification argument we obtain the same result for compactly supported distributions. The proof works directly at the level of the X-ray transform and does not use the normal operator at all. Therefore we do not need to use any unique continuation results in the proof of the partial data problem.

We briefly go through our notations. We denote by  $\mathscr{O}_M(\mathbb{R}^d)$  the space of polynomially increasing smooth functions, by  $\mathscr{S}'(\mathbb{R}^d)$  the space of tempered distributions, by  $\mathscr{O}'_C(\mathbb{R}^d)$  the space of rapidly decreasing distributions and by  $H^r(\mathbb{R}^d)$  the fractional  $L^2$ -Sobolev space of order  $r \in \mathbb{R}$ . For precise definitions see [18, 30, 43, 51]. For us it is enough to know that  $\mathscr{E}'(\mathbb{R}^d) \subset \mathscr{O}'_C(\mathbb{R}^d) \subset \mathscr{S}'(\mathbb{R}^d)$  and

$$\mathscr{O}'_C(\mathbb{R}^d) \subset \bigcup_{r \in \mathbb{R}} H^r(\mathbb{R}^d).$$

Rapidly decreasing continuous functions, i.e. continuous functions which decrease faster than any polynomial at infinity, are contained in  $\mathscr{O}'_{C}(\mathbb{R}^{d})$ . The convolution operator \* is a separately continuous map  $*: \mathscr{O}'_{C}(\mathbb{R}^{d}) \times \mathscr{S}'(\mathbb{R}^{d}) \to \mathscr{S}'(\mathbb{R}^{d})$ . This implies that the Riesz potential  $I_{\alpha}f = f * |\cdot|^{-\alpha}$  is defined as a distribution when  $f \in \mathscr{O}'_{C}(\mathbb{R}^{d})$  and  $\alpha < d$ . The Fourier transform is a bijective map from  $\mathscr{O}'_{C}(\mathbb{R}^{d})$  onto  $\mathscr{O}_{M}(\mathbb{R}^{d})$  and the usual convolution formula  $\widehat{f * g} = \widehat{f} \cdot \widehat{g}$  holds in the sense of distributions when  $f \in \mathscr{O}'_{C}(\mathbb{R}^{d})$  and  $g \in \mathscr{S}'(\mathbb{R}^{d})$ .

5.1. Using convolution approximation. In this section we prove theorem 1.1 under the assumption  $I_{\alpha}f|_{V} = 0$  first for Schwartz functions. The result follows also for rapidly decreasing distributions by considering the mollifications  $f * j_{\epsilon}$ .

**Theorem 5.1.** Let  $\alpha = d-1$  or  $\alpha \in \mathbb{R} \setminus \mathbb{Z}$  and  $\alpha < d$ . Let  $f \in \mathscr{O}'_C(\mathbb{R}^d)$  and  $V \subset \mathbb{R}^d$  any nonempty open set. If  $f|_V = I_\alpha f|_V = 0$ , then f = 0.

*Proof.* We can assume that  $0 \in V$ . Let first  $f \in \mathscr{S}(\mathbb{R}^d)$ . Like in the proof of theorem 1.1 we smoothen the kernel  $h_{\alpha}$  near the origin, let this smoothened version be  $g \in C^{\infty}(\mathbb{R}^d)$ . There is  $\epsilon > 0$  such that  $(f * g)|_{B(0,\epsilon)} = (f * h_{\alpha})|_{B(0,\epsilon)}$ . It holds that  $\partial^{\beta}(f * g) = f * \partial^{\beta}g$  where by lemma 2.2 one obtains all the polynomials p in the form  $p(K(x))h_{\alpha}(x)$  by taking finite linear combinations of  $\partial^{\beta}g$ . Since f is not supported in a ball B centered at the origin, we can use the Kelvin transform to obtain

$$0 = \int_{B^c} f(y) p(y|y|^{-2}) |y|^{-\alpha} \, \mathrm{d}y = \int_{\widetilde{B} \setminus \{0\}} f(x|x|^{-2}) p(x) |x|^{\alpha} |J_K(x)| \, \mathrm{d}x$$

where  $\widetilde{B}$  is some closed ball centered at the origin. One can calculate that  $|J_K(x)| = |x|^{-2d}$  (see [15, Remark 4.2]). Since f goes rapidly to zero at infinity, we can extend the function  $x \mapsto f(x |x|^{-2}) |x|^{\alpha} |J_K(x)|$  continuously to zero and we call this extension  $\widetilde{f}$ . We obtain

$$\int_{\widetilde{B}} \widetilde{f}(x) p(x) \mathrm{d}x = 0$$

for all polynomials p. Since  $\tilde{f}$  is continuous and  $\tilde{B}$  is compact, by the Stone-Weierstrass theorem  $\tilde{f} = 0$ . This implies f = 0.

Then let  $f \in \mathcal{O}'_C(\mathbb{R}^d)$ . Denote by  $j_{\epsilon} \in \mathcal{D}(\mathbb{R}^d)$  the standard mollifier and consider the mollifications  $f_{\epsilon} = f * j_{\epsilon} \in \mathscr{S}(\mathbb{R}^d)$ . Since  $I_{\alpha}(f * j_{\epsilon}) = I_{\alpha}f * j_{\epsilon}$  it follows that  $f_{\epsilon}|_W = I_{\alpha}f_{\epsilon}|_W = 0$  for small enough  $\epsilon > 0$  and  $W \subset V$  open. By the first part of the proof  $f_{\epsilon} = 0$  for small  $\epsilon > 0$ . This implies f = 0 since  $f_{\epsilon} \to f$  as distributions in  $\mathscr{S}'(\mathbb{R}^d)$  when  $\epsilon \to 0$ .

We remark that theorem 5.1 implies uniqueness for the partial data problem (theorem 1.2) when f is a continous function which decreases faster than any polynomial. We can thus relax the assumption of compact support to rapid decay at infinity in theorem 1.2.

5.2. Using unique continuation of the fractional Laplacian. Here we give an alternative proof for a modified version of theorem 1.1 using Fourier analysis and unique continuation of  $(-\Delta)^s$  in  $H^r(\mathbb{R}^d)$ ,  $r \in \mathbb{R}$ , when 0 < s < 1. The unique continuation of  $(-\Delta)^s$  is proved in [15].

**Theorem 5.2.** Let  $f \in \mathcal{O}'_C(\mathbb{R}^d)$ ,  $V \subset \mathbb{R}^d$  any nonempty open set and  $0 < \alpha < d$  such that  $(\alpha - d)/2 \notin \mathbb{Z}$ . If  $f|_V = 0$  and  $I_{\alpha}f|_V = 0$ , then f = 0.

*Proof.* There is  $k \in \mathbb{N}$  such that  $-k < (\alpha - d)/2 < -k + 1$ . Using the convolution property of the Fourier transform we can write

$$I_{\alpha}f = f * |\cdot|^{-\alpha} = c_d \mathcal{F}^{-1}(\mathcal{F}(f * |\cdot|^{-\alpha})) = c_d \mathcal{F}^{-1}(\hat{f} |\cdot|^{\alpha-d}) = c_d(-\Delta)^{\frac{\alpha-d}{2}}f,$$

where  $c_d > 0$  is a constant depending on dimension. Since  $(-\Delta)^{\frac{\alpha-d}{2}}f$  is a tempered distribution, again by the properties of the Fourier transform it follows that  $(-\Delta)^k(-\Delta)^{\frac{\alpha-d}{2}}f = (-\Delta)^{k+\frac{\alpha-d}{2}}f = (-\Delta)^s f$  where  $s = k + (\alpha - d)/2 \in (0, 1)$ . Since  $(-\Delta)^k$  is a local operator and  $(-\Delta)^{\frac{\alpha-d}{2}}f$  vanishes in the open set V, we obtain the conditions  $f|_V = 0$  and  $(-\Delta)^s f|_V = 0$ . Now  $f \in \mathscr{O}'_C(\mathbb{R}^d)$  which implies  $f \in H^r(\mathbb{R}^d)$  for some  $r \in \mathbb{R}$ . By [15, Theorem 1.2] we obtain f = 0.

We remark that theorem 5.2 implies the unique continuation of the normal operator of the X-ray transform in dimensions  $d \ge 2$  since in that case  $0 < d-1 = \alpha < d$ and  $(\alpha - d)/2 = -1/2 \notin \mathbb{Z}$ .

5.3. Angular Fourier series approach. In this section we give another proof of theorem 1.2. We assume without loss of generality that f is supported in  $\overline{B}(0, R') \setminus B(0, R)$  for some R' > R > 0 and that  $0 \in V$ . The proof is based on a similar idea as before, differentiation of an integral kernel and density of polynomials. However, now we study the X-ray transform directly and exploit the underlying spherical symmetry by using angular Fourier series expansion.

In the next theorem, when  $f \in C_c(\mathbb{R}^d)$  it would be enough to assume that the X-ray transform Xf vanishes to infinite order on all lines through the origin, i.e.  $\partial_r^n(Xf)(r,\theta)|_{r=0} = 0$  for all  $n \in \mathbb{N}$ . This is a similar assumption that we used in theorem 1.1.

**Theorem 5.3.** Fix any  $0 < \epsilon < R < R'$ . Let  $f \in \mathcal{E}'(\mathbb{R}^d)$  such that  $\operatorname{spt}(f) \subset \overline{B}(0, R') \setminus B(0, R)$ . If f integrates to zero over all lines in B(0, R') that meet  $B(0, \epsilon)$ , then f = 0.

*Proof.* Without loss of generality we can assume that R' = 1. Let first  $f \in C_c(\mathbb{R}^d)$ . By intersecting the origin with 2-planes it is enough to prove the result in two dimensions. The function f can be expressed as an angular Fourier series

$$f(r,\theta) = \sum_{k \in \mathbb{Z}} e^{ik\theta} a_k(r).$$

Our goal is to show that  $a_k = 0$  for all  $k \in \mathbb{Z}$ . When we parameterize the lines in  $\mathbb{R}^2$  by their closest point to the origin and use polar coordinates for these points, we find

$$Xf(r,\theta) = \sum_{k \in \mathbb{Z}} e^{ik\theta} \mathcal{A}_{|k|} a_k(r),$$

where  $\mathcal{A}_k$  is the generalized Abel transform defined by

(6) 
$$\mathcal{A}_k g(z) = 2 \int_z^1 K_k(z, y) g(y) \mathrm{d}y$$

Here the kernel is  $K_k(z, y) = T_k(z/y)[1 - (z/y)^2]^{-1/2}$  and  $T_k$  are the Chebyshev polynomials.

We know that  $f(r, \theta) = 0$  when r < R and  $Xf(r, \theta) = 0$  when  $r < \varepsilon$ . For the Fourier components  $a_k(r)$  this means that for every  $k \in \mathbb{Z}$  we have  $a_k(r) = 0$  for r < R and  $\mathcal{A}_k a_k(r) = 0$  for  $r < \varepsilon$ . Hence we get

(7) 
$$\int_{R}^{1} K_k(z, y) a_k(y) \mathrm{d}y = 0$$

for every  $z \in [0, \varepsilon)$ . Like in the proof of theorem 1.1, we differentiate the integral kernel *n* times in (7) with respect to *z* and evaluate at z = 0 to obtain

(8) 
$$\int_{R}^{1} D_{k}^{n}(y) a_{k}(y) \mathrm{d}y = 0$$

for all  $n \in \mathbb{N}$  and  $k \in \mathbb{Z}$ , where  $D_k^n(y) = \partial_z^n K_k(z, y)|_{z=0}$ .

By scaling arguments  $D_k^n(y) = A_k^n y^{-n}$  for some numbers  $A_k^n$ . The term k = 0 is

(9) 
$$A_0^n = \begin{cases} (n-1)!!^2, & n \text{ even}, \\ 0, & n \text{ odd.} \end{cases}$$

We denote the coefficient of  $x^l$  in  $T_k(x)$  by  $t_k^l$ . The *l*th derivative of  $T_k(x)$  at x = 0 is  $l!t_k^l$ . The coefficients also satisfy

(10) 
$$\sum_{l=0}^{k} t_{k}^{l} = T_{k}(1) = 1$$

By basic properties of Chebyshev polynomials  $t_k^l = 0$  if l - k is odd or l > k. Using  $K_k(z, y) = T_k(z/y)K_0(z, y)$  and the product rule of higher order derivatives we find

$$A_k^n = \sum_{l=0}^n \binom{n}{l} l! t_k^l A_0^{n-l}.$$

By parity properties it is clear that  $A_k^n$  vanishes unless both n and l are even or both are odd.

We will show that for any  $k \in \mathbb{N}$  there is a number N(k) so that  $A_k^n > 0$  when  $n \geq N(k)$  and parity is right. For k = 0 this follows from equation (9) with N(k) = 0. Consider first the case when n and k are both even and assume n > k. A calculation shows that

$$A_k^n = n! \frac{(n-1)!!}{n!!} \sum_{m=0}^{k/2} \left[ t_k^{2m} + t_k^{2m} \left( \frac{(n-2m-1)!!n!!}{(n-2m)!!(n-1)!!} - 1 \right) \right]$$

There are only finitely many terms in the sum, and for every m we have

$$\lim_{n \to \infty} \frac{(n - 2m - 1)!!n!!}{(n - 2m)!!(n - 1)!!} = 1.$$

Equation (10) implies  $\sum_{m=0}^{k/2} t_k^{2m} = 1$  so that

$$\lim_{n \to \infty} \sum_{m=0}^{k/2} \left[ t_k^{2m} + t_k^{2m} \left( \frac{(n-2m-1)!!n!!}{(n-2m)!!(n-1)!!} - 1 \right) \right] = 1.$$

Therefore  $A_k^n > 0$  for sufficiently large n as claimed. Similarly one can show for odd indices that

$$A_k^n = n! \frac{(n-2)!!}{(n-1)!!} \sum_{m=0}^{(k-1)/2} \left[ t_k^{2m+1} + t_k^{2m+1} \left( \frac{(n-2m-2)!!(n-1)!!}{(n-2m-1)!!(n-2)!!} - 1 \right) \right].$$

With the same limit argument we get  $A_k^n > 0$  for large n.

We fix any  $k \in \mathbb{Z}$  and use (8) to show that  $a_k = 0$ . By symmetry it suffices to consider  $k \ge 0$ . We found N(k) so that  $A_k^n \ne 0$  for  $n \ge N(k)$  when n - N(k) is even. We find

$$\int_{R}^{1} y^{-N(k)-2m} a_k(y) \mathrm{d}y = 0$$

19

for every  $m \in \mathbb{N}$ . By linearity

$$\int_{R}^{1} y^{-N(k)} p(y^{-2}) a_{k}(y) \mathrm{d}y = 0$$

for any polynomial p. Changing variable to  $s = y^{-2}$  and defining new coefficients  $\tilde{a}_k(s) = s^{N(k)/2-3/2} a_k(s^{-1/2})$ , we obtain

$$\int_{1}^{R^{-1/2}} p(s)\tilde{a}_k(s)\mathrm{d}s = 0.$$

By density of polynomials  $\tilde{a}_k(s) = 0$  for all  $s \in [1, R^{-1/2}]$ . This implies  $a_k = 0$  for all  $k \in \mathbb{Z}$  and hence f = 0.

Then let  $f \in \mathcal{E}'(\mathbb{R}^d)$  and consider the mollifications  $f * j_{\epsilon} \in \mathcal{D}(\mathbb{R}^d)$ . Following Helgason [16] we define the "convolution"

$$(g imes arphi)(z, heta) = \int_{\mathbb{R}^d} g(y) arphi(z - y, heta) \mathrm{d}y$$

where  $g \in \mathcal{D}(\mathbb{R}^d)$  and  $\varphi \in \mathcal{D}(\Gamma)$ . By a simple calculation one can show that  $X^*(g \times \varphi) = g * X^* \varphi$ . Using the properties of the convolutions \* and  $\times$  we obtain

$$\langle X(f*j_{\epsilon}),\varphi\rangle = \langle f*j_{\epsilon},X^*\varphi\rangle = \langle f,j_{\epsilon}*X^*\varphi\rangle = \langle f,X^*(j_{\epsilon}\times\varphi)\rangle = \langle Xf,j_{\epsilon}\times\varphi\rangle.$$

Thus for small enough  $\epsilon > 0$  and  $\tilde{R} > 0$  we get that  $(f * j_{\epsilon})|_{B(0,\tilde{R})} = 0$  and  $X(f * j_{\epsilon})$ vanishes on all lines which intersect  $B(0,\epsilon)$ . The first part of the proof implies  $f * j_{\epsilon} = 0$  for small  $\epsilon > 0$ . The claim follows since  $f * j_{\epsilon} \to f$  in  $\mathcal{E}'(\mathbb{R}^d)$  when  $\epsilon \to 0$ .

We remark that the assumption that f is supported away from the origin is crucial since it turns a Volterra integral equation into a Fredholm integral equation. This simplifies the derivatives of expression (6).

#### References

- G. E. Backus. Long-wave elastic anisotropy produced by horizontal layering. Journal of Geophysical Research (1896–1977), 67(11):4427–4440, 1962.
- J. Boman. An example of non-uniqueness for a generalized Radon transform. Journal d'Analyse Mathématique, 61(1):395-401, 1993.
- [3] J. Boman and E. T. Quinto. Support theorems for real-analytic Radon transforms. Duke Math. J., 55(4):943–948, 1987.
- [4] M. Courdurier, F. Noo, M. Defrise, and H. Kudo. Solving the interior problem of computed tomography using a priori knowledge. Inverse Problems, 24(6):065001, 2008.
- [5] S. Crampin and J. H. Lovell. A decade of shear-wave splitting in the Earth's crust: what does it mean? what use can we make of it? and what should we do next? *Geophysical Journal International*, 107(3):387–407, 1991.
- [6] K. C. Creager. Anisotropy of the inner core from differential travel times of the phases PKP and PKIKP. *Nature*, 356:309–314, 1992.
- [7] M. V. de Hoop and J. Ilmavirta. Abel transforms with low regularity with applications to Xray tomography on spherically symmetric manifolds. *Inverse Problems*, 33(12):124003, 2017.
- [8] M. V. de Hoop, J. Ilmavirta, and V. Katsnelson. Spectral rigidity for spherically symmetric manifolds with boundary. 2017. arXiv:1705.10434.
- [9] M. V. de Hoop, T. Saksala, and J. Zhai. Mixed ray transform on simple 2-dimensional Riemannian manifolds. Proc. Amer. Math. Soc., 2019. Published electronically.
- [10] A. M. Dziewonski and D. L. Anderson. Preliminary reference Earth model. Physics of the Earth and Planetary Interiors, 25(4):297–356, 1981.
- [11] M. M. Fall and V. Felli. Unique continuation property and local asymptotics of solutions to fractional elliptic equations. *Comm. Partial Differential Equations*, 39(2):354–397, 2014.
- [12] V. Felli and A. Ferrero. Unique continuation principles for a higher order fractional Laplace equation. 2018. arXiv:1809.09496.
- [13] M.-Á. García-Ferrero and A. Rüland. Strong unique continuation for the higher order fractional Laplacian. *Mathematics in Engineering*, 1(4):715–774, 2019.

- [14] T. Ghosh, A. Rüland, M. Salo, and G. Uhlmann. Uniqueness and reconstruction for the fractional Calderón problem with a single measurement. 2018. arXiv:1801.04449.
- [15] T. Ghosh, M. Salo, and G. Uhlmann. The Calderón problem for the fractional Schrödinger equation. 2016. arXiv:1609.09248.
- [16] S. Helgason. Integral Geometry and Radon Transforms. Springer, First edition, 2011.
- [17] G. Herglotz. Über die Elastizität der Erde bei Berücksichtigung ihrer variablen Dichte. Zeitschr. für Math. Phys., 52:275–299, 1905.
- [18] J. Horváth. Topological Vector Spaces and Distributions. volume I. Addison-Wesley, 1966.
- [19] J. Ilmavirta and F. Monard. Integral geometry on manifolds with boundary and applications. In R. Ramlau and O. Scherzer, editors, *The Radon Transform: The First 100 Years and Beyond.* de Gruyter, 2019.
- [20] J. Ilmavirta and G. Uhlmann. Tensor tomography in periodic slabs. Journal of Functional Analysis, 275(2):288–299, 2018.
- [21] E. Katsevich, A. Katsevich, and G. Wang. Stability of the interior problem with polynomial attenuation in the region of interest. *Inverse Problems*, 28(6):065022, 2012.
- [22] E. Klann, E. T. Quinto, and R. Ramlau. Wavelet methods for a weighted sparsity penalty for region of interest tomography. *Inverse Problems*, 31(2):025001, 22, 2015.
- [23] V. P. Krishnan. A support theorem for the geodesic ray transform on functions. J. Fourier Anal. Appl., 15(4):515–520, 2009.
- [24] V. P. Krishnan and E. T. Quinto. Microlocal Analysis in Tomography. In O. Scherzer, editor, Handbook of Mathematical Methods in Imaging, pages 847–902. Springer, New York, 2015.
- [25] P. Kuchment, K. Lancaster, and L. Mogilevskaya. On local tomography. *Inverse Problems*, 11(3):571–589, 1995.
- [26] M. Kwaśnicki. Ten equivalent definitions of the fractional Laplace operator. Fractional Calculus and Applied Analysis, 20, 2015.
- [27] N. Landkof. Foundations of Modern Potential Theory. Springer-Verlag, Berlin-Heidelberg-New York, first edition, 1972. Translated from the Russian by A. P. Doohovskoy.
- [28] M. D. Long and P. G. Silver. Shear Wave Splitting and Mantle Anisotropy: Measurements, Interpretations, and New Directions. Surveys in Geophysics, 30(4):407–461, 2009.
- [29] V. Maupin and J. Park. Theory and Observations Wave Propagation in Anisotropic Media. Treatise on Geophysics, 1:289–321, 2007.
- [30] W. McLean. Strongly Elliptic Systems and Boundary Integral Equations. Cambridge University Press, First edition, 2000.
- [31] D. Mitrea. Distributions, Partial Differential Equations, and Harmonic Analysis. Springer, New York, First edition, 2013.
- [32] Y. Mizuta. Potential theory in Euclidean spaces. GAKUTO International Series, Mathematical Sciences and Applications, volume 6, Gakkōtosho, Tokyo, 1996.
- [33] F. Monard, P. Stefanov, and G. Uhlmann. The Geodesic Ray Transform on Riemannian Surfaces with Conjugate Points. *Communications in Mathematical Physics*, 337(3):1491– 1513, 2015.
- [34] F. Natterer. The Mathematics of Computerized Tomography. SIAM, Philadelphia, 2001. Reprint.
- [35] E. Quinto. Singularities of the X-Ray Transform and Limited Data Tomography in ℝ<sup>2</sup> and ℝ<sup>3</sup>. SIAM Journal on Mathematical Analysis, 24(5):1215–1225, 1993.
- [36] E. Quinto. An Introduction to X-ray tomography and Radon Transforms. Proceedings of Symposia in Applied Mathematics, 63:1–23, 2006.
- [37] E. Quinto. Artifacts and Visible Singularities in Limited Data X-Ray Tomography. Sensing and Imaging, 18, 2017.
- [38] A. G. Ramm and A. I. Katsevich. The Radon Transform and Local Tomography. CRC Press, Boca Raton, First edition, 1996.
- [39] M. Riesz. Intégrales de Riemann-Liouville et potentiels. Acta Sci. Math. Szeged, 9(1-1):1–42, 1938.
- [40] W. Rudin. Functional Analysis. McGraw-Hill, Second edition, 1991.
- [41] A. Rüland. Unique continuation for fractional Schrödinger equations with rough potentials. Comm. Partial Differential Equations, 40(1):77–114, 2015.
- [42] A. Rüland and M. Salo. The fractional Calderón problem: Low regularity and stability. Nonlinear Analysis, 2019.
- [43] M. Salo. Fourier analysis and distribution theory. 2013. Lecture notes.
- [44] S. G. Samko. Hypersingular Integrals and Their Applications. CRC-Press, London and New York, first edition, 2001.
- [45] M. K. Savage. Seismic anisotropy and mantle deformation: What have we learned from shear wave splitting? *Reviews of Geophysics*, 37(1):65–106, 1999.

20

- [46] V. A. Sharafutdinov. Integral geometry of tensor fields. Inverse and Ill-posed Problems Series. VSP, Utrecht, 1994.
- [47] P. M. Shearer. Introduction to Seismology. Cambridge University Press, 3rd edition, 2019.
- [48] P. Stefanov and G. Uhlmann. Microlocal Analysis and Integral Geometry (working title). 2018. Draft version.
- [49] P. Stefanov, G. Uhlmann, and A. Vasy. Inverting the local geodesic X-ray transform on tensors. *Journal d'Analyse Mathématique*, 136(1):151–208, 2018.
- [50] P. Stefanov, G. Uhlmann, A. Vasy, and H. Zhou. Travel Time Tomography. Acta Mathematica Sinica, English Series, 35:1085–1114, 2019.
- [51] F. Trèves. Topological Vector Spaces, Distributions and Kernels. Academic Press, First edition, 1967.
- [52] G. Uhlmann and A. Vasy. The inverse problem for the local geodesic ray transform. *Invent. Math.*, 205(1):83–120, 2016.
- [53] G. Uhlmann and H. Zhou. Journey to the Center of the Earth. 2016. arXiv:1604.00630.
- [54] E. Wiechert and K. Zoeppritz. Üeber Erdbebenwellen. Nachrichten von der Gesellschaft der Wissenschaften zu Göttingen, Mathematisch-Physikalische Klasse, 4:415–549, 1907.
- [55] J. Yang, H. Yu, M. Jiang, and G. Wang. High-order total variation minimization for interior tomography. *Inverse Problems*, 26(3):035013, 2010.
- [56] R. Yang. On higher order extensions for the fractional Laplacian. 2013. arXiv:1302.4413.
- [57] Y. Ye, H. Yu, and G. Wang. Exact Interior Reconstruction from Truncated Limited-Angle Projection Data. International Journal of Biomedical Imaging, vol. 2008, 2008.
- [58] H. Yu and G. Wang. Compressed sensing based interior tomography. Physics in Medicine and Biology, 54(9):2791–2805, 2009.
# Unique continuation property and Poincaré inequality for higher order fractional Laplacians with applications in inverse problems

G. Covi, K. Mönkkönen and J. Railo

# First published in *Inverse Problems and Imaging* 15, no. 4 (2021) http://dx.doi.org/10.3934/ipi.2021009

https://arxiv.org/abs/2001.06210

Copyright © American Institute of Mathematical Sciences

[B]

# UNIQUE CONTINUATION PROPERTY AND POINCARÉ INEQUALITY FOR HIGHER ORDER FRACTIONAL LAPLACIANS WITH APPLICATIONS IN INVERSE PROBLEMS

GIOVANNI COVI, KEIJO MÖNKKÖNEN, AND JESSE RAILO

ABSTRACT. We prove a unique continuation property for the fractional Laplacian  $(-\Delta)^s$  when  $s \in (-n/2, \infty) \setminus \mathbb{Z}$  where  $n \geq 1$ . In addition, we study Poincaré-type inequalities for the operator  $(-\Delta)^s$  when  $s \geq 0$ . We apply the results to show that one can uniquely recover, up to a gauge, electric and magnetic potentials from the Dirichlet-to-Neumann map associated to the higher order fractional magnetic Schrödinger equation. We also study the higher order fractional Schrödinger equation with singular electric potential. In both cases, we obtain a Runge approximation property for the equation. Furthermore, we prove a uniqueness result for a partial data problem of the *d*-plane Radon transform in low regularity. Our work extends some recent results in inverse problems for more general operators.

### 1. INTRODUCTION

The fractional Laplacian  $(-\Delta)^s$ ,  $s \in (-n/2, \infty) \setminus \mathbb{Z}$ , is a non-local operator by definition and thus differs substantially from the ordinary Laplacian  $(-\Delta)$ . The non-local behaviour can be exploited when solving fractional inverse problems. In section 3.1, we prove that  $(-\Delta)^s$  admits a unique continuation property (UCP) for open sets, that is, if u and  $(-\Delta)^s u$  both vanish in a nonempty open set, then u vanishes everywhere. Clearly this property cannot hold for local operators. We give many other versions of UCPs as well.

We have also included a quite comprehensive discussion of the Poincaré inequality for the higher order fractional Laplacian  $(-\Delta)^s$ ,  $s \ge 0$ , in section 3.2. We give many proofs for the higher order fractional Poincaré inequality based on various different methods in the literature. The higher order fractional Poincaré inequality appears earlier at least in [84] for functions in  $C_c^{\infty}(\Omega)$  where  $\Omega$  is a bounded Lipschitz domain. Also similar inequalities are proved in the book [4] for homogeneous Sobolev norms but without referring to the fractional Laplacian. However, we have extended some known results, given alternative proofs, and studied a connection between the fractional and the classical Poincaré constants. We believe that section 3.2 will serve as a helpful reference on fractional Poincaré inequalities in the future.

Our main applications are fractional Schrödinger equations with and without a magnetic potential, and the *d*-plane Radon transforms with partial data. We apply the UCP result and the Poincaré inequality for higher order fractional Laplacians to show uniqueness for the associated fractional Schrödinger equation and the Runge approximation properties. UCPs have also applications in integral geometry since certain partial data inverse problems for the Radon transforms can be reduced to unique continuation problems of the normal operators. We remark that the normal operators of the Radon transforms are negative order fractional Laplacians (Riesz potentials) up to constant coefficients.

In this section, we introduce our models, discuss some related results and present our main theorems and corollaries. We start with the classical Calderón problem as a motivation.

1.1. The Calderón problem. We will study a non-local version of the famous Calderón problem called the fractional Calderón problem. A survey of the fractional Calderón problem is given in [79]. The Calderón problem is a classical inverse problem where one wants to determine the

*Date*: November 11, 2020.

Key words and phrases. Inverse problems, unique continuation, fractional Laplacian, fractional Schrödinger equation, fractional Poincaré inequality, Radon transform.

electrical conductivity on some sufficiently smooth domain by boundary measurements [77, 83]. Suppose that  $\Omega \subset \mathbb{R}^n$  is a domain with regular enough boundary  $\partial \Omega$ . The electrical conductivity is usually represented as a bounded positive function  $\gamma$ , and the conductivity equation is

(1) 
$$\begin{cases} \nabla \cdot (\gamma \nabla u) &= 0 \text{ in } \Omega \\ u|_{\partial \Omega} &= f \end{cases}$$

where f is the potential on the boundary  $\partial\Omega$  and u is the induced potential in  $\Omega$ . The data in this problem is the Dirichlet-to-Neumann (DN) map  $\Lambda_{\gamma}(f) = (\gamma \partial_{\nu} u)|_{\partial\Omega}$ , where  $\nu$  is the outer unit normal on the boundary. The DN map basically tells how the applied voltage on the boundary induces normal currents on the boundary by the electrical properties of the interior. The inverse problem is to determine  $\gamma$  from the DN map  $\Lambda_{\gamma}$ . One of the associated basic questions is the uniqueness problem, that is, whether  $\gamma_1 = \gamma_2$  follows from  $\Lambda_{\gamma_1} = \Lambda_{\gamma_2}$ .

Equation (1) can be reduced to a Schrödinger equation

(2) 
$$\begin{cases} (-\Delta + q)u &= 0 \text{ in } \Omega \\ u|_{\partial\Omega} &= f \end{cases}$$

where  $q = (\Delta \sqrt{\gamma})/\sqrt{\gamma}$  now represents the electric potential in  $\Omega$ . One typically assumes that 0 is not a Dirichlet eigenvalue of the operator  $(-\Delta + q)$  to obtain unique solutions to equation (2). The inverse problem then is to know whether one can determine the electric potential q uniquely from the DN map  $\Lambda_q$ , which can be expressed in terms of the normal derivative  $\Lambda_q f = \partial_{\nu} u |_{\partial\Omega}$  for regular enough boundaries. For more details on the classical Calderón problem and its applications to medical, seismic and industrial imaging, see [77, 83].

1.2. Fractional Schrödinger equation. In this article, we focus on the fractional Schrödinger equation and its generalization, the fractional magnetic Schrödinger equation. The main difference between the classical and fractional Schrödinger operators is that the first one is local and the second one is non-local. This can be seen since the Laplacian  $(-\Delta)$  is local as a differential operator while the fractional counterpart  $(-\Delta)^s$ ,  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$ , is a non-local Fourier integral operator. In other words, the value  $(-\Delta)^s u(x)$ ,  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$ , depends on the values of u everywhere, not just in a small neighbourhood of  $x \in \mathbb{R}^n$ . Fractional Laplacians have a close connection to Levý processes and have been used in many areas of mathematics and physics, for example to model anomalous and nonlocal diffusion, and also in the formulation of fractional quantum mechanics where the fractional Schrödinger equation arises naturally as a generalization of the ordinary Schrödinger equation [3, 7, 18, 28, 50, 51, 58, 71].

Since the fractional Laplacian is a non-local operator, it is more natural to fix exterior values for the solutions of the equation instead of just boundary values. This motivates the study of the following exterior value problem, first introduced in [28],

(3) 
$$\begin{cases} ((-\Delta)^s + q)u &= 0 \text{ in } \Omega \\ u|_{\Omega_e} &= f \end{cases}$$

where  $\Omega_e = \mathbb{R}^n \setminus \overline{\Omega}$  is the exterior of  $\Omega$ . The associated DN map for equation (3) is a bounded linear operator  $\Lambda_q \colon H^s(\Omega_e) \to (H^s(\Omega_e))^*$  which, under stronger assumptions, has an expression  $\Lambda_q f = (-\Delta)^s u|_{\Omega_e}$  [28]. We assume that the potential q is such that the following holds:

(4) If 
$$u \in H^s(\mathbb{R}^n)$$
 solves  $((-\Delta)^s + q)u = 0$  in  $\Omega$  and  $u|_{\Omega_e} = 0$ , then  $u = 0$ .

In other words, condition (4) requires that 0 is not a Dirichlet eigenvalue of the operator  $((-\Delta)^s + q)$ .

In section 5, we will prove that, under certain assumptions, one can uniquely determine the potential q in equation (3) from exterior measurements when  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$ , and we also prove a Runge approximation property for equation (3) (see also section 1.5). These generalize the results in [28, 75] to higher fractional powers of s. The proofs basically reduce to the fact that the operator  $(-\Delta)^s$  has the following UCP: if  $(-\Delta)^s u|_V = 0$  and  $u|_V = 0$  for some nonempty

open set  $V \subset \mathbb{R}^n$ , then u = 0 everywhere. This reflects the fact that  $(-\Delta)^s$  is a non-local operator since such UCP can never hold for local operators.

Unique continuation of the fractional Laplacian has been extensively studied and used to show uniqueness results for fractional Schrödinger equations [14, 27, 28, 75]. One version was already proved by Riesz [28, 70] and similar methods were used in [41] to show a UCP of Riesz potentials  $I_{\alpha}$  which can be seen as fractional Laplacians with negative exponents. See also [45] for a unique continuation result of Riesz potentials. UCP of  $(-\Delta)^s$  for functions in  $H^{r}(\mathbb{R}^{n}), r \in \mathbb{R}$ , was proved in [28] when  $s \in (0, 1)$ . The proof is based on Carleman estimates from [72] and on Caffarelli-Silvestre extension [8, 9]. Using the known result for  $s \in (0, 1)$ , we provide an elementary proof which generalizes the UCP for all  $s \in (-n/2, \infty) \setminus \mathbb{Z}$ . With the same trick we obtain several other unique continuation results. There are also strong unique continuation results for  $s \in (0,1)$  if one assumes more regularity from the function [22, 72]. In the strong UCP, one replaces the condition  $u|_V = 0$  by the requirement that u vanishes to infinite order at some point  $x_0 \in V$ . The higher order case  $s \in \mathbb{R}^+ \setminus (\mathbb{Z} \cup (0,1))$  has been studied recently by several authors [23, 26, 86]. These results however assume some special conditions on the function u, i.e. they require that u is in a Sobolev space which depends on the power sof the fractional Laplacian  $(-\Delta)^s$ . We only require that u is in some Sobolev space  $H^r(\mathbb{R}^n)$ where  $r \in \mathbb{R}$  can be an arbitrarily small (negative) number.

See also [45] where the author proves a higher order Runge approximation property by sharmonic functions in the unit ball when  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$  (compare to theorem 1.7). Here sharmonicity simply means that  $(-\Delta)^s u = 0$  in some domain  $\Omega$ . The s-harmonic approximation in the case  $s \in (0, 1)$  was already studied in [17]; similar higher regularity approximation results are proved in [11, 28] for the fractional Schrödinger equation.

1.3. Fractional magnetic Schrödinger equation. Section 6 of this paper extends the study of the fractional magnetic Schrödinger equation (FMSE) begun in [14], expanding the uniqueness result for the related inverse problem to the cases when  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$ . The direct problem for the classical magnetic Schrödinger equation (MSE) consists in finding a function u satisfying

$$\begin{cases} (-\Delta)_A u + qu &= -\Delta u - i\nabla \cdot (Au) - iA \cdot \nabla u + (|A|^2 + q)u = 0 \text{ in } \Omega \\ u|_{\partial\Omega} &= f \end{cases}$$

where  $\Omega \subset \mathbb{R}^n$  is some bounded open set with Lipschitz boundary representing a medium, f is the boundary value for the solution u, and A, q are the vector and scalar potentials of the equation. In the associated inverse problem, we are given measurements on the boundary in the form of a DN map  $\Lambda_{A,q} : H^{1/2}(\partial \Omega) \to H^{-1/2}(\partial \Omega)$ , and we are asked to recover A, q in  $\Omega$  using this information. It was shown in [60] that this is only possible up to a natural gauge: one can uniquely determine the potential q and the magnetic *field* curl A, but the magnetic *potential* A can not be determined in greater detail. The inverse problem for MSE is of great interest, because it generalizes the non-magnetic case by adding some first order terms, and shows a quite different behavior. It also possesses multiple applications in the sciences: the papers [60, 62, 56, 20, 61] and [35] give some examples of this, treating the inverse scattering problem with a fixed energy, isotropic elasticity, the Maxwell, Schrödinger and Dirac equations and the Stokes system. We refer to the survey [76] for many more references on inverse boundary value problems related to MSE.

We are interested in the study of a high order fractional version of the MSE. There have been many studies in this direction (see for instance [54, 52, 53]). In our work, we will build upon the results from [14] and generalize them to higher order. Thus, for us the direct problem for FMSE asks to find a function u which satisfies

$$\begin{cases} (-\Delta)_A^s u + qu = 0 \text{ in } \Omega \\ u|_{\Omega_e} = f \end{cases}$$

where  $\Omega$ , f, A and q play a similar role as in the local case,  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$  and  $(-\Delta)^s_A$  is the magnetic fractional Laplacian. This is a fractional version of  $(-i\nabla + A) \cdot (-i\nabla + A)$ , the magnetic Laplacian from which MSE arises. In section 6, we will construct the fractional magnetic Laplacian based

on the fractional gradient operator  $\nabla^s$ . The fractional gradient is based on the framework laid down in [18, 19], and has been studied in the papers [15, 14]. One should keep in mind that for s > 1 the fractional gradient is a tensor of order  $\lfloor s \rfloor$  rather than a vector. In the corresponding inverse problem, we assume to know the DN map  $\Lambda^s_{A,q} : H^s(\Omega_e) \to (H^s(\Omega_e))^*$ , and we wish to recover A, q in  $\Omega$ . In the cases when  $s \in (0, 1)$ , it has been shown that the pair A, q can only be recovered up to a natural gauge [14]. We generalize this result to the case  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$ . This is achieved by first proving a weak UCP and the Runge approximation property for FMSE, and then testing the Alessandrini identity for the equation with suitably chosen functions.

**Remark 1.1.** The case of the high order magnetic Schrödinger equation, that is the one in which  $s \in \mathbb{N}$ ,  $s \neq 1$ , is still open at the time of writing to the best of the authors' knowledge. Our methods are purely nonlocal, and thus cannot be applied to the integer case. It was however showed in [60], as cited above, that a uniqueness result up to a natural gauge holds when s = 1.

1.4. Radon transforms and region of interest tomography. Unique continuation results have also applications in integral geometry. It was proved in [41] that the normal operator of the X-ray transform admits a UCP in the class of compactly supported distributions. This was done by considering the normal operator as a Riesz potential. We generalize the result for the normal operator of the *d*-plane transform  $R_d$  where  $d \in \mathbb{N}$  is odd such that 0 < d < n. In the case d = 1 the transform  $R_d$  corresponds to the X-ray transform and in the case d = n - 1 to the Radon transform. The UCP of the normal operator  $N_d = R_d^* R_d$  implies uniqueness for the following partial data problem: if f integrates to zero over all d-planes which intersect some nonempty open set V and  $f|_V = 0$ , then f = 0. This can be seen as a complementary result to the Helgason support theorem for the d-plane transform [36]. Helgason's theorem says that if fintegrates to zero over all d-planes not intersecting a convex and compact set K and  $f|_{K} = 0$ , then f = 0. The *d*-plane transform  $R_d$  is injective on continuous functions which decay rapidly enough at infinity and also on compactly supported distributions [36]. The d-plane transform has been recently studied in the periodic case on the flat torus [2, 40, 67] but also in other settings [16, 37, 69]. Weighted and limited data Radon transforms (d = n - 1) have been studied recently for example in [25, 29, 30, 31].

When d = 1, partial data problems as discussed above arise for example in seismology and medical imaging. In [41], it is explained how one can use shear wave splitting data to uniquely determine the difference of the anisotropic perturbations in the S-wave speeds, and also how one can use local measurements of travel times of seismic waves to uniquely determine the conformal factor in the linearization. Both of these problems reduce to the following partial data result: if f integrates to zero over all lines which intersect some nonempty open set V and  $f|_V = 0$ , then f = 0. In medical imaging, one typically wants to reconstruct a specific part of the human body. Can this be done by using only X-rays which go though our region of interest (ROI)? Generally this is not possible even for  $C_c^{\infty}$ -functions [43, 63, 81], but if we know some information of f in the ROI, then the reconstruction can be done. For example, if the function f is piecewice constant, piecewice polynomial or analytic in the ROI, then fcan be uniquely determined from the X-ray data [42, 43, 85]. Also, if we know the X-ray data through the ROI and the values of f in an arbitrarily small open set inside the ROI, then fis uniquely determined everywhere [13, 41]. For practical applications of ROI tomography in medical imaging, see for example [87, 88]. See also [44, 65, 66] for a discussion of the difficulties of obtaining stable reconstruction in partial data problems for the X-ray transform (visible and invisible singularities).

1.5. Main results. We briefly introduce the basic notation; more details can be found in sections 2, 4, 5 and 6. Let  $H^r(\mathbb{R}^n)$  be the  $L^2$  Sobolev space of order  $r \in \mathbb{R}$  and  $\tilde{H}^r(\Omega)$  the closure of  $C_c^{\infty}(\Omega)$  in  $H^r(\mathbb{R}^n)$  when  $\Omega$  is an open set. The  $L^1$  Bessel potential space is denoted by  $H^{r,1}(\mathbb{R}^n)$ . We define  $H^r_K(\mathbb{R}^n) \subset H^r(\mathbb{R}^n)$  to be those Sobolev functions which have support in the compact set K. The fractional Laplacian is defined via the Fourier transform  $(-\Delta)^s u = \mathcal{F}^{-1}(|\cdot|^{2s}\hat{u})$ . Then  $(-\Delta)^s \colon H^r(\mathbb{R}^n) \to H^{r-2s}(\mathbb{R}^n)$  is a continuous operator when

 $s \in \mathbb{R}^+ \setminus \mathbb{Z}$ . The *d*-plane transform  $R_d$  takes a function which decreases rapidly enough at infinity and integrates it over *d*-dimensional planes where 0 < d < n. The normal operator of the *d*-plane transform is defined as  $N_d = R_d^* R_d$  where  $R_d^*$  is the adjoint operator. Further, we denote by  $\mathcal{D}'(\mathbb{R}^n)$  the space of all distributions,  $\mathcal{E}'(\mathbb{R}^n)$  the space of compactly supported distributions,  $\mathcal{O}'_C(\mathbb{R}^n)$  the space of rapidly decreasing distributions and  $C_{\infty}(\mathbb{R}^n)$  the set of rapidly decreasing continuous functions. The space of singular potentials  $Z_0^{-s}(\mathbb{R}^n)$  is a certain subset of distributions  $\mathcal{D}'(\mathbb{R}^n)$  and can be interpreted as a set of bounded multipliers from  $H^s(\mathbb{R}^n)$  to  $H^{-s}(\mathbb{R}^n)$ .

The following theorem extends a result in [28] and has a central role in this article. We call it the UCP of the operator  $(-\Delta)^s$ .

**Theorem 1.2.** Let  $n \ge 1$ ,  $s \in (-n/4, \infty) \setminus \mathbb{Z}$  and  $u \in H^r(\mathbb{R}^n)$  where  $r \in \mathbb{R}$ . If  $(-\Delta)^s u|_V = 0$ and  $u|_V = 0$  for some nonempty open set  $V \subset \mathbb{R}^n$ , then u = 0. The claim holds also for  $s \in (-n/2, -n/4] \setminus \mathbb{Z}$  if  $u \in H^{r,1}(\mathbb{R}^n)$  or  $u \in \mathcal{O}'_C(\mathbb{R}^n)$ .

Theorem 1.2 is proved in section 3.1. The UCP of  $(-\Delta)^s$  implies corresponding UCP for Riesz potentials (see corollary 3.2 and [41, Theorem 5.2]). This in turn implies the following UCP for the normal operator of the *d*-plane transform  $N_d$  when *d* is odd; the case d = 1 was already studied in [41].

**Corollary 1.3.** Let  $n \ge 2$  and let f belong to either  $\mathcal{E}'(\mathbb{R}^n)$  or  $C_{\infty}(\mathbb{R}^n)$ . Let  $d \in \mathbb{N}$  be odd such that 0 < d < n. If  $N_d f|_V = 0$  and  $f|_V = 0$  for some nonempty open set  $V \subset \mathbb{R}^n$ , then f = 0.

From the UCP of  $N_d$  we obtain the next result which is in a sense complementary to the Helgason support theorem for the *d*-plane transform [36, Theorem 6.1]. It extends a result in [41] where the authors prove a similar uniqueness property for the X-ray transform.

**Corollary 1.4.** Let  $n \ge 2$ ,  $V \subset \mathbb{R}^n$  a nonempty open set and  $f \in C_{\infty}(\mathbb{R}^n)$ . Let  $d \in \mathbb{N}$  be odd such that 0 < d < n. If  $f|_V = 0$  and  $R_d f = 0$  for all d-planes intersecting V, then f = 0. The claim holds also for  $f \in \mathcal{E}'(\mathbb{R}^n)$  when the assumption  $R_d f = 0$  for all d-planes intersecting V is understood in the sense of distributions.

If d is even, then f is uniquely determined in V by its integrals over d-planes which intersect V, i.e.  $R_d f = 0$  for all d-planes intersecting V implies  $f|_V = 0$  (see remark 4.2). The authors do not know if the result of corollary 1.4 holds when d is even. However, if d is even, then the result of corollary 1.3 cannot be true as the normal operator  $N_d$  is the inverse of a local operator. See section 4 for the proofs and the definition of the d-plane transform of distributions.

The following result is a general version of the Poincaré inequality which we need for the well-posedness of the inverse problem for the fractional Schrödinger equation.

**Theorem 1.5.** Let  $n \ge 1$ ,  $s \ge t \ge 0$ ,  $K \subset \mathbb{R}^n$  a compact set and  $u \in H^s_K(\mathbb{R}^n)$ . There exists a constant  $\tilde{c} = \tilde{c}(n, K, s) > 0$  such that

$$\left\| (-\Delta)^{t/2} u \right\|_{L^2(\mathbb{R}^n)} \le \tilde{c} \left\| (-\Delta)^{s/2} u \right\|_{L^2(\mathbb{R}^n)}$$

The constant  $\tilde{c}$  can be expressed in terms of the classical Poincaré constant when  $s \geq 1$  (see theorem 3.17. See section 3.2 for several proofs of the Poincaré inequality. From the unique continuation of  $(-\Delta)^s$  we obtain results for the higher order fractional Schrödinger equation with singular electric potential. The following theorems generalize the results in [28, 75] for higher exponents  $s \in \mathbb{R}^+ \setminus (\mathbb{Z} \cup (0, 1))$ .

**Theorem 1.6.** Let  $n \geq 1$ ,  $\Omega \subset \mathbb{R}^n$  a bounded open set,  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$ , and  $q_1, q_2 \in Z_0^{-s}(\mathbb{R}^n)$ which satisfy condition (4). Let  $W_1, W_2 \subset \Omega_e$  be open sets. If the DN maps for the equations  $(-\Delta)^s u + m_{q_i}(u) = 0$  in  $\Omega$  satisfy  $\Lambda_{q_1} f|_{W_2} = \Lambda_{q_2} f|_{W_2}$  for all  $f \in C_c^{\infty}(W_1)$ , then  $q_1|_{\Omega} = q_2|_{\Omega}$ .

**Theorem 1.7.** Let  $n \geq 1$  and  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$ . Let  $\Omega \subset \mathbb{R}^n$  be a bounded open set and  $\Omega_1 \supset \Omega$ any open set such that  $int(\Omega_1 \setminus \Omega) \neq \emptyset$ . If  $q \in Z_0^{-s}(\mathbb{R}^n)$  satisfies condition (4), then any  $g \in \widetilde{H}^s(\Omega)$  can be approximated arbitrarily well in  $\widetilde{H}^s(\Omega)$  by solutions  $u \in H^s(\mathbb{R}^n)$  to the equation  $(-\Delta)^s u + m_q(u) = 0$  in  $\Omega$  such that  $spt(u) \subset \overline{\Omega}_1$ . We remark that the approximation property in theorem 1.7 also holds in  $L^2(\Omega)$  when one takes restrictions of the solutions (see [28, Theorem 1.3]). In [17, 45] the authors prove similar approximation results:  $C^k$ -functions can be approximated (in the  $C^k$ -norm) in the unit ball by *s*-harmonic functions, i.e. functions *u* which satisfy  $(-\Delta)^s u = 0$  in  $B_1(0)$  (see also [28, Remark 7.3]). Theorems 1.6 and 1.7 are proved in section 5. The proofs are almost identical to those in [28, 75] and only slight changes need to be done. We will present the main ideas of the proofs for clarity and in order to make a comparison to the more complicated case of FMSE.

We have achieved the following result on the Calderón problem for FMSE:

**Theorem 1.8.** Let  $\Omega \subset \mathbb{R}^n$ ,  $n \geq 2$ , be a bounded open set,  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$ , and let  $A_i, q_i$  verify assumptions (a1)-(a5) in section 6 for i = 1, 2. Let  $W_1, W_2 \subset \Omega_e$  be open sets. If the DN maps for the FMSEs in  $\Omega$  relative to  $(A_1, q_1)$  and  $(A_2, q_2)$  satisfy

$$\Lambda^{s}_{A_{1},q_{1}}[f]|_{W_{2}} = \Lambda^{s}_{A_{2},q_{2}}[f]|_{W_{2}} \quad for \ all \ f \in C^{\infty}_{c}(W_{1}),$$

then  $(A_1, q_1) \sim (A_2, q_2)$ , that is, the potentials coincide up to gauge.

An in-depth clarification of the assumptions and the definition of the gauge involved in the proof are presented in section 6.

1.6. Organization of the article. This article is organized as follows. Section 2 is devoted to preliminaries. We introduce our notation and definitions of relevant quantities. In sections 3.1 and 3.2 we prove the unique continuation property of  $(-\Delta)^s$  for  $s \in (-n/2, \infty) \setminus \mathbb{Z}$  and give several proofs for the fractional Poincaré inequality. We introduce some applications in integral geometry and partial data problems of the *d*-plane transform in section 4. In section 5, we show the uniqueness and the Runge approximation results for the higher order fractional Schrödinger equation with singular electric potential. We prove the uniqueness result up to a gauge for the higher order fractional magnetic Schrödinger equation in section 6. Finally, in section 7, we discuss other problems that would now naturally continue our work. There are many potential recent results in inverse problems which perhaps can be generalized to higher order fractional Laplacians using our unique continuation result and fractional Poincaré inequality.

Acknowledgements. The authors wish to thank Yi-Hsuan Lin for suggesting to study higher order fractional Calderón problems and for his idea of reducing the UCP of higher order fractional Laplacians to the case  $s \in (0, 1)$ . The authors are grateful to Mikko Salo for proposing a proof for the fractional Poincaré inequality for n = 1 and  $s \in (1/2, 1)$ , and for many other helpful discussions. We thank Joonas Ilmavirta for discussions about integral geometry. The authors wish to thank the anonymous referees for helpful comments and suggestions to improve the article. G.C. was partially supported by the European Research Council under Horizon 2020 (ERC CoG 770924). K.M. and J.R. were supported by Academy of Finland (Centre of Excellence in Inverse Modelling and Imaging, grant numbers 284715 and 309963).

#### 2. Preliminaries

In this section, we will go through our basic notations and definitions. The following theory of distributions, Fourier analysis and Sobolev spaces can be found in many books (see for example [1, 4, 6, 38, 39, 57, 59, 78, 82]). We write  $|\cdot|$  for both the Euclidean norm of vectors and the absolute value of complex numbers. We denote by  $\mathbb{N}_0$  the set of natural numbers including zero.

2.1. Distributions and Fourier transform. We denote by  $\mathcal{E}(\mathbb{R}^n)$  the set of smooth functions equipped with the topology of uniform convergence of derivatives of all order on compact sets. We also denote by  $\mathcal{D}(\mathbb{R}^n)$  the set of compactly supported smooth functions with the topology of uniform convergence of derivatives of all order in a fixed compact set. The topological duals of these spaces are denoted by  $\mathcal{D}'(\mathbb{R}^n)$  and  $\mathcal{E}'(\mathbb{R}^n)$ . Elements in the space  $\mathcal{E}'(\mathbb{R}^n)$  can be identified as distributions in  $\mathcal{D}'(\mathbb{R}^n)$  with compact support. We also use the space of rapidly decreasing smooth functions, i.e. Schwartz functions. Define the Schwartz space as

$$\mathscr{S}(\mathbb{R}^n) = \left\{ \varphi \in C^{\infty}(\mathbb{R}^n) : \left\| \langle \cdot \rangle^N \partial^\beta \varphi \right\|_{L^{\infty}(\mathbb{R}^n)} < \infty \text{ for all } N \in \mathbb{N} \text{ and } \beta \in \mathbb{N}_0^n \right\},\$$

where  $\langle x \rangle = (1+|x|^2)^{1/2}$ , equipped with the topology induced by the seminorms  $\|\langle \cdot \rangle^N \partial^\beta \varphi\|_{L^{\infty}(\mathbb{R}^n)}$ . The continuous dual of  $\mathscr{S}(\mathbb{R}^n)$  is denoted by  $\mathscr{S}'(\mathbb{R}^n)$  and its elements are called tempered distributions. We have the continuous inclusions  $\mathcal{E}'(\mathbb{R}^n) \subset \mathscr{S}'(\mathbb{R}^n) \subset \mathcal{D}'(\mathbb{R}^n)$ . The Fourier transform of  $u \in L^1(\mathbb{R}^n)$  is defined as

$$(\mathcal{F}u)(\xi) = \hat{u}(\xi) = \int_{\mathbb{R}^n} e^{-ix\cdot\xi} u(x) \mathrm{d}x$$

and it is an isomorphism  $\mathcal{F}: \mathscr{S}(\mathbb{R}^n) \to \mathscr{S}(\mathbb{R}^n)$ . By duality the Fourier transform is also an isomorphism  $\mathcal{F}: \mathscr{S}'(\mathbb{R}^n) \to \mathscr{S}'(\mathbb{R}^n)$ . By density of  $\mathscr{S}(\mathbb{R}^n)$  in  $L^2(\mathbb{R}^n)$  the Fourier transform can be extended to an isomorphism  $\mathcal{F}: L^2(\mathbb{R}^n) \to L^2(\mathbb{R}^n)$ . The following subset of Schwartz space

$$\mathscr{S}_0(\mathbb{R}^n) = \{ \varphi \in \mathscr{S}(\mathbb{R}^n) : \hat{\varphi}|_{B(0,\epsilon)} = 0 \text{ for some } \epsilon > 0 \}$$

is used to define fractional Laplacians on homogeneous Sobolev spaces.

Finally, we denote by  $\mathscr{O}'_{C}(\mathbb{R}^{n})$  the space of rapidly decreasing distributions. One has that  $T \in \mathscr{O}'_{C}(\mathbb{R}^{n})$  if and only if for any  $N \in \mathbb{N}$  there exist  $M(N) \in \mathbb{N}$  and continuous functions  $g_{\beta}$  such that

$$T = \sum_{|\beta| \le M(N)} \partial^{\beta} g_{\beta} ,$$

where  $\langle \cdot \rangle^N g_\beta$  is a bounded function for every  $|\beta| \leq M(N)$ . Alternatively one can characterize  $\mathscr{O}'_C(\mathbb{R}^n)$  via the Fourier transform: it holds that  $\mathcal{F} \colon \mathscr{O}'_C(\mathbb{R}^n) \to \mathscr{O}_M(\mathbb{R}^n)$  is a bijective map where  $\mathscr{O}_M(\mathbb{R}^n)$  is the space of smooth functions with polynomially bounded derivatives of all orders. We have the continuous inclusions  $\mathcal{E}'(\mathbb{R}^n) \subset \mathscr{O}'_C(\mathbb{R}^n) \subset \mathscr{S}'(\mathbb{R}^n)$ . For example  $C_{\infty}(\mathbb{R}^n) \subset \mathscr{O}'_C(\mathbb{R}^n)$ , where  $f \in C_{\infty}(\mathbb{R}^n)$  if and only if f is continuous and  $\langle \cdot \rangle^N f$  is bounded for every  $N \in \mathbb{N}$ . The convolution formula for the Fourier transform  $\widehat{f*g} = \widehat{f}\widehat{g}$  holds in the sense of distributions when  $f \in \mathscr{O}'_C(\mathbb{R}^n)$  and  $g \in \mathscr{S}'(\mathbb{R}^n)$ . For more details on distributions, see the classic books [38, 39, 82].

2.2. Fractional Laplacian on Sobolev spaces. Let  $r \in \mathbb{R}$ . We define the inhomogeneous fractional  $L^2$  Sobolev space of order r to be the set

$$H^{r}(\mathbb{R}^{n}) = \{ u \in \mathscr{S}'(\mathbb{R}^{n}) : \mathcal{F}^{-1}(\langle \cdot \rangle^{r} \hat{u}) \in L^{2}(\mathbb{R}^{n}) \}$$

equipped with the norm

$$\|u\|_{H^r(\mathbb{R}^n)} = \left\|\mathcal{F}^{-1}(\langle \cdot \rangle^r \hat{u})\right\|_{L^2(\mathbb{R}^n)}.$$

The spaces  $H^r(\mathbb{R}^n)$  are Hilbert spaces for all  $r \in \mathbb{R}$ . It follows that both  $\mathscr{S}(\mathbb{R}^n)$  and  $\mathscr{S}_0(\mathbb{R}^n)$  are dense in  $H^r(\mathbb{R}^n)$  for all  $r \in \mathbb{R}$ . Note that

$$\mathscr{O}'_C(\mathbb{R}^n) \subset \bigcup_{r \in \mathbb{R}} H^r(\mathbb{R}^n).$$

If  $s \in (0, 1)$ , the fractional Laplacian can be defined in several equivalent ways [46]. We will take the Fourier transform approach which allows us to define it as a continuous map on Sobolev spaces for all  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$ . Define the fractional Laplacian of order  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$  as  $(-\Delta)^s \varphi = \mathcal{F}^{-1}(|\cdot|^{2s} \hat{\varphi})$  for  $\varphi \in \mathscr{S}(\mathbb{R}^n)$ . Then  $(-\Delta)^s \colon \mathscr{S}(\mathbb{R}^n) \to H^{r-2s}(\mathbb{R}^n)$  is linear and continuous with respect to the norm  $\|\cdot\|_{H^r(\mathbb{R}^n)}$  by a simple calculation. Thus we can uniquely extend it to a continuous linear operator  $(-\Delta)^s \colon H^r(\mathbb{R}^n) \to H^{r-2s}(\mathbb{R}^n)$  as  $(-\Delta)^s u = \lim_{k\to\infty} (-\Delta)^s \varphi_k$ , where  $\varphi_k \in \mathscr{S}(\mathbb{R}^n)$  is such that  $\varphi_k \to u$  in  $H^r(\mathbb{R}^n)$ .

On the other hand, if s > -n/4, one can always define  $(-\Delta)^s u$  for  $u \in H^r(\mathbb{R}^n)$  as the tempered distribution  $(-\Delta)^s u = \mathcal{F}^{-1}(|\cdot|^{2s}\hat{u})$ , note that we also allow integer values of s here. This can be seen in the following way: let  $\varphi_k \in \mathscr{S}(\mathbb{R}^n)$  such that  $\varphi_k \to 0$  in  $\mathscr{S}(\mathbb{R}^n)$ . It holds that

 $|\cdot|^{-\beta} \in L^1_{loc}(\mathbb{R}^n)$  if and only if  $\beta < n$ . Taking  $N \in \mathbb{N}$  large enough and using Cauchy-Schwartz we obtain

$$\begin{split} \int_{\mathbb{R}^n} |x|^{2s} |\hat{u}(x)| \, |\varphi_k(x)| \, \mathrm{d}x &\leq \left( \int_{\mathbb{R}^n} \langle x \rangle^{2r} \, |\hat{u}(x)|^2 \, \mathrm{d}x \right)^{1/2} \left( \int_{\mathbb{R}^n} |x|^{4s} \, \langle x \rangle^{-2r} \, |\varphi_k(x)|^2 \, \mathrm{d}x \right)^{1/2} \\ &\leq C \bigg( \int_{\mathbb{R}^n} \frac{|x|^{4s}}{\langle x \rangle^{2N}} \mathrm{d}x \bigg)^{1/2} \, \big\| \langle \cdot \rangle^{N-r} \varphi_k \big\|_{L^{\infty}(\mathbb{R}^n)} \to 0. \end{split}$$

Hence  $|\cdot|^{2s} \hat{u} \in \mathscr{S}'(\mathbb{R}^n)$  and also  $(-\Delta)^{su} = \mathcal{F}^{-1}(|\cdot|^{2s} \hat{u}) \in \mathscr{S}'(\mathbb{R}^n)$ . The definition can be relaxed to s > -n/2 if we assume that  $\langle \cdot \rangle^t \hat{u} \in L^{\infty}(\mathbb{R}^n)$  for some  $t \in \mathbb{R}$ . This holds for example if  $u \in \mathscr{O}'_{C}(\mathbb{R}^{n})$  or  $u \in H^{r,1}(\mathbb{R}^{n})$  (see the definition of Bessel potential spaces below). When  $s \geq 0$ , we again obtain that  $(-\Delta)^s \colon H^r(\mathbb{R}^n) \to H^{r-2s}(\mathbb{R}^n)$  is continuous. It follows from the properties of the Fourier transform that  $(-\Delta)^k (-\Delta)^s = (-\Delta)^{k+s}$  when s > -n/2 and  $k \in \mathbb{N}$ . This relation will be used many times.

Fractional Laplacians with negative powers s have a connection to Riesz potentials. Let  $\alpha \in \mathbb{R}$ such that  $0 < \alpha < n$ . We define the Riesz potential  $I_{\alpha} \colon \mathscr{O}'_{C}(\mathbb{R}^{n}) \to \mathscr{S}'(\mathbb{R}^{n})$  as  $I_{\alpha}f = f * h_{\alpha}$ , where the kernel is  $h_{\alpha}(x) = |x|^{-\alpha}$ . It follows that  $I_{\alpha}$  is continuous in the distributional sense and  $I_{\alpha} = (-\Delta)^{-s}$ , up to a constant factor, where  $s = (n - \alpha)/2$ . On the other hand, if -n/2 < s < 0, then one can write  $(-\Delta)^s f = f * |\cdot|^{-2s-n} = I_{2s+n}f$ , also up to a constant factor. Hence fractional Laplacians with negative powers correspond to Riesz potentials and vice versa.

Following [4], one can define fractional Laplacians and Riesz potentials on homogeneous Sobolev spaces. Let us define

$$\dot{H}^{r}(\mathbb{R}^{n}) = \{ u \in \mathscr{S}'(\mathbb{R}^{n}) : \hat{u} \in L^{1}_{loc}(\mathbb{R}^{n}) \text{ and } |\cdot|^{r} \, \hat{u} \in L^{2}(\mathbb{R}^{n}) \}$$

and equip it with the norm

$$||u||_{\dot{H}^{r}(\mathbb{R}^{n})} = \left(\int_{\mathbb{R}^{n}} |\xi|^{2r} |\hat{u}(\xi)|^{2} d\xi\right)^{1/2}.$$

The norm  $||u||_{\dot{H}^r(\mathbb{R}^n)}$  is homogeneous with respect to scaling  $\xi \to \lambda \xi$  in contrast to the norm  $||u||_{H^r(\mathbb{R}^n)}$ . We have the inclusions  $\dot{H}^r(\mathbb{R}^n) \subsetneq H^r(\mathbb{R}^n)$  for r < 0 and  $H^r(\mathbb{R}^n) \subsetneq \dot{H}^r(\mathbb{R}^n)$  for r > 0. If r < n/2, then  $\dot{H}^r(\mathbb{R}^n)$  is a Hilbert space and  $\mathscr{S}_0(\mathbb{R}^n)$  is dense in  $\dot{H}^r(\mathbb{R}^n)$ . Let  $s \ge 0$ and define  $(-\Delta)^s \varphi = \mathcal{F}^{-1}(|\cdot|^{2s} \hat{\varphi})$  for  $\varphi \in \mathscr{S}_0(\mathbb{R}^n)$ . Then  $(-\Delta)^s \colon \mathscr{S}_0(\mathbb{R}^n) \to \dot{H}^{r-2s}(\mathbb{R}^n)$  is an isometry with respect to the norm  $\|\cdot\|_{\dot{H}^r(\mathbb{R}^n)}$  and by density can be extended to a continuous map  $(-\Delta)^s \colon \dot{H}^r(\mathbb{R}^n) \to \dot{H}^{r-2s}(\mathbb{R}^n)$  when r < n/2. Similarly one obtains that  $I_\alpha \colon \dot{H}^r(\mathbb{R}^n) \to$  $\dot{H}^{r+n-\alpha}(\mathbb{R}^n)$  is a continuous map for  $r < \alpha - n/2$  and corresponds to fractional Laplacians with negative powers, up to a constant factor.

The fractional Laplacian can also be defined on Bessel potential spaces. Let  $1 \le p < \infty$ . We define

$$H^{r,p}(\mathbb{R}^n) = \{ u \in \mathscr{S}'(\mathbb{R}^n) : \mathcal{F}^{-1}(\langle \cdot \rangle^r \hat{u}) \in L^p(\mathbb{R}^n) \}$$

and equip it with the norm

$$\left\|u\right\|_{H^{r,p}(\mathbb{R}^n)} = \left\|\mathcal{F}^{-1}(\langle\cdot\rangle^r \hat{u})\right\|_{L^p(\mathbb{R}^n)}.$$

It follows that  $H^{r,p}(\mathbb{R}^n)$  is a Banach space and  $\mathscr{S}(\mathbb{R}^n)$  is dense in  $H^{r,p}(\mathbb{R}^n)$  for all  $r \in \mathbb{R}$ . By the Mikhlin multiplier theorem, one obtains that the operator  $(-\Delta)^s \colon H^{r,p}(\mathbb{R}^n) \to H^{r-2s,p}(\mathbb{R}^n)$ is continuous for  $s \ge 0$  and  $1 . The fractional Laplacian is also defined in the space <math>H^{r,1}(\mathbb{R}^n)$  since  $H^{r,1}(\mathbb{R}^n) \hookrightarrow H^{\frac{2r-n-\epsilon}{2}}(\mathbb{R}^n)$  for any  $\epsilon > 0$  by the continuity of the Fourier transform  $\mathcal{F}: L^1(\mathbb{R}^n) \to L^{\infty}(\mathbb{R}^n).$ 

One can define fractional Laplacians on more general spaces. It follows that if  $s \in (-n/2, 1]$ , then  $(-\Delta)^s \colon \mathscr{S}(\mathbb{R}^n) \to \mathscr{S}_s(\mathbb{R}^n)$  is continuous where  $\mathscr{S}_s(\mathbb{R}^n)$  is the set

$$\mathscr{S}_{s}(\mathbb{R}^{n}) = \{ \varphi \in C^{\infty}(\mathbb{R}^{n}) : \langle \cdot \rangle^{n+2s} \partial^{\beta} \varphi \in L^{\infty}(\mathbb{R}^{n}) \text{ for all } \beta \in \mathbb{N}_{0}^{n} \}$$

equipped with the topology induced by the seminorms  $\|\langle \cdot \rangle^{n+2s} \partial^{\beta} \varphi \|_{L^{\infty}(\mathbb{R}^{n})}$ . One can then extend  $(-\Delta)^{s}$  by duality to a continuous map  $(-\Delta)^{s} : (\mathscr{S}_{s}(\mathbb{R}^{n}))^{*} \to \mathscr{S}'(\mathbb{R}^{n})$ . See [28, 80] for more details and a characterization of the dual  $(\mathscr{S}_{s}(\mathbb{R}^{n}))^{*}$ .

2.3. Trace spaces and singular potentials. Let  $U, F \subset \mathbb{R}^n$  be an open and a closed set. We define the following Sobolev spaces

$$H^{r}(U) = \{u|_{U} : u \in H^{r}(\mathbb{R}^{n})\}$$
  

$$\widetilde{H}^{r}(U) = \text{ closure of } C^{\infty}_{c}(U) \text{ in } H^{r}(\mathbb{R}^{n})$$
  

$$H^{r}_{0}(U) = \text{ closure of } C^{\infty}_{c}(U) \text{ in } H^{r}(U)$$
  

$$H^{r}_{F}(\mathbb{R}^{n}) = \{u \in H^{r}(\mathbb{R}^{n}) : \operatorname{spt}(u) \subset F\}.$$

It is obvious that  $\widetilde{H}^r(U) \subset H^r_{\overline{U}}(\mathbb{R}^n)$  and  $\widetilde{H}^r(U) \subset H^r_0(U)$ . In nonlocal problems, we impose exterior values for the equation instead of boundary values. Therefore exterior values are considered to be the same if their difference is in the space  $\widetilde{H}^r(U)$ . For example, in equation (3) the condition  $u|_{\Omega_e} = f$  means that  $u - f \in \widetilde{H}^s(\Omega)$ , i.e. u and f are equal outside  $\overline{\Omega}$ , where  $\Omega$  is bounded open set. This motivates the definition of the abstract trace space  $X = H^r(\mathbb{R}^n)/\widetilde{H}^r(\Omega)$ which identifies functions in  $\Omega_e$ . If  $\Omega$  is a Lipschitz domain, then we have  $H^r_0(\Omega) = H^r_{\overline{\Omega}}(\mathbb{R}^n)$ when r > -1/2,  $r \notin \{1/2, 3/2, \ldots\}$ ,  $\widetilde{H}^r(\Omega) = H^r_{\overline{\Omega}}(\mathbb{R}^n)$ ,  $X = H^r(\Omega_e)$  and  $X^* = H^{-r}_{\overline{\Omega_e}}(\mathbb{R}^n)$ . Thus for more regular domains it could be more convenient to work with the spaces  $H^r_{\overline{\Omega}}(\mathbb{R}^n)$ , but in this article we do not assume any regularity of the set  $\Omega$ . For more theory of Sobolev spaces on (non-Lipschitz) domains and their properties, see [12, 57].

We also use some properties of singular potentials which were introduced in [75]. Let  $t \ge 0$ and define  $Z^{-t}(U)$  as a subspace of distributions  $\mathcal{D}'(U)$  equipped with the norm

$$||f||_{Z^{-t}(U)} = \sup\{|\langle f, u_1 u_2 \rangle_U| : u_i \in C_c^{\infty}(U), \ ||u_i||_{H^t(\mathbb{R}^n)} = 1\}$$

where  $\langle \cdot, \cdot \rangle_U$  is the dual pairing. We denote by  $Z_0^{-t}(U)$  the closure of  $C_c^{\infty}(U)$  in  $Z^{-t}(U)$ . Elements in  $Z^{-t}(\mathbb{R}^n)$  can be seen as multipliers: every  $f \in Z^{-t}(\mathbb{R}^n)$  induces a map  $m_f \colon H^t(\mathbb{R}^n) \to H^{-t}(\mathbb{R}^n)$  defined as  $\langle m_f(u), v \rangle_{\mathbb{R}^n} = \langle f, uv \rangle_{\mathbb{R}^n}$ . Also  $|\langle f, uv \rangle_{\mathbb{R}^n}| \le ||f||_{Z^{-t}(\mathbb{R}^n)} ||u||_{H^t(\mathbb{R}^n)} ||v||_{H^t(\mathbb{R}^n)}$ , and this inequality can be seen as a motivation for the definition of the space  $Z^{-t}(\mathbb{R}^n)$ . Clearly we have  $Z_0^{-t}(\mathbb{R}^n) \subset Z^{-t}(\mathbb{R}^n)$ . If U is bounded, then  $L^{\frac{n}{2t}}(U) \subset Z_0^{-t}(\mathbb{R}^n)$  for 0 < t < n/2 and  $L^{\infty}(U) \subset Z_0^{-t}(\mathbb{R}^n)$  in the sense of zero extensions. Further, it holds that  $L^p(U) \subset Z_0^{-t}(\mathbb{R}^n)$  when  $p > \max\{1, n/2t\}$  (see section 6). We will only need these basic inclusions. For a more detailed treatment of the space of singular potentials  $Z^{-t}(U)$ , see [55, 75].

# 3. UNIQUE CONTINUATION PROPERTY AND POINCARÉ INEQUALITY

3.1. Unique continuation results. In this section, we prove theorem 1.2 and give several other unique continuation results for fractional Laplacians and Riesz potentials in inhomogeneous and homogeneous Sobolev spaces. Even though we do not need all the results to solve the inverse problems considered in this article, we still state those variants since they are not given in earlier literature to the best of our knowledge. The strategy to prove results in this chapter is straightforward: if something is true for  $(-\Delta)^s$  when  $s \in (0, 1)$ , then by the splitting  $(-\Delta)^s = (-\Delta)^k (-\Delta)^{s-k}$  it should also be true for all powers s whenever the operations and claims are meaningful.

First we need a basic lemma for polyharmonic distributions, i.e. distributions which satisfy  $(-\Delta)^k g = 0$  for some integer  $k \in \mathbb{N}$ . We sketch the proof since it reflects the method of reduction we repeatedly use in this section.

**Lemma 3.1.** Let  $V \subset \mathbb{R}^n$  be any nonempty open set. If  $g \in \mathcal{D}'(\mathbb{R}^n)$  satisfies  $(-\Delta)^k g = 0$  and  $g|_V = 0$  for some  $k \in \mathbb{N}$ , then g = 0.

*Proof.* The proof is by induction. The case k = 1 is true since harmonic distributions are harmonic functions and therefore analytic [59]. Assume that the lemma holds for some k =

 $m \in \mathbb{N}$ . If  $(-\Delta)^{m+1}g = 0$  and  $g|_V = 0$ , then  $(-\Delta)^m((-\Delta)g) = 0$  and  $(-\Delta)g|_V = 0$  since  $(-\Delta)$  is a local operator. The induction assumption implies  $(-\Delta)g = 0$ , and since also  $g|_V = 0$ , we obtain g = 0 by harmonicity. This implies the claim. Alternatively one could use the fact that polyharmonic distributions are analytic [59, Theorem 7.30].

Now we can prove theorem 1.2. The idea is to reduce the general case back to the one where  $s \in (0, 1)$  and use the UCP proved in [28]. Note that the corresponding UCP cannot hold for local operators such as  $(-\Delta)^k$  when  $k \in \mathbb{N}$ . Therefore we have to assume that  $s \in \mathbb{R} \setminus \mathbb{Z}$ . For the proof of the case  $s \in (0, 1)$ , see [28, Theorem 1.2].

Proof of theorem 1.2. Because of our assumptions for u, the fractional Laplacian  $(-\Delta)^{s}u$  for  $s \in (-n/2, \infty) \setminus \mathbb{Z}$  is well-defined, see section 2.2. Assume that k - 1 < s < k for some  $k \in \mathbb{N}$ . Now we can split  $(-\Delta)^{s}u = (-\Delta)^{s-(k-1)}((-\Delta)^{k-1}u)$  where  $s - (k-1) \in (0,1)$ . Since the operator  $(-\Delta)^{k-1}$  is local, we obtain  $(-\Delta)^{s-(k-1)}((-\Delta)^{k-1}u)|_{V} = 0$  and  $(-\Delta)^{k-1}u|_{V} = 0$  where  $(-\Delta)^{k-1}u \in H^{r-2(k-1)}(\mathbb{R}^{n})$ . By the UCP of  $(-\Delta)^{s-(k-1)}$ , we have  $(-\Delta)^{k-1}u = 0$ . Since u is polyharmonic and  $u|_{V} = 0$ , lemma 3.1 implies u = 0.

If -n/2 < s < 0,  $s \notin \mathbb{Z}$ , choose  $k \in \mathbb{N}$  such that k + s > 0. Then by the locality of  $(-\Delta)^k$  we obtain  $(-\Delta)^{k+s}u|_V = 0$  and  $u|_V = 0$ . The first part of the proof implies the claim.

Note that theorem 1.2 implies UCP for equations of the type  $(-\Delta)^s u + Lu = 0$  where L is any local operator. Especially, this holds if L = P(x, D) where

$$P(x,D) = \sum_{|\alpha| \le m} a_{\alpha}(x) D^{\alpha}$$

is a differential operator of order m.

The following unique continuation result of Riesz potentials was presented in [41]. We use it to show uniqueness for partial data problems of the d-plane transform in section 4. We recall the short proof since it relies on the UCP of the fractional Laplacian.

**Corollary 3.2.** Let  $\alpha \in \mathbb{R}$  such that  $0 < \alpha < n$  and  $(\alpha - n)/2 \in \mathbb{R} \setminus \mathbb{Z}$ . Let  $f \in \mathscr{O}'_C(\mathbb{R}^n)$  and  $V \subset \mathbb{R}^n$  some nonempty open set. If  $I_{\alpha}f|_V = 0$  and  $f|_V = 0$ , then f = 0.

Proof. Recall that  $f \in H^r(\mathbb{R}^n)$  for some  $r \in \mathbb{R}$ . We can write  $I_{\alpha}f = (-\Delta)^{-s}f$  where  $s = (n-\alpha)/2$ . Choose  $k \in \mathbb{N}$  such that k-s > 0. By locality of  $(-\Delta)^k$  we obtain the conditions  $(-\Delta)^{k-s}f|_V = 0$  and  $f|_V = 0$ . Theorem 1.2 implies f = 0.

It is also independently proved in [41], without using the UCP of  $(-\Delta)^s$ , that if  $f \in \mathcal{E}'(\mathbb{R}^n)$ , then one can replace the condition  $I_{\alpha}f|_V = 0$  by the requirement  $\partial^{\beta}(I_{\alpha}f)(x_0) = 0$  for some  $x_0 \in V$  and all  $\beta \in \mathbb{N}_0^n$ . In fact, this can be used to prove a slightly stronger result for  $(-\Delta)^s$ in the case of compact support.

**Corollary 3.3.** Let  $u \in \mathcal{E}'(\mathbb{R}^n)$ ,  $V \subset \mathbb{R}^n$  some nonempty open set and  $s \in (-n/2, \infty) \setminus \mathbb{Z}$ . If  $\partial^{\beta}((-\Delta)^s u)(x_0) = 0$  and  $u|_V = 0$  for some  $x_0 \in V$  and all  $\beta \in \mathbb{N}^n_0$ , then u = 0.

Proof. Let k-1 < s < k where  $k \in \mathbb{N}$ . Now  $(-\Delta)^s = (-\Delta)^k (-\Delta)^{s-k} = (-\Delta)^k I_{\alpha}$  where  $\alpha = n + 2s - 2k \in (n-2, n)$ . Furthermore,  $\partial^{\beta} (-\Delta)^s u = \partial^{\beta} I_{\alpha} (-\Delta)^k u$  since the Riesz potential commutes with derivatives. By the locality of  $(-\Delta)^k$  we obtain the conditions  $\partial^{\beta} (I_{\alpha} (-\Delta)^k u)(x_0) = 0$  and  $(-\Delta)^k u|_V = 0$  where  $(-\Delta)^k u \in \mathcal{E}'(\mathbb{R}^n)$ . By [41, Theorem 1.1], we must have  $(-\Delta)^k u = 0$ . Since also  $u|_V = 0$ , we obtain u = 0 by lemma 3.1.

Let then  $s \in (-n/2, 0)$ ,  $s \notin \mathbb{Z}$ , and pick  $k \in \mathbb{N}$  such that s + k > 0. All the derivatives  $\partial^{\beta}((-\Delta)^{s}u)(x_{0})$  vanish, and hence  $((-\Delta)^{k}\partial^{\beta})((-\Delta)^{s}u)(x_{0}) = 0$ . Now  $((-\Delta)^{k}\partial^{\beta})((-\Delta)^{s}u) = \partial^{\beta}((-\Delta)^{s+k}u)$  and we get the conditions  $\partial^{\beta}((-\Delta)^{s+k}u)(x_{0}) = 0$  and  $u|_{V} = 0$ . The first part of the proof gives the claim.

The UCP of  $(-\Delta)^s$  also extends to homogeneous Sobolev spaces. The following result is a simple consequence of theorem 1.2. See [22, 23] for related results (strong UCP and measurable UCP in some special cases).

**Corollary 3.4.** Let  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$  and  $u \in \dot{H}^r(\mathbb{R}^n)$ , r < n/2. If  $(-\Delta)^s u|_V = 0$  and  $u|_V = 0$  for some nonempty open set  $V \subset \mathbb{R}^n$ , then u = 0.

Proof. If r < 0, then  $u \in H^r(\mathbb{R}^n)$  and the claim follows from theorem 1.2. Let r > 0 and choose  $k \in \mathbb{N}$  such that r - 2k < 0. Now  $(-\Delta)^k (-\Delta)^s = (-\Delta)^s (-\Delta)^k$  holds in  $\mathscr{S}_0(\mathbb{R}^n)$  so by the density of  $\mathscr{S}_0(\mathbb{R}^n)$  and the locality of  $(-\Delta)^k$  we obtain  $(-\Delta)^s ((-\Delta)^k u)|_V = 0$  and  $(-\Delta)^k u|_V = 0$ , where  $(-\Delta)^k u \in \dot{H}^{r-2k}(\mathbb{R}^n) \subset H^{r-2k}(\mathbb{R}^n)$ . Hence  $(-\Delta)^k u = 0$  by theorem 1.2 and since  $u|_V = 0$  we obtain u = 0 by lemma 3.1.

Since  $(-\Delta)^k (-\Delta)^{-s} = (-\Delta)^{k-s}$  also holds by the density of  $\mathscr{S}_0(\mathbb{R}^n)$ , one can reduce the case of negative exponents to the case of positive exponents. Thus one obtains the corresponding UCP for the Riesz potential  $I_{\alpha}$  in  $\dot{H}^r(\mathbb{R}^n)$  where  $r < \alpha - n/2$ . By the Sobolev embedding theorem we obtain the following unique continuation result for Bessel potential spaces when  $1 \le p \le 2$ .

**Corollary 3.5.** Let  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$ ,  $1 \le p \le 2$  and  $u \in H^{r,p}(\mathbb{R}^n)$ ,  $r \in \mathbb{R}$ . If  $(-\Delta)^s u|_V = 0$  and  $u|_V = 0$  for some nonempty open set  $V \subset \mathbb{R}^n$ , then u = 0.

Proof. If p = 1, then  $\mathcal{F}^{-1}(\langle \cdot \rangle^r \hat{u}) \in L^1(\mathbb{R}^n)$  which implies  $\langle \cdot \rangle^r \hat{u} \in L^{\infty}(\mathbb{R}^n)$  since  $\mathcal{F} \colon L^1(\mathbb{R}^n) \to L^{\infty}(\mathbb{R}^n)$  is continuous. Hence  $u \in H^t(\mathbb{R}^n)$  for some  $t \in \mathbb{R}$  and the claim follows from theorem 1.2. Let then  $1 . By the Sobolev embedding theorem (see e.g. [6, Theorem 6.5.1]) <math>H^{r,p}(\mathbb{R}^n) \hookrightarrow H^{r_1,p_1}(\mathbb{R}^n)$  when  $r_1 \leq r, 1 and$ 

$$r - \frac{n}{p} = r_1 - \frac{n}{p_1}.$$

Choose  $p_1 = 2$ . Then for any 1 the previous equality holds when

$$r_1 = \frac{2rp + n(p-2)}{2p} \le r.$$

Hence  $u \in H^{r_1,2}(\mathbb{R}^n) = H^{r_1}(\mathbb{R}^n)$  and by theorem 1.2 we obtain u = 0.

For higher exponents p, we can prove the following version of unique continuation considering the Fourier transform.

**Corollary 3.6.** Let  $r \ge 0$ ,  $2 \le p < \infty$  and  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$ . Let  $u \in H^{r,p}(\mathbb{R}^n)$  and  $V \subset \mathbb{R}^n$  some nonempty open set. If  $(-\Delta)^s \hat{u}|_V = 0$  and  $\hat{u}|_V = 0$ , then u = 0.

Proof. By the inclusion  $H^{r,p}(\mathbb{R}^n) \hookrightarrow L^p(\mathbb{R}^n)$  for  $r \ge 0$ , we can assume  $u \in L^p(\mathbb{R}^n)$ . If p = 2, then  $\hat{u} \in L^2(\mathbb{R}^n)$ . By theorem 1.2, we obtain  $\hat{u} = 0$  and hence u = 0. If  $2 , then we have that <math>\hat{u} \in H^{-t}(\mathbb{R}^n)$  where t > n(1/2 - 1/p) by [38, Theorem 7.9.3]. Again we obtain  $\hat{u} = 0$  by theorem 1.2 and eventually u = 0.

Note that if u has compact support, then by the Paley-Wiener theorem the condition  $\hat{u}|_V = 0$ already implies that u = 0.

3.2. The fractional Poincaré inequality. This subsection is dedicated to the proofs of a fractional Poincaré inequality. It serves the goal of estimating the  $L^2$ -norm of  $u \in \tilde{H}^s(\Omega)$  with that of its fractional Laplacian  $(-\Delta)^{s/2}u$ . We give five possible proofs for the fractional Poincaré inequality. We believe that giving several proofs will be helpful in subsequent works. This also illustrates some connections between methods which might have been unnoticed before.

The first proof is the most direct one and is based on splitting of frequencies on the Fourier side. The second proof utilizes several estimates (most importantly Hardy-Littlewood-Sobolev inequalities). This proof is motivated by the approach taken in [28]. Third proof uses a reduction argument to extend the inequality proved in [11] for all powers  $s \ge 0$ . Fourth proof is based on interpolation of homogeneous Sobolev spaces and it also gives an explicit constant in terms of the classical Poincaré constant. Fifth proof uses uncertainty inequalities which are treated in [24].

We begin our first proof by dividing the Fourier side into high and low frequencies. We only use simple estimates in the proof. In this approach we also get a control on the Poincaré constant. The result is basically the same as [4, Proposition 1.55].

**Theorem 3.7** (Poincaré inequality). Let  $s \ge 0$ ,  $K \subset \mathbb{R}^n$  compact set and  $u \in H^s_K(\mathbb{R}^n)$ . There exists a constant c = c(n, K, s) > 0 such that

$$\|u\|_{L^{2}(\mathbb{R}^{n})} \leq c \left\| (-\Delta)^{s/2} u \right\|_{L^{2}(\mathbb{R}^{n})}$$

*Proof.* We divide the integration into high and low frequencies

$$||u||_{L^{2}(\mathbb{R}^{n})}^{2} = \int_{|\xi| \leq \epsilon} |\hat{u}(\xi)|^{2} \,\mathrm{d}\xi + \int_{|\xi| > \epsilon} |\hat{u}(\xi)|^{2} \,\mathrm{d}\xi$$

where  $\epsilon > 0$  is determined later on. Let us analyze the first part. Since  $u \in L^2(\mathbb{R}^n)$  and has support in K, Hölder's inequality implies

$$|\hat{u}(\xi)| \le ||u||_{L^1(\mathbb{R}^n)} \le |K|^{1/2} ||u||_{L^2(\mathbb{R}^n)}$$

Thus we have

$$\int_{|\xi| \le \epsilon} |\hat{u}(\xi)|^2 \,\mathrm{d}\xi \le \int_{|\xi| \le \epsilon} |K| \, \|u\|_{L^2(\mathbb{R}^n)}^2 \,\mathrm{d}\xi = \epsilon^n \,|K| \,|B(0,1)| \,\|u\|_{L^2(\mathbb{R}^n)}^2$$

where |K| and |B(0,1)| are the measures of K and the unit ball B(0,1). For high frequencies we can do the following trick

$$\int_{|\xi|>\epsilon} |\hat{u}(\xi)|^2 \,\mathrm{d}\xi = \int_{|\xi|>\epsilon} \frac{|\xi|^{2s} |\hat{u}(\xi)|^2}{|\xi|^{2s}} \mathrm{d}\xi \le \epsilon^{-2s} \left\| (-\Delta)^{s/2} u \right\|_{L^2(\mathbb{R}^n)}^2$$

Now choose  $0 < \epsilon < (|K| |B(0,1)|)^{-1/n}$ . Then one obtains the inequality

$$\|u\|_{L^{2}(\mathbb{R}^{n})} \leq \frac{\epsilon^{-s}}{\sqrt{1-\epsilon^{n}|K||B(0,1)|}} \left\| (-\Delta)^{s/2} u \right\|_{L^{2}(\mathbb{R}^{n})}. \quad \Box$$

**Remark 3.8.** Choosing  $\epsilon = (2 |K| |B(0,1)|)^{-1/n}$  one obtains the following inequality in theorem 3.7

$$\|u\|_{L^{2}(\mathbb{R}^{n})} \leq \sqrt{2} (2|K||B(0,1)|)^{s/n} \left\| (-\Delta)^{s/2} u \right\|_{L^{2}(\mathbb{R}^{n})}$$

If K is a ball, the constant in this inequality has the same scaling with respect to the diameter of the set as in theorem 3.17, i.e.  $c \approx (diam(K))^s$ . Further, one can use similar method of proof as in theorem 3.7 to show Poincaré inequalities for more general pseudodifferential operators on certain manifolds. See [84] for details.

Provided we have the Poincaré inequality, we can prove the generalized version of it. See also [4, Corollary 1.56] for a similar inequality when K is a ball. In that case one can take  $\tilde{c} \approx (\operatorname{diam}(K))^{s-t}$ . The cases  $s \geq t \geq 1$  and  $s \geq 1 \geq t \geq 0$  are also proved for  $u \in \tilde{H}^s(\Omega)$  in theorem 3.17.

Proof of theorem 1.5. Since  $s \ge t \ge 0$  we have the continuous embeddings  $H^t(\mathbb{R}^n) \hookrightarrow \dot{H}^t(\mathbb{R}^n)$ and  $H^s(\mathbb{R}^n) \hookrightarrow H^t(\mathbb{R}^n)$ . Using the Poincaré inequality in theorem 3.7 we obtain

$$\begin{split} \left\| (-\Delta)^{t/2} u \right\|_{L^{2}(\mathbb{R}^{n})} &= \| u \|_{\dot{H}^{t}(\mathbb{R}^{n})} \leq \| u \|_{H^{t}(\mathbb{R}^{n})} \leq \| u \|_{H^{s}(\mathbb{R}^{n})} \leq 2^{\frac{s+1}{2}} \left( \left\| (u \|_{L^{2}(\mathbb{R}^{n})} + \| u \|_{\dot{H}^{s}(\mathbb{R}^{n})} \right) \\ &\leq 2^{\frac{s+1}{2}} \left( c \left\| (-\Delta)^{s/2} u \right\|_{L^{2}(\mathbb{R}^{n})} + \left\| (-\Delta)^{s/2} u \right\|_{L^{2}(\mathbb{R}^{n})} \right) \\ &= \tilde{c} \left\| (-\Delta)^{s/2} u \right\|_{L^{2}(\mathbb{R}^{n})} \end{split}$$

where the constants c and  $\tilde{c}$  do not depend on u. In the fourth step we used the elementary inequality  $(a+b)^r \leq 2^r(a^r+b^r)$  for  $a,b \geq 0$ . This concludes the proof.

We then start preparation for our second proof by stating some known lemmas:

- lemma 3.9 is the continuity of Riesz potentials,
- lemma 3.10 is the  $L^2$  boundedness of inverse of elliptic second order operators,
- lemma 3.11 is a convolution  $L^p$  estimate from below by an inhomogeneous Hölder norm,
- lemma 3.12 is a specific form of the Poincaré inequality for fractional Laplacians, and
- lemma 3.13 is a simple commutation property for the gradient and a Fourier multiplier.

**Lemma 3.9** (Theorem 4.5.3 in [38]). Let  $t \ge 0$ , 1 be such that <math>n > tp, and define  $q = \frac{np}{n-tp}$ . Then the Riesz potential  $(-\Delta)^{-t/2} : L^p(\mathbb{R}^n) \to L^q(\mathbb{R}^n)$  is continuous.

**Lemma 3.10** (Section 6 in [21]). Let  $\Omega \subset \mathbb{R}^n$  be a bounded domain and  $f \in L^2(\Omega)$ . If  $w \in H^1_0(\Omega)$  is the unique solution of the problem

$$\begin{cases} (-\Delta)w &= f \ in \ \Omega \\ w|_{\partial\Omega} &= 0 \end{cases}$$

then there exists a constant  $C = C(\Omega)$  such that

(5) 
$$||w||_{L^2(\Omega)} \le C ||f||_{L^2(\Omega)}$$
.

**Lemma 3.11** (Theorem 4.5.10 in [38]). Let  $\psi \in C^1(\mathbb{R}^n \setminus \{0\})$  be homogeneous of degree -n/a,  $p \in [1, \infty]$  and  $\gamma = n(1 - 1/a - 1/p)$  be such that  $\gamma \in (0, 1)$ . Then if  $v \in L^p(\mathbb{R}^n) \cap \mathcal{E}'(\mathbb{R}^n)$  we have

$$\sup_{x \neq y} \left\{ \frac{|(\psi * v)(x) - (\psi * v)(y)|}{|x - y|^{\gamma}} \right\} \le C ||v||_{L^{p}(\mathbb{R}^{n})},$$

where C does not depend on w.

**Lemma 3.12** (Formula (1.3) in [64]). Let  $1 and <math>f \in W^{n/p,p}(\mathbb{R}^n)$ . There is a constant C = C(n,p) such that

(6) 
$$\|f\|_{L^q(\mathbb{R}^n)} \le Cq^{1-1/p} \|(-\Delta)^{n/2p} f\|_{L^p(\mathbb{R}^n)}^{1-p/q} \|f\|_{L^p(\mathbb{R}^n)}^{p/q}$$

This estimate is proved using sharp Hardy-Littlewood-Sobolev inequalities.

**Lemma 3.13.** Let  $t \ge 0$  and  $f \in H^{1+2t}(\mathbb{R}^n)$ . Then  $[\nabla, (-\Delta)^t]f = 0$ , that is, the gradient and the fractional Laplacian of exponent t commute.

*Proof.* The proof is just a trivial computation with Fourier symbols:

$$\mathcal{F}(\nabla(-\Delta)^t f) = i\xi |\xi|^{2t} \hat{f}(\xi) = |\xi|^{2t} i\xi \hat{f}(\xi) = \mathcal{F}((-\Delta)^t (\nabla f)) .$$

We are now ready to state and prove the fractional Poincaré inequality.

**Theorem 3.14** (Poincaré inequality). Let  $\Omega \subset \mathbb{R}^n$  be a bounded domain,  $s \in [0, \infty)$  and  $u \in \widetilde{H}^s(\Omega)$ . There exists a constant  $c = c(n, \Omega, s)$  such that

$$||u||_{L^2(\mathbb{R}^n)} \le c ||(-\Delta)^{s/2}u||_{L^2(\mathbb{R}^n)}.$$

*Proof.* In the inequalities the constants (usually denoted by c, C, etc.) do not depend on the function which is being estimated and can change from line to line. We let the symbol  $s' = s - \lfloor s \rfloor$  indicate the fractional part of the exponent s, with the convention that  $s' \in [0, 1)$ . First observe that by using lemma 3.9 with p = 2 and Hölder's inequality we get the following useful estimate

(7) 
$$\|u\|_{L^{2}(\mathbb{R}^{n})} \leq C_{\Omega} \|u\|_{L^{q}(\mathbb{R}^{n})} \leq c \|(-\Delta)^{t/2} u\|_{L^{2}(\mathbb{R}^{n})}$$

when  $u \in \widetilde{H}^t(\Omega)$  where q and t are as in lemma 3.9. Our proof is divided in several cases.

Case 1:  $\lfloor s \rfloor \in 2\mathbb{Z}, s' = 0.$ 

Recall that  $\tilde{\tilde{H}}^{2h}(\Omega) \subset H_0^{2h}(\Omega)$ . We show that if  $u \in H_0^{2h}(\Omega)$  and  $h \in \mathbb{N}$  then there exists a constant  $c = c(n, \Omega, h)$  such that

(8) 
$$\|(-\Delta)^{h}u\|_{L^{2}(\mathbb{R}^{n})} \ge c\|u\|_{L^{2}(\mathbb{R}^{n})}.$$

The estimate (8) holds trivially if h = 0, while if h = 1 then (8) follows from the boundedness of the inverse lemma 3.10. Assume now that  $h \ge 2$ , and by induction that (8) holds for h - 1. Then  $(-\Delta)u \in H_0^{2h-2}(\Omega)$ , so we can apply (8) and (5) to get

$$\|(-\Delta)^{h}u\|_{L^{2}(\mathbb{R}^{n})} = \|(-\Delta)^{h-1}(-\Delta u)\|_{L^{2}(\mathbb{R}^{n})} \ge c\|(-\Delta)u\|_{L^{2}(\mathbb{R}^{n})} \ge c'\|u\|_{L^{2}(\mathbb{R}^{n})}.$$

In the next steps we consider  $s \notin \mathbb{N}$ .

Case 2:  $[s] \in 2\mathbb{Z}, s' \in (0, 1/2)$  or  $[s] \in 2\mathbb{Z}, s' \in [1/2, 1), n \ge 2$ .

Now it holds that n > 2s', and there exists  $k \in \mathbb{N}$  such that  $s \in (2k, 2k+1)$  and we can write  $(-\Delta)^{s/2}u = (-\Delta)^{s'/2}(-\Delta)^k u$ . Since  $(-\Delta)^k u \in \widetilde{H}^{s-2k}(\Omega) = \widetilde{H}^{s'}(\Omega)$ , we can apply formula (7)  $\|(-\Delta)^k u\|_{L^2(\mathbb{R}^n)} \leq c \|(-\Delta)^{s/2}u\|_{L^2(\mathbb{R}^n)}$ .

Since  $u \in H_0^s(\Omega) \subset H_0^{2k}(\Omega)$ , we can get the result using formula (8).

# Case 3: $|s| \in 2\mathbb{Z}, s' \in (1/2, 1), n = 1.$

As in the second case, there exists  $k \in \mathbb{N}$  such that  $s \in (2k, 2k+1)$  and we can write  $(-\Delta)^{s/2}u = (-\Delta)^{s'/2}(-\Delta)^k u$ . However, since now n < 2s', we cannot directly use formula (7).

Assume first that  $w \in C_c^{\infty}(\Omega)$ . Then we can take  $y_0 \in \Omega$  such that  $w(y_0) = 0$  and  $x_0 \in \Omega$  such that  $w(x_0) = \|w\|_{L^{\infty}(\Omega)}$ . With these choices and for any  $\gamma > 0$  we can write

(9) 
$$\|w\|_{L^2(\mathbb{R}^n)} \le C \|w\|_{L^\infty(\Omega)} \le C \frac{w(x_0) - w(y_0)}{|x_0 - y_0|^{\gamma}}$$

We now let  $\gamma = s' - n/2 = s' - 1/2 \in (0, 1/2)$ , and define  $\psi = |x|^{s'-1}$ ,  $v = (-\Delta)^{s'/2}w$ . By the mapping properties of the fractional Laplacian and the Mikhlin theorem, we can observe that  $v \in L^p(\mathbb{R})$  for all 1 (see [1, Theorem 7.2]). Using the continuity of the Riesz potential in lemma 3.9, we see that for a constant <math>c = c(n, s) the following holds almost everywhere:

$$w = (-\Delta)^{-s'/2}((-\Delta)^{s'/2}w) = (-\Delta)^{-s'/2}v = cI_{1-s'}v = c|x|^{s'-1} * v = c(\psi * v).$$

Let  $\chi_R$  be the characteristic function of the ball  $B_R$  of radius R > 0, and define  $w_R = c(\psi * (\chi_R v))$ , with c as above. We see that

$$w_R(x) = c(\psi * (\chi_R v))(x) = c \int_{\mathbb{R}} \psi(x - y)\chi_R(y)v(y)dy$$

and the integrand is dominated by  $|\psi(x-y)v(y)|$ . This is an integrable function, since

$$\int_{\mathbb{R}} |\psi(x-y)v(y)| dy = \int_{\mathbb{R}} \psi(x-y)|v(y)| dy = I_{1-s'}(|v|)(x) ,$$

and the Riesz potential is well defined almost everywhere on  $L^p(\mathbb{R})$  for any 1 . Now $the dominated convergence theorem gives that <math>w_R(x) \to w(x)$  as  $R \to \infty$  for almost every fixed  $x \in \mathbb{R}$ .

Let  $\epsilon > 0$  and  $x'_0, y'_0 \in \mathbb{R}$  be such that  $|x_0 - x'_0| < \epsilon$ ,  $|y_0 - y'_0| < \epsilon$  and  $w_R(x'_0)$ ,  $w_R(y'_0)$  converge to  $w(x'_0)$ ,  $w(y'_0)$  as  $R \to \infty$ . Applying lemma 3.11 with p = 2, n = 1 and a = 1 - s', we see that

$$\frac{w_R(x'_0) - w_R(y'_0)}{|x_0 - y_0|^{\gamma}} \leq \sup_{x \neq y} \left\{ \frac{w_R(x) - w_R(y)}{|x - y|^{\gamma}} \right\} \\
= c \sup_{x \neq y} \left\{ \frac{(\psi * (\chi_R v))(x) - (\psi * (\chi_R v))(y)}{|x - y|^{\gamma}} \right\} \\
\leq C \|\chi_R v\|_{L^2(\mathbb{R})} \leq C \|v\|_{L^2(\mathbb{R})} = C \|(-\Delta)^{s'/2} w\|_{L^2(\mathbb{R})}$$

We now first take the limit for  $R \to \infty$  and then the one for  $\epsilon \to 0$ . By the smoothness of w, this gives

(10) 
$$\frac{w(x_0) - w(y_0)}{|x_0 - y_0|^{\gamma}} \le C \| (-\Delta)^{s'/2} w \|_{L^2(\mathbb{R})} .$$

Combining formulas (9) and (10) we get  $||w||_{L^2(\mathbb{R}^n)} \leq C||(-\Delta)^{s'/2}w||_{L^2(\mathbb{R}^n)}$ , and the same inequality holds for  $w \in \widetilde{H}^{s'}(\Omega)$  by density. Let now  $w := (-\Delta)^k u \in \widetilde{H}^{s-2k}(\Omega) = \widetilde{H}^{s'}(\Omega)$ . The result is then obtained applying formula (8).

# Case 4: $\lfloor s \rfloor \in 2\mathbb{Z}, \, s' = 1/2, \, n = 1.$

Let  $w := (-\Delta)^k u \in \widetilde{H}^{s-2k}(\Omega) = \widetilde{H}^{s'}(\Omega)$ . Here we make use of formula (6) with p = 2, q = 3 in order to estimate

(11) 
$$\|w\|_{L^{2}(\mathbb{R}^{n})} = \|w\|_{L^{2}(\mathbb{R}^{n})}^{3} \|w\|_{L^{2}(\mathbb{R}^{n})}^{-2} \le \|w\|_{L^{3}(\mathbb{R}^{n})}^{3} \|w\|_{L^{2}(\mathbb{R}^{n})}^{-2} \le C \|(-\Delta)^{n/4}w\|_{L^{2}(\mathbb{R}^{n})}.$$

Since n/4 equals s'/2 for n = 1, the results follows from (11) and (8).

# Case 5: $\lfloor s \rfloor \notin 2\mathbb{Z}$ .

Let  $u \in C_c^{\infty}(\Omega)$ . In this case s = s' + 2k + 1 for some  $k \in \mathbb{N}$ , therefore we can calculate

(12)  
$$\begin{aligned} \|(-\Delta)^{s/2}u\|_{L^{2}(\mathbb{R}^{n})} &= \|(-\Delta)^{1/2}(-\Delta)^{(s'+2k)/2}u\|_{L^{2}(\mathbb{R}^{n})} \\ &= \|\nabla(-\Delta)^{(s'+2k)/2}u\|_{L^{2}(\mathbb{R}^{n})} \\ &= \|(-\Delta)^{(s'+2k)/2}\nabla u\|_{L^{2}(\mathbb{R}^{n})} \\ &\geq C\|\nabla u\|_{L^{2}(\mathbb{R}^{n})} \geq C\|u\|_{L^{2}(\mathbb{R}^{n})} .\end{aligned}$$

The second equality in (12) is just an  $L^2$  property of the gradient and the  $(-\Delta)^{1/2}$  operator. The third equality in (12) follows from lemma 3.13. The first inequality in (12) follows from the even cases, given that  $\lfloor s' + 2k \rfloor \in 2\mathbb{Z}$  and  $\nabla u \in \widetilde{H}^{s'+2k}(\Omega)$  componentwise. The last inequality follows from the classical Poincaré inequality. The rest follows by approximation.

**Remark 3.15.** Third way to prove the Poincaré inequality is using the known result in the case  $n \ge 1$  and  $s \in (0,1)$  [11, Lemma 2.2]. This result is proved using Caffarelli-Silvestre extension. Then one can use similar reduction argument to prove it for all  $s \ge 0$  and  $u \in C_c^{\infty}(\Omega)$ . Namely, one shows using the classical Poincaré inequality that the claim holds for all  $s \in [0,2)$ . The higher order fractional cases are obtained by splitting the fractional Laplacian as  $(-\Delta)^s = (-\Delta)^k (-\Delta)^{t/2}$  where  $t \in (0,2)$ . Boundedness of the inverse and the fractional Poincaré inequality for  $t \in (0,2)$  imply the claim for fractional exponents. Integer order exponents are obtained from the boundedness of the inverse as before. The inequality for  $u \in \widetilde{H}^s(\Omega)$  follows by approximation.

For the fourth proof we use the following interpolation lemma of homogeneous Sobolev spaces which is a simple consequence of Hölder's inequality, see [4, Proposition 1.32].

Lemma 3.16. Let 
$$s_0 \leq r \leq s_1$$
 and  $f \in \dot{H}^{s_0}(\mathbb{R}^n) \cap \dot{H}^{s_1}(\mathbb{R}^n)$ . Then  $f \in \dot{H}^r(\mathbb{R}^n)$  and  
 $\|f\|_{\dot{H}^r(\mathbb{R}^n)} \leq \|f\|_{\dot{H}^{s_0}(\mathbb{R}^n)}^{1-\theta} \|f\|_{\dot{H}^{s_1}(\mathbb{R}^n)}^{\theta}, \quad r = (1-\theta)s_0 + \theta s_1.$ 

Using the interpolation lemma and the usual Poincaré inequality we can easily prove the following theorem. Note that we also obtain explicit constant from the proof.

**Theorem 3.17** (Poincaré inequality). Let  $s \ge t \ge 1$  or  $s \ge 1 \ge t \ge 0$ ,  $\Omega \subset \mathbb{R}^n$  bounded open set and  $u \in \widetilde{H}^s(\Omega)$ . The following inequality holds

$$\left\| (-\Delta)^{t/2} u \right\|_{L^2(\mathbb{R}^n)} \le C^{s-t} \left\| (-\Delta)^{s/2} u \right\|_{L^2(\mathbb{R}^n)}$$

where  $C = C(n, \Omega)$  is the classical Poincaré constant.

*Proof.* Let  $s \ge t \ge 1$  and  $u \in C_c^{\infty}(\Omega)$ . The usual Poincaré inequality can be written in terms of the homogeneous Sobolev norm as

$$\|u\|_{L^{2}(\mathbb{R}^{n})} = \|u\|_{L^{2}(\Omega)} \leq C \|\nabla u\|_{L^{2}(\Omega)} = C \|\nabla u\|_{L^{2}(\mathbb{R}^{n})} = C \|u\|_{\dot{H}^{1}(\mathbb{R}^{n})}$$

where  $C = C(n, \Omega)$ . We use the interpolation lemma 3.16 twice. First choose  $s_0 = 0$ , r = 1 and  $s_1 = t \ge 1$ . Now  $\theta = 1/t$  and by the classical Poincaré inequality we obtain

$$\|u\|_{\dot{H}^{1}(\mathbb{R}^{n})} \leq \|u\|_{L^{2}(\mathbb{R}^{n})}^{1-\theta} \|u\|_{\dot{H}^{t}(\mathbb{R}^{n})}^{\theta} \leq C^{1-\theta} \|u\|_{\dot{H}^{1}(\mathbb{R}^{n})}^{1-\theta} \|u\|_{\dot{H}^{t}(\mathbb{R}^{n})}^{\theta}$$

From this we get the following inequality

$$\|u\|_{\dot{H}^1(\mathbb{R}^n)} \le C^{\frac{1-\theta}{\theta}} \|u\|_{\dot{H}^t(\mathbb{R}^n)}$$

for all  $u \in C_c^{\infty}(\Omega)$ . Hence

$$||u||_{L^2(\mathbb{R}^n)} \le C ||u||_{\dot{H}^1(\mathbb{R}^n)} \le C^t ||u||_{\dot{H}^t(\mathbb{R}^n)}.$$

Then choose  $s_0 = 0$ , r = t and  $s_1 = s \ge t$  in lemma 3.16. Now  $\theta = t/s$  and by the previous inequality

$$\|u\|_{\dot{H}^{t}(\mathbb{R}^{n})} \leq \|u\|_{L^{2}(\mathbb{R}^{n})}^{1-\theta} \|u\|_{\dot{H}^{s}(\mathbb{R}^{n})}^{\theta} \leq C^{t(1-\theta)} \|u\|_{\dot{H}^{t}(\mathbb{R}^{n})}^{1-\theta} \|u\|_{\dot{H}^{s}(\mathbb{R}^{n})}^{\theta}.$$

From this we obtain

$$\left\| (-\Delta)^{t/2} u \right\|_{L^2(\mathbb{R}^n)} \le C^{s-t} \left\| (-\Delta)^{s/2} u \right\|_{L^2(\mathbb{R}^n)}$$

for  $u \in C_c^{\infty}(\Omega)$ .

Let then  $s \ge 1 \ge t \ge 0$  and  $u \in C_c^{\infty}(\Omega)$ . First interpolate for  $s \ge 1 \ge t$  to obtain

$$||u||_{\dot{H}^{1}(\mathbb{R}^{n})} \leq ||u||_{\dot{H}^{t}(\mathbb{R}^{n})}^{1-\theta} ||u||_{\dot{H}^{s}(\mathbb{R}^{n})}^{\theta}, \quad \theta = \frac{1-t}{s-t}.$$

Second, interpolate for  $1 \ge t \ge 0$  and use the previous inequality and the classical Poincaré inequality to get

$$\|u\|_{\dot{H}^{t}(\mathbb{R}^{n})} \leq \|u\|_{L^{2}(\mathbb{R}^{n})}^{1-\tilde{\theta}} \|u\|_{\dot{H}^{1}(\mathbb{R}^{n})}^{\tilde{\theta}} \leq C^{1-\tilde{\theta}} \|u\|_{\dot{H}^{t}(\mathbb{R}^{n})}^{1-\theta} \|u\|_{\dot{H}^{s}(\mathbb{R}^{n})}^{\theta}, \quad \tilde{\theta} = t,$$

which eventually implies the inequality

$$\left\| (-\Delta)^{t/2} u \right\|_{L^2(\mathbb{R}^n)} = \| u \|_{\dot{H}^t(\mathbb{R}^n)} \le C^{s-t} \| u \|_{\dot{H}^s(\mathbb{R}^n)} = C^{s-t} \left\| (-\Delta)^{s/2} u \right\|_{L^2(\mathbb{R}^n)}$$

for all  $u \in C_c^{\infty}(\Omega)$ .

Then let  $u \in \widetilde{H}^{s}(\Omega)$ . By definition there is a sequence  $\varphi_{k} \in C_{c}^{\infty}(\Omega)$  such that

$$\varphi_k \to u \quad \text{in } H^s(\mathbb{R}^n).$$

The continuity of  $(-\Delta)^{t/2}$  implies that

$$(-\Delta)^{t/2}\varphi_k \to (-\Delta)^{t/2}u$$
 in  $H^{s-t}(\mathbb{R}^n)$ .

The embedding  $H^{s-t}(\mathbb{R}^n) \hookrightarrow L^2(\mathbb{R}^n)$  is continuous and thus

$$(-\Delta)^{t/2}\varphi_k \to (-\Delta)^{t/2}u$$
 in  $L^2(\mathbb{R}^n)$ .

By the continuity of the norm and  $(-\Delta)^{s/2}$  we finally obtain

$$\left\| (-\Delta)^{t/2} u \right\|_{L^2(\mathbb{R}^n)} = \lim_k \left\| (-\Delta)^{t/2} \varphi_k \right\|_{L^2(\mathbb{R}^n)} \le C^{s-t} \lim_k \left\| (-\Delta)^{s/2} \varphi_k \right\|_{L^2(\mathbb{R}^n)}$$
$$= C^{s-t} \left\| (-\Delta)^{s/2} u \right\|_{L^2(\mathbb{R}^n)}.$$

We remark that the case t = 0 and s = 1 corresponds to the classical Poincaré inequality since  $\|\nabla u\|_{L^2(\mathbb{R}^n)} = \|(-\Delta)^{1/2}u\|_{L^2(\mathbb{R}^n)}$ . Also the constant  $C^{s-t}$  is the expected one. In the usual Poincaré inequality we take one derivative and the constant is C. In the higher order version we take t and s derivatives and the constant naturally becomes  $C^{s-t}$ . The constant C can be taken to be proportional to the diameter of the set,  $C \approx \text{diam}(\Omega)$ . **Remark 3.18.** Fifth way to prove the Poincaré inequality is using uncertainty inequalities. If  $u \in L^2(\mathbb{R}^n)$ , then there is a constant c = c(n, s) such that

(13) 
$$\|u\|_{L^{2}(\mathbb{R}^{n})}^{2} \leq c \, \||\cdot|^{s} \, u\|_{L^{2}(\mathbb{R}^{n})} \, \||\cdot|^{s} \, \hat{u}\|_{L^{2}(\mathbb{R}^{n})}$$

see the discussion after theorem 4.1 in [24]. We can interpret this inequality as

$$\|u\|_{L^{2}(\mathbb{R}^{n})}^{2} \leq c \left\| (-\Delta)^{s/2} (\mathcal{F}^{-1}(u)) \right\|_{L^{2}(\mathbb{R}^{n})} \left\| (-\Delta)^{s/2} u \right\|_{L^{2}(\mathbb{R}^{n})}$$

whenever the terms on the right hand side of equation (13) are finite. If u is supported in some fixed compact set K, then one obtains similar inequality as in theorem 3.7, i.e.

$$\|u\|_{L^2(\mathbb{R}^n)} \le c' \left\| (-\Delta)^{s/2} u \right\|_{L^2(\mathbb{R}^n)}$$

holds for all  $u \in H^s_K(\mathbb{R}^n)$  and for some constant c' = c'(n, K, s).

**Remark 3.19.** The Poincaré inequality for the operator  $(-\Delta)^{s/2}$  implies also Poincaré inequality for the fractional gradient  $\nabla^s \colon H^s(\mathbb{R}^n) \to L^2(\mathbb{R}^{2n}, \mathbb{M}^{\lfloor s \rfloor + 1})$  which is defined as

$$\nabla^s u(x,y) := \frac{\mathcal{C}_{n,s'}^{1/2}}{\sqrt{2}} \frac{\nabla^{\lfloor s \rfloor} u(x) - \nabla^{\lfloor s \rfloor} u(y)}{|y-x|^{n/2+s'+1}} \otimes (y-x),$$

see section 6 for more details. If  $s \ge t \ge 0$  and  $u \in C_c^{\infty}(\Omega)$ , then

$$\left\|\nabla^{t} u\right\|_{L^{2}(\mathbb{R}^{2n},\mathbb{M}^{\lfloor s\rfloor+1})} = \left\|(-\Delta)^{t/2} u\right\|_{L^{2}(\mathbb{R}^{n})} \leq \tilde{c} \left\|(-\Delta)^{s/2} u\right\|_{L^{2}(\mathbb{R}^{n})} = \tilde{c} \left\|\nabla^{s} u\right\|_{L^{2}(\mathbb{R}^{2n},\mathbb{M}^{\lfloor s\rfloor+1})},$$

where the constant  $\tilde{c}$  does not depend on u. By approximation and the continuity of  $\nabla^s$  the previous inequality is also true for  $u \in \tilde{H}^s(\Omega)$ .

# 4. Applications to integral geometry

In this section we discuss how the UCP of Riesz potentials can be used in partial data problems in integral geometry. We follow [36] for the treatment of the *d*-plane transform, theory of X-ray transform and Radon transform can also be found in [63, 68, 81]. Let  $d \in \{1, \ldots, n-1\}$  and denote by  $\mathbf{P}^d$  the space of all *d*-dimensional affine planes in  $\mathbb{R}^n$ . We define the *d*-plane transform of a function f to be

$$R_d f(A) = \int_{x \in A} f(x) \mathrm{d}m(x)$$

where  $A \in \mathbf{P}^d$  and m is the Hausdorff measure on A. The adjoint of  $R_d$  is defined as

$$R_d^*g(x) = \int_{A \ni x} g(A) \mathrm{d}\mu(A)$$

where g is a function on  $\mathbf{P}^d$  and  $\mu$  is the associated measure. These transforms are defined for functions such that the integrals exist. The case d = 1 corresponds to the usual X-ray transform and d = n-1 to the Radon transform. The normal operator of the d-plane transform  $N_d = R_d^* R_d$ has an expression  $N_d f = c_{n,d} (f * |\cdot|^{-(n-d)})$  where  $c_{n,d}$  is a constant depending on n and d. The normal operator is well defined if f is a function that decreases rapidly enough at infinity [36]. This holds for example if  $f \in C_{\infty}(\mathbb{R}^n)$  where  $C_{\infty}(\mathbb{R}^n)$  is the space of continuous functions which decrease faster than any polynomial at infinity (see section 2.1 for a precise definition). Thus, up to a constant factor,  $N_d$  can be represented as a Riesz potential  $N_d = I_{\alpha} = (-\Delta)^{-d/2}$  where  $\alpha = n - d \in \{1, \ldots, n-1\}$ .

The transforms  $R_d$  and  $R_d^*$  can be extended to distributions by duality. Let  $f \in \mathcal{E}'(\mathbb{R}^n)$  and  $g \in \mathcal{D}'(\mathbb{R}^n)$ . Since  $R_d: \mathcal{D}(\mathbb{R}^n) \to \mathcal{D}(\mathbf{P}^d)$  and  $R_d^*: \mathcal{E}(\mathbf{P}^d) \to \mathcal{E}(\mathbb{R}^n)$  are continuous [32], we can define the following operations

$$\langle R_d f, \psi \rangle = \langle f, R_d^* \psi \rangle, \quad \psi \in \mathcal{E}(\mathbf{P}^d) \langle R_d^* g, \varphi \rangle = \langle g, R_d \varphi \rangle, \quad \varphi \in \mathcal{D}(\mathbb{R}^n).$$

Therefore  $R_d f \in \mathcal{E}'(\mathbf{P}^d)$  and  $R_d^* g \in \mathcal{D}'(\mathbb{R}^n)$ . This shows that the normal operator  $N_d = R_d^* R_d: \mathcal{E}'(\mathbb{R}^n) \to \mathcal{D}'(\mathbb{R}^n)$  is always defined and  $N_d f = c_{n,d}(f * |\cdot|^{-(n-d)})$  holds in the sense of distributions. Let  $V \subset \mathbb{R}^n$  be a nonempty open set and  $f \in \mathcal{E}'(\mathbb{R}^n)$ . We say that  $R_d f$  vanishes on all *d*-planes intersecting *V*, if  $\langle R_d f, \varphi \rangle = 0$  for all  $\varphi \in C_c^{\infty}(\mathbf{P}_V^d)$  where  $\mathbf{P}_V^d$  is the set of all *d*-planes intersecting *V*. If V = B(0, R) is a ball,  $\varphi \in C_c^{\infty}(\mathbf{P}_V^d)$  means that  $\varphi$  is smooth and  $\varphi(A) = 0$  for all *d*-planes *A* for which d(0, A) > r for some r < R. For more details on the range of the *d*-plane transform and duality in integral geometry, see [32] and [36, Chapter II].

**Remark 4.1.** The UCP of Riesz potentials (corollary 3.2) immediately implies the UCP of the normal operator of the d-plane transform when d is odd (corollary 1.3) since  $N_d \approx I_{n-d}$  and  $d/2 \notin \mathbb{Z}$ . However, such UCP cannot hold if d is even, which can be shown by contradiction. Assume that corollary 1.3 holds when d is even. Take any nonzero  $f \in C_c^{\infty}(\mathbb{R}^n)$ . By the properties of the Fourier transform and Riesz potentials we have  $(-\Delta)^{d/2}f = (-\Delta)^{-d/2}((-\Delta)^d f) =$  $N_d(-\Delta)^d f$  up to a constant factor. Since d is even  $(-\Delta)^{d/2}$  is a local operator and we obtain  $N_d(-\Delta)^d f|_V = (-\Delta)^d f|_V = 0$  where  $V \subset \mathbb{R}^n$  is an open set outside the support of f and  $(-\Delta)^d f \in C_c^{\infty}(\mathbb{R}^n)$ . By the assumption we would get that  $(-\Delta)^d f = 0$ , i.e. f is polyharmonic. But this implies f = 0 by lemma 3.1, which is a contradiction. Hence the UCP cannot hold for  $N_d$  when d even.

Using the UCP of  $N_d$  we can now prove corollary 1.4.

Proof of corollary 1.3. Consider first  $f \in C_{\infty}(\mathbb{R}^n)$ . Taking the adjoint, we get the conditions  $N_d f|_V = 0$  and  $f|_V = 0$ . By corollary 1.3 we obtain f = 0 whenever d is odd. Then let  $f \in \mathcal{E}'(\mathbb{R}^n)$ . We can assume that V = B(0, R) is a ball of radius R centered at the origin. As in [36] we define the "convolution"

$$(g \times \varphi)(A) = \int_{\mathbb{R}^n} g(y)\varphi(A-y)\mathrm{d}y$$

where  $g \in C_c^{\infty}(\mathbb{R}^n)$ ,  $\varphi \in C_c^{\infty}(\mathbf{P}^d)$ ,  $A \in \mathbf{P}^d$  and A - y is a *d*-plane shifted by  $y \in \mathbb{R}^n$ . Then one can calculate that  $R_d^*(g \times \varphi) = g * R_d^*\varphi$  (see [36, Proof of theorem 5.4]). Let  $j_{\epsilon} \in C_c^{\infty}(\mathbb{R}^n)$  be the standard mollifier and consider the mollifications  $f * j_{\epsilon} \in C_c^{\infty}(\mathbb{R}^n)$ . By the properties of the convolutions

(14) 
$$\langle R_d(f*j_{\epsilon}),\varphi\rangle = \langle f*j_{\epsilon}, R_d^*\varphi\rangle = \langle f, j_{\epsilon}*R_d^*\varphi\rangle = \langle f, R_d^*(j_{\epsilon}\times\varphi)\rangle = \langle R_df, j_{\epsilon}\times\varphi\rangle$$

Take r > 0 and  $\epsilon > 0$  small enough so that  $r + \epsilon < R$ . Let  $\varphi \in C_c^{\infty}(\mathbf{P}^d)$  such that  $\varphi(A) = 0$ for all *d*-planes which satisfy d(0, A) > r. Then  $(j_{\epsilon} \times \varphi)(A) = 0$  for all *d*-planes for which  $d(0, A) > r + \epsilon$ . Thus  $j_{\epsilon} \times \varphi \in C_c^{\infty}(\mathbf{P}_V^d)$  and by (14) it follows that  $R_d(f * j_{\epsilon}) = 0$  for all *d*-planes intersecting B(0, r). We also have  $(f * j_{\epsilon})|_{B(0,r)} = 0$  and the first part of the proof implies the claim for  $f * j_{\epsilon}$  for small  $\epsilon > 0$ . Since  $f * j_{\epsilon} \to f$  in  $\mathcal{E}'(\mathbb{R}^n)$  when  $\epsilon \to 0$ , we obtain the claim for f.

**Remark 4.2.** When d is even, corollary 1.4 does not say that the result is false. It only indicates that we cannot use the UCP of the normal operator in the proof. This boils down to the fact that  $(-\Delta)^s$  does not admit the UCP for  $s \in \mathbb{Z}$ . However, if d is even, then the function f is determined uniquely in V by its integrals over d-planes which intersect V. Namely, if  $R_d f = 0$ on all d-planes intersecting V, then  $N_d f|_V = 0$ . Since  $N_d \approx (-\Delta)^{-d/2}$ , one can invert  $N_d f$  by the local operator  $(-\Delta)^{d/2}$  to obtain  $f|_V = 0$ . Hence the ROI problem is uniquely solvable in this case without the additional knowledge of f in an open set inside the ROI.

**Remark 4.3.** We also note that unlike in the global data case lower dimensional data does not determine higher dimensional data. In other words,  $R_k f = 0$  for all k-planes intersecting V does not necessarily imply that  $R_d f = 0$  for all d-planes which intersect V where 0 < k < d < n. Thus one cannot reduce the partial data problem for k-planes to the partial data problem for d-planes.

## 5. Higher order fractional Schrödinger equation with singular potential

In this section, we study the fractional Schrödinger equation with higher order fractional Laplacian and singular potential. Let  $\Omega \subset \mathbb{R}^n$  be a bounded open set,  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$  and consider the equation

(15) 
$$\begin{cases} ((-\Delta)^s + q)u = 0 \text{ in } \Omega\\ u|_{\Omega_e} = f \end{cases}$$

where  $u \in H^s(\mathbb{R}^n)$ ,  $f \in H^s(\mathbb{R}^n)$  is the exterior value of u and  $q \in L^{\infty}(\Omega)$  represents the electric potential. When the potential q is more singular one has to interpret the product qu in a suitable way. If  $q \in Z_0^{-s}(\mathbb{R}^n)$ , then q acts as a multiplier and induces a map  $m_q \colon H^s(\mathbb{R}^n) \to H^{-s}(\mathbb{R}^n)$ defined by  $\langle m_q(u), v \rangle_{\mathbb{R}^n} = \langle q, uv \rangle_{\mathbb{R}^n}$ . Then equation (15) becomes

(16) 
$$\begin{cases} (-\Delta)^s u + m_q(u) = 0 \text{ in } \Omega \\ u|_{\Omega_e} = f. \end{cases}$$

We will prove that the generalized DN map  $\Lambda_q$  for equation (16) determines the restriction of the potential  $q \in Z_0^{-s}(\mathbb{R}^n)$  to  $\Omega$  uniquely from exterior measurements. We also obtain the Runge approximation property for equation (16): any function  $g \in \tilde{H}^s(\Omega)$  can be approximated arbitrarily well by solutions of (16).

Similar results were proved in [75] when 0 < s < 1. Our theorems generalize those results for higher order fractional Laplacians. The proofs rely essentially on the same thing: the UCP of the operator  $(-\Delta)^s$  which was proved for  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$  in section 3.1. Also the higher order Poincaré inequality is needed for the well-posedness of the inverse problem. In this section, we provide the basic ideas for the proofs of the lemmas, which are reminescent of the ones in [75] and [28]. We will mainly follow the same notation as in those articles.

The strategy to prove theorems 1.6 and 1.7 is the following. First one constructs a bilinear form and proves that unique weak solutions are obtained in the complement of a countable set of eigenvalues. One also proves that 0 is not an eigenvalue when (4) holds. Then one defines the abstract DN map and proves the Alessandrini identity using it. Using the UCP of  $(-\Delta)^s$  one obtains the Runge approximation property for equation (16). From the Runge approximation and the Alessandrini identity, one can prove the uniqueness result for the inverse problem.

If  $U \subset \mathbb{R}^n$  is open and  $u, v \in L^2(U)$ , we denote the inner product by

$$\langle u, v \rangle_U = \int_U uv \mathrm{d}x.$$

We also use the same notation  $\langle \cdot, \cdot \rangle_U$  for dual pairing.

The following lemma guarantees the existence of unique weak solutions (see [75, Lemma 2.6]).

**Lemma 5.1.** Let  $\Omega \subset \mathbb{R}^n$  be bounded open set,  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$  and  $q \in Z_0^{-s}(\mathbb{R}^n)$ . For  $v, w \in H^s(\mathbb{R}^n)$  define the bilinear form  $B_q$  as

$$B_q(v,w) = \left\langle (-\Delta)^{s/2} v, (-\Delta)^{s/2} w \right\rangle_{\mathbb{R}^n} + \left\langle m_q(v), w \right\rangle_{\mathbb{R}^n}.$$

The following claims hold:

(a) There is a countable set  $\Sigma = \{\lambda_i\}_{i=1}^{\infty} \subset \mathbb{R}, \lambda_1 \leq \lambda_2 \leq \ldots \rightarrow \infty$ , with the following property: if  $\lambda \notin \Sigma$ , then for any  $F \in (\widetilde{H}^s(\Omega))^*$  and  $f \in H^s(\mathbb{R}^n)$  there is unique  $u \in H^s(\mathbb{R}^n)$  satisfying

$$B_q(u,w) - \lambda \langle u, w \rangle_{\mathbb{R}^n} = F(w) \text{ for } w \in \widetilde{H}^s(\Omega), \quad u - f \in \widetilde{H}^s(\Omega)$$

with the norm estimate

$$||u||_{H^{s}(\mathbb{R}^{n})} \leq C\Big( ||F||_{(\widetilde{H}^{s}(\Omega))^{*}} + ||f||_{H^{s}(\mathbb{R}^{n})} \Big)$$

where C is independent of F and f.

(b) The function u in (a) is the unique  $u \in H^{s}(\mathbb{R}^{n})$  satisfying

$$((-\Delta)^s u + m_q(u) - \lambda u)|_{\Omega} = F$$

in the sense of distributions with  $u - f \in \widetilde{H}^{s}(\Omega)$ .

(c) One has  $0 \notin \Sigma$  if (4) holds. If  $q \in L^{\infty}(\Omega)$  and  $q \ge 0$ , then  $\Sigma \subset (0, \infty)$  and (4) always holds.

*Proof.* The constants in the inequalities do not depend on the function v in the proof. It is enough to solve the problem in (a) for  $u - f = v \in \tilde{H}^s(\Omega)$ . Using the fractional Poincaré inequality (theorem 1.5) we obtain

$$\|v\|_{H^{s}(\mathbb{R}^{n})}^{2} \leq C\left(\left\|(-\Delta)^{s/2}v\right\|_{L^{2}(\mathbb{R}^{n})}^{2} + \|v\|_{L^{2}(\mathbb{R}^{n})}^{2}\right) \leq C'\left\|(-\Delta)^{s/2}v\right\|_{L^{2}(\mathbb{R}^{n})}^{2}$$

Let  $0 < \epsilon < 1/C'$  where the constant C' > 0 comes from the previous inequality. Since  $q \in Z_0^{-s}(\mathbb{R}^n)$ , we can find  $q_s \in C_c^{\infty}(\mathbb{R}^n)$  and  $q_r \in Z^{-s}(\mathbb{R}^n)$  such that  $q = q_s + q_r$  and  $||q_r||_{Z^{-s}(\mathbb{R}^n)} < \epsilon$ . When we take  $\mu = ||q_s^-||_{L^{\infty}(\mathbb{R}^n)}$  where  $q_s^- = -\min\{0, q_s(x)\}$ , we obtain the coercivity condition

$$B_{q}(v,v) + \mu \langle v, v \rangle_{\mathbb{R}^{n}} \ge \left\| (-\Delta)^{s/2} v \right\|_{L^{2}(\mathbb{R}^{n})}^{2} + \langle q_{r}, vv \rangle_{\mathbb{R}^{n}} \ge \frac{1}{C'} \left\| v \right\|_{H^{s}(\mathbb{R}^{n})}^{2} - \epsilon \left\| v \right\|_{H^{s}(\mathbb{R}^{n})}^{2}.$$

Hence  $v, w \mapsto B_q(v, w) + \mu \langle v, w \rangle_{\mathbb{R}^n}$  defines an equivalent inner product in  $\widetilde{H}^s(\Omega)$ . The proof is then completed as in [28]: the Riesz representation theorem implies that for every  $\widetilde{F} \in (\widetilde{H}^s(\Omega))^*$ there is unique  $v = G_\mu \widetilde{F} \in \widetilde{H}^s(\Omega)$  such that  $B_q(v, w) + \mu \langle v, w \rangle_{\mathbb{R}^n} = \widetilde{F}(w)$  for all  $w \in \widetilde{H}^s(\Omega)$ . The map  $G_\mu : (\widetilde{H}^s(\Omega))^* \to \widetilde{H}^s(\Omega)$  induces a compact, self-adjoint and positive definite operator  $\widetilde{G}_\mu : L^2(\Omega) \to L^2(\Omega)$  by the compact Sobolev embedding theorem. The spectral theorem for the self-adjoint compact operator  $\widetilde{G}_\mu$  implies the claim in (a). Part (b) holds since  $C_c^{\infty}(\Omega)$  is dense in  $\widetilde{H}^s(\Omega)$ . The first claim in (c) follows from the Fredholm alternative. The second claim in (c) is essentially the same as in [28, Lemma 2.3] and is proved by replacing  $\widetilde{H}^s(\Omega)$  with  $H_{\overline{\Omega}}(\mathbb{R}^n)$  in the proof of (a).

Recall the definition of the abstract trace space  $X = H^s(\mathbb{R}^n)/\widetilde{H}^s(\Omega)$  which we equip with the quotient norm

$$\|[f]\|_X = \inf_{\varphi \in \widetilde{H}^s(\Omega)} \|f - \varphi\|_{H^s(\mathbb{R}^n)}, \quad f \in H^s(\mathbb{R}^n).$$

The following lemma implies that the DN map is well-defined and continuous. The result follows immediately from the definition of the bilinear form  $B_q(\cdot, \cdot)$  and from the continuity of  $(-\Delta)^{s/2} \colon H^s(\mathbb{R}^n) \to L^2(\mathbb{R}^n)$  (see [28, Lemma 2.4]).

**Lemma 5.2.** Let  $\Omega \subset \mathbb{R}^n$  be bounded open set,  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$  and  $q \in Z_0^{-s}(\mathbb{R}^n)$  which satisfies (4). Then the map  $\Lambda_q \colon X \to X^*$ ,  $\langle \Lambda_q[f], [g] \rangle = B_q(u_f, g)$ , is linear and continuous, where  $u_f \in H^s(\mathbb{R}^n)$  solves  $(-\Delta)^s u + m_q(u) = 0$  in  $\Omega$  with  $u - f \in \widetilde{H}^s(\Omega)$ . One also has the self-adjointness property  $\langle \Lambda_q[f], [g] \rangle = \langle [f], \Lambda_q[g] \rangle$  for  $f, g \in H^s(\mathbb{R}^n)$ .

Proof. Since  $u_f$  is a solution to  $(-\Delta)^s u + m_q(u) = 0$  in  $\Omega$  with  $u_f - f \in \widetilde{H}^s(\Omega)$  and solutions are unique, we obtain  $B_q(u_{f+\varphi}, g+\psi) = B_q(u_f, g)$  for  $\varphi, \psi \in \widetilde{H}^s(\Omega)$ . This implies that  $\Lambda_q$  is well-defined. Further, using continuity of  $(-\Delta)^{s/2}$  and the norm estimate for solution  $u_f$  from lemma 5.1, we obtain

$$\begin{aligned} |\langle \Lambda_q[f], [g] \rangle| &\leq \left\| (-\Delta)^{s/2} u_f \right\|_{L^2(\mathbb{R}^n)} \left\| (-\Delta)^{s/2} g \right\|_{L^2(\mathbb{R}^n)} + \|q\|_{Z^{-s}(\mathbb{R}^n)} \|u_f\|_{H^s(\mathbb{R}^n)} \|g\|_{H^s(\mathbb{R}^n)} \\ &\leq C \|f\|_{H^s(\mathbb{R}^n)} \|g\|_{H^s(\mathbb{R}^n)} \,, \end{aligned}$$

where C does not depend on f and g. By the definition of the quotient norm  $|\langle \Lambda_q[f], [g] \rangle| \leq C ||[f]||_X ||[g]||_X$ , so  $\Lambda_q$  is continuous. Choosing  $g = u_g$  we obtain by symmetry of  $B_q(\cdot, \cdot)$ 

$$\langle \Lambda_q[f], [g] \rangle = B_q(u_f, u_g) = \langle \Lambda_q[g], [u_f] \rangle = \langle [f], \Lambda_q[g] \rangle_{20}$$

where we used the fact that  $u_f - f \in \widetilde{H}^s(\Omega)$ .

We immediately obtain the Alessandrini identity from lemma 5.2 (see [75, Lemma 2.7]). We use some abuse of notation and write f instead of [f].

**Lemma 5.3** (Alessandrini identity). Let  $\Omega \subset \mathbb{R}^n$  be bounded open set,  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$  and  $q_1, q_2 \in Z_0^{-s}(\mathbb{R}^n)$  which satisfy (4). For any  $f_1, f_2 \in X$  one has

$$\langle (\Lambda_{q_1} - \Lambda_{q_2}) f_1, f_2 \rangle = \langle m_{q_1 - q_2}(u_1), u_2 \rangle_{\mathbb{R}^n}$$

where  $u_i \in H^s(\mathbb{R}^n)$  solves  $(-\Delta)^s u_i + m_{q_i}(u_i) = 0$  in  $\Omega$  with  $u_i - f_i \in \widetilde{H}^s(\Omega)$ .

*Proof.* Using the self-adjointness of  $\Lambda_q$  and the property  $B_q(u_i, g+\psi) = B_q(u_i, g)$  for  $\psi \in H^s(\Omega)$ , we obtain

$$\begin{split} \langle (\Lambda_{q_1} - \Lambda_{q_2})f_1, f_2 \rangle &= \langle \Lambda_{q_1}f_1, f_2 \rangle - \langle f_1, \Lambda_{q_2}f_2 \rangle = B_{q_1}(u_1, f_2) - B_{q_2}(u_2, f_1) \\ &= B_{q_1}(u_1, f_2 + (u_2 - f_2)) - B_{q_2}(u_2, f_1 + (u_1 - f_1)) \\ &= B_{q_1}(u_1, u_2) - B_{q_2}(u_1, u_2) = \langle m_{q_1 - q_2}(u_1), u_2 \rangle_{\mathbb{R}^n} \end{split}$$

which gives the claim.

From the UCP of  $(-\Delta)^s$  (theorem 1.2), we obtain the following approximation result (see [75, Lemma 8.1]).

**Lemma 5.4.** Let  $\Omega \subset \mathbb{R}^n$  be bounded open set,  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$  and  $q \in Z_0^{-s}(\mathbb{R}^n)$  which satisfies (4). Denote by  $P_q: X \to H^s(\mathbb{R}^n)$ ,  $P_q f = u_f$ , where  $u_f \in H^s(\mathbb{R}^n)$  is the unique solution to  $(-\Delta)^s u + m_q(u) = 0$  in  $\Omega$  with  $u - f \in \widetilde{H}^s(\Omega)$  given by lemma 5.1. Let  $W \subset \Omega_e$  be any open set and define the set

$$\mathcal{R} = \{ P_q f - f : f \in C_c^{\infty}(W) \}.$$

Then  $\mathcal{R}$  is dense in  $\widetilde{H}^{s}(\Omega)$ .

*Proof.* By the Hahn-Banach theorem it is enough to show that if  $F \in (\widetilde{H}^s(\Omega))^*$  and  $\langle F, v \rangle = 0$  for all  $v \in \mathcal{R}$ , then F = 0. Let  $F \in (\widetilde{H}^s(\Omega))^*$  and assume that

$$\langle F, P_q f - f \rangle = 0, \quad f \in C^{\infty}_c(W).$$

Let  $\varphi \in \widetilde{H}^s(\Omega)$  be the solution to

$$(-\Delta)^{s}\varphi + m_{q}(\varphi) = F \text{ in } \Omega, \quad \varphi|_{\Omega_{e}} = 0$$

which exists by lemma 5.1. This means that  $B_q(\varphi, w) = \langle F, w \rangle$  for all  $w \in \widetilde{H}^s(\Omega)$ . Let  $u_f = P_q f \in H^s(\mathbb{R}^n)$  where  $u_f - f \in \widetilde{H}^s(\Omega)$ . Now

$$\langle F, P_q f - f \rangle = B_q(\varphi, u_f - f) = -B_q(\varphi, f)$$

since  $u_f$  is a solution to  $(-\Delta)^s u + m_q(u) = 0$  in  $\Omega$  and  $\varphi \in \widetilde{H}^s(\Omega)$ . Thus  $B_q(\varphi, f) = 0$  for all  $f \in C_c^{\infty}(W)$ . Using the fact that  $\operatorname{spt}(\varphi)$  and  $\operatorname{spt}(f)$  are disjoint, we obtain

$$0 = \left\langle (-\Delta)^{s/2} \varphi, (-\Delta)^{s/2} f \right\rangle_{\mathbb{R}^n} = \left\langle (-\Delta)^s \varphi, f \right\rangle_{\mathbb{R}^n}.$$

Here we used that  $\langle (-\Delta)^{s/2}u, (-\Delta)^{s/2}v \rangle_{\mathbb{R}^n} = \langle (-\Delta)^s u, v \rangle_{\mathbb{R}^n}$  for  $u, v \in \mathscr{S}(\mathbb{R}^n)$  and the equality holds also in  $H^s(\mathbb{R}^n)$  by density. Hence  $\varphi|_W = (-\Delta)^s \varphi|_W = 0$  and theorem 1.2 implies  $\varphi = 0$  and eventually F = 0.

We remark that exactly the same proof gives the density of  $r_{\Omega}\mathcal{R}$  in  $L^2(\Omega)$  where  $r_{\Omega}$  is the restriction to  $\Omega$  (see [28, Lemma 5.1]). Now it is easy to prove theorems 1.6 and 1.7.

Proof of theorem 1.6. Since we can always shrink the sets  $W_i$ , we can assume without loss of generality that  $\overline{W}_1 \cap \overline{W}_2 = \emptyset$  and  $(\overline{W}_1 \cup \overline{W}_2) \cap \overline{\Omega} = \emptyset$ . Using the Alessandrini identity (lemma 5.3), we obtain

$$\langle m_{q_1-q_2}(u_1), u_2 \rangle_{\mathbb{R}^n} = 0$$

for any  $u_i \in H^s(\mathbb{R}^n)$  which solves  $(-\Delta)^s u_i + m_{q_i}(u_i) = 0$  in  $\Omega$  with exterior values in  $C_c^{\infty}(W_i)$ . Let  $v_1, v_2 \in \widetilde{H}^s(\Omega)$ . By lemma 5.4 there are sequences of exterior values  $f_1^k \in C_c^{\infty}(W_1)$  and  $f_2^k \in C_c^{\infty}(W_2)$  with sequences of solutions  $u_1^k, u_2^k \in H^s(\mathbb{R}^n)$  such that

- $(-\Delta)^s u_i^k + m_{q_i}(u_i^k) = 0$  in  $\Omega$
- $u_i^k f_i^k \in \widetilde{H}^s(\Omega)$
- $u_i^k = f_i^k + v_i + r_i^k$  where  $r_i^k \xrightarrow{k \to \infty} 0$  in  $\widetilde{H}^s(\Omega)$ .

When we insert the solutions  $u_i^k$  into the Alessandrini identity, use the support conditions and take the limit  $k \to \infty$ , we obtain

$$\langle m_{q_1-q_2}(v_1), v_2 \rangle_{\mathbb{R}^n} = 0.$$

Let  $\varphi \in C_c^{\infty}(\Omega)$ . Choose  $v_1 = \varphi$  and  $v_2 \in C_c^{\infty}(\Omega)$  such that  $v_2 = 1$  in a neighborhood of  $\operatorname{spt}(\varphi)$ . This implies

$$0 = \langle m_{q_1 - q_2}(v_1), v_2 \rangle_{\mathbb{R}^n} = \langle q_1 - q_2, v_1 v_2 \rangle_{\mathbb{R}^n} = \langle q_1 - q_2, \varphi \rangle_{\mathbb{R}^n}$$

which is equivalent to that  $q_1|_{\Omega} = q_2|_{\Omega}$  as distributions.

Proof of theorem 1.7. Since  $\operatorname{int}(\Omega_1 \setminus \Omega) \neq \emptyset$ , there is open set  $W \subset \Omega_e$  such that  $\overline{W} \subset \Omega_1 \setminus \overline{\Omega}$ . By lemma 5.4, the set  $\mathcal{R}$  is dense in  $\widetilde{H}^s(\Omega)$ . Hence, we can approximate any  $g \in \widetilde{H}^s(\Omega)$  arbitrarily well by solutions  $u \in H^s(\mathbb{R}^n)$  to the equation  $(-\Delta)^s u + m_q(u) = 0$  in  $\Omega$  with  $u - f \in \widetilde{H}^s(\Omega)$ . Since  $f \in C_c^{\infty}(W)$  we especially have  $\operatorname{spt}(u) \subset \overline{\Omega}_1$ .

# 6. Higher order fractional magnetic Schrödinger equation

In this section we will be dealing with the definition of the FMSE, as well as with the proof of the injectivity result for the corresponding inverse problem. For the sake of simplicity, let us fix the convention throughout this section that the symbol  $\langle \cdot, \cdot \rangle$  indicates both the scalar product (duality pairing) on  $L^2(\mathbb{R}^n)$  and the one on  $L^2(\mathbb{R}^{2n})$ , the distinction between the two being always possible by checking the number of free variables inside the brackets. We also let the norms  $\|\cdot\|_{L^2}$ ,  $\|\cdot\|_{H^s}$  etc. to denote the norms over the whole  $\mathbb{R}^n$  or  $\mathbb{R}^{2n}$  when the base set is not specified.

6.1. High order bivariate functions. Let  $l, n \in \mathbb{N}$ , and consider a family A of scalar twopoint functions indexed over the set  $\{1, ..., n\}^l$ . A generic member of the family is determined by a vector  $(i_1, ..., i_l)$  such that  $i_j \in \{1, ..., n\}$  for all  $j \in \{1, ..., l\}$ , and it is a function  $A_{i_1,...,i_l}$ :  $\mathbb{R}^{2n} \to \mathbb{R}$ . We call such family A a bivariate function of order l. One can see the family A as a function  $A : \mathbb{R}^{2n} \to \mathbb{M}^l$ , where  $\mathbb{M}^l$  is the set of all  $n \times ... \times n = n^l$  arrays of real numbers, i.e. tensors of order l.

Let  $a, b \in \mathbb{N}$ , and let A, B be bivariate functions of orders a and b respectively, in the variables  $x_1, x_2$ . The *tensor product* of A and B is the bivariate function of order a + b given by

$$(A \otimes B)_{i_1,\dots,i_a,j_1,\dots,j_b}(x_1,x_2) := A_{i_1,\dots,i_a}(x_1,x_2)B_{j_1,\dots,j_b}(x_1,x_2) \ .$$

In particular, for a vector  $\xi \in \mathbb{R}^n$  we let  $\xi^{\otimes 0} = 0$ ,  $\xi^{\otimes 1} = \xi$  and recursively  $\xi^{\otimes m} = \xi^{\otimes (m-1)} \otimes \xi$ . Let A, B as before, but assume now that  $a \geq b$ . The *contraction* of A and B is the bivariate function of order a - b given by

$$(A \cdot B)_{i_1,\dots,i_{a-b}}(x_1,x_2) := \sum_{j_1,\dots,j_b=1}^n A_{i_1,\dots,i_{a-b},j_1,\dots,j_b}(x_1,x_2) B_{j_1,\dots,j_b}(x_1,x_2) \ .$$

If A = B, then of course a = b, so that  $A \cdot A$  is a scalar function of the variables  $(x_1, x_2)$ . One sees that  $|A| := (A \cdot A)^{1/2}$  defines a norm for fixed  $x_1$  and  $x_2$ , and that this coincides with the usual one when A is a vector function.

**Lemma 6.1.** Let  $a, b \in \mathbb{N}$ , and let A, B, v be bivariate functions of orders a, b and 1 respectively, in the variables  $x_1, x_2$ . Assume that  $a \ge b + 1$ ; then

$$A \cdot (B \otimes v) = (A \cdot v) \cdot B$$
.

*Proof.* The proof is just a simple computation:

$$\begin{split} [A \cdot (B \otimes v)]_{i_1, \dots, i_{a-b-1}} &= \sum_{j_1, \dots, j_{b+1}=1}^n A_{i_1, \dots, i_{a-b-1}, j_1, \dots, j_{b+1}} (B \otimes v)_{j_1, \dots, j_{b+1}} \\ &= \sum_{j_1, \dots, j_{b+1}=1}^n A_{i_1, \dots, i_{a-b-1}, j_1, \dots, j_{b+1}} B_{j_1, \dots, j_b} v_{j_{b+1}} \\ &= \sum_{j_1, \dots, j_b=1}^n B_{j_1, \dots, j_b} \sum_{j_{b+1}=1}^n A_{i_1, \dots, i_{a-b-1}, j_1, \dots, j_{b+1}} v_{j_{b+1}} \\ &= \sum_{j_1, \dots, j_b=1}^n B_{j_1, \dots, j_b} (A \cdot v)_{i_1, \dots, i_{a-b-1}, j_1, \dots, j_b} \\ &= [(A \cdot v) \cdot B]_{i_1, \dots, i_{a-b-1}} . \end{split}$$

Let A be a bivariate function of any order. Following [14], we recall the definitions of the symmetric and antisymmetric parts of A with respect to the variables x and y and the  $L^2$  norms of A with respect to the first and second variable at point x:

$$A_s(x,y) := \frac{A(x,y) + A(y,x)}{2}, \qquad A_a(x,y) := A(x,y) - A_s(x,y),$$
$$\mathcal{J}_1 A(x) := \left( \int_{\mathbb{R}^n} |A(y,x)|^2 \, dy \right)^{1/2}, \qquad \mathcal{J}_2 A(x) := \left( \int_{\mathbb{R}^n} |A(x,y)|^2 \, dy \right)^{1/2}$$

It is easily seen that  $A \in L^2$  implies  $A_s, A_a \in L^2$ , since

(17) 
$$||A_s||_{L^2} = \left\|\frac{A(x,y) + A(y,x)}{2}\right\|_{L^2} \le ||A||_{L^2}, \quad ||A_a||_{L^2} = \left\|\frac{A(x,y) - A(y,x)}{2}\right\|_{L^2} \le ||A||_{L^2}.$$

A bivariate function A of any order will be called *symmetric* if  $A = A_s$  almost everywhere, and *antisymmetric* if  $A = A_a$  almost everywhere.

**Lemma 6.2.** Let  $A \in L^1(\mathbb{R}^{2n}, \mathbb{M}^l)$  be an antisymmetric bivariate function of order l for some  $l \in \mathbb{N}$ . Then  $\int_{\mathbb{R}^{2n}} A(x, y) \, dy \, dx = 0$ .

Proof. Let  $D^+$ ,  $D^-$  be the sets respectively above and under the diagonal  $D := \{(x, y) \in \mathbb{R}^{2n} : x = y\}$  of  $\mathbb{R}^{2n}$ . Since  $\int_{D^{\pm}} A(x, y) \, dy dx \leq \int_{D^{\pm}} |A(x, y)| \, dy dx \leq ||A||_{L^1} < \infty$ , we can decompose the integral as  $\int_{\mathbb{R}^{2n}} A(x, y) \, dy dx = \int_{D^+} A(x, y) \, dy dx + \int_{D^-} A(x, y) \, dy dx$ . Given the symmetry of the sets  $D^+$  and  $D^-$ , this can be rewritten as  $\int_{\mathbb{R}^{2n}} A(x, y) \, dy dx = \int_{D^+} (A(x, y) \, dy dx = \int_{D^+} (A(x, y) + A(y, x)) \, dy dx$ , which vanishes by virtue of the antisymmetry of A.

6.2. Fractional operators. Let  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$ ,  $u \in C_c^{\infty}(\mathbb{R}^n)$  and  $x, y \in \mathbb{R}^n$ . Let  $\lfloor s \rfloor := \sup\{n \in \mathbb{N} : n < s\}$  and  $s' := s - \lfloor s \rfloor$ , so that by definition  $s' \in (0, 1)$ . The fractional gradient of order s of u at points x and y is the following symmetric bivariate function of order  $\lfloor s \rfloor + 1$ :

4 10

$$\nabla^{s} u(x,y) := \frac{\mathcal{C}_{n,s'}^{1/2}}{\sqrt{2}} \frac{\nabla^{\lfloor s \rfloor} u(x) - \nabla^{\lfloor s \rfloor} u(y)}{|y - x|^{n/2 + s' + 1}} \otimes (y - x) .$$

Observe that this definition coincides with the usual one for  $s \in (0, 1)$ , since in this case  $\lfloor s \rfloor = 0$ and s' = s. One can compute

$$\begin{split} \|\nabla^{s} u\|_{L^{2}(\mathbb{R}^{2n},\mathbb{M}^{\lfloor s \rfloor+1})}^{2} &= \frac{\mathcal{C}_{n,s'}}{2} \int_{\mathbb{R}^{n}} \int_{\mathbb{R}^{n}} \frac{|\nabla^{\lfloor s \rfloor} u(x) - \nabla^{\lfloor s \rfloor} u(y)|^{2}}{|x - y|^{n + 2s'}} dx \, dy \\ &= \frac{\mathcal{C}_{n,s'}}{2} [\nabla^{\lfloor s \rfloor} u]_{\dot{H}^{s'}(\mathbb{R}^{n})}^{2} = \left\| (-\Delta)^{s'/2} \nabla^{\lfloor s \rfloor} u \right\|_{L^{2}(\mathbb{R}^{n})}^{2} \\ &= \left\| |\xi|^{s'} \xi^{\otimes \lfloor s \rfloor} \hat{u}(\xi) \right\|_{L^{2}(\mathbb{R}^{n})}^{2} = \left\| |\xi|^{s} \hat{u}(\xi) \right\|_{L^{2}(\mathbb{R}^{n})}^{2} \\ &= \left\| (-\Delta)^{s/2} u \right\|_{L^{2}(\mathbb{R}^{n})}^{2} \leq \|u\|_{H^{s}(\mathbb{R}^{n})}^{2} \, . \end{split}$$

Thus, by the density of  $C_c^{\infty}$  in  $H^s$ ,  $\nabla^s$  can be extended to a continuous operator  $\nabla^s : H^s(\mathbb{R}^n) \to L^2(\mathbb{R}^{2n}, \mathbb{M}^{\lfloor s \rfloor + 1})$ . One sees by density that the formula given for  $\nabla^s u$  in the case  $u \in C_c^{\infty}(\mathbb{R}^n)$  still holds almost everywhere for  $u \in H^s(\mathbb{R}^n)$ . Thus if  $u, v \in H^s$ , by the above computation,

$$\langle \nabla^s u, \nabla^s u \rangle = \| (-\Delta)^{s/2} u \|_{L^2}^2 = \langle (-\Delta)^{s/2} u, (-\Delta)^{s/2} u \rangle = \langle (-\Delta)^s u, u \rangle ,$$

so that by the polarization identity and the self-adjointness of  $(-\Delta)^s$ ,

$$\begin{split} \langle \nabla^s u, \nabla^s v \rangle &= \frac{\langle \nabla^s (u+v), \nabla^s (u+v) \rangle - \langle \nabla^s u, \nabla^s u \rangle - \langle \nabla^s v, \nabla^s v \rangle}{2} \\ &= \frac{\langle (-\Delta)^s (u+v), u+v \rangle - \langle (-\Delta)^s u, u \rangle - \langle (-\Delta)^s v, v \rangle}{2} \\ &= \frac{\langle (-\Delta)^s u, v \rangle + \langle (-\Delta)^s v, u \rangle}{2} = \langle (-\Delta)^s u, v \rangle \,. \end{split}$$

This proves that if the fractional divergence  $(\nabla \cdot)^s : L^2(\mathbb{R}^{2n}, \mathbb{M}^{\lfloor s \rfloor + 1}) \to H^{-s}(\mathbb{R}^n)$  is defined as the adjoint of  $\nabla^s$ , then weakly  $(\nabla \cdot)^s \nabla^s = (-\Delta)^s$  for  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$ . This result was already proved in [15], but only for the case  $s \in (0, 1)$ . If we define the antisymmetric bivariate vector function

$$\alpha(x,y) := \frac{\mathcal{C}_{n,s'}^{1/2}}{\sqrt{2}} \frac{y-x}{|y-x|^{n/2+s'+1}}$$

then for  $u \in H^s$  the identity

$$\nabla^{s} u(x,y) = (\nabla^{\lfloor s \rfloor} u(x) - \nabla^{\lfloor s \rfloor} u(y)) \otimes \alpha(x,y)$$

holds almost everywhere.

We now define the magnetic versions of the above operators. Fix  $p > \max\{1, n/2s\}$ , and let A be a bivariate function of order  $\lfloor s \rfloor + 1$  such that

- (a1)  $\mathcal{J}_2 A \in L^{2p}(\mathbb{R}^n)$
- (a2)  $\operatorname{spt}(A) \subset \Omega \times \Omega$ .

With such choice of p, the embedding  $H^s \times L^{2p} \hookrightarrow L^2$  always holds by [5, Theorem 6.1], recall that  $W^r(\mathbb{R}^n) = H^r(\mathbb{R}^n)$  with equivalent norms when  $r \in \mathbb{R}$  and  $W^r(\mathbb{R}^n)$  is the  $L^2$  Sobolev-Slobodecki space [5, 57]. Therefore, if  $u \in H^s$ ,

$$\begin{split} \|A(x,y)u(x)\|_{L^{2}(\mathbb{R}^{2n},\mathbb{M}^{\lfloor s\rfloor+1})} &= \left(\int_{\mathbb{R}^{n}} |u(x)|^{2} \int_{\mathbb{R}^{n}} |A(x,y)|^{2} dy \, dx\right)^{1/2} \\ &= \left(\int_{\mathbb{R}^{n}} |u(x)|^{2} \, |\mathcal{J}_{2}A(x)|^{2} \, dx\right)^{1/2} = \|u \, \mathcal{J}_{2}A\|_{L^{2}(\mathbb{R}^{n})} \\ &\leq c\|u\|_{H^{s}}\|\mathcal{J}_{2}A\|_{L^{2p}} < \infty, \end{split}$$

where c does not depend on u and A. This allows the definition of  $\nabla_A^s u(x, y) := \nabla^s u(x, y) + A(x, y)u(x)$  and its adjoint  $(\nabla \cdot)_A^s$  just as in [14], in such a way that  $\nabla_A^s : H^s(\mathbb{R}^n) \to L^2(\mathbb{R}^{2n}, \mathbb{M}^{\lfloor s \rfloor + 1})$  and  $(\nabla \cdot)_A^s : L^2(\mathbb{R}^{2n}, \mathbb{M}^{\lfloor s \rfloor + 1}) \to H^{-s}(\mathbb{R}^n)$ . By definition, the magnetic fractional Laplacian  $(-\Delta)_A^s : H^s \to H^{-s}$  will be the composition  $(\nabla \cdot)_A^s \nabla_A^s$ . Let now q be a scalar field such that

(a3)  $q \in L^p(\Omega)$ .

By [5, Theorem 8.3] we have the embedding  $H^s \times L^p \hookrightarrow H^{-s}$  and hence  $qu \in H^{-s}$  holds for  $u \in H^s$ . We can thus define the magnetic Schrödinger operator  $(-\Delta)_A^s + q : H^s \to H^{-s}$  and the fractional magnetic Schrödinger equation (FMSE)

$$(-\Delta)^s_A u + qu = 0 \; .$$

In the next Lemma we write  $(-\Delta)_A^s$  in a more convenient form. To this scope, we introduce the bivariate function of order  $\lfloor s \rfloor$  given by  $S(x,y) := A(x,y) \cdot \alpha(x,y)$ , for which we assume that

(a4)  $|S(x,y)| \leq \tilde{S}(y)$  for a.e.  $x, y \in \mathbb{R}^n$ , with  $\tilde{S} \in L^2$ , (a5)  $S(x,y) \in H^{\lfloor s \rfloor}(\mathbb{R}^{2n}, \mathbb{M}^{\lfloor s \rfloor})$ .

**Remark 6.3.** Assumption (a4) is really relevant only when  $\lfloor s \rfloor \neq 0$ , as it will be clear from the proof; in the case  $s \in (0,1)$ , this assumption can be reduced. We refer to [14] for a set of sufficient conditions in that regime. Moreover, with a more careful analysis, one could reduce the exponent of the space to which  $\tilde{S}$  belongs. However, we decided to keep  $L^2$  for the sake of simplicity.

**Lemma 6.4.** Let A be a bivariate function of order  $\lfloor s \rfloor + 1$  satisfying conditions (a1), (a2) (a4), (a5), and let  $u \in H^s$ . There exist linear operators  $\mathfrak{N}, \mathfrak{M}_\beta$  acting on bivariate functions of order  $\lfloor s \rfloor$ , with  $\beta$  a multi-index of length  $|\beta| \leq \lfloor s \rfloor$ , such that the equation

$$\begin{split} (-\Delta)^s_A u(x) = & (-\Delta)^s u(x) + \sum_{|\beta| \le \lfloor s \rfloor} \partial^\beta u(x) (\mathfrak{M}_\beta(S))(x) + \\ & + \int_{\mathbb{R}^n} u(y) (\mathfrak{N}(S))(x,y) \, dy + u(x) \int_{\mathbb{R}^n} |A(x,y)|^2 dy \end{split}$$

holds in weak sense.

*Proof.* If  $v \in H^s$ , then in weak sense

(18) 
$$\langle (-\Delta)^s_A u, v \rangle = \langle \nabla^s u, \nabla^s v \rangle + \langle \nabla^s u, Av \rangle + \langle \nabla^s v, Au \rangle + \langle Au, Av \rangle ,$$

where all the terms on the right hand side are finite, as observed above.

**Step 1.** Let us start by computing the third term on the right hand side of (18). The bivariate function  $\nabla^s v(x,y)[A(x,y)u(x)]_a$  is antisymmetric, and by Cauchy-Schwartz and formula (17) we have  $\|\nabla^s v(Au)_a\|_{L^1} \leq \|\nabla^s v\|_{L^2} \|(Au)_a\|_{L^2} \leq \|v\|_{H^s} \|Au\|_{L^2} < \infty$ . Therefore Lemma 6.2 gives  $\langle \nabla^s v, (Au)_a \rangle = 0$ , and we can use Lemma 6.1 to write

(19)  

$$\langle \nabla^{s} v, Au \rangle = \langle \nabla^{s} v, Au \rangle - \langle \nabla^{s} v, (Au)_{a} \rangle = \langle \nabla^{s} v, (Au)_{s} \rangle$$

$$= \langle (\nabla^{\lfloor s \rfloor} v(x) - \nabla^{\lfloor s \rfloor} v(y)) \otimes \alpha, (Au)_{s} \rangle$$

$$= \langle \nabla^{\lfloor s \rfloor} v(x) - \nabla^{\lfloor s \rfloor} v(y), (Au)_{s} \cdot \alpha \rangle$$

$$= \langle \nabla^{\lfloor s \rfloor} v(x) - \nabla^{\lfloor s \rfloor} v(y), (A \cdot \alpha u)_{a} \rangle$$

$$= \langle \nabla^{\lfloor s \rfloor} v(x) - \nabla^{\lfloor s \rfloor} v(y), (Su)_{a} \rangle .$$

The bivariate function  $[\nabla^{\lfloor s \rfloor} v(x) + \nabla^{\lfloor s \rfloor} v(y)][S(x, y)u(x)]_a$  is antisymmetric, and we can estimate its  $L^1$  norm by means of the triangle inequality as

$$\|(\nabla^{\lfloor s \rfloor}v(x) + \nabla^{\lfloor s \rfloor}v(y))(Su)_a\|_{L^1} \le \|(\nabla^{\lfloor s \rfloor}v(x) - \nabla^{\lfloor s \rfloor}v(y))(Su)_a\|_{L^1} + \|2\nabla^{\lfloor s \rfloor}v(x)(Su)_a\|_{L^1} + \|2\nabla^{\lfloor s \rfloor}v(x)\|_{L^1} +$$

The first term on the right hand side equals  $\|\nabla^s v(Au)_s\|_{L^1}$  by computation (19), so that it is finite by  $\|\nabla^s v(Au)_s\|_{L^1} \leq \|\nabla^s v\|_{L^2}\|(Au)_s\|_{L^2} \leq \|v\|_{H^s}\|Au\|_{L^2} < \infty$ . We estimate the other term again by triangular inequality as

(20) 
$$\|2\nabla^{\lfloor s \rfloor}v(x)(Su)_a\|_{L^1} \le \|\nabla^{\lfloor s \rfloor}v(x)S(x,y)u(x)\|_{L^1} + \|\nabla^{\lfloor s \rfloor}v(x)S(y,x)u(y)\|_{L^1} .$$

The estimation of the second term on the right hand side of (20) can be done as follows, and similarly for the other one:

(21)  

$$\begin{split} \|\nabla^{\lfloor s \rfloor} v(x) S(y, x) u(y)\|_{L^{1}} &= \int_{\mathbb{R}^{n}} |\nabla^{\lfloor s \rfloor} v(x)| \int_{\mathbb{R}^{n}} |S(y, x)| |u(y)| \, dy dx \\ &\leq \int_{\mathbb{R}^{n}} |\nabla^{\lfloor s \rfloor} v(x)| \tilde{S}(x) \int_{\Omega} |u(y)| \, dy dx \\ &\leq c \|u\|_{L^{2}} \int_{\mathbb{R}^{n}} |\nabla^{\lfloor s \rfloor} v(x)| \tilde{S}(x) dx \\ &\leq c \|u\|_{L^{2}} \|\nabla^{\lfloor s \rfloor} v(x)\|_{L^{2}} \|\tilde{S}\|_{L^{2}} \\ &\leq c \|u\|_{H^{s}} \|v\|_{H^{s}} \|\tilde{S}\|_{L^{2}} < \infty, \end{split}$$

where the constant c can change from line to line and does not depend on v, u and S. Thus we have proved that  $\|2\nabla^{\lfloor s\rfloor}v(x)(Su)_a\|_{L^1} < \infty$ , which in turn implies that  $\|(\nabla^{\lfloor s\rfloor}v(x) +$  $\nabla^{\lfloor s \rfloor} v(y))(Su)_a \|_{L^1} < \infty$ . Now we can use again Lemma 6.2 to conclude that  $\langle \nabla^{\lfloor s \rfloor} v(x) + v(x) \rangle$  $\nabla^{\lfloor s \rfloor} v(y), (Su)_a \rangle = 0$ . From this fact and formula (19), integrating by parts,

$$\begin{split} \langle \nabla^{s} v, Au \rangle &= \langle \nabla^{\lfloor s \rfloor} v(x) - \nabla^{\lfloor s \rfloor} v(y), (Su)_{a} \rangle + \langle \nabla^{\lfloor s \rfloor} v(x) + \nabla^{\lfloor s \rfloor} v(y), (Su)_{a} \rangle \\ &= 2 \langle \nabla^{\lfloor s \rfloor} v(x), (Su)_{a} \rangle = \langle \nabla^{\lfloor s \rfloor} v(x), S(x, y) u(x) - S(y, x) u(y) \rangle \\ &= \langle \nabla^{\lfloor s \rfloor} v(x), S(x, y) u(x) \rangle - \langle \nabla^{\lfloor s \rfloor} v(x), S(y, x) u(y) \rangle \\ &= (-1)^{\lfloor s \rfloor} \left\langle v, (\nabla \cdot)_{x}^{\lfloor s \rfloor} \left( u(x) \int_{\mathbb{R}^{n}} S(x, y) dy \right) \right\rangle \\ &- (-1)^{\lfloor s \rfloor} \left\langle v, (\nabla \cdot)_{x}^{\lfloor s \rfloor} \int_{\mathbb{R}^{n}} S(y, x) u(y) dy \right\rangle \,. \end{split}$$

In the last term the derivatives can pass under the integral sign by means of the dominated convergence theorem, since  $|S(x,y)u(y)| \leq S(y)|u(y)|$ , and  $\int_{\mathbb{R}^n} S(y)|u(y)|dy \leq \|S\|_{L^2} \|u\|_{L^2} < \infty$  $\infty$ . Eventually,

(22) 
$$\langle \nabla^{s} v, Au \rangle = (-1)^{\lfloor s \rfloor} \left\langle v, (\nabla \cdot)_{x}^{\lfloor s \rfloor} \left( u(x) \int_{\mathbb{R}^{n}} S(x, y) dy \right) \right\rangle$$
$$+ (-1)^{\lfloor s \rfloor + 1} \left\langle v, \int_{\mathbb{R}^{n}} u(y) (\nabla \cdot)_{x}^{\lfloor s \rfloor} S(y, x) dy \right\rangle .$$

**Step 2.** Next we compute the second term on the right hand side of (18). With a computation similar to (19), we obtain  $\langle \nabla^s u, Av \rangle = \langle \nabla^{\lfloor s \rfloor} u(x) - \nabla^{\lfloor s \rfloor} u(y), S(x, y)v(x) \rangle$ ; moreover, we have estimates similar to the ones in (21), and so we can split the integral. Eventually, we integrate by parts and get

$$\begin{split} \langle \nabla^{s} u, Av \rangle &= \langle \nabla^{\lfloor s \rfloor} u(x), S(x, y)v(x) \rangle - \langle \nabla^{\lfloor s \rfloor} u(y), S(x, y)v(x) \rangle \\ (23) &= \left\langle v(x), \nabla^{\lfloor s \rfloor} u(x) \cdot \int_{\mathbb{R}^{n}} S(x, y) dy \right\rangle - \left\langle v(x), \int_{\mathbb{R}^{n}} \nabla^{\lfloor s \rfloor} u(y) \cdot S(x, y) dy \right\rangle \\ &= \left\langle v(x), \nabla^{\lfloor s \rfloor} u(x) \cdot \int_{\mathbb{R}^{n}} S(x, y) dy \right\rangle + (-1)^{\lfloor s \rfloor + 1} \left\langle v(x), \int_{\mathbb{R}^{n}} u(y) (\nabla \cdot)^{\lfloor s \rfloor}_{y} S(x, y) dy \right\rangle \;. \end{split}$$

**Step 3.** The properties  $\langle (-\Delta)^s u, v \rangle = \langle \nabla^s u, \nabla^s v \rangle$  and  $\langle Au, Av \rangle = \langle v, u \int_{\mathbb{R}^n} |A(x, y)|^2 dy \rangle$  hold, as proved in [14]. Using this information and formulas (22), (23) we can write the fractional magnetic Schrödinger operator as

$$\begin{split} \langle (-\Delta)^s u, v \rangle + \left\langle \nabla^{\lfloor s \rfloor} u(x) \cdot \int_{\mathbb{R}^n} S(x, y) dy + (-1)^{\lfloor s \rfloor} (\nabla \cdot)_x^{\lfloor s \rfloor} \left( u(x) \int_{\mathbb{R}^n} S(x, y) dy \right), v \right\rangle + \\ + (-1)^{\lfloor s \rfloor + 1} \left\langle \int_{\mathbb{R}^n} u(y) \left( (\nabla \cdot)_x^{\lfloor s \rfloor} S(y, x) + (\nabla \cdot)_y^{\lfloor s \rfloor} S(x, y) \right) dy, v \right\rangle + \left\langle u \int_{\mathbb{R}^n} |A|^2 dy, v \right\rangle. \end{split}$$

Let us compute the left hand side of the second bracket and collect the resulting terms according to the order of their derivatives of u. For every multi-index  $\beta$  such that  $|\beta| \leq \lfloor s \rfloor$  we can find a linear operator  $\mathfrak{M}_{\beta}$  such that

$$\nabla^{\lfloor s \rfloor} u(x) \cdot \int_{\mathbb{R}^n} S(x,y) dy + (-1)^{\lfloor s \rfloor} (\nabla \cdot)_x^{\lfloor s \rfloor} \left( u(x) \int_{\mathbb{R}^n} S(x,y) dy \right) = \sum_{|\beta| \leq \lfloor s \rfloor} \partial^{\beta} u(x) \mathfrak{M}_{\beta}(S) \; .$$

We can also define the following linear operator:

$$\mathfrak{N}(S) = (-1)^{\lfloor s \rfloor + 1} \left( (\nabla \cdot)_x^{\lfloor s \rfloor} S(y, x) + (\nabla \cdot)_y^{\lfloor s \rfloor} S(x, y) \right) \ .$$

With these new definitions, we can rewrite the fractional magnetic Schrödinger operator as in the statement of the Lemma.  $\hfill \Box$ 

6.3. The bilinear form and the DN map. For every  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$  and  $u, v \in H^s$  we define the bilinear form  $B^s_{A,q} : H^s \times H^s \to \mathbb{R}$  as in [14]:

$$B^s_{A,q}(u,v) = \int_{\mathbb{R}^n} \int_{\mathbb{R}^n} \nabla^s_A u \cdot \nabla^s_A v \, dy dx + \int_{\mathbb{R}^n} quv \, dx$$

**Lemma 6.5.** There are constants  $\mu', k' > 0$  such that, for all  $u \in H^s$ ,

$$B^s_{A,q}(u,u) + \mu' \langle u, u \rangle \ge k' \|u\|^2_{H^s}$$

Proof. The formula we want to prove is called *coercivity estimate*. Using (18), we can write

$$B_{A,q}^{s}(u,u) = \int_{\mathbb{R}^{n}} \int_{\mathbb{R}^{n}} \nabla_{A}^{s} u \cdot \nabla_{A}^{s} u \, dy dx + \int_{\mathbb{R}^{n}} qu^{2} \, dx$$
  

$$= \int_{\mathbb{R}^{n}} u(-\Delta)_{A}^{s} u \, dx + \int_{\mathbb{R}^{n}} qu^{2} \, dx = \langle (-\Delta)_{A}^{s} u, u \rangle + \langle qu, u \rangle$$
  

$$= \langle (-\Delta)^{s} u, u \rangle + 2 \langle \nabla^{s} u, Au \rangle + \left\langle \left( q + \int_{\mathbb{R}^{n}} |A(x,y)|^{2} dy \right) u, u \right\rangle$$
  

$$= \langle (-\Delta)^{s} u, u \rangle + 2 \left\langle \int_{\mathbb{R}^{n}} \nabla^{s} u \cdot A \, dy, u \right\rangle + \langle Qu, u \rangle ,$$

where  $Q(x) := q(x) + \int_{\mathbb{R}^n} |A(x,y)|^2 dy$  belongs to  $L^p$  since Cauchy-Schwartz and assumptions (a1) and (a3) imply the embedding  $L^{2p} \times L^{2p} \hookrightarrow L^p$ . Since we always have  $L^p \times H^s \hookrightarrow H^{-s}$ , we get  $\langle Qu, u \rangle \leq ||u||_{H^s} ||Qu||_{H^{-s}} \leq ||Q||_{L^p} ||u||_{H^s}^2$ . For the second term on the right hand side of (24) we first perform an estimate by means of the Young inequality

$$2\left\langle \int_{\mathbb{R}^n} \nabla^s u \cdot A \, dy, u \right\rangle \leq \epsilon^{-1} \|u\|_{L^2}^2 + \epsilon \left\| \int_{\mathbb{R}^n} \nabla^s u \cdot A \, dy \right\|_{L^2}^2 \,,$$

then estimate the second term with the Cauchy-Schwartz inequality, in light of (a4):

$$\begin{split} \epsilon \left\| \int_{\mathbb{R}^n} \nabla^s u \cdot A \, dy \right\|_{L^2}^2 &= \epsilon \left\| \int_{\mathbb{R}^n} \left( (\nabla^{\lfloor s \rfloor} u(x) - \nabla^{\lfloor s \rfloor} u(y)) \otimes \alpha \right) \cdot A \, dy \right\|_{L^2}^2 \\ &= \epsilon \left\| \int_{\mathbb{R}^n} (\nabla^{\lfloor s \rfloor} u(x) - \nabla^{\lfloor s \rfloor} u(y)) \cdot (A \cdot \alpha) \, dy \right\|_{L^2}^2 \\ &= \epsilon \left\| \int_{\Omega} (\nabla^{\lfloor s \rfloor} u(x) - \nabla^{\lfloor s \rfloor} u(y)) \cdot S(x, y) \, dy \right\|_{L^2(\Omega)}^2 \\ &\leq \epsilon \left\| \left( \int_{\Omega} |\nabla^{\lfloor s \rfloor} u(x) - \nabla^{\lfloor s \rfloor} u(y)|^2 dy \right)^{1/2} \left( \int_{\Omega} |S(x, y)|^2 \, dy \right)^{1/2} \right\|_{L^2(\Omega)}^2 \\ &= \epsilon \int_{\Omega} \left( \int_{\Omega} |\nabla^{\lfloor s \rfloor} u(x) - \nabla^{\lfloor s \rfloor} u(y)|^2 dy \int_{\Omega} |S(x, y)|^2 \, dy \right) dx \end{split}$$

$$\begin{split} &\leq \epsilon \int_{\Omega} \left( \int_{\Omega} (|\nabla^{\lfloor s \rfloor} u(x)| + |\nabla^{\lfloor s \rfloor} u(y)|)^2 dy \int_{\Omega} \tilde{S}^2(y) \, dy \right) dx \\ &= \epsilon \|\tilde{S}\|_{L^2(\Omega)}^2 \int_{\Omega} \int_{\Omega} (|\nabla^{\lfloor s \rfloor} u(x)| + |\nabla^{\lfloor s \rfloor} u(y)|)^2 dy dx \\ &\leq 2\epsilon \|\tilde{S}\|_{L^2(\Omega)}^2 \int_{\Omega} \int_{\Omega} (|\nabla^{\lfloor s \rfloor} u(x)|^2 + |\nabla^{\lfloor s \rfloor} u(y)|^2) dy dx \\ &\leq 4 |\Omega| \epsilon \|\tilde{S}\|_{L^2(\Omega)}^2 \|\nabla^{\lfloor s \rfloor} u\|_{L^2}^2 \leq c \, \epsilon \|\nabla^{\lfloor s \rfloor} u\|_{H^{s'}}^2 \leq c \, \epsilon \|u\|_{H^s}^2, \end{split}$$

where the constant c can change from line to line and does not depend on u. Eventually

$$2\left\langle \int_{\mathbb{R}^n} \nabla^s u \cdot A \, dy, u \right\rangle \le \epsilon^{-1} \|u\|_{L^2}^2 + c \, \epsilon \|u\|_{H^s}^2,$$

which leads to

(25) 
$$B^{s}_{A,q}(u,u) \ge B^{s}_{0,Q}(u,u) - \epsilon^{-1} \|u\|^{2}_{L^{2}} - c \,\epsilon \|u\|^{2}_{H^{s}} \,.$$

Since  $C_c^{\infty}(\Omega)$  is dense in  $L^p(\Omega)$ , for every  $\delta > 0$  we can find functions  $Q_s, Q_r$  such that  $Q_s \in$  $C_c^{\infty}(\Omega), \|Q_r\|_{L^p(\Omega)} \leq \delta$  and  $Q = Q_s + Q_r$ . Also, if  $\phi_j \in C_c^{\infty}(\Omega)$  and  $\|\phi_j\|_{H^s} = 1$  for j = 1, 2,then  $|\langle Q_r\phi_1,\phi_2\rangle| \leq c \|\phi_1\|_{H^s} \|\phi_2\|_{H^s} \|Q_r\|_{L^p} \leq c\delta$  by the embedding  $L^p \times H^s \hookrightarrow H^{-s}$ . Therefore,

$$||Q_r||_{Z^{-s}} = \sup_{||\phi_j||_{H^s} = 1} \{ |\langle Q_r \phi_1, \phi_2 \rangle| \} \le c\delta$$

which means that Q belongs to the closure of  $C_c^{\infty}(\Omega)$  in  $Z^{-s}(\mathbb{R}^n)$ , that is  $Q \in Z_0^{-s}(\mathbb{R}^n)$ . Now by Lemma 5.1 we know the coercivity estimate for the non-magnetic high exponent case; this lets us write (25) as

$$B_{A,q}^{s}(u,u) + (\mu + \epsilon^{-1})\langle u, u \rangle \ge (k - c \epsilon) ||u||_{H^{s}}^{2},$$

which is the coercivity estimate for  $B_{A,q}^s$  as soon as  $\epsilon$  is fixed small enough and  $\mu' := \mu + \epsilon^{-1}$ ,  $k' := k - c \epsilon$  are defined.

By means of the lemma above, if we assume 0 is not an eigenvalue for the equation, we can proceed as in the proof of Lemma 2.6 from [75] and get the well-posedness of the direct problem for FMSE. This can be stated as follows: if  $F \in (H^s(\Omega))^*$ , there exists unique solution  $u \in H^s(\mathbb{R}^n)$  to  $B^s_{A,q}(u,v) = F(v)$  for all  $v \in \widetilde{H}^s(\Omega)$ , i.e. unique  $u \in H^s(\mathbb{R}^n)$  such that  $(-\Delta)^s_A u + qu = F$  in  $\Omega$ ,  $u|_{\Omega_e} = 0$ . This is also true for non-vanishing exterior value  $f \in H^s(\mathbb{R}^n)$ (see [15] and [28]), and the following estimate holds:

(26) 
$$||u||_{H^{s}(\mathbb{R}^{n})} \leq c(||F||_{(\widetilde{H}^{s}(\Omega))^{*}} + ||f||_{H^{s}(\mathbb{R}^{n})}),$$

where c does not depend on F and f.

One can prove (see Lemma 3.11 from [14]) that  $B_{A,q}^s$  also enjoys these properties:

- $\begin{array}{ll} (1) & B^{s}_{A,q}(v,w) = B^{s}_{A,q}(w,v) \text{ , for all } v,w \in H^{s}, \\ (2) & |B^{s}_{A,q}(v,w)| \leq c \|v\|_{H^{s}(\mathbb{R}^{n})} \|w\|_{H^{s}(\mathbb{R}^{n})} \text{ for all } v,w \in H^{s}, \text{ where } c \text{ does not depend on } v \end{array}$ and w.
- (3)  $B^s_{A,q}(u_1, e_2) = B^s_{A,q}(u_2, e_1)$ , for  $u_j \in H^s$  solution to the direct problem for FMSE with exterior value  $f_j \in H^s(\Omega_e)$  and  $e_j$  any extension of  $f_j$  to  $H^s$ , j = 1, 2.

**Lemma 6.6.** Let  $X = H^s(\mathbb{R}^n)/\widetilde{H}^s(\Omega)$  be the abstract quotient space, and let  $u_1 \in H^s$  be the solution to the direct problem for FMSE with exterior value  $f_1 \in H^s(\Omega_e)$ . Then

$$\langle \Lambda_{A,q}^s[f_1], [f_2] \rangle = B_{A,q}^s(u_1, f_2), \qquad f_j \in H^s, \ j = 1, 2$$

defines a bounded, linear, self-adjoint map  $\Lambda_{A,q}^s: X \to X^*$ . We call  $\Lambda_{A,q}^s$  the DN map.

*Proof.* The proof follows trivially from properties (1)-(3) of  $B_{A,q}^s$  and (26).

6.4. The gauge. Consider two couples of potentials  $(A_1, q_1)$  and  $(A_2, q_2)$ . We say that  $(A_1, q_1) \sim (A_2, q_2)$  if and only if the following conditions are met:

- $\mathfrak{N}(S_1 S_2) = 0$  for almost every  $x, y \in \mathbb{R}^n$
- $\mathfrak{M}_{(0,\dots,0)}(\tilde{S}_1 S_2) + \int_{\mathbb{R}^n} (|A_1|^2 |A_2|^2) dy + (q_1 q_2) = 0$  for almost every  $x \in \mathbb{R}^n$
- $\mathfrak{M}_{\beta}(S_1 S_2) = 0$  for all  $1 \leq |\beta| \leq \lfloor s \rfloor$  and almost every  $x \in \mathbb{R}^n$ .

It is clear from the linearity of  $\mathfrak{N}$  and  $\mathfrak{M}_{\alpha}$  that  $\sim$  is an equivalence relation, and so the set of all couples of potentials is divided into equivalence classes by  $\sim$ . We call these gauge classes, and if  $(A_1, q_1) \sim (A_2, q_2)$  we say that  $(A_1, q_1)$  and  $(A_2, q_2)$  are in gauge.

Observe that this gauge  $\sim$  coincides with the one defined in [14] if  $s \in (0, 1)$ , although it looks quite different. Since in this case  $\lfloor s \rfloor = 0$ , there is no third condition. In the language of that paper, the first condition reads

$$D = -\Re(S_1 - S_2) = S_1(y, x) + S_1(x, y) - S_2(y, x) - S_2(x, y)$$
  
=  $(A_1(x, y) - A_2(x, y)) \cdot \alpha(x, y) + (A_1(y, x) - A_2(y, x)) \cdot \alpha(y, x)$   
=  $(A_1(x, y) - A_1(y, x) - A_2(x, y) + A_2(y, x)) \cdot \alpha(x, y)$   
=  $2(A_1 - A_2)_a \cdot \alpha = 2(A_1 - A_2)_{a\parallel} \cdot \alpha$ ,

which is equivalent to  $(A_1)_{a\parallel} = (A_2)_{a\parallel}$ , since the two vectors in the last scalar product have the same direction. Given this fact, for any  $v \in H^s$  the first term in the second condition weakly is

$$\begin{split} \langle \mathfrak{M}_{(0,\dots,0)}(S_1 - S_2), v \rangle &= 2\langle S_1 - S_2, v \rangle = 2\langle \alpha \cdot (A_1 - A_2), v \rangle = 2\langle \alpha \cdot (A_1 - A_2)_{\parallel}, v \rangle \\ &= 2\langle \alpha \cdot (A_1 - A_2)_{s\parallel}, v \rangle = 2\langle \alpha v, (A_1 - A_2)_{s\parallel} \rangle = 2\langle (\alpha v)_s, (A_1 - A_2)_{s\parallel} \rangle \\ &= \langle \alpha(x, y)v(x) + \alpha(y, x)v(y), (A_1 - A_2)_{s\parallel} \rangle \\ &= \langle \alpha(x, y)(v(x) - v(y)), (A_1 - A_2)_{s\parallel} \rangle \\ &= \langle \nabla^s v, (A_1 - A_2)_{s\parallel} \rangle = \langle v, (\nabla \cdot)^s ((A_1 - A_2)_{s\parallel}) \rangle \;, \end{split}$$

which lets us rewrite the second condition as

$$(\nabla \cdot)^s (A_1)_{s\parallel} + \int_{\mathbb{R}^n} |A_1|^2 dy + q_1 = (\nabla \cdot)^s (A_2)_{s\parallel} + \int_{\mathbb{R}^n} |A_2|^2 dy + q_2 dy + q_2$$

**Remark 6.7.** Observe that the gauge enjoyed by the FMSE is quite different from the one holding for the MSE. For the sake of simplicity, we shall compare the classical case with the fractional one in the regime  $s \in (0, 1)$ , following section 3 in [14].

Given lemma 6.4, one sees that the following is an equivalent definition for the gauge  $\sim$  above:

$$(A_1, q_1) \sim (A_2, q_2) \quad \Leftrightarrow \quad (-\Delta)^s_{A_1} u + q_1 u = (-\Delta)^s_{A_2} u + q_2 u ,$$

for all  $u \in H^{s}(\mathbb{R}^{n})$ . One may also define the accessory gauge  $\approx$  as

$$(A_1, q_1) \approx (A_2, q_2) \quad \Leftrightarrow \quad \exists \phi \in G : (-\Delta)^s_{A_1}(u\phi) + q_1 u\phi = \phi((-\Delta)^s_{A_2}u + q_2 u)$$

for all  $u \in H^s(\mathbb{R}^n)$ , where  $G := \{\phi \in C^\infty(\mathbb{R}^n) : \phi > 0, \phi|_{\Omega_e} = 1\}$ . These definitions can be extended to the MSE in the natural way. It was proved in lemmas 3.9 and 3.10 of [14] that the FMSE enjoys the gauge  $\sim$ , but not  $\approx$ . In the same discussion, it was argued that the opposite holds for MSE. The reason for this surprising discrepancy should be looked for in the nonlocal structure of the FMSE. As apparent in formula (10) in [14], the coefficient of the gradient term in FMSE is not related to the whole vector potential A itself, but only to its antisymmetric part  $A_a$ . It is such antisymmetry requirement what eventually does not allow the FMSE to enjoy  $\approx$ as the MSE. As a result, the scalar potential q can not be in general uniquely determined as in the classical case.

#### 6.5. Main result.

**Remark 6.8.** Assume  $W \subseteq \Omega_e$  is an open set and  $u \in H^s$  satisfies u = 0 and  $(-\Delta)_A^s u + qu = 0$  in W. We say that the fractional magnetic Schrödinger operator enjoys the weak unique continuation property (WUCP) if we can deduce that u = 0 in  $\Omega$ . This was proved in [14] by using the UCP of the fractional Laplacian for  $s \in (0, 1)$ ; since we know by Theorem 1.2 that

UCP still holds for  $(-\Delta)^s$  in the regime  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$ , we can deduce WUCP for  $(-\Delta)^s_A + q$  by the same proof.

Proof of theorem 1.8. Step 1. Without loss of generality, let  $W_1 \cap W_2 = \emptyset$ . Let  $f_i \in C_c^{\infty}(W_i)$ , and let  $u_i \in H^s(\mathbb{R}^n)$  solve  $(-\Delta)_{A_i}^s u_i + q_i u_i = 0$  with  $u_i - f_i \in \widetilde{H}^s(\Omega)$  for i = 1, 2. Knowing that the DN maps computed on  $f \in C_c^{\infty}(W_1)$  coincide when restricted to  $W_2$ , using Lemmas 6.4 and 6.6 we write this integral identity

$$\begin{split} 0 &= \left\langle (\Lambda_{A_1,q_1}^s - \Lambda_{A_2,q_2}^s) f_1, f_2 \right\rangle = B_{A_1,q_1}^s (u_1, u_2) - B_{A_2,q_2}^s (u_1, u_2) \\ &= \left\langle u_2, \sum_{|\beta| \le \lfloor s \rfloor} \partial^\beta u_1 \mathfrak{M}_\beta (S_1 - S_2) \right\rangle + \left\langle u_2, \int_{\mathbb{R}^n} u_1(y) \mathfrak{N}(S_1 - S_2) \, dy \right\rangle + \\ &+ \left\langle u_2, u_1 \left( \int_{\mathbb{R}^n} (|A_1|^2 - |A_2|^2) dy + (q_1 - q_2) \right) \right\rangle. \end{split}$$

Since if  $x \notin \Omega$  or  $y \notin \Omega$  we have  $A_1(x, y) = A_2(x, y)$  and  $q_1(x) = q_2(x)$ , we can restrict  $u_1, u_2$ and  $\partial^{\beta} u_1$  over  $\Omega$  in the previous formula; it is also true that  $(\partial^{\beta} u_1)|_{\Omega} = \partial^{\beta}(u_1|_{\Omega})$ , and therefore

$$0 = \left\langle u_2|_{\Omega}, \sum_{|\beta| \le \lfloor s \rfloor} \partial^{\beta}(u_1|_{\Omega}) \mathfrak{M}_{\beta}(S_1 - S_2) \right\rangle + \left\langle u_2|_{\Omega}, \int_{\mathbb{R}^n} u_1|_{\Omega}(y) \mathfrak{N}(S_1 - S_2) \, dy \right\rangle + \left\langle u_2|_{\Omega}, u_1|_{\Omega} \left( \int_{\mathbb{R}^n} (|A_1|^2 - |A_2|^2) dy + (q_1 - q_2) \right) \right\rangle.$$

This is the Alessandrini identity, which now we will test with certain solutions in order to obtain information about the potentials. The appropriate test solutions will be produced by means of the Runge approximation property (RAP), which holds for the FMSE because of Remark 6.8 and Lemma 3.15 in [14]. This property says that the set  $\mathcal{R} = \{u_f|_{\Omega} : f \in C_c^{\infty}(W)\} \subset L^2(\Omega)$ of the restrictions to  $\Omega$  of those functions  $u_f$  solving FMSE for some smooth exterior value fsupported in W is dense in  $L^2(\Omega)$ .

**Step 2.** Given any  $f \in L^2(\Omega)$ , by the RAP we can find a sequence of solutions  $(u_2)_k \to f$  in  $L^2$  sense as  $k \to \infty$ . Substituting these in the Alessandrini identity and taking limits, by the arbitrarity of f we can deduce that

$$0 = \sum_{|\beta| \le \lfloor s \rfloor} \partial^{\beta}(u_1|_{\Omega}) \mathfrak{M}_{\beta}(S_1 - S_2) + \int_{\mathbb{R}^n} u_1|_{\Omega}(y) \mathfrak{N}(S_1 - S_2) \, dy + u_1|_{\Omega} \left( \int_{\mathbb{R}^n} (|A_1|^2 - |A_2|^2) dy + (q_1 - q_2) \right)$$

holds for every solution  $u_1 \in H^s$  and almost every point  $x \in \Omega$ . Fix  $x \in \Omega$ . Consider now any  $\psi \in C_c^{\infty}(\Omega)$  and let  $g(y) := e^{-1/|x-y|}\psi(y)$ , g(x) = 0. Since  $e^{-1/|x-y|}$  is smooth, it is easy to see that  $g \in C_c^{\infty}(\Omega) \subset L^2(\Omega)$ ; also, by the properties of  $e^{-1/|x-y|}$  one has that  $\partial^{\beta}g(x) = 0$  for all multi-indices  $\beta$ . By the RAP we can find a sequence of solutions  $(u_1)_k \to g$  in  $L^2$  sense as  $k \to \infty$ . Substituting these in the above identity and taking limits, we get

$$\int_{\mathbb{R}^n} e^{-1/|x-y|} \psi(y) \mathfrak{N}(S_1 - S_2) \, dy = 0 \; ,$$

which by the arbitrarity of  $\psi$  and the positivity of the exponential now implies  $\mathfrak{N}(S_1 - S_2) = 0$ for almost all  $x, y \in \Omega$ . We can now return to the above equation with this new information: for every solution  $u_1 \in H^s$  and almost every  $x \in \Omega$ ,

$$0 = \sum_{|\beta| \le \lfloor s \rfloor} \partial^{\beta}(u_1|_{\Omega}) \mathfrak{M}_{\beta}(S_1 - S_2) + u_1|_{\Omega} \left( \int_{\mathbb{R}^n} (|A_1|^2 - |A_2|^2) dy + (q_1 - q_2) \right) \,.$$

For every multi-index  $\beta$  we can consider the function  $h_{\beta}(x) = x^{\beta} = x_1^{\beta_1} \dots x_n^{\beta_n}$ , which belongs to  $L^2(\Omega)$ . Let  $(h_{\beta})_k$  be a sequence of solutions approximating  $h_{\beta}$  in  $L^2$ , which exists by the RAP. We will first substitute  $(h_{(0,\dots,0)})_k$  into the last formula, take limits and deduce

$$\mathfrak{M}_{(0,\dots,0)}(S_1 - S_2) + \int_{\mathbb{R}^n} (|A_1|^2 - |A_2|^2) dy + (q_1 - q_2) = 0 ,$$

which has the effect of reducing the equation to

$$\sum_{1 \le |\beta| \le \lfloor s \rfloor} \partial^{\beta}(u_1|_{\Omega}) \mathfrak{M}_{\beta}(S_1 - S_2) = 0.$$

If  $\lfloor s \rfloor \geq 1$ , we will repeat the last steps with each  $h_{\beta}$  such that  $|\beta| = 1$ , deducing  $\mathfrak{M}_{\beta}(S_1 - S_2) = 0$  for every such  $\beta$ , and subsequently

$$\sum_{\leq |\beta| \leq \lfloor s \rfloor} \partial^{\beta}(u_1|_{\Omega}) \mathfrak{M}_{\beta}(S_1 - S_2) = 0.$$

Repeating this process for a total of  $\lfloor s \rfloor$  times eventually leads to

$$\mathfrak{M}_{\beta}(S_1 - S_2) = 0 \quad \forall \ 1 \le |\beta| \le \lfloor s \rfloor,$$

which proves the theorem by the definition of the gauge  $\sim$ .

2

### 7. Possible generalizations and applications beyond this article

We discuss some possible directions for the future research on higher order fractional inverse problems, fractional Poincaré inequalities and unique continuation properties. It seems that now it would be the most natural to reconsider many of the recent developments in fractional inverse problems for higher order operators. We outline here some problems which we would like to see solved in the future.

We have split this section in three in order to emphasize some open problems which we find especially interesting. We do not claim that answers to all questions are positive and it would be interesting to see why and where the greatest difficulties, or even counterexamples, would show up. We first list the most natural directions to continue our work on higher order fractional Calderón problems. One could study for example the following cases:

- (i) Is reconstruction from a single measurement [15, 27] possible also in the higher order cases?
- (ii) Is there stability [75] in the higher order cases?
- (iii) Is there exponential instability [73] in the higher order cases?
- (iv) Is there uniqueness for the Calderón problem for fractional semilinear Schrödinger equations [47, 48] in the higher order cases?
- (v) Do the monotonicity methods [33, 34] extend to the higher order cases?
- (vi) Is there uniqueness for the conductivity type fractional Calderón problems [10, 15] in the higher order cases?
- (vii) Could recent results on fractional heat equations [49, 74] be generalized to the higher order cases?
- (viii) Does the higher regularity Runge approximation in [11, 28] generalize to higher order cases?

7.1. Unique continuation problems. We state here some unique continuation problems, which do not follow directly from the earlier results and the techniques that we have developed for this article.

**Question 7.1** (UCP for Bessel potentials). Let  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$ ,  $p \in [1, \infty)$  and  $r \in \mathbb{R}$ . Let  $V \subset \mathbb{R}^n$  be an open set. Suppose that  $f \in H^{r,p}(\mathbb{R}^n)$ ,  $f|_V = 0$  and  $(-\Delta)^s f|_V = 0$ . Show that  $f \equiv 0$  or give a counterexample.

The positive answer to question 7.1 is known when  $p \in [1, 2]$  (see corollary 3.5). If f has compact support, then the answer is positive for all  $p \in [1, \infty)$  (see corollary 3.3). Question 7.1 is also open for the exponents  $s \in (0, 1)$  when p > 2. See section 3.1 for details.

**Question 7.2** (Measurable UCP). Let  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$  and  $r \in \mathbb{R}$ . Let  $V \subset \mathbb{R}^n$  be an open set and  $E \subset V$  a measurable set with positive measure. Suppose that  $f \in H^r(\mathbb{R}^n)$ ,  $f|_E = 0$  and  $(-\Delta)^s f|_V = 0$ . Show that  $f \equiv 0$  or give a counterexample.

The positive answer to question 7.2 is known when  $s \in (0, 1)$  [27]. Question 7.2 with a potential q from a suitable class of functions is also an interesting and more challenging problem. See [27, Proposition 5.1] for more details.

**Question 7.3** (Alternative strong UCP). Let  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$  and  $r \in \mathbb{R}$ . Let  $V \subset \mathbb{R}^n$  be an open set. Suppose that  $f \in H^r(\mathbb{R}^n)$ ,  $f|_V = 0$  and  $\partial^{\beta}((-\Delta)^s f)(x_0) = 0$  for some  $x_0 \in V$  and all  $\beta \in \mathbb{N}_0^n$ . Show that  $f \equiv 0$  or give a counterexample.

Question 7.3 can be seen as a version of the strong unique continuation property (see e.g. [22, 26, 72]) with interchanged decay conditions. When f has compact support, the answer to question 7.3 is positive for  $s \in (-n/2, \infty) \setminus \mathbb{Z}$  (see corollary 3.3).

The problems posed in questions 7.1–7.3 for the fractional Laplacian are interesting mathematical problems on their own right, but they also have important applications in inverse problems. The UCPs can be used to show Runge approximation properties for nonlocal equations such as the fractional Schrödinger equation (see theorem 1.7), which in turn can be used to show uniqueness for the corresponding nonlocal inverse problem (see theorem 1.6). The UCPs have also applications in integral geometry, where the uniqueness of the ROI problem for the d-plane transform can be reduced to a unique continuation problem for the fractional Laplacian (see remark 4.1 and corollaries 1.3 and 1.4).

7.2. Fractional Poincaré inequality for  $L^p$ -norms. In section 3.2 we prove the fractional Poincaré inequality for  $L^2$ -norms in multiple ways. The inequality is needed for the well-posedness of the inverse problem for the fractional Schrödinger equation. One could try to extend the Poincaré inequality for general  $L^p$ -norms. This suggests the following natural question which is also interesting from the pure mathematical point of view.

**Question 7.4.** Let  $s \ge 0$ ,  $1 \le p < \infty$ ,  $K \subset \mathbb{R}^n$  compact set and  $u \in H^{s,p}(\mathbb{R}^n)$  such that  $spt(u) \subset K$ . Show that there exists a constant c = c(n, K, s, p) such that

(27) 
$$\|u\|_{L^p(\mathbb{R}^n)} \le c \left\| (-\Delta)^{s/2} u \right\|_{L^p(\mathbb{R}^n)}$$

or give a counterexample.

Since we have presented several proofs for the Poincaré inequality in the case p = 2, one could try some of our methods to solve question 7.4. However, some of our proofs are heavily based on Fourier analysis and those approaches might be difficult to generalize to the  $L^{p}$ -case when  $p \neq 2$ . Like in theorem 1.5 and in theorem 3.17, another interesting question is whether one can replace u in the left-hand side of equation (27) with  $(-\Delta)^{t/2}u$  when  $0 \leq t \leq s$ , and whether the constant c in equation (27) can be expressed in terms of the classical Poincaré constant when  $s \geq 1$ .

7.3. The Calderón problem for determining a higher order PDO. In this discussion, we try to make as simple assumptions as possible. The whole point is to introduce a new inverse problem that we think is a very natural and interesting one, at least from a pure mathematical point of view. Therefore the optimal regularity in the statement of the problem is not as important. Let  $\Omega$  be a domain with smooth boundary. Suppose that  $P(x, D) = \sum_{|\alpha| \leq m} a_{\alpha}(x)D^{\alpha}$  is a partial differential operator (PDO) of order m with smooth coefficients on  $\Omega$ . We argue in section 3.1 that the operator  $(-\Delta)^s + P(x, D)$  admits the UCP (in open sets).

It is shown in the seminal work of Ghosh, Uhlmann and Salo [28] that if P(x, D) is of order m = 0, then one can determine the zeroth order coefficient (i.e. the potential q) from the

associated DN map. It was then later shown in [11] that if P(x, D) is of order m = 1, then one can also determine the coefficients (i.e. the potential q and the magnetic drift b) from the associated DN map whenever the order of  $(-\Delta)^s$  is large enough, namely when 2s > 1. This and our work on higher order Calderón problems motivate the following inverse problem.

**Question 7.5.** Suppose that  $\Omega \subset \mathbb{R}^n$  is a bounded open domain with smooth boundary. Let  $P_j(x, D), j = 1, 2$ , be smooth PDOs of order  $m \in \mathbb{N}$  in  $\Omega$ . Let  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$  be such that 2s > m. Given any two open sets  $W_1, W_2 \subset \Omega_e$ , suppose that the DN maps  $\Lambda_{P_i}$  for the equations

$$((-\Delta)^s + P_j(x,D))u_j = 0$$
 in  $\Omega$ 

satisfy  $\Lambda_{P_1}f|_{W_2} = \Lambda_{P_2}f|_{W_2}$  for all  $f \in C_c^{\infty}(W_1)$ . Show that  $P_1(x, D) = P_2(x, D)$  or give a counterexample.

Another interesting question is whether the strong UCP [26] can be extended to higher order PDOs.

#### References

- [1] H. Abels. Pseudodifferential and Singular Integral Operators. De Gruyter, First edition, 2012.
- [2] A. Abouelaz. The *d*-plane Radon transform on the torus  $\mathbb{T}^n$ . Fract. Calc. Appl. Anal., 14(2):233–246, 2011.
- [3] F. Andreu-Vaillo, J. M. Mazón, J. D. Rossi, and J. J. Toledo-Melero. Nonlocal Diffusion Problems. American Mathematical Society, First edition, 2010.
- [4] H. Bahouri, J.-Y. Chemin, and R. Danchin. Fourier Analysis and Nonlinear Partial Differential Equations. Springer, First edition, 2011.
- [5] A. Behzadan and M. Holst. Multiplication in Sobolev spaces, revisited. 2017. arXiv:1512.07379v2.
- [6] J. Bergh and J. Löfström. Interpolation Spaces, An Introduction. Springer-Verlag, First edition, 1976.
- [7] C. Bucur and E. Valdinoci. Nonlocal Diffusion and Applications. Springer, First edition, 2016.
- [8] X. Cabré and Y. Sire. Nonlinear equations for fractional Laplacians, I: Regularity, maximum principles, and Hamiltonian estimates. Annales de l'I.H.P. Analyse Non Linéaire, 31(1):23–53, 2014.
- [9] L. Caffarelli and L. Silvestre. An Extension Problem Related to the Fractional Laplacian. Communications in Partial Differential Equations, 32, 2006.
- [10] X. Cao, Y.-H. Lin, and H. Liu. Simultaneously recovering potentials and embedded obstacles for anisotropic fractional Schrödinger operators. *Inverse Probl. Imaging*, 13(1):197–210, 2019.
- [11] M. Cekic, Y.-H. Lin, and A. Ruland. The Calderón problem for the fractional Schrödinger equation with drift. Calculus of Variations and Partial Differential Equations, 59(3):91, 2020.
- [12] S. N. Chandler-Wilde, D. P. Hewett, and A. Moiola. Sobolev Spaces on Non-Lipschitz Subsets of R<sup>n</sup> with Application to Boundary Integral Equations on Fractal Screens. *Integral Equations and Operator Theory*, 87(2):179–224, 2017.
- [13] M. Courdurier, F. Noo, M. Defrise, and H. Kudo. Solving the interior problem of computed tomography using a priori knowledge. *Inverse Problems*, 24(6):065001, 2008.
- [14] G. Covi. An inverse problem for the fractional Schrödinger equation in a magnetic field. *Inverse Problems*, 36(4):045004, 2020.
- [15] G. Covi. Inverse problems for a fractional conductivity equation. Nonlinear Analysis, 193:111418, 2020. Nonlocal and Fractional Phenomena.
- [16] A. D'Agnolo and M. Eastwood. Radon and Fourier transforms for *D*-modules. Adv. Math., 180(2):452–485, 2003.
- [17] S. Dipierro, O. Savin, and E. Valdinoci. All functions are locally s-harmonic up to a small error. Journal of the European Mathematical Society, 19(4):957–966, 2017.
- [18] Q. Du, M. Gunzburger, R. B. Lehoucq, and K. Zhou. Analysis and Approximation of Nonlocal Diffusion Problems with Volume Constraints. SIAM Rev., 54, No. 4:667–696, 2012.
- [19] Q. Du, M. Gunzburger, R. B. Lehoucq, and K. Zhou. A nonlocal vector calculus, nonlocal volume-constrained problems, and nonlocal balance laws. *Math. Models Methods Appl. Sci.*, 23, No. 3:493–540, 2013.
- [20] G. Eskin. Global uniqueness in the inverse scattering problem for the Schrödinger operator with external Yang-Mills potentials. *Communications in Mathematical Physics*, 222(3):503–531, 2001.
- [21] L. C. Evans. Partial differential equations, volume 19 of Graduate Studies in Mathematics. American Mathematical Society, Providence, RI, second edition, 2010.
- [22] M. M. Fall and V. Felli. Unique continuation property and local asymptotics of solutions to fractional elliptic equations. Comm. Partial Differential Equations, 39(2):354–397, 2014.
- [23] V. Felli and A. Ferrero. Unique continuation principles for a higher order fractional Laplace equation. Nonlinearity, 33(8):4133–4190, 2020.
- [24] G. B. Folland and A. Sitaram. The Uncertainty Principle: A Mathematical Survey. Journal of Fourier Analysis and Applications, 3(3):207–238, 1997.

- [25] J. Frikel and E. T. Quinto. Limited data problems for the generalized Radon transform in  $\mathbb{R}^n$ . SIAM J. Math. Anal., 48(4):2301–2318, 2016.
- [26] M.-Á. García-Ferrero and A. Rüland. Strong unique continuation for the higher order fractional Laplacian. Mathematics in Engineering, 1(4):715–774, 2019.
- [27] T. Ghosh, A. Rüland, M. Salo, and G. Uhlmann. Uniqueness and reconstruction for the fractional Calderón problem with a single measurement. *Journal of Functional Analysis*, 279(1):108505, 2020.
- [28] T. Ghosh, M. Salo, and G. Uhlmann. The Calderón problem for the fractional Schrödinger equation. Anal. PDE 13(2):455-475, 2020.
- [29] F. O. Goncharov. An iterative inversion of weighted radon transforms along hyperplanes. *Inverse Problems*, 33(12):124005, 20, 2017.
- [30] F. O. Goncharov and R. G. Novikov. An example of non-uniqueness for Radon transforms with continuous positive rotation invariant weights. J. Geom. Anal., 28(4):3807–3828, 2018.
- [31] F. O. Goncharov and R. G. Novikov. An example of non-uniqueness for the weighted Radon transforms along hyperplanes in multidimensions. *Inverse Problems*, 34(5):054001, 6, 2018.
- [32] F. B. Gonzalez. On the Range of the Radon d-Plane Transform and Its Dual. Transactions of the American Mathematical Society, 327(2):601–619, 1991.
- [33] B. Harrach and Y.-H. Lin. Monotonicity-based inversion of the fractional Schrödinger equation I. Positive potentials. SIAM Journal on Mathematical Analysis, 51(4):3092–3111, 2019.
- [34] B. Harrach and Y.-H. Lin. Monotonicity-based inversion of the fractional Schrödinger equation II. General potentials and stability. SIAM Journal on Mathematical Analysis, 52(1):402–436, 2020.
- [35] H. Heck, X. Li, and J.-N. Wang. Identification of Viscosity in an Incompressible Fluid. Indiana University Mathematics Journal, 56(5):2489–2510, 2007.
- [36] S. Helgason. Integral Geometry and Radon Transforms. Springer, First edition, 2011.
- [37] A. Homan and H. Zhou. Injectivity and stability for a generic class of generalized Radon transforms. J. Geom. Anal., 27(2):1515–1529, 2017.
- [38] L. Hörmander. The Analysis of Linear Partial Differential Operators I. Springer-Verlag, Second edition, 1990.
- [39] J. Horváth. Topological Vector Spaces and Distributions. volume I. Addison-Wesley, 1966.
- [40] J. Ilmavirta. On Radon transforms on tori. J. Fourier Anal. Appl., 21(2):370–382, 2015.
- [41] J. Ilmavirta and K. Mönkkönen. Unique continuation of the normal operator of the x-ray transform and applications in geophysics. *Inverse Problems*, 36(4):045014, 2020.
- [42] E. Katsevich, A. Katsevich, and G. Wang. Stability of the interior problem with polynomial attenuation in the region of interest. *Inverse Problems*, 28(6):065022, 2012.
- [43] E. Klann, E. T. Quinto, and R. Ramlau. Wavelet methods for a weighted sparsity penalty for region of interest tomography. *Inverse Problems*, 31(2):025001, 22, 2015.
- [44] V. P. Krishnan and E. T. Quinto. Microlocal Analysis in Tomography. In O. Scherzer, editor, Handbook of Mathematical Methods in Imaging, pages 847–902. Springer, New York, 2015.
- [45] N. Krylov. All functions are locally s-harmonic up to a small error. Journal of Functional Analysis, 277(8):2728 – 2733, 2019.
- [46] M. Kwaśnicki. Ten equivalent definitions of the fractional Laplace operator. Fractional Calculus and Applied Analysis, 20, 2015.
- [47] R.-Y. Lai and Y.-H. Lin. Global uniqueness for the fractional semilinear Schrödinger equation. Proc. Amer. Math. Soc., 147(3):1189–1199, 2019.
- [48] R.-Y. Lai and Y.-H. Lin. Inverse problems for fractional semilinear elliptic equations. 2020. arXiv:2004.00549.
- [49] R.-Y. Lai, Y.-H. Lin, and A. Rüland. The Calderón problem for a space-time fractional parabolic equation. SIAM Journal on Mathematical Analysis, 52(3):2655–2688, 2020.
- [50] N. Laskin. Fractional Quantum Mechanics and Lévy Path Integrals. Physics Letters A, 268(4):298–305, 2000.
- [51] N. Laskin. Fractional Quantum Mechanics. World Scientific, First edition, 2018.
- [52] L. Li. A semilinear inverse problem for the fractional magnetic Laplacian. arXiv:2005.06714, 2020.
- [53] L. Li. Determining the magnetic potential in the fractional magnetic Calderón problem. arXiv:2006.10150, 2020.
- [54] L. Li. The Calderón problem for the fractional magnetic operator. Inverse Problems, 36(7):075003, 2020.
- [55] V. G. Maz'ya and T. O. Shaposhnikova. Theory of Sobolev Multipliers. Springer, First edition, 2009.
- [56] S. R. McDowall. An electromagnetic inverse problem in chiral media. Trans. Amer. Math. Soc. 352, 2000.
- [57] W. McLean. Strongly Elliptic Systems and Boundary Integral Equations. Cambridge University Press, First edition, 2000.
- [58] R. Metzler and J. Klafter. The random walk's guide to anomalous diffusion: a fractional dynamics approach. *Physics Reports*, 339(1):1–77, 2000.
- [59] D. Mitrea. Distributions, Partial Differential Equations, and Harmonic Analysis. Springer, New York, First edition, 2013.
- [60] G. Nakamura, Z. Sun, and G. Uhlmann. Global identifiability for an inverse problem for the Schrödinger equation in a magnetic field. *Matematische Annalen*, 303(1):377-388, 1995.

- [61] G. Nakamura and T. Tsuchida. Uniqueness For An Inverse Boundary Value Problem For Dirac Operators. Communications in Partial Differential Equations, 25(7-8):557–577, 1999.
- [62] G. Nakamura and G. Uhlmann. Global uniqueness for an inverse boundary problem arising in elasticity. Invent. Math., 118, 1994.
- [63] F. Natterer. The Mathematics of Computerized Tomography. SIAM, Philadelphia, 2001. Reprint.
- [64] T. Ozawa. On critical cases of Sobolev inequalities. Hokkaido University, series 154, 1992.
- [65] E. Quinto. Singularities of the X-Ray Transform and Limited Data Tomography in R<sup>2</sup> and R<sup>3</sup>. SIAM Journal on Mathematical Analysis, 24(5):1215–1225, 1993.
- [66] E. Quinto. Artifacts and Visible Singularities in Limited Data X-Ray Tomography. Sensing and Imaging, 18, 2017.
- [67] J. Railo. Fourier Analysis of Periodic Radon Transforms. Journal of Fourier Analysis and Applications, 26(4):64, 2020.
- [68] A. G. Ramm and A. I. Katsevich. The Radon Transform and Local Tomography. CRC Press, Boca Raton, First edition, 1996.
- [69] T. Reichelt. A comparison theorem between Radon and Fourier-Laplace transforms for D-modules. Ann. Inst. Fourier (Grenoble), 65(4):1577–1616, 2015.
- [70] M. Riesz. Intégrales de Riemann-Liouville et potentiels. Acta Sci. Math. Szeged, 9(1-1):1–42, 1938.
- [71] X. Ros-Oton. Nonlocal elliptic equations in bounded domains: a survey. Publicacions Matemátiques, 60:3 26, 2015.
- [72] A. Rüland. Unique continuation for fractional Schrödinger equations with rough potentials. Comm. Partial Differential Equations, 40(1):77–114, 2015.
- [73] A. Rüland and M. Salo. Exponential instability in the fractional Calderón problem. *Inverse Problems*, 34(4):045003, 21, 2018.
- [74] A. Rüland and M. Salo. Quantitative approximation properties for the fractional heat equation. Mathematical Control & Related Fields, pages 233–249, 2019.
- [75] A. Rüland and M. Salo. The fractional Calderón problem: Low regularity and stability. Nonlinear Analysis, 2019.
- [76] M. Salo. Recovering first order terms from boundary measurements. J. Phys.: Conf. Ser., 73, 2007.
- [77] M. Salo. Calderón problem. 2008. Lecture notes.
- [78] M. Salo. Fourier analysis and distribution theory. 2013. Lecture notes.
- [79] M. Salo. The fractional Calderón problem. Journées équations aux dérivées partielles, Exp. No.(7), 2017.
- [80] L. Silvestre. Regularity of the obstacle problem for a fractional power of the Laplace operator. Communications on Pure and Applied Mathematics, 60:67 – 112, 2007.
- [81] P. Stefanov and G. Uhlmann. Microlocal Analysis and Integral Geometry (working title). 2018. Draft version.
- [82] F. Trèves. Topological Vector Spaces, Distributions and Kernels. Academic Press, First edition, 1967.
- [83] G. Uhlmann. Inverse problems: seeing the unseen. Bulletin of Mathematical Sciences, 4(2):209–279, 2014.
- [84] L. Xiaojun. A Note On Fractional Order Poincarés Inequalities. 2012.
- [85] J. Yang, H. Yu, M. Jiang, and G. Wang. High-order total variation minimization for interior tomography. *Inverse Problems*, 26(3):035013, 2010.
- [86] R. Yang. On higher order extensions for the fractional Laplacian. 2013. arXiv:1302.4413.
- [87] Y. Ye, H. Yu, and G. Wang. Exact Interior Reconstruction from Truncated Limited-Angle Projection Data. International Journal of Biomedical Imaging, vol. 2008, 2008.
- [88] H. Yu and G. Wang. Compressed sensing based interior tomography. Physics in Medicine and Biology, 54(9):2791–2805, 2009.

DEPARTMENT OF MATHEMATICS AND STATISTICS, UNIVERSITY OF JYVÄSKYLÄ, JYVÄSKYLÄ, FINLAND *Email address*: giovanni.g.covi@jyu.fi

Email address: kematamo@student.jyu.fi

Seminar for Applied Mathematics, Department of Mathematics, ETH Zurich, Zürich, Switzer-LAND

Email address: jesse.railo@math.ethz.ch
[C]

# X-ray Tomography of One-forms with Partial Data

J. Ilmavirta and K. Mönkkönen

First published in SIAM Journal on Mathematical Analysis 53, no. 3 (2021) https://doi.org/10.1137/20M1344779

https://arxiv.org/abs/2006.05790

Copyright © Society for Industrial and Applied Mathematics

# X-RAY TOMOGRAPHY OF ONE-FORMS WITH PARTIAL DATA

JOONAS ILMAVIRTA AND KEIJO MÖNKKÖNEN

ABSTRACT. If the integrals of a one-form over all lines meeting a small open set vanish and the form is closed in this set, then the one-form is exact in the whole Euclidean space. We obtain a unique continuation result for the normal operator of the X-ray transform of one-forms, and this leads to one of our two proofs of the partial data result. Our proofs apply to compactly supported covector-valued distributions.

#### 1. INTRODUCTION

Let f be a one-form on  $\mathbb{R}^n$  where  $n \ge 2$ . We define the X-ray transform (also known as the Doppler transform in this case) of f by the formula

(1) 
$$X_1 f(\gamma) = \int_{\gamma} f$$

where  $\gamma$  is a line in  $\mathbb{R}^n$ . We freely identify one-forms with vector fields, so the differential of a scalar field corresponds to its gradient. We are interested in the problem of reconstructing f from  $X_1 f$ . One-forms of the form  $f = d\phi$ where  $\phi$  goes to zero at infinity are always in the kernel of  $X_1$ . Thus one can only try to recover the solenoidal part  $f^s$  of the decomposition f = $f^s + d\phi$  from the data  $X_1 f$ . The transform  $X_1$  is known to be solenoidally injective [31, 40], i.e.  $X_1 f = 0$  implies  $f = d\phi$  for some scalar function  $\phi$ . We study whether this implication holds in the whole space also in the case where we know  $X_1 f$  only for a subset of lines.

We consider the following partial data problem for  $X_1$ . Let  $V \subset \mathbb{R}^n$  be a nonempty open set. Assume that we know  $df|_V$  and  $X_1f$  on all lines intersecting V, where df is the exterior derivative or the curl of the oneform f. Can we determine the solenoidal part  $f^s$  – find f modulo potential fields – from this data? We will study the uniqueness of the partial data problem: If  $df|_V = 0$  and  $X_1f = 0$  on all lines intersecting V, does it follow that  $f^s = 0$ ?

The partial data problem for  $X_1$  can be reduced to the following unique continuation problem for the normal operator  $N_1 = X_1^* X_1$ : if  $df|_V = 0$  and

Date: March 9, 2021.

<sup>2010</sup> Mathematics Subject Classification. 46F12, 44A12, 58A10.

Key words and phrases. Inverse problems, X-ray tomography, vector field tomography, normal operator, unique continuation.

Unit of Computing Sciences, Tampere University, Kalevantie 4, FI-33014 Tampere University, Finland; joonas.ilmavirta@tuni.fi.

Department of Mathematics and Statistics, University of Jyväskylä, P.O. Box 35 (MaD) FI-40014 University of Jyväskylä, Finland; kematamo@student.jyu.fi.

 $N_1f|_V = 0$ , does it follow that  $f^s = 0$ ? We prove that such unique continuation property holds for compactly supported covector-valued distributions under the weaker assumption that  $N_1f$  vanishes to infinite order at some point in V. The unique continuation of the normal operator implies uniqueness for the partial data problem: The solenoidal part of a one-form f is uniquely determined whenever one knows the curl of the one-form in V and the integrals of f over all lines intersecting V.

For scalar fields the uniqueness of a corresponding partial data problem and the unique continuation of the normal operator were proved in [18]. We generalize the results to one-forms using the results for scalar fields in our proofs. We also obtain partial data results and unique continuation results for the generalized X-ray transform of one-forms  $X_A = X_1 \circ A$  where A is a smooth invertible matrix-valued function. As a special case of this transform we study the transverse ray transform in  $\mathbb{R}^2$ .

We give two alternative proofs for the partial data results. The first one uses the unique continuation of the normal operator while the second one works directly at the level of the X-ray transform and is based on Stokes' theorem.

The X-ray transform of one-forms or vector fields has applications in the determination of velocity fields of moving fluids using acoustic travel time measurements [29] or Doppler backscattering measurements [30]. Medical applications include ultrasound imaging of blood flows [20, 21, 42]. The transverse ray transform of one-forms has applications in the temperature measurements of flames [4, 38]. For two-tensors the applications include also diffraction tomography [24], photoelasticity [14] and polarization tomography [40]. For a more comprehensive treatment see the reviews [36, 37, 41] and the references therein.

We will give our main results in section 1.1 and discuss related results in section 1.2. The preliminaries are covered in section 2 and finally the theorems are proven in section 3.

1.1. Main results. Here we give the main results of this paper. The proofs can be found in section 3. First we briefly go through our notation; for more detailed definitions see section 2.

Let  $\mathcal{E}'(\mathbb{R}^n)$  be the space of compactly supported distributions. By  $f \in (\mathcal{E}'(\mathbb{R}^n))^n$  we mean that  $f = (f_1, \ldots, f_n)$  where  $f_i \in \mathcal{E}'(\mathbb{R}^n)$  for all  $i = 1, \ldots, n$ . We call  $(\mathcal{E}'(\mathbb{R}^n))^n$  the space of compactly supported covectorvalued distributions. We denote by  $X_1$  the X-ray transform of one-forms and by  $N_1 = X_1^* X_1$  its normal operator; see equation (20) for an explicit formula.

We say that  $N_1 f$  vanishes to infinite order at  $x_0 \in \mathbb{R}^n$  if it is smooth in a neighborhood of  $x_0$  and  $\partial^{\beta}(N_1 f)_i(x_0) = 0$  for all  $\beta \in \mathbb{N}^n$  and  $i = 1, \ldots, n$ . We denote the exterior derivative of differential forms by d. When acting on scalars, it corresponds to the gradient.

Our first result is a unique continuation property for the normal operator  $N_1$ . The corresponding result for scalar fields and the normal operator  $N_0 = X_0^* X_0$  of the scalar X-ray transform  $X_0$  (see equation (13)) was proven in [18, Theorem 1.1]. **Theorem 1.1.** Let  $f \in (\mathcal{E}'(\mathbb{R}^n))^n$  and  $V \subset \mathbb{R}^n$  some nonempty open set. If  $df|_V = 0$  and  $N_1 f$  vanishes to infinite order at  $x_0 \in V$ , then  $f = d\phi$  for some  $\phi \in \mathcal{E}'(\mathbb{R}^n)$ .

We point out that as df vanishes in V, the distribution  $N_1 f$  is smooth in V by lemma 3.3 and the vanishing condition at a point is well-defined.

Theorem 1.1 is also true under the weaker assumption that  $df|_V = 0$  and  $d(N_1f)$  vanishes to infinite order at  $x_0$  (see the proof in section 3.1). The condition that f is closed in V (i.e.  $df|_V = 0$ ) is satisfied if, for example,  $f|_V = 0$ . When f is solenoidal (i.e.  $\operatorname{div}(f) = 0$ ), theorem 1.1 gives the following unique continuation property: if  $f|_V = N_1f|_V = 0$ , then f = 0.

The next result is stated directly at the level of the X-ray transform. The corresponding problem with full data was solved in [40, Theorem 2.5.1].

**Theorem 1.2.** Let  $f \in (\mathcal{E}'(\mathbb{R}^n))^n$  and  $V \subset \mathbb{R}^n$  some nonempty open set. Assume that  $df|_V = 0$ . Then  $X_1 f$  vanishes on all lines intersecting V if and only if  $f = d\phi$  for some  $\phi \in \mathcal{E}'(\mathbb{R}^n)$ .

**Remark 1.3.** In theorems 1.1 and 1.2 the support of the potential  $\phi$  is contained in the convex hull of spt(f).

**Remark 1.4.** We can combine the partial data result for vector fields (theorem 1.2) with the partial data result for scalar fields (lemma 3.4) to obtain the following partial data result. Let  $F: S\mathbb{R}^n \to \mathbb{R}$  be a function on the sphere bundle  $S\mathbb{R}^n = \mathbb{R}^n \times S^{n-1}$  defined as  $F(x,\xi) = g(x) + f(x) \cdot \xi$  where  $g: \mathbb{R}^n \to \mathbb{R}$  is a function on  $\mathbb{R}^n$  and f is a one-form on  $\mathbb{R}^n$ . We define the X-ray transform  $X_{S\mathbb{R}^n}$  of F as

(2) 
$$X_{S\mathbb{R}^n}F(\gamma) = \int_{\mathbb{R}} F(\gamma(t), \dot{\gamma}(t)) dt = X_0 g(\gamma) + X_1 f(\gamma)$$

where  $\gamma$  is an oriented line in  $\mathbb{R}^n$  and  $X_0$  is the X-ray transform of scalar fields (see section 2.2).

Assume that  $V \subset \mathbb{R}^n$  is a nonempty open set such that  $g|_V = df|_V = 0$ and  $X_{S\mathbb{R}^n}F(\gamma) = 0$  on all lines  $\gamma$  intersecting V. Denote by  $\overline{\gamma}$  the reversed line. Since  $X_{0g}(\overline{\gamma}) = X_{0g}(\gamma)$  and  $X_{1}f(\overline{\gamma}) = -X_{1}f(\gamma)$  we obtain  $X_{0g}(\gamma) = \frac{1}{2}(X_{S\mathbb{R}^n}F(\gamma) + X_{S\mathbb{R}^n}F(\overline{\gamma}))$  and  $X_{1}f(\gamma) = \frac{1}{2}(X_{S\mathbb{R}^n}F(\gamma) - X_{S\mathbb{R}^n}F(\overline{\gamma}))$ . Hence the partial data problem for  $X_{S\mathbb{R}^n}F$  decouples to separate partial data problems for  $X_{0g}$  and  $X_{1}f$ . Using theorem 1.2 and lemma 3.4 one obtains that g = 0 and  $f = d\phi$  for some scalar field  $\phi$ . This means that  $F = d\phi$ , i.e.  $F(x,\xi) = d\phi(x) \cdot \xi$ . See [3, 34] for similar results in the case of full data.

One can view theorems 1.1 and 1.2 in terms of the global solenoidal decomposition  $f = f^{s} + d\phi$  (see section 2.1 and equation (5)). The conclusion  $f = d\phi$  for some  $\phi \in \mathcal{E}'(\mathbb{R}^n)$  is equivalent to  $f^{s} = 0$ .

From theorem 1.2 we obtain the following local partial data result in a bounded domain  $\Omega \subset \mathbb{R}^n$ . The X-ray transform of  $f \in (L^2(\Omega))^n$  is defined to be  $X_1 f := X_1 \tilde{f}$  where  $\tilde{f}$  is the zero extension of f to  $\mathbb{R}^n$ .

**Theorem 1.5.** Let  $f \in (L^2(\Omega)))^n$  where  $\Omega \subset \mathbb{R}^n$  is a bounded and smooth convex domain and let  $V \subset \Omega$  be some nonempty open set. Assume that  $df|_V = 0$ . Then  $X_1 f = 0$  on all lines intersecting V if and only if  $f = d\phi$ for some  $\phi \in H^1_0(\Omega)$ . In terms of the local solenoidal decomposition  $f = f_{\Omega}^{s} + d\phi_{\Omega}$  (see section 2.1 and equation (6)) the conclusion  $f = d\phi$  for some  $\phi \in H_{0}^{1}(\Omega)$  is equivalent to that  $f_{\Omega}^{s} = 0$ .

From theorem 1.1 we also obtain the following unique continuation and partial data results for the transform  $X_A = X_1 \circ A$  where A = A(x) is smooth invertible matrix field. We denote by  $N_A = A^T \circ N_1 \circ A$  the normal operator of  $X_A$ . When B is the constant matrix field  $B(v_1, v_2) = (v_2, -v_1)$ where  $(v_1, v_2) \in \mathbb{R}^2$  we write  $X_B = X_{\perp}$  and call  $X_{\perp}$  the transverse ray transform.

**Corollary 1.6.** Let  $f \in (\mathcal{E}'(\mathbb{R}^n))^n$  and  $V \subset \mathbb{R}^n$  some nonempty open set. If  $d(Af)|_V = 0$  and  $N_A f|_V = 0$ , then  $f = A^{-1}(d\psi)$  for some  $\psi \in \mathcal{E}'(\mathbb{R}^n)$ .

**Corollary 1.7.** Let  $f \in (\mathcal{E}'(\mathbb{R}^n))^n$  and  $V \subset \mathbb{R}^n$  some nonempty open set. Assume that  $d(Af)|_V = 0$ . Then  $X_A f$  vanishes on all lines intersecting V if and only if  $f = A^{-1}(d\psi)$  for some  $\psi \in \mathcal{E}'(\mathbb{R}^n)$ .

In corollaries 1.6 and 1.7 the distribution  $\psi \in \mathcal{E}'(\mathbb{R}^n)$  is the potential part of the solenoidal decomposition of  $Af \in (\mathcal{E}'(\mathbb{R}^n))^n$  and  $\operatorname{spt}(\psi)$  is contained in the convex hull of  $\operatorname{spt}(f)$ . As a special case of the transform  $X_A$  we obtain the next partial data result for the transverse ray transform  $X_{\perp}$  which is similar to the full data result in [4, 11] (see also [2]).

**Corollary 1.8.** Let  $f \in (\mathcal{E}'(\mathbb{R}^2))^2$  and  $V \subset \mathbb{R}^2$  some nonempty open set. Assume that  $\operatorname{div}(f)|_V = 0$ . Then  $X_{\perp}f$  vanishes on all lines intersecting V if and only if  $\operatorname{div}(f) = 0$ .

In particular, if  $df|_V = div(f)|_V = 0$  and both  $X_1f$  and  $X_{\perp}f$  vanish on all lines intersecting V, then f = 0.

Alternatively, one can conclude in the first claim of corollary 1.8 that  $f = \operatorname{curl}(\psi)$  for some  $\psi \in \mathcal{E}'(\mathbb{R}^n)$  where  $\operatorname{curl}(\psi) = (\partial_2 \psi, -\partial_1 \psi)$ . In terms of the global solenoidal decomposition this is equivalent to that  $f = f^s$ . Also in the latter claim it is enough to know the partial data of  $X_1 f$  for  $V \subset \mathbb{R}^2$  and the partial data of  $X_{\perp} f$  for  $W \subset \mathbb{R}^2$  where V and W can be disjoint.

**Remark 1.9.** Some of the results above can be slightly generalized. Using the same proof as in theorem 1.5 one can show that corollaries 1.7 and 1.8 hold also in the local case when  $f \in (L^2(B))^n$ . Also in corollary 1.6 one can replace the condition  $N_A f|_V = 0$  with the requirement that  $N_A f$  vanishes to infinite order at  $x_0 \in V$  when A is a constant matrix field. Especially, this holds for the normal operator of the transverse ray transform. One can also see from theorem 1.2 and corollary 1.8 that the X-ray transform and the transverse ray transform provide complementary information about the one-form in  $\mathbb{R}^2$ .

We note that if A = A(x) is not invertible for all  $x \in \mathbb{R}^n$ , we can still conclude in corollary 1.7 that  $Af = d\psi$  for some potential  $\psi \in \mathcal{E}'(\mathbb{R}^n)$ . Thus we obtain the "pointwise projection" Af modulo potentials from the local data for  $X_A f$ . We also remark that in all of our results which consider the X-ray transform in  $\mathbb{R}^n$  we could replace the assumption of compact support with rapid decay at infinity. If all the derivatives of the matrix field  $A = A_{ij}(x)$  grow at most polynomially, then the results are true for one-forms whose components are Schwartz functions. This follows since the corresponding partial data result for scalar fields holds for Schwartz functions [18] and our method of proof is based on reducing the problem of one-forms to the problem of scalar fields.

1.2. Related results. Similar partial data results as in theorems 1.2 and 1.5 are previously known for scalar fields. If one knows the values of the scalar function f in an open set V, then one can uniquely reconstruct f from its local X-ray data [5, 18, 23]. In  $\mathbb{R}^2$  uniqueness is also obtained under weaker assumptions: if f is piecewise constant, piecewise polynomial or analytic in V, then one can recover f uniquely from its integrals over the lines going through V [22, 23, 48]. A complementary partial data result is the Helgason support theorem [15]. According to Helgason's theorem, if  $f|_C = 0$  and the integrals of f vanish on all lines not intersecting a compact and convex set C, then f = 0.

The normal operator of the X-ray transform of scalar fields admits a similar unique continuation property as in theorem 1.1. If f is a function which satisfies  $f|_V = 0$  and  $N_0 f$  vanishes to infinite order at some point in V, then f = 0 [18]. This is a special case of a more general unique continuation result for Riesz potentials [18] (see equation (13)). Unique continuation of Riesz potentials is related to unique continuation of fractional Laplacians [6, 12, 18] (see also equation (14)).

Unique reconstruction of the solenoidal part of a one-form or vector field with full data is known in  $\mathbb{R}^n$  [21, 29, 42, 43] and on compact simple Riemannian manifolds with boundary [17, 31]. In  $\mathbb{R}^n$  uniqueness holds for compactly supported covector-valued distributions as well [40]. Some partial data results are known for one-forms. The solenoidal part can be reconstructed by knowing  $X_1 f$  on all lines parallel to a finite set of planes [21, 35, 39]. When  $n \ge 3$ , one can locally recover one-forms up to potential fields near a strictly convex boundary point [44], and the solenoidal part can be determined from the knowledge of  $X_1 f$  on all lines intersecting a certain type of curve [47] (see also [10]). One can also obtain information about the singularities of the curl of a compactly supported covector-valued distribution from its X-ray data on lines intersecting a fixed curve [33]. There is a Helgason-type support theorem for the X-ray transform of oneforms which is in a sense complementary to our result. If f integrates to zero over all lines not intersecting a compact and convex set C, then df = 0 in the complement of C [43, Theorem 7.5]. If we further assume that  $df|_C = 0$ , then the one-form f is closed in the whole space which implies that f is exact and the solenoidal part of f vanishes. See also the discussion after the alternative proof in section 3.2.

The transverse ray transform has been studied earlier with full data in  $\mathbb{R}^2$  [4, 11, 28] and also on Riemannian manifolds [19, 40] (see also [1] for a support theorem). The transverse ray transform is a special case of a more general mixed ray transform [7, 8, 11, 40]. In higher dimensions the transverse ray transform is related to the normal Radon transform [41, 45]. In  $\mathbb{R}^2$  and on certain Riemannian manifolds the knowledge of  $X_1 f$  and  $X_{\perp} f$ fully determines the one-form [4, 11, 19]. By theorem 1.2 and corollary 1.8 this is true in  $\mathbb{R}^2$  also in the case of partial data. In higher dimensions f is determined by  $X_1 f$  and the normal Radon transform of f [45]. A similar transform to  $X_A$  was studied in [19, 32]. Recently in [2] the authors studied the so-called V-line transform of vector fields which is a generalization of the X-ray transform to V-shaped "lines" which consist of one vertex and two rays (half-lines).

Acknowledgements. J.I. was supported by the Academy of Finland (grants 332890 and 336254). K.M. was supported by Academy of Finland (Centre of Excellence in Inverse Modelling and Imaging, grant numbers 284715 and 309963). We are grateful to Lauri Oksanen for discussions. The authors want to thank the anonymous referees for their valuable comments.

#### 2. Preliminaries

In this section we give a brief introduction to the theory of X-ray tomography of scalar fields and one-forms in  $\mathbb{R}^n$ . We also define the generalized X-ray transform of one-forms. First we recall the definition and solenoidal decomposition of covector-valued distributions. We mainly follow the conventions of [9, 16, 27, 40, 43, 46] and refer the reader to them for further details.

2.1. Covector-valued distributions and solenoidal decomposition. We denote by  $\mathcal{D}(\mathbb{R}^n)$  the space of compactly supported smooth functions, by  $\mathscr{S}(\mathbb{R}^n)$  the space of rapidly decreasing smooth functions (Schwartz space) and by  $\mathcal{E}(\mathbb{R}^n)$  the space of smooth functions. All spaces are equipped with their standard topologies. The spaces  $\mathcal{D}'(\mathbb{R}^n)$ ,  $\mathscr{S}'(\mathbb{R}^n)$  and  $\mathcal{E}'(\mathbb{R}^n)$  are the corresponding topological duals. Elements of  $\mathcal{D}'(\mathbb{R}^n)$  are called distributions and  $\mathcal{E}'(\mathbb{R}^n)$  can be seen as the space of compactly supported distributions. We have the continuous inclusions  $\mathcal{E}'(\mathbb{R}^n) \subset \mathscr{S}'(\mathbb{R}^n) \subset \mathcal{D}'(\mathbb{R}^n)$ . We write the dual pairing as  $\langle f, \varphi \rangle$  when f is a distribution and  $\varphi$  is a test function.

We define the vector-valued test function space  $(\mathcal{D}(\mathbb{R}^n))^n$  such that  $\varphi \in (\mathcal{D}(\mathbb{R}^n))^n$  if and only if  $\varphi = (\varphi_1, \ldots, \varphi_n)$  and  $\varphi_i \in \mathcal{D}(\mathbb{R}^n)$  for all  $i = 1, \ldots, n$ . The topology of the space  $(\mathcal{D}(\mathbb{R}^n))^n$  is defined as follows: a sequence  $\varphi_k$  converges to zero in  $(\mathcal{D}(\mathbb{R}^n))^n$  if and only if  $(\varphi_k)_i$  converges to zero in  $\mathcal{D}(\mathbb{R}^n)$  for all  $i = 1, \ldots, n$ . We then define the space of covector-valued distributions  $(\mathcal{D}'(\mathbb{R}^n))^n$  so that  $f \in (\mathcal{D}'(\mathbb{R}^n))^n$  if and only if  $f = (f_1, \ldots, f_n)$  and  $f_i \in \mathcal{D}'(\mathbb{R}^n)$  for all  $i = 1, \ldots, n$ . The duality pairing of  $f \in (\mathcal{D}'(\mathbb{R}^n))^n$  and  $\varphi \in (\mathcal{D}(\mathbb{R}^n))^n$  becomes

(3) 
$$\langle f, \varphi \rangle = \sum_{i=1}^{n} \langle f_i, \varphi_i \rangle$$

The spaces  $(\mathcal{E}(\mathbb{R}^n))^n$ ,  $(\mathscr{S}(\mathbb{R}^n))^n$ ,  $(\mathcal{E}'(\mathbb{R}^n))^n$  and  $(\mathscr{S}'(\mathbb{R}^n))^n$  are defined in a similar way and we call  $(\mathcal{E}'(\mathbb{R}^n))^n$  the space of compactly supported covectorvalued distributions. Covector-valued distributions are a special case of currents which are continuous linear functionals in the space of differential forms [9, Section III]. The components of the exterior derivative or the curl of a one-form or covector-valued distribution are

(4) 
$$(\mathrm{d}f)_{ij} = \partial_i f_j - \partial_j f_i.$$

One can split certain covector-valued distributions into a divergence-free part and a potential part. If  $f \in (\mathcal{E}'(\mathbb{R}^n))^n$ , then we have the unique decomposition [40]

(5) 
$$f = f^{s} + d\phi, \quad \operatorname{div}(f^{s}) = 0$$

where  $\phi \in \mathscr{S}'(\mathbb{R}^n)$  and  $f^{\mathrm{s}} \in (\mathscr{S}'(\mathbb{R}^n))^n$  are smooth outside  $\operatorname{spt}(f)$  and go to zero at infinity. Here  $\phi$  is defined so that it solves the equation  $\Delta \phi = \operatorname{div}(f)$ in the sense of distributions and  $f^{\mathrm{s}} = f - \mathrm{d}\phi$ . The decomposition (5) is known as solenoidal decomposition or Helmholtz decomposition and it holds also for  $f \in (\mathscr{S}(\mathbb{R}^n))^n$  [40]. We call f solenoidal if  $\operatorname{div}(f) = 0$ . For the decomposition (5) this means that  $f = f^{\mathrm{s}}$ .

If f is supported in a fixed set, we can do the decomposition locally in that set. If  $\Omega \subset \mathbb{R}^n$  is a regular enough bounded domain and  $f \in (L^2(\Omega))^n$ , we let  $\phi_{\Omega}$  to be the unique weak solution to the Poisson equation

(6) 
$$\begin{cases} \Delta \phi = \operatorname{div}(f) \text{ in } \Omega\\ \phi \in H_0^1(\Omega). \end{cases}$$

Then we have  $f = f_{\Omega}^{s} + d\phi_{\Omega}$  where  $f_{\Omega}^{s} = f - d\phi_{\Omega} \in (L^{2}(\Omega))^{n}$  and  $\operatorname{div}(f_{\Omega}^{s}) = 0$ . If  $f \in (C^{1,\alpha}(\overline{\Omega}))^{n}$  for some  $0 < \alpha < 1$ , then there is unique classical solution  $\phi_{\Omega} \in C^{2,\alpha}(\overline{\Omega})$  to the boundary value problem (6) and the solenoidal decomposition holds pointwise [13].

2.2. The X-ray transform of scalar fields. Let  $\Gamma$  be the set of all oriented lines in  $\mathbb{R}^n$ . The X-ray transform of a function f is defined as

(7) 
$$X_0 f(\gamma) = \int_{\gamma} f \mathrm{d}s, \quad \gamma \in \Gamma$$

whenever the integrals exist. The set  $\Gamma$  can be parameterized as

(8) 
$$\Gamma = \{ (z, \theta) : \theta \in S^{n-1}, \ z \in \theta^{\perp} \}.$$

Then the X-ray transform becomes

(9) 
$$X_0 f(z,\theta) = \int_{\mathbb{R}} f(z+t\theta) dt$$

and it is a continuous map  $X_0: \mathcal{D}(\mathbb{R}^n) \to \mathcal{D}(\Gamma)$ . One can define the adjoint using the formula

(10) 
$$X_0^*\psi(x) = \int_{S^{n-1}} \psi(x - (x \cdot \theta)\theta, \theta) \mathrm{d}\theta$$

and it follows that  $X_0^* \colon \mathcal{E}(\Gamma) \to \mathcal{E}(\mathbb{R}^n)$  is continuous. By duality we can define  $X_0 \colon \mathcal{E}'(\mathbb{R}^n) \to \mathcal{E}'(\Gamma)$  and  $X_0^* \colon \mathcal{D}'(\Gamma) \to \mathcal{D}'(\mathbb{R}^n)$  as

(11) 
$$\langle X_0 f, \varphi \rangle = \langle f, X_0^* \varphi \rangle$$

(12) 
$$\langle X_0^*g,\eta\rangle = \langle g,X_0\eta\rangle.$$

The normal operator  $N_0 = X_0^* X_0$  is useful in studying the properties of the X-ray transform since it takes functions on  $\mathbb{R}^n$  to functions on  $\mathbb{R}^n$ . It has an expression

(13) 
$$N_0 f(x) = 2 \int_{\mathbb{R}^n} \frac{f(y)}{|x-y|^{n-1}} \mathrm{d}y = 2(f * |\cdot|^{1-n})(x)$$

for continuous functions f decreasing rapidly enough at infinity. By duality the formula  $N_0 f = 2(f * |\cdot|^{1-n})$  holds also for compactly supported distributions and the normal operator becomes a map  $N_0: \mathcal{E}'(\mathbb{R}^n) \to \mathcal{D}'(\mathbb{R}^n)$ . One can invert f from its X-ray transform using the normal operator by

(14) 
$$f = c_{0,n} (-\Delta)^{1/2} N_0 f,$$

where  $c_{0,n} = (2\pi |S^{n-2}|)^{-1}$  is a constant depending on the dimension and  $(-\Delta)^{1/2}$  is the fractional Laplacian of order 1/2. The inversion formula (14) holds for  $f \in \mathcal{E}'(\mathbb{R}^n)$  and for continuous functions f decreasing rapidly enough at infinity.

2.3. The X-ray transform of one-forms. Let f be a one-form on  $\mathbb{R}^n$ . We define its X-ray transform as

(15) 
$$X_1 f(\gamma) = \int_{\gamma} f, \quad \gamma \in \Gamma$$

whenever the integrals exist. The formula (15) is understood as the integral of the one-form f over the (oriented) one-dimensional submanifold  $\gamma$ . Using the parametrization (8) for  $\Gamma$  we can write

(16) 
$$X_1 f(z, \theta) = \int_{\mathbb{R}} f(z + t\theta) \cdot \theta dt.$$

It follows that  $X_1: (\mathcal{D}(\mathbb{R}^n))^n \to \mathcal{D}(\Gamma)$  is continuous. The adjoint is defined as

(17) 
$$(X_1^*\psi)_i(x) = \int_{S^{n-1}} \theta_i \psi(x - (x \cdot \theta)\theta, \theta) \mathrm{d}\theta$$

and  $X_1^* \colon \mathcal{E}(\Gamma) \to (\mathcal{E}(\mathbb{R}^n))^n$  is also continuous. Thus we can define  $X_1 \colon (\mathcal{E}'(\mathbb{R}^n))^n \to \mathcal{E}'(\Gamma)$  and  $X_1^* \colon \mathcal{D}'(\Gamma) \to (\mathcal{D}'(\mathbb{R}^n))^n$  as

(18) 
$$\langle X_1 f, \varphi \rangle = \langle f, X_1^* \varphi \rangle$$

(19) 
$$\langle X_1^*g,\eta\rangle = \langle g,X_1\eta\rangle$$

If  $f \in (L^p(\Omega))^n$  where  $\Omega \subset \mathbb{R}^n$  is a bounded domain and  $p \ge 1$ , we define its X-ray transform as  $X_1 f := X_1 \tilde{f}$  where  $\tilde{f} \in (\mathcal{E}'(\mathbb{R}^n))^n$  is the zero extension of f.

Like in the scalar case we define the normal operator  $N_1 = X_1^* X_1$  and it satisfies the formula

(20) 
$$(N_1 f)_i = \sum_{j=1}^n \frac{2x_i x_j}{|x|^{n+1}} * f_j.$$

The normal operator can be extended to a map  $N_1 : (\mathcal{E}'(\mathbb{R}^n))^n \to (\mathcal{D}'(\mathbb{R}^n))^n$ and the formula (20) holds for  $f \in (\mathcal{E}'(\mathbb{R}^n))^n$  and also for continuous oneforms decreasing rapidly enough at infinity. One can invert the solenoidal part of f using the normal operator by

(21) 
$$f^{s} = c_{1,n} (-\Delta)^{1/2} N_1 f,$$

where  $c_{1,n} = |S^n|$  is a constant depending on the dimension and  $(-\Delta)^{1/2}$  operates componentwise. The formula (21) holds for  $f \in (\mathcal{E}'(\mathbb{R}^n))^n$  and also for continuous one-forms decreasing rapidly enough at infinity.

2.4. The generalized X-ray transform of one-forms. Let A = A(x) be a smooth matrix-valued function on  $\mathbb{R}^n$  such that for each  $x \in \mathbb{R}^n$  the matrix A(x) is invertible. We define the transform  $X_A$  of a one-form f as

(22) 
$$X_A f(\gamma) = \int_{-\infty}^{\infty} A(\gamma(t)) f(\gamma(t)) \cdot \dot{\gamma}(t) dt = X_1(Af)(\gamma), \quad \gamma \in \Gamma.$$

Thus  $X_A$  can be seen as the X-ray transform of the "rotated" one-form Af. The transform  $X_A$  can also be defined on compactly supported covectorvalued distributions. We first let  $\langle Af, \varphi \rangle = \langle f, A^T \varphi \rangle$  for  $f \in (\mathcal{D}'(\mathbb{R}^n))^n$  and a test function  $\varphi$  where  $A^T$  is the pointwise transpose of A and  $(A^T \varphi)(x) =$  $A^T(x)\varphi(x)$ . Then clearly A is a map  $A: (\mathcal{E}'(\mathbb{R}^n))^n \to (\mathcal{E}'(\mathbb{R}^n))^n$ . Therefore we can define  $X_A: (\mathcal{E}'(\mathbb{R}^n))^n \to \mathcal{D}'(\Gamma)$  as  $X_A f = X_1(Af)$ . One easily sees that the adjoint is  $X_A^* = A^T \circ X_1^*$  and the normal operator becomes  $N_A =$  $A^T \circ N_1 \circ A$ . By the discussion above the normal operator can be extended to a map  $N_A: (\mathcal{E}'(\mathbb{R}^n))^n \to (\mathcal{D}'(\mathbb{R}^n))^n$ .

Let B be the constant matrix field on  $\mathbb{R}^2$  defined as  $B(v_1e_1 + v_2e_2) = v_2e_1 - v_1e_2$  where  $\{e_1, e_2\}$  is any orthonormal basis of  $\mathbb{R}^2$ . The matrix B corresponds to a clockwise rotation by 90 degrees. We then define the transverse ray transform  $X_{\perp}$  by letting  $X_{\perp} = X_B$ . It follows that the transverse ray transform provides complementary information about the solenoidal decomposition compared to the X-ray transform, i.e.  $X_1$  determines the solenoidal part and  $X_{\perp}$  determines the potential part of a one-form [4, 11] (see also theorem 1.2 and corollary 1.8).

#### 3. Proofs of the main results

We give two alternative proofs for the partial data results. The first proof uses the unique continuation of the normal operator and the second proof works directly at the level of the X-ray transform. Both proofs are based on the corresponding results for scalar fields.

3.1. **Proofs using the unique continuation of the normal operator.** In this section we prove our main results using the unique continuation property of the normal operator. We need the following lemmas in our proofs.

**Lemma 3.1** ([18, Theorem 1.1]). Let  $V \subset \mathbb{R}^n$  be some nonempty open set and  $g \in \mathcal{E}'(\mathbb{R}^n)$ . If  $g|_V = 0$  and  $\partial^{\beta}(N_0g)(x_0) = 0$  for some  $x_0 \in V$  and all  $\beta \in \mathbb{N}^n$ , then g = 0.

**Lemma 3.2** (Poincaré lemma). Let  $g \in (\mathcal{D}'(\mathbb{R}^n))^n$  such that dg = 0. Then there is  $\eta \in \mathcal{D}'(\mathbb{R}^n)$  such that  $d\eta = g$ . If  $g \in (\mathcal{E}'(\mathbb{R}^n))^n$ , then  $\eta \in \mathcal{E}'(\mathbb{R}^n)$ .

The proof of lemma 3.2 can be found in [16, 25]. We first prove the unique continuation result for the normal operator. The proof is based on the fact that we can reduce the unique continuation problem of  $N_1$  to a unique continuation problem of  $N_0$  acting on the components of df.

The assumptions of theorem 1.1 come in two stages. We first assume that  $df|_V = 0$ . To make sense of the next assumption that  $N_1 f$  vanishes at  $x_0$  to infinite order, we need to ensure that it is smooth near this point. This is given by the next lemma.

**Lemma 3.3.** Let  $V \subset \mathbb{R}^n$  be an open set and  $f \in (\mathcal{E}'(\mathbb{R}^n))^n$ . If  $df|_V = 0$ , then  $N_1 f|_V$  is smooth.

*Proof.* Take any  $x_0 \in V$  and a small open ball B centered at it and contained in V. As  $df|_B = 0$ , the Poincaré lemma applied in the ball B (lemma 3.2 is applicable because B is diffeomorphic to  $\mathbb{R}^n$ ) gives  $f|_B = dh$  for some  $h \in \mathcal{D}'(B)$ . Let  $B' \subset B$  be a smaller ball with the same center, and let  $\chi \in \mathcal{D}(B)$  be a bump function so that  $\chi|_{B'} \equiv 1$ . If we let  $h' = \chi h \in \mathcal{E}'(\mathbb{R}^n)$ , then f = dh' + g, where  $g \in (\mathcal{E}'(\mathbb{R}^n))^n$  with  $g|_{B'} = 0$ .

As  $X_1(dh') = 0$  (cf. (27)), we have  $N_1f = N_1g$ . Because  $g|_{B'} = 0$ , it follows from properties of convolutions that  $N_1f$  is smooth in B'. Now that  $N_1f$  is smooth in a neighborhood of any point in V, the claim follows.

*Proof of theorem 1.1.* The normal operator has an expression

(23) 
$$(N_1f)_i = \sum_{j=1}^n \frac{2x_i x_j}{|x|^{n+1}} * f_j.$$

We can write the kernel as

(24) 
$$\frac{2x_i x_j}{|x|^{n+1}} = \frac{2}{n-1} \left( \delta_{ij} |x|^{1-n} - \partial_i (x_j |x|^{1-n}) \right)$$

and we obtain

(25) 
$$(N_1f)_i = \frac{2}{n-1} \left( \frac{1}{2} N_0 f_i - \sum_{j=1}^n x_j |x|^{1-n} * \partial_i f_j \right).$$

We can calculate that

(26) 
$$\partial_k (N_1 f)_i - \partial_i (N_1 f)_k = \frac{1}{n-1} N_0 (\partial_k f_i - \partial_i f_k).$$

This can be interpreted as  $d(N_1f) = (n-1)^{-1}N_0(df)$ , where the scalar normal operator  $N_0$  acts on the 2-form df componentwise to produce another 2-form. The normal operator commutes with the exterior derivative in this sense.

Since  $N_1 f$  vanishes to infinite order at  $x_0 \in V$  also  $N_0(\partial_k f_i - \partial_i f_k)$  vanishes to infinite order at  $x_0$ . Using lemma 3.1 we obtain df = 0. By lemma 3.2 there is  $\phi \in \mathcal{E}'(\mathbb{R}^n)$  such that  $d\phi = f$ . This concludes the proof.  $\Box$ 

Lemma 3.1 is false if no restrictions are imposed on  $g|_V$  [23, 27], and the assumption  $g|_V = 0$  is the most convenient. Consequently, the assumption  $df|_V = 0$  in theorem 1.1 is important. This condition is invariant under gauge transformations of the field f.

If  $f^{s}|_{V} = N_{1}f|_{V} = 0$ , then one can alternatively use the unique continuation of the fractional Laplacian  $(-\Delta)^{s}$ ,  $s \in (0, 1)$ , to prove the unique continuation of the normal operator [12]. This follows since  $(-\Delta)^{1/2}f^{s} =$  $c_{1,n}(-\Delta)N_{1}f$  where  $f^{s} \in (H^{r}(\mathbb{R}^{n}))^{n}$  for some  $r \in \mathbb{R}$  when  $f \in (\mathcal{E}'(\mathbb{R}^{n}))^{n}$ . One can also make use of the fact that  $(-\Delta)^{-1/2}$  is a Riesz potential and use its unique continuation properties [18] (see equation (21)).

The rest of the results follow easily from theorem 1.1.

Proof of theorem 1.2. Let  $f = d\phi$  where  $\phi \in \mathcal{E}'(\mathbb{R}^n)$ . Then  $d\phi \in (\mathcal{E}'(\mathbb{R}^n))^n$ and using the definition of the X-ray transform on distributions we obtain

(27) 
$$\langle X_1(\mathrm{d}\phi),\varphi\rangle = \langle \mathrm{d}\phi, X_1^*\varphi\rangle = \langle \phi, \mathrm{div}(X_1^*\varphi)\rangle = 0.$$

Here we used the fact that  $\operatorname{div}(X_1^*\varphi) = 0$  which follows from a straightforward computation. This shows that  $X_1f = X_1(\mathrm{d}\phi) = 0$ , and especially  $X_1f$  vanishes on all lines intersecting V. Assume then that  $\mathrm{d}f|_V = 0$ . Since  $X_1f = 0$  on all lines intersecting V we obtain  $N_1f|_V = 0$ . Theorem 1.1 implies that  $f = \mathrm{d}\phi$  for some  $\phi \in \mathcal{E}'(\mathbb{R}^n)$ . This concludes the proof.  $\Box$ 

Proof of theorem 1.5. If  $f = d\phi$  where  $\phi \in H_0^1(\Omega)$ , then using the same argument as in the proof of theorem 1.2 and the fact that  $H_0^1(\Omega) \subset \mathcal{E}'(\mathbb{R}^n)$ in the sense of zero extension we obtain that  $X_1 f = 0$ , and especially  $X_1 f$ vanishes on all lines intersecting V. Then assume that  $df|_V = 0$  and  $X_1 f =$ 0 on all lines intersecting V. Let  $\tilde{f} \in (\mathcal{E}'(\mathbb{R}^n))^n$  be the zero extension of f. The assumptions imply that  $d\tilde{f}|_V = 0$  and  $X_1\tilde{f} = 0$  on all lines intersecting V. Theorem 1.2 implies that  $\tilde{f} = d\phi$  for some  $\phi \in \mathcal{E}'(\mathbb{R}^n)$ . Since  $\Delta \phi = \operatorname{div}(\tilde{f}) \in H^{-1}(\mathbb{R}^n)$  we have  $\phi \in H^1(\mathbb{R}^n)$  by elliptic regularity. On the other hand,  $\operatorname{spt}(\phi) \subset \overline{\Omega}$  and hence  $\phi \in H_0^1(\Omega)$  [26, Theorem 3.33]. The claim follows from the fact that  $d\phi = \tilde{f} = f$  in  $\Omega$ .

Proof of corollary 1.6. We know that the normal operator is  $N_A = A^T \circ N_1 \circ A$ . The assumptions imply that  $d(Af)|_V = N_1(Af)|_V = 0$ . By theorem 1.1 we obtain that  $Af = d\psi$  for some  $\psi \in \mathcal{E}'(\mathbb{R}^n)$ . This gives the claim.  $\Box$ 

Proof of corollary 1.7. The claim follows directly from corollary 1.6 and from the fact that  $X_A = X_1 \circ A$ .

In theorems 1.1 and 1.2 one has  $\operatorname{spt}(\phi) \subset \operatorname{Conv}(\operatorname{spt}(f))$  where  $\operatorname{Conv}(\operatorname{spt}(f))$  is the convex hull of  $\operatorname{spt}(f)$ . This follows from the fact that  $\phi$  has compact support and  $d\phi$  vanishes in the connected set  $\operatorname{Conv}(\operatorname{spt}(f))^c$ . This was pointed out in remark 1.3.

In corollaries 1.6 and 1.7 one also has  $\operatorname{spt}(\psi) \subset \operatorname{Conv}(\operatorname{spt}(f))$ . This holds since  $d\psi$  vanishes in the connected set  $\operatorname{Conv}(\operatorname{spt}(Af))^c$  and  $\operatorname{spt}(Af) = \operatorname{spt}(f)$ .

Proof of corollary 1.8. Assume first that  $\operatorname{div}(f) = 0$ . Since f is a covectorvalued distribution in  $\mathbb{R}^2$  we can identify  $\mathrm{d}f = \partial_1 f_2 - \partial_2 f_1$ . It follows that  $\mathrm{d}(Bf) = -\operatorname{div}(f) = 0$  and thus  $Bf = \mathrm{d}\eta$  for some  $\eta \in \mathcal{E}'(\mathbb{R}^n)$  by lemma 3.2. Therefore  $X_{\perp}f = X_1(Bf) = X_1(\mathrm{d}\eta) = 0$ . Assume then that  $\operatorname{div}(f)|_V = 0$  and  $X_{\perp}f = 0$  on all lines intersecting V. As above we obtain that  $\mathrm{d}(Bf)|_V = 0$  and  $X_{\perp}f = 0$  on all lines intersecting V. Corollary 1.7 implies that  $f = B^{-1}(\mathrm{d}\psi)$  for some  $\psi \in \mathcal{E}'(\mathbb{R}^n)$ . From this we obtain that  $\operatorname{div}(f) = 0$ .

Assume then that  $df|_V = \operatorname{div}(f)|_V = 0$  and both  $X_1 f$  and  $X_{\perp} f$  vanish on all lines intersecting V. By the discussion above we obtain that  $\operatorname{div}(f) = 0$ . On the other hand, theorem 1.2 implies that  $f = d\phi$  for some  $\phi \in \mathcal{E}'(\mathbb{R}^2)$ . Therefore  $\Delta \phi = 0$  and since  $\phi$  has compact support we must have  $\phi = 0$ , i.e. f = 0. 3.2. **Proofs based on Stokes' theorem.** In this section we give alternative proofs for the partial data results using Stokes' theorem in  $\mathbb{R}^n$ . A similar approach was used in [21, 42] in the case of full data, and also recently in [2] for the generalized V-line transform. We prove the results first for compactly supported smooth one-forms and then use standard mollification argument to prove them for compactly supported covector-valued distributions. We only need to prove theorem 1.2 since the rest of the partial data results follow from it. We will use the following lemma.

**Lemma 3.4** ([18, Theorem 1.2]). Let  $V \subset \mathbb{R}^n$  be some nonempty open set and  $g \in \mathcal{E}'(\mathbb{R}^n)$ . If  $g|_V = 0$  and  $X_0g = 0$  on all lines intersecting V, then g = 0.

Alternative proof of theorem 1.2. By lemma 3.2 it suffices to show that df = 0. Assume first that n = 2 and  $f \in (\mathcal{D}(\mathbb{R}^2))^2$ . Let  $\gamma$  be any (oriented) line going through V and  $\nu$  the counterclockwise rotated normal to  $\gamma$ . We denote by  $\gamma_h = h\nu + \bar{\gamma}$  the reversed parallel line shifted by h > 0 in the direction of  $\nu$  so that  $\gamma_h$  also intersects V. By assumption  $\int_{\gamma} f = \int_{\gamma_h} f = 0$ .

We form a closed loop  $\tilde{\gamma}_h$  enclosing counterclockwise a rectangular region  $R_h$  such that the ends are outside  $\operatorname{spt}(f)$  (see figure 1). When considered as chains, we have  $\tilde{\gamma}_h = \partial R_h$ . As the chains  $\gamma - \gamma_h$  and  $\tilde{\gamma}_h$  differ only outside the support of f, the integrals coincide. By Stokes' theorem

(28) 
$$0 = \int_{\gamma} f - \int_{\gamma_h} f = \int_{\widetilde{\gamma}_h} f = \int_{\partial R_h} f = \int_{R_h} \mathrm{d}f = \int_{R_h} \star \mathrm{d}f \,\mathrm{d}\mu,$$

where  $\star$  is the Hodge star and  $\mu$  is the 2-Hausdorff measure.

We aim to show that the scalar function  $\star \mathrm{d} f$  vanishes. Scaling with h, we find

(29) 
$$0 = \lim_{h \to 0} \frac{1}{h} \int_{R_h} \star \mathrm{d}f \,\mathrm{d}\mu = \int_{\gamma} \star \mathrm{d}f \,\mathrm{d}s.$$

Now that  $\star df|_V = 0$  and  $X_0(\star df)(\gamma) = 0$  for all lines  $\gamma$  meeting V, lemma 3.4 implies that  $\star df = 0$  and thus also df = 0 in the whole plane.

Consider then the case  $n \geq 3$  for a compactly supported smooth oneform f. Let  $P \subset \mathbb{R}^n$  be any two-plane meeting V and  $\iota_P \colon P \to \mathbb{R}^n$  the corresponding inclusion. By the argument above for the two-form  $\iota_P^* f$  in the plane P we have that  $\iota_P^* df = 0$  for all such planes.

Take any point  $z \in \mathbb{R}^n$ . For any plane P through z that intersects V we have  $\iota_P^* df = 0$ . This is an open subset of the Grassmannian of 2-planes through z, so df(z) = 0. As the point z was arbitrary, we have df = 0.

Finally, let  $f \in (\mathcal{E}'(\mathbb{R}^n))^n$  and define  $f_{\epsilon} = f * j_{\epsilon} = (f_1 * j_{\epsilon}, \dots, f_n * j_{\epsilon})$ where  $j_{\epsilon} \in \mathcal{D}(\mathbb{R}^n)$  is the standard mollifier. Then  $f_{\epsilon} \in (\mathcal{D}(\mathbb{R}^n))^n$  and  $\langle X_1(f * j_{\epsilon}), \varphi \rangle = \langle X_1 f, X_0 j_{\epsilon} \circledast \varphi \rangle$  where

(30) 
$$(h \circledast g)(z, \theta) = \int_{\theta^{\perp}} h(z - y, \theta) g(y, \theta) \mathrm{d}y.$$

Hence there is a nonempty open set  $W \subset V$  such that for small  $\epsilon > 0$  we have  $f_{\epsilon}|_{W} = 0$  and  $X_{1}f_{\epsilon} = 0$  on all lines intersecting W. Using the above reasoning for smooth one-forms we obtain  $0 = df_{\epsilon} = df * j_{\epsilon}$  for small  $\epsilon > 0$ . Taking  $\epsilon \to 0$  we get df = 0.



FIGURE 1. Basic idea of the alternative proof of theorem 1.5 when n = 2. We may assume that f is supported in a ball  $B_r$ . We form a closed loop  $\tilde{\gamma}_h$  with the lines  $\gamma$  and  $\gamma_h$  (dashed) enclosing the rectangular region  $R_h$ . Then we apply Stokes' theorem and a limit argument  $h \to 0$  together with a known partial data result for scalar fields to obtain that df = 0.

Now the proof of theorem 1.5 follows in the same way from theorem 1.2 as before using the zero extension  $\tilde{f}$ . Corollaries 1.7 and 1.8 are also direct consequences of theorem 1.2 since  $X_A = X_1 \circ A$ .

Moreover, the above alternative proof can be used to prove a complementary support theorem for the transform  $X_A$ : if  $d(Af)|_C = 0$  and  $X_A f = X_1(Af) = 0$  on all lines not intersecting a convex and compact set C, then  $f = A^{-1}(d\psi)$  for some potential  $\psi$  (see [43, Theorem 7.5] for a similar support theorem for the X-ray transform  $X_1$ ). Indeed, if  $\gamma$  is any line not intersecting C, then we can form a closed loop  $\tilde{\gamma}_h$  as in figure 1 so that the loop is completely contained in  $C^c$  and the ends are outside the support of f. Using Stokes' theorem and a limit argument  $h \to 0$  as in the alternative proof above we obtain that  $X_0(\star d(Af)) = 0$  on all lines not intersecting C. Now we can use the Helgason support theorem for scalar fields (see e.g. [15, Corollary 6.1] and [43, Section 5.2]) to conclude that d(Af) = 0 in  $C^c$ . Since also  $d(Af)|_C = 0$  we get that Af is a closed one-form and thus exact, i.e. there is a scalar field  $\psi$  such that  $Af = d\psi$ .

#### References

- A. Abhishek. Support theorems for the transverse ray transform of tensor fields of rank m. J. Math. Anal. Appl., 485(2):123828, 2020.
- [2] G. Ambartsoumian, M. J. L. Jebelli, and R. K. Mishra. Generalized V-line transforms in 2D vector tomography. *Inverse Problems*, 36(10):104002, 2020.
- [3] Y. M. Assylbekov and N. S. Dairbekov. The X-ray Transform on a General Family of Curves on Finsler Surfaces. J. Geom. Anal., 28(2):1428–1455, 2018.

- [4] H. Braun and A. Hauck. Tomographic Reconstruction of Vector Fields. *IEEE Trans.* Signal Process., 39(2):464–471, 1991.
- [5] M. Courdurier, F. Noo, M. Defrise, and H. Kudo. Solving the interior problem of computed tomography using a priori knowledge. *Inverse Problems*, 24(6):065001, 2008.
- [6] G. Covi, K. Mönkkönen, and J. Railo. Unique continuation property and Poincaré inequality for higher order fractional Laplacians with applications in inverse problems. *Inverse Probl. Imaging*, 2020. To appear.
- [7] M. V. de Hoop, T. Saksala, G. Uhlmann, and J. Zhai. Generic uniqueness and stability for mixed ray transform. 2019. arXiv:1909.11172.
- [8] M. V. de Hoop, T. Saksala, and J. Zhai. Mixed ray transform on simple 2-dimensional Riemannian manifolds. Proc. Amer. Math. Soc., 147(11):4901–4913, 2019.
- [9] G. de Rham. Differentiable Manifolds. Springer-Verlag, First edition, 1984.
- [10] A. Denisjuk. Inversion of the x-ray transform for 3D symmetric tensor fields with sources on a curve. *Inverse Problems*, 22(2):399–411, 2006.
- [11] E. Y. Derevtsov and I. Svetov. Tomography of tensor fields in the plain. Eurasian J. Math. Comput. Appl., 3(2):24–68, 2015.
- [12] T. Ghosh, M. Salo, and G. Uhlmann. The Calderón problem for the fractional Schrödinger equation. Anal. PDE, 13:455–475, 2020.
- [13] D. Gilbarg and N. S. Trudinger. Elliptic Partial Differential Equations of Second Order. Springer, Second edition, 2001. Reprint.
- [14] H. Hammer and B. Lionheart. Application of Sharafutdinov's Ray Transform in Integrated Photoelasticity. J. Elasticity, 75(3):229–246, 2004.
- [15] S. Helgason. Integral Geometry and Radon Transforms. Springer, First edition, 2011.
- [16] J. Horváth. Topological Vector Spaces and Distributions, volume I. Addison-Wesley, 1966.
- [17] J. Ilmavirta and F. Monard. Integral geometry on manifolds with boundary and applications. In R. Ramlau and O. Scherzer, editors, *The Radon Transform: The First 100 Years and Beyond.* de Gruyter, 2019.
- [18] J. Ilmavirta and K. Mönkkönen. Unique continuation of the normal operator of the x-ray transform and applications in geophysics. *Inverse Problems*, 36(4):045014, 2020.
- [19] J. Ilmavirta, K. Mönkkönen, and J. Railo. On tensor decompositions and algebraic structure of the mixed and transverse ray transforms. 2020. arXiv:2009.01043.
- [20] T. Jansson, M. Almqvist, K. Stråhlén, R. Eriksson, G. Sparr, H. W. Persson, and K. Lindström. Ultrasound Doppler vector tomography measurements of directional blood flow. *Ultrasound Med. Biol.*, 23(1):47–57, 1997.
- [21] P. Juhlin. Principles of Doppler Tomography. Technical report, Center for Mathematical Sciences, Lund Institute of Technology, S-221 00 Lund, Sweden, 1992.
- [22] E. Katsevich, A. Katsevich, and G. Wang. Stability of the interior problem with polynomial attenuation in the region of interest. *Inverse Problems*, 28(6):065022, 2012.
- [23] E. Klann, E. T. Quinto, and R. Ramlau. Wavelet methods for a weighted sparsity penalty for region of interest tomography. *Inverse Problems*, 31(2):025001, 2015.
- [24] W. R. B. Lionheart and P. J. Withers. Diffraction tomography of strain. *Inverse Problems*, 31(4):045005, 2015.
- [25] S. Mardare. On Poincaré and de Rham's theorems. Rev. Roumaine Math. Pures Appl., 53(5-6):523–541, 2008.
- [26] W. McLean. Strongly Elliptic Systems and Boundary Integral Equations. Cambridge University Press, First edition, 2000.
- [27] F. Natterer. The Mathematics of Computerized Tomography. SIAM, Philadelphia, 2001. Reprint.
- [28] F. Natterer and F. Wübbeling. Mathematical Methods in Image Reconstruction. SIAM, Philadelphia, 2001.
- [29] S. J. Norton. Tomographic Reconstruction of 2-D Vector Fields: Application to Flow Imaging. *Geophys. J. Int.*, 97(1):161–168, 1989.
- [30] S. J. Norton. Unique Tomographic Reconstruction of Vector Fields Using Boundary Data. *IEEE Trans. Image Process.*, 1(3):406–412, 1992.

14

- [31] G. P. Paternain, M. Salo, and G. Uhlmann. Tensor tomography: Progress and challenges. Chin. Ann. Math. Ser. B, 35(3):399–428, 2014.
- [32] J. Prince. Tomographic Reconstruction of 3-D Vector Fields Using Inner Product Probes. *IEEE Trans. Image Process.*, 3(2):216–219, 1994.
- [33] K. Ramaseshan. Microlocal Analysis of the Doppler Transform on R<sup>3</sup>. J. Fourier Anal. Appl., 10(1):73–82, 2004.
- [34] M. Salo and G. Uhlmann. The Attenuated Ray Transform on Simple Surfaces. J. Differential Geom., 88(1):161–187, 2011.
- [35] T. Schuster. The 3D Doppler transform: elementary properties and computation of reconstruction kernels. *Inverse Problems*, 16(3):701–722, 2000.
- [36] T. Schuster. 20 years of imaging in vector field tomography: a review. In Y. Censor, M. Jiang, and A. K. Louis, editors, *Mathematical Methods in Biomedical Imaging and Intensity-Modulated Dariation Therapy (IMRT)*, Publications of the Scuola Normale Superiore, CRM Series, volume 7. Birkhäuser, 2008.
- [37] T. Schuster. The importance of the Radon transform in vector field tomography. In R. Ramlau and O. Scherzer, editors, *The Radon Transform: The First 100 Years and Beyond.* de Gruyter, 2019.
- [38] A. Schwarz. Multi-tomographic flame analysis with a schlieren apparatus. Meas. Sci. Technol., 7(3):406–413, 1996.
- [39] V. Sharafutdinov. Slice-by-slice reconstruction algorithm for vector tomography with incomplete data. *Inverse Problems*, 23(6):2603–2627, 2007.
- [40] V. A. Sharafutdinov. Integral geometry of tensor fields. Inverse and Ill-posed Problems Series. VSP, Utrecht, 1994.
- [41] G. Sparr and K. Stråhlén. Vector field tomography, an overview. Technical report, Centre for Mathematical Sciences, Lund Institute of Technology, Lund, Sweden, 1998.
- [42] G. Sparr, K. Stråhlén, K. Lindström, and H. W. Persson. Doppler tomography for vector fields. *Inverse Problems*, 11(5):1051–1061, 1995.
- [43] P. Stefanov and G. Uhlmann. Microlocal Analysis and Integral Geometry (working title). 2018. Draft version.
- [44] P. Stefanov, G. Uhlmann, and A. Vasy. Inverting the local geodesic X-ray transform on tensors. J. Anal. Math., 136(1):151–208, 2018.
- [45] K. Stråhlén. Reconstructions from Doppler Radon transforms. In Proceedings of 3rd IEEE International Conference on Image Processing, volume 2, pages 753–756, 1996.
- [46] F. Trèves. Topological Vector Spaces, Distributions and Kernels. Academic Press, First edition, 1967.
- [47] L. B. Vertgeim. Integral geometry problems for symmetric tensor fields with incomplete data. J. Inverse Ill-Posed Probl., 8(3):355–364, 2000.
- [48] J. Yang, H. Yu, M. Jiang, and G. Wang. High-order total variation minimization for interior tomography. *Inverse Problems*, 26(3):035013, 2010.

[D]

# The higher order fractional Calderón problem for linear local operators: uniqueness

G. Covi, K. Mönkkönen, J. Railo and G. Uhlmann

Preprint (2020)

https://arxiv.org/abs/2008.10227

# THE HIGHER ORDER FRACTIONAL CALDERÓN PROBLEM FOR LINEAR LOCAL OPERATORS: UNIQUENESS

GIOVANNI COVI, KEIJO MÖNKKÖNEN, JESSE RAILO, AND GUNTHER UHLMANN

ABSTRACT. We study an inverse problem for the fractional Schrödinger equation (FSE) with a local perturbation by a linear partial differential operator (PDO) of the order smaller than the order of the fractional Laplacian. We show that one can uniquely recover the coefficients of the PDO from the Dirichlet-to-Neumann (DN) map associated to the perturbed FSE. This is proved for two classes of coefficients: coefficients which belong to certain spaces of Sobolev multipliers and coefficients which belong to fractional Sobolev spaces with bounded derivatives. Our study generalizes recent results for the zeroth and first order perturbations to higher order perturbations.

#### 1. INTRODUCTION

Let  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$ ,  $\Omega \subset \mathbb{R}^n$  a bounded open set where  $n \geq 1$ ,  $\Omega_e = \mathbb{R}^n \setminus \overline{\Omega}$  its exterior and P(x, D) a linear partial differential operator (PDO) of order  $m \in \mathbb{N}$ 

$$P(x,D) = \sum_{|\alpha| \le m} a_{\alpha}(x) D^{\alpha}$$

where the coefficients  $a_{\alpha} = a_{\alpha}(x)$  are functions defined in  $\Omega$ . We study a nonlocal inverse problem for the perturbed fractional Schrödinger equation

(1) 
$$\begin{cases} (-\Delta)^s u + P(x, D)u = 0 \text{ in } \Omega \\ u = f \text{ in } \Omega_e \end{cases}$$

where  $(-\Delta)^s$  is a nonlocal pseudo-differential operator  $(-\Delta)^s u = \mathcal{F}^{-1}(|\cdot|^{2s} \hat{u})$  in contrast to the local operator P(x, D). In the inverse problem, one aims to recover the local operator P from the associated Dirichlet-to-Neumann map.

We always assume that 0 is not a Dirichlet eigenvalue of the operator  $((-\Delta)^s + P(x, D))$ , i.e.

If 
$$u \in H^s(\mathbb{R}^n)$$
 solves  $((-\Delta)^s + P(x,D))u = 0$  in  $\Omega$  and  $u|_{\Omega_e} = 0$ , then  $u = 0$ 

Our data for the inverse problem is the Dirichlet-to-Neumann (DN) map  $\Lambda_P \colon H^s(\Omega_e) \to (H^s(\Omega_e))^*$  which maps Dirichlet exterior values to a nonlocal version of the Neumann boundary value (see section 2 and 3.1). The main question that we study in this article is whether the DN map  $\Lambda_P$  determines uniquely the coefficients  $a_{\alpha}$  in  $\Omega$ . In other words, does  $\Lambda_{P_1} = \Lambda_{P_2}$  imply that  $a_{1,\alpha} = a_{2,\alpha}$  in  $\Omega$  for all  $|\alpha| \leq m$ ? We prove that the answer is positive under certain restrictions on the coefficients  $a_{\alpha}$  and the order of the PDOs.

This gives positive answer to the uniqueness problem [10, Question 2.5] posed by the first three authors in a previous work. The precise statement in [10] asks to prove uniqueness for the higher order fractional Calderón problem in the case of a bounded domain with smooth boundary and PDOs with smooth coefficients (up to the boundary). The positive answer to this question follows from theorem 1.2. The study of the fractional Calderón problem was initiated by Ghosh, Salo and Uhlmann in the work [15] where the uniqueness for the associated inverse problem is proved when  $m = 0, s \in (0, 1)$  and  $a_0 \in L^{\infty}(\Omega)$ .

Date: April 22, 2021.

Key words and phrases. Inverse problems, fractional Calderón problem, fractional Schrödinger equation, Sobolev multipliers.

We briefly note that by Peetre's theorem any linear operator  $L: C_c^{\infty}(\Omega) \to C_c^{\infty}(\Omega)$  which does not increase supports, i.e.  $\operatorname{spt}(Lf) \subset \operatorname{spt}(f)$  for all  $f \in C_c^{\infty}(\Omega)$ , is in fact a differential operator [30] (see also the original work [32]). Therefore our results apply to any local operator satisfying such properties and it is enough to study PDOs only. For a more general formulation of Peetre's theorem on the level of vector bundles, see [31].

1.1. Main results. We denote by  $M(H^{s-|\alpha|} \to H^{-s})$  the space of all bounded Sobolev multipliers between the Sobolev spaces  $H^{s-|\alpha|}(\mathbb{R}^n)$  and  $H^{-s}(\mathbb{R}^n)$ . We denote by  $M_0(H^{s-|\alpha|} \to H^{-s}) \subset M(H^{s-|\alpha|} \to H^{-s})$  the space of bounded Sobolev multipliers that can be approximated with smooth compactly supported functions in the multiplier norm of  $M(H^{s-|\alpha|} \to H^{-s})$ . We also write  $H^{r,\infty}(\Omega)$  for the local Bessel potential space with bounded derivatives. See section 2 for more detailed definitions.

Our first theorem is a generalization of [36, Theorem 1.1] which considered the case m = 0 with  $s \in (0, 1)$ . It also generalizes [10, Theorem 1.5] which considered the higher order cases  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$  when m = 0.

**Theorem 1.1.** Let  $\Omega \subset \mathbb{R}^n$  be a bounded open set where  $n \ge 1$ . Let  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$  and  $m \in \mathbb{N}$  be such that 2s > m. Let

$$P_j = \sum_{|\alpha| \le m} a_{j,\alpha} D^{\alpha}, \quad j = 1, 2,$$

be linear PDOs of order m with coefficients  $a_{j,\alpha} \in M_0(H^{s-|\alpha|} \to H^{-s})$ . Given any two open sets  $W_1, W_2 \subset \Omega_e$ , suppose that the DN maps  $\Lambda_{P_i}$  for the equations  $((-\Delta)^s + P_j)u = 0$  in  $\Omega$ satisfy

$$\Lambda_{P_1} f|_{W_2} = \Lambda_{P_2} f|_{W_2}$$

for all  $f \in C_c^{\infty}(W_1)$ . Then  $P_1|_{\Omega} = P_2|_{\Omega}$ .

In theorem 1.1 one can pick the lower order coefficients  $(|\alpha| < s)$  from  $L^p(\Omega)$  for high enough p (especially from  $L^{\infty}(\Omega)$ ) and higher order coefficients  $(s < |\alpha| < 2s)$  from the closure of  $C_c^{\infty}(\Omega)$  in  $H^{r,\infty}(\Omega)$  for certain values of  $r \in \mathbb{R}$ . Lemmas 2.8 and 2.9 give more examples of Sobolev spaces which belong to the space of multipliers  $M_0(H^{s-|\alpha|} \to H^{-s})$ . We also note that when  $|\alpha| = 0$ , then the space of multipliers  $M_0(H^s \to H^{-s})$  coincides with the one studied in [36].

It follows that the space of multipliers is trivial for higher order operators, i.e.  $M(H^{s-|\alpha|} \to H^{-s}) = \{0\}$  when  $s - |\alpha| < -s$ . It would be possible to state theorem 1.1 for higher order PDOs, but that forces  $a_{\alpha} = 0$  for all  $|\alpha| > 2s$ . For this reason we only consider PDOs whose order is m < 2s. See lemma 2.5 and the related remarks for more details.

Our second theorem generalizes [7, Theorem 1.1] and [15, Theorem 1.1] where similar results are proved when m = 0, 1 and  $s \in (0, 1)$ . It also generalizes [10, Theorem 1.5] where the case m = 0 and  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$  was studied.

**Theorem 1.2.** Let  $\Omega \subset \mathbb{R}^n$  be a bounded Lipschitz domain where  $n \ge 1$ . Let  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$  and  $m \in \mathbb{N}$  be such that 2s > m. Let

$$P_j(x,D) = \sum_{|\alpha| \le m} a_{j,\alpha}(x) D^{\alpha}, \quad j = 1, 2,$$

be linear PDOs of order m with coefficients  $a_{i,\alpha} \in H^{r_{\alpha},\infty}(\Omega)$  where

(2) 
$$r_{\alpha} := \begin{cases} 0 & if & |\alpha| - s < 0, \\ |\alpha| - s + \delta & if & |\alpha| - s \in \{1/2, 3/2, ...\}, \\ |\alpha| - s & if & otherwise \end{cases}$$

for any fixed  $\delta > 0$ . Given any two open sets  $W_1, W_2 \subset \Omega_e$ , suppose that the DN maps  $\Lambda_{P_i}$  for the equations  $((-\Delta)^s + P_j(x, D))u = 0$  in  $\Omega$  satisfy

$$\Lambda_{P_1} f|_{W_2} = \Lambda_{P_2} f|_{W_2}$$
  
for all  $f \in C_c^{\infty}(W_1)$ . Then  $P_1(x, D) = P_2(x, D)$ .

Our first theorem is formulated for general bounded open sets and the second theorem for Lipschitz domains. The difference arises in the proof of the well-posedness of the inverse problem. We note that theorem 1.2 holds for coefficients  $a_{\alpha}$  which are smooth up to the boundary  $(a_{\alpha} = g|_{\Omega}$  where  $g \in C^{\infty}(\mathbb{R}^n)$ ). The conditions (2) imply that one can choose  $a_{\alpha} \in L^{\infty}(\Omega)$  for every  $\alpha$  such that  $|\alpha| < s$ . The case  $|\alpha| = s$  never happens, as s is assumed not to be an integer. If  $|\alpha| > s$ , we have  $a_{\alpha} \in H^{|\alpha|-s,\infty}(\Omega)$  when  $|\alpha| - s \notin \{1/2, 3/2, ...\}$ . Thus the conditions (2) coincide with [7, 15] when m = 0, 1 and  $s \in (0, 1)$ .

Our article is roughly divided into two parts. The first part of the article (theorem 1.1 and section 3) generalizes the study of the uniqueness problem for singular potentials in [36] and the second part (theorem 1.2 and section 4) generalizes the uniqueness problem for bounded first order perturbations in [7].

The approach to prove theorems 1.1 and 1.2 is the following. First one shows that the inverse problem is well-posed and the corresponding bilinear forms are bounded. This leads to the boundedness of the DN maps and an Alessandrini identity. By a unique continuation property of the higher order fractional Laplacian one obtains a Runge approximation property for equation (1). Using the Runge approximation and the Alessandrini identity for suitable test functions one proves the uniqueness of the inverse problem.

1.2. On the earlier literature. Equation (1) and theorems 1.1 and 1.2 are related to the Calderón problem for the fractional Schrödinger equation first introduced in [15]. There one tries to uniquely recover the potential q in  $\Omega$  by doing measurements in the exterior  $\Omega_e$ . This is a nonlocal (fractional) counterpart of the classical Calderón problem arising in electrical impedance tomography where one obtains information about the electrical properties of some bounded domain by doing voltage and current measurements on the boundary [39, 40]. In [36] the study of the fractional Calderón problem is extended for "rough" potentials q, i.e. potentials which are in general bounded Sobolev multipliers. First order perturbations were studied in [7] assuming that the fractional part dominates the equation, i.e.  $s \in (1/2, 1)$ , and that the perturbations have bounded fractional derivatives. A higher order version ( $s \in \mathbb{R}^+ \setminus \mathbb{Z}$ ) of the fractional Calderón problem was introduced and studied in [10]. These three articles [7, 10, 36] motivate the study of higher order (rough) perturbations to the fractional Laplacian  $(-\Delta)^s$  in equation (1). The natural restriction for the order of P(x, D) in theorems 1.1 and 1.2

The fractional Calderón problem for  $s \in (0, 1)$  has been studied in many settings. We refer to the survey [37] for a more detailed treatment. In the work [36] stability was proved for singular potentials, and in [34] the related exponential instability was shown. The fractional Calderón problem has also been solved under single measurement [14]. The perturbed equation is related to the fractional magnetic Schrödinger equation which is studied in [9, 24, 25, 26]. See also [4] for a fractional Schrödinger equation with a lower order nonlocal perturbation. Other variants of the fractional Calderón problem include semilinear fractional (magnetic) Schrödinger equation [19, 20, 24, 25], fractional heat equation [21, 35] and fractional conductivity equation [8] (see also [6, 13] for equations arising from a nonlocal Schrödinger-type elliptic operator). In the recent work [10], the first three authors of this article studied higher order versions ( $s \in \mathbb{R}^+ \setminus \mathbb{Z}$ ) of the fractional Calderón problem and proved uniqueness for the Calderón problem for the fractional magnetic Schrödinger equation (up to a gauge). This article continues these studies by showing uniqueness for the fractional Schrödinger equation with higher order perturbations and gives positive answer to the question 2.5 posed in [10].

1.3. Examples of fractional models in the sciences. Equations involving fractional Laplacians like (1) have applications in mathematics and natural sciences. Fractional Laplacians appear in the study of anomalous and nonlocal diffusion, and these diffusion phenomena can be used in many areas such as continuum mechanics, graph theory and ecology just to mention a few [2, 5, 12, 27, 33]. Another place where the fractional counterpart of the classical Laplacian naturally shows up is the formulation of fractional quantum mechanics [22, 23]. For more applications of fractional mathematical models, see [5] and the references therein.

1.4. The organization of the article. In section 2 we introduce the notation and give preliminaries on Sobolev spaces and fractional Laplacians. We also define the spaces of rough coefficients (Sobolev multipliers) and discuss some of the basic properties. In section 3 we prove theorem 1.1 in detail. Finally, in section 4 we prove theorem 1.2 but as the proofs of both theorems are very similar we do not repeat all identical steps and we keep our focus in the differences of the proofs.

Acknowledgements. G.C. was partially supported by the European Research Council under Horizon 2020 (ERC CoG 770924). K.M. and J.R. were supported by Academy of Finland (Centre of Excellence in Inverse Modelling and Imaging, grant numbers 284715 and 309963). G.U. was partly supported by NSF, a Walker Family Endowed Professorship at UW and a Si Yuan Professorship at IAS, HKUST. G.C. would like to thank Angkana Rüland for helpful discussions and her hospitality during his visit to Max Planck Institute for Mathematics in the Sciences. J.R. and G.U. wish to thank Maarten V. de Hoop, Rice University and Simons Foundation for providing the support to participate 2020 MATH + X Symposium on Inverse Problems and Deep Learning, Mitigating Natural Hazards, where this project was initiated.

#### 2. Preliminaries

In this section we recall some basic theory of Sobolev spaces, Fourier analysis and fractional Laplacians on  $\mathbb{R}^n$ . We also introduce the spaces of Sobolev multipliers and prove a few properties for them. Some auxiliary lemmas which are needed in the proofs of our main theorems are given as well. We follow the references [1, 15, 29, 28, 38, 41] (see also section 3 in [10]).

2.1. Sobolev spaces. The (inhomogeneous) fractional  $L^2$ -based Sobolev space of order  $r \in \mathbb{R}$  is defined to be

$$H^{r}(\mathbb{R}^{n}) = \{ u \in \mathscr{S}'(\mathbb{R}^{n}) : \mathcal{F}^{-1}(\langle \cdot \rangle^{r} \hat{u}) \in L^{2}(\mathbb{R}^{n}) \}$$

equipped with the norm

$$\|u\|_{H^r(\mathbb{R}^n)} = \left\|\mathcal{F}^{-1}(\langle\cdot\rangle^r \hat{u})\right\|_{L^2(\mathbb{R}^n)}.$$

Here  $\hat{u} = \mathcal{F}(u)$  is the Fourier transform of a tempered distribution  $u \in \mathscr{S}'(\mathbb{R}^n)$ ,  $\mathcal{F}^{-1}$  is the inverse Fourier transform and  $\langle x \rangle = (1 + |x|^2)^{1/2}$ . We define the fractional Laplacian of order  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$  as  $(-\Delta)^s \varphi = \mathcal{F}^{-1}(|\cdot|^{2s} \hat{\varphi})$  where  $\varphi \in \mathscr{S}(\mathbb{R}^n)$  is a Schwartz function. Then  $(-\Delta)^s$  extends to a bounded operator  $(-\Delta)^s \colon H^r(\mathbb{R}^n) \to H^{r-2s}(\mathbb{R}^n)$  for all  $r \in \mathbb{R}$  by density of  $\mathscr{S}(\mathbb{R}^n)$  in  $H^r(\mathbb{R}^n)$ .

Let  $\Omega \subset \mathbb{R}^n$  be an open set and  $F \subset \mathbb{R}^n$  a closed set. We define the following Sobolev spaces

 $\begin{aligned} H^r_F(\mathbb{R}^n) &= \{ u \in H^r(\mathbb{R}^n) : \operatorname{spt}(u) \subset F \} \\ \widetilde{H}^r(\Omega) &= \text{ closure of } C^\infty_c(\Omega) \text{ in } H^r(\mathbb{R}^n) \\ H^r(\Omega) &= \{ u|_{\Omega} : u \in H^r(\mathbb{R}^n) \} \\ H^r_0(\Omega) &= \text{ closure of } C^\infty_c(\Omega) \text{ in } H^r(\Omega). \end{aligned}$ 

It follows that  $\widetilde{H}^r(\Omega) \subset H^r_0(\Omega)$ ,  $\widetilde{H}^r(\Omega) \subset H^r_{\overline{\Omega}}(\mathbb{R}^n)$ ,  $(\widetilde{H}^r(\Omega))^* = H^{-r}(\Omega)$  and  $(H^r(\Omega))^* = \widetilde{H}^{-r}(\Omega)$  for any open set  $\Omega$  and  $r \in \mathbb{R}$ . If  $\Omega$  is in addition a Lipschitz domain, then we have  $\widetilde{H}^r(\Omega) = H^r_{\overline{\Omega}}(\mathbb{R}^n)$  for all  $r \in \mathbb{R}$  and  $H^r_0(\Omega) = H^r_{\overline{\Omega}}(\mathbb{R}^n)$  when r > -1/2 such that  $r \notin \{\frac{1}{2}, \frac{3}{2}, \frac{5}{2} \dots\}$ . More generally, let  $1 \leq p \leq \infty$  and  $r \in \mathbb{R}$ . We define the Bessel potential space

$$H^{r,p}(\mathbb{R}^n) = \{ u \in \mathscr{S}'(\mathbb{R}^n) : \mathcal{F}^{-1}(\langle \cdot \rangle^r \hat{u}) \in L^p(\mathbb{R}^n) \}$$

equipped with the norm

$$\|u\|_{H^{r,p}(\mathbb{R}^n)} = \left\|\mathcal{F}^{-1}(\langle\cdot\rangle^r \hat{u})\right\|_{L^p(\mathbb{R}^n)}.$$

We also write  $\mathcal{F}^{-1}(\langle \cdot \rangle^r \hat{u}) =: J^r u$  where the Fourier multiplier  $J = (\mathrm{Id} - \Delta)^{1/2}$  is called the Bessel potential. We have the continuous inclusions  $H^{r,p}(\mathbb{R}^n) \hookrightarrow H^{t,p}(\mathbb{R}^n)$  whenever  $r \geq t$  [41]. By the

Mikhlin multiplier theorem one can show that  $(-\Delta)^s : H^{r,p}(\mathbb{R}^n) \to H^{r-2s,p}(\mathbb{R}^n)$  is continuous whenever  $s \ge 0$  and  $1 . The local version of the space <math>H^{r,p}(\mathbb{R}^n)$  is defined as earlier by the restrictions

$$H^{r,p}(\Omega) = \{ u |_{\Omega} : u \in H^{r,p}(\mathbb{R}^n) \}$$

where  $\Omega \subset \mathbb{R}^n$  is any open set. This space is equipped with the quotient norm

$$\|v\|_{H^{r,p}(\Omega)} = \inf\{\|w\|_{H^{r,p}(\mathbb{R}^n)} : w \in H^{r,p}(\mathbb{R}^n), \ w|_{\Omega} = v\}.$$

We have the continuous inclusions  $H^{r,p}(\Omega) \hookrightarrow H^{t,p}(\Omega)$  whenever  $r \ge t$  by the definition of the quotient norm.

We also define the spaces

$$H_F^{r,p}(\mathbb{R}^n) = \{ u \in H^{r,p}(\mathbb{R}^n) : \operatorname{spt}(u) \subset F \}$$
  
$$\widetilde{H}^{r,p}(\Omega) = \text{ closure of } C_c^{\infty}(\Omega) \text{ in } H^{r,p}(\mathbb{R}^n)$$
  
$$H_0^{r,p}(\Omega) = \text{ closure of } C_c^{\infty}(\Omega) \text{ in } H^{r,p}(\Omega)$$

where  $F \subset \mathbb{R}^n$  is a closed set. Note that  $\widetilde{H}^{r,p}(\Omega) \subset H_0^{r,p}(\Omega)$  since the restriction map  $|_{\Omega} \colon H^{r,p}(\mathbb{R}^n) \to H^{r,p}(\Omega)$  is by definition continuous. One can also see that  $\widetilde{H}^{r,p}(\Omega) \subset H_{\overline{\Omega}}^{r,p}(\mathbb{R}^n)$ . If  $\Omega$  is a bounded  $C^{\infty}$ -domain and 1 , then we have [38, Theorem 1 in section 4.3.2]

$$\begin{split} \widetilde{H}^{r,p}(\Omega) &= H^{r,p}_{\overline{\Omega}}(\mathbb{R}^n), \quad s \in \mathbb{R} \\ H^{r,p}_0(\Omega) &= H^{r,p}(\Omega), \quad s \leq \frac{1}{p}. \end{split}$$

Some authors (especially in [7, 36]) use the notation  $W^{r,p}(\Omega)$  for Bessel potential spaces. We have decided to use the notation  $H^{r,p}(\Omega)$  so that these spaces are not confused with the Sobolev-Slobodeckij spaces which are in general different from the Bessel potential spaces [11].

The equation (1) we study is nonlocal. Instead of putting boundary conditions we impose exterior values for the equation. This can be done by saying that u = f in  $\Omega_e$  if  $u - f \in \widetilde{H}^s(\Omega)$ . Motivated by this we define the (abstract) trace space  $X = H^r(\mathbb{R}^n)/\widetilde{H}^r(\Omega)$ , i.e. functions in X are the same (have the same trace) if they agree in  $\Omega_e$ . If  $\Omega$  is a Lipschitz domain, then we have  $X = H^r(\Omega_e)$  and  $X^* = H_{\overline{\Omega}_e}^{-r}(\mathbb{R}^n)$ .

2.2. **Properties of the fractional Laplacian.** The fractional Laplacian admits two important properties which we need in our proofs. The first one is unique continuation property (UCP) which is used in proving the Runge approximation property.

**Lemma 2.1** (UCP). Let  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$ ,  $r \in \mathbb{R}$  and  $u \in H^r(\mathbb{R}^n)$ . If  $(-\Delta)^s u|_V = 0$  and  $u|_V = 0$  for some nonempty open set  $V \subset \mathbb{R}^n$ , then u = 0.

Lemma 2.1 is proved in [10] for s > 1 by reducing the problem to the UCP result for  $s \in (0, 1)$  in [15]. Note that such property is not true for local operators like the classical Laplacian  $(-\Delta)$ . The second property we need is the Poincaré inequality, which is used in showing that the fractional Calderón problem is well-posed.

**Lemma 2.2** (Poincaré inequality). Let  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$ ,  $K \subset \mathbb{R}^n$  compact set and  $u \in H^s_K(\mathbb{R}^n)$ . There exists a constant c = c(n, K, s) > 0 such that

$$\|u\|_{L^2(\mathbb{R}^n)} \le c \left\| (-\Delta)^{s/2} u \right\|_{L^2(\mathbb{R}^n)}$$

Many different proofs for lemma 2.2 are given in [10]. We note that in the literature, the fractional Poincaré inequality is typically considered only when  $s \in (0, 1)$ .

Finally, we recall the fractional Leibniz rule, also known as the Kato-Ponce inequality. It is used to show the boundedness of the bilinear forms associated to the perturbed fractional Schrödinger equation in the case when the coefficients of the PDO have bounded fractional derivatives. **Lemma 2.3** (Kato-Ponce inequality). Let  $s \ge 0$ ,  $1 < r < \infty$ ,  $1 < q_1 \le \infty$  and  $1 < p_2 \le \infty$  such that  $\frac{1}{r} = \frac{1}{p_1} + \frac{1}{q_1} = \frac{1}{p_2} + \frac{1}{q_2}$ . If  $f \in L^{p_2}(\mathbb{R}^n)$ ,  $J^s f \in L^{p_1}(\mathbb{R}^n)$ ,  $g \in L^{q_1}(\mathbb{R}^n)$  and  $J^s g \in L^{q_2}(\mathbb{R}^n)$ , then  $J^s(fg) \in L^r(\mathbb{R}^n)$  and

$$\|J^{s}(fg)\|_{L^{r}(\mathbb{R}^{n})} \leq C(\|J^{s}f\|_{L^{p_{1}}(\mathbb{R}^{n})} \|g\|_{L^{q_{1}}(\mathbb{R}^{n})} + \|f\|_{L^{p_{2}}(\mathbb{R}^{n})} \|J^{s}g\|_{L^{q_{2}}(\mathbb{R}^{n})})$$

where  $J^s$  is the Bessel potential of order s and  $C = C(s, n, r, p_1, p_2, q_1, q_2)$ .

The proof of lemma 2.3 can be found in [17] (see also [16, 18]).

2.3. Spaces of rough coefficients. Following [28, Ch. 3], we introduce the space of multipliers  $M(H^r \to H^t)$  between pairs of Sobolev spaces. Here we are assuming that  $r, t \in \mathbb{R}$ . The coefficients of P(x, D) in theorem 1.1 will be picked from such spaces of multipliers.

If  $f \in \mathcal{D}'(\mathbb{R}^n)$  is a distribution, we say that  $f \in M(H^r \to H^t)$  whenever the norm

$$|f||_{r,t} := \sup\{|\langle f, uv \rangle| ; u, v \in C_c^{\infty}(\mathbb{R}^n), ||u||_{H^r(\mathbb{R}^n)} = ||v||_{H^{-t}(\mathbb{R}^n)} = 1\}$$

is finite. Here  $\langle \cdot, \cdot \rangle$  is the duality pairing. By  $M_0(H^r \to H^t)$  we indicate the closure of  $C_c^{\infty}(\mathbb{R}^n)$ in  $M(H^r \to H^t) \subset \mathcal{D}'(\mathbb{R}^n)$ . If  $f \in M(H^r \to H^t)$  and  $u, v \in C_c^{\infty}(\mathbb{R}^n)$  are both non-vanishing, we have the multiplier inequality (3)

$$|\langle f, uv \rangle| = \left| \left\langle f, \frac{u}{\|u\|_{H^{r}(\mathbb{R}^{n})}} \frac{v}{\|v\|_{H^{-t}(\mathbb{R}^{n})}} \right\rangle \right| \|u\|_{H^{r}(\mathbb{R}^{n})} \|v\|_{H^{-t}(\mathbb{R}^{n})} \leq \|f\|_{r,t} \|u\|_{H^{r}(\mathbb{R}^{n})} \|v\|_{H^{-t}(\mathbb{R}^{n})}.$$

By density (3) can be extended to act over  $u \in H^r(\mathbb{R}^n), v \in H^{-t}(\mathbb{R}^n)$ . Moreover, each  $f \in M(H^r \to H^t)$  gives rise to a multiplication map  $m_f : H^r(\mathbb{R}^n) \to H^t(\mathbb{R}^n)$  defined as

$$\langle m_f(u), v \rangle := \langle f, uv \rangle$$
 for all  $u \in H^r(\mathbb{R}^n), v \in H^{-t}(\mathbb{R}^n).$ 

We have as well the unique adjoint multiplication map  $m_f^*: H^{-t}(\mathbb{R}^n) \to H^{-r}(\mathbb{R}^n)$  such that

$$\langle m_f^*(v), u \rangle := \langle f, uv \rangle$$
 for all  $u \in H^r(\mathbb{R}^n), v \in H^{-t}(\mathbb{R}^n)$ 

Since one sees that the adjoint of  $m_f$  is  $m_f^*$ , the chosen notation is justified. For convenience, in the rest of the paper we will just write fu for both  $m_f(u)$  and  $m_f^*(u)$ .

**Remark 2.4.** The spaces of rough coefficients we use are generalizations of the ones considered in [36]. In fact, the space  $Z^{-s}(\mathbb{R}^n)$  used there coincides with our space  $M(H^s \to H^{-s})$ .

In the next lemma we state some elementary properties of the spaces of multipliers. Other interesting properties may be found in [28].

**Lemma 2.5.** Let  $\lambda, \mu \geq 0$  and  $r, t \in \mathbb{R}$ . Then

- (i)  $M(H^r \to H^t) = M(H^{-t} \to H^{-r})$ , and the norms associated to the two spaces also coincide.
- (ii)  $M(H^{r-\lambda} \to H^{t+\mu}) \hookrightarrow M(H^r \to H^t)$  continuously.
- (iii)  $M(H^r \to H^t) = \{0\}$  whenever r < t.

*Proof.* (i) Let  $f \in \mathcal{D}'(\mathbb{R}^n)$  be a distribution. Then by just using the definition we see that

$$||f||_{r,t} = \sup\{|\langle f, uv\rangle| \; ; \; u, v \in C_c^{\infty}(\mathbb{R}^n), ||u||_{H^r(\mathbb{R}^n)} = ||v||_{H^{-t}(\mathbb{R}^n)} = 1\}$$

$$= \sup\{|\langle f, vu \rangle| ; v, u \in C_c^{\infty}(\mathbb{R}^n), \|v\|_{H^{-t}(\mathbb{R}^n)} = \|u\|_{H^{-(-r)}(\mathbb{R}^n)} = 1\} = \|f\|_{-t, -r}.$$

(ii) Observe that the given definition of  $||f||_{r,t}$  is equivalent to the following:

$$||f||_{r,t} = \sup\{|\langle f, uv \rangle| ; u, v \in C_c^{\infty}(\mathbb{R}^n), ||u||_{H^r(\mathbb{R}^n)} \le 1, ||v||_{H^{-t}(\mathbb{R}^n)} \le 1\}$$

Since  $\lambda, \mu \geq 0$ , we also have

$$\|u\|_{H^{r-\lambda}(\mathbb{R}^n)} \le \|u\|_{H^r(\mathbb{R}^n)}, \quad \|v\|_{H^{-(t+\mu)}(\mathbb{R}^n)} \le \|v\|_{H^{-t}(\mathbb{R}^n)}$$

This implies  $||f||_{r,t} \leq ||f||_{r-\lambda,t+\mu}$ , which in turn gives the wanted inclusion.

(iii) If  $0 \le r < t$ , then this was considered in [28, Ch. 3]. The proof given there recalls the easier one for Sobolev spaces ([28, Sec. 2.1]), which is based on the explicit computation of derivatives of aptly chosen exponential functions.

If  $r < t \leq 0$ , then by point (i) we have  $M(H^r \to H^t) = M(H^{-t} \to H^{-r})$ . We need to show that  $M(H^{-t} \to H^{-r}) = \{0\}$  whenever  $0 \leq -t < -r$ . This reduces the problem back to the case of non-negative Sobolev scales.

If  $r \leq 0 < t$ , then  $-r \geq 0$ . Now by point (ii), we have  $M(H^r \to H^t) \subseteq M(H^{r+(-r)} \to H^t) = M(L^2 \to H^t)$ . It is therefore enough to show that this last space is trivial, which again immediately follows from the case of non-negative Sobolev scales.

If  $r < 0 \le t$ , then the problem can be reduced again to the earlier cases.

**Remark 2.6.** We also have  $M_0(H^{r-\lambda} \to H^{t+\mu}) \subseteq M_0(H^r \to H^t)$  whenever  $\lambda, \mu \ge 0$ , since the inclusion in (ii) is continuous.

**Remark 2.7.** In light of lemma 2.5 (ii) we are only interested in  $M(H^r \to H^t)$  in the case  $r \ge t$ , the case r < t being trivial. For our theorem 1.1, this translates into the condition  $m \le 2s$ . We decided not to consider the limit case m = 2s in this work, as our machinery (in particular, the coercivity estimate (25)) breaks down in this case. However, it should be noted that since by assumption we have  $m \in \mathbb{Z}$  and  $s \notin \mathbb{Z}$ , the equality m = 2s can only arise if m is odd, which forces s = 1/2 + k with  $k \in \mathbb{Z}$ . This case was excluded in [7, 15] as well.

The next lemmas relate our spaces of multipliers with some special Bessel potential spaces. This is interesting since in the coming section 3 we will consider the inverse problem for coefficients coming from such spaces. We start with a general result.

**Lemma 2.8.** Let  $\Omega \subset \mathbb{R}^n$  be an open set and let  $t \in \mathbb{R}$  and  $r \in \mathbb{R}$  be such that  $t > \max\{0, r\}$ . The following inclusions hold:

- (i)  $\widetilde{H}^{r',\infty}(\Omega) \subset M_0(H^{-r} \to H^{-t})$  whenever  $r' \ge \max\{0,r\}$ .
- (ii)  $H_0^{r',\infty}(\Omega) \subset M_0(H^{-r} \to H^{-t})$  whenever  $r' \ge \max\{0,r\}$  such that  $r' \notin \{\frac{1}{2}, \frac{3}{2}, \frac{5}{2}, \dots\}$  and  $\Omega$  is a Lipschitz domain.
- (iii)  $\widetilde{H}^{r'}(\Omega) \subset M_0(H^{-r} \to H^{-t})$  whenever  $r' \geq t$  and r' > n/2. The same holds true for  $H_{\overline{\Omega}}^{r'}(\mathbb{R}^n)$  if  $\Omega$  is a Lipschitz domain, and for  $H_0^{r'}(\Omega)$  when  $\Omega$  is a Lipschitz domain and  $r' \notin \{\frac{1}{2}, \frac{3}{2}, \frac{5}{2}, \dots\}$ .

Proof. Throughout the proof we assume that  $u, v \in C_c^{\infty}(\mathbb{R}^n)$  such that  $||u||_{H^{-r}(\mathbb{R}^n)} = ||v||_{H^t(\mathbb{R}^n)} = 1$ . In parts (i) and (ii) we can assume that r' < t since if  $r' \ge t$ , then we have the continuous inclusion  $H^{r',\infty}(\Omega) \hookrightarrow H^{r'',\infty}(\Omega)$  where  $\max\{0,r\} \le r'' < t$  (such r'' always exists since  $t > \max\{0,r\}$ ).

(i) Let 
$$f \in H^{r',\infty}(\Omega)$$
. Now  $f = f_1 + f_2$  where  $f_1 \in C_c^{\infty}(\Omega)$  and  $||f_2||_{H^{r',\infty}(\mathbb{R}^n)} \leq \epsilon$ . Then

$$\begin{aligned} \langle f_2, uv \rangle &| \le \| f_2 v \|_{H^{r'}(\mathbb{R}^n)} \| u \|_{H^{-r'}(\mathbb{R}^n)} \le C \| f_2 \|_{H^{r',\infty}(\mathbb{R}^n)} \| v \|_{H^{r'}(\mathbb{R}^n)} \| u \|_{H^{-r}(\mathbb{R}^n)} \\ &\le C \epsilon \| v \|_{H^t(\mathbb{R}^n)} = C \epsilon. \end{aligned}$$

Here we used the Kato-Ponce inequality (lemma 2.3)

$$\begin{split} \left\| J^{r'}(f_2 v) \right\|_{L^2(\mathbb{R}^n)} &\leq C(\|f_2\|_{L^{\infty}(\mathbb{R}^n)} \left\| J^{r'} v \right\|_{L^2(\mathbb{R}^n)} + \left\| J^{r'} f_2 \right\|_{L^{\infty}(\mathbb{R}^n)} \|v\|_{L^2(\mathbb{R}^n)}) \\ &\leq C \|f_2\|_{H^{r',\infty}(\mathbb{R}^n)} \|v\|_{H^{r'}(\mathbb{R}^n)} \end{split}$$

and the assumption  $\max\{0, r\} \leq r' < t$ . Therefore  $||f - f_1||_{-r, -t} = ||f_2||_{-r, -t} \leq C\epsilon$  which shows that  $f \in M_0(H^{-r} \to H^{-t})$ .

(ii) Let  $f \in H_0^{r',\infty}(\Omega)$ . Now  $f = f_1 + f_2$  where  $f_1 \in C_c^{\infty}(\Omega)$  and  $||f_2||_{H^{r',\infty}(\Omega)} \leq \epsilon$ . By the definition of the quotient norm  $||\cdot||_{H^{r',\infty}(\Omega)}$  we can take  $F \in H^{r',\infty}(\mathbb{R}^n)$  such that  $F|_{\Omega} = f_2$  and  $||F||_{H^{r',\infty}(\mathbb{R}^n)} \leq 2 ||f_2||_{H^{r',\infty}(\Omega)}$ . The assumptions imply the duality  $(H^{-r'}(\Omega))^* = H_0^{r'}(\Omega) \subset$ 

 $H^{r'}(\Omega)$ . Using the Kato-Ponce inequality for the extension F we obtain as in the proof of part (i) that

$$\left\| J^{r'}(Fv) \right\|_{L^2(\mathbb{R}^n)} \le C \left\| F \right\|_{H^{r',\infty}(\mathbb{R}^n)} \left\| v \right\|_{H^{r'}(\mathbb{R}^n)} \le 2C \left\| f_2 \right\|_{H^{r',\infty}(\Omega)} \left\| v \right\|_{H^t(\mathbb{R}^n)} \le 2C\epsilon$$

and hence

$$\begin{aligned} |\langle f_2, uv \rangle| &\leq \|f_2 v\|_{(H^{-r'}(\Omega))^*} \|u\|_{H^{-r'}(\Omega)} \leq \|f_2 v\|_{H^{r'}(\Omega)} \|u\|_{H^{-r}(\mathbb{R}^n)} \\ &\leq \left\| J^{r'}(Fv) \right\|_{L^2(\mathbb{R}^n)} \leq 2C\epsilon. \end{aligned}$$

This shows that  $f \in M_0(H^{-r} \to H^{-t})$ .

(iii) Let  $f \in \widetilde{H}^{r'}(\Omega)$ . Now  $f = f_1 + f_2$  where  $f_1 \in C_c^{\infty}(\Omega)$  and  $||f_2||_{H^{r'}(\mathbb{R}^n)} \leq \epsilon$ . Now [3, Theorem 7.3] implies the continuity of the multiplication  $H^{r'}(\mathbb{R}^n) \times H^t(\mathbb{R}^n) \hookrightarrow H^t(\mathbb{R}^n)$  when  $r' \geq t$  and r' > n/2. We obtain

$$|\langle f_2, uv \rangle| \le \|f_2v\|_{H^t(\mathbb{R}^n)} \|u\|_{H^{-t}(\mathbb{R}^n)} \le C \|f_2\|_{H^{r'}(\mathbb{R}^n)} \|v\|_{H^t(\mathbb{R}^n)} \|u\|_{H^{-r}(\mathbb{R}^n)} \le C\epsilon.$$

Hence  $f \in M_0(H^{-r} \to H^{-t})$ . If  $\Omega$  is a Lipschitz domain, then  $H_{\overline{\Omega}}^{r'}(\mathbb{R}^n) = \widetilde{H}^{r'}(\Omega)$ . If in addition  $r' \notin \{\frac{1}{2}, \frac{3}{2}, \frac{5}{2}, \dots\}$ , we also have  $H_0^{r'}(\Omega) = \widetilde{H}^{r'}(\Omega)$ .

Note that the assumptions in theorem 1.1 satisfy the conditions of the previous lemma since then  $r = |\alpha| - s$  and t = s. The following lemma gives examples of spaces of lower order coefficients  $(|\alpha| \leq s)$ .

**Lemma 2.9.** Let  $\Omega \subset \mathbb{R}^n$  be an open set and t > 0. The following inclusions hold:

- (i)  $L^p(\Omega) \subset M_0(H^0 \to H^{-t})$  whenever  $2 \leq p < \infty$  and p > n/t. Especially, if  $\Omega$  is bounded, then  $L^{\infty}(\Omega) \subset M_0(H^0 \to H^{-t})$ .
- (ii)  $\widetilde{H}^r(\Omega) \subset M_0(H^0 \to H^{-t})$  whenever  $r \ge 0$  and r > n/2 t. The same holds true for  $H^r_{\overline{\Omega}}(\mathbb{R}^n)$  if  $\Omega$  is a Lipschitz domain, and for  $H^r_0(\Omega)$  when  $\Omega$  is Lipschitz domain and  $r \notin \{\frac{1}{2}, \frac{3}{2}, \frac{5}{2}, \ldots\}$ .

*Proof.* Throughout the proof we assume that  $u, v \in C_c^{\infty}(\mathbb{R}^n)$  such that  $||u||_{L^2(\mathbb{R}^n)} = ||v||_{H^t(\mathbb{R}^n)} = 1$ .

(i) Let  $f \in L^p(\Omega)$ . By density of  $C_c^{\infty}(\Omega)$  in  $L^p(\Omega)$  we have  $f = f_1 + f_2$  where  $f_1 \in C_c^{\infty}(\Omega)$ and  $\left\| \widetilde{f_2} \right\|_{L^p(\mathbb{R}^n)} \leq \epsilon$  where  $\widetilde{f_2}$  is the zero extension of  $f_2 \in L^p(\Omega)$ . The assumptions on p imply the continuity of the multiplication  $L^p(\mathbb{R}^n) \times H^t(\mathbb{R}^n) \hookrightarrow L^2(\mathbb{R}^n)$  ([3, Theorem 7.3]) and we have

$$\left|\left\langle \widetilde{f}_{2}, uv \right\rangle\right| \leq \left\|\widetilde{f}_{2}v\right\|_{L^{2}(\mathbb{R}^{n})} \|u\|_{L^{2}(\mathbb{R}^{n})} \leq C \left\|\widetilde{f}_{2}\right\|_{L^{p}(\mathbb{R}^{n})} \|v\|_{H^{t}(\mathbb{R}^{n})} \leq C\epsilon$$

This gives that  $f \in M_0(H^0 \to H^{-t})$ . If  $\Omega$  is bounded, we have  $L^{\infty}(\Omega) \hookrightarrow L^p(\Omega)$  for all  $1 \leq p < \infty$ , giving the second claim.

(ii) Let  $f \in \widetilde{H}^r(\Omega)$ . Now we have  $f = f_1 + f_2$  where  $f_1 \in C_c^{\infty}(\Omega)$  and  $||f_2||_{H^r(\mathbb{R}^n)} \leq \epsilon$ . The assumptions on r imply that the multiplication  $H^r(\mathbb{R}^n) \times H^t(\mathbb{R}^n) \to L^2(\mathbb{R}^n)$  is continuous ([3, Theorem 7.3]). We obtain

$$|\langle f_2, uv \rangle| \le \|f_2v\|_{L^2(\mathbb{R}^n)} \|u\|_{L^2(\mathbb{R}^n)} \le C \|f_2\|_{H^r(\mathbb{R}^n)} \|v\|_{H^t(\mathbb{R}^n)} \le C\epsilon$$

and therefore  $f \in M_0(H^0 \to H^{-t})$ . The claims for  $H^r_{\overline{\Omega}}(\mathbb{R}^n)$  and  $H^r_0(\Omega)$  follow as in the proof of part (iii) of lemma 2.8 from the usual identifications for Lipschtiz domains.

As mentioned above we put t = s > 0 in theorem 1.1 and the condition in lemma 2.9 is satisfied. Note that under the assumption  $|\alpha| \leq s$  we have  $M_0(H^0 \to H^{-s}) \subset M_0(H^{s-|\alpha|} \to H^{-s})$ . Hence we can choose the lower order coefficients from a less regular space in theorem 1.1 (compare to lemma 2.8).

#### 3. Main theorem for singular coefficients

In this section, to shorten the notation, we will write  $\|\cdot\|_{H^s}$ ,  $\|\cdot\|_{L^2}$  and so on for the global norms in  $\mathbb{R}^n$  when the base set is not written explicitly.

#### 3.1. Well-posedness of the inverse problem. Consider the problem

(4) 
$$(-\Delta)^{s}u + \sum_{|\alpha| \le m} a_{\alpha}(D^{\alpha}u) = F \quad \text{in } \Omega,$$
$$u = f \quad \text{in } \Omega_{e}$$

and the corresponding *adjoint-problem* 

(5) 
$$(-\Delta)^s u^* + \sum_{|\alpha| \le m} (-1)^{|\alpha|} D^{\alpha}(a_{\alpha} u^*) = F^* \quad \text{in } \Omega,$$
$$u^* = f^* \quad \text{in } \Omega_e.$$

Note that if  $u, u^* \in H^s(\mathbb{R}^n)$  and  $a_\alpha \in M(H^{s-|\alpha|} \to H^{-s}) = M(H^s \to H^{|\alpha|-s})$ , then  $a_\alpha(D^\alpha u) \in H^{-s}(\mathbb{R}^n)$  and  $D^\alpha(a_\alpha u^*) \in H^{-s}(\mathbb{R}^n)$  matching with  $(-\Delta)^s u, (-\Delta)^s u^* \in H^{-s}(\mathbb{R}^n)$ .

The problems (4) and (5) are associated to the bilinear forms

(6) 
$$B_P(v,w) := \langle (-\Delta)^{s/2}v, (-\Delta)^{s/2}w \rangle_{\mathbb{R}^n} + \sum_{|\alpha| \le m} \langle a_\alpha, (D^\alpha v)w \rangle_{\mathbb{R}^n}$$

and

(7) 
$$B_P^*(v,w) := \langle (-\Delta)^{s/2}v, (-\Delta)^{s/2}w \rangle_{\mathbb{R}^n} + \sum_{|\alpha| \le m} \langle a_\alpha, v(D^\alpha w) \rangle_{\mathbb{R}^n}$$

defined on  $v, w \in C_c^{\infty}(\mathbb{R}^n)$ . In the latter terms of the bilinear forms we have written the dual pairing as  $\langle \cdot, \cdot \rangle_{\mathbb{R}^n}$  since  $a_{\alpha}$  is now a distribution in the whole space  $\mathbb{R}^n$  in contrast to section 4 where  $a_{\alpha}$  is an object defined only in  $\Omega$ .

**Remark 3.1.** Observe that  $B_P$  is not symmetric, which motivates the introduction of the bilinear form  $B_P^*$ . Moreover, one sees by simple inspection that  $B_P(v, w) = B_P^*(w, v)$  for all  $v, w \in C_c^{\infty}(\mathbb{R}^n)$ . This identity holds for  $v, w \in H^s(\mathbb{R}^n)$  as well by density, thanks to the following boundedness lemma.

**Lemma 3.2** (Boundedness of the bilinear forms). Let  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$  and  $m \in \mathbb{N}$  such that  $2s \geq m$ , and let  $a_{\alpha} \in M(H^{s-|\alpha|} \to H^{-s})$ . Then  $B_P$  and  $B_P^*$  extend as bounded bilinear forms on  $H^s(\mathbb{R}^n) \times H^s(\mathbb{R}^n)$ .

Proof of lemma 3.2. We only prove the boundedness of  $B_P$ , as for  $B_P^*$  one can proceed in the same way. The proof is a simple calculation following from inequality (3). Let  $u, v \in C_c^{\infty}(\mathbb{R}^n)$ . We can then estimate that

$$|B_{P}(v,w)| \leq |\langle (-\Delta)^{s/2}v, (-\Delta)^{s/2}w \rangle_{\mathbb{R}^{n}}| + \sum_{|\alpha| \leq m} |\langle a_{\alpha}, D^{\alpha}vw \rangle_{\mathbb{R}^{n}}|$$
  
$$\leq ||w||_{H^{s}(\mathbb{R}^{n})} ||v||_{H^{s}(\mathbb{R}^{n})} + \sum_{|\alpha| \leq m} ||a_{\alpha}||_{s-|\alpha|,-s} ||D^{\alpha}v||_{H^{s-|\alpha|}(\mathbb{R}^{n})} ||w||_{H^{s}(\mathbb{R}^{n})}$$
  
$$\leq \left(1 + \sum_{|\alpha| \leq m} ||a_{\alpha}||_{s-|\alpha|,-s}\right) ||w||_{H^{s}(\mathbb{R}^{n})} ||v||_{H^{s}(\mathbb{R}^{n})}.$$

Now the claim follows from the density of  $C_c^{\infty}(\mathbb{R}^n)$  in  $H^s(\mathbb{R}^n)$ .

Next we shall prove existence and uniqueness of solutions for the problems (4) and (5). To this end, we will use the following form of Young's inequality, which holds for all  $a, b, \eta \in \mathbb{R}^+$ 

and  $p, q \in (1, \infty)$  such that 1/p + 1/q = 1:

(8) 
$$ab \le \frac{(q\eta)^{-p/q}}{p}a^p + \eta b^q$$

The validity of (8) is easily proved by choosing  $a_1 = a(q\eta)^{-1/q}$  and  $b_1 = b(q\eta)^{1/q}$  in Young's inequality  $a_1b_1 \leq a_1^p/p + b_1^q/q$ .

**Lemma 3.3** (Well-posedness). Let  $\Omega \subset \mathbb{R}^n$  be a bounded open set. Let  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$  and  $m \in \mathbb{N}$ be such that 2s > m, and let  $a_\alpha \in M_0(H^{s-|\alpha|} \to H^{-s})$ . There exist a real number  $\mu > 0$  and a countable set  $\Sigma \subset (-\mu, \infty)$  of eigenvalues  $\lambda_1 \leq \lambda_2 \leq ... \to \infty$  such that if  $\lambda \in \mathbb{R} \setminus \Sigma$ , for any  $f \in H^s(\mathbb{R}^n)$  and  $F \in (\widetilde{H}^s(\Omega))^*$  there exists unique  $u \in H^s(\mathbb{R}^n)$  such that  $u - f \in \widetilde{H}^s(\Omega)$  and

$$B_P(u,v) - \lambda \langle u, v \rangle_{\Omega} = F(v) \text{ for all } v \in H^s(\Omega).$$

One has the estimate

$$||u||_{H^{s}(\mathbb{R}^{n})} \leq C\left(||f||_{H^{s}(\mathbb{R}^{n})} + ||F||_{(\widetilde{H}^{s}(\Omega))^{*}}\right).$$

The function u is also the unique  $u \in H^s(\mathbb{R}^n)$  satisfying

$$r_{\Omega}\left((-\Delta)^{s} + \sum_{|\alpha| \le m} a_{\alpha}D^{\alpha} - \lambda\right)u = F$$

in the sense of distributions in  $\Omega$  and  $u - f \in \widetilde{H}^{s}(\Omega)$ . Moreover, if (14) holds then  $0 \notin \Sigma$ .

Proof. Let  $\tilde{u} := u - f$ . The above problem is reduced to finding a unique  $\tilde{u} \in \tilde{H}^s(\Omega)$  such that  $B_P(\tilde{u}, v) - \lambda \langle \tilde{u}, v \rangle_{\Omega} = \tilde{F}(v)$ , where  $\tilde{F} := F - B_P(f, \cdot) + \lambda \langle f, \cdot \rangle_{\Omega}$ . Observe that the modified functional  $\tilde{F}$  belongs to  $(\tilde{H}^s(\Omega))^*$  as well, since by lemma 3.2 we have for all  $v \in \tilde{H}^s(\Omega)$ 

$$|\tilde{F}(v)| \le |F(v)| + |B_P(f,v)| + |\lambda| |\langle f,v \rangle_{\Omega}| \le (||F||_{(\tilde{H}^s(\Omega))^*} + (C+|\lambda|)||f||_{H^s(\mathbb{R}^n)}) ||v||_{H^s(\mathbb{R}^n)}.$$

Since  $a_{\alpha} \in M_0(H^{s-|\alpha|} \to H^{-s})$ , for any  $\epsilon > 0$  we can write  $a_{\alpha} = a_{\alpha,1} + a_{\alpha,2}$ , where  $a_{\alpha,1} \in C_c^{\infty}(\mathbb{R}^n) \cap M(H^{s-|\alpha|} \to H^{-s})$  and  $||a_{\alpha,2}||_{s-|\alpha|,-s} < \epsilon$ . Thus by formula (3), the continuity of the multiplication  $H^r(\mathbb{R}^n) \times H^s(\mathbb{R}^n) \hookrightarrow H^s(\mathbb{R}^n)$  for large enough  $r \in \mathbb{R}$  (see [3, Theorem 7.3]) and the fact that  $a_{\alpha,1} \in C_c^{\infty}(\mathbb{R}^n) \subset H^r(\mathbb{R}^n)$  for all  $r \in \mathbb{R}$  we obtain

$$\begin{aligned} |\langle a_{\alpha}, D^{\alpha}vw \rangle| &\leq |\langle a_{\alpha,1}, D^{\alpha}vw \rangle| + |\langle a_{\alpha,2}, D^{\alpha}vw \rangle| \\ &\leq \|a_{\alpha,1}\|_{H^{r}(\mathbb{R}^{n})} \|D^{\alpha}v\|_{H^{-s}(\mathbb{R}^{n})} \|w\|_{H^{s}(\mathbb{R}^{n})} + \|a_{\alpha,2}\|_{s-|\alpha|,-s} \|D^{\alpha}v\|_{H^{s-|\alpha|}(\mathbb{R}^{n})} \|w\|_{H^{s}(\mathbb{R}^{n})} \\ &\leq c \|w\|_{H^{s}(\mathbb{R}^{n})} \left( \|a_{\alpha,1}\|_{H^{r}(\mathbb{R}^{n})} \|v\|_{H^{|\alpha|-s}(\mathbb{R}^{n})} + \epsilon \|v\|_{H^{s}(\mathbb{R}^{n})} \right) \end{aligned}$$

where  $r \in \mathbb{R}$  is large enough  $(r > \max\{s, n/2\}$  is sufficient). If  $|\alpha| < s$ , from formulas (9) and (8) with p = q = 2 we get directly

(10) 
$$\begin{aligned} |\langle a_{\alpha}, D^{\alpha}vv\rangle| &\leq C\left(\|v\|_{H^{s}(\mathbb{R}^{n})}\|v\|_{L^{2}(\mathbb{R}^{n})} + \epsilon\|v\|_{H^{s}(\mathbb{R}^{n})}^{2}\right) \\ &\leq C(\epsilon^{-1}\|v\|_{L^{2}(\mathbb{R}^{n})}^{2} + \epsilon\|v\|_{H^{s}(\mathbb{R}^{n})}^{2}) \end{aligned}$$

for a constant C independent of  $v, w, \epsilon$ . If instead  $|\alpha| > s$  (observe that we can not have  $|\alpha| = s$ , because s can not be an integer), we use the interpolation inequality

$$\|v\|_{H^{|\alpha|-s}(\mathbb{R}^n)} \le C \|v\|_{L^2(\mathbb{R}^n)}^{1-(|\alpha|-s)/s} \|v\|_{H^s(\mathbb{R}^n)}^{(|\alpha|-s)/s} = C \|v\|_{L^2(\mathbb{R}^n)}^{2-|\alpha|/s} \|v\|_{H^s(\mathbb{R}^n)}^{|\alpha|/s-1}$$

in order to get

$$|\langle a_{\alpha}, D^{\alpha}vw\rangle| \leq C \|w\|_{H^{s}(\mathbb{R}^{n})} \left( \|v\|_{L^{2}(\mathbb{R}^{n})}^{2-|\alpha|/s} \|v\|_{H^{s}(\mathbb{R}^{n})}^{|\alpha|/s-1} + \epsilon \|v\|_{H^{s}(\mathbb{R}^{n})} \right).$$

Then by formula (8) with

$$a = \|v\|_{L^2(\mathbb{R}^n)}^{2-|\alpha|/s}, \quad b = \|v\|_{H^s(\mathbb{R}^n)}^{|\alpha|/s-1}, \quad p = \frac{s}{2s - |\alpha|}, \quad q = \frac{s}{|\alpha| - s}, \quad \eta = \epsilon$$

we obtain

$$|\langle a_{\alpha}, D^{\alpha}vw\rangle| \leq C \|w\|_{H^{s}(\mathbb{R}^{n})} \left(\epsilon^{\frac{s-|\alpha|}{2s-|\alpha|}} \|v\|_{L^{2}(\mathbb{R}^{n})} + \epsilon \|v\|_{H^{s}(\mathbb{R}^{n})}\right)$$

for a constant C independent of  $v, w, \epsilon$ . Now we use formula (8) again, but this time we choose

$$a = \|v\|_{L^2(\mathbb{R}^n)}, \quad b = \|v\|_{H^s(\mathbb{R}^n)}, \quad q = p = 2, \quad \eta = \epsilon^{s/(2s - |\alpha|)}.$$

This leads to

(11) 
$$\begin{aligned} |\langle a_{\alpha}, D^{\alpha}vv\rangle| &\leq C\left(\epsilon^{\frac{s-|\alpha|}{2s-|\alpha|}} \|v\|_{L^{2}(\mathbb{R}^{n})} \|v\|_{H^{s}(\mathbb{R}^{n})} + \epsilon \|v\|_{H^{s}(\mathbb{R}^{n})}^{2}\right) \\ &\leq C\left(\epsilon^{\frac{-|\alpha|}{2s-|\alpha|}} \|v\|_{L^{2}(\mathbb{R}^{n})}^{2} + 2\epsilon \|v\|_{H^{s}(\mathbb{R}^{n})}^{2}\right) \\ &\leq C\left(\epsilon^{\frac{-|\alpha|}{2s-|\alpha|}} \|v\|_{L^{2}(\mathbb{R}^{n})}^{2} + \epsilon \|v\|_{H^{s}(\mathbb{R}^{n})}^{2}\right) \\ &\leq C'\left(\epsilon^{\frac{-m}{2s-m}} \|v\|_{L^{2}(\mathbb{R}^{n})}^{2} + \epsilon \|v\|_{H^{s}(\mathbb{R}^{n})}^{2}\right) \end{aligned}$$

where C, C' are constants changing from line to line. Observe that C' can be taken independent of  $\alpha$ . Eventually, using (10) and (11) we get

(12) 
$$B_P(v,v) \ge \|(-\Delta)^{s/2}v\|_{L^2(\mathbb{R}^n)}^2 - \sum_{|\alpha| \le m} |\langle a_{\alpha}, D^{\alpha}vv\rangle| \\ \ge \|(-\Delta)^{s/2}v\|_{L^2(\mathbb{R}^n)}^2 - C'\left((\epsilon^{\frac{-m}{2s-m}} + \epsilon^{-1})\|v\|_{L^2(\mathbb{R}^n)}^2 + \epsilon\|v\|_{H^s(\mathbb{R}^n)}^2\right).$$

By the higher order Poincaré inequality (lemma 2.2) (24) turns into

$$B_{P}(v,v) \geq c \left( \|(-\Delta)^{s/2}v\|_{L^{2}(\mathbb{R}^{n})}^{2} + \|v\|_{L^{2}(\mathbb{R}^{n})}^{2} \right) - C' \left( (\epsilon^{\frac{-m}{2s-m}} + \epsilon^{-1}) \|v\|_{L^{2}(\mathbb{R}^{n})}^{2} + \epsilon \|v\|_{H^{s}(\mathbb{R}^{n})}^{2} \right)$$
$$\geq c \|v\|_{H^{s}(\mathbb{R}^{n})}^{2} - C' \left( (\epsilon^{\frac{-m}{2s-m}} + \epsilon^{-1}) \|v\|_{L^{2}(\mathbb{R}^{n})}^{2} + \epsilon \|v\|_{H^{s}(\mathbb{R}^{n})}^{2} \right)$$

for some constant  $c = c(\Omega, n, s)$  changing from line to line. For  $\epsilon$  small enough, this eventually gives the coercivity estimate

(13) 
$$B_P(v,v) \ge c_0 \|v\|_{H^s(\mathbb{R}^n)}^2 - \mu \|v\|_{L^2(\mathbb{R}^n)}^2$$

for some constants  $c_0, \mu > 0$  independent of v.

As a consequence of the coercivity estimate,  $B_P(\cdot, \cdot) + \mu \langle \cdot, \cdot \rangle_{L^2(\mathbb{R}^n)}$  is an equivalent inner product on  $\widetilde{H}^s(\Omega)$ . Therefore, by the Riesz representation theorem there exists a bounded linear operator  $G_{\mu} : (\widetilde{H}^s(\Omega))^* \to \widetilde{H}^s(\Omega)$  associating each functional in  $(\widetilde{H}^s(\Omega))^*$  to its unique representative in the inner product  $B_P(\cdot, \cdot) + \mu \langle \cdot, \cdot \rangle_{L^2(\mathbb{R}^n)}$  on  $\widetilde{H}^s(\Omega)$ . Thus  $\widetilde{u} := G_{\mu} \widetilde{F}$  verifies

$$B_P(\tilde{u}, v) + \mu \langle \tilde{u}, v \rangle_{L^2(\mathbb{R}^n)} = \tilde{F}(v) \text{ for all } v \in \tilde{H}^s(\Omega)$$

and it is the required unique solution  $\tilde{u} \in \tilde{H}^s(\Omega)$ . Moreover,  $G_{\mu}$  induces a compact, self-adjoint and positive operator  $\tilde{G}_{\mu} : L^2(\Omega) \to L^2(\Omega)$  by the compact Sobolev embedding theorem. The remaining claims follow from the spectral theorem for  $\tilde{G}_{\mu}$  and from the Fredholm alternative as in [15].

By the above lemma 3.3, both problems (4) and (5) have a countable set of Dirichlet eigenvalues. Throughout the paper we will assume that the coefficients  $a_{\alpha}$  are such that 0 is not a

Dirichlet eigenvalue for either of the problems. That is, we assume that

(14) 
$$\begin{cases} \text{if } u \in H^s(\mathbb{R}^n) \text{ solves } (-\Delta)^s u + \sum_{|\alpha| \le m} a_\alpha D^\alpha u = 0 \text{ in } \Omega \text{ and } u|_{\Omega_e} = 0, \\ \text{then } u \equiv 0 \end{cases}$$

and

(15) 
$$\begin{cases} \text{if } u^* \in H^s(\mathbb{R}^n) \text{ solves } (-\Delta)^s u^* + \sum_{|\alpha| \le m} (-1)^{|\alpha|} D^{\alpha}(a_{\alpha}u^*) = 0 \text{ in } \Omega \text{ and } u^*|_{\Omega_e} = 0, \\ \text{then } u^* \equiv 0. \end{cases}$$

With this in mind, we shall define the DN maps. Consider the abstract trace space  $X := H^s(\mathbb{R}^n)/\widetilde{H}^s(\Omega)$  equipped with the quotient norm

$$\|[f]\|_X := \inf_{\phi \in \widetilde{H}^s(\Omega)} \|f - \phi\|_{H^s(\mathbb{R}^n)}, \quad f \in H^s(\mathbb{R}^n)$$

and its dual space  $X^*$ . We use these in order to define the DN maps associated to the problems (4) and (5), which we study in the following lemma.

**Lemma 3.4** (DN maps). Let  $\Omega \subset \mathbb{R}^n$  be a bounded open set. Let  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$  and  $m \in \mathbb{N}$  such that 2s > m, and let  $a_\alpha \in M_0(H^{s-|\alpha|} \to H^{-s})$ . There exist two continuous linear maps

$$\Lambda_P: X \to X^*$$
 defined by  $\langle \Lambda_P[f], [g] \rangle := B_P(u_f, g)$ 

and

$$\Lambda_P^*: X \to X^*$$
 defined by  $\langle \Lambda_P^*[f], [g] \rangle := B_P^*(u_f^*, g)$   
the unique solutions to the equations

where  $u_f, u_f^*$  are the unique solutions to the equations

$$(-\Delta)^{s}u + \sum_{|\alpha| \le m} a_{\alpha}D^{\alpha}u = 0 \quad in \ \Omega, \quad u - f \in \tilde{H}^{s}(\Omega)$$

and

$$(-\Delta)^{s}u^{*} + \sum_{|\alpha| \le m} (-1)^{|\alpha|} D^{\alpha}(a_{\alpha}u^{*}) = 0 \quad in \ \Omega, \quad u^{*} - f \in \widetilde{H}^{s}(\Omega)$$

with  $f, g \in H^s(\mathbb{R}^n)$ . Moreover, the identity  $\langle \Lambda_P[f], [g] \rangle = \langle [f], \Lambda_P^*[g] \rangle$  holds.

*Proof.* We show well-definedness and continuity only for  $\Lambda_P$ , the proof being similar for  $\Lambda_P^*$ . We note that such unique solutions exist by lemma 3.3.

If  $\phi \in H^s(\Omega)$ , then  $u_f|_{\Omega_e} = f = u_{f+\phi}|_{\Omega_e}$ , and also  $u_f$ ,  $u_{f+\phi}$  both solve  $(-\Delta)^s u + Pu = 0$  in  $\Omega$ . By unicity of solutions, we must then have that  $u_f$  and  $u_{f+\phi}$  coincide. On the other hand, if  $\psi \in \widetilde{H}^s(\Omega)$ , then  $\psi|_{\Omega_e} = 0$ . These two facts imply the well-definedness of  $\Lambda_P$ , since

$$B_P(u_{f+\phi}, g+\psi) = B_P(u_f, g) + B_P(u_f, \psi) = B_P(u_f, g).$$

The continuity of  $\Lambda_P$  is an easy consequence of lemma 3.2 and the estimate in lemma 3.3. If  $f, g \in H^s(\mathbb{R}^n)$  and  $\phi, \psi \in \widetilde{H}^s(\Omega)$ , then

$$|\langle \Lambda_P[f], [g] \rangle| = |B_P(u_{f-\phi}, g-\psi)| \le C ||u_{f-\phi}||_{H^s} ||g-\psi||_{H^s} \le C ||f-\phi||_{H^s} ||g-\psi||_{H^s}.$$

By taking the infimum on both sides with respect to  $\phi$  and  $\psi$ , we end up with

$$|\langle \Lambda_P[f], [g] \rangle| \le C \inf_{\phi \in \widetilde{H}^s(\Omega)} \|f - \phi\|_{H^s} \inf_{\psi \in \widetilde{H}^s(\Omega)} \|g - \psi\|_{H^s} = C \|[f]\|_X \|[g]\|_X.$$

The well-posedness result proved above implies that for all  $f, g \in H^s(\mathbb{R}^n)$  we have  $\langle \Lambda_P[f], [g] \rangle = B_P(u_f, e_g)$ , where  $e_g$  is a generic extension of  $g|_{\Omega_e}$  from  $\Omega_e$  to  $\mathbb{R}^n$ . In particular,  $\langle \Lambda_P[f], [g] \rangle = B_P(u_f, u_g^*)$ . By lemma 3.2 this leads to

$$\langle \Lambda_P[f], [g] \rangle = B_P(u_f, u_g^*) = B_P^*(u_g^*, u_f) = \langle \Lambda_P^*[g], [f] \rangle$$

which conlcudes the proof.

**Remark 3.5.** We should observe at this point that a priori  $\Lambda_P^*$  has no reason to be the adjoint of  $\Lambda_P$ , as the symbols would suggest. However, the identity we proved in lemma 3.4 shows that this is in fact true, and thus there is no abuse of notation.

3.2. **Proof of injectivity.** The proof of injectivity is based on an Alessandrini identity and the Runge approximation property for our operator, following the scheme developed in [15].

**Lemma 3.6** (Alessandrini identity). Let  $\Omega \subset \mathbb{R}^n$  be a bounded open set. Let  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$  and  $m \in \mathbb{N}$  such that 2s > m. For j = 1, 2, let  $a_{j,\alpha} \in M_0(H^{s-|\alpha|} \to H^{-s})$ . For any  $f_1, f_2 \in H^s(\mathbb{R}^n)$ , let  $u_1, u_2^* \in H^s(\mathbb{R}^n)$  respectively solve

$$(-\Delta)^{s}u_{1} + \sum_{|\alpha| \le m} a_{1,\alpha} D^{\alpha} u_{1} = 0 \quad in \ \Omega, \quad u_{1} - f_{1} \in \widetilde{H}^{s}(\Omega)$$

and

$$(-\Delta)^{s} u_{2}^{*} + \sum_{|\alpha| \le m} (-1)^{|\alpha|} D^{\alpha}(a_{2,\alpha} u_{2}^{*}) = 0 \quad in \ \Omega, \quad u_{2}^{*} - f_{2} \in \widetilde{H}^{s}(\Omega).$$

Then we have the integral identity

$$\langle (\Lambda_{P_1} - \Lambda_{P_2})[f_1], [f_2] \rangle = \sum_{|\alpha| \le m} \langle (a_{1,\alpha} - a_{2,\alpha}), D^{\alpha} u_1 u_2^* \rangle.$$

*Proof.* The proof is a simple computation following from lemma 3.4

$$\langle (\Lambda_{P_1} - \Lambda_{P_2})[f_1], [f_2] \rangle = \langle \Lambda_{P_1}[f_1], [f_2] \rangle - \langle \Lambda_{P_2}[f_1], [f_2] \rangle = \langle \Lambda_{P_1}[f_1], [f_2] \rangle - \langle [f_1], \Lambda_{P_2}^*[f_2] \rangle$$

$$= B_{P_1}(u_1, u_2^*) - B_{P_2}^*(u_2^*, u_1) = \sum_{|\alpha| \le m} \langle (a_{1,\alpha} - a_{2,\alpha}), D^{\alpha}u_1u_2^* \rangle.$$

**Lemma 3.7** (Runge approximation property). Let  $\Omega, W \subset \mathbb{R}^n$  respectively be a bounded open set and a non-empty open set such that  $\overline{W} \cap \overline{\Omega} = \emptyset$ . Let  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$  and  $m \in \mathbb{N}$  be such that 2s > m, and let  $a_\alpha \in M_0(H^{s-|\alpha|} \to H^{-s})$ . Moreover, let  $\mathcal{R} := \{u_f - f : f \in C_c^\infty(W)\} \subset \widetilde{H}^s(\Omega)$ where  $u_f$  solves

$$(-\Delta)^{s}u_{f} + \sum_{|\alpha| \le m} a_{\alpha}D^{\alpha}u_{f} = 0 \quad in \ \Omega, \quad u_{f} - f \in \widetilde{H}^{s}(\Omega)$$

and  $\mathcal{R}^* := \{ u_f^* - f : f \in C_c^{\infty}(W) \} \subset \widetilde{H}^s(\Omega) \text{ where } u_f^* \text{ solves }$ 

$$(-\Delta)^{s} u_{f}^{*} + \sum_{|\alpha| \le m} (-1)^{|\alpha|} D^{\alpha}(a_{\alpha} u_{f}^{*}) = 0 \quad in \ \Omega, \quad u_{f}^{*} - f \in \widetilde{H}^{s}(\Omega).$$

Then  $\mathcal{R}$  and  $\mathcal{R}^*$  are dense in  $\widetilde{H}^s(\Omega)$ .

*Proof.* The proofs of the two statements are similar, so we show only the density of  $\mathcal{R}$  in  $\widetilde{H}^{s}(\Omega)$ . By the Hahn-Banach theorem, it is enough to prove that any functional F acting on  $\widetilde{H}^{s}(\Omega)$  that vanishes on  $\mathcal{R}$  must be identically 0. Thus, let  $F \in (\widetilde{H}^{s}(\Omega))^{*}$  and assume  $F(u_{f} - f) = 0$  for all  $f \in C_{c}^{\infty}(W)$ . Let  $\phi$  be the unique solution of

(16) 
$$(-\Delta)^{s}\phi + \sum_{|\alpha| \le m} (-1)^{|\alpha|} D^{\alpha}(a_{\alpha}\phi) = -F \quad \text{in } \Omega, \quad \phi \in \widetilde{H}^{s}(\Omega).$$

In other words,  $\phi$  is the unique function in  $\widetilde{H}^{s}(\Omega)$  such that  $B_{P}^{*}(\phi, w) = -F(w)$  for all  $w \in \widetilde{H}^{s}(\Omega)$ . Then we can compute

(17)  
$$0 = F(u_f - f) = -B_P^*(\phi, u_f - f) = B_P^*(\phi, f)$$
$$= \langle (-\Delta)^{s/2} f, (-\Delta)^{s/2} \phi \rangle + \sum_{|\alpha| \le m} \langle a_{\alpha}, D^{\alpha} f \phi \rangle$$
$$= \langle f, (-\Delta)^s \phi \rangle.$$

On the first line of (17) we used that  $\phi \in \widetilde{H}^s(\Omega)$  and  $u_f$  solves the equation in  $\Omega$ , and on the last line we used the support condition for f. By the arbitrariety of  $f \in C_c^{\infty}(W)$  we have obtained that  $(-\Delta)^s \phi = 0$  in W, and on the same set we also have  $\phi = 0$ . Using the unique continuation result for the higher order fractional Laplacian given in lemma 2.1 we deduce  $\phi \equiv 0$  on all of  $\mathbb{R}^n$ . The vanishing of the functional F now follows easily from the definition of  $\phi$ .

**Remark 3.8.** We remark that using the same proof one can show that  $r_{\Omega}\mathcal{R} \subset L^2(\Omega)$  and  $r_{\Omega}\mathcal{R}^* \subset L^2(\Omega)$  are dense in  $L^2(\Omega)$ , where  $r_{\Omega}$  is the restriction to  $\Omega$ . If  $F \in L^2(\Omega)$ , then F induces an element in  $(\tilde{H}^s(\Omega))^*$  via the inner product  $F(w) := \langle F, r_{\Omega}w \rangle_{\Omega}$ , where  $w \in \tilde{H}^s(\Omega)$ . Hence one can choose the solution  $\phi$  in equation (16) with F as a source term and complete the proof as in equation (17) showing that  $(r_{\Omega}\mathcal{R})^{\perp} = \{0\}$  in  $L^2(\Omega)$  (similarly  $(r_{\Omega}\mathcal{R}^*)^{\perp} = \{0\}$ ).

We are ready to prove the main result of the paper.

Proof of theorem 1.1. Step 1. Since one can always shrink the sets  $W_1$  and  $W_2$  if necessary, we can assume without loss of generality that  $\overline{W_1} \cap \overline{W_2} = \emptyset$ . Let  $v_1, v_2 \in C_c^{\infty}(\Omega)$ . By the Runge approximation property proved in lemma 3.7 we can find two sequences of functions  $\{f_{j,k}\}_{k\in\mathbb{N}} \subset C_c^{\infty}(W_j), j = 1, 2$ , such that

$$u_{1,k} = f_{1,k} + v_1 + r_{1,k}, \quad u_{2,k}^* = f_{2,k} + v_2 + r_{2,k}$$

where  $u_{1,k}, u_{2,k}^* \in \widetilde{H}^s(\Omega)$  respectively solve

$$(-\Delta)^s u_{1,k} + \sum_{|\alpha| \le m} a_{1,\alpha} D^{\alpha} u_{1,k} = 0 \quad \text{in } \Omega, \quad u_{1,k} - f_{1,k} \in \widetilde{H}^s(\Omega)$$

and

$$(-\Delta)^{s} u_{2,k}^{*} + \sum_{|\alpha| \le m} (-1)^{|\alpha|} D^{\alpha}(a_{2,\alpha} u_{2,k}^{*}) = 0 \quad \text{in } \Omega, \quad u_{2,k}^{*} - f_{2,k} \in \widetilde{H}^{s}(\Omega)$$

and  $r_{1,k}$ ,  $r_{2,k} \to 0$  in  $\widetilde{H}^s(\Omega)$  as  $k \to \infty$ . By the assumption on the DN maps and the Alessandrini identity from lemma 3.6 we have

(18) 
$$0 = \langle (\Lambda_{P_1} - \Lambda_{P_2})[f_{1,k}], [f_{2,k}] \rangle = \sum_{|\alpha| \le m} \langle (a_{1,\alpha} - a_{2,\alpha}), D^{\alpha} u_{1,k} u_{2,k}^* \rangle$$
$$= \sum_{|\alpha| \le m} \langle (a_{1,\alpha} - a_{2,\alpha}), D^{\alpha} r_{1,k} u_{2,k}^* \rangle + \sum_{|\alpha| \le m} \langle (a_{1,\alpha} - a_{2,\alpha}), D^{\alpha} v_1 r_{2,k} \rangle$$
$$+ \sum_{|\alpha| \le m} \langle (a_{1,\alpha} - a_{2,\alpha}), D^{\alpha} v_1 v_2 \rangle.$$

However, for the first two terms on the right hand side of (18) we can deduce

$$\begin{aligned} \left| \sum_{|\alpha| \le m} \langle (a_{1,\alpha} - a_{2,\alpha}), D^{\alpha} r_{1,k} u_{2,k}^* \rangle \right| &\le \sum_{|\alpha| \le m} |\langle (a_{1,\alpha} - a_{2,\alpha}), D^{\alpha} r_{1,k} u_{2,k}^* \rangle| \\ &\le C \|u_{2,k}^*\|_{H^s} \|r_{1,k}\|_{H^s} \sum_{|\alpha| \le m} \|a_{1,\alpha} - a_{2,\alpha}\|_{s-|\alpha|,-s} \to 0 \end{aligned}$$

and

$$\begin{split} |\sum_{|\alpha| \le m} \langle (a_{1,\alpha} - a_{2,\alpha}), D^{\alpha} v_{1} r_{2,k} \rangle | \le \sum_{|\alpha| \le m} |\langle (a_{1,\alpha} - a_{2,\alpha}), D^{\alpha} v_{1} r_{2,k} \rangle | \\ \le C \|r_{2,k}\|_{H^{s}} \|v_{1}\|_{H^{s}} \sum_{|\alpha| \le m} \|a_{1,\alpha} - a_{2,\alpha}\|_{s - |\alpha|, -s} \to 0 \end{split}$$

as  $k \to \infty$ . Thus by taking the limit in formula (18) we obtain

(19) 
$$\sum_{|\alpha| \le m} \langle (a_{1,\alpha} - a_{2,\alpha}), D^{\alpha} v_1 v_2 \rangle = 0 \quad \text{for all} \quad v_1, v_2 \in C_c^{\infty}(\Omega).$$

**Step 2.** Assume that we have  $a_{1,\alpha}|_{\Omega} = a_{2,\alpha}|_{\Omega}$  for all  $\alpha$  such that  $|\alpha| < N$  for some  $N \in \mathbb{N}$ . We show that the equality of the coefficients also holds for  $\alpha$  for which  $|\alpha| = N$  and this will prove the theorem by the principle of complete induction.

To this end, consider  $v_2 \in C_c^{\infty}(\Omega)$ , and then take  $v_1 \in C_c^{\infty}(\Omega)$  such that  $v_1(x) = x^{\alpha}$ on  $\operatorname{supp}(v_2) \Subset \Omega$ . Recall that since  $\alpha = (\alpha_1, \alpha_2, ..., \alpha_n) \in \mathbb{N}^n$  is a multi-index and  $x = (x_1, x_2, ..., x_n) \in \mathbb{R}^n$ , the symbol  $x^{\alpha}$  is intended to mean  $x_1^{\alpha_1} x_2^{\alpha_2} \dots x_n^{\alpha_n}$ . With this choice of  $v_1, v_2$ , equation (19) becomes

$$(20) \qquad 0 = \sum_{|\beta| \le m} \langle (a_{1,\beta} - a_{2,\beta}), D^{\beta} v_1 v_2 \rangle = \sum_{N \le |\beta| \le m} \langle (a_{1,\beta} - a_{2,\beta}), D^{\beta} (x^{\alpha}) v_2 \rangle$$
$$= \sum_{N < |\beta| \le m} \langle (a_{1,\beta} - a_{2,\beta}), D^{\beta} (x^{\alpha}) v_2 \rangle + \sum_{|\beta| = N, \beta \ne \alpha} \langle (a_{1,\beta} - a_{2,\beta}), D^{\beta} (x^{\alpha}) v_2 \rangle$$
$$+ \langle (a_{1,\alpha} - a_{2,\alpha}), D^{\alpha} (x^{\alpha}) v_2 \rangle.$$

If  $|\beta| > N = |\alpha|$ , then there must exist  $k \in \{1, 2, ..., n\}$  such that  $\beta_k > \alpha_k$ . This is true also if  $|\beta| = N$  with  $\beta \neq \alpha$ . In both cases we can compute

$$D^{\beta}(x^{\alpha}) = (\partial_{x_1}^{\beta_1} x_1^{\alpha_1}) \ (\partial_{x_2}^{\beta_2} x_2^{\alpha_2}) \ \dots \ (\partial_{x_n}^{\beta_n} x_n^{\alpha_n}) = 0$$

because  $\partial_{x_k}^{\beta_k} x_k^{\alpha_k} = 0$ . Therefore formula (20) becomes

$$0 = \langle (a_{1,\alpha} - a_{2,\alpha}), D^{\alpha}(x^{\alpha})v_2 \rangle_{\mathbb{R}^n} = \alpha! \langle a_{1,\alpha} - a_{2,\alpha}, v_2 \rangle_{\mathbb{R}^n}$$

which by the arbitrariety of  $v_2 \in C_c^{\infty}(\Omega)$  implies  $a_{1,\alpha}|_{\Omega} = a_{2,\alpha}|_{\Omega}$  also for  $\alpha$  for which  $|\alpha| = N$ .

**Step 3.** We have proved that  $a_{1,\alpha}|_{\Omega} = a_{2,\alpha}|_{\Omega}$  for all  $\alpha$  of the order  $|\alpha| \leq m$ . Since this entails  $P_1|_{\Omega} = P_2|_{\Omega}$ , the proof is complete.

## 4. Main theorem for bounded coefficients

We shall now study the case when the coefficients of PDOs are from the bounded spaces  $H^{r_{\alpha},\infty}(\Omega)$ . It should be noted, however, that most of the considerations of the previous section still apply identically.

4.1. Well-posedness of the inverse problem. We shall define the bilinear forms for the problems (4) and (5) respectively by (6) and (7), just as in the case of singular coefficients. These will turn out to be bounded in  $H^s(\mathbb{R}^n) \times H^s(\mathbb{R}^n)$  as well, but the proof we give of this fact is a fortiori different. Since now we assume that  $a_{\alpha} \in H^{r_{\alpha},\infty}(\Omega) \subset L^{\infty}(\Omega)$  for  $r_{\alpha} \geq 0$ , the duality pairing  $\langle a_{\alpha}, D^{\alpha}vw \rangle$  becomes an inner product over  $\Omega$  and we write  $\langle a_{\alpha}(x)D^{\alpha}v,w \rangle_{\Omega}$  to emphasize that the coefficients  $a_{\alpha} = a_{\alpha}(x)$  are now functions defined in  $\Omega$ .

**Lemma 4.1** (Boundedness of the bilinear forms). Let  $\Omega \subset \mathbb{R}^n$  be a bounded Lipschitz domain and  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$ ,  $m \in \mathbb{N}$  such that 2s > m. Let  $a_\alpha \in H^{r_\alpha,\infty}(\Omega)$ , with  $r_\alpha$  defined as in (2). Then  $B_P$  and  $B_P^*$  extend as bounded bilinear forms on  $H^s(\mathbb{R}^n) \times H^s(\mathbb{R}^n)$ .

**Remark 4.2.** Since  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$  and  $|\alpha| \leq m < 2s$ , we also have that  $\max(0, |\alpha| - s) \leq r_\alpha < s$  for  $\delta > 0$  small (see equation (2)).

Proof of lemma 4.1. We only prove the boundedness of  $B_P$ , as for  $B_P^*$  one can proceed in the same way. If  $v, w \in C_c^{\infty}(\mathbb{R}^n)$ , then

$$|\langle a_{\alpha}(x)D^{\alpha}v,w\rangle_{\Omega}| = \left|\int_{\Omega} a_{\alpha}wD^{\alpha}v\,dx\right| \le ||a_{\alpha}w||_{(H^{-r_{\alpha}}(\Omega))^{*}}||D^{\alpha}v||_{H^{-r_{\alpha}}(\Omega)}.$$

Since  $\Omega$  is a Lipschitz domain and  $r_{\alpha} > -1/2$ ,  $r_{\alpha} \notin \{\frac{1}{2}, \frac{3}{2}, \frac{5}{2}...\}$ , we have  $(H^{-r_{\alpha}}(\Omega))^* = H_0^{r_{\alpha}}(\Omega) \subset H^{r_{\alpha}}(\Omega)$ . Therefore

$$(21) \quad |\langle a_{\alpha}(x)D^{\alpha}v,w\rangle_{\Omega}| \leq C ||a_{\alpha}w||_{H^{r_{\alpha}}(\Omega)} ||D^{\alpha}v||_{H^{-r_{\alpha}}(\Omega)} \leq C ||A_{\alpha}w||_{H^{r_{\alpha}}(\mathbb{R}^{n})} ||D^{\alpha}v||_{H^{-r_{\alpha}}(\Omega)}$$
$$\leq C ||J^{r_{\alpha}}(A_{\alpha}w)||_{L^{2}(\mathbb{R}^{n})} ||v||_{H^{|\alpha|-r_{\alpha}}(\Omega)}$$

where  $J = (\mathrm{Id} - \Delta)^{1/2}$  is the Bessel potential and  $A_{\alpha}$  is an extension of  $a_{\alpha}$  from  $\Omega$  to  $\mathbb{R}^n$  such that  $A_{\alpha}|_{\Omega} = a_{\alpha}$  and  $||A_{\alpha}||_{H^{r_{\alpha},\infty}(\mathbb{R}^n)} \leq 2||a_{\alpha}||_{H^{r_{\alpha},\infty}(\Omega)}$ . Since  $r_{\alpha} \geq 0$ , we may estimate the last term of (21) by the Kato-Ponce inequality given in lemma 2.3

$$\begin{aligned} \|J^{r_{\alpha}}(A_{\alpha}w)\|_{L^{2}(\mathbb{R}^{n})} &\leq C\left(\|A_{\alpha}\|_{L^{\infty}(\mathbb{R}^{n})}\|J^{r_{\alpha}}w\|_{L^{2}(\mathbb{R}^{n})} + \|J^{r_{\alpha}}A_{\alpha}\|_{L^{\infty}(\mathbb{R}^{n})}\|w\|_{L^{2}(\mathbb{R}^{n})}\right) \\ &\leq C\|A_{\alpha}\|_{H^{r_{\alpha},\infty}(\mathbb{R}^{n})}\|w\|_{H^{r_{\alpha}}(\mathbb{R}^{n})} \leq C\|a_{\alpha}\|_{H^{r_{\alpha},\infty}(\Omega)}\|w\|_{H^{r_{\alpha}}(\mathbb{R}^{n})}.\end{aligned}$$

Substituting this into (21) gives

(22) 
$$\begin{aligned} |\langle a_{\alpha}(x)D^{\alpha}v,w\rangle_{\Omega}| &\leq C ||a_{\alpha}||_{H^{r_{\alpha},\infty}(\Omega)} ||w||_{H^{r_{\alpha}}(\mathbb{R}^{n})} ||v||_{H^{|\alpha|-r_{\alpha}}(\Omega)} \\ &\leq C ||a_{\alpha}||_{H^{r_{\alpha},\infty}(\Omega)} ||w||_{H^{s}(\mathbb{R}^{n})} ||v||_{H^{s}(\mathbb{R}^{n})} \end{aligned}$$

given that both  $r_{\alpha} < s$  and  $|\alpha| - r_{\alpha} \leq s$  hold by remark 4.2. Eventually we obtain

$$|B_P(v,w)| \leq |\langle (-\Delta)^{s/2}v, (-\Delta)^{s/2}w\rangle_{\mathbb{R}^n}| + \sum_{|\alpha|\leq m} |\langle a_\alpha D^\alpha v, w\rangle_{\mathbb{R}^n}|$$
  
$$\leq ||w||_{H^s(\mathbb{R}^n)} ||v||_{H^s(\mathbb{R}^n)} + \sum_{|\alpha|\leq m} C ||a_\alpha||_{H^{r_\alpha,\infty}(\Omega)} ||w||_{H^s(\mathbb{R}^n)} ||v||_{H^s(\mathbb{R}^n)}$$
  
$$\leq C ||w||_{H^s(\mathbb{R}^n)} ||v||_{H^s(\mathbb{R}^n)}.$$

Next we shall prove existence and uniqueness of solutions for the problems (4) and (5). The reasoning is similar to the one for the proof of lemma 3.3, but the details of the computations are quite different.

**Lemma 4.3** (Well-posedness). Let  $\Omega \subset \mathbb{R}^n$  be a bounded Lipschitz domain and  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$ ,  $m \in \mathbb{N}$  such that 2s > m. Let  $a_{\alpha} \in H^{r_{\alpha},\infty}(\Omega)$ , with  $r_{\alpha}$  defined as in (2). There exist a real number  $\mu > 0$  and a countable set  $\Sigma \subset (-\mu, \infty)$  of eigenvalues  $\lambda_1 \leq \lambda_2 \leq ... \rightarrow \infty$  such that if  $\lambda \in \mathbb{R} \setminus \Sigma$ , for any  $f \in H^s(\mathbb{R}^n)$  and  $F \in (\widetilde{H}^s(\Omega))^*$  there exists a unique  $u \in H^s(\mathbb{R}^n)$  such that  $u - f \in \widetilde{H}^{s}(\Omega)$  and

$$B_P(u,v) - \lambda \langle u, v \rangle_{\Omega} = F(v) \text{ for all } v \in \widetilde{H}^s(\Omega).$$

One has the estimate

$$\|u\|_{H^s(\mathbb{R}^n)} \le C\left(\|f\|_{H^s(\mathbb{R}^n)} + \|F\|_{(\widetilde{H}^s(\Omega))^*}\right).$$

The function u is also the unique  $u \in H^s(\mathbb{R}^n)$  satisfying

$$r_{\Omega}\left((-\Delta)^{s} + \sum_{|\alpha| \le m} a_{\alpha}(x)D^{\alpha} - \lambda\right)u = F$$

in the sense of distributions in  $\Omega$  and  $u - f \in \widetilde{H}^{s}(\Omega)$ . Moreover, if (14) holds then  $0 \notin \Sigma$ .

*Proof.* Again it is enough to find unique  $\tilde{u} \in \tilde{H}^s(\Omega)$  such that  $B_P(\tilde{u}, v) - \lambda \langle \tilde{u}, v \rangle_{\Omega} = \tilde{F}(v)$ , where  $\tilde{F} := F - B_P(f, \cdot) + \lambda \langle f, \cdot \rangle_{\Omega}$ . Consider  $v, w \in C_c^{\infty}(\Omega)$  and  $r_{\alpha} \neq 0$ . Since  $0 < r_{\alpha} < s$ , the interpolation inequality

$$||w||_{H^{r_{\alpha}}(\mathbb{R}^{n})} \leq C ||w||_{L^{2}(\mathbb{R}^{n})}^{1-r_{\alpha}/s} ||w||_{H^{s}(\mathbb{R}^{n})}^{r_{\alpha}/s}$$

holds. Using this and formula (22) we get, for a constant  $C = C(\Omega, n, s, r_{\alpha})$  which may change from line to line,

$$\begin{aligned} (23) \quad |\langle a_{\alpha}(x)D^{\alpha}v,w\rangle_{\Omega}| &\leq C ||a_{\alpha}||_{H^{r_{\alpha},\infty}(\Omega)} ||v||_{H^{s}(\mathbb{R}^{n})} ||w||_{H^{r_{\alpha}}(\mathbb{R}^{n})} \\ &\leq C ||a_{\alpha}||_{H^{r_{\alpha},\infty}(\Omega)} ||v||_{H^{s}(\mathbb{R}^{n})} ||w||_{L^{2}(\mathbb{R}^{n})}^{1-r_{\alpha}/s} ||w||_{H^{s}(\mathbb{R}^{n})} \\ &\leq ||a_{\alpha}||_{H^{r_{\alpha},\infty}(\Omega)} ||v||_{H^{s}(\mathbb{R}^{n})} \left(C\epsilon^{r_{\alpha}/(r_{\alpha}-s)} ||w||_{L^{2}(\mathbb{R}^{n})} + \epsilon ||w||_{H^{s}(\mathbb{R}^{n})}\right). \end{aligned}$$

In the last step of (23) we used formula (8) with

$$q = \frac{s}{r_{\alpha}}, \quad p = \frac{s}{s - r_{\alpha}}, \quad b = \|w\|_{H^{s}(\mathbb{R}^{n})}^{r_{\alpha}/s}, \quad a = C\|w\|_{L^{2}(\mathbb{R}^{n})}^{1 - r_{\alpha}/s}, \quad \eta = \epsilon.$$

If instead  $r_{\alpha} = 0$ , just by formula (22) we already have

$$|\langle a_{\alpha}(x)D^{\alpha}v,w\rangle_{\Omega}| \leq C ||a_{\alpha}||_{L^{\infty}(\Omega)} ||v||_{H^{s}(\mathbb{R}^{n})} ||w||_{L^{2}(\mathbb{R}^{n})}.$$

Moreover, the two estimates above also hold for  $v, w \in \widetilde{H}^{s}(\Omega)$  by the density of  $C_{c}^{\infty}(\Omega)$  in  $\widetilde{H}^{s}(\Omega)$ . Now we use formula (8) again, but this time we choose

$$q = p = 2, \quad b = \|v\|_{H^s(\mathbb{R}^n)}, \quad a = \|v\|_{L^2(\mathbb{R}^n)}, \quad \eta = \epsilon^{s/(s-r_\alpha)}.$$

This leads to

$$\begin{aligned} |\langle a_{\alpha}(x)D^{\alpha}v,v\rangle_{\Omega}| &\leq \|a_{\alpha}\|_{H^{r_{\alpha},\infty}(\Omega)}\|v\|_{H^{s}(\mathbb{R}^{n})} \left(C\epsilon^{r_{\alpha}/(r_{\alpha}-s)}\|v\|_{L^{2}(\mathbb{R}^{n})} + \epsilon\|v\|_{H^{s}(\mathbb{R}^{n})}\right) \\ &= \|a_{\alpha}\|_{H^{r_{\alpha},\infty}(\Omega)} \left(C\epsilon^{r_{\alpha}/(r_{\alpha}-s)}\|v\|_{L^{2}(\mathbb{R}^{n})}\|v\|_{H^{s}(\mathbb{R}^{n})} + \epsilon\|v\|_{H^{s}(\mathbb{R}^{n})}^{2}\right) \\ &\leq \|a_{\alpha}\|_{H^{r_{\alpha},\infty}(\Omega)} \left(C\epsilon^{\frac{r_{\alpha}+s}{r_{\alpha}-s}}\|v\|_{L^{2}(\mathbb{R}^{n})}^{2} + \epsilon(C+1)\|v\|_{H^{s}(\mathbb{R}^{n})}^{2}\right) \\ &\leq C\|a_{\alpha}\|_{H^{r_{\alpha},\infty}(\Omega)} \left(\epsilon^{\frac{r_{\alpha}+s}{r_{\alpha}-s}}\|v\|_{L^{2}(\mathbb{R}^{n})}^{2} + \epsilon\|v\|_{H^{s}(\mathbb{R}^{n})}^{2}\right) \\ &\leq C'\|a_{\alpha}\|_{H^{r_{\alpha},\infty}(\Omega)} \left(\epsilon^{\frac{M+s}{M-s}}\|v\|_{L^{2}(\mathbb{R}^{n})}^{2} + \epsilon\|v\|_{H^{s}(\mathbb{R}^{n})}^{2}\right) \end{aligned}$$

where  $C = C(\Omega, n, s, r_{\alpha})$  and  $C' = C'(\Omega, n, s)$  are constants changing from line to line and  $M \in [0, s)$  is defined by  $M := \max_{|\alpha| \le m} r_{\alpha}$ . Eventually

$$(24) \quad B_P(v,v) \ge \|(-\Delta)^{s/2}v\|_{L^2(\mathbb{R}^n)}^2 - \sum_{|\alpha| \le m} |\langle a_\alpha(x)D^\alpha v, v\rangle_{\Omega}| \ge \|(-\Delta)^{s/2}v\|_{L^2(\mathbb{R}^n)}^2 - C'\left(\epsilon^{\frac{M+s}{M-s}} \|v\|_{L^2(\mathbb{R}^n)}^2 + \epsilon \|v\|_{H^s(\mathbb{R}^n)}^2\right) \sum_{|\alpha| \le m} \|a_\alpha\|_{H^{r_\alpha,\infty}(\Omega)} = \|(-\Delta)^{s/2}v\|_{L^2(\mathbb{R}^n)}^2 - C'C''\left(\epsilon^{\frac{M+s}{M-s}} \|v\|_{L^2(\mathbb{R}^n)}^2 + \epsilon \|v\|_{H^s(\mathbb{R}^n)}^2\right)$$

where  $C'' := \sum_{|\alpha| \le m} ||a_{\alpha}||_{H^{r_{\alpha},\infty}(\Omega)}$  is a constant independent of  $\epsilon$  and v. By the higher order Poincaré inequality (lemma 2.2) (24) turns into

$$B_{P}(v,v) \geq c \left( \| (-\Delta)^{s/2} v \|_{L^{2}(\mathbb{R}^{n})}^{2} + \| v \|_{L^{2}(\mathbb{R}^{n})}^{2} \right) - C'C'' \left( \epsilon^{\frac{M+s}{M-s}} \| v \|_{L^{2}(\mathbb{R}^{n})}^{2} + \epsilon \| v \|_{H^{s}(\mathbb{R}^{n})}^{2} \right)$$
$$\geq c \| v \|_{H^{s}(\mathbb{R}^{n})}^{2} - C'C'' \left( \epsilon^{\frac{M+s}{M-s}} \| v \|_{L^{2}(\mathbb{R}^{n})}^{2} + \epsilon \| v \|_{H^{s}(\mathbb{R}^{n})}^{2} \right)$$

for some constant  $c = c(\Omega, n, s)$  changing from line to line. For  $\epsilon$  small enough (notice that M - s < 0), this eventually gives the coercivity estimate

(25) 
$$B_P(v,v) \ge c_0 \|v\|_{H^s(\mathbb{R}^n)}^2 - \mu \|v\|_{L^2(\mathbb{R}^n)}^2$$

for some constants  $c_0, \mu > 0$  independent of v. The proof is now concluded as in lemma 3.3.

Assuming as in Section 3 that both (14) and (15) hold, by means of the above lemma 4.3 we can define the DN-maps  $\Lambda_P, \Lambda_P^*$  just as in lemma 3.4. We also arrive at the same Alessandrini identity and Runge approximation property which we get in lemmas 3.6 and 3.7.

**Lemma 4.4** (DN maps). Let  $\Omega \subset \mathbb{R}^n$  be a bounded Lipschitz domain and  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$ ,  $m \in \mathbb{N}$  such that 2s > m. Let  $a_\alpha \in H^{r_\alpha,\infty}(\Omega)$ , with  $r_\alpha$  defined as in (2). There exist two continuous linear maps

$$\Lambda_P: X \to X^*$$
 defined by  $\langle \Lambda_P[f], [g] \rangle := B_P(u_f, g)$ 

and

$$\Lambda_P^*: X \to X^* \quad defined \ by \quad \langle \Lambda_P^*[f], [g] \rangle := B_P^*(u_f^*, g)$$

where  $u_f, u_f^*$  are the unique solutions to the equations

$$(-\Delta)^{s}u + \sum_{|\alpha| \le m} a_{\alpha}(x)D^{\alpha}u = 0 \quad in \ \Omega, \quad u - f \in \widetilde{H}^{s}(\Omega)$$

and

$$(-\Delta)^{s}u^{*} + \sum_{|\alpha| \le m} (-1)^{|\alpha|} D^{\alpha}(a_{\alpha}(x)u^{*}) = 0 \quad in \ \Omega, \quad u^{*} - f \in \widetilde{H}^{s}(\Omega)$$

with  $f, g \in H^s(\mathbb{R}^n)$ . Moreover, the identity  $\langle \Lambda_P[f], [g] \rangle = \langle [f], \Lambda_P^*[g] \rangle$  holds.

**Lemma 4.5** (Alessandrini identity). Let  $\Omega \subset \mathbb{R}^n$  be a bounded Lipschitz domain and  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$ ,  $m \in \mathbb{N}$  such that 2s > m. Let  $a_\alpha \in H^{r_\alpha,\infty}(\Omega)$ , with  $r_\alpha$  defined as in (2). For any  $f_1, f_2 \in H^s(\mathbb{R}^n)$ , let  $u_1, u_2^* \in H^s(\mathbb{R}^n)$  respectively solve

$$(-\Delta)^{s}u_{1} + \sum_{|\alpha| \le m} a_{1,\alpha}(x)D^{\alpha}u_{1} = 0 \quad in \ \Omega, \quad u_{1} - f_{1} \in \widetilde{H}^{s}(\Omega)$$

and

$$(-\Delta)^{s}u_{2}^{*} + \sum_{|\alpha| \le m} (-1)^{|\alpha|} D^{\alpha}(a_{2,\alpha}(x)u_{2}^{*}) = 0 \quad in \ \Omega, \quad u_{2}^{*} - f_{2} \in \widetilde{H}^{s}(\Omega).$$

Then we have the integral identity

$$\langle (\Lambda_{P_1} - \Lambda_{P_2})[f_1], [f_2] \rangle = \sum_{|\alpha| \le m} \langle (a_{1,\alpha} - a_{2,\alpha}) D^{\alpha} u_1, u_2^* \rangle_{\Omega}.$$

**Lemma 4.6** (Runge approximation property). Let  $\Omega, W \subset \mathbb{R}^n$  respectively be a bounded Lipschitz domain and a non-empty open set such that  $\overline{W} \cap \overline{\Omega} = \emptyset$ . Let  $s \in \mathbb{R}^+ \setminus \mathbb{Z}$ ,  $m \in \mathbb{N}$  such that 2s > m. Let  $a_\alpha \in H^{r_\alpha,\infty}(\Omega)$ , with  $r_\alpha$  defined as in (2). Moreover, let  $\mathcal{R} := \{u_f - f : f \in C_c^\infty(W)\} \subset \widetilde{H}^s(\Omega)$ , where  $u_f$  solves

$$(-\Delta)^{s}u_{f} + \sum_{|\alpha| \le m} a_{\alpha}(x)D^{\alpha}u_{f} = 0 \quad in \ \Omega, \quad u_{f} - f \in \widetilde{H}^{s}(\Omega)$$

and  $\mathcal{R}^* := \{ u_f^* - f : f \in C_c^\infty(W) \} \subset \widetilde{H}^s(\Omega), \text{ where } u_f^* \text{ solves }$ 

I.

$$(-\Delta)^{s}u_{f}^{*} + \sum_{|\alpha| \le m} (-1)^{|\alpha|} D^{\alpha}(a_{\alpha}(x)u_{f}^{*}) = 0 \quad in \ \Omega, \quad u_{f}^{*} - f \in \widetilde{H}^{s}(\Omega).$$

Then  $\mathcal{R}$  and  $\mathcal{R}^*$  are dense in  $\widetilde{H}^s(\Omega)$ .

## 4.2. Proof of injectivity.

*Proof of theorem 1.2.* The proof is virtually identical to the one of theorem 1.1, the unique difference being in the way the error terms of the Runge approximation are estimated. We make use of (22), which relied on the Kato-Ponce inequality instead of multiplier space estimates. In this way we get

$$\begin{aligned} \left| \sum_{|\alpha| \le m} \langle (a_{1,\alpha} - a_{2,\alpha}) D^{\alpha} r_{1,k}, u_{2,k}^* \rangle_{\mathbb{R}^n} \right| &\le \sum_{|\alpha| \le m} |\langle (a_{1,\alpha} - a_{2,\alpha}) D^{\alpha} r_{1,k}, u_{2,k}^* \rangle_{\mathbb{R}^n} | \\ &\le C \|u_{2,k}^*\|_{H^s(\mathbb{R}^n)} \|r_{1,k}\|_{H^s(\mathbb{R}^n)} \sum_{|\alpha| \le m} \|a_{1,\alpha} - a_{2,\alpha}\|_{H^{r_\alpha,\infty}(\Omega)} \to 0 \end{aligned}$$

and

ī.

$$\begin{split} |\sum_{|\alpha| \le m} \langle (a_{1,\alpha} - a_{2,\alpha}) D^{\alpha} v_1, r_{2,k} \rangle_{\mathbb{R}^n} | \le \sum_{|\alpha| \le m} |\langle (a_{1,\alpha} - a_{2,\alpha}) D^{\alpha} v_1, r_{2,k} \rangle_{\mathbb{R}^n} | \\ \le C \| r_{2,k} \|_{H^s(\mathbb{R}^n)} \| v_1 \|_{H^s(\mathbb{R}^n)} \sum_{|\alpha| \le m} \| a_{1,\alpha} - a_{2,\alpha} \|_{H^{r_\alpha,\infty}(\Omega)} \to 0. \Box \end{split}$$

#### References

- [1] H. Abels. Pseudodifferential and Singular Integral Operators. De Gruyter, First edition, 2012.
- [2] F. Andreu-Vaillo, J. M. Mazón, J. D. Rossi, and J. J. Toledo-Melero. Nonlocal Diffusion Problems. American Mathematical Society, First edition, 2010.
- [3] A. Behzadan and M. Holst. Multiplication in Sobolev spaces, revisited. 2017. arXiv:1512.07379v2.
- [4] S. Bhattacharyya, T. Ghosh, and G. Uhlmann. Inverse Problem for Fractional-Laplacian with Lower Order Non-local Perturbations. *Transactions of the AMS*, 2020. To appear.
- [5] C. Bucur and E. Valdinoci. Nonlocal Diffusion and Applications. Springer, First edition, 2016.
- [6] X. Cao, Y.-H. Lin, and H. Liu. Simultaneously recovering potentials and embedded obstacles for anisotropic fractional Schrödinger operators. *Inverse Probl. Imaging*, 13(1):197–210, 2019.
- [7] M. Cekić, Y.-H. Lin, and A. Rüland. The Calderón problem for the fractional Schrödinger equation with drift. Calculus of Variations and Partial Differential Equations, 59(3):91, 2020.
- [8] G. Covi. Inverse problems for a fractional conductivity equation. Nonlinear Analysis, 2018.
- [9] G. Covi. An inverse problem for the fractional Schrödinger equation in a magnetic field. *Inverse Problems*, 36(4):045004, 2020.
- [10] G. Covi, K. Mönkkönen, and J. Railo. Unique continuation property and Poincaré inequality for higher order fractional Laplacians with applications in inverse problems. 2020. arXiv:2001.06210.
- [11] E. Di Nezza, G. Palatucci, and E. Valdinoci. Hitchhiker's guide to the fractional Sobolev spaces. Bulletin des Sciences Mathématiques, 136(5):521–573, 2012.
- [12] Q. Du, M. Gunzburger, R. B. Lehoucq, and K. Zhou. Analysis and Approximation of Nonlocal Diffusion Problems with Volume Constraints. SIAM Rev., 54, No. 4:667–696, 2012.
- [13] T. Ghosh, Y.-H. Lin, and J. Xiao. The Calderón problem for variable coefficients nonlocal elliptic operators. Communications in Partial Differential Equations, 42(12):1923–1961, 2017.
- [14] T. Ghosh, A. Rüland, M. Salo, and G. Uhlmann. Uniqueness and reconstruction for the fractional Calderón problem with a single measurement. *Journal of Functional Analysis*, 279(1):108505, 2020.
- [15] T. Ghosh, M. Salo, and G. Uhlmann. The Calderón problem for the fractional Schrödinger equation. Anal. PDE 13(2):455-475, 2020.
- [16] L. Grafakos and S. Oh. The Kato-Ponce inequality. Communications in Partial Differential Equations, 39(6):1128-1157, 2014.
- [17] A. Gulisashvili and M. A. Kon. Exact Smoothing Properties of Schrödinger Semigroups. American Journal of Mathematics, 118(6):1215–1248, 1996.
- [18] T. Kato and G. Ponce. Commutator Estimates and the Euler and Navier-Stokes Equations. Communications on Pure and Applied Mathematics, 41(7):891–907, 1988.
- [19] R.-Y. Lai and Y.-H. Lin. Global uniqueness for the fractional semilinear Schrödinger equation. Proc. Amer. Math. Soc., 147(3):1189–1199, 2019.
- [20] R.-Y. Lai and Y.-H. Lin. Inverse problems for fractional semilinear elliptic equations. 2020. arXiv:2004.00549.
- [21] R.-Y. Lai, Y.-H. Lin, and A. Rüland. The Calderón Problem for a Space-Time Fractional Parabolic Equation, 2020.
- [22] N. Laskin. Fractional Quantum Mechanics and Lévy Path Integrals. Physics Letters A, 268(4):298 305, 2000.
- [23] N. Laskin. Fractional Quantum Mechanics. World Scientific, First edition, 2018.
- [24] L. Li. A Semilinear Inverse Problem For The Fractional Magnetic Laplacian. 2020. arXiv:2005.06714.
- [25] L. Li. Determining The Magnetic Potential In The Fractional Magnetic Calderón Problem. 2020. arXiv:2006.10150.
- [26] L. Li. The Calderón problem for the fractional magnetic operator. Inverse Problems, 36(7):075003, 2020.
- [27] A. Massaccesi and E. Valdinoci. Is a nonlocal diffusion strategy convenient for biological populations in competition? *Journal of Mathematical Biology*, 74(1):113–147, 2017.
- [28] V. G. Maz'ya and T. O. Shaposhnikova. Theory of Sobolev Multipliers. Springer, First edition, 2009.
- [29] W. McLean. Strongly Elliptic Systems and Boundary Integral Equations. Cambridge University Press, First edition, 2000.
- [30] M. Mišur. A Refinement of Peetre's Theorem. Results in Mathematics, 74(4):199, 2019.
- [31] J. Navarro and J. B. Sancho. Peetre-Slovák's theorem revisited. 2014. arXiv:1411.7499.
- [32] J. Peetre. Une caractérisation abstraite des opérateurs différentiels. Mathematica Scandinavica, 7:211–218, 1959.
- [33] X. Ros-Oton. Nonlocal elliptic equations in bounded domains: a survey. Publicacions Matemátiques, 60:3 26, 2015.
- [34] A. Rüland and M. Salo. Exponential instability in the fractional Calderón problem. *Inverse Problems*, 34(4):045003, 21, 2018.
- [35] A. Rüland and M. Salo. Quantitative approximation properties for the fractional heat equation. Mathematical Control & Related Fields, pages 233–249, 2019.
- [36] A. Rüland and M. Salo. The fractional Calderón problem: Low regularity and stability. Nonlinear Analysis, 2019.
- [37] M. Salo. The fractional Calderón problem. Journées équations aux dérivées partielles, Exp. No.(7), 2017.
- [38] H. Triebel. Interpolation Theory, Function Spaces, Differential Operators. North-Holland Publishing Company, 1978.
- [39] G. Uhlmann. Electrical impedance tomography and Calderón's problem. Inverse Problems, 25(12):123011, 2009.
- [40] G. Uhlmann. Inverse problems: seeing the unseen. Bulletin of Mathematical Sciences, 4(2):209–279, 2014.
- [41] M. W. Wong. An Introduction to Pseudo-Differential Operators. World Scientific, Third edition, 2014.

DEPARTMENT OF MATHEMATICS AND STATISTICS, UNIVERSITY OF JYVÄSKYLÄ, JYVÄSKYLÄ, FINLAND *Email address*: giovanni.g.covi@jyu.fi

Email address: kematamo@student.jyu.fi

Seminar for Applied Mathematics, Department of Mathematics, ETH Zurich, Zürich, Switzer-Land

Email address: jesse.railo@math.ethz.ch

Department of Mathematics, University of Washington, Seattle, USA / Jockey Club Institute for Advanced Study, HKUST, Hong Kong

Email address: gunther@math.washington.edu

[E]

# On tensor decompositions and algebraic structure of the mixed and transverse ray transforms

J. Ilmavirta, K. Mönkkönen and J. Railo

Preprint (2020)

https://arxiv.org/abs/2009.01043

#### ON TENSOR DECOMPOSITIONS AND ALGEBRAIC STRUCTURE OF THE MIXED AND TRANSVERSE RAY TRANSFORMS

JOONAS ILMAVIRTA

Unit of Computing Sciences Tampere University FI-33014 Tampere University, Finland joonas.ilmavirta@tuni.fi

#### KEIJO MÖNKKÖNEN

Department of Mathematics and Statistics University of Jyväskylä P.O. Box 35 (MaD) FI-40014 University of Jyväskylä, Finland kematamo@student.jyu.fi

#### JESSE RAILO

Seminar for Applied Mathematics, Department of Mathematics ETH Zurich Rämistrasse 101, CH-8092 Zürich, Switzerland jesse.railo@math.ethz.ch

ABSTRACT. The geodesic ray transform, the mixed ray transform and the transverse ray transform of a tensor field on a manifold can all be seen as what we call mixing ray transforms, compositions of the geodesic ray transform and an invertible linear map on tensor fields. We show that the characterization of the kernel and the stability of a mixing ray transform can be reduced to the same properties of any other mixing ray transform. Our approach applies to various geometries and ray transforms, including the light ray transform. In particular, we extend studies in de Hoop–Saksala–Zhai (2019) from compact simple surfaces to orientable surfaces with solenoidally injective geodesic ray transform. Our proofs are based on algebraic arguments.

Date: Friday 18<sup>th</sup> September, 2020.

<sup>2010</sup> Mathematics Subject Classification. 44A12, 65R32, 53A99.

Key words and phrases. Geodesic ray transform, integral geometry, inverse problems.

#### 1. INTRODUCTION

This article gives an algebraic point of view to various geodesic ray transforms of tensor fields, unifying the Riemannian X-ray transform, the transverse ray transform, and the mixed ray transform, and the light ray transforms on Lorentzian manifolds. This comes with a natural notion of symmetry, which is not the same as the symmetry of the covariant tensor field whose integral transforms are under study, but arises from the structure of the relevant transform. For example, in the case of light ray tomography of 2-tensor fields on Lorentzian manifolds, the concept of symmetry automatically includes the "conformal gauge freedom".

When two transforms differ from each other by a so-called mixing, they have the same injectivity properties by theorem 3.3. Mixings turn mixed ray transforms into regular tensor transforms in two dimensions. In corollary 3.7 we recast the injectivity result [7] of the mixed ray transform on simple surfaces in our language and we provide a reproof in corollary 4.1. These results are also extended to Cartan–Hadamard manifolds in corollary 4.2.

The tensor tomography results [8] on globally hyperbolic Lorentzian manifolds have a different kind of kernel than their Riemannian counterpart. The kernel, when operating on symmetric tensor fields of order  $m \ge 2$ , contains both potential fields and conformal multiples of the metric. In the present approach the conformal gauge is absorbed into the concept of symmetry, making the statements of solenoidal injectivity (s-injectivity) fully analogous on Riemannian and Lorentzian manifolds; see corollary 3.9.

A number of corollaries of the method are given in this article, and we refrain from listing them all here. Consequently we have a great amount of notation, and we have collected the key items in appendix A to help the reader.

1.1. Mixing ray transforms. Let M be a Riemannian manifold of dimension  $n \geq 2$ . Let  $f \in \mathfrak{X}(T_m M)$  be a covariant *m*-tensor field (not necessarily symmetric) where  $m \geq 1$ . We completely exclude the scalar case m = 0 from our discussion. Let  $A: \mathfrak{X}(T_m M) \to \mathfrak{X}(T_m M)$  be an invertible linear map such that

(1.1) 
$$(Af)_x(v_1, \dots, v_m) = f_x(A_1(x)v_1, \dots, A_m(x)v_m)$$

where  $A_i(x): T_x M \to T_x M$  are linear isomorphisms. The linear maps  $\mathfrak{X}(T_m M) \to \mathfrak{X}(T_m M)$  of this form are called *mixings* in this article.

We study the class of geodesic ray transforms, called *mixing ray transforms*, defined by the formula

$$\begin{split} I_A f(x,v) &\coloneqq \int_{\tau_-(x,v)}^{\tau_+(x,v)} (Af)_{\gamma_{x,v}(t)} (\dot{\gamma}_{x,v}(t)^{\otimes m}) \mathrm{d}t \\ &= \int_{\tau_-(x,v)}^{\tau_+(x,v)} f_{\gamma_{x,v}(t)} (A_1(\gamma_{x,v}(t)) \dot{\gamma}_{x,v}(t), \dots, A_m(\gamma_{x,v}(t)) \dot{\gamma}_{x,v}(t)) \mathrm{d}t, \end{split}$$

where  $\gamma_{x,v}: [\tau_{-}(x,v), \tau_{+}(x,v)] \to M$  is the maximal unit speed geodesic through  $(x,v) \in SM$ . Formula (1.2) is invariant under the geodesic flow  $\varphi_t(x,v) = (\gamma_{x,v}(t), \dot{\gamma}_{x,v}(t))$ , that is,  $I_A f(x,v) = I_A f(\varphi_t(x,v))$  for any  $t \in$   $\mathbb{R}$  in the maximal domain of  $\gamma_{x,v}$ . This definition allows to define  $I_A$  on Riemannian manifolds without boundary, provided that the tensor field fis sufficiently integrable. We remark that if  $A_i = \text{Id}$  for every  $i = 1, \ldots, m$ , then  $I_A$  is the usual geodesic ray transform of tensor fields. Other special cases of the mixing ray transforms in two dimensions have been studied earlier in [5, 7, 26], and somewhat related geodesic ray transforms in higher dimensions have been studied recently in [1, 6, 17]. We remark that the mixing ray transforms are defined for all  $n \geq 2$  but they do not include the higher dimensional transforms  $(n \geq 3)$  studied in [1, 6, 17].

The main problems that we study are uniqueness and stability for recovering  $f \in \mathfrak{X}(T_m M)$  from the knowledge of  $I_A f$ . The main point of this work is an algebraic view of the mixing ray transforms. We present many applications of the method and instead of having a main theorem we have a main idea how to study the mixing ray transforms. We show in theorem 3.3 and corollary 3.4 that the related inverse problems for  $I_A$  and  $I_{\widetilde{A}}$  with two different mixings A and  $\widetilde{A}$  can be reduced to each other. Especially, this allows us to derive new uniqueness and stability results for the mixed and transverse ray transforms in two dimensions using the known results for the geodesic ray transform. These results are given in corollaries 4.1, 4.2, 4.5 and 4.7. Moreover, we show in corollaries 4.9 and 4.13 that on compact simple surfaces and on certain Cartan–Hadamard manifolds the geodesic ray transform and the transverse ray transform together determine one-forms uniquely. This extends results in [5] to more general Riemannian manifolds.

Furthermore, we study tensor decompositions and their symmetries with respect to these integral transforms. These considerations lead us to corollaries 3.7 and 3.9 which show how the earlier kernel characterizations of the mixed ray transform on compact simple surfaces and the light ray transform on static globally hyperbolic Lorentzian manifolds can be seen as sinjectivity results under the correct notions of symmetry.

1.2. Related problems. The geodesic ray transform has been studied extensively on Riemannian manifolds and s-injectivity is known in many cases. For example, the geodesic ray transform is s-injective on tensor fields of any order on two-dimensional compact simple manifolds [21] and on simply connected compact manifolds with strictly convex boundary and non-positive curvature [20, 23, 26]. S-injectivity is also known on non-compact Cartan– Hadamard manifolds for all tensor fields which satisfy certain decay conditions [15, 16]. We refer to the surveys [12, 22] for a more comprehensive treatment of the geodesic ray transform and s-injectivity. The mixed ray transform has been studied mainly on two- and three-dimensional compact simple manifolds, and the kernel is known in these cases for a certain class of tensor fields [6, 7] (see also [26]). There are a few results for the transverse ray transform: in  $\mathbb{R}^2$  the kernel of the transverse ray transform consists of curls of scalar fields [19] and in higher dimensions the transform is even injective on certain manifolds [26] (see also [1] for a support theorem).

The usual applications of the geodesic ray transform are medical imaging [18, 19], Doppler tomography [24, 27] and seismic imaging [26, 30]. The transverse ray transform has applications in polarization tomography [26],

photoelasticity [11], diffraction tomography [17] and also in the determination of the refractive index of gases [5, 25]. The mixed ray transform arises in seismology as a linearization of elastic travel time tomography problem [6, 26].

**Organization of the article.** In section 2 we recall the preliminaries on the geodesic ray transform and the mixed ray transform. In section 3 we define the mixing ray transforms and study their basic properties using an algebraic approach. In section 4 we apply our methods for the mixed ray transform and the transverse ray transform on orientable two-dimensional Riemannian manifolds which admit s-injectivity of the geodesic ray transform. We have included some of our notation in appendix A.

Acknowledgements. The authors wish to thank Teemu Saksala for helpful discussions related to the mixed ray transform. J.I. was supported by Academy of Finland (grants 332890 and 336254). K.M. and J.R. were supported by Academy of Finland (Centre of Excellence in Inverse Modelling and Imaging, grant numbers 284715 and 309963). J.R. was also supported by Swiss National Science Foundation (SNF Grant 184698).

#### 2. Preliminaries

We mainly follow the reference [26] for the integral geometry part of this section. Basic theory of differential geometry can be found in [13, 14] and basic theory of Sobolev spaces of tensor fields on manifolds can be found for example in [3, 33]. We always assume that (M, g) is a connected Riemannian manifold, and we can sometimes allow it to be pseudo-Riemannian.

2.1. Notation. If E is a vector bundle, we denote by  $\mathfrak{X}(E)$  the space of all smooth sections of E. We use this notation whenever the regularity is unimportant.

We let  $T_{m_1}^{m_2}M = T^*M^{\otimes m_1} \otimes TM^{\otimes m_2}$  be the bundle of tensors of type  $(m_2, m_1)$  over M. Then  $\mathfrak{X}(T_{m_1}^{m_2}M)$  is the space of all  $(m_2, m_1)$ -tensor fields on M. We also write  $\mathfrak{X}(T_mM) \coloneqq \mathfrak{X}(T_m^0M)$ .

We denote by  $S_m M \subset \mathfrak{X}(T_m M)$  the space of all symmetric covariant tensor fields. When we want to emphasize the regularity of the tensor field, we replace  $\mathfrak{X}$  with the regularity in question; for example  $C^q(T_m M) \subset \mathfrak{X}(T_m M)$ ,  $q \in \mathbb{N}$ , is the space of all  $C^q$ -smooth (0, m)-tensor fields on M. For symmetric tensor fields we write  $C^q(S_m M)$  and so on. We use the Einstein summation convention, where every repeated index (both as a subscript and superscript) is implicitly summed over.

2.2. Sobolev norms of tensor fields. Let  $f, h \in \mathfrak{X}(T_m M)$  be tensor fields and  $m \geq 1$ . We define the fiberwise inner product as

(2.1) 
$$g_x(f,h) = g^{i_1 j_1}(x) \dots g^{i_m j_m}(x) f_{i_1 \dots i_m}(x) h_{j_1 \dots j_m}(x)$$

and the fiberwise norm is denoted by  $|f|_{g_x} = \sqrt{g_x(f, f)}$ . If m = 0, we simply let  $|f|_{q_x} := |f(x)|$ .

Let  $dV_g(x)$  be the Riemannian volume measure on M. If M is orientable, then  $dV_g(x)$  is given by the Riemannian volume form and  $dV_g(x) = \sqrt{\det g(x)} dx^1 \wedge \ldots \wedge dx^n$  where  $(x^1, \ldots, x^n)$  are any positively oriented

4

smooth coordinates. We define the  $L^p$ -norm,  $1 \leq p < \infty$ , of a tensor field  $f \in \mathfrak{X}(T_m M)$  by

(2.2) 
$$||f||_{p} = \left(\int_{M} |f|_{g_{x}}^{p} \,\mathrm{d}V_{g}(x)\right)^{1/p}$$

whenever the integral exists.

Denote by  $\nabla^k f \in C^{q-k}(T_{m+k}M)$  the *k*th iterated covariant derivative of the tensor field  $f \in C^q(T_mM)$  whenever  $q \ge k \ge 0$  and  $k, q \in \mathbb{N}$ . We define the Sobolev norm  $\|\cdot\|_{k,p}$  as

(2.3) 
$$\|f\|_{k,p} = \left(\sum_{i=0}^{k} \|\nabla^{i}f\|_{p}^{p}\right)^{1/p}$$

where  $\nabla^0 f := f$ . Let  $C_{k,p}^{\infty}(T_m M)$  be the set of smooth tensor fields f for which  $||f||_{k,p} < \infty$ . The Sobolev space  $W^{k,p}(T_m M)$  is defined to be the completion of  $C_{k,p}^{\infty}(T_m M)$  with respect to the norm  $||\cdot||_{k,p}$ . We are mainly interested in the space  $W^{k,2}(T_m M) =: H^k(T_m M)$ . Then  $H^k(T_m M)$  is a Hilbert space with the inner product (2.4)

$$\langle f,h\rangle_{H^k(T_mM)} = \sum_{i=0}^k \left\langle \nabla^i f,\nabla^i h \right\rangle_{L^2(T_{m+i}M)} = \sum_{i=0}^k \int_M g_x(\nabla^i f,\nabla^i h) \mathrm{d} V_g(x).$$

Similarly one defines the Sobolev space  $H^k(S_mM) \subset H^k(T_mM)$  as the completion of  $C^{\infty}_{k,2}(S_mM)$  with respect to the norm induced by the inner product  $\langle \cdot, \cdot \rangle_{H^k(T_mM)}$ .

2.3. Hodge star on orientable Riemannian surfaces. Assume that (M, g) is two-dimensional orientable Riemannian manifold. For example, M is orientable if there is a smooth mapping  $F: M \to N$  such that F is a local diffeomorphism and N is orientable, or if M is simply connected [13]. The Hodge star  $\star$  is an operator on one-forms  $\star: \mathfrak{X}(T_1M) \to \mathfrak{X}(T_1M)$  and it corresponds to a 90 degree rotation counterclockwise. Orientability of M guarantees that  $\star$  is a well-defined global operator. Since we can identify one-forms with vector fields by the musical isomorphisms  $\flat$  and  $\sharp$ , we can also rotate vector fields. To shorten the notation, we simply let  $\sharp \star \flat =: \star$  and locally we have

(2.5) 
$$\star (v^1 e_1 + v^2 e_2) \coloneqq -v^2 e_1 + v^1 e_2$$

in any positively oriented local orthonormal frame  $\{e_1, e_2\}$ .

2.4. The geodesic ray transform. For any set X we denote by  $\mathcal{F}(X)$  the space of all complex-valued functions  $X \to \mathbb{C}$ . We define the map  $\lambda \colon \mathfrak{X}(T_m M) \to \mathcal{F}(TM)$  as

(2.6) 
$$(\lambda f)(x,v) \coloneqq f_x(v,\ldots,v) = f_{i_1\ldots i_m}(x)v^{i_1}\ldots v^{i_m}$$

where  $f_{i_1...i_m}(x)$  are the components of the tensor field  $f \in \mathfrak{X}(T_m M)$  in any local coordinates. We let  $SM = \bigcup_{x \in M} S_x M$  be the sphere bundle where the fibers are the unit spheres  $S_x M = \{v \in T_x M : |v|_{g_x} = 1\}$  of the tangent spaces  $T_x M$ . The unit sphere bundle SM is not to be confused with the space  $S_m M$  of symmetric covariant tensor fields of order m. The geodesic flow is defined as  $\varphi_t(x,v) = (\gamma_{x,v}(t), \dot{\gamma}_{x,v}(t))$  where  $\gamma_{x,v}(t)$  is the unique geodesic such that  $(\gamma_{x,v}(0), \dot{\gamma}_{x,v}(0)) = (x, v) \in SM$ . If M has boundary  $\partial M$ , we denote by  $\tau(x, v)$  the first time when the geodesic  $\gamma_{x,v}$  reaches  $\partial M$ .

Assume that (M, g) is compact and non-trapping Riemannian manifold with boundary. Non-trapping means that  $\tau(x, v) < \infty$  for all  $(x, v) \in SM$ . We denote by  $\partial_{in}SM \subset \partial SM$  the inward-pointing unit vectors. We define the geodesic ray transform to be the operator  $I : \mathcal{X} \to \mathcal{F}(\partial_{in}SM)$  given by the formula

(2.7) 
$$If(x,v) = \int_0^{\tau(x,v)} (\lambda f)(\varphi_t(x,v)) dt, \quad (x,v) \in \partial_{\rm in} SM$$

where  $\mathcal{X} \subset \mathfrak{X}(T_m M)$  is any set such that the integral in (2.7) is well-defined. Typically we choose  $\mathcal{X} = C_c^{\infty}(T_m M)$  or  $\mathcal{X} = H^k(T_m M)$ . We note that two definitions (1.2) and (2.7) agree when A = Id and  $(x, v) \in \partial_{\text{in}} SM$  (in that case  $\tau_+(x, v) = \tau(x, v)$  and  $\tau_-(x, v) = 0$ ). One can also write  $If = I_{SM}(\lambda f|_{SM})$  where the geodesic ray transform of a function  $h: SM \to \mathbb{R}$  is

(2.8) 
$$I_{SM}h(x,v) = \int_0^{\tau(x,v)} h(\varphi_t(x,v)) \mathrm{d}t, \quad (x,v) \in \partial_{\mathrm{in}} SM.$$

One can then define an adjoint  $I^*$  by duality using an  $L^2$ -inner product. However, there are different measures on  $\partial_{in}SM$  which lead to different adjoints. We use the weighted measure defined in [22] which is invariant under the scattering relation, and the normal operator  $N = I^*I$  is defined with respect to this measure.

If  $f \in H^k(S_m M)$  and M is a compact Riemannian manifold with boundary, then there is the solenoidal decomposition [26, Theorem 3.3.2]

(2.9) 
$$f = f^{s} + \sigma \nabla p, \quad \operatorname{div}(f^{s}) = 0, \quad p|_{\partial M} = 0$$

where  $f^{s} \in H^{k}(S_{m}M)$ ,  $p \in H^{k+1}(S_{m-1}M)$  and  $m \geq 1$ . Moreover, if  $f \in C^{\infty}(S_{m}M)$ , then  $f^{s} \in C^{\infty}(S_{m}M)$  and  $p \in C^{\infty}(S_{m-1}M)$ . Here  $\sigma$  is the symmetrization of tensor fields (see section 3.1 for details) and div( $\cdot$ ) is the covariant divergence. The tensor field  $f^{s}$  is the solenoidal part and  $\sigma \nabla p$  is the potential part of f. By the fundamental theorem of calculus one sees that  $I(\sigma \nabla p) = 0$  since p vanishes on the boundary. Therefore potentials are always in the kernel of I and we can only try to recover the solenoidal part of f from I. When  $m \geq 1$  we say that I is solenoidally injective (s-injective) if for sufficiently regular  $f \in S_{m}M$  it holds that If = 0 if and only if  $f = \sigma \nabla p$  for some (sufficiently regular)  $p \in S_{m-1}M$  vanishing on the boundary.

One particular class of manifolds where one usually studies the geodesic ray transform is the class of compact simple manifolds. The manifold (M, g) is simple if it is non-trapping, has no conjugate points and the boundary  $\partial M$  is strictly convex (the second fundamental form on  $\partial M$  is positive definite). Each compact simple manifold is diffeomorphic to the Euclidean unit ball. It also follows that compact simple manifolds are simply connected and hence orientable [21, 32].

One can also study the geodesic ray transform on certain non-compact manifolds. The manifold (M, g) without boundary is a Cartan-Hadamard manifold if it is complete, simply connected and its sectional curvature is nonpositive. Cartan–Hadamard manifolds are always non-compact, orientable and diffeomorphic to  $\mathbb{R}^n$ . Basic examples of Cartan–Hadamard manifolds are Euclidean and hyperbolic spaces. On such manifolds the geodesic ray transform is defined as

(2.10) 
$$If(x,v) = \int_{-\infty}^{\infty} (\lambda f)(\varphi_t(x,v)) dt, \quad (x,v) \in SM.$$

Note that completeness implies that geodesics are defined on all times by the Hopf–Rinow theorem [14]. We will use the following classes of tensor fields on Cartan–Hadamard manifolds:

$$E_{\eta}(T_{m}M) = \{f \in C^{1}(T_{m}M) : \\ |f|_{g_{x}} \leq Ce^{-\eta d(x,o)} \text{ for some } C > 0\}, \\ E_{\eta}^{1}(T_{m}M) = \{f \in C^{1}(T_{m}M) : \\ |f|_{g_{x}} + |\nabla f|_{g_{x}} \leq Ce^{-\eta d(x,o)} \text{ for some } C > 0\}, \\ (2.11) \qquad P_{\eta}(T_{m}M) = \{f \in C^{1}(T_{m}M) : \\ |f|_{g_{x}} \leq C(1 + d(x,o))^{-\eta} \text{ for some } C > 0\}, \\ P_{\eta}^{1}(T_{m}M) = \{f \in C^{1}(T_{m}M) : \\ |f|_{g_{x}} \leq C(1 + d(x,o))^{-\eta} \text{ and} \\ |\nabla f|_{g_{x}} \leq C(1 + d(x,o))^{-\eta-1} \text{ for some } C > 0\}. \end{cases}$$

Here  $o \in M$  is fixed reference point and  $\eta > 0$ . The spaces defined above are independent of the choice of this point.

2.5. The mixed and transverse ray transforms. Define  $S_k M \otimes S_l M \subset \mathfrak{X}(T_{k+l}M)$  to be the set of (k+l)-tensor fields which are symmetric in the first k and last l variables. Let  $S_k(T_xM)$  denote the space of symmetric (0, k)-tensors on  $T_xM$  for any fixed  $x \in M$ . If  $f \in S_kM \otimes S_lM$ , then  $f_x \in S_k(T_xM) \otimes S_l(T_xM)$ . Let  $\pi: \partial_{\mathrm{in}}SM \to M$  be the restriction of the projection of the tangent bundle. Let  $\pi^*(S_kM)$  be the pullback bundle of symmetric k-tensor fields so that for every  $\varphi \in \mathfrak{X}(\pi^*(S_kM))$  and  $(x, v) \in \partial_{\mathrm{in}}SM$  we have  $\varphi_{x,v} \in S_k(T_xM)$ .

Let  $v \in S_x M$ . We define the projection operator  $p_v : T_x M \to v^{\perp} \subset T_x M$ as

(2.12) 
$$p_v(w) \coloneqq w - g_x(w, v)v = \left(\delta_j^i - v_j v^i\right) w^j e_i$$

where the latter formula holds in any local coordinates. We then define the projection operator  $P_v^k \colon S_k(T_xM) \to S_k(T_xM)$  by the formula

(2.13) 
$$(P_v^k h)(v_1, \dots, v_k) \coloneqq h(p_v(v_1), \dots, p_v(v_k))$$

for any  $v_1, \ldots, v_k \in T_x M$ , and one can write in any local coordinates that

(2.14) 
$$(P_v^k h)_{i_1 \dots i_k} = (\delta_{i_1}^{j_1} - v^{j_1} v_{i_1}) \dots (\delta_{i_k}^{j_k} - v^{j_k} v_{i_k}) h_{j_1 \dots j_k}$$

We can identify  $p_v$  as a (1,1)-tensor by setting  $\tilde{p}_v(\alpha, w) \coloneqq \alpha(p_v(w))$  where  $w \in T_x M$  and  $\alpha \in T_x^* M$ . We note that also  $P_v^k h = \tilde{p}_v^{\otimes k} h$  where the product on the right hand side is a contraction of  $\tilde{p}_v^{\otimes k}$  by h.

We define the contraction of  $f \in S_k(T_xM) \otimes S_l(T_xM)$  by  $v \in T_xM$  with respect to the last l arguments as a mapping  $\Lambda_v^l \colon S_k(T_xM) \otimes S_l(T_xM) \to S_k(T_xM)$  by

(2.15) 
$$(\Lambda_v^l f)_{i_1...i_k} = f_{i_1...i_k j_1...j_l} v^{j_1} \dots v^{j_l}$$

Let us denote by  $\mathcal{T}_{\gamma}^{t\to s}$  the parallel transport along  $\gamma$  from  $\gamma(t)$  to  $\gamma(s)$ whenever  $s, t \in \mathbb{R}$  belong to the maximal domain of  $\gamma$ . The mixed ray transform is the map  $L_{k,l}: S_k M \otimes S_l M \to \pi^*(S_k M)$  defined as

(2.16) 
$$L_{k,l}f(x,v) \coloneqq \int_0^{\tau(x,v)} \mathcal{T}_{\gamma_{x,v}}^{t\to 0}(P_{\dot{\gamma}_{x,v}(t)}^k \Lambda^l_{\dot{\gamma}_{x,v}(t)} f_{\gamma_{x,v}(t)}) \mathrm{d}t$$

for any  $(x, v) \in \partial_{in}SM$ , whenever the integral is well-defined. We note that

(2.17) 
$$P_{\dot{\gamma}_{x,v}(t)}^{k} \Lambda_{\dot{\gamma}_{x,v}(t)}^{l} f_{\gamma_{x,v}(t)}(w_{1},\ldots,w_{k}) \\ = f_{\gamma_{x,v}(t)}(p_{\dot{\gamma}_{x,v}(t)}w_{1},\ldots,p_{\dot{\gamma}_{x,v}(t)}w_{k},\dot{\gamma}_{x,v}(t),\ldots,\dot{\gamma}_{x,v}(t))$$

for any  $w_1, \ldots, w_k \in T_{\gamma_{x,v}(t)}M$ .

Using (2.17) and the definition of the parallel transport  $\mathcal{T}_{\gamma}^{t \to s}$ , one can show that the mixed ray transform acts on  $(x, v) \in \partial_{\text{in}}SM$  as

(2.18)  
$$\begin{cases} \left\langle L_{k,l}f(x,v), (\eta+av)^{\otimes k} \right\rangle \\ = \int_{0}^{\tau(x,v)} f_{i_{1}\dots i_{k}j_{1}\dots j_{l}}(\gamma_{x,v}(t))\eta_{x,v}^{i_{1}}(t)\dots\eta_{x,v}^{i_{k}}(t)\dot{\gamma}_{x,v}^{j_{1}}(t)\dots\dot{\gamma}_{x,v}^{j_{l}}(t)dt \end{cases}$$

where  $(\eta + av)^{\otimes k}$  is the tensor product of  $\eta + av$  with itself k times,  $a \in \mathbb{R}$  and  $\eta_{x,v}(t)$  is the parallel transport of a vector  $\eta = \eta_{x,v}(0) \in T_x M$  orthogonal to  $v = \dot{\gamma}_{x,v}(0)$ , see [26, Chapter 5.2] for details.

The mixed ray transform is considerably simpler when M is orientable and n = 2. Then  $v^{\perp}$  is one-dimensional for all  $(x, v) \in \partial_{\text{in}}SM$  and there is only one possible choice (modulo sign) for the vector  $\eta$  which is parallel transported along  $\gamma$ . We choose the orthogonal vector field as  $\eta(t) = (\star \dot{\gamma})(t)$ . It is clear that  $\star \dot{\gamma} \perp \dot{\gamma}$  at every point on the geodesic  $\gamma$  and that  $D_t^{\gamma}(\star \dot{\gamma}) = 0$ where  $D_t^{\gamma}$  is the covariant derivative along the geodesic  $\gamma$ . Therefore  $\star \dot{\gamma}$  is parallel along  $\gamma$ . Now using formula (2.18) the mixed ray transform can be seen as a composition  $L_{k,l} = I \circ A_{k,l}$  where

(2.19) 
$$(A_{k,l}f)_x(v_1,\ldots,v_m) = f_x(A_1v_1,\ldots,A_mv_m)$$

and  $A_i = \star$  when i = 1, ..., k and  $A_i = \text{Id}$  when i = k + 1, ..., k + l. Thus in two dimensions the mixed ray transform operates as

(2.20) 
$$L_{k,l}f(x,v) = \int_0^{\tau(x,v)} (\lambda(A_{k,l}f))(\varphi_t(x,v)) \mathrm{d}t, \quad (x,v) \in \partial_{\mathrm{in}}SM,$$

and with these choices of  $A_{k,l}$  we have  $L_{k,l} = I_{A_{k,l}}$  where the transform  $I_{A_{k,l}}$ is given by formula (1.2). If k = 0, then  $L_{0,l}$  reduces to the geodesic ray transform *I*. If l = 0, we call  $L_{k,0}$  the transverse ray transform and use the notation  $I_{\perp} := L_{k,0}$ . In higher dimensions n > 2 the operator  $\star$  cannot be used to define the mixed ray transform since it maps k-forms into (n - k)forms. 3.1. Decompositions of tensor fields. Let  $\sigma: \mathfrak{X}(T_m M) \to S_m M$  be the usual symmetrization map of tensor fields where  $m \geq 2$ . We remind that if m = 1, then any  $f \in \mathfrak{X}(T_m M)$  is symmetric. The components of  $\sigma f$  at a point  $x \in M$  are

(3.1) 
$$(\sigma f)_{i_1...i_m}(x) = \frac{1}{m!} \sum_{\tau \in \Pi_m} f_{i_{\tau(1)}...i_{\tau(m)}}(x)$$

where  $\Pi_m$  is the group of permutations. The symmetrization  $\sigma$  is a projection  $\mathfrak{X}(T_m M) \to S_m M$ , and it turns out to be orthogonal at every point with respect to any Riemannian metric by proposition 3.1. In particular,  $\sigma$  is idempotent and we can decompose the space  $\mathfrak{X}(T_m M)$  as

(3.2) 
$$\operatorname{Ker}(\sigma) \oplus \operatorname{Im}(\sigma) = \mathfrak{X}(T_m M)$$

by letting  $f = (f - \sigma f) + \sigma f$ . The decomposition (3.2) can be done on any differentiable manifold M. The set  $\text{Ker}(\sigma)$  can be identified with antisymmetric tensor fields when m = 2 and for m > 2 the antisymmetric tensor fields are a strict subset of  $\text{Ker}(\sigma)$ .

Recall that the map  $\lambda \colon \mathfrak{X}(T_m M) \to \mathcal{F}(TM)$  was defined as

(3.3) 
$$(\lambda f)(x,v) = f_x(v,\ldots,v)$$

where  $\mathcal{F}(TM)$  is the space of all complex-valued functions on TM. We note that the restriction  $\lambda f|_{SM}$  determines  $\lambda f$  completely since  $f_x$  is homogeneous of degree m. It follows directly from the definitions that  $\lambda \circ \sigma = \lambda$ . It is true that  $\operatorname{Ker}(\sigma) = \operatorname{Ker}(\lambda)$  (see proposition 3.1) and  $\operatorname{Ker}(\lambda) \subset \operatorname{Ker}(I)$ . Hence we call  $\operatorname{Ker}(\lambda) \subset \mathfrak{X}(T_mM)$  the set of  $\lambda$ -antisymmetric tensor fields or trivial part of the kernel of the geodesic ray transform depending on the context.

We denote by  $\lambda_x \colon (T_x^*M)^{\otimes m} \to \mathcal{F}(T_xM)$  the map  $(\lambda_x\omega)(v) = \omega(v, \ldots, v)$ , i.e.  $(\lambda f)(x, v) = (\lambda_x f_x)(v)$ . We let  $\sigma_x$  be the symmetrization of *m*-tensors in  $(T_x^*M)^{\otimes m}$  and  $S_m(T_xM)$  is the space of symmetric *m*-tensors in  $(T_x^*M)^{\otimes m}$ . We have the following proposition which summarizes some important connections between the different concepts introduced above.

**Proposition 3.1.** Suppose that  $m \ge 2$  and let M be a Riemannian (or pseudo-Riemannian) manifold. Let  $x \in M$  and define the sets

(a)  $V_1 = S_m(T_x M), V_2 = \text{Im}(\sigma_x) \text{ and } V_3 = \text{Ker}(\lambda_x)^{\perp}.$ 

(b)  $W_1 = (S_m(T_x M))^{\perp}, W_2 = \text{Ker}(\sigma_x) \text{ and } W_3 = \text{Ker}(\lambda_x).$ 

Then  $V_1 = V_2 = V_3$ ,  $W_1 = W_2 = W_3$ , and  $V_i \oplus W_j = (T_x^* M)^{\otimes m}$  for any i, j = 1, 2, 3.

*Proof.* It follows directly from the definitions that  $V_1 = V_2$ . Suppose that  $W_2 = W_3$  and  $V_3 \subset V_1$ . This implies that

(3.4) 
$$(T_x^*M)^{\otimes m} = V_2 \oplus W_2 = V_3 \oplus W_3 = V_3 \oplus W_2$$

Since  $V_3 \subset V_1 = V_2$ , we get that  $V_2 = V_3$ . It then follows that  $V_1 = V_2 = V_3$ ,  $W_1 = W_2 = W_3$ , and  $V_i \oplus W_j = (T_x^* M)^{\otimes m}$  for any i, j = 1, 2, 3. Hence it is sufficient to show that  $W_2 = W_3$  and  $V_3 \subset V_1$ .

Let us first prove that  $W_2 = W_3$ . It is clear that  $\operatorname{Ker}(\sigma_x) \subset \operatorname{Ker}(\lambda_x)$  since  $\lambda_x \circ \sigma_x = \lambda_x$ . Let  $f \in \operatorname{Ker}(\lambda_x)$ . It now follows that  $\sigma_x f \in \operatorname{Ker}(\lambda_x)$ . The

polarization identity for symmetric multilinear maps [31, Theorem 1] states that a symmetric multilinear map is uniquely determined by its restriction to the diagonal. Since  $\lambda_x \sigma_x f$  is the restriction of  $\sigma_x f: (T_x M)^m \to \mathbb{C}$  to the diagonal of  $(T_x M)^m, \sigma_x f \in S_m(T_x M)$  and  $\lambda_x \sigma_x f = 0$ , we obtain that  $\sigma_x f =$ 0. This shows that  $\operatorname{Ker}(\lambda_x) \subset \operatorname{Ker}(\sigma_x)$ , and we conclude that  $W_2 = W_3$ .

Let us then prove that  $V_3 \subset V_1$ . Let  $f \in V_3$ . Fix some indices  $j'_1, \ldots, j'_m$ and define the components of the tensor  $h \in (T^*_x M)^{\otimes m}$  as

(3.5) 
$$h_{j_1\dots j_m} = \left(\delta_{j_1}^{j_1'}\delta_{j_m}^{j_m'} - \delta_{j_m}^{j_1'}\delta_{j_1}^{j_m'}\right)\delta_{j_2}^{j_2'}\cdots\delta_{j_{m-1}}^{j_{m-1}'}.$$

Then  $h \in \text{Ker}(\lambda_x)$  and  $g_x(f,h) = 0$  implies that  $f_{i_1...i_m} = f_{i_m...i_1}$ . By switching the order of the indices  $j_k$  in the definition of h one sees that  $f_{i_1...i_m}$ has to be symmetric with respect to all indices. Hence  $V_3 \subset V_1$ . This completes the proof.

Remark 3.2. We obtain a somewhat surprising implication that the orthogonal complement of  $S_m(T_xM)$  does not depend on the Riemannian metric  $g_x$  at the point  $x \in M$ . This follows from proposition 3.1 since the mapping  $\sigma_x$  does not depend on  $g_x$ . Proposition 3.1 also shows that the symmetrization  $\sigma: \mathfrak{X}(T_mM) \to S_mM$  is an orthogonal projection when M is equipped with any Riemannian or pseudo-Riemannian metric.

By proposition 3.1 we have many choices for the decomposition of the space  $(T_x^*M)^{\otimes m}$ . We will use the orthogonal complement so that  $\operatorname{Ker}(\lambda_x) \oplus \operatorname{Ker}(\lambda_x)^{\perp} = (T_x^*M)^{\otimes m}$ . This allows us to decompose the space  $\mathfrak{X}(T_mM)$  in the following way. We define the space  $\operatorname{Ker}(\lambda)^{\perp}$  by saying that  $f \in \operatorname{Ker}(\lambda)^{\perp}$  if and only if  $f_x \in \operatorname{Ker}(\lambda_x)^{\perp}$  for all  $x \in M$ . Define the projection  $\widehat{\sigma} : \mathfrak{X}(T_mM) \to \operatorname{Ker}(\lambda)^{\perp}$  such that  $(\widehat{\sigma}f)_x = P_{\operatorname{Ker}(\lambda_x)^{\perp}}f_x$  where  $P_{\operatorname{Ker}(\lambda_x)^{\perp}}$  is the orthogonal projection  $P_{\operatorname{Ker}(\lambda_x)^{\perp}} : (T_x^*M)^{\otimes m} \to \operatorname{Ker}(\lambda_x)^{\perp}$ . Then  $f = (f - \widehat{\sigma}f) + \widehat{\sigma}f$  where  $f - \widehat{\sigma}f \in \operatorname{Ker}(\lambda)$  and  $\widehat{\sigma}f \in \operatorname{Ker}(\lambda)^{\perp}$ . Hence we have the orthogonal decomposition

(3.6) 
$$\operatorname{Ker}(\lambda) \oplus \operatorname{Ker}(\lambda)^{\perp} = \mathfrak{X}(T_m M)$$

where orthogonality is understood pointwise. We call the map  $\hat{\sigma}$  a  $\lambda$ -symmetrization. Note that  $\operatorname{Ker}(\lambda) = \operatorname{Ker}(\sigma)$  and  $\operatorname{Ker}(\lambda)^{\perp} = S_m M$  by proposition 3.1.

Another way to view  $\lambda$ -symmetric tensor fields is to take the quotient space  $\operatorname{Coim}(\lambda) = \mathfrak{X}(T_m M)/\operatorname{Ker}(\lambda)$  which identifies all tensor fields which differ by an element of  $\operatorname{Ker}(\lambda)$ . This definition is natural for the geodesic ray transform in the sense that If = Ih whenever  $f \sim h$ . It follows that if V is any algebraic complement of  $\operatorname{Ker}(\lambda)$ , i.e.  $\operatorname{Ker}(\lambda) \oplus V = \mathfrak{X}(T_m M)$ , then  $V \cong \operatorname{Coim}(\lambda)$  via the map  $v \mapsto [v]$  where [v] is the equivalence class of v. This shows that one can realize the abstract quotient space  $\operatorname{Coim}(\lambda)$  as a complementary subspace of  $\operatorname{Ker}(\lambda)$  and that all complementary subspaces are isomorphic.

More generally, let  $\Omega = \bigcup_{x \in M} \Omega_x \subset TM$  where  $\Omega_x \subset T_x M$ . Let  $r_x$  be the restriction of a multilinear map on  $T_x M$  to  $\Omega_x$ . As before we can decompose  $(T_x^*M)^{\otimes m} = \operatorname{Ker}(\lambda_{r,x}) \oplus \operatorname{Ker}(\lambda_{r,x})^{\perp}$  where  $\lambda_{r,x} = r_x \circ \lambda_x$ . This splitting can be done globally as follows. Denote by  $r: \mathcal{F}(TM) \to \mathcal{F}(\Omega)$  the restriction

to  $\Omega$  and define  $\lambda_r = r \circ \lambda$ . Then we have the decomposition

(3.7) 
$$\operatorname{Ker}(\lambda_r) \oplus \operatorname{Ker}(\lambda_r)^{\perp} = \mathfrak{X}(T_m M)$$

by writing  $f = (f - \hat{\sigma}_r f) + \hat{\sigma}_r f$  where  $\hat{\sigma}_r \colon \mathfrak{X}(T_m M) \to \operatorname{Ker}(\lambda_r)^{\perp}$  is defined as  $(\hat{\sigma}_r f)_x = P_{\operatorname{Ker}(\lambda_{r,x})^{\perp}} f_x$  and the space  $\operatorname{Ker}(\lambda_r)^{\perp}$  is defined pointwise as earlier. We call the projection  $\hat{\sigma}_r$  a  $\lambda_r$ -symmetrization. As above  $\operatorname{Ker}(\lambda_r)^{\perp}$  can be seen as a realization of the quotient space  $\operatorname{Coim}(\lambda_r) = \mathfrak{X}(T_m M) / \operatorname{Ker}(\lambda_r)$ . It follows that  $\operatorname{Ker}(\lambda) \subset \operatorname{Ker}(\lambda_r)$  and  $\operatorname{Ker}(\lambda_r)^{\perp} \subset \operatorname{Ker}(\lambda)^{\perp} \subset S_m M$  by proposition 3.1.

Note that if r is the restriction to SM, then  $\operatorname{Ker}(\lambda_r) = \operatorname{Ker}(\lambda)$ . The geodesic ray transform can then be seen as a composition  $I = I_{SM} \circ \lambda_r$ . We will generalize this approach in the next subsection.

3.2. The mixing ray transform. Let  $\operatorname{Aut}(TM)$  be the automorphism bundle of the tangent bundle. A section B of this bundle, called an automorphism field, is a field whose value B(x) at any  $x \in M$  is an automorphism (a linear self-bijection) of  $T_x M$ . In local coordinates B can be expressed as

$$(3.8) B(x) = B_k^j(x) dx^k \otimes \partial_j$$

where  $B_k^j(x)$  is an invertible matrix at every point x.

Let  $A_i$ , i = 1, ..., m, be smooth automorphism fields. Their tensor product  $A = A_1 \otimes \cdots \otimes A_m$  is a mapping of tensor fields,  $A: \mathfrak{X}(T_m M) \to \mathfrak{X}(T_m M)$ . From an invariant point of view it operates on a tensor field  $f \in \mathfrak{X}(T_m M)$  as

(3.9) 
$$(Af)_x(v_1,\ldots,v_m) = f_x(A_1(x)v_1,\ldots,A_m(x)v_m),$$

and it can be written in local coordinates as

(3.10) 
$$(Af)_{i_1...i_m}(x) = (A_1)_{i_1}^{j_1}(x) \dots (A_m)_{i_m}^{j_m}(x) f_{j_1...j_m}(x).$$

Since each  $A_i(x)$  (or  $(A_i)_k^j(x)$ ) is invertible and  $A_i$  is smooth also A is invertible and smooth. We call such map A an *admissible mixing of degree m*.

Let  $r: \mathcal{F}(TM) \to \mathcal{F}(\Omega)$  be the restriction to  $\Omega = \bigcup_{x \in M} \Omega_x \subset TM$  where  $\Omega_x \subset T_x M$  and  $\lambda_r = r \circ \lambda$ . Let Z be a vector space and  $J: \mathcal{F}(\Omega) \to Z$  a linear mapping. We define the abstract ray transform  $I_{A,r}: \mathfrak{X}(T_m M) \to Z$  as

$$(3.11) I_{A,r} = J \circ \lambda_r \circ A.$$

Usually r is the restriction to SM and J is the geodesic ray transform on SM. We call  $I_{A,r}$  the mixing ray transform when these assumptions hold and write  $\lambda := \lambda_r$  and  $I_A := I_{A,r}$  to simplify our notation. Next, we decompose the space  $\mathfrak{X}(T_mM)$  into symmetric and antisymmetric parts with respect to  $\lambda_r \circ A$ . Assume we have the decomposition  $\operatorname{Ker}(\lambda_r) \oplus$  $\operatorname{Ker}(\lambda_r)^{\perp} = \mathfrak{X}(T_mM)$ . Since A is bijective linear map, we have  $\operatorname{Ker}(\lambda_r \circ A) =$  $A^{-1}(\operatorname{Ker}(\lambda_r))$  and

(3.12) 
$$\mathfrak{X}(T_m M) = A^{-1}(\operatorname{Ker}(\lambda_r) \oplus \operatorname{Ker}(\lambda_r)^{\perp}) \\ = \operatorname{Ker}(\lambda_r \circ A) \oplus A^{-1}(\operatorname{Ker}(\lambda_r)^{\perp}).$$

Hence we choose  $A^{-1}(\operatorname{Ker}(\lambda_r)^{\perp})$  as the space of  $(\lambda_r \circ A)$ -symmetric tensor fields. The symmetrization map  $\widehat{\sigma}_{A,r}$  is a projection onto  $A^{-1}(\operatorname{Ker}(\lambda_r)^{\perp})$ 

and it has the expression  $\widehat{\sigma}_{A,r} = A^{-1} \circ \widehat{\sigma}_r \circ A$  where  $\widehat{\sigma}_r$  is a projection onto  $\operatorname{Ker}(\lambda_r)^{\perp}$ .

One can also naturally define the mixing ray transform on a quotient space as a mapping

(3.13) 
$$I_{A,r}^q: \mathfrak{X}(T_m M) / \operatorname{Ker}(\lambda_r \circ A) \to Z$$

such that

(3.14) 
$$I^q_{A,r}[f] = I_{A,r}f,$$

where  $[f] \in \mathfrak{X}(T_m M) / \operatorname{Ker}(\lambda_r \circ A)$  is the equivalence class of  $f \in \mathfrak{X}(T_m M)$ . The transform  $I_{A,r}^q$  is well-defined, i.e. it does not depend on the representative.

We conclude this subsection with the following theorem which basically says that it is enough to know the properties of one mixing ray transform since any other mixing ray transform can be reduced to the known case.

**Theorem 3.3.** Let  $E_1, E_2, E_3 \subset \mathfrak{X}(T_mM)$  be subspaces and  $m \geq 1$ . Assume that A and  $\widetilde{A}$  are admissible mixings of degree m and let  $\mathcal{D} = A^{-1} \circ \widetilde{A}$ . Then the following properties hold:

(a) **Kernel characterization:** Let  $\mathcal{H} = \operatorname{Id} -\widehat{\sigma}_{A,r}$  and  $Y = \operatorname{Ker}(I_{\widetilde{A},r}) \cap \widetilde{A}^{-1}(\operatorname{Ker}(\lambda_r)^{\perp})$ . Then  $f \in \operatorname{Ker}(I_{A,r})$  if and only if  $f = \mathcal{H}f + \mathcal{D}w$  for some  $w \in Y$ . We have the decomposition

(3.15) 
$$\operatorname{Ker}(I_{A,r}) = \operatorname{Im}(\mathcal{H}) \oplus \operatorname{Im}(\mathcal{D}|_Y) = \operatorname{Ker}(\lambda_r \circ A) \oplus \operatorname{Im}(\mathcal{D}|_Y)$$

where  $\operatorname{Ker}(I_{A,r}), \operatorname{Im}(\mathcal{H}), \operatorname{Im}(\mathcal{D}|_Y) \subset \mathfrak{X}(T_m M).$ 

- (b) **Reconstruction:** Let  $\mathcal{R}_{\widetilde{A},r}: Z \to S$  be a left inverse of  $I_{\widetilde{A},r}: S \to Z$ , where  $S \subset \widetilde{A}^{-1}(\operatorname{Ker}(\lambda_r)^{\perp})$ . Then  $\mathcal{R}_{A,r} = \mathcal{D} \circ \mathcal{R}_{\widetilde{A},r}: Z \to \mathcal{D}(S)$  is a left inverse of  $I_{A,r}: \mathcal{D}(S) \to Z$  where  $\mathcal{D}(S) \subset A^{-1}(\operatorname{Ker}(\lambda_r)^{\perp})$ .
- (c) **Stability:** Let  $(Z, \|\cdot\|_Z)$  and  $(E_1, \|\cdot\|_{E_1})$  be normed spaces. Also assume that  $\mathcal{D}$  is bounded on  $(E_1, \|\cdot\|_{E_1})$  and that the estimate  $\|f\|_{E_1} \leq C \|I_{\widetilde{A},r}f\|_Z$  holds for some subset  $S' \subset \widetilde{A}^{-1}(\operatorname{Ker}(\lambda_r)^{\perp})$ . Then the estimate

(3.16) 
$$\|f\|_{E_1} \le C \|\mathcal{D}\|_{E_1} \|I_{A,r}f\|_Z$$

holds for all  $f \in \mathcal{D}(S') \subset A^{-1}(\operatorname{Ker}(\lambda_r)^{\perp}).$ 

(d) Adjoint and normal operator: Let  $(Z, \langle \cdot, \cdot \rangle_Z)$  and  $(E_2, \langle \cdot, \cdot \rangle_{E_2})$  be Hilbert spaces. Assume that  $\mathcal{D}^{-1}$  is bounded in  $(E_2, \langle \cdot, \cdot \rangle_{E_2})$  and that  $I_{\widetilde{A},r} \colon E_2 \to Z$  is bounded. Then the adjoints and the normal operators of  $I_{A,r}$  and  $I_{\widetilde{A},r}$  satisfy the formulas

(3.17) 
$$I_{A,r}^* = (\mathcal{D}^{-1})^* I_{\widetilde{A},r}^*, \quad N_{A,r} = (\mathcal{D}^{-1})^* N_{\widetilde{A},r} \mathcal{D}^{-1}.$$

(e) Stability with normal operators: Suppose that the assumptions of (d) hold and let  $\|\cdot\|_{E_3}$  be a norm on  $E_3$ . Assume also that  $\mathcal{D}^*$  is bounded in

12

 $(E_3, \|\cdot\|_{E_3})$  and that the estimate  $\|f\|_{E_2} \leq C \|N_{\widetilde{A}, r}f\|_{E_3}$  holds for some subset  $S'' \subset \widetilde{A}^{-1}(\operatorname{Ker}(\lambda_r)^{\perp})$ . Then the estimate

(3.18) 
$$\|f\|_{E_2} \le C \|\mathcal{D}\|_{E_2} \|\mathcal{D}^*\|_{E_3} \|N_{A,r}f\|_{E_3}$$

holds for all  $f \in \mathcal{D}(S'') \subset A^{-1}(\operatorname{Ker}(\lambda_r)^{\perp}).$ 

*Proof.* (a) If f is of the form  $f = \mathcal{H}f + \mathcal{D}w$  for some  $w \in Y$ , then clearly  $f \in \operatorname{Ker}(I_{A,r})$ . For the converse, let  $w = \widehat{\sigma}_{\widetilde{A},r}\mathcal{D}^{-1}f = \mathcal{D}^{-1}\widehat{\sigma}_{A,r}f$ . We can write

(3.19) 
$$f = (f - \widehat{\sigma}_{A,r}f) + \widehat{\sigma}_{A,r}f = \mathcal{H}f + \mathcal{D}\mathcal{D}^{-1}\widehat{\sigma}_{A,r}f = \mathcal{H}f + \mathcal{D}w.$$

Clearly  $w \in \widetilde{A}^{-1}(\operatorname{Ker}(\lambda_r)^{\perp})$  and

(3.20) 
$$I_{\widetilde{A},r}w = I_{\widetilde{A},r}\mathcal{D}^{-1}\widehat{\sigma}_{A,r}f = I_{A,r}\widehat{\sigma}_{A,r}f = I_{A,r}f = 0$$

so  $w \in \operatorname{Ker}(I_{\widetilde{A},r})$ . Assume then that  $f \in \operatorname{Im}(\mathcal{H}) \cap \operatorname{Im}(\mathcal{D}|_Y)$ . Now  $f \in \operatorname{Ker}(\lambda_r \circ A)$  and  $f = \mathcal{D}w$  where  $w \in \widetilde{A}^{-1}(\operatorname{Ker}(\lambda_r)^{\perp})$ . But this implies  $w = \mathcal{D}^{-1}f \in \operatorname{Ker}(\lambda_r \circ \widetilde{A})$  and hence w = 0. Therefore  $\operatorname{Im}(\mathcal{H}) \cap \operatorname{Im}(\mathcal{D}|_Y) = \{0\}$ .

(b) Clearly  $\mathcal{D}(S) \subset \mathcal{D}(\widetilde{A}^{-1}(\operatorname{Ker}(\lambda_r)^{\perp})) = A^{-1}(\operatorname{Ker}(\lambda_r)^{\perp})$ . Let  $f \in \mathcal{D}(S)$ . Then

(3.21) 
$$\mathcal{DR}_{\widetilde{A},r}I_{A,r}f = \mathcal{DR}_{\widetilde{A},r}I_{\widetilde{A},r}\mathcal{D}^{-1}f = \mathcal{D}\mathcal{D}^{-1}f = f$$

implying that  $\mathcal{D} \circ \mathcal{R}_{\widetilde{A}_r}$  is a left inverse of  $I_{A,r}$  on  $\mathcal{D}(S)$ .

(c) For  $f \in \mathcal{D}(S')$  we find

$$(3.22) ||f||_{E_1} = ||\mathcal{D}\mathcal{D}^{-1}f||_{E_1} \le ||\mathcal{D}||_{E_1} ||\mathcal{D}^{-1}f||_{E_1} \le C ||\mathcal{D}||_{E_1} ||I_{\widetilde{A},r}\mathcal{D}^{-1}f||_{Z}$$
  
(3.23) 
$$= C ||\mathcal{D}||_{E_1} ||I_{A,r}f||_{Z}$$

as claimed.

(d) Using the definitions of adjoints, we obtain

$$\langle I_{A,r}f,h\rangle_{Z} = \left\langle I_{\widetilde{A},r}\mathcal{D}^{-1}f,h\right\rangle_{Z} = \left\langle \mathcal{D}^{-1}f,I_{\widetilde{A},r}^{*}h\right\rangle_{E_{2}} = \left\langle f,(\mathcal{D}^{-1})^{*}I_{\widetilde{A},r}^{*}h\right\rangle_{E_{2}}$$

Hence  $I_{A,r}^* = (\mathcal{D}^{-1})^* I_{\widetilde{A}r}^*$  and the normal operator becomes

(3.25) 
$$N_{A,r} = I_{A,r}^* I_{A,r} = (\mathcal{D}^{-1})^* I_{\widetilde{A},r}^* I_{\widetilde{A},r} \mathcal{D}^{-1} = (\mathcal{D}^{-1})^* N_{\widetilde{A},r} \mathcal{D}^{-1}.$$

(e) If  $f \in \mathcal{D}(S'')$ , then we have

(3.26) 
$$\|f\|_{E_2} \le \|\mathcal{D}\|_{E_2} \|\mathcal{D}^{-1}f\|_{E_2} \le C \|\mathcal{D}\|_{E_2} \|N_{\widetilde{A},r}\mathcal{D}^{-1}f\|_{E_3}$$

(3.27) 
$$= C \|\mathcal{D}\|_{E_2} \left\| \mathcal{D}^* (\mathcal{D}^{-1})^* N_{\widetilde{A},r} \mathcal{D}^{-1} f \right\|_{E_3} \\ \leq C \|\mathcal{D}\|_{E_2} \|\mathcal{D}^*\|_{E_3} \|N_{A,r} f\|_{E_3}.$$

3.3. Solenoidal injectivity. In this section we analyze more closely the kernel characterization given in theorem 3.3(a). Specifically, we apply our methods to show s-injectivity of general mixing ray transforms when s-injectivity of the geodesic ray transform is known. We also use our approach to show that the earlier results about the kernel of the mixed ray transform on compact simple surfaces and the kernel of the light ray transform on static globally hyperbolic Lorentzian manifolds can be seen as s-injectivity results under correct notions of symmetry.

3.3.1. General results. Let r be the restriction to SM and  $J = I_{SM}$  so that  $I = I_{SM} \circ \lambda_r$  and  $I_{A,r} = I \circ A$ . Since now  $\operatorname{Ker}(\lambda_r) = \operatorname{Ker}(\lambda)$  we use an abuse of notation and write  $\lambda := \lambda_r$ . By proposition 3.1 we can choose  $\operatorname{Ker}(\lambda)^{\perp} = S_m M$  so that  $\widehat{\sigma}_A = A^{-1} \circ \sigma \circ A$  is a projection onto  $A^{-1}(S_m M)$ . Further, we define the covariant derivative  $\nabla^A = A^{-1} \circ \nabla$ . The derivative  $\nabla^A$  is natural for the transform  $I_A$  since if  $v|_{\partial M} = 0$ , then  $I_A(\widehat{\sigma}_A \nabla^A v) = 0$ .

We say that the mixing ray transform  $I_A$  is s-injective on a compact Riemannian manifold with boundary if the following property holds for all  $f \in C^{\infty}(T_m M)$ :  $I_A f = 0$  if and only if  $\widehat{\sigma}_A f = \widehat{\sigma}_A \nabla^A u$  for some  $u \in C^{\infty}(S_{m-1}M)$  vanishing on the boundary. S-injectivity allows one to decompose the kernel of  $I_A$  as

(3.28) 
$$\operatorname{Ker}(I_A|_{C^{\infty}(T_mM)}) = \operatorname{Im}(\mathcal{H}|_{C^{\infty}(T_mM)}) \oplus \operatorname{Im}(\widehat{\sigma}_A \nabla^A|_Y)$$

where  $\mathcal{H} = \mathrm{Id} - \widehat{\sigma}_A$  and

(3.29) 
$$Y = \{ u \in C^{\infty}(S_{m-1}M) : u |_{\partial M} = 0 \}$$

It follows that s-injectivity of any mixing ray transform implies s-injectivity for all mixing ray transforms.

**Corollary 3.4.** Let  $m \ge 1$  and (M, g) be a compact Riemannian manifold with boundary so that the transform  $I_A$  is s-injective for some A of degree m. Then  $I_{\widetilde{A}}$  is s-injective for all  $\widetilde{A}$  of degree m.

*Proof.* Let us denote  $\hat{\sigma}_A = A^{-1}\sigma A$  and  $\hat{\sigma}_{\widetilde{A}} = \widetilde{A}^{-1}\sigma \widetilde{A}$  for the projections. Using the solenoidal injectivity for  $I_A$  we easily obtain

$$(3.30) I_{\widetilde{A}}f = I_A(A^{-1}\widetilde{A}f) = 0 \Leftrightarrow \exists u \in Y : \widehat{\sigma}_A A^{-1}\widetilde{A}f = \widehat{\sigma}_A \nabla^A u$$
$$\Leftrightarrow \exists u \in Y : \widehat{\sigma}_{\widetilde{A}}f = \widehat{\sigma}_{\widetilde{A}} \nabla^{\widetilde{A}}u. \Box$$

We immediately obtain the following corollary from the previous corollary.

**Corollary 3.5.** Take any  $m \ge 1$ . Assume that (M,g) is a compact Riemannian manifold with boundary so that the geodesic ray transform is s-injective on m-tensor fields. Then  $I_A$  is s-injective for all A of degree m.

We have similar results for the mixing ray transform in the quotient space  $\mathfrak{X}(T_m M)/\operatorname{Ker}(\lambda \circ A)$ . We denote by  $[\cdot]_A$  the corresponding equivalence classes and say that the quotient transform defined as  $I_A^q[f]_A = I_A f$  (see section 3.2) is s-injective if for all  $[f]_A \in C^{\infty}(T_m M)/\operatorname{Ker}(\lambda \circ A)$  we have  $I_A^q[f]_A = 0$  if and only if  $[f]_A = [\nabla^A u]_A$  for some  $u \in C^{\infty}(S_{m-1}M)$  vanishing on the boundary.

**Corollary 3.6.** Let  $m \ge 1$  and A be a mixing of degree m. Assume that (M,g) is a compact Riemannian manifold with boundary. Then  $I_A$  is s-injective if and only if  $I_A^q$  is s-injective.

*Proof.* Assume first that  $I_A$  is s-injective. We obtain

(3.31) 
$$I_A f = I_A^q [f]_A = 0 \Leftrightarrow \widehat{\sigma}_A f = \widehat{\sigma}_A \nabla^A u$$

which in turn implies

(3.32) 
$$[f]_A = [\widehat{\sigma}_A f]_A = [\widehat{\sigma}_A \nabla^A u]_A = [\nabla^A u]_A.$$

Assume then that  $I_A^q$  is s-injective. Now

(3.33) 
$$I_A^q[f]_A = I_A f = 0 \Leftrightarrow [f]_A = [\nabla^A u]_A$$

which implies  $f - \nabla^A u = h \in \text{Ker}(\lambda \circ A)$ . Hence  $\widehat{\sigma}_A f = \widehat{\sigma}_A \nabla^A u$ .

The previous results imply that if  $I_A^q$  is s-injective for some A of degree  $m \ge 1$ , then  $I_{\widetilde{A}}^q$  is s-injective for all  $\widetilde{A}$  of degree m. We remark that if  $A = \operatorname{Id}$ , then  $I_A^q$  corresponds to the geodesic ray transform in the quotient space  $\mathfrak{X}(T_m M)/\operatorname{Ker}(\lambda)$ . Especially, s-injectivity of I on m-tensor fields implies s-injectivity for  $I_A^q$  where A is any mixing of degree m.

3.3.2. Mixed ray transform on compact simple surfaces. Let us then consider the mixed ray transform  $L_{k,l} = I_{SM} \circ \lambda \circ A_{k,l}$  on a compact simple surface (M,g) where  $A_{k,l}$  is defined via equation (2.19). Define the operators  $\lambda' w = \sigma_{k,l}(g \otimes w)$  and  $d'u = \sigma_l \nabla u$  where  $\sigma_l$  is the symmetrization with respect to the last l indices and  $\sigma_{k,l}$  is the symmetrization with respect to the first kand the last l indices. In coordinates

(3.34) 
$$(\lambda'w)_{i_1\dots i_k j_1\dots j_l} = \sigma_{k,l}(g_{i_1j_1}w_{i_2,\dots i_k j_2\dots j_l})$$

(3.35) 
$$(\mathbf{d}'u)_{i_1\dots i_k j_1\dots j_l} = \sigma_l((\nabla_{e_{j_1}} u)_{i_1\dots i_k j_2\dots j_l}).$$

We compare our approach to the kernel characterization done in [7]. Especially, we obtain the following alternative result for s-injectivity.

**Corollary 3.7.** Let (M, g) be two-dimensional compact simple Riemannian manifold and  $f \in C^{\infty}(T_{k+l}M)$ . Then  $L_{k,l}f = 0$  if and only if  $\widehat{\sigma}_{A_{k,l}}f = \widehat{\sigma}_{A_{k,l}}\sigma_l \nabla u$  where  $u \in C^{\infty}(S_k M \otimes S_{l-1}M)$  such that  $u|_{\partial M} = 0$ .

*Proof.* Assume that  $L_{k,l}f = 0$ . Since

(3.36) 
$$\widehat{\sigma}_{A_{k,l}} f \in A_{k,l}^{-1}(C^{\infty}(S_m M)) \subset C^{\infty}(S_k M \otimes S_l M)$$

we obtain  $L_{k,l}\widehat{\sigma}_{A_{k,l}}f = L_{k,l}f = 0$ . By [7, Theorem 1] we have  $\widehat{\sigma}_{A_{k,l}}f = \sigma_l \nabla u + \sigma_{k,l}(g \otimes w)$  for some  $u \in C^{\infty}(S_k M \otimes S_{l-1}M)$ ,  $u|_{\partial M} = 0$ , and  $w \in C^{\infty}(S_{k-1}M \otimes S_{l-1}M)$ . Now  $\sigma_{k,l}(g \otimes w) \in \operatorname{Ker}(\lambda \circ A_{k,l})$  and hence  $\widehat{\sigma}_{A_{k,l}}f = \widehat{\sigma}_{A_{k,l}}\sigma_l \nabla u$ .

Then assume that  $\widehat{\sigma}_{A_{k,l}}f = \widehat{\sigma}_{A_{k,l}}\sigma_l \nabla u$  for some u vanishing on the boundary. Since  $L_{k,l} = I_{SM} \circ \lambda \circ A_{k,l}$  and  $\lambda \circ \sigma = \lambda$  we obtain

$$(3.37) \quad L_{k,l}f = L_{k,l}\widehat{\sigma}_{A_{k,l}}f = (I_{SM} \circ \lambda \circ A_{k,l})(A_{k,l}^{-1}\sigma A_{k,l}\sigma_l\nabla u) = L_{k,l}\sigma_l\nabla u = 0$$

where the last equality follows from the fundamental theorem of calculus.  $\hfill\square$ 

Remark 3.8. The previous s-injectivity result is similar to what we obtained earlier, i.e.  $I_{A_{k,l}}f = 0$  if and only if  $\hat{\sigma}_{A_{k,l}}f = \hat{\sigma}_{A_{k,l}}\nabla^{A_{k,l}}u$  for some  $u \in C^{\infty}(S_{m-1}M)$  vanishing on the boundary. We thus have the following alternative characterizations of the kernel of the mixed ray transform

(3.38) 
$$\operatorname{Ker}(L_{k,l}|_{C^{\infty}(T_mM)}) = \operatorname{Im}(\mathcal{H}|_{C^{\infty}(T_mM)}) \oplus \operatorname{Im}(\widehat{\sigma}_{A_{k,l}}\nabla^A|_Y)$$

(3.39) 
$$\operatorname{Ker}(L_{k,l}|_{C^{\infty}(T_mM)}) = \operatorname{Im}(\mathcal{H}|_{C^{\infty}(T_mM)}) \oplus \operatorname{Im}(\widehat{\sigma}_{A_{k,l}}d'|_{Y'})$$

where  $\mathcal{H} = \mathrm{Id} - \widehat{\sigma}_{A_{k,l}}$  and

(3.40) 
$$Y = \{ u \in C^{\infty}(S_{m-1}M) : u|_{\partial M} = 0 \}$$

(3.41) 
$$Y' = \{ u \in C^{\infty}(S_k M \otimes S_{l-1} M) : u |_{\partial M} = 0 \}.$$

Compare these to the decomposition of the kernel in [7]

(3.42) 
$$\operatorname{Ker}(L_{k,l}|_{C^{\infty}(S_{k}M\otimes S_{l}M)}) = \operatorname{Im}(\lambda'|_{C^{\infty}(S_{k-1}M\otimes S_{l-1}M)}) + \operatorname{Im}(\mathbf{d}'|_{Y'}).$$

Our decompositions split any tensor field uniquely into the trivial part and non-trivial part of the kernel. The uniqueness of decomposition (3.42) is not known and it only applies to tensor fields with certain symmetries.

3.3.3. *Light ray transform on Lorentzian manifolds.* We quickly review the relevant definitions for the light ray transform on static globally hyperbolic Lorentzian manifolds. More details can be found in [8].

Let  $(\mathcal{N}, \overline{g})$  be a smooth globally hyperbolic Lorentzian manifold of dimension 1 + n with signature  $(-, +, \ldots, +)$ . Let  $\beta$  be a maximal light-like geodesic so that

(3.43) 
$$\overline{\nabla}_{\dot{\beta}(s)}\dot{\beta}(s) = 0, \quad \overline{g}(\dot{\beta}(s), \dot{\beta}(s)) = 0$$

where  $\overline{\nabla}$  is the covariant derivative with respect to  $\overline{g}$ . We define the light ray transform of  $f \in C_c^{\infty}(T_m \mathcal{N})$  as

(3.44) 
$$\mathcal{L}_{\beta}f = \int_{-\infty}^{\infty} (\lambda f)(\beta(s), \dot{\beta}(s)) \mathrm{d}s,$$

where  $\beta$  ranges over all lightlike geodesics of  $\mathcal{N}$ . Since  $(\mathcal{N}, \overline{g})$  is globally hyperbolic, there exists a Cauchy hypersurface  $N \subset \mathcal{N}$ , i.e. a hypersurface such that any causal curve intersects N exactly once. We define  $g = \overline{g}|_N$ ; note that (N, g) becomes a Riemannian manifold. We will focus on static Lorentzian manifolds. It follows that if  $\mathcal{N}$  is static, then for any Cauchy hypersurface  $N \subset \mathcal{N}$  there exists an isometric embedding  $\Phi \colon \mathbb{R} \times N \to \mathcal{N}$ so that  $\Phi^* \overline{g} = -\kappa dt^2 + g$  where  $\kappa$  is a smooth positive function on N. We let  $g_c = \kappa^{-1} g$ .

Let r be the restriction to the set

(3.45) 
$$\Omega = \{(x, v) \in T\mathcal{M} : \overline{g}_x(v, v) = 0\}$$

where  $\mathcal{M} = \Phi(\mathbb{R} \times M)$ ,  $M \subset N$  is a compact submanifold with smooth boundary and  $\lambda_r = r \circ \lambda$  as before. We define the quotient light ray transform  $\mathcal{L}^q_\beta$  in  $C^\infty_c(T_m\mathcal{M})/\operatorname{Ker}(\lambda_r)$  as  $\mathcal{L}^q_\beta[f] = \mathcal{L}_\beta f$ ; note that the definition does not depend on the representative. We obtain the following s-injectivity result for  $\mathcal{L}^q_\beta$ . **Corollary 3.9.** Let  $(\mathcal{N}, \overline{g})$  be static globally hyperbolic Lorentzian manifold of dimension 1 + n and let  $\mathcal{N} \subset \mathcal{N}$  be a fixed Cauchy hypersurface. Let  $\mathcal{M} = \Phi(\mathbb{R} \times M)$  where  $M \subset \mathcal{N}$  is a compact n-dimensional submanifold with smooth boundary and  $\Phi$  is the isometric embedding introduced earlier. Assume that the geodesic ray transform is s-injective on  $(M, g_c)$  and let  $[f] \in C_c^{\infty}(T_m \mathcal{M})/\operatorname{Ker}(\lambda_r)$ . Then  $\mathcal{L}_{\beta}^q[f] = 0$  for all maximal  $\beta$  in  $(\mathcal{M}, \overline{g})$  if and only if  $[f] = [\overline{\nabla}T]$  for some  $T \in C_c^{\infty}(S_{m-1}\mathcal{M})$ .

Proof. Assume that  $\mathcal{L}_{\beta}^{q}[f] = 0$ . Then  $\mathcal{L}_{\beta}(\sigma f) = \mathcal{L}_{\beta}f = 0$  and by [8, Theorem 2] we obtain  $\sigma f = \sigma \overline{\nabla}T + \sigma(g \otimes U)$  for some  $T \in C_{c}^{\infty}(S_{m-1}\mathcal{M})$  and  $U \in C_{c}^{\infty}(S_{m-2}\mathcal{M})$ . Hence

(3.46) 
$$[f] = [\sigma f] = [\sigma \overline{\nabla} T] + [\sigma(g \otimes U)] = [\overline{\nabla} T]$$

where we used the fact that  $\operatorname{Ker}(\lambda) \subset \operatorname{Ker}(\lambda_r)$  and  $\sigma(g \otimes U) \in \operatorname{Ker}(\lambda_r)$ . This gives the other direction of the claim. Assume than that  $[f] = [\overline{\nabla}T]$  for some  $T \in C_c^{\infty}(S_{m-1}\mathcal{M})$ . The fundamental theorem of calculus implies that  $\mathcal{L}^q_{\beta}[f] = \mathcal{L}^q_{\beta}[\overline{\nabla}T] = \mathcal{L}_{\beta}(\sigma\overline{\nabla}T) = 0$ . This concludes the proof.

Remark 3.10. One can realize the quotient space  $\mathfrak{X}(T_m\mathcal{M})/\operatorname{Ker}(\lambda_r)$  as a complementary subspace  $V_r \subset \mathfrak{X}(T_m\mathcal{M})$  which satisfies  $\operatorname{Ker}(\lambda_r) \oplus V_r = \mathfrak{X}(T_m\mathcal{M})$ . This can be done for example by taking the orthogonal complement  $V_r = \operatorname{Ker}(\lambda_r)^{\perp}$  with respect to a Riemannian metric on  $\mathcal{M}$  (see section 3.1). Then corollary 3.9 implies that we have the decomposition

$$(3.47) \qquad \operatorname{Ker}(\mathcal{L}_{\beta}|_{C_{c}^{\infty}(T_{m}\mathcal{M})}) = \operatorname{Im}(\mathcal{H}|_{C_{c}^{\infty}(T_{m}\mathcal{M})}) \oplus \operatorname{Im}(\widehat{\sigma}_{r}\nabla|_{C_{c}^{\infty}(S_{m-1}\mathcal{M})})$$

where  $\hat{\sigma}_r$  is the orthogonal projection onto  $\operatorname{Ker}(\lambda_r)^{\perp}$  and  $\mathcal{H} = \operatorname{Id} - \hat{\sigma}_r$ .

3.4. Boundedness and pointwise estimates of mixings. In this section we give sufficient conditions which imply pointwise norm estimates and continuity of A in Sobolev spaces. Boundedness and pointwise estimates are used in section 4 to prove stability estimates and injectivity results for the mixed ray transform on two-dimensional orientable Riemannian manifolds.

**Lemma 3.11.** Let  $f \in C^q(T_mM)$  where  $m \ge 1$  and  $q \in \mathbb{N}$ . Then the following properties hold:

(a) If  $A_i$  satisfy the relation  $|A_iv|_{g_x} \leq C_i(x) |v|_{g_x}$  for all  $v \in T_xM$ , then we have the pointwise estimate

(3.48) 
$$|Af|_{q_x} \le n^m C_1(x) \dots C_m(x) |f|_{q_x}.$$

Especially, if  $C_i = C_i(x)$  are all bounded, then A extends into a bounded mapping A:  $L^2(T_m M) \to L^2(T_m M)$ .

(b) If in addition  $|\nabla_{e_j} A_i|_{g_x} \leq C'_i(x)$  for any local frame  $\{e_j\}$ , then we have the pointwise estimate

(3.49) 
$$|\nabla(Af)|_{q_x} \le C''(x)(|f|_{q_x} + |\nabla f|_{q_x})$$

where C'' = C''(x) can be expressed in terms of  $C_i$  and  $C'_i$ . Especially, if  $C_i, C'_i$  are all bounded, then A extends into a bounded mapping  $A: H^1(T_mM) \to H^1(T_mM).$  (c) If (M, g) is a two-dimensional orientable Riemannian manifold, then the operator  $A_{k,l}$  defined in (2.19) satisfies

$$(3.50) \qquad \qquad \left|\nabla^p(A_{k,l}f)\right|_{g_x} = \left|\nabla^p f\right|_{g_x}$$

for all  $p \in \mathbb{N}$ ,  $p \leq q$ . In particular, the mixing  $A_{k,l}$  extends into an isometry  $A_{k,l}: H^p(T_mM) \to H^p(T_mM)$  for all  $p \in \mathbb{N}$ .

*Proof.* (a) Choose normal coordinates in a neighborhood of x. The boundedness assumption for  $A_i$  implies

(3.51) 
$$|(A_i)_k^j(x)| \le \left(\sum_{j=1}^n \left|(A_i)_k^j(x)\right|^2\right)^{1/2} = |A_i e_k|_{g_x} \le C_i(x)$$

where  $(A_i)_k^j$  are the components of  $A_i$  in these coordinates. Now we can estimate the norm as

$$|Af|_{g_x}^2 = \sum_{i_1...i_m=1}^n ((Af)_{i_1...i_m}(x))^2$$
  
(3.52) 
$$= \sum_{i_1...i_m=1}^n \left(\sum_{j_1...j_m=1}^n (A_1)_{i_1}^{j_1}(x) \dots (A_m)_{i_m}^{j_m}(x) f_{j_1...j_m}(x)\right)^2$$
  
$$\leq n^m C_1^2(x) \dots C_m^2(x) \sum_{i_1...i_m=1}^n \sum_{j_1...j_m=1}^n |f_{j_1...j_m}(x)|^2$$
  
$$\leq n^{2m} C_1^2(x) \dots C_m^2(x) |f|_{g_x}^2.$$

If  $C_i$  are all bounded, then  $A: L^2(T_m M) \to L^2(T_m M)$  is bounded by definition and approximation by smooth tensor fields.

(b) By choosing normal coordinates, the covariant derivative at the point  $x \in M$  reduces to the ordinary derivative. Now

(3.53) 
$$\left|\nabla_{e_j} A_i\right|_{g_x}^2 = \sum_{k,l=1}^n (\partial_j (A_i)_l^k(x))^2,$$

which implies  $\left|\partial_j(A_i)_l^k(x)\right| \le C'_i(x)$ . Using the Leibniz rule we obtain (3.54)

$$\begin{split} |\nabla (Af)|_{g_x}^2 &= \sum_{k,i_1\dots i_m=1}^n (\partial_k (Af)_{i_1\dots i_m} (x))^2 \\ &\leq n^{2m+1} ((C_1')^2 (x) \dots C_m^2 (x) + \dots + C_1^2 (x) \dots (C_m')^2 (x)) \left| f \right|_{g_x}^2 \\ &+ n^{2m} C_1^2 (x) \dots C_m^2 (x) \left| \nabla f \right|_{g_x}^2 \\ &= \widehat{C}(x) \left| f \right|_{g_x}^2 + \widetilde{C}(x) \left| \nabla f \right|_{g_x}^2 . \end{split}$$

By taking  $C''(x) = \sqrt{2} \max\{\widehat{C}^{1/2}(x), \widetilde{C}^{1/2}(x)\}$  we get the desired inequality. If  $C_i, C'_i$  are all bounded, then  $A: H^1(T_mM) \to H^1(T_mM)$  is bounded.

(c) Again using normal coordinates, one can calculate that

(3.55) 
$$g_x(A_{k,l}f, A_{k,l}f) = g_x(f, A_{k,l}^{-1}A_{k,l}f) = g_x(f, f),$$

18

where we used the relations  $(A_i)_m^j = -(A_i)_m^m$  for  $i = 1, \ldots k$ ,  $(A_i)_m^j = \delta_m^j$ for  $i = k + 1, \ldots k + l$  and  $(-1)^k A_{k,l} = A_{k,l}^{-1}$ . For the derivatives we get

(3.56) 
$$g_x(\nabla^p(Af), \nabla^p(Af)) = g_x(\nabla^p f, \nabla^p f)$$

using the fact that  $\sum_{j=1}^{n} (A_i)_j^m (A_i)_j^q = \delta_q^m$  for all  $i = 1, \dots, k+l$ .

Remark 3.12. In a similar fashion as in part (b) one obtains the boundedness of  $A: H^k(T_m M) \to H^k(T_m M)$  if one assumes boundedness of the derivatives up to order  $k \in \mathbb{N}$ , i.e.  $|\nabla_{\alpha} A_i|_{g_x} \leq C_i^{\alpha}(x)$  for all  $|\alpha| \leq k$  where  $C_i^{\alpha} = C_i^{\alpha}(x)$  is bounded,  $\nabla_{\alpha} = \nabla_{e_1}^{\alpha_1} \dots \nabla_{e_n}^{\alpha_n}$  and  $|\alpha| = \alpha_1 + \dots + \alpha_n$ .

4. The mixed and transverse ray transforms on two-dimensional orientable Riemannian manifolds

4.1. Solenoidal injectivity on compact and non-compact surfaces. First we state a general result on s-injectivity of the mixed ray transform on compact orientable surfaces with boundary. This follows from corollary 3.5. We use the notation introduced in section 3.3.1. Note that  $A_{k,l}f \in C^p(T_mM)$  whenever  $f \in C^p(T_mM)$  where  $p \in \mathbb{N}$ .

**Corollary 4.1.** Let  $m \geq 1$ . Let (M,g) be compact two-dimensional orientable Riemannian manifold with boundary such that the geodesic ray transform is s-injective on  $C^{\infty}(S_m M)$  and let  $f \in C^{\infty}(T_m M)$ . Then  $L_{k,l}f = 0$ if and only if  $\widehat{\sigma}_{A_{k,l}}f = \widehat{\sigma}_{A_{k,l}}\nabla^{A_{k,l}}h$  for some  $h \in C^{\infty}(S_{m-1}M)$  vanishing on the boundary  $\partial M$ .

We note that the previous result holds on a wide class of two-dimensional orientable manifolds. These include for example compact simple surfaces [21] and simply connected compact surfaces with strictly convex boundary and non-positive sectional curvature [20, 26]. See [12, 22] for more manifolds with s-injective geodesic ray transform.

We have the following corollary for the mixed ray transform on Cartan– Hadamard manifolds which is a simple consequence of the pointwise estimates for  $A_{k,l}$  and the results in [16]. We denote by K(x) the Gaussian curvature of (M, g) at  $x \in M$ .

**Corollary 4.2.** Let (M, g) be a two-dimensional Cartan–Hadamard manifold and let  $m \ge 1$ . The following claims are true:

- (a) Let  $-K_0 \leq K \leq 0$  for some  $K_0 > 0$  and  $f \in E_{\eta}^1(T_m M)$  for some  $\eta > \frac{3}{2}\sqrt{K_0}$ . Then  $L_{k,l}f = 0$  if and only if  $\widehat{\sigma}_{A_{k,l}}f = \widehat{\sigma}_{A_{k,l}}\nabla^{A_{k,l}}h$  for some  $h \in S_{m-1}M$  such that  $h \in E_{\eta-\varepsilon}(T_{m-1}M)$  for all  $\varepsilon > 0$ .
- (b) Let  $K \in P_{\kappa}(M)$  for some  $\kappa > 2$  and  $f \in P_{\eta}^{1}(T_{m}M)$  for some  $\eta > 2$ . 2. Then  $L_{k,l}f = 0$  if and only if  $\widehat{\sigma}_{A_{k,l}}f = \widehat{\sigma}_{A_{k,l}}\nabla^{A_{k,l}}h$  for some  $h \in S_{m-1}M \cap P_{\eta-1}(T_{m-1}M)$ .

*Proof.* (a) If  $f \in E^1_{\eta}(T_m M)$ , then from the pointwise estimates for the transform  $A_{k,l}$  we obtain that

(4.1) 
$$|\sigma A_{k,l}f|_{g_x} \le (m!)^{1/2} |A_{k,l}f|_{g_x} = (m!)^{1/2} |f|_{g_x} \le Ce^{-\eta d(x,o)}$$

for some C > 0 and

(4.2) 
$$|\nabla(\sigma A_{k,l}f)|_{g_x} \le (m!)^{1/2} |\nabla(A_{k,l}f)|_{g_x} = (m!)^{1/2} |\nabla f|_{g_x} \le C' e^{-\eta d(x,o)}$$

for some C' > 0. Hence  $\sigma A_{k,l}f \in S_m M \cap E^1_{\eta}(T_m M)$  for some  $\eta > \frac{3}{2}\sqrt{K_0}$ . Since  $I(\sigma A_{k,l}f) = L_{k,l}f = 0$ , we must have  $\sigma A_{k,l}f = \sigma \nabla h$  for some  $h \in S_{m-1}M$  where  $h \in E_{\eta-\varepsilon}(T_{m-1}M)$  for all  $\varepsilon > 0$  by [16, Theorem 1.1]. This gives the claim for the first part.

(b) Similarly using the pointwise estimates one obtains that  $\sigma A_{k,l}f \in S_m M \cap P^1_{\eta}(T_m M)$  for some  $\eta > 2$ . Now [16, Theorem 1.2] implies that  $\sigma A_{k,l}f = \sigma \nabla h$  for some  $h \in S_{m-1}M \cap P_{\eta-1}(T_{m-1}M)$ . This proves the second part.

Remark 4.3. One can study the mixed ray transform on asymptotically hyperbolic surfaces [9]. Let (M, g) be an asymptotically hyperbolic surface,  $\overline{M}$  the compactification of M and  $\rho$  a geodesic boundary defining function as defined in [9]. One usually assumes that  $f \in \rho^{1-m}C^{\infty}(S_m\overline{M})$  to obtain s-injectivity results for the geodesic ray transform. It then follows that  $\sigma A_{k,l}f \in \rho^{1-m}C^{\infty}(S_m\overline{M})$  and similar s-injectivity result as in corollary 4.1 holds under certain assumptions on (M, g); we refer to [9] for a more detailed discussion. One can also study the mixed ray transform on asymptotically conic surfaces (M', g'). One obtains s-injectivity for tensor fields  $f \in A_{k,l}^{-1}\rho'^r C^{\infty}(S_m^s\overline{M'})$  where  $\rho'$  is the boundary defining function, r > n/2 + 1 and  $S_m^{sc}\overline{M'} \subset S_m\overline{M'}$  is the set of scattering tensor fields on the compactification  $\overline{M'}$ . See [10] for more details.

4.2. Stability results on compact surfaces. In this section we obtain stability estimates for the mixed ray transform. We begin with the following lemma.

**Lemma 4.4.** Let (M, g) be a compact simple surface. Then the normal operator of the mixed ray transform  $L_{k,l}$  is  $N_{k,l} = (-1)^k A_{k,l} N A_{k,l}$  where N is the normal operator of the geodesic ray transform I on (k+l)-tensor fields.

*Proof.* By theorem 3.3 part (d) we only need to calculate  $(\mathcal{D}^{-1})^* = A_{k,l}^*$ . Now for the matrix representations of  $A_i$  we have that  $(A_i)_m^j = -(A_i)_j^m$  for  $i = 1, \ldots k$  and  $(A_i)_m^j = \delta_m^j$  for  $i = k + 1, \ldots k + l$ . Using this one obtains

(4.3) 
$$g_x(A_{k,l}f,h) = g_x(f,(-1)^k A_{k,l}h)$$

and thus

(4.4) 
$$\langle A_{k,l}f,h\rangle_{L^2(T_mM)} = \left\langle f,(-1)^k A_{k,l}h\right\rangle_{L^2(T_mM)}$$

Hence  $A_{k,l}^* = (-1)^k A_{k,l}$  which gives the claim.

The next estimates are direct consequences of the results in [20, 28, 29]. We denote by  $\operatorname{Sol}(T_m M)$  the set of solenoidal tensor fields. For the definition of the tangential norm  $\|\cdot\|_{H^{1/2}_{T}(\partial_{\operatorname{in}}SM)}$  see [20].

**Corollary 4.5.** For any compact simple surface (M, g) and nonnegative integers k and l there is a constant C > 0 so that:

(a) Let k + l = 1. Let g be extended to a simple metric in  $M_1 \supset \supset M$ . Then the estimate

(4.5) 
$$\|f\|_{L^2(T_1M)} / C \le \|N_{k,l}f\|_{H^1(T_1M_1)} \le C \|f\|_{L^2(T_1M)}$$

20

holds for all  $f \in A_{kl}^{-1}(\operatorname{Sol}(T_1M)) \cap L^2(T_1M)$ .

(b) Let k + l = 2. Let g be extended to a simple metric in  $M_1 \supset \supset M$ . Then the estimate

(4.6) 
$$\|f\|_{L^2(T_2M)}/C \le \|N_{k,l}f\|_{H^1(T_2M_1)} \le C \|f\|_{L^2(T_2M)}$$

holds for all  $f \in A_{k,l}^{-1}(\operatorname{Sol}(T_2M) \cap H^1(S_2M))$ .

(c) Let  $m \coloneqq k+l \ge 1$ . Assume further that (M, g) has non-positive sectional curvature. Then the estimate

(4.7) 
$$\|f\|_{L^{2}(T_{m}M)} \leq C \|L_{k,l}f\|_{H^{1/2}_{T}(\partial_{\mathrm{in}}SM)}$$

holds for all  $f \in A_{kl}^{-1}(\operatorname{Sol}(T_m M) \cap H^1(S_m M))$ .

*Proof.* (a) We know that the stability estimate holds for the geodesic ray transform [29, Theorem 4]. Now  $A_{k,l}: L^2(T_1M) \to L^2(T_1M)$  and  $A_{k,l}^* = A_{k,l}^{-1}: H^1(T_1M_1) \to H^1(T_1M_1)$  are isometries by lemma 3.11 part (c). By theorem 3.3 part (e) we obtain

(4.8) 
$$\|f\|_{L^2(T_1M)}/C \le \|N_{k,l}f\|_{H^1(T_1M_1)} \le C \|f\|_{L^2(T_1M)}$$

(b) By [28, Theorem 1] the stability estimate holds for the geodesic ray transform if we know s-injectivity. But s-injectivity holds on twodimensional simple manifolds for tensor fields of all order [21, Theorem 1.1]. Using the fact that  $A_{k,l}: L^2(T_2M) \to L^2(T_2M)$  and  $A_{k,l}^*: H^1(T_2M) \to H^1(T_2M)$  are isometries we obtain the stability estimate as in part (a) above.

(c) We know that the stability estimate is true for the geodesic ray transform [20, Theorem 1.3]. Since  $A_{k,l}: L^2(T_mM) \to L^2(T_mM)$  is an isometry theorem 3.3 part (c) implies

(4.9) 
$$||f||_{L^2(T_mM)} \le C ||L_{k,l}f||_{H^{1/2}_T(\partial_{\text{in}}SM)}.$$

This concludes the proof.

*Remark* 4.6. Note that for example the estimate

(4.10) 
$$\|f\|_{L^2}/C \le \|N_{k,l}f\|_{H^1} \le C \|f\|_{L^2}$$

holds for all  $f \in A_{k,l}^{-1}(S'')$  if and only if the estimate

$$(4.11) ||h||_{L^2}/C \le ||Nh||_{H^1} \le C ||h||_{L^2}$$

holds for all  $h \in S''$ . This follows since  $A_{k,l} \colon H^p(T_mM) \to H^p(T_mM)$  is an isometry for all  $p \in \mathbb{N}$  and  $N_{k,l} = (-1)^k A_{k,l} N A_{k,l}$ . Therefore the sets defined in corollary 4.5 are in a sense largest sets where such stability estimates can hold. A similar sharp stability estimate as in part (c) of corollary 4.5 can be proved on compact simple surfaces when m = 1, 2 [2, Theorem 1.1] (see also [4] for the Euclidean case).

4.3. Transverse ray transform of one-forms. Next we study the kernel of the transverse ray transform on one-forms in two dimensions. The result which we obtain is previously known in  $\mathbb{R}^2$  [19]. We recall that in our notation the transverse ray transform is  $I_{\perp}f = I_A f$  where  $A_i = \star$  for all  $i \in \{1, \ldots, m\}$ . For a scalar field  $\phi$ , we define  $\operatorname{curl}(\phi) = e_2(\phi)e^1 - e_1(\phi)e^2$  where  $\{e_1, e_2\}$  is any positively oriented local orthonormal frame and  $\{e^1, e^2\}$  its coframe.

**Corollary 4.7.** Let (M, g) be two-dimensional orientable Riemannian manifold with boundary such that the geodesic ray transform is s-injective on smooth one-forms and let  $f \in C^{\infty}(T_1M)$ . Then  $I_{\perp}f = 0$  if and only if  $f = \operatorname{curl}(\phi)$  for some smooth function  $\phi$  vanishing on the boundary.

Proof. If  $f = \operatorname{curl}(\phi)$  where  $\phi$  vanishes on the boundary, then  $Af = d\phi$ and  $I_{\perp}f = I(Af) = 0$  by the fundamental theorem of calculus. For the converse, if  $I_{\perp}f = 0$ , then I(Af) = 0. By solenoidal injectivity we have that  $Af = d\phi$  for some smooth scalar function  $\phi$  vanishing on the boundary  $\partial M$ . This implies that  $f = A^{-1}d\phi$  which in local positively oriented orthonormal frame  $\{e_1, e_2\}$  means  $f_1 = e_2(\phi)$  and  $f_2 = -e_1(\phi)$ , i.e.  $f = \operatorname{curl}(\phi)$ .

Remark 4.8. We note that on two-dimensional Cartan–Hadamard manifolds one can also deduce from  $I_{\perp}f = 0$  that  $f = \operatorname{curl}(\phi)$  if one of the following assumptions holds

(a)  $-K_0 \leq K \leq 0$  for some  $K_0 > 0$  and  $f \in E_{\eta}^1(T_1M)$  for some  $\eta > \frac{3}{2}\sqrt{K_0}$ 

(b)  $K \in P_{\kappa}(M)$  for some  $\kappa > 2$  and  $f \in P_{\eta}^{1}(T_{1}M)$  for some  $\eta > 2$ .

If we combine the data from the geodesic ray transform If and the transverse ray transform  $I_{\perp}f$ , we can uniquely reconstruct any smooth one-form on two-dimensional compact simple manifolds. This result is also known previously in  $\mathbb{R}^2$  [5]. Recall that  $\Delta_g u = \operatorname{div}((\operatorname{grad}(u))$  where  $\operatorname{grad}(u) = (\operatorname{d} u)^{\sharp}$ .

**Corollary 4.9.** Let (M, g) be a compact simple surface. Then the geodesic ray transform and the transverse ray transform together determine  $f \in C^{\infty}(T_1M)$  uniquely, i.e. if both If = 0 and  $I_{\perp}f = 0$ , then f = 0.

Proof. Since (M, g) is simple, the solenoidal injectivity of I (see [21]) implies that f = du for some smooth function u vanishing on the boundary. On the other hand,  $I_{\perp}f = 0$  gives that  $f = \operatorname{curl}(\phi)$  for some smooth scalar function  $\phi$  by corollary 4.7. But this implies that  $\operatorname{div}(f) = 0$ . Therefore  $\Delta_g u = \operatorname{div}(f) = 0$  so u is a harmonic function vanishing on the boundary. We obtain u = 0 and hence f = 0.

Remark 4.10. One could also use solenoidal decomposition to prove the previous corollary. By the solenoidal decomposition  $f = f^{s} + du$ . Now If = 0 implies that  $f^{s} = 0$ . On the other hand,  $I_{\perp}f = 0$  implies that f is solenoidal, i.e.  $f = f^{s} = 0$ .

The previous corollary holds also on two-dimensional Cartan–Hadamard manifolds as we will prove next. We first state and prove a version of Liouville's theorem on Cartan–Hadamard manifolds.

**Lemma 4.11.** Let (M, g) be a two-dimensional Cartan–Hadamard manifold and u harmonic function on M, i.e.  $\Delta_g u = 0$ . Fix any point  $o \in M$ . Assume that one of the following conditions hold:  $\begin{array}{ll} (a) & -K_0 \leq K \leq 0 \ for \ some \ K_0 > 0 \ and \\ (4.12) & |u(x)| \ |du(x)| \leq Ce^{-\eta d(x,o)} \\ for \ some \ C > 0 \ and \ some \ \eta > \sqrt{K_0}. \\ (b) \ The \ curvature \ satisfies \\ (4.13) & |K(x)| \leq C(1 + d(x,o))^{-\kappa} \\ for \ some \ C > 0 \ and \ \kappa > 2 \ and \ the \ function \ satisfies \\ (4.14) & |u(x)| \ |du(x)| \leq C(1 + d(x,o))^{-\eta} \\ for \ some \ \eta > 1. \end{array}$ 

Then u is constant.

We point out that the conditions above are independent of the choice of  $o \in M$  as in the definition of the spaces in (2.11). Moving the point will only change the constants.

*Proof.* Assume first that (a) holds. Let  $B_r(o)$  be the geodesic ball of radius r > 0 centered at o. Using the integration by parts formula (see [14]) we obtain

$$(4.15) \qquad 0 = \int_{M} u \Delta u dV_g$$
$$= \lim_{r \to \infty} \int_{B_r(o)} u \Delta u dV_g$$
$$= \lim_{r \to \infty} \left( -\int_{B_r(o)} |\operatorname{grad}(u)|_{g_x}^2 dV_g + \int_{S_r(o)} u N(u) d\widehat{V}_g \right)$$

where  $d\hat{V}_g$  is the induced volume form on the geodesic sphere  $S_r(o) = \partial B_r(o)$ and N is the outward unit normal vector field. We focus on the second term. Since  $N(u) = g_x(\operatorname{grad}(u), N)$  and  $|\operatorname{grad}(u)|_{g_x} = |\mathrm{d}u|_{g_x}$ , we can estimate that  $|uN(u)| \leq |u| |\operatorname{grad}(u)|_{g_x} = |u| |\mathrm{d}u|_{g_x}$ . The volume form can be expressed in polar coordinates as  $d\hat{V}_g = J_o(r, \theta) \mathrm{d}\theta$  where  $|J_o(r, \theta)| \leq Ce^{\sqrt{K_0}r}$  [15, Lemma 4.7]. Therefore we obtain

(4.16) 
$$\left| \int_{S_r(o)} u N(u) \mathrm{d}\widehat{V}_g \right| \le C' e^{(-\eta + \sqrt{K_0})r} \xrightarrow{r \to \infty} 0.$$

This implies  $|du|_{g_x} = |\operatorname{grad}(u)|_{g_x} = 0$  and hence du = 0. Connectedness of M implies that u is constant.

If (b) holds, then  $|J_o(r,\theta)| \leq Cr$  [15, Lemma 4.7]. The claim is proved identically as in part (a).

Remark 4.12. One can prove the previous lemma in the exact same way for Cartan–Hadamard manifolds of dimension n > 2 using the growth estimates for the Jacobi fields proved in [16]. In the condition (a) one requires  $\eta > (n-1)\sqrt{K_0}$  and in the condition (b) one requires  $\eta > n-1$ .

**Corollary 4.13.** Let (M, g) be two-dimensional Cartan–Hadamard manifold. Assume that one of the following conditions holds:

(a)  $-K_0 \leq K \leq 0$  for some  $K_0 > 0$  and  $f \in E_n^1(T_1M)$  for some  $\eta > \frac{3}{2}\sqrt{K_0}$ .

(b) The curvature satisfies the estimate (4.13) for some C > 0 and  $\kappa > 2$ and  $f \in P_n^1(T_1M)$  for some  $\eta > 2$ .

Then the geodesic ray transform and the transverse ray transform together determine the one-form f uniquely, i.e. if both If = 0 and  $I_{\perp}f = 0$ , then f = 0.

Proof. Assume that (a) holds. The condition If = 0 implies that f = dh for some  $h \in E_{\eta-\varepsilon}(M)$  where  $\varepsilon > 0$  is arbitrary (see [16]). On the other hand,  $I_{\perp}f = 0$  implies that  $\Delta_g h = \operatorname{div}(f) = 0$ . Hence h is harmonic and satisfies the decay estimate in lemma 4.11. Thus h is constant and f = 0. The proof under the assumption (b) is identical.

#### Appendix A. Notation

#### A.1. Integral transforms.

- If, the geodesic X-ray transform of a tensor field f of order m. See section 2.4 and equations (2.7) and (2.10).
- $I_{SM}h$ , the geodesic ray transform of a function  $h: SM \to \mathbb{R}$ . See section 2.4 and equation (2.8).
- $I_{A,r}f$ , the (abstract) mixing ray transform with a mixing A of degree m, operating on a tensor field f of order m. See section 3.2 and equation (3.11).
- $L_{k,l}f = I_{A_{k,l}}f$ , the mixed ray transform of a tensor field f of order k + l on two-dimensional orientable Riemannian manifold. See section 2.5 and equations 2.19 and 2.20.
- $I_{\perp}f$ , the transverse ray transform of a tensor field f of order k, corresponding to the mixed ray transform with l = 0. See section 2.5 and equation 2.19.
- $I_{A,r}^q[f] = I_{A,r}f$ , the quotient transform of an equivalence class of tensor field f of degree m. See section 3.2.
- $\mathcal{L}_{\beta}f$ , the light ray transform of a (compactly supported) tensor field of order *m*. See section 3.3.3 and equation (3.44).
- $\mathcal{L}_{\beta}^{q}[f] = \mathcal{L}_{\beta}f$ , the quotient light ray transform of an equivalence class of a (compactly supported) tensor field f of degree m. See section 3.3.3.

#### A.2. Other operators on tensor fields.

- A, a mixing composed of automorphisms of the tangent bundle. See section 3.2 and equation (3.9).
- $A_i$ , automorphisms (fiberwise linear bijections) of the tangent bundle. See the beginning of section 3.2.
- $\lambda$  and  $\lambda_x$ , operators converting *m*-tensor field and *m*-tensor into a function on the tangent bundle and tangent space. See section 3.1 and equation (3.3).
- $\lambda_r = r \circ \lambda$  and  $\lambda_{r,x} = r_x \circ \lambda_x$ , where r and  $r_x$  are the restriction operators on the tangent bundle and tangent space. See section 3.1.
- $A_{k,l}$ , the mixing corresponding to the mixed ray transform  $L_{k,l}$ . See section 2.5 and equation 2.19.
- $\sigma$ , the usual symmetrization operator of tensor fields. See section 3.1 and equation (3.1).

24

- $\hat{\sigma}_{A,r}$ , the projection operator onto  $A^{-1}(\text{Ker}(\lambda_r)^{\perp})$ , related to the mixing ray transform  $I_{A,r}$ . See sections 3.1 and 3.2, and equations (3.7) and (3.12).
- $\mathcal{H} = \text{Id} \hat{\sigma}_{A,r}$ , an operator projecting *m*-tensor field onto  $\text{Ker}(\lambda_r \circ A)$ . See sections 3.2 and 3.3, and theorem 3.3.
- $\mathcal{D} = A^{-1} \circ \widetilde{A}$ , an auxiliary operator related to two admissible mixings A and  $\widetilde{A}$  of degree m. See section 3.2 and theorem 3.3.
- $\nabla^{A} = A^{-1} \circ \nabla$ , the weighted covariant derivative of a *m*-tensor field where A is an admissible mixing of degree *m*. See section 3.3.1.
- $N_{k,l}$ , the normal operator of the mixed ray transform  $L_{k,l}$  on compact simple surfaces. See section 4.2 and lemma 4.4.

#### A.3. Other.

- $\mathcal{F}(X)$ , the set of all functions  $X \to \mathbb{C}$ .
- M or (M, g), a connected (pseudo-) Riemannian manifold of dimension  $n \ge 2$ .
- *SM*, the sphere bundle whose fibers are unit spheres of the tangent spaces. See section 2.4.
- $\mathfrak{X}(T_m M)$ , the space of all covariant *m*-tensor fields. See section 2.1.
- $S_m M$ , the space of symmetric *m*-tensor fields. See sections 2.1 and 3.1.
- $C^q(T_mM)$  and  $C^q(S_mM)$ , the set of  $C^q$ -smooth (symmetric) *m*-tensor fields where  $q \in \mathbb{N}$ . See section 2.1.
- $H^k(T_m M)$  and  $H^k(S_m M)$ , the  $L^2$ -Sobolev space of (symmetric) *m*-tensor field where  $k \in \mathbb{N}$ . See section 2.2.
- $P_{\eta}(T_m M)$  and  $P_{\eta}^1(T_m M)$ , the spaces of polynomially decaying *m*-tensor fields on Cartan-Hadamard manifolds. See section 2.4 and equation (2.11).
- $E_{\eta}(T_m M)$  and  $E_{\eta}^1(T_m M)$ , the spaces of exponentially decaying *m*-tensor fields on Cartan–Hadamard manifolds. See section 2.4 and equation (2.11).
- [f] and  $[f]_A$ , the equivalence class of the tensor field f, under the relation  $f \sim h$  if and only if  $f h \in \text{Ker}(\lambda_r \circ A)$ . See sections 3.1, 3.2, 3.3.1 and 3.3.3.

#### References

- A. Abhishek. Support theorems for the transverse ray transform of tensor fields of rank m. Journal of Mathematical Analysis and Applications, 485(2):123828, 2020.
- [2] Y. M. Assylbekov and P. Stefanov. Sharp stability estimate for the geodesic ray transform. *Inverse Problems*, 36(2):025013, 2020.
- [3] A. Behzadan and M. Holst. Sobolev-Slobodeckij Spaces on Compact Manifolds, Revisited. 2017. arXiv:1704.07930.
- [4] J. Boman and V. Sharafutdinov. Stability estimates in tensor tomography. *Inverse Problems and Imaging*, 12(5):1245–1262, 2018.
- [5] H. Braun and A. Hauck. Tomographic Reconstruction of Vector Fields. *IEEE Trans*actions on Signal Processing, 39(2):464–471, 1991.
- [6] M. V. de Hoop, T. Saksala, G. Uhlmann, and J. Zhai. Generic uniqueness and stability for mixed ray transform. 2019. arXiv:1909.11172.
- [7] M. V. de Hoop, T. Saksala, and J. Zhai. Mixed ray transform on simple 2-dimensional Riemannian manifolds. Proc. Amer. Math. Soc., 147(11):4901–4913, 2019.

- [8] A. Feizmohammadi, J. Ilmavirta, and L. Oksanen. The Light Ray Transform in Stationary and Static Lorentzian Geometries. *The Journal of Geometric Analysis*, 2020.
- [9] C. R. Graham, C. Guillarmou, P. Stefanov, and G. Uhlmann. X-Ray Transform and Boundary Rigidity for Asymptotically Hyperbolic Manifolds. *Annales de l'Institut Fourier*, 69(7):2857–2919, 2019.
- [10] C. Guillarmou, M. Lassas, and L. Tzou. X-Ray Transform in Asymptotically Conic Spaces. 2019. arXiv:1910.09631.
- [11] H. Hammer and B. Lionheart. Application of Sharafutdinov's Ray Transform in Integrated Photoelasticity. *Journal of Elasticity*, 75(3):229–246, 2004.
- [12] J. Ilmavirta and F. Monard. Integral geometry on manifolds with boundary and applications. In R. Ramlau and O. Scherzer, editors, *The Radon Transform: The First 100 Years and Beyond.* de Gruyter, 2019.
- [13] J. M. Lee. Introduction to Smooth Manifolds. Springer-Verlag New York, second edition, 2012.
- [14] J. M. Lee. Introduction to Riemannian Manifolds. Springer International Publishing, second edition, 2018.
- [15] J. Lehtonen. The geodesic ray transform on two-dimensional Cartan-Hadamard manifolds. 2016. arXiv:1612.04800.
- [16] J. Lehtonen, J. Railo, and M. Salo. Tensor tomography on Cartan–Hadamard manifolds. *Inverse Problems*, 34(4):044004, 2018.
- [17] W. R. B. Lionheart and P. J. Withers. Diffraction tomography of strain. *Inverse Problems*, 31(4):045005, 2015.
- [18] F. Natterer. The Mathematics of Computerized Tomography. SIAM, Philadelphia, 2001. Reprint.
- [19] F. Natterer and F. Wübbeling. Mathematical Methods in Image Reconstruction. SIAM, Philadelphia, 2001.
- [20] G. P. Paternain and M. Salo. A sharp stability estimate for tensor tomography in non-positive curvature. 2020. arXiv:2001.04334.
- [21] G. P. Paternain, M. Salo, and G. Uhlmann. Tensor tomography on surfaces. *Inven*tiones mathematicae, 193(1):229–247, 2013.
- [22] G. P. Paternain, M. Salo, and G. Uhlmann. Tensor tomography: Progress and challenges. *Chinese Annals of Mathematics, Series B*, 35(3):399–428, 2014.
- [23] L. N. Pestov and V. A. Sharafutdinov. Integral geometry of tensor fields on a manifold of negative curvature. *Siberian Mathematical Journal*, 29(3):427–441, 1988.
- [24] T. Schuster. The importance of the Radon transform in vector field tomography. In R. Ramlau and O. Scherzer, editors, *The Radon Transform: The First 100 Years and Beyond.* de Gruyter, 2019.
- [25] A. Schwarz. Multi-tomographic flame analysis with a schlieren apparatus. Measurement Science and Technology, 7(3):406–413, 1996.
- [26] V. A. Sharafutdinov. Integral geometry of tensor fields. Inverse and Ill-posed Problems Series. VSP, Utrecht, 1994.
- [27] G. Sparr and K. Stråhlén. Vector field tomography, an overview. Technical report, Centre for Mathematical Sciences, Lund Institute of Technology, Lund, Sweden, 1998.
- [28] P. Stefanov. A sharp stability estimate in tensor tomography. Journal of Physics: Conference Series, 124:012007, 2008.
- [29] P. Stefanov and G. Uhlmann. Stability estimates for the X-ray transform of tensor fields and boundary rigidity. *Duke Mathematical Journal*, 123(3):445–467, 2004.
- [30] P. Stefanov, G. Uhlmann, A. Vasy, and H. Zhou. Travel Time Tomography. Acta Mathematica Sinica, English Series, 35:1085–1114, 2019.
- [31] E. G. Thomas. A polarization identity for multilinear maps. Indagationes Mathematicae, 25(3):468–474, 2014.
- [32] G. Thorbergsson. Closed geodesics on non-compact Riemannian manifolds. Mathematische Zeitschrift, 159(3):249–258, 1978.
- [33] K. Wehrheim. Uhlenbeck Compactness. EMS Series of Lectures in Mathematics. European Mathematical Society, 2004.

## [F]

### Boundary rigidity for Randers metrics

K. Mönkkönen

To appear in Annales Fennici Mathematici

https://arxiv.org/abs/2010.11484

Copyright © Suomen matemaattinen yhdistys

#### BOUNDARY RIGIDITY FOR RANDERS METRICS

KEIJO MÖNKKÖNEN

ABSTRACT. If a non-reversible Finsler norm is the sum of a reversible Finsler norm and a closed 1-form, then one can uniquely recover the 1-form up to potential fields from the boundary distance data. We also show a boundary rigidity result for Randers metrics where the reversible Finsler norm is induced by a Riemannian metric which is boundary rigid. Our theorems generalize Riemannian boundary rigidity results to some non-reversible Finsler manifolds. We provide an application to seismology where the seismic wave propagates in a moving medium.

#### 1. INTRODUCTION

In this article we study a certain type of inverse problem for a special class of Finsler norms. The inverse problem we consider is known as the boundary rigidity problem: does the boundary distance data determine the Finsler norm uniquely up to the natural gauge in question? Here we present the problem and our results in a general level; more detailed information can be found in sections 1.1, 1.2 and 2.

Let M be a smooth manifold with boundary  $\partial M$ . A Finsler norm F on M is a non-negative function on the tangent bundle  $F: TM \to [0, \infty)$  such that for each  $x \in M$  the map  $y \mapsto F(x, y)$  defines a positively homogeneous norm in  $T_x M$ . In general, Finsler norms are homogeneous only in positive scalings and they induce a distance function on M which is not necessarily symmetric in contrast to the Riemannian distance function.

Let  $\beta$  be a smooth 1-form on M and  $F_r$  a reversible Finsler norm, i.e.  $F_r(x, -y) = F_r(x, y)$  for all  $x \in M$  and  $y \in T_x M$ . If the norm of  $\beta$  with respect to  $F_r$  is small enough, we can define the non-reversible Finsler norm  $F = F_r + \beta$ . The Finsler norm F is non-reversible in the sense that F(x, -y) = F(x, y) for all  $x \in M$  and  $y \in T_x M$  if and only if  $\beta = 0$ . We can thus think that  $\beta$  is an anisotropic perturbation to the reversible Finsler norm  $F_r$ . We further assume that  $\beta$  is closed (d $\beta = 0$ ) which implies that F and  $F_r$  have the same geodesics as point sets and that F has reversible geodesics.

Suppose we know the boundary distance data of  $F = F_r + \beta$ , i.e. we know the lengths of all geodesics of F connecting two points on the boundary  $\partial M$ . The question is: can we say something about  $\beta$  and  $F_r$  from this information? We prove that if M is simply connected, then one can uniquely

Date: March 26, 2021.

<sup>2020</sup> Mathematics Subject Classification. 53C24, 53A35, 86A22.

Key words and phrases. Inverse problems, boundary rigidity, travel time tomography. Department of Mathematics and Statistics, University of Jyväskylä, P.O. Box 35 (MaD) FI-40014 University of Jyväskylä, Finland; kematamo@student.jyu.fi.

recover the 1-form  $\beta$  (up to potential fields) and the boundary distance data of  $F_r$  from the boundary distance data of F (see theorem 1.3 for a precise statement).

Riemannian metrics form a special class of reversible Finsler norms. Suppose that  $F_r$  is induced by a Riemannian metric g and write  $F_r = F_g$ . If  $\|\beta\|_g < 1$ , then  $F = F_g + \beta$  defines a non-reversible Finsler norm called Randers metric. We say that the Riemannian manifold (M, g) is boundary rigid, if the boundary distance data determines the metric g uniquely up to boundary preserving diffeomorphism. We prove that if M is simply connected and (M, g) is boundary rigid, then (M, F) is also boundary rigid in the sense that one can uniquely recover the 1-form  $\beta$  up to potential fields and the Riemannian metric g up to boundary preserving diffeomorphism from the boundary distance data of F. See theorem 1.5 for a precise statement.

Our proofs are mainly based on the following two facts. First, if two Finsler norms differ only by a closed 1-form, then they are projectively equivalent (they have the same geodesics modulo orientation preserving reparametrizations). Second, since  $F = F_r + \beta$ , we can express the length of any curve  $\gamma$  with respect to  $F_r$  in terms of the symmetric part of the length functional  $L_F(\gamma)$ . Similarly, the integral  $\int_{\gamma} \beta$  can be expressed in terms of the antisymmetric part of  $L_F(\gamma)$ . This allows us to reduce the boundary rigidity problem of F to the boundary rigidity problem of  $F_r$ .

Boundary rigidity has been studied earlier mainly on Riemannian manifolds. Boundary rigidity is known for example for simple subspaces of Euclidean space [34], simple subspaces of symmetric spaces of constant negative curvature [10], conformal simple metrics which agree on the boundary [26, 55] and for certain two-dimensional manifolds including compact simple surfaces [25, 43, 45, 50]. It is also conjectured that compact simple manifolds of any dimension are boundary rigid [43]. Our results generalize the boundary rigidity results to certain Randers metrics whenever the boundary rigidity of the unperturbed Riemannian manifold is known (see theorem 1.5). For a more comprehensive treatment of the boundary rigidity problem in Riemannian geometry, see the review [55].

Closest to our main theorems are rigidity results for magnetic geodesics on Riemannian manifolds. In [27] the authors prove boundary rigidity in the presence of a magnetic field (see also [7] for a generalization). Magnetic geodesics can be seen as geodesics of a Randers metric under additional assumptions for the vector potential which induces the magnetic field (the magnetic field has to be "weak") [36, 56]. There is also a correspondence between Randers metrics and stationary Lorentzian metrics [16, 17, 40] (see [57] for a boundary rigidity result on stationary Lorentzian manifolds). We note that projectively flat Finsler norms (geodesics of the Finsler norm are segments of straight lines) on compact convex domains in  $\mathbb{R}^2$  are completely determined by their boundary distance data [2, 3, 41]. In fact, this holds for a more general class of projective metrics in the plane [41].

Some geometric results similar to the boundary rigidity are known on Finsler manifolds. It was shown in [30] that the collection of boundary distance maps, which measure distances from the interior to the boundary, determines the topological and differential structures of the Finsler manifold. Further, it was shown in [31] that the broken scattering relation (lengths of all geodesics with endpoints on the boundary and reflecting once in the interior) determines the isometry class of reversible Finsler manifolds admitting a strictly convex foliation.

The boundary rigidity problem is known in seismology as the travel time tomography problem where one tries to recover the speed of sound inside the Earth by measuring travel times of seismic waves on the surface. The ray paths of the seismic waves correspond to geodesics and the travel times correspond to lengths of the geodesics. The travel time tomography problem was solved in the beginning of 20th century for spherically symmetric metrics  $g = c^{-2}(r)e$  where e is the Euclidean metric and c = c(r) is a radial sound speed satisfying the Herglotz condition (see equation (1)) [35, 58]. Our results apply to the situation where the seismic wave propagates in a moving medium: one can uniquely recover both the sound speed and the velocity of the medium up to potential fields from travel time measurements (see theorem 1.5 and section 1.2). The linearization of the boundary rigidity or travel time tomography problem leads to tensor tomography where one wants to characterize the kernel of the geodesic ray transform on symmetric 2-tensor fields [52]. For results in this direction and a general overview of tensor tomography, see the reviews [37, 48].

1.1. The main results. Before stating our main results let us briefly introduce some notation; more details can be found in section 2. The proofs of the main theorems can be found in section 3.

We denote by M an n-dimensional smooth manifold with boundary  $\partial M$ where  $n \geq 2$ . We let F be a Finsler norm and  $F_r$  refers to a reversible Finsler norm, i.e.  $F_r(x, -y) = F_r(x, y)$  for all  $x \in M$  and  $y \in T_x M$ . Riemannian metrics are a special case of Finsler norms: if g is a Riemannian metric, then it induces a reversible Finsler norm  $F_g$  as  $F_g(x, y) = \sqrt{g_{ij}(x)y^iy^j}$ . We denote by  $\beta$  a smooth closed 1-form  $(d\beta = 0)$  and  $\|\beta\|_{F^*} = \sup_{x \in M} F^*(x, \beta(x))$  is the dual norm of  $\beta$  with respect to the co-Finsler norm  $F^*$  in  $T^*M$ .

We say that the Finsler norm F is admissible, if for every two boundary points  $x, x' \in \partial M$  there is unique geodesic  $\gamma$  of F with finite length going from x to x'. If F is admissible, then we define the (not necessarily symmetric) map  $d_F(\cdot, \cdot) : \partial M \times \partial M \to [0, \infty)$  by setting  $d_F(x, x') = L_F(\gamma)$ where  $L_F(\gamma)$  denotes the length of the curve  $\gamma$  with respect to F. We call the map  $d_F(\cdot, \cdot)$  the boundary distance data of F. Finally, we say that the Riemannian manifolds  $(M, g_1)$  and  $(M, g_2)$  are boundary rigid, if  $d_{g_1}(x, x') = d_{g_2}(x, x')$  for all  $x, x' \in \partial M$  implies that  $g_2 = \Psi^* g_1$  where  $\Psi \colon M \to M$  is a diffeomorphism such that  $\Psi|_{\partial M} = \text{Id}$ . In other words,  $g_1$ and  $g_2$  are isometric as Riemannian metrics.

We recall that a diffeomorphism  $\Psi: (M, F_2) \to (M, F_1)$  is an isometry between Finsler manifolds if  $\Psi^*F_1 = F_2$ , or equivalently  $\Psi$  preserves the Finslerian distance [6]. We make the following observations before giving our first theorem.

**Remark 1.1.** We note that Finsler norms are very flexible with respect to the boundary distance data, i.e. they are not usually boundary rigid in the same sense as Riemannian metrics. Let  $\Psi: M \to M$  be a diffeomorphism which is identity on the boundary. If  $F_1$  is an admissible

#### KEIJO MÖNKKÖNEN

Finsler norm and  $\phi$  is a scalar field which is constant on the boundary and its differential d $\phi$  has sufficiently small norm with respect to  $\Psi^*F_1$ , then  $F_1$  and  $F_2 = \Psi^*F_1 + d\phi$  give the same boundary distance data (Finslerian isometries preserve geodesics [6] and addition of d $\phi$  only changes parametrizations of geodesics [21]). Especially, if  $F_1$  is reversible, then  $\{\Psi^*F_1 + d\phi : \Psi|_{\partial M} = \text{Id and } \phi|_{\partial M} = \text{constant}\}$  provides a large family of Finsler norms which give the same boundary distances but are not isometric to  $F_1$  (since  $\Psi^*F_1 + d\phi$  is non-reversible whenever  $\phi$  is not constant). See also [12, 22, 23, 38] for results and constructions of non-isometric Finsler norms giving the same boundary distances.

**Remark 1.2.** Finsler norms  $F_1$  and  $F_2$  which satisfy  $F_2 = \Psi^*F_1 + d\phi$ for some scalar field  $\phi$  and diffeomorphism  $\Psi$  are sometimes called almost isometric Finsler norms and the map  $\Psi : (M, F_2) \to (M, F_1)$  is called almost isometry [14, 28, 36, 39]. We show in theorem 1.5 that under certain assumptions the boundary distance data determines Randers metrics up to an almost isometry (see also remark 1.6). Almost isometries have many good properties: they for example are projective transformations which preserve (minimizing) geodesics up to reparametrization [39]. Almost isometries can also be defined on general quasi-metric spaces (X, d). It follows that if  $\Psi : (X_1, d_1) \to (X_2, d_2)$  is an almost isometry between quasi-metric spaces, then  $\Psi$  is an isometry between the metric spaces  $(X_1, \tilde{d}_1) \to (X_2, \tilde{d}_2)$  where  $\tilde{d}_i(p,q) = \frac{1}{2}(d_i(p,q) + d_i(q,p))$  is the symmetrized metric [14, 39]. Especially, in the case of metric spaces almost isometries are isometries.

Our first theorem says that one can uniquely recover (up to potential fields) the perturbation  $\beta$  and the boundary distance data of  $F_r$  from the boundary distance data of  $F = F_r + \beta$ .

**Theorem 1.3.** Let M be a compact and simply connected smooth manifold with boundary. For  $i \in \{1, 2\}$  let  $F_i = F_{r,i} + \beta_i$  be admissible Finsler norms where  $F_{r,i}$  is an admissible and reversible Finsler norm and  $\beta_i$  is a smooth closed 1-form such that  $\|\beta_i\|_{F_{r,i}^*} < 1$ . Then the following are equivalent:

- (i)  $d_{F_1}(x, x') = d_{F_2}(x, x')$  for all  $x, x' \in \partial M$ .
- (ii) There is unique scalar field  $\phi$  vanishing on the boundary such that  $\beta_2 = \beta_1 + d\phi$ , and  $d_{F_{r,1}}(x, x') = d_{F_{r,2}}(x, x')$  for all  $x, x' \in \partial M$ .

**Remark 1.4.** Since  $\beta_i$  is closed and M is simply connected, it follows that  $\beta_i = d\phi_i$  for some scalar field  $\phi_i$ . Thus  $F_{r,i}$  and  $F_i = F_{r,i} + \beta_i = F_{r,i} + d\phi_i$  are almost isometric (but not isometric) Finsler norms (see remark 1.2). Trivially one can define  $\phi = \phi_2 - \phi_1$  so that  $d\phi = \beta_2 - \beta_1$ . The assumption  $d_{F_1}(x, x') = d_{F_2}(x, x')$  for all  $x, x' \in \partial M$  is then used to show that  $\phi$  is constant on the boundary (and one can choose this constant to be zero).

Let us clarify some of our assumptions in theorem 1.3. We need the assumption  $\|\beta_i\|_{F_{r,i}^*} < 1$  to guarantee that the sum  $F_{r,i} + \beta_i$  defines a Finsler norm. Reversibility of  $F_{r,i}$  is needed so that any curve has the same length with respect to  $F_{r,i}$  as any of its reversed reparametrizations. The condition that  $\beta_i$  is closed is used in three places. First, it is equivalent to that  $F_i$  and  $F_{r,i}$  have the same geodesics up to orientation preserving reparametrizations ( $F_i$  and  $F_{r,i}$  are projectively equivalent, see lemma 2.2). Second, closedness of  $\beta_i$  is also equivalent to that  $F_i$  has reversible geodesics ( $F_i$  is projectively reversible, see lemma 2.1). Third,  $d\beta_i = 0$  implies that  $\beta_i$  is exact since M is assumed to be simply connected. All these properties are in a crucial role in our proofs.

The existence of unique geodesics connecting boundary points is used in the proof as well and for this reason we assume that the Finsler norms are admissible. We note that since  $F_i$  and  $F_{r,i}$  are projectively equivalent, the admissibility of  $F_{r,i}$  implies the admissibility of  $F_i$ , and vice versa. We also note that  $d\phi$  is closed so the conclusion  $\beta_2 = \beta_1 + d\phi$  is compatible with the assumptions on  $\beta_i$ . The conclusion that  $\beta_i$  differ only by a potential is similar to the solenoidal injectivity result for the geodesic ray transform of 1-forms [4, 48].

As an application of theorem 1.3 we have the following boundary rigidity result for Randers metrics (see [27, Theorem 6.4] for a similar result).

**Theorem 1.5.** Let M be a compact and simply connected smooth manifold with boundary. For  $i \in \{1, 2\}$  let  $F_i = F_{g_i} + \beta_i$  be admissible Finsler norms where  $g_i$  is an admissible Riemannian metric and  $\beta_i$  is a smooth closed 1form such that  $\|\beta_i\|_{g_i} < 1$ . Assume that  $(M, g_i)$  is boundary rigid. Then the following are equivalent:

- (a)  $d_{F_1}(x, x') = d_{F_2}(x, x')$  for all  $x, x' \in \partial M$ .
- (b) There is unique scalar field  $\phi$  vanishing on the boundary and a diffeomorphism  $\Psi$  which is identity on the boundary such that  $\beta_2 = \beta_1 + d\phi$ and  $g_2 = \Psi^* g_1$ .
- (c) There is unique scalar field  $\phi$  vanishing on the boundary and a diffeomorphism  $\Psi$  which is identity on the boundary such that  $\beta_2 = \Psi^* \beta_1 + d\phi$  and  $g_2 = \Psi^* g_1$ .

**Remark 1.6.** Theorem 1.5 part (c) implies that  $F_2 = \Psi^* F_1 + d\phi$ , i.e. the Randers metrics  $F_1$  and  $F_2$  are almost isometric (see remark 1.2). Hence we obtain a boundary rigidity result for Randers metrics in the special case when the 1-form  $\beta$  is closed and the Riemannian metric g is boundary rigid. This generalizes earlier boundary rigidity results to non-reversible (and hence non-Riemannian) Finsler norms. Note that the diffeomorphism  $\Psi: (M, F_2) \rightarrow (M, F_1)$  in part (c) is an almost isometry but not an isometry since this would require that  $\Psi^*\beta_1 = \beta_2$  [9]. Also note that if  $\beta_1 = 0$  and  $\beta_2 \neq 0$ , then  $F_1$  and  $F_2$  can not be isometric since  $F_1$  is reversible and  $F_2$ is non-reversible.

The assumptions of theorem 1.5 are the same as in theorem 1.3 except that we also assume the boundary rigidity of (M,g). We can simultaneously recover the metric g and the 1-form  $\beta$  from the boundary distance data  $d_F(\cdot, \cdot)$  since the reversibility of  $F_g$  implies that the data for  $\beta_i$  and  $g_i$ "decouple": for any curve  $\gamma$  one can obtain  $\int_{\gamma} \beta$  from the antisymmetric part and  $L_g(\gamma)$  from the symmetric part of the length functional  $L_F(\gamma)$ . We note that in theorems 1.3 and 1.5 we only use the lengths of geodesics connecting boundary points as data.

#### KEIJO MÖNKKÖNEN

Admissible Finsler norms as we have defined are closely related to simple Finsler norms and simple Riemannian metrics. A Riemannian metric g on a smooth manifold M with boundary is simple if it is non-trapping (geodesics have finite length), geodesics have no conjugate points and the boundary  $\partial M$  is strictly convex with respect to g (the second fundamental form on  $\partial M$  is positive definite). See [47, Section 3.7] for many equivalent definitions of simple Riemannian metrics. The concept of a simple Finsler norm can be defined analogously [13, 38]. The simplicity of the Finsler norm or Riemannian metric implies that there exists unique minimizing geodesic between any two points of the manifold [13, 38, 47]. More generally, if the manifold admits a convex function which has a minimum point, then there is a finite number of geodesics between any two non-conjugate points [15, 32] (see also [49]).

We remark that one can take  $(M, g_i)$  to be a compact simple surface in theorem 1.5 since simple Riemannian metrics are admissible and in two dimensions they are boundary rigid [50]. If  $g_1$  and  $g_2$  are simple metrics which are conformal and agree on the boundary, then they are boundary rigid in any dimension  $n \geq 2$  [26, 45, 55].

Theorem 1.5 has an application to Randers metrics arising in seismology (see section 1.2 for more details). Let  $M = \overline{B}(0, R)$  be a closed ball of radius R > 0 and  $g = c^{-2}(r)e$  where e is the Euclidean metric and c = c(r) is a radial sound speed satisfying the Herglotz condition

(1) 
$$\frac{\mathrm{d}}{\mathrm{d}r}\left(\frac{r}{c(r)}\right) > 0, \quad r \in [0, R]$$

It follows that (M, g) is a non-trapping Riemannian manifold with strictly convex boundary [44, 55]. Let us further assume that g has no conjugate points, i.e.  $g = c^{-2}(r)e$  is a simple Riemannian metric. Then g is admissible and one can recover c and hence g uniquely in theorem 1.5 (see [47, Remark 2.10] and [52, 55]). Especially, the diffeomorphism  $\Psi$  becomes identity in this case ( $\Psi = \text{Id}$  also for general conformal simple metrics which agree on the boundary). However,  $\Psi$  can be a nontrivial diffeomorphism for general spherically symmetric Riemannian metrics g (see [29, Appendix C]). We also note that there are sound speeds c satisfying the Herglotz condition (1) such that g has conjugate points (and g is not admissible anymore, see [44, Section 3.3.2 and figure 6]). In section 1.2 we give a physical interpretation for the 1-form  $\beta$  in theorem 1.5 ( $\beta$  corresponds to the flow field of a moving medium).

1.2. Application in seismology. Here we give an application of theorems 1.3 and 1.5 to seismology where the seismic wave propagates in a moving medium. Assume that we have an object moving on a Riemannian manifold (M,g) with constant speed  $||U||_g = 1$ . The speed is fixed, but the object can change the direction of the velocity vector U arbitrarily. Let Wbe a vector field which can be interpreted as the additional velocity resulting from a time-independent external force field acting on the object. The net velocity is U + W and we assume  $||W||_g < 1$  so that the object can move freely in any direction.

Given any two points  $p, q \in M$  we would like to know which path gives the least time when traveling from p to q taking the drift W into account. This is known as the Zermelo's navigation problem (see [9, 20, 54]). It turns out that the unique solution is given by a geodesic of the Randers metric  $F = F_{\alpha} + \beta$  where (see [20, Section 2.2])

$$\alpha_{ij} = \frac{g_{ij}}{\lambda} + \frac{W_i}{\lambda} \frac{W_j}{\lambda}, \quad \beta_i = -\frac{W_i}{\lambda}$$
$$W_i = g_{ij} W^j, \quad \lambda = 1 - \|W\|_g^2$$

and we have left the dependence on  $x \in M$  implicit. Especially, if the parameter of a piecewise smooth curve  $\gamma \colon [0,T] \to M$  represents time, then (see [54, Lemma 3.1] and [21, Lemma 1.4.1])

(2) 
$$T = L_F(\gamma).$$

Let us interpret the object as a seismic wave (or ray) propagating in a moving medium. The manifold M corresponds to the Earth which can be modelled as a compact and simply connected smooth manifold with boundary (a ball). By the Fermat's principle the path of the ray is a critical point of the travel time functional [5, 11, 19]. But since this functional equals to the length functional  $L_F(\gamma)$  of the Randers metric  $F = F_{\alpha} + \beta$  by equation (2), the ray paths of seismic waves correspond to geodesics of F which is the unique solution to the Zermelo's navigation problem.

If our Riemannian metric is of the form  $g = c^{-2}e$  where e is the Euclidean metric and c = c(x) is the sound speed, then  $||U||_g = 1$  is equivalent to  $||U||_e = c$  where  $||\cdot||_e$  is the Euclidean norm of vectors. Thus U corresponds to the velocity of the propagating wave and the medium moves with velocity W for which  $||W||_e < c$ . The components of the Randers metric take the form

$$\begin{aligned} \alpha_{ij} &= \frac{c^{-2} \delta_{ij}}{1 - c^{-2} \|W\|_e^2} + \frac{c^{-4} W^i W^j}{(1 - c^{-2} \|W\|_e^2)^2} \\ \beta_i &= -\frac{c^{-2} W^i}{1 - c^{-2} \|W\|_e^2}. \end{aligned}$$

Note that here we have identified  $W^i = \delta_{ij} W^j$ . Now if the 1-form  $\beta$  is closed, then theorem 1.3 implies that one can uniquely recover  $\beta$  up to potential fields from travel time measurements of seismic waves (assuming admissibility of  $\alpha$ ). In addition, if the Riemannian manifold  $(M, \alpha)$  is boundary rigid, then by theorem 1.5 one can also uniquely recover the Riemannian metric  $\alpha$ up to boundary preserving diffeomorphism from the travel time data.

Let us do the following approximation. If we assume that  $\left\|W\right\|_{e}/c\ll 1,$  then

$$\begin{aligned} \alpha_{ij} &\approx c^{-2} \delta_{ij} + \frac{W^i}{c^2} \frac{W^j}{c^2} \\ \beta_i &\approx -\frac{W^i}{c^2}. \end{aligned}$$

When we only work to first order in  $||W||_e/c$ , the Riemannian metric  $\alpha$  reduces to

$$\alpha_{ij} \approx c^{-2} \delta_{ij} = g_{ij}$$

and the ray paths of seismic waves correspond to geodesics of the Randers metric  $F = F_g + \beta$ . Similar linearization result is obtained in [33] for sound
waves propagating in air under the influence of wind. We also note that the same result can be obtained from the linearization of travel time measurements [46].

If the sound speed c = c(r) is radial, c satisfies the Herglotz condition (1) and  $g = c^{-2}(r)e$  has no conjugate points, then theorem 1.5 implies that in the first order approximation (with respect to  $||W||_e/c$ ) one can uniquely recover the sound speed c and the velocity of the medium W up to potential fields from travel time measurements. If the speed of sound c is constant, then the condition  $d(W/c^2) = 0$  reduces to dW = 0, which in the case of a fluid flow means that W is irrotational (or curl-free). Note that in the approximation we identify  $W^i = \delta_{ij}W^j$ . In general, if c is not constant, then the condition  $d(W/c^2) = 0$  only means that the scaled flow field  $W/c^2$ is irrotational.

To summarize this section: our results (theorems 1.3 and 1.5) apply to the propagation of seismic waves in a moving medium. Under certain assumptions one can recover the velocity of the medium from travel time measurements, and at the same time one reduces the travel time tomography problem in moving medium to the case where no flow field is involved. This allows one to recover the speed of sound as well in the first order approximation.

#### 2. FINSLER MANIFOLDS

In this section we give a brief introduction to Finsler geometry. We only go through definitions and results which are needed in proving our main theorems. Basic theory of Finsler geometry can be found for example in [1, 8, 21, 53]. We use the Einstein summation convention, i.e. indices which appear both as a subscript and superscript are implicitly summed over.

Let M be a smooth manifold. We denote by  $x \in M$  the base point on the manifold and by  $y \in T_x M$  the tangent vectors. A Finsler norm F on M is a non-negative function on the tangent bundle  $F: TM \to [0, \infty)$  such that

- (F1) F is smooth in  $TM \setminus \{0\}$  (smoothness outside zero section)
- (F2) F(x, y) = 0 if and only if y = 0 (positivity)
- (F3)  $F(x, \lambda y) = \lambda F(x, y)$  for all  $\lambda \ge 0$  (positive homogeneity of degree 1)
- (F4)  $\frac{1}{2} \frac{\partial^2 F^2(x,y)}{\partial u^i \partial y^j}$  is positive definite whenever  $y \neq 0$  (convexity).

The pair (M, F) is called a Finsler manifold. If F is a Finsler norm, then one can define the reversed Finsler norm  $\overleftarrow{F}$  by setting  $\overleftarrow{F}(x, y) = F(x, -y)$ . It follows that  $\overleftarrow{F}$  also satisfies the properties (F1)–(F4).

The conditions (F1)–(F4) imply that for every  $x \in M$  the map  $y \mapsto F(x,y)$  defines a positively homogeneous norm in  $T_xM$ . If F(x,-y) = F(x,y) for all  $x \in M$  and  $y \in T_xM$ , we say that the Finsler norm F is reversible (or absolutely homogeneous). If F is reversible, then the map  $y \mapsto F(x,y)$  defines a norm in  $T_xM$ . Every Riemannian metric g = g(x) on M induces a reversible Finsler norm  $F_g$  on M by setting

$$F_g(x,y) = \sqrt{g_{ij}(x)y^i y^j}.$$

The condition (F4) allows us to define the local metric  $g_{ij} = g_{ij}(x, y)$  as

$$g_{ij}(x,y) = \frac{1}{2} \frac{\partial^2 F^2(x,y)}{\partial y^i \partial y^j}.$$

One can then define the Legendre transformation  $L: TM \to T^*M$  using the local metric  $g_{ij}$  (see for example [53, Chapter 3.1]). If  $F: TM \to [0, \infty)$  is a Finsler norm, then by using the Legendre transformation one obtains the dual norm (or co-Finsler norm)  $F^*: T^*M \to [0, \infty)$  satisfying the properties (F1)–(F4) in  $T^*M$ . The dual norm of a covector  $\omega \in T^*_xM$  becomes

$$F^*(x,\omega) = \sup_{\substack{y \in T_x M \\ F(x,y)=1}} \omega(y).$$

If  $F = F_g$  where g = g(x) is a Riemannian metric, then  $g_{ij}(x, y) = g_{ij}(x)y^iy^j$ and the Legendre transformation L and its inverse correspond to the musical isomorphisms.

In this article we study a class of non-reversible Finsler norms. Let  $F_1$  be a Finsler norm on M and  $\beta$  a smooth nonzero 1-form on M. Assume that the dual norm of  $\beta$  satisfies

$$\|\beta\|_{F_1^*} := \sup_{x \in M} F_1^*(x,\beta(x)) < 1.$$

Then  $F = F_1 + \beta$  defines also a Finsler norm on M (see [53, Example 6.3.1] and [8, Chapter 11.1]). We study the special case  $F = F_r + \beta$  where  $F_r$  is a reversible Finsler norm. It follows that Finsler norms of this kind are nonreversible since F(x, -y) = F(x, y) for all  $x \in M$  and  $y \in T_x M$  if and only if  $\beta \equiv 0$ . If  $F_r = F_g$  where g is a Riemannian metric, then  $F = F_g + \beta$  is called a Randers metric (see [51] for the original definition of a Randers metric). Randers metrics are examples of Finsler norms which are not induced by any Riemannian metric (since Riemannian metrics are always reversible).

The length of a piecewise smooth curve  $\gamma \colon [a, b] \to M$  is defined to be

$$L_F(\gamma) = \int_a^b F(\gamma(t), \dot{\gamma}(t)) \mathrm{d}t.$$

In general,  $L_F(\gamma)$  is invariant only in orientation preserving reparametrizations. If in addition F is reversible, then  $L_F(\gamma)$  is also invariant in orientation reversing reparametrizations. When F is induced by a Riemannian metric g, then we simply write  $L_g := L_{F_g}$ . If F is a Finsler norm such that  $F = F_1 + \beta$  where  $F_1$  is a Finsler norm and  $\beta$  is a 1-form, then for any piecewise smooth curve  $\gamma$  we have

$$L_F(\gamma) = L_{F_1}(\gamma) + \int_{\gamma} \beta.$$

Note that for the term coming from the 1-form  $\beta$  we have

$$\int_{\widetilde{\gamma}}\beta=\pm\int_{\gamma}\beta$$

where the plus sign corresponds to reparametrizations  $\tilde{\gamma}$  of  $\gamma$  preserving the orientation and the minus sign corresponds to reparametrizations reversing the orientation.

A smooth curve  $\gamma$  on (M, F) is a geodesic, if it satisfies the geodesic equation

$$\ddot{\gamma}^i(t) + 2G^i(\gamma(t), \dot{\gamma}(t)) = 0$$

where the spray coefficients  $G^i = G^i(x, y)$  are given by

$$G^{i}(x,y) = \frac{1}{4}g^{il}(x,y) \left( y^{k} \frac{\partial^{2}F^{2}(x,y)}{\partial x^{k} \partial y^{l}} - \frac{\partial F^{2}(x,y)}{\partial x^{l}} \right).$$

Here  $g^{ij}(x, y)$  are the components of the inverse matrix of  $g_{ij}(x, y)$ . Geodesics correspond to straightest possible paths in (M, F) and they are locally minimizing. Geodesics are also critical points of the length functional  $L_F(\gamma)$ .

We say that two Finsler norms  $F_1$  and  $F_2$  on a smooth manifold M are projectively equivalent, if  $F_1$  and  $F_2$  have the same geodesics as point sets. More precisely,  $F_1$  and  $F_2$  are projectively equivalent, if for any geodesic  $\gamma$ of  $F_1$  there is an orientation preserving reparametrization  $\eta$  of  $\gamma$  such that  $\eta$ is a geodesic of  $F_2$ , and vice versa. We also say that a Finsler norm F has reversible geodesics (or is projectively reversible), if for any geodesic  $\gamma$  of Fthe reversed curve  $\overline{\gamma}$  is also a geodesic of F up to orientation preserving reparametrization. In other words, F has reversible geodesics if F and  $\overline{F}$ are projectively equivalent.

In general, if  $\gamma$  is a geodesic of F, then the reversed curve  $\overleftarrow{\gamma}$  is not necessarily a geodesic of F. If F is reversible, then  $\overleftarrow{\gamma}$  is also a geodesic. The following lemma says that the same holds (modulo orientation preserving reparametrization) if we perturb a reversible Finsler norm with a closed 1-form (see also [42, Theorem 7.1] for a more general version of the lemma).

**Lemma 2.1** ([24, p. 406]). Let  $F = F_r + \beta$  be a Finsler norm where  $F_r$  is a reversible Finsler norm and  $\beta$  is a 1-form such that  $\|\beta\|_{F_r^*} < 1$ . Then Fhas reversible geodesics (is projectively equivalent to  $\overleftarrow{F}$ ) if and only if  $\beta$  is closed (d $\beta = 0$ ).

The next lemma has a central role in the proofs of our main theorems. It says that if we perturb a Finsler norm with a closed 1-form, then the geodesics change only by an orientation preserving reparametrization (see also [18, Theorem 3.3] and [1, Example 2.11]).

**Lemma 2.2** ([21, Theorem 3.3.1 and Example 3.3.2]). Let  $F_1$  be a Finsler norm on a smooth manifold M. Let  $F_2 = F_1 + \beta$  be another Finsler norm where  $\beta$  is a 1-form such that  $\|\beta\|_{F_1^*} < 1$ . Then  $(M, F_1)$  and  $(M, F_2)$  are projectively equivalent if and only if  $\beta$  is closed  $(d\beta=0)$ .

#### 3. PROOFS OF THE MAIN THEOREMS

In this section we prove our main results. The proofs are based on lemmas 2.1 and 2.2 which imply that  $F_i$  and  $F_{r,i}$  have the same geodesics up to orientation preserving reparametrizations and that  $F_i$  has reversible geodesics. This allows us to express the integrals of the 1-forms  $\beta_i$  in terms of the boundary distance data of  $F_i$ . Similarly we can express the boundary distance data of  $g_i$  in terms of the boundary distance data of  $F_i$ , which implies that  $g_1$  and  $g_2$  differ only by a boundary preserving diffeomorphism since the underlying manifolds  $(M, g_i)$  are assumed to be boundary rigid.

10

We are now ready to prove our main theorems. Recall that a Finsler norm F is admissible if every two boundary points can be joined by unique geodesic of F with finite length.

Proof of theorem 1.3. Let us first prove the direction (i) $\Rightarrow$ (ii). We note that if  $\gamma$  is any curve on M and  $\overleftarrow{\gamma}$  any of its reversed reparametrizations, then reversibility of  $F_{r,i}$  implies that  $L_{F_{r,i}}(\gamma) = L_{F_{r,i}}(\overleftarrow{\gamma})$  and

$$\int_{\gamma} \beta_i = \frac{L_{F_i}(\gamma) - L_{F_i}(\overleftarrow{\gamma})}{2}$$

Let  $x, x' \in \partial M$  and  $\gamma_i$  be the unique geodesic of  $F_{r,i}$  connecting x to x' (see figure 1). By lemma 2.2 there is an orientation preserving reparametrization  $\eta_i$  of  $\gamma_i$  such that  $\eta_i$  is a geodesic of  $F_i$ . Note that since  $\eta_i$  connects xto x' we have  $L_{F_i}(\eta_i) = d_{F_i}(x, x')$  by admissibility of  $F_i$ . Using lemma 2.1 let  $\overline{\eta_i}$  be the reversed curve which is a geodesic of  $F_i$ . Now again by admissibility of  $F_i$  we have  $L_{F_i}(\overline{\eta_i}) = d_{F_i}(x', x)$  since  $\overline{\eta_i}$  connects x' to x. We obtain

$$\int_{\gamma_1} \beta_1 = \int_{\eta_1} \beta_1 = \frac{L_{F_1}(\eta_1) - L_{F_1}(\overline{\eta_1})}{2} = \frac{d_{F_1}(x, x') - d_{F_1}(x', x)}{2}$$
$$= \frac{d_{F_2}(x, x') - d_{F_2}(x', x)}{2} = \frac{L_{F_2}(\eta_2) - L_{F_2}(\overline{\eta_2})}{2} = \int_{\eta_2} \beta_2 = \int_{\gamma_2} \beta_2.$$

The closed 1-form  $\beta_i$  is exact because M is simply connected, i.e.  $\beta_i = d\phi_i$  for some scalar field  $\phi_i$ . Since  $\gamma_1$  and  $\gamma_2$  both connect x to x', we obtain that

$$\phi_1(x') - \phi_1(x) = \int_{\gamma_1} \beta_1 = \int_{\gamma_2} \beta_2 = \phi_2(x') - \phi_2(x).$$

It follows that  $\phi_2 - \phi_1$  is constant on the boundary. Let this constant be  $c \in \mathbb{R}$ and define the scalar field  $\phi = \phi_2 - \phi_1 - c$ . Then  $\phi$  satisfies  $d\phi = \beta_2 - \beta_1$ and  $\phi|_{\partial M} = 0$ . If there is another scalar field  $\phi'$  such that  $d\phi' = \beta_2 - \beta_1$ and  $\phi'|_{\partial M} = 0$ , then  $d(\phi - \phi') = 0$  and  $\phi - \phi' = \text{constant} = 0$  since M is connected and both scalar fields vanish on the boundary. This proves the first claim of the first implication.

For the second claim we note that for any curve  $\gamma$  and any of its reversed reparametrization  $\overleftarrow{\gamma}$  it holds that

$$L_{F_{r,i}}(\gamma) = \frac{L_{F_i}(\gamma) + L_{F_i}(\overleftarrow{\gamma})}{2}.$$

Now let  $\gamma_i$  and  $\eta_i$  be as in the beginning of the proof. It follows that

$$d_{F_{r,1}}(x,x') = L_{F_{r,1}}(\gamma_1) = L_{F_{r,1}}(\eta_1) = \frac{L_{F_1}(\eta_1) + L_{F_1}(\overline{\eta_1})}{2}$$
  
=  $\frac{d_{F_1}(x,x') + d_{F_1}(x',x)}{2} = \frac{d_{F_2}(x,x') + d_{F_2}(x',x)}{2}$   
=  $\frac{L_{F_2}(\eta_2) + L_{F_2}(\overline{\eta_2})}{2} = L_{F_{r,2}}(\eta_2) = L_{F_{r,2}}(\gamma_2) = d_{F_{r,2}}(x,x')$ 

proving the second claim of the first implication.

Let us then prove the implication (ii) $\Rightarrow$ (i). First we note that for any curve  $\gamma$  it holds that

$$L_{F_i}(\gamma) = L_{F_{r,i}}(\gamma) + \int_{\gamma} \beta_i.$$

Let  $x, x' \in \partial M$  and  $\eta_i$  be the unique geodesic of  $F_i$  connecting x to x'. By lemma 2.2 there is an orientation preserving reparametrization  $\sigma_i$  of  $\eta_i$  such that  $\sigma_i$  is a geodesic of  $F_{r,i}$ . Simply connectedness of M and the assumptions on  $\beta_i$  imply that  $\int_{\eta_2} \beta_2 = \int_{\eta_1} \beta_1$ . Using the assumption  $d_{F_{r,1}}(x, x') = d_{F_{r,2}}(x, x')$  and the admissibility of  $F_{r,i}$  it follows that

$$d_{F_2}(x, x') = L_{F_2}(\eta_2) = L_{F_{r,2}}(\eta_2) + \int_{\eta_2} \beta_2 = L_{F_{r,2}}(\sigma_2) + \int_{\eta_1} \beta_1$$
  
=  $d_{F_{r,2}}(x, x') + \int_{\eta_1} \beta_1 = d_{F_{r,1}}(x, x') + \int_{\eta_1} \beta_1$   
=  $L_{F_{r,1}}(\sigma_1) + \int_{\eta_1} \beta_1 = L_{F_{r,1}}(\eta_1) + \int_{\eta_1} \beta_1$   
=  $L_{F_1}(\eta_1) = d_{F_1}(x, x').$ 

This concludes the proof.



FIGURE 1. A picture illustrating the proof of theorem 1.3. Here  $\gamma_i$  is a geodesic of  $F_{r,i}$  connecting the boundary point  $x \in \partial M$  to the boundary point  $x' \in \partial M$ . The curves  $\gamma_i$  are also geodesics of  $F_i$  up to orientation preserving reparametrization by lemma 2.2. The picture is highly simplified; in reality the curves  $\gamma_1$  and  $\gamma_2$  can for example cross each other.

12

Proof of theorem 1.5. If  $d_{F_1}(x, x') = d_{F_2}(x, x')$  for all  $x, x' \in \partial M$ , then by theorem 1.3 there is unique scalar field  $\phi$  vanishing on the boundary such that  $\beta_2 = \beta_1 + d\phi$ , and  $d_{g_1}(x, x') = d_{g_2}(x, x')$  for all  $x, x' \in \partial M$ . Since we assume that  $(M, g_i)$  are boundary rigid, there is a diffeomorphism  $\Psi \colon M \to$ M which is identity on the boundary such that  $g_2 = \Psi^* g_1$ . This proves the implication (a) $\Rightarrow$ (b). The implication (b) $\Rightarrow$ (a) is proved in the same way as the implication (ii) $\Rightarrow$ (i) in theorem 1.3 using the fact that  $\Psi$  is a Riemannian isometry fixing boundary points.

Let us then prove the equivalence (b) $\Leftrightarrow$ (c). Let  $\Psi: M \to M$  be a diffeomorphism which is identity on the boundary. Since  $\beta_1$  is closed and the pullback commutes with the differential, we have that  $\Psi^*\beta_1$  is also closed. This implies that  $\beta_1 - \Psi^*\beta_1$  is closed and hence exact because M is simply connected, i.e. there is a scalar field  $\tilde{\phi}$  such that  $\beta_1 - \Psi^*\beta_1 = d\tilde{\phi}$ . Let  $x, x' \in \partial M$  be any two boundary points and  $\gamma$  any curve connecting x to x'. Since  $\Psi$  is identity on the boundary and  $\beta_1$  is exact we have

$$0 = \int_{\gamma} (\beta_1 - \Psi^* \beta_1) = \int_{\gamma} d\widetilde{\phi} = \widetilde{\phi}(x') - \widetilde{\phi}(x).$$

Therefore  $\phi$  is constant on the boundary and we can subtract this constant to obtain a scalar field  $\phi'$  such that  $\beta_1 - \Psi^* \beta_1 = d\phi'$  and  $\phi'|_{\partial M} = 0$ . Thus  $\beta_1$  and  $\Psi^* \beta_1$  differ only by a potential which vanishes on the boundary, concluding the proof.

Acknowledgements. The author was supported by Academy of Finland (Centre of Excellence in Inverse Modelling and Imaging, grant numbers 284715 and 309963). The author is grateful to Joonas Ilmavirta for helpful discussions and suggestions to improve the article. The author wants to thank Jesse Railo and Teemu Saksala for discussions, and Árpád Kurusa for providing access to the article [41]. The author also wishes to thank the anonymous referee for helpful comments.

#### References

- T. Aikou and L. Kozma. Global aspects of Finsler geometry. In D. Krupka and D. Saunders, editors, *Handbook of Global Analysis*, pages 1–39. Elsevier, Amsterdam, 2008.
- R. Alexander. Planes for which the lines are the shortest paths between points. Illinois J. Math., 22(2):177–190, 1978.
- [3] R. V. Ambartzumian. A note on pseudo-metrics on the plane. Z. Wahrscheinlichkeitstheorie verw. Gebiete, 37(2):145–155, 1976.
- [4] Y. E. Anikonov and V. G. Romanov. On uniqueness of determination of a form of first degree by its integrals along geodesics. J. Inverse Ill-Posed Probl., 5(6):487–490, 1997.
- [5] P. L. Antonelli, A. Bóna, and M. A. Slawiński. Seismic rays as Finsler geodesics. Nonlinear Anal. Real World Appl., 4(5):711–722, 2003.
- [6] B. Aradi and D. C. Kertész. Isometries, submetries and distance coordinates on Finsler manifolds. Acta Math. Hungar., 143(2):337–350, 2014.
- [7] Y. M. Assylbekov and H. Zhou. Boundary and scattering rigidity problems in the presence of a magnetic field and a potential. *Inverse Probl. Imaging*, 9(4):935–950, 2015.
- [8] D. Bao, S.-S. Chern, and Z. Shen. An Introduction to Riemann-Finsler Geometry. Springer-Verlag, first edition, 2000.

#### KEIJO MÖNKKÖNEN

- [9] D. Bao, C. Robles, and Z. Shen. Zermelo navigation on Riemannian manifolds. J. Differential Geom., 66(3):377–435, 2004.
- [10] G. Besson, G. Courtois, and S. Gallot. Entropies et rigidités des espaces localement symétriques de courbure strictement négative. *Geom. Funct. Anal.*, 5(5):731–799, 1995.
- [11] A. Bóna and M. A. Slawiński. Fermat's principle for seismic rays in elastic media. J. Appl. Geophys., 54(3):445–451, 2003.
- [12] D. Burago and S. Ivanov. Boundary rigidity and filling volume minimality of metrics close to a flat one. Ann. of Math. (2), 171(2):1183–1211, 2010.
- [13] D. Burago and S. Ivanov. Boundary distance, lens maps and entropy of geodesic flows of Finsler metrics. *Geom. Topol.*, 20(1):469–490, 2016.
- [14] J. Cabello and J. A. Jaramillo. A functional representation of almost isometries. J. Math. Anal. Appl., 445(2):1243–1257, 2017. A special issue of JMAA dedicated to Richard Aron.
- [15] E. Caponio, M. A. Javaloyes, and A. Masiello. Finsler geodesics in the presence of a convex function and their applications. J. Phys. A: Math. Theor., 43(13):135207, 2010.
- [16] E. Caponio, M. A. Javaloyes, and A. Masiello. On the energy functional on Finsler manifolds and applications to stationary spacetimes. *Math. Ann.*, 351(2):365–392, 2011.
- [17] E. Caponio, M. A. Javaloyes, and M. Sánchez. On the interplay between Lorentzian Causality and Finsler metrics of Randers type. *Rev. Mat. Iberoamericana*, 27(3):919– 952, 2011.
- [18] C. J. Catone. Projective equivalence of Finsler and Riemannian surfaces. Differential Geom. Appl., 26(4):404–418, 2008.
- [19] V. Cerveny. Seismic Ray Theory. Cambridge University Press, 2001.
- [20] X. Cheng and Z. Shen. Finsler Geometry, An Approach via Randers Spaces. Springer, 2012.
- [21] S.-S. Chern and Z. Shen. Riemann-Finsler Geometry. World Scientific, 2005.
- [22] B. Colbois, F. Newberger, and P. Verovic. Some smooth Finsler deformations of hyperbolic surfaces. Ann. Global Anal. Geom., 35(2):191–226, 2009.
- [23] D. Cooper and K. Delp. The marked length spectrum of a projective manifold or orbifold. Proc. Am. Math. Soc., 138(9):3361–3376, 2010.
- [24] M. Crampin. Randers spaces with reversible geodesics. Publ. Math. Debrecen, 67(3– 4):401–409, 2005.
- [25] C. B. Croke. Rigidity for surfaces of non-positive curvature. Comment. Math. Helv., 65(1):150–169, 1990.
- [26] C. B. Croke. Rigidity and the distance between boundary points. J. Differential Geom., 33(2):445-464, 1991.
- [27] N. S. Dairbekov, G. P. Paternain, P. Stefanov, and G. Uhlmann. The boundary rigidity problem in the presence of a magnetic field. *Adv. Math.*, 216(2):535–609, 2007.
- [28] A. Daniilidis, J. A. Jaramillo, and F. Venegas M. Smooth semi-Lipschitz functions and almost isometries between Finsler manifolds. J. Funct. Anal., 279(8):108662, 2020.
- [29] M. V. de Hoop, J. Ilmavirta, and V. Katsnelson. Spectral rigidity for spherically symmetric manifolds with boundary. 2017. arXiv:1705.10434.
- [30] M. V. de Hoop, J. Ilmavirta, M. Lassas, and T. Saksala. Inverse problem for compact Finsler manifolds with the boundary distance map. 2019. arXiv:1901.03902.
- [31] M. V. de Hoop, J. Ilmavirta, M. Lassas, and T. Saksala. A foliated and reversible Finsler manifold is determined by its broken scattering relation. 2020. arXiv:2003.12657.
- [32] F. Giannoni, A. Masiello, and P. Piccione. Convexity and the finiteness of the number of geodesics. Applications to the multiple-image effect. *Class. Quantum Grav.*, 16(3):731–748, 1999.
- [33] G. Gibbons and C. Warnick. The geometry of sound rays in a wind. Contemp. Phys., 52(3):197–209, 2011.
- [34] M. Gromov. Filling Riemannian manifolds. J. Differential Geom., 18(1):1-147, 1983.

14

- [35] G. Herglotz. Über die Elastizität der Erde bei Berücksichtigung ihrer variablen Dichte. Zeitschr. für Math. Phys., 52:275–299, 1905.
- [36] J. Herrera and M. A. Javaloyes. Stationary-Complete Spacetimes with non-standard splittings and pre-Randers metrics. J. Geom. Phys, 163:104120, 2021.
- [37] J. Ilmavirta and F. Monard. Integral geometry on manifolds with boundary and applications. In R. Ramlau and O. Scherzer, editors, *The Radon Transform: The First 100 Years and Beyond.* de Gruyter, 2019.
- [38] S. Ivanov. Local monotonicity of Riemannian and Finsler volume with respect to boundary distances. *Geom. Dedicata*, 164(1):83–96, 2013.
- [39] M. A. Javaloyes, L. Lichtenfelz, and P. Piccione. Almost isometries of non-reversible metrics with applications to stationary spacetimes. J. Geom. Phys., 89:38–49, 2015.
- [40] M. A. Javaloyes, E. Pendas-Recondo, and M. Sanchez. Applications of cone structures to the anisotropic rheonomic Huygens' principle. *Nonlinear Anal.*, 2020. To appear.
- [41] Á. Kurusa and T. Ódor. Boundary-rigidity of projective metrics and the geodesic X-ray transform. 2020. Preprint.
- [42] I. M. Masca, V. S. Sabau, and H. Shimada. Reversible geodesics for  $(\alpha, \beta)$ -metrics. Internat. J. Math., 21(08):1071–1094, 2010.
- [43] R. Michel. Sur la rigidité imposée par la longueur des géodésiques. Invent. Math., 65(1):71–83, 1981.
- [44] F. Monard. Numerical Implementation of Geodesic X-Ray Transforms and Their Inversion. SIAM J. Imaging Sci., 7(2):1335–1357, 2014.
- [45] R. G. Mukhometov. The reconstruction problem of a two-dimensional Riemannian metric, and integral geometry (Russian). Dokl. Akad. Nauk SSSR, 232(1):32–35, 1977.
- [46] S. J. Norton. Tomographic Reconstruction of 2-D Vector Fields: Application to Flow Imaging. *Geophys. J. Int.*, 97(1):161–168, 1989.
- [47] G. Paternain, M. Salo, and G. Uhlmann. Geometric inverse problems in 2D. 2020. Draft version.
- [48] G. P. Paternain, M. Salo, and G. Uhlmann. Tensor tomography: Progress and challenges. *Chin. Ann. Math. Ser. B*, 35(3):399–428, 2014.
- [49] G. P. Paternain, M. Salo, G. Uhlmann, and H. Zhou. The geodesic X-ray transform with matrix weights. Amer. J. Math., 141(6):1707–1750, 2019.
- [50] L. Pestov and G. Uhlmann. Two dimensional compact simple Riemannian manifolds are boundary distance rigid. Ann. of Math. (2), 161(2):1093–1110, 2005.
- [51] G. Randers. On an Asymmetrical Metric in the Four-Space of General Relativity. *Phys. Rev.*, 59:195–199, 1941.
- [52] V. A. Sharafutdinov. Ray Transform on Riemannian Manifolds. In K. Bingham, Y. V. Kurylev, and E. Somersalo, editors, New Analytic and Geometric Methods in Inverse Problems, pages 187–259. Springer, 2004.
- [53] Z. Shen. Lectures on Finsler Geometry. World Scientific, 2001.
- [54] Z. Shen. Finsler Metrics with K=0 and S=0. Canad. J. Math., 55(1):112–132, 2003.
- [55] P. Stefanov, G. Uhlmann, A. Vasy, and H. Zhou. Travel Time Tomography. Acta Math. Sin. (Engl. Ser.), 35:1085–1114, 2019.
- [56] S. Tabachnikov. Remarks on magnetic flows and magnetic billiards, Finsler metrics and a magnetic analog of Hilbert's fourth problem. 2003. arXiv:0302288.
- [57] G. Uhlmann, Y. Yang, and H. Zhou. Travel Time Tomography in Stationary Spacetimes. J. Geom. Anal., 2021. Published online.
- [58] E. Wiechert and K. Zoeppritz. Über Erdbebenwellen. Nachr. Königl. Ges. Wiss. Göttingen, 4:415–549, 1907.

[G]

# Partial data problems and unique continuation in scalar and vector field tomography

J. Ilmavirta and K. Mönkkönen

Preprint (2021)

https://arxiv.org/abs/2103.14385

# PARTIAL DATA PROBLEMS AND UNIQUE CONTINUATION IN SCALAR AND VECTOR FIELD TOMOGRAPHY

#### JOONAS ILMAVIRTA AND KEIJO MÖNKKÖNEN

ABSTRACT. We prove that if P(D) is some constant coefficient partial differential operator and f is a scalar field such that P(D)f vanishes in a given open set, then the integrals of f over all lines intersecting that open set determine the scalar field uniquely everywhere. This is done by proving a unique continuation property of fractional Laplacians which implies uniqueness for the partial data problem. We also apply our results to partial data problems of vector fields.

## 1. INTRODUCTION

Let f be a scalar field and  $V \subset \mathbb{R}^n$  a nonempty open set where  $n \geq 2$ . We study the following partial data problem in X-ray tomography: can we say something about f if we know the integrals of f over all lines intersecting V? Especially, we are interested in the uniqueness problem which can be formulated in terms of the X-ray transform  $X_0$  as follows: if  $X_0 f = 0$ on all lines which intersect V, does it follow that f = 0? In general, the answer is "no" [29] and one has to put some conditions on  $f|_V$ . We prove that if there is some constant coefficient partial differential operator P(D)such that  $P(D)f|_V = 0$  and  $X_0f = 0$  on all lines intersecting V, then f = 0. This generalizes a recent partial data result in [20]. As a special case we obtain that if f is for example polynomial or (poly)harmonic in V, then fis uniquely determined by its partial X-ray data.

The partial data result is proved by using a unique continuation property of fractional Laplacian  $(-\Delta)^s$ . We prove that if  $s \in (-n/2, \infty) \setminus \mathbb{Z}$  and there is some constant coefficient partial differential operator P(D) such that  $P(D)f|_V = (-\Delta)^s f|_V = 0$ , then f = 0. This generalizes earlier results about unique continuation of fractional Laplacians [6, 14]. The unique continuation of  $(-\Delta)^s$  implies a unique continuation result for the normal operator  $N_0$  of the X-ray transform  $X_0$ , and the uniqueness for the partial data problem follows directly from the unique continuation of  $N_0$ . This approach which uses the unique continuation of the normal operator in proving uniqueness for partial data problems was also used in [6, 20, 21].

Date: April 22, 2021.

<sup>2020</sup> Mathematics Subject Classification. 44A12, 46F12, 58A10.

Key words and phrases. Inverse problems, X-ray tomography, vector field tomography, region of interest tomography, unique continuation.

Unit of Computing Sciences, Tampere University, FI-33014 Tampere University, Finland; joonas.ilmavirta@tuni.fi.

Department of Mathematics and Statistics, University of Jyväskylä, P.O. Box 35 (MaD) FI-40014 University of Jyväskylä, Finland; kematamo@student.jyu.fi.

We also study partial data problems of vector fields. Let F be a vector field and denote by dF its exterior derivative or curl which components are  $(dF)_{ij} = \partial_i F_j - \partial_j F_i$ . We prove that if there are some constant coefficient partial differential operators  $P_{ij}(D)$  such that  $P_{ij}(D)(dF)_{ij}|_V = 0$  and the integrals of F over all lines intersecting V vanish, then F must be a potential field (F is the gradient of some scalar field). This is a generalization of a recent result in [21]. The partial data result is proved by using a relation between the normal operator of the X-ray transform of scalar fields and the normal operator of the X-ray transform of vector fields (see lemma 4.4). This allows one to reduce the partial data problem for the vector field F to partial data problems for the scalar fields  $(dF)_{ij}$ . As a special case we obtain that if F is for example componentwise polynomial or (poly)harmonic in V, then the solenoidal part of F is uniquely determined by the partial X-ray data of F.

The partial data problems we study have a relation to the region of interest (ROI) tomography [4, 23, 24, 29, 46]. The main goal in such imaging problems is to determine the attenuation inside a small part of a human body (region of interest) by using only the X-ray data on lines which go through the ROI. This for example reduces the needed X-ray dose which is given to the patient. Our results imply that if the attenuation f satisfies  $P(D)f|_V = 0$  for some open subset V of the ROI and some constant coefficient partial differential operator P(D), then f is uniquely determined by its partial X-ray data on lines which intersect the ROI. Note that f is uniquely determined not only in the ROI but also outside the ROI. This holds for example if the attenuation is polynomial or (poly)harmonic in a small subregion of the ROI. In general, f does not have to be smooth and it can have singularities in the ROI. We also note that our proof for uniqueness does not give stability for the partial data problem. Especially, outside the ROI we have invisible singularities which cannot be seen by the X-ray data and the reconstruction of such singularities is not stable (see remark 1.5 and [25, 34, 35]).

Similar ROI tomography problems can be studied in the case of vector fields. In vector field tomography the usual objective is to determine the velocity field of a fluid flow using acoustic travel time or Doppler backscattering measurements [30, 31, 39]. Assuming that the velocity of the fluid flow is much smaller than the speed of the propagating signal one can linearize the problem. Linearization then leads to the X-ray transform of the velocity field. Our results imply that if the velocity field F satisfies  $P_{ij}(D)(dF)_{ij}|_V = 0$  for some open subset V of the ROI and some constant coefficient partial differential operators  $P_{ij}(D)$ , then the solenoidal part of Fis uniquely determined everywhere by the partial X-ray data of F on lines intersecting the ROI. Examples of such velocity fields are those which are componentwise polynomial or (poly)harmonic in a small subregion of the ROI. As in the scalar case, F can have singularities in the ROI, and our proof does not give stability for the partial data problem (since it is based on reduction to the scalar case).

The article is organized as follows. In section 1.1 we introduce our notation, in section 1.2 we give our main theorems and in section 1.3 we discuss some related results. We go through the theory of distributions and the X-ray transform in section 2, and study the space of admissible functions in section 3. Finally, we prove our main results in section 4.

1.1. Notation. We quickly go through the notation used in our main theorems. More detailed information about distributions and the X-ray transform of scalar and vector fields can be found in section 2.

We denote by f a scalar field. The set  $\mathscr{O}'_{C}(\mathbb{R}^{n})$  is the space of rapidly decreasing distributions and the space  $\mathscr{E}'(\mathbb{R}^{n}) \subset \mathscr{O}'_{C}(\mathbb{R}^{n})$  consists of compactly supported distributions. The subset  $C_{\infty}(\mathbb{R}^{n}) \subset \mathscr{O}'_{C}(\mathbb{R}^{n})$  is the set of all continuous functions which decay faster than any polynomial at infinity. We let  $X_{0}$  be the X-ray transform of scalar fields and it maps a function to its line integrals. The normal operator is  $N_{0} = X_{0}^{*}X_{0}$  where  $X_{0}^{*}$  is the adjoint of  $X_{0}$ .

We let  $H^r(\mathbb{R}^n)$  be the fractional  $L^2$ -Sobolev space of order  $r \in \mathbb{R}$  and  $H^{-\infty}(\mathbb{R}^n) = \bigcup_{r \in \mathbb{R}} H^r(\mathbb{R}^n)$ . We define the fractional Laplacian as  $(-\Delta)^s f = \mathcal{F}^{-1}(|\cdot|^{2s} \hat{f})$  where  $\hat{f} = \mathcal{F}(f)$  is the Fourier transform of f and  $\mathcal{F}^{-1}$  is the inverse Fourier transform. The fractional Laplacian  $(-\Delta)^s$  is well-defined in  $\mathscr{O}'_C(\mathbb{R}^n)$  for all  $s \in (-n/2, \infty) \setminus \mathbb{Z}$  and in  $H^r(\mathbb{R}^n)$  for all  $s \in (-n/4, \infty) \setminus \mathbb{Z}$ .

We denote by  $\mathcal{P}$  the set of all polynomials in  $\mathbb{R}^n$  with complex coefficients with the convention that the zero polynomial  $P \equiv 0$  does not belong to  $\mathcal{P}$ . A polynomial  $P \in \mathcal{P}$  of degree  $m \in \mathbb{N}$  induces a constant coefficient partial differential operator P(D) of order  $m \in \mathbb{N}$  by setting  $P(D) = \sum_{|\alpha| \leq m} a_{\alpha} D^{\alpha}$ where  $a_{\alpha} \in \mathbb{C}$ ,  $D^{\alpha} = D_1^{\alpha_1} \cdots D_n^{\alpha_n}$ ,  $D_j = -i\partial_j$  and  $\alpha = (\alpha_1, \ldots, \alpha_n) \in \mathbb{N}^n$  is a multi-index such that  $|\alpha| = \alpha_1 + \ldots + \alpha_n$ . The set of admissible functions  $\mathcal{A}_V$ is defined as

(1) 
$$\mathcal{A}_V = \{ f \in H^{-\infty}(\mathbb{R}^n) : P(D)f|_V = 0 \text{ for some } P \in \mathcal{P} \}$$

where  $V \subset \mathbb{R}^n$  is some nonempty open set.

We denote by F a vector field. The notation  $F \in (\mathcal{E}'(\mathbb{R}^n))^n$  means that  $F = (F_1, \ldots, F_n)$  where  $F_i \in \mathcal{E}'(\mathbb{R}^n)$  for all  $i = 1, \ldots, n$ . The exterior derivative of F is written in components as  $(dF)_{ij} = \partial_i F_j - \partial_j F_i$ . For scalar fields  $\phi$  the notation  $d\phi$  denotes the gradient of  $\phi$ . We let  $X_1$  be the X-ray transform of vector fields which maps a vector field to its line integrals. The normal operator is  $N_1 = X_1^* X_1$  where  $X_1^*$  is the adjoint of  $X_1$ .

1.2. Main results. In this section we give our main theorems. The proofs of the results can be found in section 4.

Our main theorem is the following unique continuation result for the fractional Laplacian.

**Theorem 1.1.** Let  $n \ge 1$ ,  $s \in (-n/4, \infty) \setminus \mathbb{Z}$  and  $f \in \mathcal{A}_V$  where  $V \subset \mathbb{R}^n$  is some nonempty open set. If  $(-\Delta)^s f|_V = 0$ , then f = 0. If  $f \in \mathscr{O}'_C(\mathbb{R}^n) \cap \mathcal{A}_V$ , then the claim holds for  $s \in (-n/2, \infty) \setminus \mathbb{Z}$ .

Theorem 1.1 generalizes the result in [6] (see lemma 4.1) where one assumes that  $(-\Delta)^s f|_V = f|_V = 0$ . In fact, theorem 1.1 is proved by reducing the claim to the case treated in [6, Theorem 1.1] (see section 4). The meaning of the condition  $f \in \mathcal{A}_V$  is discussed in section 3 (see remark 3.3). When  $s \in (-n/2, -n/4] \setminus \mathbb{Z}$ , we need to have  $f \in \mathcal{O}'_C(\mathbb{R}^n)$  so that  $(-\Delta)^s f$ is well-defined and we can use lemma 4.1 in the proof of theorem 1.1. **Remark 1.2.** If  $f \in \mathcal{E}'(\mathbb{R}^n)$ , then instead of assuming  $(-\Delta)^s f|_V = 0$  in theorem 1.1 we could only require that  $(-\Delta)^s f$  vanishes to infinite order at some point  $x_0 \in V$ , i.e.  $\partial^{\beta}((-\Delta)^s f)(x_0) = 0$  for all  $\beta \in \mathbb{N}^n$ . This follows since a corresponding unique continuation result is known for  $f \in \mathcal{E}'(\mathbb{R}^n)$ under the assumptions  $f|_V = 0$  and  $\partial^{\beta}((-\Delta)^s f)(x_0) = 0$  for all  $\beta \in \mathbb{N}^n$  (see corollary 4 on page 12 in [6]), and constant coefficient partial differential operators P(D) commute with fractional Laplacians and ordinary derivatives. Therefore we can use the same proof to prove this slightly stronger result (see the proof of theorem 1.1).

From the unique continuation of fractional Laplacians we immediately obtain the following unique continuation result for the normal operator of the X-ray transform of scalar fields. The reason is that the normal operator can be written as  $N_0 = (-\Delta)^{-1/2}$  up to a constant factor (see section 2.2).

**Theorem 1.3.** Let  $n \geq 2$  and  $f \in \mathcal{E}'(\mathbb{R}^n) \cap \mathcal{A}_V$  or  $f \in C_{\infty}(\mathbb{R}^n) \cap \mathcal{A}_V$  where  $V \subset \mathbb{R}^n$  is some nonempty open set. If  $N_0 f|_V = 0$ , then f = 0.

Theorem 1.3 is a generalization of the result in [20] where one assumes  $N_0f|_V = f|_V = 0$ . When  $f \in \mathcal{E}'(\mathbb{R}^n) \cap \mathcal{A}_V$ , we could replace the assumption  $N_0f|_V = 0$  with the requirement that  $N_0f$  vanishes to infinite order at some point  $x_0 \in V$  (see remark 1.2). In order to use theorem 1.1 in the case s = -1/2 and  $n \geq 2$ , and to guarantee that  $N_0f$  is well-defined, we need to have  $f \in \mathcal{E}'(\mathbb{R}^n) \subset \mathcal{O}'_C(\mathbb{R}^n)$  or  $f \in C_{\infty}(\mathbb{R}^n) \subset \mathcal{O}'_C(\mathbb{R}^n)$  in theorem 1.3.

The unique continuation of  $N_0$  implies uniqueness for the following partial data problem.

**Theorem 1.4.** Let  $n \ge 2$  and  $f \in \mathcal{E}'(\mathbb{R}^n) \cap \mathcal{A}_V$  or  $f \in C_{\infty}(\mathbb{R}^n) \cap \mathcal{A}_V$  where  $V \subset \mathbb{R}^n$  is some nonempty open set. If  $X_0 f = 0$  on all lines intersecting V, then f = 0.

Theorem 1.4 generalizes theorem 1.2 in [20], where one assumes  $f|_V = 0$ , to the case  $P(D)f|_V = 0$  for some  $P \in \mathcal{P}$ . We note that if f is polynomial in V, then there is  $P \in \mathcal{P}$  such that  $P(D)f|_V = 0$ . Hence those scalar fields which are polynomial in V can be uniquely determined from their X-ray data on lines intersecting V. This special case of theorem 1.4 is previously known in two dimensions [23, 46]. We also note that theorem 1.4 includes much larger class of functions than just polynomials. The scalar field f can be (poly)harmonic in V and f can also have singularities in V if f is for example a non-smooth solution to the wave equation (see section 3 for more examples of admissible functions).

It is important to notice that from the vector space structure of admissible functions  $\mathcal{A}_V$  it follows that theorem 1.4 is indeed a uniqueness result: if  $f_1$ and  $f_2$  satisfy  $P_1(D)f_1|_V = P_2(D)f_2|_V = 0$  for some  $P_1, P_2 \in \mathcal{P}$  and  $X_0f_1 = X_0f_2$  on all lines intersecting V, then  $f_1 = f_2$  (see proposition 3.4 and remark 3.5 for more details). Especially, the equality of the X-ray data on all lines intersecting V implies that the scalar fields are equal everywhere even though  $f_1$  and  $f_2$  a priori can have very different behaviour in V since  $P_1(D)$ can be different from  $P_2(D)$ .

**Remark 1.5.** We note that our proof for theorem 1.4 gives only uniqueness but not stability for the partial data problem. In theorem 1.4 we eventually

have to assume that f is not supported in V since otherwise we would have P(D)f = 0 everywhere and therefore f = 0 without assuming anything about the X-ray data (see the proof of theorem 1.1). When f is supported outside V we do not have access to all singularities of f via the X-ray data, i.e. we have invisible singularities outside V. It is known that the recovery of such invisible singularities is not stable [25, 34, 35].

**Remark 1.6.** We note that similar results as in theorems 1.3 and 1.4 also hold for the d-plane transform when d is odd (see corollaries 1 and 2 on page 6 in [6]). The d-plane transform takes a scalar field and integrates it over d-dimensional affine planes where 0 < d < n. The case d = 1 corresponds to the X-ray transform. The normal operator  $N_d$  of the d-plane transform can be expressed as  $N_d = (-\Delta)^{-d/2}$  up to a constant factor (see [6, 16]). Hence  $N_d$  admits the same unique continuation property as in theorem 1.1 for functions in  $\mathcal{E}'(\mathbb{R}^n) \cap \mathcal{A}_V$  or  $C_{\infty}(\mathbb{R}^n) \cap \mathcal{A}_V$  provided d is odd. The unique continuation of  $N_d$  then implies a similar uniqueness result as in theorem 1.4 for a partial data problem of the d-plane transform when d is odd.

From the unique continuation of fractional Laplacians we also obtain a partial data result for the X-ray transform of vector fields. The normal operators satisfy the relationship  $d(N_1F) = N_0(dF)$  up to a constant factor (see lemma 4.4). Hence the unique continuation and partial data problems of vector fields can be reduced to the corresponding problems for scalar fields, namely the components  $(dF)_{ij}$ .

The next theorem generalizes the result in [21] where the authors assume that  $dF|_V = 0$  instead of  $(dF)_{ij} \in \mathcal{A}_V$ .

**Theorem 1.7.** Let  $n \ge 2$  and  $F \in (\mathcal{E}'(\mathbb{R}^n))^n$  such that  $(dF)_{ij} \in \mathcal{A}_V$  for all i, j = 1, ..., n where  $V \subset \mathbb{R}^n$  is some nonempty open set. If  $X_1F = 0$  on all lines intersecting V, then dF = 0. Especially,  $F = d\phi$  for some  $\phi \in \mathcal{E}'(\mathbb{R}^n)$ .

The conclusion  $F = d\phi$  in theorem 1.7 is equivalent to that the solenoidal part  $F^{s}$  vanishes in the solenoidal decomposition  $F = F^{s} + d\phi$  (see e.g. [41]). Therefore theorem 1.7 can be seen as a solenoidal injectivity result in terms of partial data (see [21] and [33, 41]). Theorem 1.7 holds also for vector fields  $F \in (\mathscr{S}(\mathbb{R}^{n}))^{n}$  which components are Schwartz functions since in that case  $(dF)_{ii} \in C_{\infty}(\mathbb{R}^{n}) \cap \mathcal{A}_{V}$ .

We note that if the components  $F_i$  are all polynomial in V, then also  $(dF)_{ij}$  are all polynomial in V. Hence there are some  $P_{ij} \in \mathcal{P}$  such that  $P_{ij}(D)(dF)_{ij}|_V = 0$  and therefore  $(dF)_{ij} \in \mathcal{A}_V$ . This means that solenoidal vector fields which are polynomial in V can be uniquely determined from their X-ray data on lines intersecting V. However, this is only a small subset of admissible vector fields: F can be for example componentwise (poly)harmonic in V and more generally F can also have singularities in V.

**Remark 1.8.** Theorems 1.1 and 1.3 imply a unique continuation result for  $N_1$ : if  $F \in (\mathcal{E}'(\mathbb{R}^n))^n$  satisfies  $(dF)_{ij} \in \mathcal{A}_V$  for all i, j = 1, ..., n and  $N_1F|_V = 0$ , then dF = 0. This follows since  $d(N_1F) = N_0(dF)$  up to a constant factor (see lemma 4.4) and one can use theorem 1.3 for the components  $(dF)_{ij} \in \mathcal{E}'(\mathbb{R}^n)$ . One also obtains a stronger version where one can replace the assumption  $N_1F|_V = 0$  with the requirement that  $d(N_1F)$  vanishes componentwise to infinite order at some point  $x_0 \in V$  (see remark 1.2). 1.3. Related results. There are some earlier unique continuation and partial data results for scalar and vector fields. The partial data problem for scalar fields has a unique solution if  $f|_V$  vanishes [4, 20, 24],  $f|_V$  is polynomial or piecewise polynomial [23, 24, 46] or  $f|_V$  is real analytic [23]. A complementary result is the Helgason support theorem: if the integrals of fvanish on all lines not intersecting a given compact and convex set, then fhas to vanish outside that set [16, 42]. The normal operator of the X-ray transform of scalar fields has a unique continuation property under the assumptions  $N_0 f|_V = f|_V = 0$  [20]. This is a special case of a more general unique continuation property of fractional Laplacians [6, 14]. There are also partial data and unique continuation results for the *d*-plane transform of scalar fields when *d* is odd, including the X-ray transform as a special case d = 1 (see [6] and remark 1.6).

The partial data problem of vector fields is known to be uniquely solvable up to potential fields, if  $dF|_V = 0$  [21]. Similarly, the normal operator of the X-ray transform of vector fields has a unique continuation property under the assumptions  $N_1F|_V = dF|_V = 0$  [21]. There are other partial data results for vector fields where one knows the integrals of F over lines which are parallel to a finite set of planes [22, 38, 40] or which intersect a certain type of curve [9, 36, 44]. There is also a Helgason-type support theorem for vector fields: if the integrals of F vanish on all lines not intersecting a given compact and convex set, then dF vanishes outside that set [21, 42].

The normal operator of scalar fields, the normal operator of vector fields and the fractional Laplacian all admit stronger versions of the unique continuation property (see [6, 11, 12, 13, 20, 21, 37, 47] and remarks 1.2 and 1.8). Other applications of unique continuation of fractional Laplacians include fractional inverse problems. Especially, the unique continuation of  $(-\Delta)^s$ is used to prove uniqueness for different versions of the fractional Calderón problem (see e.g. [1, 2, 5, 6, 7, 14]).

### 2. The X-ray transform and distributions

In this section we define the X-ray transform of scalar and vector fields, and introduce the distribution spaces we use in our main theorems. The basic theory of distributions and Sobolev spaces can be found in [15, 17, 27, 28, 43] and the X-ray transform is treated for example in [29, 41, 42].

2.1. Distributions and Sobolev spaces. We let  $\mathcal{E}(\mathbb{R}^n)$  be the space of smooth functions,  $\mathscr{S}(\mathbb{R}^n)$  is the Schwartz space and  $\mathcal{D}(\mathbb{R}^n)$  is the space of compactly supported smooth functions. We equip all these spaces with their standard topologies. The corresponding duals are denoted by  $\mathcal{E}'(\mathbb{R}^n)$ ,  $\mathscr{S}'(\mathbb{R}^n)$  and  $\mathcal{D}'(\mathbb{R}^n)$ . Elements in  $\mathcal{E}'(\mathbb{R}^n)$  are identified with distributions of compact support and elements in  $\mathscr{S}'(\mathbb{R}^n)$  are called tempered distributions.

We define the space of rapidly decreasing distributions  $\mathscr{O}'_{C}(\mathbb{R}^{n}) \subset \mathscr{S}'(\mathbb{R}^{n})$ as follows:  $f \in \mathscr{O}'_{C}(\mathbb{R}^{n})$  if and only if  $\hat{f} \in \mathscr{O}_{M}(\mathbb{R}^{n})$  where  $\hat{f} = \mathcal{F}(f)$  is the Fourier transform of tempered distributions. Here  $\mathscr{O}_{M}(\mathbb{R}^{n})$  is the space of polynomially growing smooth functions, i.e.  $f \in \mathscr{O}_{M}(\mathbb{R}^{n})$  if f and all its derivatives are polynomially bounded. We note that the Fourier transform is an isomorphism  $\mathcal{F} \colon \mathscr{S}'(\mathbb{R}^{n}) \to \mathscr{S}'(\mathbb{R}^{n})$  and also extends to an isomorphism  $\mathcal{F} \colon L^{2}(\mathbb{R}^{n}) \to L^{2}(\mathbb{R}^{n})$ . We have the inclusions  $\mathcal{E}'(\mathbb{R}^{n}) \subset \mathscr{O}'_{C}(\mathbb{R}^{n}) \subset$   $\mathscr{S}'(\mathbb{R}^n) \subset \mathcal{D}'(\mathbb{R}^n)$ . As a special case we have  $\mathscr{S}(\mathbb{R}^n) \subset C_{\infty}(\mathbb{R}^n) \subset \mathscr{O}'_C(\mathbb{R}^n)$  where  $C_{\infty}(\mathbb{R}^n)$  is the set of all continuous functions which decay faster than any polynomial at infinity.

The fractional  $L^2$ -Sobolev space of order  $r \in \mathbb{R}$  is defined as

(2) 
$$H^{r}(\mathbb{R}^{n}) = \{ f \in \mathscr{S}'(\mathbb{R}^{n}) : \langle \cdot \rangle^{r} \hat{f} \in L^{2}(\mathbb{R}^{n}) \}$$

where  $\langle \xi \rangle = (1 + |\xi|^2)^{1/2}$ . The space  $H^r(\mathbb{R}^n)$  is equipped with the norm

(3) 
$$\|f\|_{H^r(\mathbb{R}^n)} = \left\|\langle \cdot \rangle^r \hat{f}\right\|_{L^2(\mathbb{R}^n)}$$

and  $H^r(\mathbb{R}^n)$  becomes a separable Hilbert space for every  $r \in \mathbb{R}$ . It follows that the spaces are nested, i.e.  $H^r(\mathbb{R}^n) \hookrightarrow H^t(\mathbb{R}^n)$  continuously when  $r \ge t$ . One can isomorphically identify  $H^{-r}(\mathbb{R}^n)$  with the dual  $(H^r(\mathbb{R}^n))^*$  for all  $r \in \mathbb{R}$ . We define the following spaces

(4) 
$$H^{\infty}(\mathbb{R}^n) = \bigcap_{r \in \mathbb{R}} H^r(\mathbb{R}^n), \quad H^{-\infty}(\mathbb{R}^n) = \bigcup_{r \in \mathbb{R}} H^r(\mathbb{R}^n)$$

It holds that  $\mathscr{O}'_{C}(\mathbb{R}^{n}) \subset H^{-\infty}(\mathbb{R}^{n}) \subset \mathscr{S}'(\mathbb{R}^{n})$  and  $\mathscr{S}(\mathbb{R}^{n}) \subset H^{\infty}(\mathbb{R}^{n})$ . Further, using the Sobolev embedding one can see that  $H^{\infty}(\mathbb{R}^{n}) = C^{\infty}_{L^{2}}(\mathbb{R}^{n})$  where  $f \in C^{\infty}_{L^{2}}(\mathbb{R}^{n})$  if f is smooth and f and all its derivatives belong to  $L^{2}(\mathbb{R}^{n})$  (see [15, Theorem 6.12]).

The fractional Laplacian is defined as

(5) 
$$(-\Delta)^s f = \mathcal{F}^{-1}(|\cdot|^{2s} \hat{f})$$

where  $\mathcal{F}^{-1}$  is the inverse Fourier transform of tempered distributions. It follows that  $(-\Delta)^s f$  is well-defined as a tempered distribution for  $f \in \mathscr{O}'_C(\mathbb{R}^n)$ when  $s \in (-n/2, \infty) \setminus \mathbb{Z}$ , and for  $f \in H^r(\mathbb{R}^n)$  when  $s \in (-n/4, \infty) \setminus \mathbb{Z}$ (see [6, Section 2.2]). We have that  $(-\Delta)^s \colon H^r(\mathbb{R}^n) \to H^{r-2s}(\mathbb{R}^n)$  is continuous whenever  $s \in (0, \infty) \setminus \mathbb{Z}$  and  $(-\Delta)^s$  also admits a Poincaré-type inequality for  $s \in (0, \infty) \setminus \mathbb{Z}$  (see [6]). We note that  $(-\Delta)^s$  is a non-local operator in contrast to the ordinary Laplacian  $(-\Delta)$ . The non-locality implies a unique continuation property (see theorem 1.1 and lemma 4.1) which cannot hold for local operators.

We also use local versions of distributions and fractional Sobolev spaces. Let  $\Omega \subset \mathbb{R}^n$  be an open set and  $r \in \mathbb{R}$ . We denote by  $\mathcal{D}(\Omega)$ ,  $\mathcal{D}'(\Omega)$  etc. the test function and distribution spaces defined in  $\Omega$ . We define the local Sobolev space  $H^r(\Omega)$  as

(6) 
$$H^{r}(\Omega) = \{g \in \mathcal{D}'(\Omega) : g = f|_{\Omega} \text{ for some } f \in H^{r}(\mathbb{R}^{n})\}.$$

In other words, the space  $H^r(\Omega)$  consists of restrictions of distributions  $f \in H^r(\mathbb{R}^n)$ . The local Sobolev space is equipped with the quotient norm

(7) 
$$||g||_{H^r(\Omega)} = \inf\{||f||_{H^r(\mathbb{R}^n)} : f \in H^r(\mathbb{R}^n) \text{ such that } f|_{\Omega} = g\}.$$

Then  $H^r(\Omega)$  becomes a separable Hilbert space and the restriction map  $|_{\Omega}: H^r(\mathbb{R}^n) \to H^r(\Omega)$  is continuous. If  $r \geq t$ , then  $H^r(\Omega) \hookrightarrow H^t(\Omega)$  continuously. One can also isomorphically identify  $H^{-r}(\Omega)$  as the dual  $(\tilde{H}^r(\Omega))^*$ for every  $r \in \mathbb{R}$  where  $\tilde{H}^r(\Omega)$  is the closure of  $\mathcal{D}(\Omega)$  in  $H^r(\mathbb{R}^n)$  (see [3] and [27]). If  $r \geq 0$ , then  $H^r(\Omega) \subset W^r(\Omega)$  where  $W^r(\Omega)$  is the Sobolev-Slobodeckij space which is defined by using weak derivatives of  $L^2$ -functions (see [27] for a precise definition). If  $\Omega$  is a Lipschitz domain, then we have the equality  $H^r(\Omega) = W^r(\Omega)$  for all  $r \ge 0$ .

More generally, we define the vector-valued test function space  $(\mathcal{D}(\mathbb{R}^n))^n$ by saying that  $\varphi \in (\mathcal{D}(\mathbb{R}^n))^n$  if and only if  $\varphi = (\varphi_1, \ldots, \varphi_n)$  and  $\varphi_i \in \mathcal{D}(\mathbb{R}^n)$  for all  $i = 1, \ldots, n$ . A sequence converges to zero in  $(\mathcal{D}(\mathbb{R}^n))^n$  if and only if all its components converge to zero in  $\mathcal{D}(\mathbb{R}^n)$ . We then define the space of vector-valued distributions  $(\mathcal{D}'(\mathbb{R}^n))^n$  by saying that  $F \in (\mathcal{D}'(\mathbb{R}^n))^n$ if and only if  $F = (F_1, \ldots, F_n)$  where  $F_i \in \mathcal{D}'(\mathbb{R}^n)$  for all  $i = 1, \ldots, n$ . The duality pairing is defined as  $\langle F, \varphi \rangle = \sum_{i=1}^n \langle F_i, \varphi_i \rangle$ . The test function spaces  $(\mathcal{E}(\mathbb{R}^n))^n$  and  $(\mathscr{S}(\mathbb{R}^n))^n$ , and the corresponding distribution spaces  $(\mathcal{E}'(\mathbb{R}^n))^n$  and  $(\mathscr{S}'(\mathbb{R}^n))^n$  are defined analogously. The elements in  $(\mathcal{E}'(\mathbb{R}^n))^n$ are called compactly supported vector-valued distributions. Vector-valued distributions are a special case of currents (continuous linear functionals in the space of differential forms, see [8, Section III]).

For  $F \in (\mathcal{D}'(\mathbb{R}^n))^n$  we define the exterior derivative or curl of F as a matrix which components are  $(dF)_{ij} = \partial_i F_j - \partial_j F_i$ . It follows from the Poincaré lemma (see e.g. [26, Theorem 2.1] and lemma 4.2) that if dF = 0, then  $F = d\phi$  for some  $\phi \in \mathcal{D}'(\mathbb{R}^n)$  where  $d\phi$  is the distributional gradient of  $\phi$ .

2.2. The X-ray transform of scalar fields. Let  $f \in \mathcal{D}(\mathbb{R}^n)$  be a scalar field. The X-ray transform  $X_0$  is defined as

(8) 
$$X_0 f(\gamma) = \int_{\gamma} f \mathrm{d}s$$

where  $\gamma$  is an oriented line in  $\mathbb{R}^n$ . When we parameterize the set of all oriented lines with the set

(9) 
$$\Gamma = \{(z,\theta) : \theta \in S^{n-1}, \ z \in \theta^{\perp}\}$$

the X-ray transform becomes

(10) 
$$X_0 f(z,\theta) = \int_{\mathbb{R}} f(z+s\theta) \mathrm{d}s$$

The adjoint or back-projection  $X_0^*$  is defined as

(11) 
$$X_0^*\psi(x) = \int_{S^{n-1}} \psi(x - (x \cdot \theta)\theta, \theta) \mathrm{d}\theta$$

where  $\psi \in \mathcal{E}(\Gamma)$ . One then sees that  $X_0: \mathcal{D}(\mathbb{R}^n) \to \mathcal{D}(\Gamma)$  and  $X_0^*: \mathcal{E}(\Gamma) \to \mathcal{E}(\mathbb{R}^n)$  are continuous maps. Using duality we can define  $X_0: \mathcal{E}'(\mathbb{R}^n) \to \mathcal{E}'(\Gamma)$ and  $X_0^*: \mathcal{D}'(\Gamma) \to \mathcal{D}'(\mathbb{R}^n)$  as

(12) 
$$\langle X_0 f, \varphi \rangle = \langle f, X_0^* \varphi \rangle$$

(13) 
$$\langle X_0^*\psi,\eta\rangle = \langle \psi, X_0\eta\rangle$$

where  $\langle \cdot, \cdot \rangle$  is the dual pairing.

The normal operator is  $N_0 = X_0^* X_0$  and it can be expressed as the convolution

(14) 
$$N_0 f(x) = 2(f * |\cdot|^{1-n})(x).$$

Using duality the normal operator extends to a map  $N_0: \mathcal{E}'(\mathbb{R}^n) \to \mathcal{D}'(\mathbb{R}^n)$ and the convolution formula holds in the sense of distributions. The normal operator can be seen as the fractional Laplacian  $(-\Delta)^{-1/2}$  up to a constant factor and we have the reconstruction formula

(15) 
$$f = c_{0,n} (-\Delta)^{1/2} N_0 f$$

where  $c_{0,n}$  is a constant which depends on dimension. Both  $X_0$  and  $N_0$  are also defined for functions  $f \in C_{\infty}(\mathbb{R}^n)$ .

2.3. The X-ray transform of vector fields. Let  $F \in (\mathcal{D}(\mathbb{R}^n))^n$  be a vector field. The X-ray transform  $X_1$  is defined as

(16) 
$$X_1 F(\gamma) = \int_{\gamma} F \cdot \mathrm{d}\bar{s}$$

where  $\gamma$  is an oriented line. Using the parametrization  $\Gamma$  for oriented lines (see equation (9)) we have

(17) 
$$X_1 F(z, \theta) = \int_{\mathbb{R}} F(z + s\theta) \cdot \theta \mathrm{d}s.$$

We define the adjoint  $X_1^*$  as the vector-valued operator

(18) 
$$(X_1^*\psi)_i(x) = \int_{S^{n-1}} \theta_i \psi(x - (x \cdot \theta)\theta, \theta) \mathrm{d}\theta$$

where  $\psi \in \mathcal{E}(\Gamma)$  is a scalar field in the space of oriented lines. One sees that  $X_1: (\mathcal{D}(\mathbb{R}^n))^n \to \mathcal{D}(\Gamma)$  and  $X_1^*: \mathcal{E}(\Gamma) \to (\mathcal{E}(\mathbb{R}^n))^n$  are continuous and by duality we can define  $X_1: (\mathcal{E}'(\mathbb{R}^n))^n \to \mathcal{E}'(\Gamma)$  and  $X_1^*: \mathcal{D}'(\Gamma) \to (\mathcal{D}'(\mathbb{R}^n))^n$  by setting

(19) 
$$\langle X_1 F, \varphi \rangle = \langle F, X_1^* \varphi \rangle$$

(20) 
$$\langle X_1^*\psi,\eta\rangle = \langle\psi,X_1\eta\rangle$$

We define the normal operator as  $N_1 = X_1^* X_1$  and it can be expressed in terms of convolution

(21) 
$$(N_1F)_i = \sum_{j=1}^n \frac{2x_i x_j}{|x|^{n+1}} * F_j.$$

The normal operator extends to a map  $N_1: (\mathcal{E}'(\mathbb{R}^n))^n \to (\mathcal{D}'(\mathbb{R}^n))^n$  by duality and the convolution formula holds in the sense of distributions. One has the reconstruction formula for the solenoidal part  $F^s$  in the solenoidal decomposition  $F = F^s + d\phi$  (see for example [41, 42])

(22) 
$$F^{s} = c_{1,n}(-\Delta)^{1/2} N_1 F$$

where  $c_{1,n}$  is a constant depending on dimension and  $(-\Delta)^{1/2}$  operates componentwise on  $N_1F$ . Both  $X_1$  and  $N_1$  are also defined for vector fields  $F \in (\mathscr{S}(\mathbb{R}^n))^n$ .

#### 3. PARTIAL DIFFERENTIAL OPERATORS AND ADMISSIBLE FUNCTIONS

In this section we introduce constant coefficient partial differential operators and also study the space of admissible functions  $\mathcal{A}_V$  in more detail. A comprehensive treatment of constant coefficient partial differential operators can be found in Hörmander's book [18].

Let us denote by  $\mathcal{P}$  the set of all polynomials in  $\mathbb{R}^n$  with complex coefficients excluding the zero polynomial  $P \equiv 0$ . A polynomial  $P \in \mathcal{P}$  of degree

 $m\in\mathbb{N}$  can be identified with the constant coefficient partial differential operator P(D) of order  $m\in\mathbb{N}$  as

(23) 
$$P(D) = \sum_{|\alpha| \le m} a_{\alpha} D^{\alpha}, \quad a_{\alpha} \in \mathbb{C},$$

where  $D^{\alpha} = D_1^{\alpha_1} \cdots D_n^{\alpha_n}$ ,  $D_j = -i\partial_j$  and  $\alpha = (\alpha_1, \ldots, \alpha_n) \in \mathbb{N}^n$  is a multiindex such that  $|\alpha| = \alpha_1 + \ldots + \alpha_n$ . In fact, using the Fourier transform one sees that

(24) 
$$\widehat{P(D)} = P(\xi) = \sum_{|\alpha| \le m} a_{\alpha} \xi^{\alpha}$$

where  $\xi \in \mathbb{R}^n$  and  $\xi^{\alpha} = \xi^{\alpha_1} \cdots \xi^{\alpha_n}$ . The polynomial  $P(\xi)$  is also known as the full symbol of P(D). If  $g \in \mathcal{D}'(\Omega)$  where  $\Omega \subset \mathbb{R}^n$  is an open set, then one can define the distributional derivative  $P(D)g \in \mathcal{D}'(\Omega)$  by duality. Further, it holds that  $P(D): H^r(\Omega) \to H^{r-m}(\Omega)$  is continuous with respect to the quotient norm [28, Theorem 12.15] (see equation (7)).

The set of admissible functions  $\mathcal{A}_V$  which we use in our main theorems can be written as the union

(25) 
$$\mathcal{A}_{V} = \bigcup_{\substack{P \in \mathcal{P} \\ r \in \mathbb{R}}} \mathcal{H}_{P,V}^{r}(\mathbb{R}^{n}) = \bigcup_{\substack{P \in \mathcal{P} \\ r \in \mathbb{R}}} \{f \in H^{r}(\mathbb{R}^{n}) : P(D)f|_{V} = 0\}$$

where  $V \subset \mathbb{R}^n$  is some nonempty open set and  $\mathcal{H}^r_{P,V}(\mathbb{R}^n) = \{f \in H^r(\mathbb{R}^n) : P(D)f|_V = 0\}$ . We note that  $\mathcal{A}_V \subset H^{-\infty}(\mathbb{R}^n)$ . The following proposition implies that the sets  $\mathcal{H}^r_{P,V}(\mathbb{R}^n)$  in the union (25) are also Hilbert spaces.

**Proposition 3.1.** The subset  $\mathcal{H}_{P,V}^r(\mathbb{R}^n) \subset H^r(\mathbb{R}^n)$  is a separable Hilbert space for all  $r \in \mathbb{R}$ ,  $P \in \mathcal{P}$  and nonempty open set  $V \subset \mathbb{R}^n$ .

Proof. Clearly  $\mathcal{H}_{P,V}^r(\mathbb{R}^n)$  is a linear subspace of  $H^r(\mathbb{R}^n)$ . Let  $f_k \in \mathcal{H}_{P,V}^r(\mathbb{R}^n)$ be a sequence such that  $f_k \to f$  in  $H^r(\mathbb{R}^n)$ . Then by the continuity of the restriction map  $|_V : H^r(\mathbb{R}^n) \to H^r(V)$  we have that  $f_k|_V \to f|_V$  in  $H^r(V)$ . From the continuity of  $P(D) : H^r(V) \to H^{r-m}(V)$  we obtain that  $0 = P(D)f_k|_V \to P(D)f|_V$  in  $H^{r-m}(V)$ , implying that  $f \in \mathcal{H}_{P,V}^r(\mathbb{R}^n)$ . Therefore  $\mathcal{H}_{P,V}^r(\mathbb{R}^n)$  is a closed subspace of the separable Hilbert space  $H^r(\mathbb{R}^n)$ and hence itself a separable Hilbert space.

**Remark 3.2.** We note that in the smooth case we have that  $\mathcal{E}_{P,V}(\mathbb{R}^n) = \{f \in \mathcal{E}(\mathbb{R}^n) : P(D)f|_V = 0\} \subset \mathcal{E}(\mathbb{R}^n)$  is a closed subspace of  $\mathcal{E}(\mathbb{R}^n)$ and hence a Fréchet space. More generally,  $\mathcal{D}'_{P,V}(\mathbb{R}^n) = \{f \in \mathcal{D}'(\mathbb{R}^n) : P(D)f|_V = 0\} \subset \mathcal{D}'(\mathbb{R}^n)$  is sequentially closed in  $\mathcal{D}'(\mathbb{R}^n)$  under the weak<sup>\*</sup> convergence. These two facts follow from the continuity of  $P(D) : \mathcal{E}(\mathbb{R}^n) \to \mathcal{E}(\mathbb{R}^n)$  and  $P(D) : \mathcal{D}'(\mathbb{R}^n) \to \mathcal{D}'(\mathbb{R}^n)$  with respect to the standard topologies. More topological properties of kernels of constant coefficient partial differential operators can be found in [45].

**Remark 3.3.** The interpretation of the condition  $f \in \mathcal{A}_V$  is the following. If  $f \in \mathcal{A}_V$ , then there is some  $r \in \mathbb{R}$  and some  $P \in \mathcal{P}$  such that  $f \in H^r(\mathbb{R}^n)$ and  $P(D)f|_V = 0$ . The distributional derivatives commute with restrictions, i.e.  $P(D)f|_V = P(D)(f|_V)$  where  $f|_V \in \mathcal{D}'(V)$ . Since  $f \in H^r(\mathbb{R}^n)$  we see that  $f|_V$  is not only a distribution but in addition  $f|_V \in H^r(V)$  for some  $r \in \mathbb{R}$ . Therefore the existence of  $r \in \mathbb{R}$  and  $P \in \mathcal{P}$  for which  $P(D)f|_V = 0$ means that  $f|_V \in H^r(V)$  and  $f|_V$  is a weak solution to some homogeneous constant coefficient partial differential equation. In other words,  $f|_V$  satisfies

(26) 
$$\sum_{|\alpha| \le m} a_{\alpha} D^{\alpha}(f|_{V}) = 0, \quad f|_{V} \in \bigcup_{r \in \mathbb{R}} H^{r}(V),$$

for some coefficients  $a_{\alpha} \in \mathbb{C}$  and some integer  $m \in \mathbb{N}$ .

The following proposition is important in the uniqueness of the partial data problem.

**Proposition 3.4.** The set  $\mathcal{A}_V \subset H^{-\infty}(\mathbb{R}^n)$  is a vector space for every nonempty open set  $V \subset \mathbb{R}^n$ .

Proof. Let  $f_1, f_2 \in \mathcal{A}_V$  and  $\lambda \in \mathbb{C}$ . This means that  $f_1 \in H^{r_1}(\mathbb{R}^n)$ ,  $f_2 \in H^{r_2}(\mathbb{R}^n)$  and  $P_1(D)f_1|_V = P_2(D)f_2|_V = 0$  for some  $r_1, r_2 \in \mathbb{R}$  and  $P_1, P_2 \in \mathcal{P}$ . It follows that  $f_1 + \lambda f_2 \in H^r(\mathbb{R}^n)$  where  $r = \min\{r_1, r_2\}$  since the spaces  $H^t(\mathbb{R}^n), t \in \mathbb{R}$ , are nested vector spaces. We also have that  $P_1(D)P_2(D)(f_1 + \lambda f_2)|_V = 0$  since the distributional derivatives commute  $P_1(D)P_2(D) = P_2(D)P_1(D)$ . This implies that  $f_1 + \lambda f_2 \in \mathcal{A}_V$ , i.e.  $\mathcal{A}_V$  is a linear subspace of the vector space  $H^{-\infty}(\mathbb{R}^n) \subset \mathscr{S}'(\mathbb{R}^n)$ .

**Remark 3.5.** The vector space structure of  $\mathcal{A}_V$  is important since it implies that the partial data results we have proved in this article are indeed uniqueness results. Namely, if  $f_1, f_2 \in \mathcal{E}'(\mathbb{R}^n) \cap \mathcal{A}_V$  (or  $f_1, f_2 \in C_{\infty}(\mathbb{R}^n) \cap \mathcal{A}_V$ ) such that  $X_0f_1 = X_0f_2$  on all lines intersecting V, then  $f_1 - f_2 \in \mathcal{E}'(\mathbb{R}^n) \cap \mathcal{A}_V$ (or  $f_1 - f_2 \in C_{\infty}(\mathbb{R}^n) \cap \mathcal{A}_V$ ) and  $X_0(f_1 - f_2) = 0$  on all lines intersecting V. Theorem 1.4 then implies that  $f_1 - f_2 = 0$ , i.e. the solution to the partial data problem is unique.

We list some examples of admissible functions. We have that the function  $f \in H^{-\infty}(\mathbb{R}^n)$  belongs to  $\mathcal{A}_V$ , if

- f is polyharmonic in V, i.e.  $(-\Delta)^k f|_V = 0$  for some  $k \in \mathbb{N}$ .
- f is polynomial in V.
- f is independent of one of the variables  $x_1, \ldots, x_n$  in V.
- $f(x) = q(x)e^{ix\cdot\zeta}$  in V where q(x) is a suitable polynomial and  $\zeta \in \mathbb{C}^n$ is a generalized frequency. Especially, if f is of the form  $f(x) = e^{ix\cdot\xi_0}$ in V where  $\xi_0 \in \mathbb{C}^n$  is a zero of  $P \in \mathcal{P}$ , then  $P(D)f|_V = 0$ .

Further, it holds that for convex sets V and a fixed  $P \in \mathcal{P}$  the linear span of solutions of the form  $q(x)e^{ix\cdot\zeta}$  is dense in the space of all smooth solutions of P(D)g = 0 in V (see [17, Theorem 7.3.6] and a more general result [18, Theorem 10.5.1]).

We note that if P(D) is a hypoelliptic operator, then the condition  $P(D)f|_V = 0$  already implies that f is smooth in V (see [18, 28]). Basic examples of hypoelliptic operators are elliptic operators such as integer powers of Laplacians  $((-\Delta)^k$  where  $k \in \mathbb{N})$  and also the non-elliptic heat operator  $\partial_t - \Delta$ . However, there are non-smooth distributions  $f|_V$  which satisfy the condition  $P(D)f|_V = 0$  for some  $P \in \mathcal{P}$  and therefore f can have singularities in V. For example, the wave operator  $\partial_t^2 - \Delta$  is not hypoelliptic and has non-smooth weak solutions.

#### 4. PROOFS OF THE MAIN THEOREMS

In this section we prove our main theorems. We need a few auxiliary results. The first one is a unique continuation result for fractional Laplacians and the second one is the Poincaré lemma for compactly supported vectorvalued distributions.

**Lemma 4.1** ([6, Theorem 1.1]). Let  $n \ge 1$ ,  $s \in (-n/4, \infty) \setminus \mathbb{Z}$  and  $u \in H^t(\mathbb{R}^n)$  where  $t \in \mathbb{R}$ . If  $(-\Delta)^s u|_V = 0$  and  $u|_V = 0$  for some nonempty open set  $V \subset \mathbb{R}^n$ , then u = 0. The claim holds also for  $s \in (-n/2, -n/4] \setminus \mathbb{Z}$  if  $u \in \mathscr{O}'_C(\mathbb{R}^n)$ .

**Lemma 4.2** (Poincaré lemma). Let  $U \in (\mathcal{E}'(\mathbb{R}^n))^n$  such that dU = 0. Then there is  $\phi \in \mathcal{E}'(\mathbb{R}^n)$  such that  $U = d\phi$ .

The proof of lemma 4.2 can be found for example in [19, 26]. The third lemma is a known result about the zero set of multivariate polynomials.

**Lemma 4.3** ([32, Lemma on p.1]). Let Q = Q(x) be a non-zero multivariate polynomial of order  $m \in \mathbb{N}$ 

(27) 
$$Q(x) = \sum_{|\alpha| \le m} b_{\alpha} x^{\alpha} = \sum_{|\alpha| \le m} b_{\alpha} x_1^{\alpha_1} \cdots x_n^{\alpha_n}, \quad b_{\alpha} \in \mathbb{C},$$

where  $\alpha = (\alpha_1, \ldots, \alpha_n) \in \mathbb{N}^n$  is a multi-index such that  $|\alpha| = \alpha_1 + \ldots + \alpha_n$ . Then the set  $Z_Q = \{x \in \mathbb{R}^n : Q(x) = 0\}$  has Lebesgue measure zero.

Lemma 4.3 is proved in [32] for real coefficients but the result holds also for complex coefficients by splitting  $b_{\alpha} \in \mathbb{C}$  to its real and imaginary parts. We note that the set  $Z_Q$  is Zariski closed but not the whole space  $\mathbb{R}^n$ . From the coarseness of the Zariski topology (i.e. there are relatively few closed sets) one can already deduce that the set  $Z_Q$  must be small in topological sense (see e.g. [10, Chapter 15.2]).

The next lemma shows how the normal operator of the X-ray transform of vector fields is related to the normal operator of scalar fields (see also [21, Proof of theorem 1.1]).

**Lemma 4.4.** Let  $F \in (\mathcal{E}'(\mathbb{R}^n))^n$ . Then  $d(N_1F) = (n-1)^{-1}N_0(dF)$  holds componentwise where  $N_0$  acts on the components  $(dF)_{ij} \in \mathcal{E}'(\mathbb{R}^n)$ .

*Proof.* The normal operator has the expression

(28) 
$$(N_1F)_i = \sum_{j=1}^n \frac{2x_i x_j}{|x|^{n+1}} * F_j.$$

Rewrite the kernel as

(29) 
$$\frac{2x_i x_j}{|x|^{n+1}} = \frac{2}{n-1} \left( \delta_{ij} |x|^{1-n} - \partial_i (x_j |x|^{1-n}) \right)$$

which implies that

(30) 
$$(N_1F)_i = \frac{2}{n-1} \left( \frac{1}{2} N_0 F_i - \sum_{j=1}^n x_j |x|^{1-n} * \partial_i F_j \right).$$

Calculating the components of  $d(N_1F)$  we obtain

(31) 
$$\partial_k (N_1 F)_i - \partial_i (N_1 F)_k = \frac{1}{n-1} N_0 (\partial_k F_i - \partial_i F_k).$$

This means that  $d(N_1F) = (n-1)^{-1}N_0(dF)$  where  $N_0$  acts componentwise on dF, giving the claim

Now we are ready to prove our results. We start with the main theorem.

Proof of theorem 1.1. Let  $f \in \mathcal{A}_V$  and  $s \in (-n/4, \infty) \setminus \mathbb{Z}$ . This means that  $f \in H^r(\mathbb{R}^n)$  for some  $r \in \mathbb{R}$  and  $P(D)f|_V = 0$  for some constant coefficient partial differential operator P(D) of order  $m \in \mathbb{N}$  and nonempty open set  $V \subset \mathbb{R}^n$ . In particular, we have  $f \in \mathscr{S}'(\mathbb{R}^n)$  such that  $\hat{f} = \langle \cdot \rangle^r g$ where  $g \in L^2(\mathbb{R}^n)$ . Using the properties of the Fourier transform we see that  $P(D)((-\Delta)^s f) = (-\Delta)^s (P(D)f)$  because P(D) has constant coefficients. Since P(D) is a local operator we obtain the conditions  $P(D)f|_V =$  $(-\Delta)^s (P(D)f)|_V = 0$ . Now  $P(D): H^r(\mathbb{R}^n) \to H^{r-m}(\mathbb{R}^n)$  is continuous (see e.g. [28, Theorem 12.7]) and we have  $P(D)f \in H^{r-m}(\mathbb{R}^n)$ . We can use lemma 4.1 for P(D)f to obtain that P(D)f = 0 as a tempered distribution. Taking the Fourier transform this is equivalent to that  $P(\xi)\hat{f}(\xi) =$  $P(\xi)\langle\xi\rangle^{-r}g(\xi) = 0$  almost everywhere where  $P(\xi)$  is a multivariate polynomial of order  $m \in \mathbb{N}$ . Since  $\langle \cdot \rangle^{-r} \neq 0$  everywhere and  $P(\xi) \neq 0$  almost everywhere by lemma 4.3, we have that g = 0 almost everywhere. This implies that  $\hat{f} = 0$  as a tempered distribution and hence f = 0.

Let then  $f \in \mathscr{O}'_{C}(\mathbb{R}^{n}) \cap \mathcal{A}_{V}$  and  $s \in (-n/2, \infty) \setminus \mathbb{Z}$ . Using the same arguments as above we obtain that  $P(D)f|_{V} = (-\Delta)^{s}(P(D)f)|_{V} = 0$  for some constant coefficient partial differential operator P(D) and nonempty open set  $V \subset \mathbb{R}^{n}$ . We know that  $f \in \mathscr{O}'_{C}(\mathbb{R}^{n})$  is equivalent to that  $\hat{f} \in \mathscr{O}_{M}(\mathbb{R}^{n})$ . Now  $\mathcal{F}(P(D)f)(\xi) = P(\xi)\hat{f}(\xi)$  where  $P(\xi)$  is a multivariate polynomial of order  $m \in \mathbb{N}$ . It follows from the Leibnitz product rule for multivariable functions that  $\mathcal{F}(P(D)f) \in \mathscr{O}_{M}(\mathbb{R}^{n})$  since  $P(\xi)$  is polynomial and the derivatives of  $\hat{f}$  are polynomially growing. This is equivalent to that  $P(D)f \in \mathscr{O}'_{C}(\mathbb{R}^{n})$  and we can use lemma 4.1 to deduce that P(D)f = 0 as a tempered distribution. Taking the Fourier transform this is equivalent to that  $P(\xi)\hat{f}(\xi) = 0$  almost everywhere. As a polynomial  $P(\xi) \neq 0$  almost everywhere and we obtain that  $\hat{f} = 0$ .

The rest of the results are then direct consequences of theorem 1.1.

Proof of theorem 1.3. If  $f \in \mathcal{E}'(\mathbb{R}^n) \cap \mathcal{A}_V$  or  $f \in C_{\infty}(\mathbb{R}^n) \cap \mathcal{A}_V$ , then also  $f \in \mathscr{O}'_C(\mathbb{R}^n) \cap \mathcal{A}_V$ . Since  $N_0 = (-\Delta)^{-1/2}$  up to a constant factor and  $n \geq 2$  we have that  $-1/2 \in (-n/2, \infty) \setminus \mathbb{Z}$  and we can use theorem 1.1 to obtain that f = 0.

Proof of theorem 1.4. The assumption  $X_0 f = 0$  on all lines intersecting V implies that  $N_0 f|_V = 0$ . Since we also assume that  $f \in \mathcal{E}'(\mathbb{R}^n) \cap \mathcal{A}_V$  or  $f \in C_{\infty}(\mathbb{R}^n) \cap \mathcal{A}_V$  we obtain f = 0 by theorem 1.3.

Proof of theorem 1.7. The assumption  $X_1F = 0$  on all lines intersecting V implies that  $N_1F|_V = 0$ . By lemma 4.4 we have  $d(N_1F) = N_0(dF)$  componentwise up to a constant factor. The locality of the exterior derivative implies that  $(dF)_{ij} \in \mathcal{E}'(\mathbb{R}^n)$  and  $N_0(dF)_{ij}|_V = 0$ . Since  $(dF)_{ij} \in \mathcal{E}'(\mathbb{R}^n) \cap \mathcal{A}_V$  we can use theorem 1.3 for the components  $(dF)_{ij}$  to obtain that dF = 0. Lemma 4.2 implies that  $F = d\phi$  for some  $\phi \in \mathcal{E}'(\mathbb{R}^n)$ .

Acknowledgements. J.I. was supported by Academy of Finland (grants 332890 and 336254). K.M. was supported by Academy of Finland (Centre of Excellence in Inverse Modelling and Imaging, grant numbers 284715 and 309963).

#### References

- S. Bhattacharyya, T. Ghosh, and G. Uhlmann. Inverse Problem for Fractional-Laplacian with Lower Order Non-local Perturbations. *Trans. Amer. Math. Soc.*, 2020. To appear.
- [2] M. Cekić, Y.-H. Lin, and A. Rüland. The Calderón problem for the fractional Schrödinger equation with drift. *Calc. Var. Partial Differential Equations*, 59(3):Paper No. 91, 46, 2020.
- [3] S. N. Chandler-Wilde, D. P. Hewett, and A. Moiola. Sobolev spaces on non-Lipschitz subsets of ℝ<sup>n</sup> with application to boundary integral equations on fractal screens. *Integral Equations Operator Theory*, 87(2):179–224, 2017.
- [4] M. Courdurier, F. Noo, M. Defrise, and H. Kudo. Solving the interior problem of computed tomography using a priori knowledge. *Inverse Problems*, 24(6):065001, 2008.
- [5] G. Covi. An inverse problem for the fractional Schrödinger equation in a magnetic field. *Inverse Problems*, 36(4):045004, 24, 2020.
- [6] G. Covi, K. Mönkkönen, and J. Railo. Unique continuation property and Poincaré inequality for higher order fractional Laplacians with applications in inverse problems. *Inverse Probl. Imaging*, 2020. To appear.
- [7] G. Covi, K. Mönkkönen, J. Railo, and G. Uhlmann. The higher order fractional Calderón problem for linear local operators: uniqueness. 2020. arXiv:2008.10227.
- [8] G. de Rham. Differentiable Manifolds. Grundlehren der mathematischen Wissenschaften. Springer-Verlag Berlin Heidelberg, first edition, 1984.
- [9] A. Denisjuk. Inversion of the x-ray transform for 3D symmetric tensor fields with sources on a curve. *Inverse Problems*, 22(2):399–411, 2006.
- [10] D. S. Dummit and R. M. Foote. Abstract Algebra. Wiley, 2003.
- [11] M. M. Fall and V. Felli. Unique continuation property and local asymptotics of solutions to fractional elliptic equations. *Comm. Partial Differential Equations*, 39(2):354–397, 2014.
- [12] V. Felli and A. Ferrero. Unique continuation principles for a higher order fractional Laplace equation. *Nonlinearity*, 33(8):4133–4191, 2020.
- [13] M. A. García-Ferrero and A. Rüland. Strong unique continuation for the higher order fractional Laplacian. *Math. Eng.*, 1(4):715–774, 2019.
- [14] T. Ghosh, M. Salo, and G. Uhlmann. The Calderón problem for the fractional Schrödinger equation. Anal. PDE, 13(2):455–475, 2020.
- [15] G. Grubb. Distributions and Operators. Graduate Texts in Mathematics. Springer-Verlag New York, first edition, 2009.
- [16] S. Helgason. Integral Geometry and Radon transforms. Springer, New York, 2011.
- [17] L. Hörmander. The Analysis of Linear Partial Differential Operators I. Distribution Theory and Fourier Analysis. Classics in Mathematics. Springer-Verlag Berlin Heidelberg, second edition, 2003. Reprint of the 2nd edition 1990.
- [18] L. Hörmander. The Analysis of Linear Partial Differential Operators II. Differential Operators with Constant Coefficients. Classics in Mathematics. Springer-Verlag

Berlin Heidelberg, first edition, 2005. Reprint of the 1983 Edition (Grundlehren der mathematischen Wissenschaften Vol. 257).

- [19] J. Horváth. Topological Vector Spaces and Distributions. Vol. I. Addison-Wesley Publishing Co., Reading, Mass.-London-Don Mills, Ont., 1966.
- [20] J. Ilmavirta and K. Mönkkönen. Unique continuation of the normal operator of the x-ray transform and applications in geophysics. *Inverse Problems*, 36(4):045014, 2020.
- [21] J. Ilmavirta and K. Mönkkönen. X-ray tomography of one-forms with partial data. SIAM J. Math. Anal., 2020. To appear.
- [22] P. Juhlin. Principles of Doppler Tomography. Technical report, Center for Mathematical Sciences, Lund Institute of Technology, S-221 00 Lund, Sweden, 1992.
- [23] E. Katsevich, A. Katsevich, and G. Wang. Stability of the interior problem with polynomial attenuation in the region of interest. *Inverse Problems*, 28(6):065022, 2012.
- [24] E. Klann, E. T. Quinto, and R. Ramlau. Wavelet methods for a weighted sparsity penalty for region of interest tomography. *Inverse Problems*, 31(2):025001, 2015.
- [25] P. Kuchment, K. Lancaster, and L. Mogilevskaya. On local tomography. *Inverse Problems*, 11(3):571–589, 1995.
- [26] S. Mardare. On Poincaré and de Rham's theorems. Rev. Roumaine Math. Pures Appl., 53(5-6):523–541, 2008.
- [27] W. McLean. Strongly Elliptic Systems and Boundary Integral Equations. Cambridge University Press, Cambridge, 2000.
- [28] D. Mitrea. Distributions, Partial Differential Equations, and Harmonic Analysis. Universitext. Springer International Publishing, 2nd edition, 2018.
- [29] F. Natterer. The Mathematics of Computerized Tomography, volume 32 of Classics in Applied Mathematics. Society for Industrial and Applied Mathematics (SIAM), Philadelphia, PA, 2001. Reprint of the 1986 original.
- [30] S. J. Norton. Tomographic Reconstruction of 2-D Vector Fields: Application to Flow Imaging. *Geophys. J. Int.*, 97(1):161–168, 1989.
- [31] S. J. Norton. Unique Tomographic Reconstruction of Vector Fields Using Boundary Data. *IEEE Trans. Image Process.*, 1(3):406–412, 1992.
- [32] M. Okamoto. Distinctness of the Eigenvalues of a Quadratic form in a Multivariate Sample. Ann. Statist., 1(4):763–765, 1973.
- [33] G. P. Paternain, M. Salo, and G. Uhlmann. Tensor tomography: Progress and challenges. Chin. Ann. Math. Ser. B, 35(3):399–428, 2014.
- [34] E. Quinto. Singularities of the X-Ray Transform and Limited Data Tomography in R<sup>2</sup> and R<sup>3</sup>. SIAM J. Math. Anal., 24(5):1215–1225, 1993.
- [35] E. Quinto. Artifacts and Visible Singularities in Limited Data X-Ray Tomography. Sens. Imaging, 18, 2017.
- [36] K. Ramaseshan. Microlocal Analysis of the Doppler Transform on R<sup>3</sup>. J. Fourier Anal. Appl., 10(1):73–82, 2004.
- [37] A. Rüland. Unique continuation for fractional Schrödinger equations with rough potentials. Comm. Partial Differential Equations, 40(1):77–114, 2015.
- [38] T. Schuster. The 3D Doppler transform: elementary properties and computation of reconstruction kernels. *Inverse Problems*, 16(3):701–722, 2000.
- [39] T. Schuster. The importance of the Radon transform in vector field tomography. In R. Ramlau and O. Scherzer, editors, *The Radon Transform: The First 100 Years and Beyond*. de Gruyter, 2019.
- [40] V. Sharafutdinov. Slice-by-slice reconstruction algorithm for vector tomography with incomplete data. *Inverse Problems*, 23(6):2603–2627, 2007.
- [41] V. A. Sharafutdinov. Integral geometry of tensor fields. Inverse and Ill-posed Problems Series. VSP, Utrecht, 1994.
- [42] P. Stefanov and G. Uhlmann. Microlocal Analysis and Integral Geometry (working title). 2018. Draft version.
- [43] F. Trèves. Topological Vector Spaces, Distributions and Kernels. Academic Press, New York-London, 1967.
- [44] L. B. Vertgeim. Integral geometry problems for symmetric tensor fields with incomplete data. J. Inverse Ill-Posed Probl., 8(3):355–364, 2000.

- [45] J. Wengenroth. Topological properties of kernels of partial differential operators. *Rocky Mountain J. Math.*, 44(3):1037–1052, 2014.
- [46] J. Yang, H. Yu, M. Jiang, and G. Wang. High-order total variation minimization for interior tomography. *Inverse Problems*, 26(3):035013, 2010.
- [47] R. Yang. On higher order extensions for the fractional Laplacian. 2013. arXiv:1302.4413.

16